

ORDERED in the Southern District of Florida on May 6, 2015.

Laurel M. Isicoff, Judge United States Bankruptcy Court

UNITED STATES BANKRUPTCY COURT SOUTHERN DISTRICT OF FLORIDA MIAMI DIVISION

www.flsb.uscourts.gov

In re: : Chapter 11 Cases

ADINATH CORP. AND SIMPLY : Case No. 15-16885-LMI FASHION STORES, LTD., 1 : Jointly Administered

Debtor.

ORDER APPROVING SALE OF ALL OR SUBSTANTIALLY ALL OF DEBTOR'S ASSETS AND GRANTING RELATED RELIEF

This case came on to be heard on the 6th day of May, 2015 at 8:00 a.m., in Miami, Florida, upon the *Debtor's Emergency Motion Pursuant to Sections 105(A),363, and 365 Of The Bankruptcy Code and Bankruptcy Rules 6004 and 6006, and Local Rule 6004-1, For (I) Approval of Procedures in Connection With the Sale of All or Substantially All of the Debtor's Assets; (II) Authorization to Enter Into Stalking Horse Agreement in Connection Therewith; (III)*

¹ The Debtors in these Chapter 11 cases, along with the last four digits of each Debtor's federal tax identification are: (i) Adinath Corp., 2110 N.W.95th Avenue, Miami FL 33172 (4843); and (ii) Simply Fashion Stores, Ltd., 2500 Crestwood Boulevard, Birmingham, AL 35210 (6230)...

Approval of the Payment of Stalking Horse Protections; and (IV) the Setting of Related Auction and Hearing Dates [ECF 39] (the "Motion");² and it appearing that the relief requested is in the best interests of Simply Fashion Stores, Ltd. (the "Debtor"), its estate, its creditors, and other parties in interest; and it appearing that this Court has jurisdiction over this matter pursuant to 28 U.S.C. §§ 157 and 1334; and it appearing that consideration of the Motion and the relief requested therein is a core proceeding pursuant to 28 U.S.C. § 157; and adequate notice of the Motion having been given in the cicumstances and it appearing that no other notice need be given; and the Debtor and a joint venture consisting of Hilco Merchant Resources, LLC and Gordon Brothers Retail Partners, LLC (collectively, the "Agent") having agreed upon terms and conditions for the Agent to act as the Debtor's exclusive agent to conduct sales (the "Sale") of certain of the Debtor's assets, including, without limitation, the Merchandise and the owned FF&E (the "Assets"), pursuant to and in accordance with the terms and conditions set forth in that certain Agency Agreement, by and between the Agent and Debtor, as amended by the First Amendment, the final form of which is attached hereto as Composite Exhibit A (the "Agency Agreement"); and the transactions represented by the Agency Agreement having been determined to be the highest and best offer for the Assets; and a hearing having been held on April 20, 2015, on the Motion whereupon the Court entered an Order approving, inter alia, the Break-Up Fee, and bidding procedures [ECF No. 83] (the "Bidding Procedures Order"); and a sale hearing having been held on May 6, 2015 (the "Sale Hearing") to consider the remaining relief requested in the Motion and approval of the Agency Agreement; and appearances of all interested parties having been noted on the record of the Sale Hearing; and upon the Declaration in Support of First Day Pleadings (the "First Day Declaration"); and upon all of the proceedings

² All capitalized terms not otherwise defined in this Order have the meaning ascribed to them in the Motion or in the Agency Agreement.

had before the Court (including but not limited to the testimony and other evidence proffered or adduced at the Sale Hearing); and the Court having found and determined that the relief sought in the Motion is in the best interests of the Debtor, its estate, its creditors, and all parties in interest and that the legal and factual bases set forth in the Motion establish just cause for the relief granted herein; and after due deliberation and sufficient cause appearing therefor, it is hereby

FOUND AND DETERMINED THAT:³

- A. **Jurisdiction:** This Court has jurisdiction to consider the Motion and the relief requested therein pursuant to 28 U.S.C. §§ 157 and 1134. Approval of the Debtor's entry into the Agency Agreement, and the transactions contemplated thereby is a core proceeding under 28 U.S.C. §§ 157(b)(2)(A), (D), (N) and (O).
- B. Venue: Venue of these cases in this district is proper pursuant to 28 U.S.C. § 1409(a).
- C. **Statutory Predicates:** The statutory predicates for the approval of the Agency Agreement and transactions contemplated therein are sections 105, 363, 364 and 554 of the Bankruptcy Code and Rules 2002, 4001, 6004 and 9014 of the Federal Rules of Bankruptcy Procedure (the "Bankruptcy Rules").
- D. **Notice:** Proper, timely, adequate and sufficient notice of the Motion and the Sale Hearing under the circumstances has been provided in accordance with the Bidding Procedures Order, sections 102(1), 105(a), and 363 of the Bankruptcy Code, Bankruptcy Rules 2002, 4001

3

The findings of fact and the conclusions of law stated herein shall constitute the Court's findings of fact and conclusions of law pursuant to Bankruptcy Rule 7052, made applicable to this proceeding pursuant to Bankruptcy Rule 9014. To the extent any finding of fact shall be determined to be a conclusion of law, it shall be so deemed, and to the extent any conclusion of law shall be determined to be a finding of fact, it shall be so deemed.

and 6004, and in compliance with the Bidding Procedures Order. No other or further notice is required.

- E. Opportunity to be Heard: A reasonable opportunity to object or be heard regarding the relief requested in the Motion and the transactions pursuant thereto has been afforded to all interested persons and entities, including, without limitation, the following: (i) the Office of the United States Trustee, (ii) counsel to JNS INVT, LLC (in its capacity as both the prepetition senior secured lender and the debtor-in-possession lender, collectively, the "Lender"), (iii) the Office of the United States Attorney General, (iv) counsel to the Creditors' Committee, (v) all parties who are known to assert any lien, claim, interest or encumbrance in or upon any of the Assets, (vi) all lessors of leases for the Stores, (vii) all applicable federal, state, and local taxing authorities (collectively, the "Taxing Authorities"), (viii) all applicable state attorneys general, and (ix) all other applicable parties in interest, including all entities on the general case service list as of the date of entry of the Bidding Procedures Order ((i) through (ix) collectively, the "Notice Parties"). Objections, if any, to the Motion have been withdrawn or resolved and, to the extent not withdrawn or resolved, are hereby OVERRULED.
- F. Marketing Process: As demonstrated by: (i) the First Day Declaration, (ii) the testimony and other evidence proffered or adduced at the Sale Hearing, and (iii) the representations of counsel made on the record at the Sale Hearing, the Debtor has thoroughly marketed the Assets and has conducted the bidding solicitation fairly, with adequate opportunity for parties that either expressed an interest in acquiring or liquidating the Assets, or who the Debtor believed may have an interest in acquiring or liquidating the Assets, to submit competing bids. The Debtor and the Agent have respectively negotiated and undertaken their roles leading

to the Sale and entry into the Agency Agreement in a diligent, noncollusive, fair and good faith manner.

- G. **Highest and Best Offer:** The Agency Agreement attached hereto as Composite Exhibit A, including the form and total consideration to be realized by the Debtor pursuant to the Agency Agreement, (i) is the highest and best offer received by the Debtor for the Assets, (ii) is fair and reasonable, and (iii) is in the best interests of the Debtor, its estate, its creditors and all other parties in interest. There is no legal or equitable reason to delay entry into the Agency Agreement, and the transactions contemplated therein, including, without limitation, the Sale.
- H. **Business Judgment:** The Debtor's decision to (i) enter into the Agency Agreement, and (ii) perform under and make payments required by the Agency Agreement, as supported by Soneet Kapila, in his capacity as the Debtor's Chief Restructuring Officer (the "CRO"), is a reasonable exercise of the Debtor's and the CRO's sound business judgment consistent with their fiduciary duties and is in the best interests of the Debtor, its estate, its creditors, and all other parties in interest.
- I. **Personally Identifiable Information:** The transactions contemplated by the Agency Agreement do not include the sale or lease of personally identifiable information, as defined in section 101(41A) of the Bankruptcy Code ("<u>Personally Identifiable Information</u>") (or Assets containing personally identifiable information).
- J. **Time of the Essence:** Time is of the essence in effectuating the Agency Agreement and proceeding with the Sale contemplated therein without interruption. Based on the record of the Sale Hearing, and for the reasons stated on the record at the Sale Hearing, the Sale under the Agency Agreement must be commenced on the first day following entry of this Order to maximize the value that the Agent may realize from the Sale, and the value that the

Debtor may realize from entering into the Agency Agreement. Accordingly, cause exists to lift the stay to the extent necessary, as contemplated by Bankruptcy Rules 4001(a) and 6004(h) and permit the immediate effectiveness of this Order.

K. **Sale Free and Clear:** A sale of the Assets other than one free and clear of liens, claims, encumbrances, defenses (including, without limitation, rights of setoff and recoupment) and interests, including, without limitation, security interests of whatever kind or nature, mortgages, conditional sales or title retention agreements, pledges, deeds of trust, hypothecations, liens, encumbrances, assignments, preferences, debts, easements, charges, suits, licenses, options, rights-of-recovery, judgments, orders and decrees of any court or foreign or domestic governmental entity, taxes (including foreign, state and local taxes), licenses, covenants, restrictions, indentures, instruments, leases, options, off-sets, claims for reimbursement, contribution, indemnity or exoneration, successor, product, environmental, tax, labor, ERISA, CERCLA, alter ego and other liabilities, causes of action, contract rights and claims, to the fullest extent of the law, in each case, of any kind or nature (including, without limitation, all "claims" as defined in section 101(5) of the Bankruptcy Code), known or unknown, whether pre-petition or post-petition, secured or unsecured, choate or inchoate, filed or unfiled, scheduled or unscheduled, perfected or unperfected, liquidated or unliquidated, noticed or unnoticed, recorded or unrecorded, contingent or non-contingent, material or non-material, statutory or non-statutory, matured or unmatured, legal or equitable (collectively, "Encumbrances") and without the protections of this Order would hinder the Debtor's ability to obtain the consideration provided for in the Agency Agreement and, thus, would impact materially and adversely the value that the Debtor's estate would be able to obtain for the sale of such Assets in the liquidation of its stores in conjunction with the Agent. But for the protections

afforded to the Agent under the Bankruptcy Code, this Order and the Agency Agreement, the Agent would not have offered to pay the consideration contemplated in the Agency Agreement. In addition, each entity with an Encumbrance upon the Assets, (i) has consented to the Sale, including as set forth below, or is deemed to have consented to the Sale, (ii) could be compelled in a legal or equitable proceeding to accept money satisfaction of such interest, or (iii) otherwise falls within the provisions of section 363(f) of the Bankruptcy Code, and therefore, in each case, one or more of the standards set forth in section 363(f)(1) - (5) of the Bankruptcy Code has been satisfied. Those holders of Encumbrances who did not object or who withdrew their objections to this Motion, or whose objections were resolved through the terms of this Order, are deemed to have consented pursuant to section 363(f)(2) of the Bankruptcy Code. Therefore, approval of the Agency Agreement and the consummation of the Sale free and clear of Encumbrances is appropriate pursuant to section 363(f) of the Bankruptcy Code and is in the best interests of the Debtor's estate, its creditors and other parties in interest; provided, however, that the Encumbrances shall attach to all amounts due to the Debtors under the Agency Agreement (the "Consideration"), including, without limitation, the Guaranteed Amount, the Additional Agent Goods Fee, the Sharing Amount (if any), the proceeds realized upon the sale of Owned FF&E (less the FF&E Commission), the Additional DC/CO Guaranteed Amount and the Additional Guaranteed Amount, in each case in accordance with the order of priority of such Encumbrances under applicable law as exists on the date hereof, including in respect of the liens and security interests granted to the Lender, in its capacity as a debtor-in-possession lender under this Court's Interim Order, dated April 17, 2015 [ECF 42] and, as approved as a final order at the Sale Hearing (the "DIP Order"). Notwithstanding the foregoing the consent of the DIP Lender to the sale is conditioned upon the payment of the DIP Loan pursuant to the DIP Order.

- L. Arm's-length Sale: The consideration to be paid by the Agent under the Agency Agreement was negotiated at arm's-length and constitutes reasonably equivalent value and fair and adequate consideration for the Assets in the liquidation of its Stores in conjunction with the Agent under the Bankruptcy Code, the Uniform Fraudulent Transfer Act, the Uniform Fraudulent Conveyance Act and the laws of the United States, any state, territory, possession thereof or the District of Columbia. The terms and conditions set forth in the Agency Agreement are fair and reasonable under these circumstances and were not entered into for the purpose of, nor do they have the effect of, hindering, delaying or defrauding the Debtor or its creditors under any applicable laws.
- M. Good Faith: The Debtor, its management, including, without limitation, its CRO, its board of directors, and the Agent, its members and its officers, directors, employees, agents and representatives, actively participated in the bidding process and acted in good faith. The Agency Agreement between the Agent and the Debtor was negotiated and entered into based upon arm's length bargaining, without collusion or fraud, and in good faith as that term is used in sections 363(m) and 364(e) of the Bankruptcy Code. The Agent shall be protected by sections on appeal. The Debtor was free to deal with any other party interested in buying or selling on behalf of the Debtor's estate some or all of the Assets. Neither the Debtor nor the Agent has engaged in any conduct that would cause or permit the Sale, the Agency Agreement, or any related action or the transactions contemplated thereby to be avoided under section 363(n) of the Bankruptcy Code, or that would prevent the application of sections 363(m) or 364(e) of the Bankruptcy Code. The Agent has not violated section 363(n) of the Bankruptcy Code by any

action or inaction. Specifically, the Agent has not acted in a collusive manner with any person and was not controlled by any agreement among bidders. The Agent's prospective performance and payment of amounts owing under the Agency Agreement are in good faith and for valid business purposes and uses.

- N. **Insider Status:** The Agent is not an "insider" as that term is defined in section 101(31) of the Bankruptcy Code. No common identity of directors or controlling stockholders exists between the Agent and the Debtor.
- O. Security Interests: The liens provided for in the Agency Agreement and this Order to secure the obligations of the Debtor under the Agency Agreement to the Agent are necessary to induce the Agent to agree to terms for the Agency Agreement that maximize value for the Debtor's estates. The absence of such protections would impact materially and adversely the value available to the Debtor in the liquidation of its stores in conjunction with a liquidation agent. But for the protections afforded to the Agent under the Bankruptcy Code, this Order and the Agency Agreement, the Agent would not have agreed to pay the Debtor the compensation provided for under the Agency Agreement. In addition, the Lender, which claims a first lien and security interest in substantially all of the Debtor's property as more particularly outlined in the DIP Order (including by virture of the liens granted to the Lender under the DIP Order for its post-petition financing) to which the Agent's security interests attach pursuant to the terms of the Agency Agreement and this Order, has consented to the security interests provided to the Agent in the Agency Agreement, in accordance with and subject to the satisfaction of the conditions set forth in the Agency Agreement and in Paragraph 31 of this Order.
- P. Corporate Authority: The Debtor (i) has full corporate or other power to execute, deliver and perform their obligations under the Agency Agreement and all other

transactions contemplated thereby (including without limitation, reaching an agreement and resolution regarding the final reconciliation contemplated by the Agency Agreement), and entry into the Agency Agreement has been duly and validly authorized by all necessary corporate or similar action, (ii) has all of the corporate or other power and authority necessary to consummate the transactions contemplated by the Agency Agreement, and (iii) has taken all actions necessary to authorize and approve the Agency Agreement and the transactions contemplated thereby. No consents or approvals, other than those expressly provided for herein or in the Agency Agreement, are required for the Debtor to consummate such transactions.

- Q. **No Successor Liability:** No sale, transfer or other disposition of the Assets pursuant to the Agency Agreement or entry into the Agency Agreement will subject the Agent to any liability for claims, obligations or Encumbrances asserted against the Debtor or the Debtor's interest in such Assets by reason of such transfer under any laws, including, without limitation, any bulk-transfer laws or any theory of successor or transferee liability, antitrust, environmental, product line, de facto merger or substantial continuity or similar theories. The Agent is not a successor to the Debtor or its estates.
- R. **No Sub Rosa Plan:** Entry into the Agency Agreement and the transactions contemplated thereby neither impermissibly restructure the rights of the Debtor's creditors, nor impermissibly dictates the terms of a liquidating plan of reorganization for the Debtor. Entry into the Agency Agreement does not constitute a sub rosa chapter 11 plan.
- S. **Bankruptcy Rule 6003:** The Court finds that the relief requested in this Order is necessary to avoid immediate and irreparable harm to the Debtors' estates if not entered today, and therefore the Court has found sufficient cause exists to enter this Order prior to the expiration of 21 days from the Petition Date.

NOW, THEREFORE, IT IS HEREBY ORDERED, ADJUDGED AND DECREED THAT:

A. Motion Granted, Objections Overruled

- 1. The relief requested in the Motion is granted as set forth herein.
- 2. Any remaining objections to the Motion or the relief requested therein that have not been withdrawn, waived, or settled, and all reservations of rights included in such objections are overruled in all respects and denied.

B. Agency Agreement Approved and Authorized

3. The Agency Agreement is approved pursuant to section 363 of the Bankruptcy Code. The Debtor is hereby authorized and empowered to enter into and perform under the Agency Agreement, and the Agency Agreement (and each of the transactions contemplated therein (including without limitation, reaching an agreement and resolution regarding the final reconciliation contemplated by the Agency Agreement, which agreement and resolution shall be binding on all parties (including, without limitation, the Debtor, the Committee, the Lender, any successor chapter 7 or chapter 11 trustee, and all other parties in interest) without further order of the Court)) is hereby approved in its entirety and is incorporated herein by reference. The failure to include specifically any particular provision of the Agency Agreement in this Order shall not diminish or impair the effectiveness of such provisions, it being the intent of the Court that the Agency Agreement and all of its provisions, payments and transactions, be authorized and approved in their entirety. Likewise, all of the provisions of this Order are nonseverable and mutually dependent.

- 4. All amounts payable to the Agent under the Agency Agreement shall be payable to the Agent without the need for any application of the Agent therefor or any further order of the Court.
- 5. Subject to the provisions of this Order, the Debtor and the Agent are hereby authorized, pursuant to sections 105(a) and 363(b)(1) of the Bankruptcy Code, to conduct the Sale in accordance with the Agency Agreement and the sale guidelines (the "GOB Sale Guidelines") attached hereto as Exhibit B, which GOB Sale Guidelines are hereby approved in their entirety.
- 6. Pursuant to section 363(b) of the Bankruptcy Code, the Debtor, the Agent and each of their respective officers, employees and agents are hereby authorized and directed to execute such documents and to do such acts as are necessary or desirable to carry out the Sale and effectuate the Agency Agreement and each of the transactions and related actions contemplated or set forth therein. The CRO, is specifically authorized to act on behalf of the Debtor in connection with the Sale and no other consents or approvals are necessary or required for the Debtor to carry out the Sale, effectuate the Agency Agreement and each of the transactions and related actions contemplated or set forth therein.

C. Order Binding

7. This Order shall be binding upon and shall govern the acts of all entities, including, without limitation, all filing agents, filing officers, title agents, title companies, recorders of mortgages, recorders of deeds, registrars of deeds, administrative agencies, governmental departments, secretaries of state, federal, state and local officials, and all other persons and entities who may be required by operation of law, the duties of their office, or

contract, to accept, file, register or otherwise record or release any documents or instruments, or who may be required to report or insure any title or state of title in or to the Assets.

8. This Order and the terms and provisions of the Agency Agreement shall be binding on all of the Debtor's creditors (whether known or unknown), the Debtor, the Agent, and their respective affiliates, successors and assigns, and any affected third parties including, but not limited to, all persons asserting an interest in the Assets, notwithstanding any subsequent appointment of any trustee, party, entity or other fiduciary under any section of the Bankruptcy Code with respect to the forgoing parties, and as to such trustee, party, entity or other fiduciary, such terms and provisions likewise shall be binding. The provisions of this Order and the terms and provisions of the Agency Agreement, and any actions taken pursuant hereto or thereto shall survive the entry of any order which may be entered confirming or consummating any plan(s) of the Debtor or converting the Debtor's case from chapter 11 to chapter 7, and the terms and provisions of the Agency Agreement, as well as the rights and interests granted pursuant to this Order and the Agency Agreement, shall continue in these or any superseding cases and shall be binding upon the Debtor, the Agent and their respective successors and permitted assigns, including any trustee or other fiduciary hereafter appointed as a legal representative of the Debtor under chapter 7 or chapter 11 of the Bankruptcy Code. Any trustee appointed in this case shall be and hereby is authorized to operate the business of the Debtor to the fullest extent necessary to permit compliance with the terms of this Order and the Agency Agreement, and Agent and the trustee shall be and hereby are authorized to perform under the Agency Agreement upon the appointment of the trustee without the need for further order of this Court.

D. Good Faith.

9. Entry into the Agency Agreement is undertaken by the parties thereto in good faith, as that term is used in sections 363(m) and 364(e) of the Bankruptcy Code, and Agent shall be protected by sections 363(m) and 364(e) of the Bankruptcy Code in the event that this Order is reversed or modified on appeal. The reversal or modification on appeal of the authorization provided herein to enter into the Agency Agreement and consummate the transactions contemplated thereby shall not affect the validity of such transactions, unless such authorization is duly stayed pending such appeal. The Agent is entitled to all of the benefits and protections afforded by sections 363(m) and 364(e) of the Bankruptcy Code. The transactions contemplated by the Agency Agreement are not subject to avoidance pursuant to section 363(n) of the Bankruptcy Code.

E. Conduct of the Sale

10. Except as otherwise provided in the Agency Agreement, pursuant to section 363(f) of the Bankruptcy Code, the Agent shall be authorized to sell all Merchandise, the Owned FF&E and other Assets to be sold pursuant to the Agency Agreement free and clear of any and all Encumbrances, including, without limitation, the liens and security interests, as the same may have been amended from time to time, of the Lender, whether arising by agreement, any statute or otherwise and whether arising before, on or after the date on which these chapter 11 cases were commenced, with any presently existing liens encumbering all or any portion of the Assets or the Proceeds thereof (including, but not limited to, the first-priority security interest of the Lender), with all such Encumbrances attaching to the Consideration, including, without limitation, the Guaranteed Amount, the Additional Agent Goods Fee, the Sharing Amount (if any), the proceeds realized upon the sale of Owned FF&E (less the FF&E Commission), the Additional DC/CO Guaranteed Amount and the Additional Guaranteed Amount, in each case in

accordance with the order of priority of such Encumbrances under applicable law as exists on the Petition Date, including in respect of the liens and security interests granted to the Lender, in its capacity as a debtor-in-possession lender under this Court's DIP Order. For the sake of clarity, however, nothing in this paragraph is intended to diminish the liens in favor of the Agent, as reflected in the Agency Agreement and this Order, that attach to, among other things, the Proceeds of the Sale.

- 11. If any person or entity that has filed financing statements, mortgages, construction or mechanic's liens, lis pendens or other documents or agreement evidencing liens on or interests in the Assets shall not have delivered to the Debtor, in proper form for filing and executed by the appropriate parties, termination statements, instruments of satisfaction, or releases of any Encumbrances which the person or entity has with respect to the Assets, each such person or entity is hereby directed to deliver all such statements, instruments and releases and the Debtor and the Agent are hereby authorized to execute and file such statements, instruments, releases and other documents on behalf of the person or entity asserting the same and the Agent is authorized to file a copy of this Order which, upon filing, shall be conclusive evidence of the release and termination of such interest. Each and every federal, state and local governmental unit is hereby directed to accept any and all documents and instruments necessary or appropriate to give effect to the Sale and related transactions. For the sake of clarity, the UCC financing statements recorded by Iberiabank are not affected by the provisions of this paragraph, as the assets that are secured by Iberiabank's UCC-1 financing statement are not being sold pursuant to the terms of this Order.
- 12. All entities that are presently in possession of some or all of the Assets or other property in which the Debtor holds an interest that are or may be subject to the Agency

Agreement hereby are directed to surrender possession of such Assets or other property to the Agent.

- 13. The Debtor and the Agent shall not extend the Sale Termination Date beyond June 30, 2015, unless extended by mutual written agreement of the Debtor and the Agent following a commensurate extension of the expiration date of the Letter of Credit.
- 14. Unless otherwise ordered by the Court, all newspapers and other advertising media in which the Sale may be advertised and all landlords are directed to accept this Order as binding authority so as to authorize the Debtor and the Agent to consummate the Agency Agreement and to consummate the transactions contemplated therein, including, without limitation, to conduct and advertise the Sale in the manner contemplated by the Agency Agreement, including, without limitation, conducting and advertising of the Sale (at the contractual rates charged to the Debtor prior to the Petition Date) in accordance with the Agency Agreement, the GOB Sale Guidelines, and this Order.
- 15. Nothing in this Order or the Agency Agreement releases, nullifies, or enjoins the enforcement of any liability to a governmental unit under environmental laws or regulations (or any associated liabilities for penalties, damages, cost recovery, or injunctive relief) that any entity would be subject to as the owner, lessor, lessee, or operator of the property after the date of entry of this Order. Nothing contained in this Order or in the Agency Agreement shall in any way (i) diminish the obligation of any entity to comply with environmental laws, or (ii) diminish the obligations of the Debtor to comply with environmental laws consistent with its rights and obligations as a debtor in possession under the Bankruptcy Code. Nothing herein shall be construed to be a debtor determination that the Agent is an operator with respect to any environmental law or regulation. Moreover, the Sale shall not be exempt from, and the Agent

shall be required to comply with, laws of general applicability, including, without limitation, public health and safety, criminal, tax, labor, employment, environmental, antitrust, fair competition, traffic and consumer protection laws, including consumer laws regulating deceptive practices and false advertising (collectively, "General Laws"). Nothing in this Order shall alter or affect the Debtor's and Agent's obligations to comply with all applicable federal safety laws and regulations. Nothing in this Order shall be deemed to bar any Governmental Unit (as defined in Bankruptcy Code section 101(27)) from enforcing General Laws in the applicable non-bankruptcy forum, subject to the Debtor's or the Agent's right to assert in that forum or before this Court that any such laws are not in fact General Laws or that such enforcement is impermissible under the Bankruptcy Code, this Order, or otherwise, pursuant to Paragraph 18 hereunder. Notwithstanding any other provision in this Order, no party waives any rights to argue any position with respect to whether the conduct was in compliance with this Order and/or any applicable law, or that enforcement of such applicable law is preempted by the Bankruptcy Code. Nothing in this Order shall be deemed to have made any rulings on any such issues.

16. <u>Disputes Between Government Units and the Debtor or the Agent.</u> To the extent that the Sale is subject to any federal, state or local statute, ordinance, or rule, or licensing requirement solely directed at regulating "going out of business," "store closing," similar inventory liquidation sales, or bulk sale laws (each a "GOB Law," and together, the "GOB Laws"), including laws restricting safe, professional and non-deceptive, customary advertising such as signs, banners, posting of signage, and use of sign-walkers solely in connection with the Sale and including ordinances establishing license or permit requirements, waiting periods, time limits or bulk sale restrictions that would otherwise apply solely to Sale (collectively, the "Liquidation Laws"), the following provisions shall apply:

- a. Provided that the Sale is conducted in accordance with the terms of this Order, the Agency Agreement and the GOB Sale Guidelines, and in light of the provisions in the laws of many Governmental Units that exempt court-ordered sales from their provisions, the Debtor shall be presumed to be in compliance with any GOB Laws and Liquidation Laws and, subject to Paragraphs 16 and 17 herein, are authorized to conduct the Sale in accordance with the terms of this Order and the GOB Sale Guidelines without the necessity of further showing compliance with any such GOB Laws and Liquidation Laws.
- b. Within five (5) business days of entry of this Order, the Debtor shall serve copies of this Order, the Agency Agreement and the Sale Guidelines via e-mail, facsimile or regular mail, on the parties set forth on Exhibit C.
- c. To the extent there is a dispute arising from or relating to the Sale, this Order, the Agency Agreement, or the GOB Sale Guidelines, which dispute relates to any GOB Laws or Liquidation Laws (a "Reserved Dispute"), the Court shall retain exclusive jurisdiction to resolve the Reserved Dispute. Any time within fifteen (15) days following service of this Order, any Governmental Unit may assert that a Reserved Dispute exists by serving written notice of such Reserved Dispute to counsel for the Debtor and counsel for the Agent at the addresses set forth in the Agency Agreement so as to ensure delivery thereof within one (1) business day thereafter. If the Debtor, the Agent and the Governmental Unit are unable to resolve the Reserved Dispute within fifteen (15) days of service of the notice, the aggrieved party may file a motion with this Court requesting that this Court resolve the Reserved Dispute (a "Dispute Resolution Motion").
- d. In the event a Dispute Resolution Motion is filed, nothing in this Order shall preclude the Debtor, a landlord, the Agent or other interested party from asserting (i) that the provisions of any GOB Laws and/or Liquidation Laws are preempted by the Bankruptcy Code or

- (ii) that neither the terms of this Order, nor the Debtor or the Agent's conduct pursuant to this Order, violates such GOB Laws and/or Liquidation Laws. Filing a Dispute Resolution Motion as set forth herein shall not be deemed to affect the finality of this Order or to limit or interfere with the Debtor's or the Agent's ability to conduct or to continue to conduct the Sale pursuant to this Order and the Agency Agreement, absent further order of this Court. The Court grants authority for the Debtor and the Agent to conduct the Sale pursuant to the terms of this Order, the Agency Agreement, and/or the GOB Sale Guidelines attached hereto and to take all actions reasonably related thereto or arising in connection therewith. The Governmental Unit shall be entitled to assert any jurisdictional, procedural, or substantive arguments it wishes with respect to the requirements of its Liquidation Laws or the lack of any preemption of such GOB Laws and/or Liquidation Laws by the Bankruptcy Code. Nothing in this Order shall constitute a ruling with respect to any issues to be raised in any Dispute Resolution Motion.
- e. If, at any time, a dispute arises between the Debtor and/or the Agent and a Governmental Unit as to whether a particular law is a GOB Law and/or Liquidation Law, and subject to any provisions contained in this Order related to GOB Laws and/or Liquidation Laws, then any party to that dispute may utilize the provisions of Subparagraphs (b) and (c) hereunder by serving a notice to the other party and proceeding thereunder in accordance with those Paragraphs. Any determination with respect to whether a particular law is a GOB Law and/or Liquidation Law shall be made de novo.
- f. Notwithstanding anything herein to the contrary, and in view of the importance of the use of sign-walkers, banners, and other advertising to the Sale, to the extent that disputes arise during the course of the Sale regarding laws regulating the use of sign-walkers and banner advertising and the Debtor and the Agent are unable to resolve the matter consensually with the

Governmental Unit, any party may request an immediate telephonic hearing with this Court pursuant to these provisions. Such hearing will, to the extent practicable, be scheduled initially within two (2) business days of such request. This scheduling shall not be deemed to preclude additional hearings for the presentation of evidence or arguments as necessary.

- 17. Except to the extent of the reserved rights of Governmental Units expressly granted elsewhere in this Order, the Debtor and Agent are hereby authorized to take such actions as may be necessary and appropriate to implement the Agency Agreement and to conduct the Sale without necessity of further order of this Court as provided in the Agency Agreement, the GOB Sale Guidelines or any Side Letter (as defined below), including, but not limited to, advertising the Sale as "going out of business," "total liquidation," "store-closing" or similar-themed sales through the posting of signs (including the use of exterior banners at non-enclosed mall Stores, and at enclosed mall Stores to the extent the applicable Store entrance does not require entry into the enclosed mall common area), use of signwalkers and street signage.
- 18. Except as expressly provided in the Agency Agreement, the Sale shall be conducted by the Debtor and the Agent notwithstanding any restrictive provision of any lease, sublease or other agreement relative to occupancy affecting or purporting to restrict the conduct of the Sale, the rejection of leases, abandonment of assets or "going dark" provisions. The Agent and landlords of the Stores are authorized to enter into agreements ("Side Letters") between themselves modifying the GOB Sale Guidelines without further order of the Court, and such Side Letters shall be binding as among the Agent and any such landlords, provided that nothing in such Side Letters affects the provisions of Paragraphs 16 and 17. In the event of any conflict between the GOB Sale Guidelines and any Side Letter, the terms of such Side Letter shall control.

- 19. Except as expressly provided for herein or in the GOB Sale Guidelines and Side Letters, and except with respect to any Governmental Unit (as to which Paragraphs 16 and 17 shall apply), no person or entity, including but not limited to any landlord, licensor, or creditor, shall take any action to directly or indirectly prevent, interfere with, or otherwise hinder consummation of the Sale, or the advertising and promotion (including the posting of signs and exterior banners or the use of signwalkers) of such Sale, and all such parties and persons of every nature and description, including landlords, licensors, creditors and utility companies and all those acting for or on behalf of such parties, are prohibited and enjoined from (i) interfering in any way with, or otherwise impeding, the conduct of the Sale and/or (ii) instituting any action or proceeding in any court or administrative body seeking an order or judgment against, among others, the Debtor, the Agent, or the landlords at the Debtor's closing stores that might in any way directly or indirectly obstruct or otherwise interfere with or adversely affect the conduct of the Sale or other liquidation sales at the stores and/or seek to recover damages for breach(es) of covenants or provisions in any lease, sublease or license based upon any relief authorized herein.
- 20. The Agent shall have the right to use the Debtor's closing stores and all related store services, furniture, fixtures, equipment and other assets of the Debtor for the purpose of conducting the Sale, free of any interference from any entity or person, subject to compliance with the GOB Sale Guidelines, any Side Letters and this Order and subject to Paragraphs 16 and 17 of this Order.
- 21. Nothing in this Order shall (a) alter or affect the Debtor's obligations to comply with section 365(d)(3) of the Bankruptcy Code or (b) alter or modify the rights of any lessor or other counterparty to a lease with the Debtor to file an appropriate motion or otherwise seek appropriate relief if the Debtor fails to comply with section 365(d)(3) of the Bankruptcy Code

other than as set forth in the *Debtor's Motion for an Order Authorizing and Approving Procedures for Rejection of Certain Unexpired Leases and Executory Contracts* [ECF No. 92]; provided that the conduct of the Sale in accordance with the Sale Guidelines shall not be a violation of section 365(d)(3) of the Bankruptcy Code.

- 22. Pursuant to section 554(a) of the Bankruptcy Code, the Debtor and the Agent, as applicable, are permitted to abandon property of the Debtor's estate in accordance with the terms and provisions of the Agency Agreement, without incurring liability to any person or entity; provided, however, that, unless the Agent otherwise consents, the Debtor may only abandon property located in any store (and, if applicable, the distribution center) on or after the applicable Sale Termination Date. In the event of any such abandonment, all applicable landlords shall be authorized to dispose of such property without any liability to any individual or entity that may claim an interest in such abandoned property, and such abandonment shall be without prejudice to any landlord's right to assert any claim based on such abandonment and without prejudice to the Debtor or other party in interest to object thereto.
- 23. Before any sale, abandonment or other disposition of the Debtor's computers (including software) and/or cash registers and any other point of sale FF&E located at the Stores (collectively, "POS Equipment") that may contain customer lists, identifiable personal and/or confidential information about the Debtor's employees and/or customers, or credit card numbers ("Confidential Information") takes effect, the Debtor shall remove or cause to be removed the Confidential Information from the POS Equipment.
- 24. The Agent shall accept the Debtor's validly-issued gift certificates and gift cards that were issued by the Debtor prior to the Sale Commencement Date, and the Debtor shall reimburse Agent for such amounts during the weekly sale reconciliation provided for and subject

to the limitations set forth in Section 8.7 of the Agency Agreement. The Agent shall accept returns of Merchandise sold by the Debtor prior to the Sale Commencement Date, provided that such return is otherwise in compliance with the Debtor's return policies in effect as of the date such item was purchased and the customer is not repurchasing the same item so as to take advantage of the sale price being offered by the Agent, and the Debtor shall reimburse Agent for such amounts during the weekly sale reconciliation provided for and subject to the limitations set forth in Section 8.7 of the Agency Agreement.

- 25. All state and federal laws relating to implied warranties for latent defects shall be complied with and are not superseded by the sale of said goods or the use of the terms "as is" or "final sales." The Debtor and/or the Agent shall accept return of any goods purchased during the Sale that contain a defect which the lay consumer could not reasonably determine was defective by visual inspection prior to purchase for a full refund, provided that the consumer must return the merchandise within twenty-one (21) days of their purchase, the consumer must provide a receipt, and the asserted defect must in fact be a "latent" defect. The Debtor shall promptly reimburse Agent in cash for any refunds Agent is required to issue to customers in respect of any goods purchased during the Sale that contain such a latent defect.
- During the Sale Term, the Agent shall be granted a limited license and right to use the trade names, logos and customer, mailing and e-mail lists, websites and social media relating to and used in connection with the operation of the stores as identified in the Agency Agreement, solely for the purpose of advertising the Sale in accordance with the terms of the Agency Agreement; provided, however, that the Agent shall not receive Personally Identifiable Information from the Debtor. In addition, the Agent is prohibited from using the websites, social

media sites, e-mail lists, customer lists and e-commerce sites owned by or related to New Dots, LLC or the "Dots" brand.

- 27. Except as expressly provided for in the Agency Agreement, nothing in this Order or the Agency Agreement, and none of the Agent's actions taken in respect of the Sale shall be deemed to constitute an assumption by Agent of any of the Debtor's obligations relating to any of the Debtor's employees. Moreover, the Agent shall not become liable under any collective bargaining or employment agreement or be deemed a joint or successor employer with respect to such employees.
- 28. The Agent shall not be liable for sales taxes except as expressly provided in the Agency Agreement and the payment of any and all sales taxes is the responsibility of the Debtor. The Debtor are directed to remit all taxes arising from the Sale to the applicable Taxing Authorities as and when due, provided that in the case of a bona fide dispute the Debtor are only directed to pay such taxes upon the resolution of the dispute, if and to the extent that the dispute is decided in favor of the Taxing Authority. For the avoidance of doubt, Sales Taxes collected and held in trust by the Debtor shall not be used to pay any creditor or any other party, other than the Taxing Authority for which the Sales Taxes are collected. The Agent shall collect, remit to the Debtor and account for Sales Taxes as and to the extent provided in the Agency Agreement. This Order does not enjoin, suspend or restrain the assessment, levy or collection of any tax under State law, and does not constitute a declaratory judgment with respect to any party's liability for taxes under State law.
- 29. Subject to the terms set forth in the Agency Agreement, the Debtor and/or the Agent (as the case may be) are authorized and empowered to transfer Assets among the closing stores and the Distribution Center. The Agent is authorized to sell the Debtor's furniture,

fixtures and equipment and abandon the same, in each case, as provided for and in accordance with the terms of the Agency Agreement.

F. Liens Granted To Agent

- 30. Pursuant to Bankruptcy Code section 364(d), in accordance with and subject to the terms of the Agency Agreement, the Debtor has granted and the Agent has first priority, senior security interests in and liens upon (i) the Agent Collateral consisting of: (a) the Merchandise; (b) to the extent of the Debtor's ownership therein, the Additional Agent Goods; (c) all Proceeds (including, without limitation, credit card Proceeds); (d) the commission regarding the sale or other disposition of the Debtor's Consignment Goods under section 5.4 of the Agency Agreement; (e) the commission regarding the sale or other disposition of Owned FF&E under section 7 of the Agency Agreement; and (f) any Sharing Amount, but only up to the amount of Agent's percentage share of such Sharing Amount under section 3.2(a) of the Agency Agreement; and (2) all "proceeds" (within the meaning of Florida Statutes § 679.1021(a)(lll)) of each of the foregoing, to secure the full payment and performance of all obligations of the Debtor to Agent under the Agency Agreement. Upon entry of this Order, payment of the Initial Guaranty Payment, and delivery of the Letter of Credit to the Lender, the security interests and liens granted to Agent under the Agency Agreement and this Order shall be deemed properly perfected without the necessity of filing UCC-1 financing statement or any other documentation.
- 31. Without any further act by or on behalf of Agent or any other party (including, without limitation, the Lender and the Debtor), Agent's security interests and liens in the Agent Collateral are (i) validly created, (ii) effective upon entry of this Order, perfected, and (iii) senior to all other liens and security interests, <u>provided</u>, <u>however</u>, that (i) until the Debtor receive payment in full of the Guaranteed Amount, the Additional Agent Goods Fee, the Sharing

Amount (if any), Expenses, the proceeds realized upon the sale of Owned FF&E (less the FF&E Commission) or Additional Guaranteed Amount, and such other amounts due to the Debtor under the Agency Agreement, the security interest granted to the Agent under the Agency Agreement shall remain junior and subordinate in all respects to the security interests of the Lender in the Agent Collateral (other than the Additional Agent Goods and proceeds thereof in which the Debtor have no property or other interest and the Lender has has no security interest or other lien), but solely to the extent and amount of the unpaid portion of the Guaranteed Amount, the Additional Agent Goods Fee, the Sharing Amount (if any), Expenses, the proceeds realized upon the sale of Owned FF&E (less the FF&E Commission) or the Additional Guaranteed Amount, and such other amounts due to the Debtor under the Agency Agreement, and (ii) upon payment in full of the Guaranteed Amount, the Additional Agent Goods Fee, the Sharing Fee (if any), Expensess, the proceeds realized upon the sale of Owned FF&E (less the FF&E Commission) or the Additional Guaranteed Amount, and such other amounts due the Debtor under the Agency Agreement, any security interest or lien of the Lender in the Agent Collateral shall be junitor and subordinate in all respects to the security interest and liens of the Agent. The Debtor shall cooperate with Agent with respect to all filings (including, without limitation, UCC-1 financing statements) and other actions to the extent reasonably requested by Agent in connection with the security interests and liens granted under the Agency Agreement.

G. Other Provisions

32. The Agent shall not be liable for any claims against the Debtor, and the Debtor shall not be liable for any claims against Agent, in each case, other than as expressly provided for in the Agency Agreement. The Agent shall have no successor liability whatsoever with respect to any Encumbrances or claims of any nature that may exist against the Debtor,

including, without limitation, the Agent shall not be, or to be deemed to be: (i) a successor in interest or within the meaning of any law, including any revenue, successor liability, pension, labor, ERISA, bulk-transfer, products liability, tax or environmental law, rule or regulation, or any theory of successor or transferee liability, antitrust, environmental, product line, de facto merger or substantial continuity or similar theories; or (ii) a joint employer, co-employer or successor employer with the Debtor, and the Agent shall have no obligation to pay the Debtor's wages, bonuses, severance pay, vacation pay, WARN act claims (if any), benefits or any other payments to employees of the Debtor, including pursuant to any collective bargaining agreement, employee pension plan, or otherwise, except as expressly set forth in the Agency Agreement.

- 33. The Agent is a party in interest and shall have the ability to appear and be heard on all issues related to or otherwise connected to this Order, the various procedures contemplated herein, any issues related to or otherwise connected to the Sale, and the Agency Agreement.
- 34. Nothing contained in any plan confirmed in the Debtor's chapter 11 case or any order of this Court confirming such plan or in any other order in this chapter 11 case (including any order entered after any conversion of this case to a case under chapter 7 of the Bankruptcy Code) shall alter, conflict with, or derogate from, the provisions of the Agency Agreement or the terms of this Order.
- 35. The Agency Agreement and related documents may be modified, amended or supplemented by the parties thereto in accordance with the terms thereof without further order of this Court, provided that the Creditors Committee and the Lender consent, which consent shall not be unreasonably withheld or delayed.
- 35. Except with respect to any Governmental Unit (as to which the provisions of Paragraph 16 and 17 shall apply), this Court shall retain exclusive jurisdiction with regard to all

issues or disputes relating to this Order or the Agency Agreement, including, but not limited to, (i) any claim or issue relating to any efforts by any party or person to prohibit, restrict or in any way limit banner and signwalker advertising, including with respect to any allegations that such advertising is not being conducted in a safe, professional and non-deceptive manner, (ii) any claim of the Debtor, the landlords and/or the Agent for protection from interference with the Sale, (iii) any other disputes related to the Sale, and (iv) to protect the Debtor and/or the Agent against any assertions of Encumbrances. No such parties or persons shall take any action against the Debtor, the Agent, the landlords or the Sale until this Court has resolved such dispute. This Court shall hear the request of such parties or persons with respect to any such disputes on an expedited basis, as may be appropriate under the circumstances.

36. Notwithstanding Bankruptcy Rules 4001 and 6004, or any other law that would serve to stay or limit the immediate effect of this Order, this Order shall be effective and enforceable immediately upon entry and its provisions shall be self-executing. In the absence of any person or entity obtaining a stay pending appeal, the Debtor and the Agent are free to perform under the Agency Agreement at any time, subject to the terms of the Agency Agreement.

37. To the extent that anything contained in this Order explicitly conflicts with a provision in the Agency Agreement or the GOB Sale Guidelines, this Order shall govern and control.

#

Submitted by:

Paul Steven Singerman, Esq. Berger Singerman LLP 1450 Brickell Avenue, Suite 1900 Miami, FL 33131 Tel. (305) 755-9500

Fax: (305) 714-4340

E-mail: singerman@bergersingerman.com

Copy furnished to:

Paul Steven Singerman, Esq.

(Attorney Singerman is directed to serve a signed copy of this Order upon all interested parties and to file a Certificate of Service with the Court.)

Composite Exhibit A

First Amendment and Agency Agreement

FIRST AMENDMENT TO AGENCY AGREEMENT

This First Amendment (the "First Amendment") to the Agency Agreement dated as of April 17, 2015 (the "Agreement") is dated as of the 6th day of May, 2015, by and among SIMPLY FASHION STORES LTD (the "Merchant") and a contractual joint venture comprised of HILCO MERCHANT RESOURCES, LLC and GORDON BROTHERS RETAIL PARTNERS, LLC ("Agent"). Merchant and Agent are referred to, collectively, as "Parties" and each, individually, as a "Party."

RECITALS

WHEREAS, the Parties entered into the Agreement; and

WHEREAS, the Parties desire to amend the Agreement.

NOW, THEREFORE, in consideration of the premises and the mutual covenants herein contained, and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged by the Parties, the Parties hereby agree as follows:

AMENDMENT

- **SECTION 1.1** Amendment to Exhibit 1. Exhibit 1 to the Agreement is hereby deleted in its entirety and replaced with Exhibit 1 to this First Amendment.
- **SECTION 1.2** Amendment to Exhibit 4.1. Exhibit 4.1 to the Agreement is hereby deleted in its entirety and replaced with Exhibit 4.1 to this First Amendment.
- **SECTION 1.3** Amendment to Section 5.3(a). Section 5.3(a) is hereby amended by adding the following at the end of the existing text:

For purposes of determining Retail Price, if an item of Merchandise in a Store has more than one ticketed price, marked price, shelf price, or "buy one get one", "BOGO", "two for", or similar promotion price, or if multiple items of the same STYLE have different ticketed prices, marked prices, shelf prices, or "buy one get one", "BOGO", "two for", or similar promotion prices, the lowest ticketed price, marked price, shelf price, "buy one get one", "BOGO", "two for", or similar promotion price on any such item shall prevail for such item or for all such items within the same STYLE, as the case may be, that are located within the same Store (as the case may be, the "Lowest Store Price"), unless it is reasonably determined by Merchant or Agent that the applicable Lowest Store Price was mismarked or such item was priced because it was damaged or marked "as is," in which case the correct price (or in the case of a damaged good that is Defective Merchandise, the agreed upon price under Section 5.3(b)) shall control.

SECTION 1.4 <u>Amendment to Section 6.1</u>. Section 6.1 is hereby amended by adding the following at the end of the existing text:

A Vacate Notice is effective on the day on which such Vacate Notice is provided

to the Merchant if such Vacate Notice is provided to the Merchant and Lender by the Agent in accordance with section 17.1 of this Agreemeth by no later than 12 noon ET on such day. If such Vacate Notice is provided to the Merchant and Lender by the Agent after 12 noon ET on such day, such Vacate Notice shall be effective on the day following the day on which such notice was provided to the Merchant and Lender.

SECTION 1.5 Entire Agreement; Amendment. This First Amendment, the Agreement, all prior amendments, and the documents delivered pursuant hereto constitute the entire agreement and understanding among the Parties hereto and supersede and revoke any prior agreement or understanding relating to the subject matter thereof. No change, amendment, termination or attempted waiver of any of the provisions hereof shall be binding upon the other Party unless reduced to writing and signed by the Party against whom such change, amendment, termination or waiver is sought to be enforced.

SECTION 1.6 Counterparts and Execution. This First Amendment may be executed in two or more counterparts, each of which shall be deemed an original and all of which together shall constitute one and the same instrument. A manual signature on this First Amendment or other documents to be delivered pursuant to this First Amendment, an image of which shall have been transmitted electronically, will constitute an original signature for all purposes. The delivery of copies of this First Amendment or other documents to be delivered pursuant to this First Amendment, including executed signature pages where required, by electronic transmission will constitute effective delivery of this First Amendment or such other document for all purposes.

IN WITNESS WHEREOF, the undersigned parties have entered into and executed this First Amendment to be effective as of the day and year first above written.

	MERCHANT:
	Name: Catherine E. Youngman Soneet R. Kapila As Chi'd Restaucturing Office AGENT:
	HILCO MERCHANT RESOURCES, LLC By: Name: Ian Fredericks Its: VP & Assistant General Counsel, Managing Member
	GORDON BROTHERS RETAIL PARTNERS,
	Name: Microsa Currora Its: Given Canal
CONSENTED TO BY: JNS INVT, LLC	
By: Name: Its:	

IN WITNESS WHEREOF, the undersigned parties have entered into and executed this First Amendment to be effective as of the day and year first above written.

	MERCHANT:
	Name:
	AGENT:
	By: Name: Its: VP & Assistant General Counsel, Managing Membe GORDON BROTHERS RETAIL PARTNERS, LLC
	Name: Microse Connect Its: Given Count
CONSENTED TO BY: JNS INVT, LLC By: Name: Swapni 5 Shaw Its: Manager	



Simply Fashions Exhibit 1

Store List

Store #	Location Type	Name	Address	City	State	Zip	Phone	Open Date	Selling Sq. Ft.
14	SIMPLY FASHIONS	LEE STREET SHOPPING CTR	4523 LEE STREET	ALEXANDRIA	LA	71302	318/619-9940	05/29/98	3000
15	SIMPLY FASHIONS	INWOOD CENTRAL	8161 ANTOINE	HOUSTON	TX	77088	281/931-5450	03/15/02	3,272
16	SIMPLY FASHIONS	PEMBERTON SQUARE	3505 PEMBERTON SQUARE BLVD #43	VICKSBURG	MS	39180	601/629-9562	08/28/98	4,573
18	SIMPLY FASHIONS	GATEWAY SHOPPING CENTER	5238-14 NORWOOD AVENUE	JACKSONVILLE	FL	32208	904/768-1488	02/20/93	4532
19	SIMPLY FASHIONS	AMERICAN WAY PLAZA	4667 AMERICAN WAY	MEMPHIS	TN	38118	901/362-2545	03/19/93	4800
20	SIMPLY FASHIONS	JEFFERSON SQUARE S/C	2801 OLIVE ST STE. 8	PINE BLUFF	AR	71603	870/534-1432	05/28/93	4,672
21	DOTS	EISENHOWER CROSSING	4673 PRESIDENTIAL PKWY, SUITE M	MACON	GA	31206	478/960-3928	7/3/14	4,368
22	SIMPLY FASHIONS	FREESTANDING	69-73 W. CHELTEN AVENUE	PHILADELPHIA	PA	19144	215/435-1317	3/21/14	4,858
24	SIMPLY FASHIONS	CENTER CITY	1629 POPLAR AVE	MEMPHIS	TN	38104	901/274-9551	7/16/05	3,800
26	SIMPLY FASHIONS	TAMPA FESTIVAL CENTRE	2525 E. HILLSBOROUGH AVE. STE.101	TAMPA	FL	33610	813/237-5150	10/28/93	3,900
27	SIMPLY FASHIONS	GENTILLY S/C	3155 GENTILLY BLVD	NEW ORLEANS	LA	70122	504/286-0793	10/01/93	3,018
30	SIMPLY FASHIONS	MERIDIAN MARKETPLACE	2526 67TH AVE LOOP, #106	MERIDIAN	MS	39307	601/553-2990	5/2/08	3,067
31	DOTS	EASTGATE S/C	26200 EASTGATE BLVD	ROSEVILLE	MI	48066	586/675-0852	6/7/14	7,450
32	SIMPLY FASHIONS	FOWLERS PLAZA	1433 S. EISENHOWER PKWY	MACON	GA	31206	478/781-9672	5/11/07	4,300
33	SIMPLY FASHIONS	CHATHAM VILLAGE SQUARE	8658 S. COTTAGE GROVE AVE, UNIT #402	CHICIAGO	IL	60619	312/485-4518	1/31/15	3,060
36	SIMPLY FASHIONS	LINDELL MARKETPLACE	4167 LINDELL BLVD	ST. LOUIS	MO	63108	314/296-2106	01/30/15	3,850
37	SIMPLY FASHIONS	GENTILLY S/C	4800 CHEF MENTEUR HWY SUITE F	NEW ORLEANS	LA	70126	504/940-2255	10/05/10	3,000
39	SIMPLY FASHIONS	FREESTANDING	456 MEETING STREET	CHARLESTON	SC	29403	843/722-2526	06/30/00	3880
40	DOTS	BRICKTOWN SQUARE	6560 WEST FULLERTON AVE	CHICAGO	IL	60707	224/213-0590	1/31/15	6,000
41	DOTS	EAST FOREST PLAZA	5424 FOREST DRIVE	COLUMBIA	SC	29206	803/239-6545	1/30/15	4,723
43	SIMPLY FASHIONS	LINCOLN S/C	26150 GREENFIELD RD	OAK PARK	MI	48237	248/514-1483	7/25/14	7,800
44	SIMPLY FASHIONS	N/A	2111 HWY 82 EAST, SUITE C	GREENVILLE	MS	38710	662/334-3138	02/08/02	3,000
45	SIMPLY FASHIONS	NORTH EASTWOOD PLAZA	8939 EAST 38TH STREET	INDIANAPOLIS	IN	46226	317/897-8935	02/08/02	3,300
46	SIMPLY FASHIONS	REDFORD PLAZA	9189 TELEGRAPH RD, SPACE #260	REDFORD	MI	48239	313/255-2376	4/10/09	3,660
47	DOTS	SALEM CONSUMER SQUARE	5477 SALEM AVE	DAYTON	OH	45426	937/823-4770	1/30/15	3,500
48	SIMPLY FASHIONS	SOUTHGATE SHOPPING CENTER	1631 GORDON HWY., SPACE 11	AUGUSTA	GA	30906	706/790-0669	05/27/94	6,570
49	SIMPLY FASHIONS	OVERBROOK PLAZA	5610 LANCASTER AVE UNIT #900 A&B	PHILADELPHIA	PA	19131	267/317-0453	11/26/14	4,370
52	DOTS	VICTORY CROSSING S/C	4010 A VICTORY BLVD	PORTSMOUTH	VA	23701	757/532-6617	1/30/15	5,250
53	SIMPLY FASHIONS	95TH & JEFFREY	2023 E. 95TH STREET	CHICAGO	IL	60617	773/933-8660	10/3/08	2,717
54	SIMPLY FASHIONS	SUPER 1 FOODS S/C	2600 WAGGONER AVE., STE. 208	SHREVEPORT	LA	71108	318/636-1354	03/27/04	12,006
55	SIMPLY FASHIONS	NORTHWOOD S/C	4445 NORTH STATE ST.	JACKSON	MS	39206	601/362-0420	04/28/04	7,135
58	SIMPLY FASHIONS	FAMILY DOLLAR BUILDING	1024 NORTH STATE STREET	CLARKSDALE	MS	38614	662/627-1040	8/27/03	3,000
59	SIMPLY FASHIONS	LINWOOD SQUARE S/C	3135 PROSPECT	KANSAS CITY	MO	64128	816/861-1015	7/15/94	8000
62	SIMPLY FASHIONS	MT. CLARE JUNCTION	1229 W. PRATT STREET, SUITE B	BALTIMORE	MD	21233	410/244-8440	10/04/06	3,000
63	SIMPLY FASHIONS	THE LANDINGS	16743 TORRENCE AVE	LANSING	IL	60438	708/418-3830	5/4/09	6,000
66	SIMPLY FASHIONS	N.W. JUNCTION S/C	3188 W. NORTHSIDE DR.	JACKSON	MS	39213	601/366-7365	9/25/89	2847
68	SIMPLY FASHIONS	WYANDOTTE PLAZA	7722 STATE AVE	KANSAS CITY	KS	66112	913/302-7118	4/4/14	4,490
70	DOTS	OAKLAND POINTE PLAZA	332 N.TELEGRAPH RD	PONTIAC	MI	48341	248/309-0487	1/30/15	8,600
70 72	SIMPLY FASHIONS	NORTHSIDE PLAZA	3018 MEMORIAL PKWY NW	HUNTSVILLE	AL	48341 35810	256/859-8004	2/14/09	4,000
72 75	DOTS	MARCUS PLAZA	624 WEST MAIN STREET	NORWICH	CT				3,850
						06360	860/222-4126	11/14/14	
76 77	DOTS DOTS	WATERTOWER PLAZA GATEWAY MARKETPLACE	40 WATERTOWER PLAZA 1355 W. 8 MILE RD	LEOMINSTER HIGHLAND PARK	MA MI	01453	774/364-2450	1/30/15	3,600
						48203	248/228-7987	6/7/14	4,000
79	DOTS	WHITEHALL S/C	2180 MACARTHUR RD SPACE 6-A	WHITEHALL	PA	18052	484/523-0298	1/30/15	4,250
80	DOTS	ROUNDTREE PLACE	2477 ELLSWORTH RD	YPSILANTI	MI	48197	734/740-2991	1/30/15	4,000
82	SIMPLY FASHIONS	SOUTHLAND MALL	20505 S. DIXIE HWY, SUITE #829	CUTLER BAY	FL	33189	305/234-6131	5/1/14	5,373
85	SIMPLY FASHIONS	K-MART PLAZA	2107 SOUTH US HWY 1	FT PIERCE	FL	34950	561/489-8068	03/31/00	2500
86	SIMPLY FASHIONS	BURLINGTON PLAZA	3753 AUSTELL RD, SUITE 160 & 170	AUSTELL	GA	30106	770/819-8150	6/26/10	3,200
91	SIMPLY FASHIONS	CHICAGO HEIGHTS HILLTOP PLAZA	1333 HILLTOP AVE, UNIT D	CHICAGO HEIGHTS	IL	60411	708/481-8730	12/12/13	3,708

Hilco Merchant Resources, LLC 5/5/2015

Simply Fashions Exhibit 1

Store List

Store #	Location Type	Name	Address	City	State	Zip	Phone	Open Date	Selling Sq. Ft.
92	DOTS	NEWPORT S/C	1765 MONMOUTH STREET	NEWPORT	KY	41071	859/414-4068	10/25/14	4,000
94	SIMPLY FASHIONS	PINE HILLS MARKETPLACE	5311 WEST COLONIAL DRIVE	ORLANDO	FL	32808	407/578-6079	02/12/03	3,600
95	SIMPLY FASHIONS	CAHOKIA VILLAGE S/C	1024 CAMP JACKSON ROAD	CAHOKIA	IL	62206	618/337-6106	11/11/94	2908
97	SIMPLY FASHIONS	THE SHOPS AT WHITE OAK VILLAGE	4501 SOUTH LABURNUM AVE SPACE 145	RICHMOND	VA	23231	804/222-2117	9/17/10	3,965
99	SIMPLY FASHIONS	GREENWOOD WEST	2304 HWY. 82 W.	GREENWOOD	MS	38930	662/453-7437	8/01/90	2880
100	DOTS	CHERRYDALE POINT S/C	1510 POINSETT HWY	GREENVILLE	SC	29609	864/270-3583	1/30/15	3,900
101	SIMPLY FASHIONS	NEWMARKET SOUTH S/C	105 NEW MARKET SQUARE EAST	NEWPORT NEWS	VA	23605	757/244-5041	2/10/95	4538
102	DOTS	FOOD 4 LESS S/C	6930 S. ASHLAND AVE, SUITE 3	CHICAGO	IL	60636	312/550-0866	6/10/14	3,900
104	SIMPLY FASHIONS	SOUTHLAKE MALL	1000 SOUTHLAKE CIRCLE, SUITE 2409	MORROW	GA	30260	404/615-5158	5/31/14	3,824
106	DOTS	RACEWAY PARK S/C	13053 ASHLAND AVE	CALUMET PARK	IL	60827	312/848-7624	5/31/14	4,000
107	SIMPLY FASHIONS	POINCIANA PLAZA	1225 W. 45TH STREET, SUITE #303 & #304	MAGNOLIA PARK	FL	33407	561/494-0705	2/25/2011	2,400
109	DOTS	HERITAGE PLACE S/C	12551 OLIVE BLVD	CREVE COEUR	MO	63141	314/304-4598	1/30/15	4,800
110	SIMPLY FASHIONS	POPLAR PLAZA S/C	4930 POPLAR LEVEL ROAD	LOUISVILLE	KY	40219	502/968-7435	2/28/95	5000
111	SIMPLY FASHIONS	LANGLEY SQUARE	39 WEST MERCURY BLVD	HAMPTON	VA	23669	757/723-1536	11/12/99	3440
114	SIMPLY FASHIONS	NORTH POINTE S/C	5430G NORTH TRYON STREET	CHARLOTTE	NC	28213	704/599-9777	05/17/96	3200
116	DOTS	SPENCER SQUARE	3552 SPENCER HWY	PASADENA	TX	77504	713/205-2976	1/30/15	4,000
118	DOTS	VILLAGE S/C	3604 VILLAGE COURT	GARY	IN	46408	219/742-8550	6/27/14	6,500
122	DOTS	LYNNHAVEN NORTH S/C	2720 N. MALL DR SUITE 200	VIRGINIA BEACH	VA	23452	757/266-1324	1/30/15	5,000
123	DOTS	CEDAR HILLS S/C	3606 BLANDING BLVD	JACKSONVILLE	FL	32210	904/404-6926	11/8/14	6,000
124	SIMPLY FASHIONS	ROEBUCK MARKETPLACE	9164 PARKWAY EAST	BIRMINGHAM	AL	35206	205/815-1173	8/15/08	3,000
126	SIMPLY FASHIONS	FREESTANDING	9 N 52ND ST	PHILADELPHIA	PA	19139	267/449-4409	1/30/15	6,336
127	SIMPLY FASHIONS	RIVER COMMONS	4021 BEHRMAN PLACE, SPACE 1-2	NEW ORLEANS	LA	70114	504/366-3967	05/01/04	3000
128	SIMPLY FASHIONS	EASTLAND HILL CENTER	605 GALLATIN ROAD	NASHVILLE	TN	37206	615/226-1066	06/30/00	4200
129	SIMPLY FASHIONS	WEST TOWN PLAZA	4263 WEST THIRD STREET	DAYTON	ОН	45417	937/263-6167	03/20/04	4,350
130	SIMPLY FASHIONS	GRAND BOULEVARD PLAZA	5401 S. WENTWORTH AVE., UNIT #7	CHICAGO	IL	60609	773/924-3379	5/01/95	3500
133	SIMPLY FASHIONS	CAPITOL PLAZA	2288 EAST SOUTH BLVD	MONTGOMERY	AL	36116	334/284-3433	2/28/93	3421
137	SIMPLY FASHIONS	PARKWAY PLAZA	2001 NW EVANGELIN THRUWAY, SUITE 400	LAFAYETTE	LA	70501	337/235-8980	04/04/03	3.247
138	SIMPLY FASHIONS	N/A	5963 PLANK ROAD, SUITE B	BATON ROUGE	LA	70805	225/355-1336	10/06/06	10,140
139	SIMPLY FASHIONS	DODGE PLAZA	7748 LANDOVER ROAD	LANDOVER	MD	20785	301/773-3691	11/10/95	2400
141	SIMPLY FASHIONS	FOOD WORLD PLAZA SOUTH	4200 MCFARLAND BOULEVARD	TUSCALOOSA	AL	35405	205/349-4360	01/01/92	2600
144	SIMPLY FASHIONS	FRAYSER PLAZA	2146 FRAYSER BOULEVARD	MEMPHIS	TN	38127	901/354-8886	04/10/98	3000
145	SIMPLY FASHIONS	IVES DAIRY CROSSING S/C	19925 NW 2ND AVE, BAY 1	MIAMI GARDENS	FL	33169	785/510-2707	10/3/14	3,000
148	SIMPLY FASHIONS	SHOPS AT NORTHEAST VILLAGE	14477 GRATIOT AVENUE	DETROIT	MI	48205	313/527-0282	03/01/01	6012
151	SIMPLY FASHIONS	TOWN & COUNTRY PLAZA	3300 N. PACE BLVD. #35	PENSACOLA	FL	32505	850/432-5711	01/01/92	4312
153	DOTS	SHOPPES AT GARNER	4446 FAYETTEVILLE RD	RALEIGH	NC	27603	919/368-1387	1/30/15	4,200
154	DOTS	NORTHLINE CROSSING S/C	4400 NORTH FREEWAY E 1000	HOUSTON	TX	77022	281/507-8484	1/31/15	4,200
155	SIMPLY FASHIONS	CANE RUN	3230 CRUMS LANE	LOUISVILLE	KY	40216	502/449-0883	02/23/96	3000
157	SIMPLY FASHIONS	EASTOVER S/C	5037 Indian Head Hwy	Oxon Hill	MD	20745	301/749-1435	4/26/96	3200
158	DOTS	STEELYARD COMMONS S/C	3495 STEELYARD DRIVE	CLEVELAND	OH	44109	216/212-2560	6/21/14	6.500
159	SIMPLY FASHIONS	WESTERN HILLS MALL	7201 AARON ARONOV DRIVE, STE. 36A	FAIRFIELD	AL	35064	205/925-6004	9/29/06	3.850
160	SIMPLY FASHIONS	THE SHOPPES OF LIBERTY CITY	1156 NW 54TH STREET	MIAMI	FL	33127	305/757-0797	08/24/00	3000
161	SIMPLY FASHIONS	BLUE PARKWAY TOWN CENTER	4307 EAST 50TH TERRACE, SPACE A & B	KANSAS CITY	MO	64130	816/921-2810	2/3/06	3.176
162	SIMPLY FASHIONS	GRIGGS ROAD S/C	4421 GRIGGS ROAD	HOUSTON	TX	77021	713/741-4604	2/15/03	5.000
163	SIMPLY FASHIONS	MADISON SQUARE	2585 MADISON AVENUE	MONTGOMERY	AL	36107	334/241-8995	10/24/08	4,000
164	SIMPLY FASHIONS	SELMA MALL	1391 HIGHLAND AVE., SUITE 113	SELMA	AL	36703	334/418-8884	03/13/96	3000
165	SIMPLY FASHIONS	FAMILY DOLLAR CENTER	3111 W. CAPITOL STREET	JACKSON	MS	39209	601/354-5564	02/27/04	3.297
166	SIMPLY FASHIONS	LANTANA VILLAGE SQUARE	1301 S. DIXIE HWY	LANTANA	FL	33462	561/586-1959	2/13/09	3,297
167	SIMPLY FASHIONS	AZALEA SHOPPING CENTER	2121 BEMISS ROAD	VALDOSTA	GA	31602	912/245-8040	3/15/96	3,500 2920
107	SIIVIPLT FASHIONS	ALALEA SHUPPING CENTER	ZIZI BEIVIISS KUAD	VALDUSTA	GA	31002	312/245-8040	3/12/30	2920

Hilco Merchant Resources, LLC 5/5/2015

Store List

Store #	Location Type	Name	Address	City	State	Zip	Phone	Open Date	Selling Sq. Ft.
170	DOTS	TAYLOR COMMONS	1341 S. MAIN AVE SUITE 210	SCRANTON	PA	18504	570/209-0637	1/30/15	7,040
173	SIMPLY FASHIONS	EUTAW VILLAGE	2710 BRAGG BLVD	FAYETTEVILLE	NC	28303	910/321-9483	07/21/2000	6,500
174	SIMPLY FASHIONS	NORTHSIDE SHOPPING CENTRE	7900 NORTHWEST 27TH AVE., STE. 11B	MIAMI	FL	33147	305/691-1604	07/24/96	4,000
175	SIMPLY FASHIONS	LAMAR AIRWAYS S/C	2236 LAMAR AVENUE	MEMPHIS	TN	38114	901/454-9377	05/25/96	4033
176	DOTS	STONY ISLAND PLAZA	1627 E. 95TH ST	CHICAGO	IL	60617	312/914-1902	1/30/15	3,500
178	DOTS	WEST ALLIS	6822 W. GREENFIELD AVE. #A-114	WEST ALLIS	WI	53214	414-207-2971	03/13/15	-
180	SIMPLY FASHIONS	WEST RIDGE	3050 MARTIN LUTHER KING JR DR. SPACE F1	ATLANTA	GA	30311	404/699-7395	4/14/04	2,886
182	SIMPLY FASHIONS	LAUDERHILL MALL	1341 N.W. 40TH AVENUE	LAUDERHILL	FL	33313	954/587-2859	09/01/1989	3600
183	SIMPLY FASHIONS	SAVE A LOT CTR	1181 E MAIN STREET	COLUMBUS	ОН	43205	614/253-3779	5/21/08	3,000
184	DOTS	MODEL T PLAZA	14110 WOODWARD AVENUE	HIGHLAND PARK	MI	48203	248/915-8625	8/29/14	3,840
185	SIMPLY FASHIONS	KING SHOPPING CENTER	7057 MARTIN LUTHER KING DR	LANDOVER	MD	20785	301/773-5117	6/21/99	4,360
186	DOTS	SOUTHGATE USA S/C	21640 LIBBY ROAD	MAPLE HEIGHTS	OH	44137	216/347-3418	1/31/15	4,400
187	DOTS	THRUWAY PLAZA	150 THRUWAY PLAZA DR	CHEEKTOWAGA	NY	14225	716/462-2133	1/30/15	4.978
188	DOTS	MIRACLE MILE S/C	4128 WILLIAM PENN HIGHWAY	MONROEVILLE	PA	15146	412/552-9383	11/17/14	4,700
189	SIMPLY FASHIONS	MARSHFIELD PLAZA	11612 S. MARSHFIELD AVE	CHICAGO	IL	60643	773/660-8840	9/18/09	3,000
190	DOTS	SOUTHERN PLAZA	4200 SOUTH EAST ST #31	INDIANAPOLIS	IN	46227	317/650-5238	1/30/15	5,003
191	DOTS	FONDREN SOUTHWEST VILLAGE	11168 FONDREN ROAD	HOUSTON	TX	77096	832/544-0371	8/29/14	5,500
192	SIMPLY FASHIONS	VILLAGE GREEN	1126 280 BYPASS	PHENIX CITY	AL	36867	334/298-1897	2/25/11	4,200
193	SIMPLY FASHIONS	N/A	3163 CLEVELAND AVENUE	COLUMBUS	OH	43224	614/268-9112	08/15/96	4,200
196	SIMPLY FASHIONS	DOLPHIN PLAZA	17235 NW 27TH AVENUE	MIAMI GARDENS	FL	33056	954/540-4395	9/12/14	3,022
197	SIMPLY FASHIONS	BOSSIER TOWN CENTER	3143 EAST TEXAS STREET	BOSSIER CITY	LA	71111	318/747-5568	10/01/89	3,568
198	SIMPLY FASHIONS	GATEWAY	3250 JACKSON AVE	MEMPHIS	TN	38122	901/458-7111	10/01/89	4,979
199	SIMPLY FASHIONS			DETROIT	MI	48234		08/30/96	3300
	DOTS	BEL AIR S/C	8780 E. 8 MILE ROAD 801 INDUSTRIAL BLVD SUITE 140		TN		313/892-1683		
200		MARKETPLACE AT SMYRNA		SMYRNA	MI	37167	615/427-8693	1/31/15	5,000
201	SIMPLY FASHIONS	GALLERY AT WARREN CONNER	4905 CONNER AVE	DETROIT		48215	313/824-6314	2/11/05	4,290
203	SIMPLY FASHIONS	ELLIS AVENUE PLAZA	1335 ELLIS AVENUE, #15	JACKSON	MS	39204	601/949-8448	8/1/97	3071
204	SIMPLY FASHIONS	RIVERVIEW PLAZA	264 WEST GENESSEE STREET	SAGINAW	MI	48602	989/753-1260	11/05/04	3,000
205	SIMPLY FASHIONS	CULLEN PLAZA	9426 CULLEN BLVD	HOUSTON	TX	77051-3321	713/738-8552	03/08/02	3000
207	SIMPLY FASHIONS	THE COMMONS ROUTE 20	4417 WEST 5TH AVENUE	GARY	IN	46406	219/949-6428	09/25/04	4,032
208	DOTS	JEFFERSON VILLAGE	11226 JEFFERSON AVENU	DETROIT	MI	48214	313/815-0246	11/7/14	3,600
212	SIMPLY FASHIONS	ROYAL TOWN CENTER	8900 B WEST EIGHT MILE ROAD	FERNDALE	MI	48220	248/546-6565	03/07/03	3,612
213	SIMPLY FASHIONS	COLLEGE PARK COMMONS	7940 WEST OUTER DRIVE	DETROIT	MI	48235	313/533-7030	6/5/09	3,602
215	SIMPLY FASHIONS	SOUTH SLAPPEY VILLAGE	1030 WEST GORDON AVENUE, SUITE F	ALBANY	GA	31701	229/446-9956	9/12/97	5840
216	SIMPLY FASHIONS	WALNUT HILLS PLAZA	2047 CRATER RD #104	PETERSBURG	VA	23805	804/862-8529	09/24/99	4680
218	DOTS	SHOPS AT WHITE OAK VILLAGE	4501 SOUTH LABURNUM AVE SUITE 560	RICHMOND	VA	23231	757/603-7020	1/30/15	4,552
219	DOTS	COUNTRY CLUB PLAZA	490 LINCOLN ST	WORCESTER	MA	01613	508/735-9290	1/31/15	5,355
221	SIMPLY FASHIONS	THE CROSSINGS AT HALLS FERRY	10835 OLD HALLS FERRY RD, STE.A	ST. LOUIS	MO	63136	314/867-7885	03/02/99	3000
223	DOTS	CHICAGO	741 & 743 EAST 47TH STREET	CHICAGO	IL	60653	773-717-0299	02/27/15	-
224	SIMPLY FASHIONS	SQUARE 67	2550 RED BIRD LANE, SUITE 207	DALLAS	TX	75237	214/330-9354	08/26/98	3720
225	SIMPLY FASHIONS	GREENS CROSSROADS	221 WEST GREENS ROAD	HOUSTON	TX	77067	281/872-4744	9/25/09	3,915
227	SIMPLY FASHIONS	PLAZA ON THE BLV	8035 WEST FLORISSANT AVE, SUITE F & G	JENNINGS	MO	63136	314/385-9597	11/3/06	3,200
228	DOTS	NORTHGATE S/C	339 SQUIRE ROAD	REVERE	MA	02151	617/716-9973	11/22/14	4,200
229	SIMPLY FASHIONS	JOYLAND S/C	15550 JOY ROAD	DETROIT	MI	48228	313/838-3836	2/27/98	3025
230	SIMPLY FASHIONS	SILVER MILL COURT S/C	6115 NORTH TEUTONIA AVE	MILWAUKEE	WI	53209	414/527-4252	01/31/03	4,300
231	SIMPLY FASHIONS	MARLOW HEIGHTS S/C	3923 BRANCH AVENUE	TEMPLE HILLS	MD	20748	301/316-2020	6/25/98	2979
233	SIMPLY FASHIONS	WESTPORT COMMONS S/C	3264 W. 87TH STREET	CHICAGO	IL	60652	708/710-9119	9/19/14	3,500
234	SIMPLY FASHIONS	TOWNE SOUTH	2525 SOUTH MONROE ST. STE. 20 & 21	TALLAHASSEE	FL	32301	850/656-7602	02/13/03	3,500
236	SIMPLY FASHIONS	LAWNDALE PLAZA	3230 WEST ROOSEVELT ROAD	CHICAGO	IL	60624	773/638-1565	11/04/98	3200

Store List

Store #	Location Type	Name	Address	City	State	Zip	Phone	Open Date	Selling Sq. Ft.
237	SIMPLY FASHIONS	BALTIMORE	2331 CLEANLEIGH DR.	BALTIMORE	MD	21234	443.826.8891	03/27/15	-
239	SIMPLY FASHIONS	SOUTH WIND PLAZA	5335 N. MILITARY TRAIL SPACE 31&32	WEST PALM BEACH	FL	33407	561/684-5399	04/10/98	3600
240	SIMPLY FASHIONS	BUCKNER COMMONS	9208 EAST RL THORNTON FREEWAY, STE.110	DALLAS	TX	75228	214/321-3733	02/27/04	4,500
241	DOTS	TINLEY PARK PLAZA	15967 SOUTH HARLEM AVE	TINLEY PARK	IL	60477	312/718-1235	1/30/15	5,000
242	SIMPLY FASHIONS	SOUTH PHILADELPHIA S/C	2426 SOUTH 23RD STREET	PHILADELPHIA	PA	19145	215/764-0260	3/1/14	2,380
246	SIMPLY FASHIONS	CITY PLAZA	3447 UNION BLVD	ST. LOUIS	MO	63115	314/381-9088	11/09/98	3000
248	SIMPLY FASHIONS	FREESTANDING	2000 E. 71ST STREET	CHICAGO	IL	60649	773/324-9770	4/16/10	4,320
249	DOTS	CROSSROADS CENTER	6777 SPENCER HWY	PASADENA	TX	77505	832/392-5687	1/30/15	3,000
252	SIMPLY FASHIONS	JEWEL OSCO S/C	117 WEST 87TH STREET	CHICAGO	IL	60620	773/874-6063	05/12/98	6562
254	DOTS	30TH STREET S/C	3034 CROMER AVE	CANTON	ОН	44709	234/804-2546	1/30/15	3,279
255	DOTS	MANCHESTER PARKADE	370 WEST MIDDLE TURNPIKE	MANCHESTER	СТ	06040	860/306-5958	2/11/15	4,691
256	DOTS	SOUTHFIELD	20176 W. 8 MILE ROAD	SOUTHFIELD	MI	48075	248-945-0282	02/27/15	-
257	SIMPLY FASHIONS	Baseline Plaza	20212 W EIGHT MILE RD	SOUTHFIELD	MI	48075	248/945-0282	09/25/98	2800
258	SIMPLY FASHIONS	SHOPPES ON MAIN	3662 EAST MAIN STREET, UNIT D	COLUMBUS	ОН	43213	614/231-2324	4/22/11	3,000
259	SIMPLY FASHIONS	SOUTH DEKALB PLAZA	2774 CANDLER RD	DECATUR	GA	30034	404/212-3880	6/23/99	3000
262	SIMPLY FASHIONS	MEADOWBROOK EAST	6707 MEADOWBROOK DRIVE	FT. WORTH	TX	76112	817/457-0580	9/19/03	4,875
263	SIMPLY FASHIONS	CHARITON SQUARE	4236 SOUTH BROADWAY	ST. LOUIS	MO	63111	314/832-9969	09/19/03	2,160
264	SIMPLY FASHIONS	LAMAR CROSSING	2938 LAMAR AVE, SUITE 116	MEMPHIS	TN	38114	901/743-8466	5/19/06	4,264
265	SIMPLY FASHIONS	WYOMING VILLAGE	1242 28TH STREET SW	WYOMING	MI	49509	616/531-6051	6/20/08	4,650
266	SIMPLY FASHIONS	STATE STREET	2524 STATE STREET	EAST ST. LOUIS	IL.	62205	618/875-5461	03/31/00	3000
268	SIMPLY FASHIONS	EASTLAND CENTER	18000 VERNIER RD SPACE #741	HARPER WOODS	MI	48225	313/527-1949	12/10/10	4,154
269	SIMPLY FASHIONS	ST. STEPHENS SQUARE	2312 ST. STEPHENS RD. SUITE 6	MOBILE	AL	36617	334/452-9331	06/23/00	2,977
270	SIMPLY FASHIONS	HIGHLAND LAKES CENTER	7459 WEST COLONIAL DRIVE	ORLANDO	FL	32818	407/298-5200	6/3/11	3.032
271	DOTS	CARRIAGE CROSSING MARKETPLACE	10233 EAST SHELBY DR	COLLIERVILLE	TN	38017	901/229-7596	1/30/15	4,080
272	SIMPLY FASHIONS	ERDMAN PLAZA	3933 ERMAN AVE	BALTIMORE	MD	21213	410/276-1441	3/28/97	4,000
273	DOTS	NORTHWEST CROSSING	5762 HOLLISTER STREET	HOUSTON	TX	77040	281/543-9144	1/30/15	4,000
274	SIMPLY FASHIONS	WOODLAND VILLAGE	6046-6048 WOODLAND AVENUE	PHILADELPHIA	PA	19142	215/990-8309	11/20/14	2,964
275	SIMPLY FASHIONS	CROSS COUNTRY PLAZA	3201 MACON ROAD	COLUMBUS	GA	31906	706/569-5800	05/04/02	3,906
278	SIMPLY FASHIONS	EDENS PLAZA	4105 WEST BELTLINE BLVD #50	COLUMBIA	SC	29204	803/251-2445	1/29/10	3,000
281	SIMPLY FASHIONS	WHITEHAVEN PLAZA	4275 ELVIS PRESLEY BLVD	MEMPHIS	TN	38116	901/332-2273	05/04/02	3,410
282	SIMPLY FASHIONS	RACEWAY PARK S/C	13065 S. ASHLAND AVENUE	CALUMET PARK	IL	60827	312/502-5380	10/03/14	4,000
283	SIMPLY FASHIONS	ALBANY CROSSING	244 CORDELE RD, SUITE 125	ALBANY	GA	31705	229/883-7722	5/13/11	2,800
284	SIMPLY FASHIONS	OAKLAND POINTE	362 TELEGRAPH ROAD	PONTIAC	MI	48341	248/333-3131	9/15/06	5,340
285	SIMPLY FASHIONS	ONE & OLNEY SQUARE	5675 NORTH FRONT ST #100	PHILADELPHIA	PA	19120	267/353-3188	4/18/14	4.042
287	DOTS	NORTHWEST CROSSING	130 BLACKHORSE PIKE	AUDUBON	NJ	08106	609/314-5040	1/30/15	4.037
288	SIMPLY FASHIONS	CLEVELAND AVENUE CROSSING	854 CLEVELAND AVENUE, SUITE 108	EAST POINT	GA	30344	404/780-4596	10/22/14	3,000
301	SIMPLY FASHIONS	CLOVERLEAF S/C	1814-A N. FRONTAGE ROAD	MERIDIAN	MS	39301	601/693-6801	01/01/92	7380
302	SIMPLY FASHIONS	BATESVILLE S/C	103 KEATING ROAD	BATESVILLE	MS	38606	662/563-0217	01/01/92	3000
303	SIMPLY FASHIONS	NORTHTOWN PLAZA	5510 NORTH FREEWAY	HOUSTON	TX	77076	713/697-3451	7/05/90	3835
317	SIMPLY FASHIONS	HAMMOND AIRE PLAZA	9626 AIRLINE HIGHWAY, STE. C-3	BATON ROUGE	LA	70827	225/927-6216	03/30/93	6000
327	SIMPLY FASHIONS	WALZEM PLAZA	5334 WALZEM ROAD	SAN ANTONIO	TX	78218	210/656-7141	01/01/92	4400
337	DOTS	TOWN & COUNTRY	3816 E. BROAD STREET	COLUMBUS	OH	43213	740/919-1112	8/29/14	5,887
338	DOTS	SOUTH PHILADELPHIA S/C	2300 W. PASSYUNK AVE	PHILADELPHIA	PA	19145	267/670-1143	8/1/14	4.428
343	DOTS	EASTOWN S/C	3954 LINDEN AVE	DAYTON	ОН	45432	937/307-9240	8/29/14	4,045
343	DOTS	NORTHERN LIGHTS	3487 CLEVELAND AVE	COLUMBUS	OH	43432	614/802-7240	8/29/14	4,500
348	DOTS	CENTURY CENTER	373 MEMORIAL AVENUE	W. SPRINGFIELD	MA	01089	413/313-4745	11/8/14	3,550
360	SIMPLY FASHIONS	FIVE POINTS WEST S/C	2209 BESSEMER ROAD	BIRMINGHAM	AL	35208	205/780-4206	08/30/90	4000
365	SIMPLY FASHIONS	CROSS CREEK	5648 GEORGETOWN ROAD	INDIANAPOLIS	IN	46254	317/388-0444	09/19/03	2,400
303	SHALL I WOLLOWS	CHOSS CILLIN	30-30 GEORGETOWN ROAD	HADIAINAF OLIS	IIN	40234	317/300-0444	03/13/03	2,400

Store List

Store #	Location Type	Name	Address	City	State	Zip	Phone	Open Date	Selling
31016 #	••				Juice				Sq. Ft.
369	SIMPLY FASHIONS	FREESTANDING	2419 MURPHY MILL RD	DOTHAN	AL	36303	334/792-7332	10/15/2010	4,030
377	DOTS	LINCOLN S/C	26130 GREENFIELD ROAD	OAK PARK	MI	48237	313/410-4264	8/29/14	4,800
500	SIMPLY FASHIONS	ANDERSON ROAD PLAZA	2600 ANDERSON ROAD	GREENVILLE	SC	29611	864/269-1204	01/01/92	5,900
501	SIMPLY FASHIONS	METRO PLAZA	1634 EAST 63RD STREET	KANSAS CITY	MO	64110	816/363-0780	12/15/12	3,000
504	SIMPLY FASHIONS	HIGHLAND PARK PLACE	14515 SOUTH WOODWARD AVENUE	DETROIT	MI	48203	313/865-6550	08/04/05	5111
506	DOTS	LIVONIA MARKETPLACE	29508 SEVEN MILE ROAD	LIVONIA	MI	48152	248/412-3564	10/24/14	4,000
507	SIMPLY FASHIONS	EDISTO VILLAGE SHOPPING CENTER	403 RIVERSIDE DRIVE S.W.	ORANGEBURG	SC	29115	803/533-0700	07/09/05	3328
508	DOTS	WESTSHORE PLAZA	1795 E. SHERMAN BLVD	MUSKEGON	MI	49444	231/215-6635	02/13/15	6,400
509	DOTS	KLINE PLAZA	130 B KLINE VILLAGE	HARRISBURG	PA	17104	717/317-3256	11/19/14	5,200
510	DOTS	CLINTON POINTE	33816 GRATIOT AVENUE	CLINTON TOWNSHIP	MI	48035	586/348-2803	10/24/14	3,250
511	SIMPLY FASHIONS	OAK STREET SHOPPING CENTER	845 OAK STREET	ATLANTA	GA	30310	404/753-0408	3/9/1995	3420
512	DOTS	NORGATE PLAZA	7235 N. KEYSTONE AVE. SUITE H	INDIANAPOLIS	IN	46240	317/448-3811	11/8/14	4,500
513	SIMPLY FASHIONS	MLK PLAZA	1300 NORTH GRAND AVE	SAINT LOUIS	MO	63113	314-535-5758	6/13/2003	3000
514	SIMPLY FASHIONS	NORTH CHARLESTON CENTER	5900 RIVERS AVE-SUITE H	NORTH CHARLESTON	SC	29406	843-554-0730	4/1/1998	3200
515	SIMPLY FASHIONS	SUMTER SQUARE SHOPPING CENTER	1009 BROAD STREET	SUMTER	SC	29150	803/773-0972	5/28/1992	2875
516	DOTS	COTTMAN & BUSELTON S/C	2121 COTTMAN AVE	PHILADELPHIA	PA	19149	267/572-0157	1/31/15	4,426
520	SIMPLY FASHIONS	BORDEAUX SHOPPING CENTER	3200 CLARKSVILLE HIGHWAY	NASHVILLE	TN	37208	615/726-1184	5/13/1992	3000
523	SIMPLY FASHIONS	WESTLAND SHOPPING CENTER	31284 MICHIGAN AVENUE	WESTLAND	MI	48185	734/728-3634	3/29/2000	3000
524	DOTS	THE SHOPS AT STERLING PONDS	33329 VAN DYKE AVE	STERLING HEIGHTS	MI	48312	586/212-3674	10/24/14	4,803
525	SIMPLY FASHIONS	NORTHWEST PLAZA	3269 W. SIEBENTHALER AVE., UNIT 5	DAYTON	ОН	45406	937/274-0096	1/28/1996	2864
526	SIMPLY FASHIONS	DORCHESTER SQUARE	4391 DORCHESTER ROAD	NORTH CHARLESTON	SC	29405	843-566-0028	08/01/05	2970
527	DOTS	CANTON CENTRE	4060 TUSCARAWAS STREET W.	CANTON	ОН	44708	234/850-4204	10/3/14	5,086
528	SIMPLY FASHIONS	GEORGETOWN PLAZA SHOPPING CENTER	4941 WEST 38TH STREET	INDIANAPOLIS	IN	46254	317/328-8958	4/1/1998	4200
529	SIMPLY FASHIONS	EASTGATE SHOPPING CENTER	3134 LOUISVILLE AVENUE	MONROE	LA	71201	318/362-1206	7/31/2003	4500
531	DOTS	MIDDLESEX S/C	1310 EASTERN BLVD	BALTIMORE	MD	21221	410/999-5097	10/25/14	3,000
535	SIMPLY FASHIONS	CRICHTON PLAZA	3038 A SPRINGHILL AVE.	MOBILE	AL	36607	251/479-5811	8/24/1995	3300
537	SIMPLY FASHIONS	CHESAPEAKE CROSSING SHOPPING CENTER	1949 S. MILITARY HWY.	CHESAPEAKE	VA	23320	757/545-1863	6/27/1999	3000
538	SIMPLY FASHIONS	CROSSROADS PLAZA	1958 STONE ROSE DRIVE	ROCKY MOUNT	NC	27802	252/446-6001	5/26/1993	4000
539	SIMPLY FASHIONS	HATTIESBURG SHOPPING CENTER	5891 U.S. HIGHWAY 49	HATTIESBURG	MS	39402	601/268-3725	10/23/2000	2700
541	DOTS	LIBERTY CENTER	3728 LIBERTY STREET	ERIE	PA	16508	814/450-9296	11/25/14	5,796
543	SIMPLY FASHIONS	LEHMBERG CROSSING	993 EAST ALABAMA STREET	EAST COLUMBUS	MS	39702	662/329-3237	4/26/2001	3200
544	DOTS	SHOP CITY PLAZA	179 SHOP CITY PLAZA	SYRACUSE	NY	13206	315/412-5800	10/10/14	4,959
545	DOTS	GREAT SOUTHERN S/C	3831 SOUTH HIGH STREET	COLUMBUS	OH	43207	614/867-2491	10/3/14	4,500
549	SIMPLY FASHIONS	FREEDOM SQUARE SHOPPING CENTER	1615 SOUTH IRBY STREET	FLORENCE	SC	29505	843-661-7847	10/26/1995	4,300 3440
554	DOTS	TOWN CENTER MARKET	2821 CHASTAIN MEADOWS PKWY NW #120	MARIETTA	GA	30066	404/858-2805	10/26/1993	4.800
555	SIMPLY FASHIONS	IROQUOIS MANOR	5326 SOUTH THIRD STREET	LOUISVILLE	KY	40214	502/368-7228	9/27/1990	3360
	SIMPLY FASHIONS				GA		•		2800
556 558	SIMPLY FASHIONS	WESLEY CHAPEL SQUARE SHOPPING CENTER SHELBY PLAZA	2389 WESLEY CHAPEL ROAD 4673 ELVIS PRESLEY BLVD.	DECATUR MEMPHIS	TN	30035 38116	770/593-0905	8/02/2005 7/28/1993	2800
558 559	DOTS				OH		901/396-7433		
		AUSTINTOWN PLAZA	6000 MAHONING AVENUE #252	YOUNGSTOWN		44515	330/369-6989	10/29/14	4,500
562	SIMPLY FASHIONS	NORTH POINT CENTER	3409 DICKERSON PIKE, STE 106	NASHVILLE	TN	37207	615/227-8765	5/13/2004	4000
563	DOTS	CONSUMER SQUARE WEST	3650 SOLDANO BLVD	COLUMBUS	OH	43228	516/507-0202	2/13/15	3,685
565	SIMPLY FASHIONS	BROADWAY CENTER	2043-A BROADWAY CENTER	KNOXVILLE	TN	37917	865/637-7025	8/4/1994	3000
571	SIMPLY FASHIONS	WESTWOOD SHOPPING CENTER	3709 JEWELLA AVE	SHREVEPORT	LA	71109	318/631-0970	4/19/2000	4055
575	SIMPLY FASHIONS	VILLAGE SHOPPING CENTER	3552 VILLAGE COURT	GARY	IN	46408	219/884-5180	6/14/2000	4550
576	SIMPLY FASHIONS	SOUTHGATE SHOPPING CENTER	4439 WEST FUQUA ROAD	HOUSTON	TX	77045	713/413-3397	9/26/2000	3300
577	SIMPLY FASHIONS	HICKORY RIDGE CROSSING	3641 HICKORY HILL	MEMPHIS	TN	38115	901/566-9443	8/05/00	4000
581	SIMPLY FASHIONS	GAINSVILLE SHOPPING CENTER	1208 NORTH MAIN STREET, UNIT 1208	GAINESVILLE	FL	32608	352/264-7990	7/9/05	3000
584	SIMPLY FASHIONS	SOUTHGATE PLAZA	303-B SOUTH SLAPPEY DRIVE	ALBANY	GA	31701	229/446-2340	4/25/2002	3456

			Store List						
Store #	Location Type	Name	Address	City	State	Zip	Phone	Open Date	Selling Sq. Ft.
595	SIMPLY FASHIONS	PARKWAY PLAZA	1223 SILAS CREEK PARKWAY	WINSTON-SALEM	NC	27127	336/724-3676	6/24/06	5,630
599	SIMPLY FASHIONS	COURTHOUST SQUARE	100 WEST ARIZONA AVE	RUSTON	LA	71270	318/255-5955	6/17/05	3,315
617	SIMPLY FASHIONS	ASHER	6420 COLONEL GLENN ROAD, SUITE 5	LITTLE ROCK	AR	72204	501/568-7334	7/1/05	2,400
243	-								5 125

15		RENT	AD VALOREM TAX	FILTER SERVICE	INSURANCE - BUSINESS	LICENSE - CITY	STATE & COUNTY	MAINTENANC E	SECURITY	TELEPHONE	UTILITIES	WASTE PICK UP	Total Occupancy	% Rent
	LEE STREET SHOPPING CTR	49	2	-	4	-	2		1	2	23	3	89	
16	INWOOD CENTRAL	-	3	-	4	-	0	4	1	2	17	2	32	5%
10	PEMBERTON SQUARE	55	2	-	4	0	-	4		2	16	-	83	
18	GATEWAY SHOPPING CENTER	82	6	-	4	0	1	2	1	2	42	-	140	
	AMERICAN WAY PLAZA	115	1	-	4	1	1	42	1	2	28	3	198	
	JEFFERSON SQUARE S/C	66	2	-	4	0	-	1	1	2	17	7	99	
	EISENHOWER CROSSING	166	17	-	8	2	-	26		1	27	1	249	
	FREESTANDING	86	-	-	3	0	-	4	1	2	44	8	148	
24	CENTER CITY	136	16	-	4	1	1	13	1	2	18	3	195	
26	TAMPA FESTIVAL CENTRE	115	21	-	10	1	0	24	1	2	25	8	208	
27	GENTILLY S/C	167	16	-	4	3	0	34	1	2	18	6	251	
30	MERIDIAN MARKETPLACE	66	23	-	7	0	-	6	1	2	23	3	130	
31	EASTGATE S/C	85	-	-	4	-	-	4	1	1	50	2	146	
32	FOWLERS PLAZA	-	2	-	4	1	-	8	1	2	24	3	43	3%
33	CHATHAM VILLAGE SQUARE	275	8	-	5	1	0	13	1	2	24	2	331	
36	LINDELL MARKETPLACE	122	8		5	1	0	13	1	2	24	2	178	
37	GENTILLY S/C	115	14		12	2	0	17	1	2	19	-	181	
39	FREESTANDING	79	28	-	4	2	-	8	1	2	41	4	168	
40	BRICKTOWN SQUARE	304	8	-	5	1	0	13	1	2	24	2	360	
41	EAST FOREST PLAZA	193	8	-	5	1	0	13	1	2	24	2	250	
43	LINCOLN S/C	-	29	-	30	1	-	32		1	36	2	131	4%
44	N/A	73	10	-	5	0	-	7	1	2	22	3	122	
	NORTH EASTWOOD PLAZA	-	1	-	4	0	-	3	1	2	21	2	34	5%
46	REDFORD PLAZA	121	13	-	8	-	-	30	1	2	34	2	212	
47	SALEM CONSUMER SQUARE	-	8	-	5	1	0	13	1	2	24	2	56	5%
48	SOUTHGATE SHOPPING CENTER	104	1	-	4	0	-	4	1	2	39	2	157	
49	OVERBROOK PLAZA	122	5	-	1	1	-	3	-	-	31	-	163	
52	VICTORY CROSSING S/C	184	8		5	1	0	13	1	2	24	2	240	
	95TH & JEFFREY	75	23		4	0		24	1	2	15	4	148	
	SUPER 1 FOODS S/C	115	7		4	3	0	5	1	2	32	4	173	
	NORTHWOOD S/C	137	4		4	0		4	1	2	38		190	
	FAMILY DOLLAR BUILDING	46	3		4	0		3	1	4	26	3	89	
	LINWOOD SQUARE S/C	187	1		4	1		9	1	2	68		272	
	MT. CLARE JUNCTION		5		4	0	1	5	1	2	19		37	7%
	THE LANDINGS	190			4	-			1	2	17	3	216	
	N.W. JUNCTION S/C	83	8		6	0		17	1	2	17	2	136	
	WYANDOTTE PLAZA	98			3	0		4	1	2	27	3	138	
	OAKLAND POINTE PLAZA	204	8	_	5	1	0	13	1	2	24	2	260	
	NORTHSIDE PLAZA	93	1	_	4	-	0		1	2	33	3	141	
	MARCUS PLAZA	36			4			12			9	0	61	
	WATERTOWER PLAZA	203	8		5	1	0		1	2	24	2	260	
	GATEWAY MARKETPLACE	203	49	-	5	1			1	1	22	_	282	
	WHITEHALL S/C	220	8		5	1	0	13	1	2	24	2	276	
	ROUNDTREE PLACE	178	8		5	1	0		1	2	24	2	234	
	SOUTHLAND MALL	121	8		4	1	1		-	1	27	-	170	
	K-MART PLAZA	49	8		5	1	0		1	2	24	2	106	
	BURLINGTON PLAZA	82	7		5		1	10	1	2	27	2	136	
	CHICAGO HEIGHTS HILLTOP PLAZA	137	,		4	1		10	1	2	8	3	166	
	NEWPORT S/C	56			3	1		3		1	8	-	71	
	PINE HILLS MARKETPLACE	58	4		4		0		1	2	29	2	102	
	CAHOKIA VILLAGE S/C	66	-		4	1	-	4	1	2	19	2	99	
	THE SHOPS AT WHITE OAK VILLAGE	82	2		4	1	1	3	1	2	16	4	114	
	GREENWOOD WEST	49	3	_	4	0	-	7	1	2	11	-	77	
	CHERRYDALE POINT S/C	203	8		5	1	0	-	1	2	24	2	260	
	NEWMARKET SOUTH S/C	77	6		4	2	-	13	1	2	15	2	122	
	FOOD 4 LESS S/C	253	0		4	2		4	1	1		2	278	
102	SOUTHLAKE MALL	113	-		4	4	-	7	-	2	16 20	3	151	

107 109 110			TAX	SERVICE	INSURANCE - BUSINESS	LICENSE - CITY	STATE & COUNTY	MAINTENANC E	SECURITY	TELEPHONE	UTILITIES	WASTE PICK UP	Total Occupancy	% Rent
109 110	RACEWAY PARK S/C	148	49		3	1		20	1	. 1	21	3	247	
110	POINCIANA PLAZA	-	17	-	7	1	0	13	1	. 2	26	4	70	3%
	HERITAGE PLACE S/C	206	8	-	5	1	0	13	1	. 2	24	2	262	
	POPLAR PLAZA S/C	79	3	-	4	-	-	13	1	. 2	26	2	130	
	LANGLEY SQUARE	100	1	-	4	1	-	3	1	. 2	16	-	128	
	NORTH POINTE S/C	90	4	-	4	1		12	1	. 2	17	2	132	
	SPENCER SQUARE	220	8	-	5	1	0	13	1	. 2	24	2	276	
	VILLAGE S/C		- 8	-	3	1	0	5		1	40	3	54	8%
	LYNNHAVEN NORTH S/C	166 93	8	-	5	1	0	13	1	. 2	24 33	2	222	
	CEDAR HILLS S/C ROEBUCK MARKETPLACE	93	1	-	4	5	0	4	1	. 2	23	0	139 130	
			8	-	5	1	0		1	. 2		2		
	FREESTANDING RIVER COMMONS	165 56	10	-	7	2	0	13 5	1	. 2	24 18	4	222 104	
			10	-	4	2	1	23	1	. 2		4		
	EASTLAND HILL CENTER WEST TOWN PLAZA	99 47	19	-	4	2	1	35	1	. 2	22 25	2	176 119	
	GRAND BOULEVARD PLAZA	149	34	-	4	0	-	28	1	. 2	23	4	244	
	CAPITOL PLAZA	47	0	-	4	2	0	18	1	. 2	33	3	111	
	PARKWAY PLAZA	38	1	-	4	1	0	5	1	. 2	21	3	75	
-	N/A	168	3		4	1	0	3	1	. 2	28	4	213	
	DODGE PLAZA	104	3		4	0	0	2	1	. 2	15	3	135	
	FOOD WORLD PLAZA SOUTH	70	4		4	2	0	10	1	. 2	23	3	118	
	FRAYSER PLAZA	74	7		5	1	1	16	1	. 2	24	3	133	
	IVES DAIRY CROSSING S/C	104	7	-	3	7		9	-	0	15	2	147	
	SHOPS AT NORTHEAST VILLAGE	110	1		4	3		4	1	. 2	32	-	157	
	TOWN & COUNTRY PLAZA	66	10		9	-	0	9	1		26	4	127	
	SHOPPES AT GARNER	188	8		5	1	0	13	1	. 2	24	2	244	
	NORTHLINE CROSSING S/C	267	8	-	5	1	0	13	1	. 2	24	2	324	
155	CANE RUN	90	2	-	4	-	-	2	1	. 2	30	2	133	
	EASTOVER S/C	192	17	-	5	2	0	20	1	. 2	25	5	269	
158	STEELYARD COMMONS S/C	339	81	-	3	-	-	54		1	40	9	529	
159	WESTERN HILLS MALL	98	0	-	4	1	0	3		2	38	1	148	
160	THE SHOPPES OF LIBERTY CITY	152	35	-	16	4	0	18	1	. 2	23	-	252	
161	BLUE PARKWAY TOWN CENTER	104	14	-	5	1	-	26	1	. 2	18	-	170	
162	GRIGGS ROAD S/C	181	17	-	13	2	-	27	1	. 2	24	4	272	
163	MADISON SQUARE	69	7	-	7	2	0	10	1	. 2	34	4	135	
164	SELMA MALL	54	0	-	4	1	0	15	1	. 2	22	4	104	
	FAMILY DOLLAR CENTER	42	4	-	4	0	-	3	1	. 2	28	2	86	
166	LANTANA VILLAGE SQUARE	-	1	-	4	1	0	3	1	. 2	26	5	42	3%
	AZALEA SHOPPING CENTER	-	4	-	5	1	-	13	1	. 2	19	4	48	3%
	TAYLOR COMMONS	192	8	-	5	1	0	13	1	. 2	24	2	248	
	EUTAW VILLAGE	119	1	-	4	0	-	7	1	. 2	21	3	157	
	NORTHSIDE SHOPPING CENTRE	176	20	-	13	0	0	30	1	. 2	29	-	271	
	LAMAR AIRWAYS S/C	103	13	-	6	1	1	23	1	. 2	22	3	174	
	STONY ISLAND PLAZA	219	8	-	5	1	0	13	1	. 2	24	2	276	
	WEST ALLIS	216	8	-	5	1	0	13	1	. 2	24	2	272	
	WEST RIDGE	99	12		6	-		19	1	. 2	25	2	164	
	LAUDERHILL MALL	151	9	-	4	2	0	4	1	. 2	30	6	208	
	SAVE A LOT CTR	88	-	-	4	0	-	-	1	. 2	22	-	117	
	MODEL T PLAZA	179	-		3	2	-	26		1	16	2	230	
	KING SHOPPING CENTER	95	25		4	-	0	32	1	. 2	39	1	199	
	SOUTHGATE USA S/C	121 227	8		5 5	1	0	13	1	. 2	24 24	2	177 283	
	THRUWAY PLAZA		8	-		1	0	13 5	1	. 2		2		
	MIRACLE MILE S/C	237	- 20		2	-	-		1	. 2	15	4	264	
189 190	MARSHFIELD PLAZA SOUTHERN PLAZA	165 196	38 8		5	1	0	37 13	1	. 2	21 24	2	275 252	
				-		_	0		1					
	FONDREN SOUTHWEST VILLAGE VILLAGE GREEN	188 69	14	-	5	2	0	16 4	1	1 2	15 34	4 2	245 119	

Store #	Store Name	RENT	AD VALOREM TAX	FILTER SERVICE	INSURANCE - BUSINESS	LICENSE - CITY	LICENSE- STATE & COUNTY	MAINTENANC E	SECURITY	TELEPHONE	UTILITIES	WASTE PICK UP	Total Occupancy	% Rent
	N/A	86	-	-	4	-		22	1	2	29	3	147	
196	DOLPHIN PLAZA	51	10	-	6	7	1	13	-	1	22	5	117	
	BOSSIER TOWN CENTER	127	4	-	4	2	0	17	1	2	25	4	186	
198	GATEWAY	50	11	-	7	1	1	17	1	2	30	3	120	
199	BEL AIR S/C	71	0	-	4	4	-	4	1	2	23	2	111	
200	MARKETPLACE AT SMYRNA	275	8	-	5	1	0	13	1	2	24	2	331	
201	GALLERY AT WARREN CONNER	178	13	-	4	0	-	21	1	2	32	4	255	
203	ELLIS AVENUE PLAZA	77	5	-	4	0	-	3	1	2	24	3	119	
204	RIVERVIEW PLAZA	56	0	-	4	0	-	3	1	2	25	1	93	
205	CULLEN PLAZA	127	15	-	9	1	-	25	1	2	15	1	197	
207	THE COMMONS ROUTE 20	43	1	-	4	1	-	4	1	2	34	2	91	
208	JEFFERSON VILLAGE	134		-	4	-	-	2	-	-	14		154	
212	ROYAL TOWN CENTER	137	1	-	4	-	-	3	1	2	22	3	172	
213	COLLEGE PARK COMMONS	95	2	-	4	-	-	3	1	2	20	3	129	
215	SOUTH SLAPPEY VILLAGE	78	10	-	7	0	-	9	1	2	29	2	138	
216	WALNUT HILLS PLAZA	107	2	-	4	2	3	6	1	2	22	2	150	
218	SHOPS AT WHITE OAK VILLAGE	244	8	-	5	1	0	13	1	2	24	2	300	
219	COUNTRY CLUB PLAZA	209	8	-	5	1	0	13	1	2	24	2	265	
221	THE CROSSINGS AT HALLS FERRY	133	23	-	5	1	-	21	1	2	14	2	201	
223	CHICAGO	258	8	-	5	1	0	13	1	2	24	2	314	
224	SQUARE 67	68	10	-	7	0	-	18	1	0	17	2	124	
225	GREENS CROSSROADS	93	3	-	4	1	-	4	1	2	19	2	130	
227	PLAZA ON THE BLV	155	20	-	4	1	-	32	1	2	15	-	229	
228	NORTHGATE S/C	289	35	-	2	5	-	25	-	-	5	0	360	
229	JOYLAND S/C	130	1	-	4	2	-	3	1	2	27	3	173	
230	SILVER MILL COURT S/C	80	0	-	4	-	0	24	1	2	41	4	157	
231	MARLOW HEIGHTS S/C	118	2	-	4	-	0	3	1	2	18	5	153	
233	WESTPORT COMMONS S/C	112	-	-	3	1	-	6	-	1	18	-	141	
234	TOWNE SOUTH	-	2	-	4	0	-	4	1	2	27	1	42	3%
236	LAWNDALE PLAZA	211	31	-	4	0	-	54	1	2	19	-	322	
237	BALTIMORE	187	8	-	5	1	0	13	1	2	24	2	243	
239	SOUTH WIND PLAZA	78	5	-	4	1	0	3	1	2	16	-	109	
240	BUCKNER COMMONS	-	15	-	5	0	-	17	1	2	22	-	63	3%
241	TINLEY PARK PLAZA	187	8	-	5	1	0	13	1	2	24	2	243	
242	SOUTH PHILADELPHIA S/C	78	6	-	3	2	-	8	1	2	24	6	129	
246	CITY PLAZA	151	22	-	6	1	1	53	1	2	17	2	255	
248	FREESTANDING	66	-	-	4	0	-	2	1	2	28	4	108	
249	CROSSROADS CENTER	132	8	-	5	1	0	13	1	2	24	2	188	
252	JEWEL OSCO S/C	165	43	-	4	0	-	102	1	2	39	7	363	
254	30TH STREET S/C	110	8	-	5	1	0	13	1	2	24	2	166	
255	MANCHESTER PARKADE	184	8	-	5	1	0	13	1	2	24	2	240	
256	SOUTHFIELD	-	8	-	5	1	0	13	1	2	24	2	56	4%
257	Baseline Plaza	102	7	-	4	0	-	24	1	2	16	-	155	
258	SHOPPES ON MAIN	88	-	-	4	-	-	5	1	2	19	-	118	
259	SOUTH DEKALB PLAZA	88	9	-	4	-	1	21	1	2	24	3	152	
262	MEADOWBROOK EAST	41	15	-	8	0	-	21	1	2	23	3	113	
263	CHARITON SQUARE	-	1	-	4	1	1	8	1	2	13	2	32	3%
264	LAMAR CROSSING	102	27	-	6	1	1	21	1	2	30	3	195	
265	WYOMING VILLAGE	-	0	-	4	0	-	-	1	2	28	-	35	3%
266	STATE STREET	116	7	-	5	0	-	10	1	2	15	2	159	
268	EASTLAND CENTER	82	1	-	4	1	-	3	-	2	40	-	132	
269	ST. STEPHENS SQUARE	82	5	-	6	2	0	11	1	2	25	3	136	
270	HIGHLAND LAKES CENTER	62	14	-	4	-	0	14	1	2	17	-	114	
271	CARRIAGE CROSSING MARKETPLACE	-	8	-	5	1	0	13	1	2	24	2	56	5%
272	ERDMAN PLAZA	110	13	-	6	-	1	19	1	2	23	2	176	
273	NORTHWEST CROSSING	267	8	-	5	1	0	13	1	2	24	2	323	
274	WOODLAND VILLAGE	147	14	-	2	2	-	30	-	-	21	3	219	

Store #	Store Name	RENT	AD VALOREM TAX	FILTER SERVICE	INSURANCE - BUSINESS	LICENSE - CITY	LICENSE- STATE & COUNTY	MAINTENANC E	SECURITY	TELEPHONE	UTILITIES	WASTE PICK UP	Total Occupancy	% Rent
275	CROSS COUNTRY PLAZA	118	13		. 7	2		3	1	. 2	29	-	175	
278	EDENS PLAZA	47	3	-	. 4	1	-	. 3	1	. 2	30	2	92	
281	WHITEHAVEN PLAZA	77	1	-	. 4	1	1	. 3	1	. 2	25	3	117	
282	RACEWAY PARK S/C	204	60	-	. 3	1	-	23	-	0	13	2	306	
283	ALBANY CROSSING	101	5	-	15	0	-	15	1	. 2	18	2	159	
284	OAKLAND POINTE	99	0	-	. 4	1	-	. 4	1	. 2	23	-	134	
285	ONE & OLNEY SQUARE	185	38	-	. 7	0	-	18	1	. 1	32	7	290	
287	NORTHWEST CROSSING	232	8	-	. 5	1	0	13	1	. 2	24	2	289	
288	CLEVELAND AVENUE CROSSING	110	-	-	. 3	5	-	. 2	-	1	29	-	150	
301	CLOVERLEAF S/C	108	4	-	. 4	0	-	. 4	1	. 2	32	3	157	
302	BATESVILLE S/C	66	8	-	. 5	0	-	. 9	1	. 2	16	4	110	
303	NORTHTOWN PLAZA	136	24	-	. 6	0	-	21	1	. 2	17	4	211	
317	HAMMOND AIRE PLAZA	132	9	-	- 11	1	0	31	1	. 2	32	4	223	
327	WALZEM PLAZA	57	5		. 4	0		13	1	. 2	23	3	108	
337	TOWN & COUNTRY	173	12		. 5	1	0			. 1	19	4	231	
338	SOUTH PHILADELPHIA S/C	147	10		. 3	0		. 2		. 1	24	5	192	
343	EASTOWN S/C	38			. 2	-	0			. 1	23	2	73	
344	NORTHERN LIGHTS	151			. 2		0			1	24	-	186	
348	CENTURY CENTER	185	8		. 5	1	-	. 6			14	4	224	
360	FIVE POINTS WEST S/C	159	1		. 4	2	0	-	1	2	27	2	202	
365	CROSS CREEK	69	1		. 4	-	-	13	1	_	17	-	107	
369	FREESTANDING	111	1		. 4	2	0		1		26	4	156	
377	LINCOLN S/C	147	8		10	2	U	10	1	1	20	2	198	
500	ANDERSON ROAD PLAZA	14/	4		. 4			31	1	_	25	3	69	5%
501	METRO PLAZA	110	1		. 4	0		. 2	1	_	32	2	154	370
504	HIGHLAND PARK PLACE	165	0		4	1		. 5	1		32	2	210	
504	LIVONIA MARKETPLACE	103	U	-	. 4	1	-	. 5	1	1	16	-	130	
	EDISTO VILLAGE SHOPPING CENTER		13		. 5	1		. 6	1	_				
507 508		41	8	-	. 5	-	0	-	1	_	20	2	89 270	
	WESTSHORE PLAZA	214		-	. 5	1	U		1	. 2	24			
509	KLINE PLAZA	172 69	43	-		3	-	34			19 3	5	282	
510	CLINTON POINTE		-	-	. 3		-	. 3		1		2	80	
511 512	OAK STREET SHOPPING CENTER	88 136	2	-	. 4	1	0	-	1	. 2	34 13	2	137 153	
	NORGATE PLAZA		-	-		-	-	-		-		U		
513	MLK PLAZA	123	25	-	. 9	2	1		1		18	-	209	
514	NORTH CHARLESTON CENTER	103	9	-	. 7		2		1		24	5	163	
515	SUMTER SQUARE SHOPPING CENTER	85	11	-	. 5	1	-	11	1		17	2	135	
516	COTTMAN & BUSELTON S/C	236	8	-	. 5	1	0		1		24	2	292	
520	BORDEAUX SHOPPING CENTER	87	-	-	. 6	1	1		1	_	26	3	154	
523	WESTLAND SHOPPING CENTER	74	17	-	. 5	-	-	- 11	1	_	18	3	130	
524	THE SHOPS AT STERLING PONDS	146	2		. 3	6	-	. 9		1	11	1	179	
525	NORTHWEST PLAZA	55	-	-	. 8	0	_	3	1		20	2	90	
526	DORCHESTER SQUARE	73	11	-	. 7	-	2		1	_	28	4	145	
527	CANTON CENTRE	227	5	-	. 3	-	0		-	. 0	19	-	267	
528	GEORGETOWN PLAZA SHOPPING CENTER	61	11	-	. 5	-	-	16	1		29	2	127	
529	EASTGATE SHOPPING CENTER	105	8	-	. 6	1	0		1		21	2	158	
531	MIDDLESEX S/C	196	10	-	. 3	2	2		-	. 1	9	5	250	
535	CRICHTON PLAZA	73	1		. 4	2	0		1	_	32	3	137	
537	CHESAPEAKE CROSSING SHOPPING CENTER	90	7	-	. 4	2	-	10	1		23	2	140	
538	CROSSROADS PLAZA	-	1	-	. 4	0	-	. 5	1		31	3	47	4%
539	HATTIESBURG SHOPPING CENTER	104	4	-	. 4	0	-	28	1	. 2	19	2	164	
541	LIBERTY CENTER	100	-	-	. 2	-	-	- 4	-	-	5	0	112	
543	LEHMBERG CROSSING	-	3	-	. 4	-	0	-	1	. 2	15	2	29	4%
544	SHOP CITY PLAZA	104	15		. 4	1	-	23	-	1	12	-	160	
545	GREAT SOUTHERN S/C	104	-		. 3	-	0		-	0	12	-	149	
549	FREEDOM SQUARE SHOPPING CENTER	103	4		. 4	2	-	4	1	. 2	20	2	141	
554	TOWN CENTER MARKET	169	5	-	. 5	-	1	. 11	-	1	18	2	213	
555	IROQUOIS MANOR	62	9	-	. 5	-	-	11	1	. 2	27	2	118	l .

Occupancy Per Diem

Store #	Store Name	RENT	AD VALOREM TAX	FILTER SERVICE	INSURANCE - BUSINESS	LICENSE - CITY	LICENSE- STATE & COUNTY	MAINTENANC E	SECURITY	TELEPHONE	UTILITIES	WASTE PICK UP	Total Occupancy	% Rent
556	WESLEY CHAPEL SQUARE SHOPPING CENTER	108	2	-	4	-	1	39	1	2	26	3	186	
558	SHELBY PLAZA	88	1	-	4	1	1	15	1	2	13	3	128	
559	AUSTINTOWN PLAZA	128	-	-	3	2	1	2	-	1	10	3	148	
562	NORTH POINT CENTER	132	1	-	4	1	1	4	1	2	28	3	176	
563	CONSUMER SQUARE WEST	160	8	-	5	1	0	13	1	2	24	2	216	
565	BROADWAY CENTER	89	1	-	4	1	1	-	1	2	16	2	115	
571	WESTWOOD SHOPPING CENTER	113	19	-	8	2	0	18	1	2	36	3	202	
575	VILLAGE SHOPPING CENTER	-	7	-	5	1	-	17	1	2	35	2	69	8%
576	SOUTHGATE SHOPPING CENTER	129	14	-	10	1	-	29	1	2	21	-	207	
577	HICKORY RIDGE CROSSING	58	1	-	4	1	1	3	1	2	22	3	94	
581	GAINSVILLE SHOPPING CENTER	84	13	-	4	1	-	20	1	2	22	3	150	
584	SOUTHGATE PLAZA	121	7	-	7	0	-	10	1	2	26	2	174	
595	PARKWAY PLAZA	91	17	-	10	0	-	17	1	2	24	2	164	
599	COURTHOUST SQUARE	86	2	-	4	-	0	2	1	2	12	3	111	
617	ASHER	59	3	-	4	1	-	11	1	2	18	2	100	
243	Total	27,705	2,081	-	1,195	210	50	3,121	195	377	5,724	571	41,230	

Note:
For stores indicating percentage rent amounts, these are straight percentage rent locations; there is no base rent associated with these stores and the percentage rent amount is a straight percentage of sales with no minimum threshold or breakpoint.

AGENCY AGREEMENT

This Agency Agreement ("<u>Agreement</u>") is made as of April 17 2015, by and between SIMPLY FASHION STORES, LTD. ("<u>Merchant</u>") and a contractual joint venture composed of Hilco Merchant Resources, LLC and Gordon Brothers Retail Partners, LLC (together, "<u>Agent</u>"), it being understood and agreed that each entity comprising Agent shall be jointly and severally liable for all obligations of Agent hereunder.

Section 1. Recitals.

WHEREAS, Merchant operates retail stores and desires that the Agent act as Merchant's exclusive agent for the limited purposes of: (a) selling all of the Merchandise (as hereinafter defined) from Merchant's two hundred forty-seven (247) retail store locations identified on Exhibit 1 attached hereto (each individually a "Store," and collectively the "Stores") by means of a "store closing", "sale on everything", "going out of business", "everything must go", or similar sale (as further described below, the "Sale"); and (b) disposing of the Owned FF&E in the Stores and Distribution Centers.

WHEREAS, Merchant intends to file voluntary petitions for relief under chapter 11 of title 11 of the United States Code, 11 U.S.C. §§ 101-1330, et seq. (the "Bankruptcy Code") in the United States Bankruptcy Court for the Southern District of Florida (such case, the "Bankruptcy Case" and such court, or other court of competent jurisdiction in which the Bankruptcy Case is filed, the "Bankruptcy Court").

NOW, THEREFORE, in consideration of the mutual covenants and agreements set forth herein, and for other good and valuable consideration, the receipt and sufficiency of which is hereby acknowledged, the Agent and Merchant hereby agrees as follows:

Section 2. Appointment of Agent/Approval Order.

- (a) Subject to entry of an order authorizing Merchant to enter into this Agreement and authorizing Merchant to conduct the Sale in accordance with the terms of this Agreement (the "Approval Order"), Merchant hereby appoints the Agent, and the Agent hereby agrees to serve, as Merchant's exclusive agent for the limited purpose of conducting the Sale in accordance with the terms and conditions of this Agreement.
- (b) The Approval Order shall provide, in a form reasonably satisfactory to Merchant and Agent, *inter alia*, that (i) this Agreement is in the best interest of the Merchant, Merchant's estates, creditors, and other parties in interest, (ii) this Agreement (and each of the transactions contemplated hereby) is approved in its entirety; (iii) Merchant and Agent shall be authorized to continue to take any and all actions as may be necessary or desirable to implement this Agreement and each of the transactions contemplated hereby; (iv) upon payment of the Initial Guaranty Payments (as defined below) and delivery of the Letter of Credit (as defined below), Agent shall be entitled to sell all Merchandise and Owned FF&E hereunder free and clear of all liens, claims or encumbrances thereon, with any presently existing liens encumbering all or any portion of the Merchandise or Owned FF&E or the Proceeds (as defined below) thereof attaching only to the Guaranteed Amounts and other amounts to be received by Merchant under this Agreement; (v) Agent shall have the right to use the Stores and all related Store services, furniture, fixtures, equipment and other assets of the Merchant as designated hereunder for the purpose of conducting the Sale, free of any interference from any entity or person, subject to compliance with the Sale Guidelines (as defined below) and Approval Order; (vi) Agent, as agent for Merchant, is authorized

to conduct, advertise, post signs, utilize signwalkers, and otherwise promote the Sale as a "store closing", "sale on everything", "going out of business", "everything must go", or similar themed sale, in accordance with the Sale Guidelines (as the same may be modified and approved by the Bankruptcy Court) and without further compliance with the Liquidation Sale Laws (as defined below), subject to compliance with the Sale Guidelines and Approval Order; (vii) Agent shall be granted a limited license and right to use until the Sale Termination Date the trade names, logos, e-mail lists, customer lists, ecommerce sites (including (without limitation) websites and social media sites such as Facebook, and Twitter) relating to and used in connection with the operation of the Stores, solely for the purpose of advertising the Sale in accordance with the terms of the Agreement; provided, however, Agent is prohibited from using the websites, social media sites, e-mail lists, customer lists and e-commerce sites owned by or related to New Dots, LLC or the "Dots" brand (collectively, the "Dots Carveout"); (viii) all newspapers and other advertising media in which the Sale is advertised shall be directed to accept the Approval Order as binding and to allow Merchant and Agent to consummate the transactions provided for in this Agreement, including, without limitation, the conducting and advertising of the Sale in the manner contemplated by this Agreement; (ix) all utilities, landlords, creditors and other interested parties and all persons acting for or on their behalf shall not interfere with or otherwise impede the conduct of the Sale, or institute any action in any court (other than in the Bankruptcy Court) with respect to Merchandise or the Owned FF&E or before any administrative body which in any way directly or indirectly interferes with or obstructs or impedes the conduct of the Sale; (x) the Bankruptcy Court shall retain jurisdiction over the parties to enforce this Agreement; (xi) Agent shall not be liable for any claims against Merchant other than as expressly provided for in this Agreement; (xii) Agent shall be granted a valid, binding, enforceable and perfected security interest as provided for in Section 16 hereof (without the necessity of filing financing statements to perfect the security interests); (xiii) the Bankruptcy Court finds that time is of the essence in effectuating this Agreement and proceeding with the Sale at the Stores uninterrupted; (xiv) the Bankruptcy Court finds that the Merchant's decisions to (a) enter into this Agreement and (b) perform under and make payments required by this Agreement is a reasonable exercise of the Merchant's sound business judgment consistent with its fiduciary duties and is in the best interests of the Merchant, its estate, its creditors, and other parties in interest; (xv) the Bankruptcy Court finds that this Agreement was negotiated in good faith and at arms' length between the Merchant and Agent and that Agent is entitled to the protection of section 363(m) of the Bankruptcy Code; (xvi) the Bankruptcy Court finds that Agent's performance under this Agreement will be, and payment of the Guaranteed Amounts under this Agreement will be made, in good faith and for valid business purposes and uses, as a consequence of which Agent is entitled to the protection and benefits of sections 363(m) and 364(e) of the Bankruptcy Code; (xvii) this Agreement is approved pursuant to Bankruptcy Code section 363; and (xviii) in the event any of the provisions of the Approval Order are modified, amended or vacated by a subsequent order of the Bankruptcy Court or any other court, Agent shall be entitled to the protections provided in Bankruptcy Code sections 363(m) and 364(e) and, no such appeal, modification, amendment or vacatur shall affect the validity and enforceability of the sale or the liens or priority authorized or created under this Agreement or the Approval Order.

Sale as a "store closing," "sale on everything", "going out of business", "everything must go", or similar-themed sale, and the Approval Order shall provide that Agent shall be required to comply with applicable federal, state and local laws, regulations and ordinances, including, without limitation, all laws and regulations relating to advertising, privacy, consumer protection, occupational health and safety and the environment, together with all applicable statutes, rules, regulations and orders of, and applicable restrictions imposed by, governmental authorities (collectively, the "Applicable General Laws"), other than all applicable laws, rules and regulations in respect of "going out of business", "store closing" or similar-themed sales and permitting (collectively, the "Liquidation Sale Laws"), provided that such Sale is

conducted in accordance with the terms of this Agreement, the Sale Guidelines and Approval Order; and <u>provided</u> further that the Approval Order shall provide that so long as the Sale is conducted in accordance with the Sale Guidelines and the Approval Order and in a safe and professional manner, Agent shall be deemed to be in compliance with any Applicable General Laws.

Section 3. Consideration to Merchant and Agent.

3.1 Payments to Merchant.

- Subject to Merchant's compliance with all representations, warranties, covenants, (a) terms and conditions set forth in this Agreement, as a guaranty of Agent's performance hereunder, Agent guarantees that Merchant shall receive twenty-seven and one-half percent (27.5%) (the "Guaranty Percentage") of the aggregate Retail Price of the Merchandise (the "Guaranteed Amount"), which Guaranteed Amount shall be paid at such times and in such manner as shall hereinafter be provided. The Agent shall pay to Merchant the Guaranteed Amount and the Sharing Amount due to Merchant (if any) in the manner and at the times specified in Section 3.3. The Guaranteed Amount will be calculated based upon the aggregate Retail Price of the Merchandise as determined by multiplying (x) the Guaranty Percentage by (y)(A) the final certified report of the Inventory Taking Service after verification and reconciliation thereof by Agent and Merchant showing the aggregate Retail Price of the Merchandise as of the Sale Commencement Date, in consultation with Lender, (B) the aggregate Retail Price of the Merchandise subject to Gross Rings (as adjusted for shrinkage per this Agreement), (C) the aggregate Retail Price of Returned Merchandise not otherwise included in the Inventory Taking, and (D) any other adjustments to Retail Price as expressly contemplated by this Agreement. Agent shall pay to Merchant (or its designee) the Guaranteed Amount in the manner and at the times specified in Section 3.3 below.
- (b) The Guaranty Percentage has been fixed based upon the aggregate Retail Price of the Merchandise included in the Sale being not less than \$16,000,000 (the "Merchandise Threshold") as of the Sale Commencement Date. To the extent that the aggregate Retail Price of the Merchandise included in the Sale is less than the Merchandise Threshold, then such deviation shall not constitute a breach of any representation or warranty, or an Event of Default; provided, however, that, the Guaranty Percentage shall be adjusted in accordance with Exhibit 3.1(b) annexed hereto, which adjustment in the Guaranty Percentage shall result in an dollar adjustment to the Guaranteed Amount the actual amount of which will be determined in connection with the Final Inventory Report; provided that, solely for purposes of determining whether the aggregate Retail Price of the Merchandise included in the Sale is less than the Merchandise Threshold, no adjustment shall be made to the applicable Retail Price to account for the effect of any prevailing discount adjustment or the following discounts or price adjustments offered by the Merchant: (i) employee discounts; (ii) member or customer appreciation points or coupons; (iii) coupons (Merchant's or competitors'); (iv) customer savings pass discounts or "bounce back" coupons, or discounts for future purchases based on dollar value of past purchases; (v) obvious ticketing or marking errors; (vi) instant (in-store) or mail in rebates.
- (c) To ensure accurate sales audit functions, as well as accurate calculations of the Sharing Amount, if any, Agent shall use Merchant's existing point-of-sale system for recording all sales (including any sales of Additional Agent Goods) in the Stores.

3.2 Compensation to Agent, Sharing:

(a) After Proceeds are used to repay Agent for amounts paid on account of the Guaranteed Amount and to pay Expenses, all remaining Proceeds shall be allocated in the following order

of priority: FIRST: to Agent in an amount equal to two percent (2.0%) of the aggregate Retail Price of the Merchandise ("<u>Agent's Fee</u>"); and THEREAFTER: fifty percent (50%) to Agent and fifty percent (50%) to Merchant ("<u>Sharing Amounts</u>"). Agent shall pay the Sharing Amount, if any, to Merchant (or its designee), on the first business day after the completion of the Final Reconciliation conducted pursuant to Section 8.7 and shall remit such payment to the Merchant's Designated Account.

(the "Remaining Merchandise"), subject to Merchand's right to participate in the Proceeds from the disposition thereof, such Remaining Merchandise shall be deemed transferred to Agent free and clear of all liens, claims, and encumbrances to the extent provided in the Approval Order, and Agent shall use diligent and commercially reasonable efforts to dispose of all such Remaining Merchandise by means of bulk sale/wholesale or otherwise. Agent and its affiliates shall be authorized to sell or otherwise dispose of the Remaining Merchandise with all logos, brand names, and other intellectual property on the Merchandise intact (subject to the Dots Carveout), and shall be authorized to advertise the sale of the Remaining Merchandise using Merchant's name and logo. The proceeds received by Agent from such disposition shall constitute Proceeds hereunder.

3.3 Proceeds; Time of Payments; Control of Proceeds.

- (a) For purposes of this Agreement, "Proceeds" shall mean the aggregate of (a) the total amount (in dollars) of all sales of Merchandise made under this Agreement and all service revenue, in each case during the Sale Term and exclusive of Sales Taxes; (b) all proceeds of Merchant's insurance for loss or damage to Merchandise arising from events occurring during the Sale Term; and (c) any and all proceeds received by Agent from the disposition of Remaining Merchandise. For the avoidance of doubt: (1) proceeds from the sales at Merchant's Stores or through the e-commerce platform for periods prior to the Sale Commencement Date; (2) the proceeds from the sale of Merchant Consignment Goods pursuant to Section 5.4 hereof (subject to Agent's right to receive the commission under Section 5.4 below); (3) all proceeds of Merchant's insurance for loss or damage to goods arising from events occurring prior to the Sale Commencement Date; (4) proceeds from the sale or other disposition of Owned FF&E (subject to Agent's right to receive the FF&E Commission under Section 7.1 below); (5) proceeds from the sale of Additional Agent Goods; and (6) payments made by Agent on account of the Guaranteed Amount, the Sharing Amounts (if any), Expenses, or the Letter of Credit, shall, in each case, not constitute "Proceeds" hereunder.
- (b) On the first business day following the entry of the Approval Order (the "Payment Date"), Agent shall pay to Merchant an amount equal to eighty-five percent (85%) of the estimated Guaranteed Amount of the Retail Price of Merchandise in the Stores as of the Sale Commencement Date (based upon Merchant's books and records maintained in the ordinary course as of the date immediately preceding the Payment Date taking into account estimated shrink, but in no event shall such estimate of the Retail Price of Merchandise be less than the Merchandise Threshold) (the "Initial Guaranty Payment") by wire transfer of immediately available funds to an account designated in writing by Merchant on Exhibit 3.3(b) attached hereto ("Merchant's Account"). The balance of the Guaranteed Amount, if any, shall be paid by Agent by wire transfer of immediately available funds to the Merchant's Account on the earlier of (i) the date that is thirty (30) days after the Sale Commencement Date and (ii) the second business day following the issuance of the final report of the aggregate Retail Price of the Merchandise included in the Sale by the Inventory Taking Service, after review, reconciliation and mutual written verification thereof by Agent and Merchant, in consultation with the Lender (the "Final Inventory Report" with the date of completion of such reconciliation and issuance of such Final Inventory Report to be referred to as the "Inventory Reconciliation Date") (and in the case of

"(i)" herein, Agent shall tender payment of only the undisputed portion on account of any remaining portion of the Guaranteed Amount). To the extent that Merchant is entitled to receive any funds on account of the Sharing Amount due to Merchant, Agent shall pay such Sharing Amount as part of the Final Reconciliation under Section 8.7. To the extent that the Guaranteed Amount has not been paid in full by the date of the Final Reconciliation, Agent shall pay the unpaid portion of the Guaranteed Amount to Merchant as part of the Final Reconciliation. In the event of a dispute as to the calculation of the portion of the Guaranteed Amount, any such dispute shall be resolved in the manner and at the times set forth in Section 8.7 hereof. To the extent the Agent fails to pay when due the balance or an undisputed portion, if any, of any remaining portion of the Guaranteed Amount in accordance with this section 3.3(b), Merchant and/or Lender shall have the rights granted under Section 3.4 hereof to the extent of such balance or undisputed portion. Merchant and Agent shall exercise commercially reasonable best efforts to reconcile the Inventory Taking within ten (10) days after its completion by the Inventory Taking Service.

- (c) All Proceeds shall be controlled by Agent in the manner provided for below.
- Agent may (but shall not be required to) establish its own accounts (i) (including without limitation credit card accounts and systems) with five (5) business days advance notice to Merchant, dedicated solely for the deposit of the Proceeds and the disbursement of amounts payable to Agent hereunder (the "Agency Accounts"), and Merchant shall use commercially reasonable procedures, upon Agent's reasonable request, execute and deliver all necessary documents to open and maintain the Agency Accounts; provided, however, Agent shall have the right, in its sole and absolute discretion, to continue to use Merchant's Designated Deposit Accounts (as defined below) as the Agency Accounts in which case Merchant's Designated Deposit Accounts shall be deemed to be Agency Accounts. Agent shall exercise sole signatory authority and control with respect to the Agency Accounts. The Agency Accounts shall be dedicated solely to the deposit of Proceeds and other amounts contemplated by this Agreement and the distribution of amounts payable hereunder. Agent shall deliver to Merchant, not later than three (3) business days following receipt thereof, copies of all bank statements and other information relating to such accounts. The Merchant shall not be responsible for, and Agent shall pay as an Expense hereunder, all bank fees and charges, including wire transfer charges, related to the Sale and the Agency Accounts, whether received during or after the Sale Term. Upon Agent's notice to Merchant of Agent's designation of the Agency Accounts (other than Merchant's Designated Deposit Accounts), all Proceeds of the Sale (including credit card Proceeds) shall be deposited into the Agency Accounts.
- (ii) Agent shall have the right to use Merchant's credit card facilities, including Merchant's credit card terminals and processor(s), credit card processor coding, Merchant's identification number(s) and existing bank accounts for credit card transactions relating solely to the Sale. In the event that Agent elects to use Merchant's credit card facilities, Merchant shall process credit card transactions on behalf of Agent and for Agent's account, applying customary practices and procedures. Without limiting the foregoing, Merchant shall cooperate with Agent to download data from all credit card terminals each day during the Sale Term to effect settlement with Merchant's credit card processor(s), and shall take such other actions necessary to process credit card transactions on behalf of Agent under Merchant's identification number(s). At Agent's request, Merchant shall cooperate with Agent to establish Merchant's identification numbers under Agent's name to enable Agent to process all such credit card Proceeds for Agent's account. Merchant shall not be responsible for, and Agent shall pay as an Expense hereunder, all credit card fees, charges, and chargebacks related to the Sale, whether received during or after the Sale Term. Agent shall not be responsible for, as an Expense or otherwise, any credit card fees, charges, or chargebacks that do not relate to the Sale, whether received prior to, during or after the Sale Term.

- (iii) Unless and until Agent establishes its own Agency Accounts (other than Merchant's Designated Deposit Accounts), all Proceeds and other amounts contemplated by this Agreement (including credit card Proceeds), shall be collected by Merchant and deposited on a daily basis into depository accounts designated by, and owned and in the name of, Merchant for the Stores, which accounts shall be designated solely for the deposit of Proceeds and other amounts contemplated by this Agreement (including credit card Proceeds), and the disbursement of amounts payable to or by Agent hereunder (the "Designated Deposit Accounts"). The Designated Deposit Accounts shall be cash collateral accounts, with all cash, credit card payments, checks and similar items of payment, deposits and any other amounts in such accounts being Proceeds or other amounts contemplated hereunder, and Merchant hereby grants to Agent a first priority senior security interest in each Designated Deposit Account and all proceeds (including Proceeds) in such accounts from and after the Sale Commencement Date. If requested by Agent, each account shall be subject to an agreement between and among the Agent, the Merchant and the subject bank (in form and content reasonably satisfactory to each party), providing for, among other things, that such bank will comply with instructions originated by Agent directing the disposition of funds in such account without further consent of the Merchant (a "Control Agreement"). If, notwithstanding the provisions of this Section 3.3(c), Merchant or Lender receives or otherwise has dominion over or control of any Proceeds or other amounts due to Agent, Merchant and Lender shall be deemed to hold such Proceeds and other amounts "in trust" for Agent and shall not commingle Proceeds or other amounts due to Agent with any of Merchant's or Lender's other funds or deposit such Proceeds or other amounts in any account except a Designated Deposit Account or as otherwise instructed by Agent.
- (iv) On each business day, Merchant shall promptly pay to Agent by wire funds transfer all funds in the Designated Deposit Accounts (including, without limitation, Proceeds, Proceeds from credit card sales, and all other amounts) deposited into the Designated Deposit Accounts for the prior day(s) without any offset or netting of Expenses or other amounts that may be due to Merchant. Agent shall notify Merchant and JNS INVT, LLC ("Lender") of any shortfall in such payment, in which case, Merchant (or Lender, as the case may be) shall promptly pay to Agent funds in the amount of such shortfall.
- (d) In the event that the Agent funds or pays all or any portion of Merchant's obligations under this Agreement, such funding or payment cannot be recovered by the Agent under section 3.3(e) by means of an offset, and, as a result of such funding or payment, Merchant (or Lender) received more value than Merchant (or Lender) would have otherwise received under this Agreement had Agent not funded or paid such obligations, Lender shall pay all such funded or paid amounts to Agent within two (2) business days of Agent's request. If and to the extent the Agent over-funds any amounts in respect of the Guaranteed Amount hereunder and Merchant for any reason fails to refund Agent such overfunded amount after two (2) business days written demand by Agent, Lender shall, within two (2) business days of written demand by Agent, pay to the Agent the over-funded amount. Notwithstanding anything to the contrary in this Agreement, the Lender shall be required to pay to Agent any amount. in excess of the amounts actually received by such Lender pursuant to this Agreement, unless the Lender is the proximate cause of an Event of Default such as the filing of a motion for appointment of a chapter 11 trustee or conversion of the Debtors' cases to cases under chapter 7 of the Bankruptcy Code or declares or an "event of default" under any debtor-in-possession financing or use of cash collateral in which case the foregoing limitation shall not be applicable.
- (e) Merchant and Agent further agree that if at any time during the Sale Term, (i) Agent holds any amounts due to Merchant under this Agreement, Agent may, in its discretion, after five (5) business days' notice to Merchant, offset such amounts being held by Agent against any undisputed

amounts due and owing by, or required to be paid by, Merchant hereunder, and (ii) Merchant holds any amounts due to Agent under this Agreement, Merchant may, in its discretion, after five (5) business days' notice to Agent, offset such amounts being held by Merchant against any undisputed amounts due and owing by, or required to be paid by, Agent hereunder.

- (f) In addition to the Guaranteed Amount, in connection with the first Weekly Sale Reconciliation, Agent shall purchase and pay Merchant by wire transfer to the account designated in writing by Merchant all cash in the Stores on and as of the start of business on the Sale Commencement Date on a dollar for dollar basis.
- Security. In order to secure Agent's obligations under this Agreement to pay the balance of any unpaid portion of the Guaranteed Amount, Sharing Amount (if any), Expenses and other amounts due Merchant hereunder, no later than the Payment Date, Agent shall deliver to JNS INVT, LLC (as defined herein, the Lender), as Merchant's designee, an irrevocable standby Letter of Credit, substantially in the form of Exhibit 3.4 attached hereto, naming Lender and Merchant as co-beneficiaries (each, a "Beneficiary") in the aggregate original face amount equal to the sum of: (a) fifteen percent (15%) of the estimated Guaranteed Amount (based upon Merchant's books and records maintained in the ordinary course as of the date immediately preceding the Payment Date), plus (b) the parties' mutually agreed upon estimate of three weeks of Expenses, which shall be substantially in the form of Exhibit 3.4 hereof (the "Letter of Credit"). The Letter of Credit shall be delivered to JNS INVT, LLC (as defined herein, the Lender), as Merchant's designee, no later than the second business day following the Payment Date, and shall be issued by a U.S. national bank selected by Agent and reasonable acceptable to Merchant and the Lender, which shall include Bank of America, N.A. and its affiliates. In the event that Agent fails to timely pay any undisputed amount hereunder in respect of the Guaranteed Amount, Sharing Amount (if any), Additional Agent Goods Fee (if any) and/or Expenses as required under this Agreement, Merchant and the Lender shall be entitled to draw on the Letter of Credit to fund such undisputed amount or obligation after five (5) business days' written notice to Agent, provided Agent does not pay such undisputed amount within such five-day period. The Letter of Credit shall have an expiry date of no earlier than sixty (60) days after the Sale Termination Date. Upon Lender's receipt of payment in full of its claims against the Merchant, Lender shall promptly deliver the Letter of Credit to Merchant and take all steps necessary to remove itself as a named co-beneficiary thereunder. Unless the parties shall have mutually agreed that they have completed the Final Reconciliation under this Agreement, then, at least ten (10) business days prior to the initial or any subsequent expiry date, Agent shall cause the Beneficiaries to receive an amendment to the Letter of Credit solely extending (or further extending, as the case may be) the expiry date by at least thirty (30) days. If Agent fails to provide the Beneficiaries such amendment to the Letter of Credit no later than ten (10) business days before the expiry date, then the Beneficiaries shall be permitted to draw the full amount under the Letter of Credit to hold as security for amounts that may become due and payable to Merchant. In the event that Agent, after receipt of five (5) business days' written notice, fails to pay any undisputed portion of the Guaranteed Amount or Expenses, the Beneficiaries may draw on the Letter of Credit in an amount equal to the unpaid, past due, amount of the Guaranteed Amount or Expenses that is not the subject of a reasonable dispute. Lender, Merchant and Agent agree that, from time to time upon Agent's request, the face amount of the Letter of Credit shall be reduced by the aggregate amount of payments made by Agent on account of the Guaranteed Amount to the time of each such request (and Merchant and Lender shall cooperate with respect to each such request); provided, however, in no event shall the face amount of the Letter of Credit be reduced to an amount less than the parties' mutually agreed upon estimate of three weeks of estimated Expenses until the Final Reconciliation.

Section 4. Expenses of the Sale.

- 4.1 <u>Expenses</u>. Agent shall be unconditionally responsible for all "Expenses," which expenses shall be paid by Agent in accordance with Section 4.2 below. Agent and/or Merchant and/or Lender may review or audit the Expenses at any reasonable time. As used herein, "Expenses" shall mean the Store-level operating expenses of the Sale which arise during the Sale Term and are attributable to the Sale, limited to the following:
- (a) actual payroll (including wages and commissions (if applicable)) with respect to all Retained Employees used in connection with conducting the Sale for actual days/hours worked at a Store during the Sale Term (excluding hours worked during the Inventory Taking), as well as payroll for any temporary labor engaged for the Sale provided, that Agent shall only be obligated to pay 50% of the payroll wages for Store-level Retained Employees used during the Inventory Taking, and Merchant shall pay the remaining 50% of the payroll wages for Store-level Retained Employees used during the Inventory Taking;
- (b) any amounts payable by Merchant for benefits for Retained Employees (including FICA, unemployment taxes, workers' compensation and healthcare insurance, but excluding Excluded Payroll Benefits) for Retained Employees used in the Sale, in an amount not to exceed 18.0% of the base payroll in the aggregate for the Retained Employees in the Stores (the "Payroll Benefits Cap");
- the actual Occupancy Expenses for the Stores on a per location and per diem basis in an amount up to the aggregate per diem per location amount set forth on Exhibit 4.1(c), including but not limited to, (i) the portion of any percentage rent obligations attributable to the sale of Merchandise during the Sale Term, plus (ii) the portion of any percentage rent obligations attributable to the sale of Additional Agent Goods during the Sale Term (in the case of 4.1(c)(i) and (ii), as determined in the manner described in the definition of "Occupancy Expenses" below in this Section 4.1); provided, however, that in respect of section 4.1(c)(i) and (ii), percentage rent shall only be an "Expense" if and to the extent such percentage rent obligation for a Store is reflected on Exhibit 4.1(c) (through the designation of a specific percentage (e.g., 5%)), and, if no percentage rent obligation is reflected on Exhibit 4.1(c) for each a Store, Agent shall not be liable for any percentage rent associated with each such Store and percentage rent shall not be an "Expense" for each such Store; in all cases Occupancy Expenses shall be limited on a per Store, per category, and per diem basis not to exceed the respective categories and amounts shown on Exhibit 4.1(c);
- (d) Retention Bonuses for Retained Employees, as provided for in Section 9.4 below;
- (e) advertising and direct mailings relating to the Sale, Store interior and exterior signage and banners, and signwalkers, in each case relating to the Sale;
- (f) credit card fees, bank card fees, and chargebacks and credit/bank card discounts with respect to Merchandise sold in the Sale;
- (g) bank service charges (for Store, corporate accounts, and Agency Accounts), check guarantee fees, and bad check expenses to the extent attributable to the Sale;
- (h) costs for additional Supplies at the Stores necessary to conduct the Sale as requested by Agent;

- (i) all fees and charges required to comply with applicable laws in connection with the Sale as agreed to by Agent;
 - (j) Store cash theft and other store cash shortfalls in the registers;
- (k) all costs and expenses associated with Agent's on-site supervision of the Stores and Distribution Centers, including (but not limited to) any and all fees, wages, taxes, third party payroll costs and expenses, and deferred compensation of Agent's field personnel, travel to, from or between the Stores and Distribution Centers, and out-of-pocket and commercially reasonable expenses relating thereto (including reasonable and documented corporate travel to monitor and manage the Sale);
- (l) postage, courier and overnight mail charges requested by Agent to the extent relating to the Sale;
- (m) fifty percent (50%) of cost of the Inventory Taking Service; provided, however, if the cost of the Inventory Taking Service exceeds \$180,000, Agent shall pay 100% of the cost of the Inventory Taking Service in excess of \$180,000 and such cost shall be considered Expenses;
- (n) Agent's actual cost of capital (including Letter of Credit fees) and insurance attributable to the Sale, and Merchant's insurance under Section 12 prorated to include amounts attributable to the Sale Term:
- (o) Agent's reasonable out-of-pocket costs and expenses associated with this Agreement, the Sale, or the transactions contemplated by this Agreement, including, but not limited to, legal fees and expenses incurred in connection with the review of data, preparation, negotiation, and execution of this Agreement and any ancillary documents, and the Sale;
 - (p) third party payroll processing expenses associated with the Sale;
- (q) costs of transfers initiated by Agent of Merchandise between and among the Stores during the Sale Term, including delivery and freight costs, it being understood that Agent shall be responsible for coordinating such transfer of Merchandise, subject, however, to the provisions of section 8.1(e); and
- (r) \$7,500 per week, pro rated for partial weeks, on account of Central Service Expenses;
- (s) Subject to Merchant's written consent, which shall not be unreasonably withheld or delayed, the documented actual costs and expenses of Agent providing such additional services as the Agent reasonably deems appropriate for the Sale.

Notwithstanding anything herein to the contrary, to the extent that any Expense category listed in section 4.1 is also included on Exhibit 4.1(c), Exhibit 4.1(c) shall control and such Expenses shall not be double counted. There will be no double counting or payment of Expenses to the extent that Expenses appear or are contained in more than one Expense category.

As used herein, the following terms have the following respective meanings:

- (i) "<u>Central Service Expenses</u>" means costs and expenses for Merchant's central administrative services necessary for the Sale, including, but not limited to, internal payroll processing, MIS services, cash and inventory reconciliation, data processing and reporting, email preparation and distribution, information technology and e-commerce site updates and maintenance, and accounting (collectively, "<u>Central Services</u>").
- (ii) "Excluded Payroll Benefits" means (i) the following benefits arising, accruing or attributable to the period prior to, during, or after the Sale Term: (w) vacation days or vacation pay, (x) sick days or sick leave or any other form of paid time off, (y) maternity leave or other leaves of absence and (z) ERISA coverage and similar contributions and/or (ii) any other benefits in excess of the Payroll Benefits Cap, including, without limitation, any payments due under the WARN Act.
- (iii) "Occupancy Expenses" means, with respect to the Stores, base rent, percentage rent, HVAC, utilities, CAM (common area maintenance), landlord promotional fees, storage costs, real estate and use taxes, Merchant's association dues and expenses, utilities expenses, cash register maintenance (including point-of-sale systems maintenance), routine repairs and maintenance, building maintenance, trash and snow removal, housekeeping and cleaning expenses, pest control services, telecom/telephone charges (including local and long-distance telephone) and internet/wifi expenses (including broadband internet), satellite broadband connections, T-1 lines, security (including, without limitation, security systems, courier and guard service, building alarm service and alarm service maintenance), taxes and licenses, rental for furniture, fixtures and equipment, and all other categories of expenses at the Stores as set forth on Exhibit 4.1(c) attached hereto and in an amount up to the specific amounts set forth on Exhibit 4.1(c) attached hereto and calculated in accordance with Section 4.1(c), plus any percentage rent obligations incurred by Merchant under applicable leases or occupancy agreements that are allocable to the sales as part of the Sale during the Sale Term of: (x) Merchandise and (y) Additional Agent Goods included in the Sale.
- (iv) "<u>Third Party</u>" means, with reference to any Expenses to be paid to a Third Party, a party which is not affiliated with or related to the Merchant.
- (v) Notwithstanding any other provision of this Agreement to the contrary, "Expenses" shall not include: (i) Excluded Payroll Benefits; (ii) Central Service Expenses, (iii) Occupancy Expenses or any occupancy-related expenses of any kind or nature in excess of the respective per Store, per diem occupancy-related categories and amounts expressly provided for as an Expense under Section 4.1(c) above to the extent actually incurred; (iv) any expenses of any kind relating to or arising from Merchant's home office or Distribution Centers, including (without limitation) the Distribution Center Expenses (except as set forth in Section 4.3), and/or (v) any other costs, expenses or liabilities payable by Merchant not provided for herein, all of which shall be paid solely by Merchant promptly when due, subject to the provisions of the Bankruptcy Code and the Approval Order.

4.2 Payment of Expenses.

Agent shall be responsible for the payment of all Expenses out of Proceeds (or from Agent's own accounts if and to the extent there are insufficient Proceeds) after the payment of the Guaranteed Amount. All Expenses incurred during each week of the Sale (*i.e.* Sunday through Saturday) shall be paid by Agent to or on behalf of Merchant, or paid by Merchant and thereafter reimbursed by Agent as provided for herein, immediately following the Weekly Sale Reconciliation; provided, however, in the event that the actual amount of an Expense is unavailable on the date of the reconciliation (such as

payroll), Merchant and Agent shall agree to an estimate of such amounts, which amounts will be reconciled once the actual amount of such Expense becomes available. Agent and/or Merchant may review or audit the Expenses at any time.

4.3 Distribution Center Expenses

From and after the Sale Commencement Date, Agent shall be responsible for allocating and designating the shipment of Merchandise, including Agent's Additional Agent Goods, from the Merchant's Distribution Centers to the Stores. From and after the date on which this Agreement is executed and pending Bankruptcy Court approval hereof, Merchant and Agent shall cooperate with each other and shall mutually agree upon a schedule and allocation to the Stores of the Merchandise located at the Distribution Centers. In any event, Agent shall provide Merchant with the schedule and allocation of Merchandise to the Stores no later than the Sale Commencement Date. Notwithstanding the foregoing and except as otherwise provided herein, all costs and expenses of operating the Distribution Centers, including, but not limited to, use and occupancy expenses, Distribution Center employee payroll and other obligations, and/or processing, transferring, consolidating, shipping, and/or delivering goods within or from the Distribution Centers (the "Distribution Center Expenses"), shall remain the sole obligation of the Merchant. Agent shall have no right to use the Distribution Center for Additional Agent Goods unless otherwise agreed to by Merchant and Agent. Notwithstanding anything contained in this Section 4.3 to the contrary, from and after the date on which there is no Merchandise remaining in the Distribution Center, the Merchant may terminate and discontinue use of the Distribution Centers in its discretion and without penalty hereunder.

Section 5. Gross Rings; Merchandise.

5.1 <u>Inventory Taking.</u>

Commencing on the Sale Commencement Date, Merchant and Agent shall cause to be taken a STYLE level and Retail Price physical inventory of the Merchandise located in the Stores (collectively, the "Inventory Taking"), which Inventory Taking shall be completed in each of the Stores as soon as practicable (the date of the Inventory Taking at each Store being the "Inventory Date" for each such Store), but in any event no later than twenty-one (21) days after the Sale Commencement Date (subject to the availability of the Inventory Taking Service), and, with respect to Merchandise located in Merchant's Distribution Centers, upon transfer to the Stores. Merchant and Agent shall jointly employ RGIS or other mutually agreed upon national inventory taking service (the "Inventory Taking Service") to conduct the Inventory Taking. The Inventory Taking shall be conducted in accordance with the procedures and instructions set forth on Exhibit 5.1(a) (the "Inventory Taking Instructions"). As an Expense, Agent shall be responsible for fifty percent (50%) of cost of the Inventory Taking Service. Merchant shall be responsible for fifty percent (50%) of cost of the Inventory Taking Service (capped at \$90,000). Except as provided in the immediately preceding two sentences, Merchant and Agent shall each bear their respective costs and expenses relative to the Inventory Taking. Merchant, Lender, and Agent may each have representatives present during the Inventory Taking, and shall each have the right to review and verify the listing and tabulation of the Inventory Taking Service. Merchant agrees that during the conduct of the Inventory Taking (which the Merchant and Agent anticipate should not reasonably take more than one (1) day at any Store), the applicable Store shall be closed to the public, and no sales or other transactions shall be conducted within the applicable Store. Merchant and Agent further agree that until the Inventory Taking in a particular Store is completed, neither the Merchant nor Agent shall: (i) move Merchandise within or about the Store so as to make any such items unavailable for counting as part of the Inventory Taking, provided, that, for the avoidance of doubt, the foregoing shall not restrict the Agent's ability to transfer Merchandise between Stores subject to the restriction set forth in section 8.1(e); or (ii) remove or add any hang tags, price tickets, inventory control tags affixed to any Merchandise or any other kind of in-store pricing signage within the Store. Merchant agrees to cooperate with Agent to conduct the Inventory Taking (including without limitation by making available to Agent information relating to sales, units, costs and Retail Price, and making available to Agent Merchant's books, records, work papers and personnel to the extent reasonably necessary to calculate the Retail Price of the Merchandise). Each Store will be closed during the Inventory Taking; provided, however, that the parties agree that the Inventory Taking will commence at a time that will minimize the number of hours that the Stores will be closed for business. The Inventory Taking, including, but not limited to the Final Inventory Report, shall be reviewed, reconciled, and mutually verified by the Merchant and Agent in writing as soon as practicable following the Inventory Taking.

(b) At each Store, for the period from the Sale Commencement Date until the Inventory Date for such Store, Agent and Merchant shall jointly keep (i) a strict count of gross register receipts less applicable Sales Taxes but excluding any prevailing discounts ("Gross Rings"), and (ii) cash reports of sales within such Store. Register receipts shall show for each item sold the Retail Price for such item and the markdown or discount, if any, specifically granted by Agent in connection with such Sale. Agent shall pay that portion of the Guaranteed Amount calculated on the Gross Rings basis, to account for shrinkage, on the basis of 101.0% of the aggregate Retail Price of Merchandise sold during the Gross Rings Period. All such records and reports shall be made available to Agent and Merchant during regular business hours upon reasonable notice. Any Merchandise included in the Sale using the Gross Rings method shall be included as Merchandise.

5.2 <u>Merchandise Subject to This Agreement.</u>

- For purposes of this Agreement, "Merchandise" shall mean all (i) new, finished, (a) first-quality saleable goods in the ordinary course of business located at the Stores as of the Sale Commencement Date (including Merchandise subject to Gross Rings), (ii) Defective Merchandise, and (iii) Distribution Center Merchandise received at the Stores no later than seven (7) days after the Sale Commencement Date, provided that if such goods are received at the Stores after such seven-day period, but on or before fourteen (14) days after the Sale Commencement Date (the "Receipt Deadline"), such goods shall be included in the Sale as Merchandise at the Retail Price of each good multiplied by the inverse of the then prevailing Sale discount for each such good. "Merchandise" shall not include: (1) goods which belong to sublessees, licensees, department lessees, or concessionaires of Merchant; (2) goods held by Merchant on memo, on consignment, or as bailee; (3) Excluded Defective Merchandise; (4) Merchant's Consignment Goods; (5) Additional Agent Goods; (6) furniture, furnishings, trade fixtures, machinery, equipment, office supplies, Supplies, conveyor systems, racking, rolling stock, and other personal property (but expressly excluding cash registers and point-of-sale system) (collectively without such exclusions, "FF&E") or improvements to real property; provided that Agent shall be permitted to sell Owned FF&E as set forth in Section 7 below; or (7) Distribution Center Merchandise received at the Stores after the Receipt Deadline or in-transit or on order goods.
- (b) As used in this Agreement, the following terms have the respective meanings set forth below:

"<u>Defective Merchandise</u>" means any item of Merchandise which is not new, finished, first-quality, saleable goods sold in the normal or ordinary course. Examples of Defective Merchandise include but are not limited to goods that are used, damaged, defective, scratched, soiled, dented, shopworn, out of box (if normally sold as new in-the-box), out of season, missing pieces, mismatched,

mismated, parts, items typically sold as a set which are incomplete, or gift with purchase items; <u>provided</u>, <u>however</u>, that display items and other clothing and accessories that can be sold individually shall not *per se* constitute "Defective Merchandise".

"Excluded Defective Merchandise" means any item of (i) Defective Merchandise that is not saleable in the ordinary course because it is so damaged or defective that it cannot reasonably be used for its intended purpose or for which the parties cannot mutually agree upon a Retail Price or (ii) obsolete or discontinued goods or goods that have been marked out of stock or have received a similar designation. Excluded Defective Merchandise shall be identified as such during the Inventory Taking.

"<u>Distribution Center Merchandise</u>" means all new, finished, first-quality saleable goods in the ordinary course of business located at Merchant's warehouse/distribution centers on the Sale Commencement Date (collectively, the "<u>Distribution Centers</u>") and reflected on Exhibit 5.2(b).

"Merchandise File" shall mean the following files, together with all updates and additional files received by Agent from Merchant through the Sale Commencement Date: "Dots Accessory Dept 4-5 (Item Level).xlsx", "Dots Junior Dept 4-5 (Item Level).xlsx", "Dots Plus Dept 4-5 (Item Level).xlsx", "Simply Fashions Accessory Dept 4-5 (Item Level).xlsx", "Simply Fashions Closeouts 4-5 (Item Report).xlsx", "Simply Fashions Junior Dept 4-5 (Item Level).xlsx", "Simply Fashions Plus Dept 4-5 (Item Level).xlsx", "Simply Fashions Super Plus Dept 4-5 (Item Level).xlsx".

5.3 <u>Valuation.</u>

- (a) For purposes of this Agreement, "Retail Price" shall be determined as of the Sale Commencement Date and shall mean with respect to each item of Merchandise, the lower of the lowest (i) ticketed price, (ii) marked price, (iii) shelf price, (iv) Merchandise File price, (iv) other file price as reflected in Merchant's books and records for such items, and (v) "buy one get one", "BOGO", "two for", or similar promotion price (which price shall be determined after taking into account such "buy one get one", "BOGO", "two for", or similar promotional pricing) of such item of Merchandise.
- (b) Notwithstanding the provisions of Section 5.3(a), with respect to each item of Defective Merchandise, the parties shall mutually agree upon the "Retail Price" (and if Agent and Merchant are unable to mutually agree on the Retail Price of any one or more items of Defective Merchandise, such items shall be deemed Excluded Defective Merchandise).
- 5.4 Excluded Goods. Merchant shall retain all responsibility for any goods not included as "Merchandise" hereunder. If the Merchant elects at the beginning of the Sale Term, Agent shall accept goods not included as "Merchandise" hereunder for sale at prices mutually agreed upon by Agent and Merchant (such goods, "Merchant's Consignment Goods"). The Agent shall retain 20% of the receipts (net of Sales Taxes) for all sales of Merchant's Consignment Goods, and Merchant shall receive 80% of the receipts (net of Sales Taxes) in respect of such sales. Merchant shall receive its share of the receipts of sales of Merchant's Consignment Goods on a weekly basis, immediately following the Weekly Sale Reconciliation. If Merchant does not elect to have Agent sell goods not included as Merchandise or Merchant and Agent are unable to agree upon prices, then all such items will be removed by Merchant from the Closing Stores at Merchant's expense as soon as practicable and shall not be shipped to the Closing Stores from the Distribution Centers absent Agent's express written consent. Agent shall have no cost, expense or responsibility in connection with any goods not included in Merchandise.

Section 6. Sale Term.

- 6.1 <u>Term.</u> Subject to satisfaction of the conditions precedent set forth in Section 10 hereof, the Sale shall commence at each Store on the date that is one (1) day after the Bankruptcy Court enters the Approval Order, but in no event later than May 7, 2015 (such date, the "<u>Sale Commencement Date</u>"). Agent shall complete the Sale at each Store no later than June 30, 2015 (the "<u>Sale Termination Date</u>", and the period from the Sale Commencement Date to the Sale Termination Date as to each Store being the "<u>Sale Term</u>"). Notwithstanding the foregoing, Agent may, in its discretion, earlier terminate the Sale on a Store-by-Store basis upon not less than seven (7) days' prior written notice (a "<u>Vacate Notice</u>") to Merchant (the "<u>Vacate Date</u>"), <u>provided</u>, that it being understood that the Agent's obligations to pay all Expenses, including Occupancy Expenses, for each Store subject to a Vacate Notice shall continue until the applicable Vacate Date for such Store.
- 6.2 <u>Vacating the Store</u>. At the conclusion of the Sale, Agent agrees to leave each Store in "broom clean" condition and otherwise in at least as good condition as on the Sale Commencement Date, ordinary wear and tear excepted, except for unsold items of Owned FF&E which may be abandoned by Agent in place in a neat and orderly manner pursuant to Section 7 below. Agent shall vacate each Store on or before the Sale Termination Date as provided for herein, at which time Agent shall surrender and deliver the Store premises, and Store keys, to Merchant. Agent's obligations to pay all Expenses for the Stores shall continue until the applicable Vacate Date for each Store.

Section 7. FF&E.

- 7.1 Owned FF&E. Agent shall sell all FF&E owned by Merchant located at the Stores, whether or not reflected on Exhibit 7.1 but including (without limitation) all items set forth on Exhibit 7.1 (the "Owned FF&E"). Merchant shall have the option, which option shall be exercised no later than one business day before the Sale Commencement Date, to have Agent sell the Owned FF&E on a commission or guaranteed basis. Merchant hereby represents, warrants, covenants, and agrees in favor of Agent that: (i) all Owned FF&E may be sold by Agent on Merchant's behalf, free and clear of all liens, claims, and encumbrances; (ii) Merchant owns all Owned FF&E and has good and marketable title to and all right, power, and authority to sell the Owned FF&E, and (iii) all Owned FF&E is devoid of hazardous materials and substances. As used herein, unless otherwise and later agreed to by the Merchant and the Agent in writing, the term Owned FF&E shall not include (i) any FF&E located in any premise other than the Stores (including the Corporate Office or any Distribution Center) and (ii) the point of sale system equipment, wheresover located.
- FF&E Guaranteed Option. If Merchant elects to have Agent sell the Owned FF&E on a guaranteed basis, Agent shall pay Merchant an amount equal to \$728,650 (i.e., \$2,950 per Store for 247 Stores) for the sole and exclusive right to sell or otherwise dispose of the Owned FF&E in the Stores (with no exclusions other than Merchant's cash registers and point-of-sale system, which are excluded from the definition of FF&E) (the "Additional Guaranteed Amount"), shall be reduced by \$2,950 per Store for any Store that is closed prior to the Sale Commencement Date and which Additional Guaranteed Amount shall be paid by Agent on the Payment Date; provided, further, that, if Merchant elects to have Agent sell the Owned FF&E on a guaranteed basis at the Distribution Center and/or Corporate Office, Merchant and Agent shall mutually agree upon the guaranteed amount for the Distribution Center and/or Corporate Office (the "Additional DC/CO Guaranteed Amount"), which Additional DC/CO Guaranteed Amount shall be paid at time(s) mutually acceptable to the Parties. In consideration for the payment of the Additional Guaranteed Amount and, if applicable the Additional DC/CO Guaranteed Amount, Agent shall be authorized to sell the Owned FF&E and retain all proceeds

(net of Sales Taxes, which shall be handled in accordance with section 8.3) from the sale of all Owned FF&E at the Stores and, if applicable, the Distribution Center and/or Corporate Office (the "FF&E Proceeds") for Agent's sole and exclusive benefit, which FF&E Proceeds shall not constitute Proceeds, and, in such circumstances, Agent shall be responsible for the payment of all costs and expenses associated with the disposition of the applicable Owned FF&E, other than Distribution Centers Expenses or any costs or expenses associated with the Corporate Offices, all of which shall be paid when due by Merchant.

- 7.3 <u>FF&E Fee Option</u>. If Merchant elects to have Agent sell the Owned FF&E on a fee basis, Agent shall be entitled to receive a commission equal to twenty percent (20%) of the gross proceeds from the sale of the Owned FF&E (net only of Sales Taxes), and Merchant shall reimburse Agent for Agent's documented costs and expenses associated with selling the Owned FF&E pursuant to a mutually agreed upon budget. Merchant shall be solely responsible for all costs and expenses associated with the sale of the FF&E, including (without limitation) Agent's costs and expenses.
- 7.4 <u>Abandonment of FF&E</u>. Anything in this Agreement to the contrary notwithstanding, Agent shall be authorized to abandon any and all FF&E, whether owned or not by Merchant, in place without any cost or liability to any party. For the avoidance of doubt, Agent shall have no responsibility whatsoever with respect to FF&E that is not owned by Merchant.

Section 8. Conduct of the Sale.

- Rights of Agent. In addition to any other rights granted to Agent elsewhere in this Agreement, Agent shall be permitted to conduct the Sale as a "store closing", "sale on everything", "going out of business", "everything must go", or similar themed sale throughout the Sale Term without compliance with any Liquidation Sale Laws. The Agent shall conduct the Sale in the name of and on behalf of the Merchant in a commercially reasonable manner and in compliance with the terms of this Agreement and subject to the Approval Order. Agent shall conduct the Sale in accordance with the sale guidelines attached hereto as Exhibit 8.1(the "Sale Guidelines"). Subject to the Approval Order and Sale Guidelines, in addition to any other rights granted to Agent hereunder in conducting the Sale, the Agent, in the exercise of its reasonable discretion shall have the right:
 - (a) to establish Sale prices and discounts and Store hours;
- (b) except as otherwise expressly included as an Expense, to use without charge during the Sale Term all FF&E, computer hardware and software, existing Supplies, intangible assets (including Merchant's name, logo and tax identification numbers), Store keys, case keys, security codes and safe and lock combinations required to gain access to and operate the Stores, and any other assets of the Merchant located at the Stores or the Distribution Centers (whether owned, leased, or licensed);
- (c) (i) to be provided by Merchant (at no additional cost to Agent) with central office facilities, central administrative services and personnel to process and perform Central Services and provide other central office services reasonably necessary for the Sale; but in no event shall Merchant be required to (x) provide any such services or facilities at a level in excess of the levels of such services and facilities customarily provided in connection with Merchant's operation of the Stores or (y) hire or retain personnel not presently employed or engaged by Merchant; (ii) to use not to exceed three reasonably sized offices located at Merchant's office facility located in Birmingham, Alabama to effect the Sale; and (iii) to use, subject to the Dots Carveout, all customer lists, mailing lists, email lists, and web and social networking sites utilized by Merchant in connection with its business (but solely in connection with the

Sale and pursuant to such reasonable restrictions requested by Merchant in order for Merchant to comply with its privacy policy and applicable laws governing the use and dissemination of confidential consumer personal data).

- (d) to establish and implement advertising, signage and promotion programs consistent with the "store closing", "sale on everything", "going out of business", "everything must go", or similar themed including without limitation by means of media advertising, and similar interior and exterior signs and banners, and the use of sign walkers;
- (e) to transfer Merchandise between and among the Stores at Agent's expense; <u>provided</u>, <u>however</u>, the Agent shall not transfer Merchandise between and among Stores so as to make the Merchandise unavailable for purposes of the Inventory Taking; and
- (f) subject to the provisions of Section 8.10 below, to include Additional Agent Goods as part of the Sale.
- 8.2 Terms of Sales to Customers; Final/As Is Sales. All sales of Merchandise will be "final sales" and "as is," and appropriate signage and sales receipts will reflect the same. Agent shall not warrant the Merchandise in any manner, but will, to the extent legally permissible, pass on all manufacturers' warranties to customers. All sales will be made only for cash or nationally recognized bank credit cards. Agent shall accept and honor coupons during the Sale Term, including but not limited to, Merchant's membership program as well as Merchant's employee discount terms as are in effect immediately prior to the commencement of the Sale Term. The Agent shall clearly mark all receipts for the Merchandise sold at the Stores during the Sale Term so as to distinguish such Merchandise from the goods sold prior to the Sale Commencement Date. Merchant shall reimburse Agent in cash for all amounts related to coupons, including, but not limited to, Merchant's membership programs as well as Merchant's employee discount terms, during each Weekly Sale Reconciliation provided for in Section 8.7.

8.3 Sales Taxes.

During the Sale Term, all sales, excise, gross receipts and other taxes attributable (a) to sales of Merchandise and Additional Agent Goods, as indicated on Merchant's point of sale equipment (other than taxes on income, but specifically including, without limitation, gross receipts taxes) payable to any taxing authority having jurisdiction (collectively, "Sales Taxes") shall be added to the sales price of Merchandise and Additional Agent Goods and collected by Agent, on Merchant's behalf and in trust for Merchant and for the benefit of the taxing authorities, at the time of sale and paid over to Merchant. All Sales Taxes shall be deposited into a segregated account designated by Merchant and Agent solely for the deposit of such Sales Taxes (the "Sales Taxes Account"). If Agent does not timely remit Sales Taxes to Merchant, Merchant shall be permitted to immediately draw on the Letter of Credit in unremitted amount of Sales Taxes collected by Agent in the preceding week. Merchant shall prepare and file all applicable reports and documents required by the applicable taxing authorities, and Merchant shall promptly pay all Sales Taxes collected by Merchant or turned by Agent to Merchant from the Sales Taxes Account. Merchant will be given access to the computation of gross receipts for verification of all such Sales Tax collections. If Agent calculates Sales Taxes using systems other than Mercahnt's systems, Agent shall reimburse Merchant for any additional Sales Taxes, interest, fines, penalties, and similar amounts payable to any taxing authority as the result of a Sales Tax audit conducted by or on behalf of such authority which discloses that the Sales Takes collected by Agent and paid over to Merchant for any period during the Sale Term were less than those mandated by applicable law for the sale of Merchandise, Additional

Agent Goods and/or Owned FF&E, if any, that is sold by Agent under this Agreement (any such additional Sales Taxes and other amounts are collectively referred to herein as "Additional Taxes and Penalties"). Provided that Agent performs its responsibilities in accordance with this Section 8.3, Agent shall have no further obligation to the Merchant, the Lender, any taxing authority, or any other party, and Merchant (and Lender to the extent it has received any funds on account of Sales Taxes) shall indemnify and hold harmless Agent and its officers, directors, employees, agents and independent contractors (collectively, "Agent Indemnified Parties") from and against any and all costs, including, but not limited to, reasonable attorneys' fees, assessments, fines or penalties which Agent sustains or incurs as a result or consequence of the failure by Merchant to promptly pay such taxes to the proper taxing authorities and/or the failure by Merchant to promptly file with such taxing authorities all reports and other documents required by applicable law to be filed with or delivered to such taxing authorities. If Agent fails to perform its responsibilities in accordance with this Section 8.3, and provided Merchant complies with its obligations hereunder, Agent shall indemnify and hold harmless Merchant and its officers, directors, employees, agents and independent contractors (collectively, "Merchant Indemnified Parties") from and against any and all costs, including, but not limited to, reasonable attorneys' fees, assessments, fines or penalties (including but not limited to all Additional Taxes and Penalties) which Merchant sustains or incurs as a result or consequence of the failure by Agent to collect Sales Taxes and/or the failure by Agent to promptly deliver any and all reports and other documents required to enable Merchant to file any requisite returns with such taxing authorities.

- (b) Without limiting the generality of Section 8.3(a) hereof, it is hereby agreed that, as Agent is conducting the Sale solely as agent for the Merchant, various payments that this Agreement contemplates that one party may make to the other party (including the payment by Agent of the Guaranteed Amount) do not represent the sale of tangible personal property and, accordingly, are not subject to Sales Taxes.
- 8.4 <u>Supplies</u>. Agent shall have the right to use, without charge, all existing supplies located at the Stores, Distribution Centers and corporate office(s), including, without limitation, boxes, bags, paper, twine and similar sales materials (collectively, "<u>Supplies</u>"). In the event that additional Supplies are required in any of the Stores during the Sale, Merchant agrees to reasonably cooperate with Agent to obtain the same, if reasonably available, for which Agent shall reimburse the Merchant at Merchant's cost therefor in connection with each Weekly Sale Reconciliation.
- Returns of Merchandise. Agent shall accept returns of goods sold by Merchant prior to the Sale Commencement Date ("Returned Merchandise"), provided that such return is in compliance with Merchant's return policy in effect immediately prior to the Sale Commencement Date. If such Returned Merchandise is otherwise "Merchandise" it shall be included in the Sale at its Retail Price, multiplied by the inverse of the then prevailing Sale discount on the date of the return. To the extent Returned Merchandise is salable as first quality merchandise, it shall be included in Merchandise and for purposes of the calculation of the Guaranteed Amount and (i) if returned during the first seven (7) days of the Sale, shall be valued at the Retail Price applicable to such item or (ii) if returned after such seven-day period, shall be valued at the Retail Price applicable to such item multiplied by the inverse of the then prevailing Sale discount. Subject to Merchant's reimbursement to Agent of the amount of any store credit or refund granted for any such Returned Merchandise, the aggregate Retail Price of the Merchandise shall be increased by the applicable Retail Price of any Returned Merchandise, and the Guaranteed Amount shall be adjusted accordingly. If the Returned Merchandise is not first, quality goods, Merchant and Agent shall negotiate in good faith to determine an appropriate Retail Price applicable to such merchandise for purposes of determining the Retail Price attributable thereto; provided that, in the event Merchant and Agent cannot agree on the Retail Price to be attributed to any particular

items) of Returned Merchandise, than such items) shall be segregated form Merchandise and excluded from the Sale and treated as Excluded Defective Merchandise for all purposes hereunder. Any reimbursements due to Agent as a result of Returned Merchandise shall be accounted for and paid by Merchant immediately following the Weekly Sale Reconciliation pursuant to Section 8.7 hereof. Any increases in payment on account of the Guaranteed Amount as a result of Returned Merchandise shall be paid by Agent as part of the Final Reconciliation provided for under Section 8.7 hereof. In brief, the aggregate Retail Price of the Merchandise shall be increased by the adjusted Retail Price of any Returned Merchandise included in Merchandise (determined in accordance with this Section 8.5). In addition, Merchant shall promptly reimburse Agent in cash or credit for any refunds Agent is required to issue to customers in respect of any Returned Merchandise during each Weekly Sale Reconciliation provided for in Section 8.7. Except to the extent that Merchant and Agent agree that Merchant's POS or other applicable systems can account for returns of goods, all returns must be noted and described in a mutually agreeable Returned Merchandise log on a weekly basis during the Sale.

8.6 <u>Gift Certificates; Membership Program.</u>

- (a) During the Sale Term, Agent shall accept Merchant's gift certificates, gift cards, return credits, and similar merchandise credits issued by Merchant in accordance with Merchant's polities and the terms and conditions applicable to such items (collectively, the "Gift Certificates"), which terms and conditions will be made available by Merchant to Agent prior to the Sale Commencement Date; and Merchant shall reimburse Agent in cash for such amounts during the Weekly Sale Reconciliation provided for in Section 8.7. Agent shall not sell any Gift Certificates.
- (b) During the Sale Term, customers may elect to take advantage of (i) discounts afforded customers in connection with Merchant's membership program benefits and/or Merchant's then valid coupons (collectively, "Merchant Discounts"); or (ii) the then-prevailing Sale discounts being offered by Agent, but not both on a cumulative basis. To the extent the customer elects an applicable Merchant Discount, then Merchant shall reimburse Agent in cash during the Weekly Sale Reconciliation for the value/differential between the applicable Merchant Discount and the then-prevailing Sale discounts being offered by Agent.
- 8.7 Sale Reconciliation. On each Wednesday during the Sale Term, Agent and Merchant shall cooperate to reconcile Expenses of the Sale, make payments/setoffs on account of the Guaranteed Amount, Agent's Fee, Sharing Amounts, sales of Additional Agent Goods, and Owned FF&E, and reconcile such other Sale-related items as either party shall reasonably request, in each case for the prior week or partial week (i.e. Sunday through Saturday), all pursuant to procedures agreed upon by Merchant and Agent (the "Weekly Sale Reconciliation"). On a weekly basis, Agent shall also provide Merchant (and a copy to the Lender) with a report (in electronic format reasonably acceptable to Merchant) of all sales of Additional Agent Goods, which report shall detail by Store gross sales (net of Sales Taxes) and, if available with Merchant's systems, items sold. Within thirty (30) days after the end of the Sale Term, or as soon as practicable thereafter, Agent and Merchant shall jointly prepare a final reconciliation of the Sale, including, without limitation, a summary of Proceeds, Sales Taxes, Expenses, and any other accountings required hereunder (the "Final Reconciliation"), the written results of which shall be certified by representatives of each of the Merchant and Agent as a final settlement of accounts between the Merchant and Agent. Within five (5) days after the completion of the Final Reconciliation and execution of a settlement letter including an appropriate mutual release, Agent shall pay to Merchant, or Merchant shall pay to Agent, as the case may be, any and all amounts due the other pursuant to the Final Reconciliation. Once executed by Merchant and Agent, such settlement and Final Reconciliation shall be deemed approved without further order of the Bankruptcy Court (other than the Approval Order).

During the Sale Term, and thereafter until all of Merchant's and Agent's obligations under this Agreement have been satisfied, Merchant and Agent shall have reasonable access to Merchant's and Agent's records with respect to the Sale (including, but not limited to, Retail Price, Merchandise, Expenses, and Proceeds) to review and audit such records. In the event that there is any dispute with respect to either (x) the determination of the aggregate Retail Price of the Merchandise as reflected in the Final Inventory Report and/or (y) the Final Reconciliation, such dispute shall be promptly (and in no event later than the fifth (5th) business day following a request by either Merchant or Agent) submitted to the Bankruptcy Court for resolution. In the event of a dispute as to (x) or (y) above, Agent shall extend the Letter of Credit in accordance with the provisions of Sections 3.4 hereof, as applicable. If Agent has for any reason not so extended the expiration date of the Letter of Credit by the date that is ten (10) business days prior to the applicable expiration date (as may have been extended previously), Merchant and/or Lender shall have the right to make a drawing under the Letter of Credit in an amount or amounts equal to the undisputed amounts Merchant asserts are then owing to Merchant.

- 8.8 Force Majeure. If any casualty, act or threatened act of terrorism, or act of God prevents or substantially inhibits the conduct of business in the ordinary course at any of the Stores for a period of five (5) consecutive days, the Merchandise located at such Store shall, in Agent's reasonable discretion (after consultation with the Merchant), be eliminated from the Sale and considered to be deleted from this Agreement as of the date of such event, and Agent and Merchant shall have no further rights or obligations hereunder with respect thereto; provided, however, that (i) the proceeds of any insurance attributable to such Merchandise shall constitute Proceeds hereunder, and (ii) the Guaranteed Amount shall be reduced to account for any Merchandise eliminated from the Sale which is not the subject of insurance proceeds, and Merchant (or Lender to the extent it has received any funds on account of the Guaranteed Amount) shall within five (5) business days following written demand by Agent reimburse Agent for the amount the Guaranteed Amount is so reduced.
- 8.9 <u>Right to Monitor</u>. Merchant shall have the right to monitor the Sale and activities attendant thereto and to be present in the Stores during the hours when the Stores are open for business; provided that Merchant's presence does not unreasonably disrupt the conduct of the Sale. Merchant shall also have a right of access to the Stores at any time in the event of an emergency situation and shall promptly notify Agent of such emergency.

8.10 Additional Agent Goods.

(a) Agent shall have the right to supplement the Merchandise in the Sale with additional goods procured by Agent which are of like kind, and no lesser quality to the Merchandise in the Sale ("Additional Agent Goods"). The Additional Agent Goods shall be purchased by Agent as part the Sale at Agent's sole expense (and such purchase price shall not constitute an Expense). Agent shall have no right to use the Distribution Center for Additional Agent Goods unless otherwise agreed to by Merchant and Agent. Sales of Additional Agent Goods shall be run through Merchant's point of sale systems, provided however, Agent shall mark the Additional Agent Goods using either a "dummy" SKU or department number, or in such other manner so as to distinguish the sale of Additional Agent Goods from the sale of Merchandise. Agent and Merchant shall also cooperate so as to ensure that the Additional Agent Goods are marked in such a way that a reasonable consumer could identify the Additional Agent Goods as non-Merchant goods. Additionally, Agent shall provide, at Agent's sole expense (and as an Expense), signage in the Stores notifying customers that the Additional Agent Goods have been included in the Sale.

- (b) In addition to all other compensation due to Merchant under this Agreement, Agent shall pay to Merchant an amount equal to five percent (5.0%) percent of the gross proceeds (excluding Sale Taxes) from the sale of the Additional Agent Goods (the "Additional Agent Goods Fee") and Agent shall retain all remaining amounts from the sale of the Additional Agent Goods. Agent shall pay Merchant its Additional Agent Goods Fee in connection with each Weekly Sale Reconciliation with respect to sales of Additional Agent Goods sold by Agent during each then prior week (or at such other mutually agreed upon time in the case of Additional Agent Goods sold pursuant to Section 3.2(b) above).
- (c) Agent and Merchant intend that the transactions relating to the Additional Agent Goods are, and shall be construed as, a true consignment from Agent to Merchant in all respects and not a consignment for security purposes. Subject solely to Agent's obligations to pay to Merchant the Additional Agent Goods Fee, at all times and for all purposes the Additional Agent Goods and their proceeds shall be the exclusive property of Agent, and no other person or entity shall have any claim against any of the Additional Agent Goods or their proceeds. The Additional Agent Goods shall at all times remain subject to the exclusive control of Agent.
- (d) Merchant shall, at Agent's sole expense (and not as an Expense), insure the Additional Agent Goods and, if required, promptly file any proofs of loss with regard to same with Merchant's insurers. Agent shall be responsible for payment of any deductible under any such insurance in the event of any casualty affecting the Additional Agent Goods, which amount shall be deemed an Additional Agent Goods expense.
- (e) Merchant acknowledges, and the Approval Order shall provide, that the Additional Agent Goods shall be consigned to Merchant as a true consignment under Article 9 of the Code. Agent is hereby granted a first priority security interest in and lien upon (i) the Additional Agent Goods and (ii) the Additional Agent Goods proceeds, which security interest shall be deemed perfected pursuant to the Approval Order without the requirement of filing UCC financing statements or providing notifications to any prior secured parties (provided that Agent is hereby authorized to deliver all required notices and file all necessary financing statements and amendments thereof under the applicable UCC identifying Agent's interest in the Additional Agent Goods as consigned goods thereunder and the Merchant as the consignee therefor, and Agent's security interest in and lien upon such Additional Agent Goods and Additional Agent Goods proceeds).
- (f) Lender hereby acknowledges receipt of notice of the consignment of the Additional Agent Goods and consents to the payments to Agent of Additional Agent Goods proceeds.
- 8.11 <u>Collective Bargaining Agreements; Leases.</u> Agent shall not be obligated to comply with any of Merchant's collective bargaining agreements or leases/occupancy agreements; except as provided for in sections 4.1(a) and 4.1(c) herein.

Section 9. Employee Matters.

9.1 Merchant's Employees. Agent may use Merchant's employees in the conduct of the Sale to the extent Agent deems necessary for the Sale, and Agent may select and schedule the number and type of Merchant's employees required for the Sale. Agent shall identify any such employees to be used in connection with the Sale (each such employee, a "Retained Employee"). Notwithstanding the foregoing, Merchant's employees shall at all times remain employees of the Merchant. Agent's selection and scheduling of Merchant's employees shall at all times comply with all applicable laws and regulations. Merchant and Agent agree that, except to the extent that wages and benefits of Retained

Employees constitute Expenses hereunder, nothing contained in this Agreement and none of Agent's actions taken in respect of the Sale shall be deemed to constitute an assumption by Agent of any of Merchant's obligations relating to any of Merchant's employees including, without limitation, Excluded Payroll Benefits, Worker Adjustment Retraining Notification Act ("WARN Act") claims and other termination type claims and obligations, or any other amounts required to be paid by statute or law; nor shall Agent become liable under any employment agreement, collective bargaining agreement, or be deemed a joint or successor employer with respect to such employees. Merchant shall not, without the prior consent of Agent, raise the salary or wages or increase the benefits for, or pay any bonuses or other extraordinary payments to, any Store employees prior to the Sale Termination Date. Merchant shall not transfer any employee in anticipation of the Sale nor any Retained Employee during the Sale Term, in each case without Agent's prior consent.

- Employee at any time during the Sale, subject to the conditions provided for herein. In the event that Agent desires to cease using any Retained Employee, Agent shall notify Merchant at least seven (7) days prior thereto; provided, however, that, in the event that Agent determines to cease using an employee "for cause" (such as dishonesty, fraud or breach of employee duties), the seven (7) day notice period shall not apply; provided, further, however, that Agent shall immediately notify Merchant of the basis for such "cause." From and after the date of this Agreement and until the Sale Termination Date, Merchant shall not transfer or dismiss employees of the Stores except "for cause" without Agent's prior consent. Notwithstanding the foregoing, Agent shall not have the right to terminate the actual employment of any employee, but rather may only cease using such employee in the Sale and paying any Expenses with respect to such employee (and all decisions relating to the termination or non-termination of such employees shall at all times rest solely with Merchant).
- 9.3 <u>Payroll Matters</u>. During the Sale Term, Merchant shall process the payroll for all Retained Employees and any former employees and temporary labor engaged for the Sale. Each Wednesday (or such other date as may be reasonably requested by Merchant to permit the funding of the payroll accounts before such payroll is due and payable) during the Sale Term, Agent shall transfer to Merchant's payroll accounts an amount equal to the base payroll for Retained Employees plus related payroll taxes, workers' compensation and benefits for such week, to the extent such amount constitutes Expenses hereunder.
- 9.4 <u>Employee Retention Bonuses</u>. Subject to approval by the Bankruptcy Court, Agent may pay, as an Expense, retention bonuses ("<u>Retention Bonuses</u>") (which bonuses shall be inclusive of payroll taxes, but as to which no benefits shall be payable), up to a maximum of ten percent (10%) of base payroll for all Retained Employees, to such Retained Employees who do not voluntarily leave employment and are not terminated "for cause," as Agent may determine in its discretion. Subject to approval by the Bankruptcy Court, the amount of such Retention Bonuses shall be in an amount to be determined by Agent, in its discretion, and shall be payable within thirty (30) days after the Sale Termination Date, and shall be processed through Merchant's payroll system.

Section 10. Conditions Precedent and Subsequent.

(a) The willingness of Agent to enter into the transactions contemplated under this Agreement is directly conditioned upon the satisfaction of the following conditions at the time or during the time periods indicated, unless specifically waived in writing by Agent:

- (i) All representations and warranties of the Merchant hereunder shall be true and correct in all material respects and no Event of Default shall have occurred at and as of the date hereof and on the Sale Commencement Date.
- (ii) Lender has executed this Agreement (with all Exhibits attached hereto) solely to bind itself to the provisions specifically relating to Lender herein.
- (iii) The Bankruptcy Court shall have entered the Approval Order by May 6, 2015, or such later date as mutually agreed upon by the Merchant and the Agent (the "<u>Approval Order</u> Deadline").
- (b) The willingness of Merchant to enter into the transactions contemplated under this Agreement is directly conditioned upon the satisfaction of the following conditions at the time or during the time periods indicated, unless specifically waived in writing by the Merchant:
- (i) All representations and warranties of Agent hereunder shall be true and correct in all material respects and no Event of Default shall have occurred at and as of the date hereof and on the Sale Commencement Date.
- (ii) Lender has executed this Agreement (with all Exhibits attached hereto), subject to the Approval Order, solely to bind itself to the provisions specifically relating to such Lender herein, subject to the Approval Order.

Section 11. Representations, Warranties and Covenants.

- 11.1 <u>Merchant's Representations, Warranties and Covenants</u>. Merchant hereby represents, warrants and covenants in favor of Agent as follows:
- (a) Merchant (i) are entities duly organized, validly existing and in good standing under the laws of the State of Florida and the State of Alabama; (ii) subject to the applicable provisions of the Bankruptcy Code, has all requisite corporate power and authority to own, lease and operate its assets and properties and to carry on its business as presently conducted; and (iii) is, and during the Sale Term will continue to be, duly authorized and qualified to do business and in good standing in each jurisdiction where the nature of its business or properties requires such qualification, including all jurisdictions in which the Stores are located, except, in each case, to the extent that the failure to be in good standing or so qualified could not reasonably be expected to have a material adverse effect on the ability of the Merchant to execute and deliver this Agreement and perform fully its obligations hereunder.
- (b) Subject to entry of the Approval Order, the Merchant has the right, power and authority to execute and deliver this Agreement and each other document and agreement contemplated hereby (collectively, together with this Agreement, the "Agency Documents") and to perform fully its obligations thereunder. Subject to entry of the Approval Order, Merchant has taken all necessary actions required to authorize the execution, delivery and performance of the Agency Documents, and no further consent or approval is required for Merchant to enter into and deliver the Agency Documents, to perform its obligations thereunder and to consummate the Sale, except for any such consent the failure of which to be obtained could not reasonably be expected to have a material adverse effect on the ability of Merchant to execute and deliver this Agreement and perform fully its obligations hereunder. Subject to entry of the Approval Order, each of the Agency Documents has been duly executed and delivered by the Merchant

and constitutes the legal, valid and binding obligation of the Merchant enforceable in accordance with its terms.

- (c) Merchant owns, and will own at all times during the Sale Term, good and marketable title to all of the Merchandise and Owned FF&E to be included in the Sale, free and clear of all security interests, liens, claims and encumbrances of any nature (other than the security interests and liens of the Agent hereunder and Lender). Merchant shall not create, incur, assume or suffer to exist any security interest, lien or other charge or encumbrance upon or with respect to any of the Merchandise or the Proceeds (or the Owned FF&E and its proceeds), in each case other than the liens and security interests of Agent hereunder and of the Lender and any other liens and security interests granted pursuant to Merchant's debtor-in-possession financing (if any) and cash collateral usage.
- (d) Merchant has maintained its pricing files (including (without limitation) the Merchandise File) in the ordinary course of business, and prices charged to the public for goods are the same in all material respects as set forth in such pricing files for the periods indicated therein, and all pricing files and records are true and accurate in all material respects as to the actual cost to Merchant for purchasing the goods referred to therein and as to the selling price to the public for such goods without consideration of any point of sale discounts, as of the dates and for the periods indicated therein. Merchant represents that (i) the ticketed prices of all items of Merchandise do not and shall not include any Sales Taxes and (ii) all registers located at the Stores are programmed to correctly compute all Sales Taxes required to be paid by the customer under applicable law, as such calculations have been identified to Merchant by its retained service provider.
- (e) Through the Sale Commencement Date, Merchant has ticketed or marked, and shall continue to ticket or mark, all items of inventory received at the Stores in a manner consistent with similar Merchandise located at the Stores, and in accordance with Merchant's ordinary course past practices and policies relative to pricing and marking inventory. Since April 1, 2015, Merchant has not removed any POS promotions, sale stickers, or other markings indicating items are on sale or on clearance from the Merchandise prior to the Sale Commencement Date, and have not raised, and will not raise, prices of any Merchandise in contemplation of the Sale.
- (f) Since April 1, 2015, Merchant has not, and through the Sale Commencement Date Merchant shall not, purchase for or transfer to or from the Stores any merchandise or goods outside the ordinary course; provided, however, that in no event shall Merchant transfer any goods into the Stores without Agent's consent from and after the date hereof.
- (g) To Merchant's knowledge after reasonable inquiry, all Merchandise is in compliance with all applicable federal, state and local product safety laws, rules and standards. Merchant shall provide Agent with its historic policies and practices, if any, regarding product recalls prior to the Sale Commencement Date.
- (h) Subject to the provisions of the Approval Order, Agent shall have the right to the unencumbered use and occupancy of, and peaceful and quiet possession of, the Stores, the assets currently located at the Stores, and the utilities and other services provided at the Stores. Merchant shall, throughout the Sale Term, maintain in good working order, condition and repair all cash registers, heating systems, air conditioning systems, elevators, escalators and all other mechanical devices necessary or appropriate for the conduct of the Sale at the Stores. Except as otherwise restricted by the Bankruptcy Code or as provided herein and absent a bona fide dispute, throughout the Sale Term Merchant shall remain current on all expenses and payables necessary or appropriate for the conduct of the Sale.

- (i) Subject to approval by the Bankruptcy Court, Merchant has paid, and will continue to pay throughout the Sale Term, all self-insured or Merchant-funded employee benefit programs for Store employees, including health and medical benefits and insurance and all proper claims made or to be made in accordance with such programs.
- (j) Since April 1, 2015, Merchant has not taken, and shall not throughout the Sale Term take, any actions with the intent of increasing the Expenses of the Sale, including without limitation increasing salaries or other amounts payable to employees; except to the extent an employee was due an annual raise in the ordinary course.
- (k) Since April 1, 2015, Merchant has operated, and, except as otherwise restricted by the Bankruptcy Code or as provided herein and absent a bona fide dispute, through the Sale Commencement Date Merchant covenants to continue to operate, the Stores in all material respects in the ordinary course of business including without limitation by: (i) selling inventory during such period at customary prices consistent with the ordinary course of business; (ii) not promoting or advertising any sales or in-store promotions (including POS promotions) to the public outside of the Merchant's ordinary course of business; (iii) except as may occur in the ordinary course of business, not returning inventory to vendors and not transferring inventory or supplies out of or to the Stores; (iv) except as may occur in the ordinary course of business, not making any management personnel moves or changes at the Stores; and (v) replenishing the Stores in the ordinary course of business.
- (l) Prior to the execution of this Agreement, Merchant has provided Agent reasonable access to all pricing and cost files, computer hardware, software and data files, inter-Stores transfer logs, markdown schedules, invoices, style runs and all other documents relative to the price, mix and quantities of inventory located at the Stores and the Distribution Centers or on order or in transit.
- (m) Merchant has provided representatives of Agent with access to (and the opportunity to copy) all collective bargaining agreements, and all amendments, modifications, and supplements thereto.
- (n) To Merchant's knowledge after reasonable inquiry, all documents, information and supplements provided by Merchant to Agent in connection with Agent's due diligence and the negotiation of this Agreement were true and accurate in all material respects at the time provided.
- (o) On and as of the Sale Commencement Date, the level of goods (as to quantity) and the mix of goods (as to type, category, style, brand and description) at the Stores shall be in all material respects, subject to ordinary course of business changes and adjustments, consistent with the level and mix set forth on Exhibit 11.1(o).
- (p) Merchant has not and shall not purchase or transfer to or from the Stores any merchandise or goods outside the ordinary course in anticipation of the Sale or of the Inventory Taking.
- (q) Merchant has not since April 1, 2015 shipped any Excluded Defective Merchandise from the Distribution Centers to the Stores. Merchant will not ship any Excluded Defective Merchandise from the date of this Agreement from the Distribution Centers to the Stores.
- (r) Other than filing the Bankruptcy Case, no action, arbitration, suit, notice, or legal, administrative or other proceeding before any court or governmental body has been instituted by or against the Merchant, or has been settled or resolved, or to Merchant's knowledge, is threatened against or

affects Merchant, relative to Merchant's business or properties, or which questions the validity of this Agreement, or that if adversely determined, would adversely affect the conduct of the Sale.

- (s) Merchant shall not, prior to the Sale Termination Date, offer any promotions or discounts at the Stores, any other retail store location, or any other retail sales channel, except as detailed on Exhibit 11.1(s).
- (t) Merchant hereby acknowledges that, as of and after the date of this Agreement, Agent intends to begin to make arrangements for the purchase of Additional Agent Goods from Agent's suppliers of goods as well as Merchant's existing suppliers of goods, all at no cost or expense whatsoever to Merchant and at Agent's sole risk. To assist Agent in that regard, from and after the date of this Agreement, Merchant shall use Merchant's commercially reasonable efforts to assist and cause Merchant's employees to assist Agent with Agent's efforts to negotiate orders with Merchant's suppliers for goods that Agent is interested in purchasing as Additional Agent Goods, provided that (i) such assistance does not unreasonably interfere with or disrupt Merchant's business operations, (ii) the appropriate employees to perform such activities then remain employees of Merchant, it being agreed that Merchant shall in no event be required to retain any such employee for the purpose of rendering such assistance, (iii) Merchant shall have no liability whatsoever in connection with any orders placed for such Additional Agent Goods or otherwise in connection with the activities contemplated by this subpart (t).
- 11.2 <u>Agent's Representations, Warranties and Covenants</u>. Agent hereby represents, warrants and covenants in favor of Merchant as follows:
- (a) Each entity composing Agent: (i) is a limited liability company duly and validly existing and in good standing under the laws of the State of Delaware; (ii) has all requisite power and authority to carry on its business as presently conducted and to consummate the transactions contemplated hereby; (iii) is, and during the Sale Term will continue to be, duly authorized and qualified to do business and in good standing in each jurisdiction where the nature of its business or properties requires such qualification, including all jurisdictions in which the Stores are located, except, in each case, to the extent that the failure to be in good standing or so qualified could not reasonably be expected to have a material adverse effect on the ability of Agent to execute and deliver this Agreement and perform fully its obligations hereunder.
- (b) Agent has the right, power and authority to execute and deliver each of the Agency Documents to which it is a party and to perform fully its obligations thereunder. Agent has taken all necessary actions required to authorize the execution, delivery and performance of the Agency Documents, and no further consent or approval is required on the part of Agent for Agent to enter into and deliver the Agency Documents, to perform its obligations thereunder and to consummate the Sale. Each of the Agency Documents has been duly executed and delivered by the Agent and constitutes the legal, valid and binding obligation of Agent enforceable in accordance with its terms. No court order or decree of any federal, state or local governmental authority or regulatory body is in effect that would prevent or impair, or is required for, Agent's consummation of the transactions contemplated by this Agreement, and no consent of any third party which has not been obtained is required therefor, other than as provided herein. No contract or other agreement to which Agent is a party or by which Agent is otherwise bound will prevent or impair the consummation of the transactions contemplated by this Agreement.
- (c) No action, arbitration, suit, notice or legal administrative or other proceeding before any court or governmental body has been instituted by or against Agent, or has been settled or resolved or, to Agent's knowledge, has been threatened against or affects Agent, which questions the

validity of this Agreement or any action taken or to be taken by Agent in connection with this Agreement or which, if adversely determined, would have a material adverse effect upon Agent's ability to perform its obligations under this Agreement.

- (d) The Sale shall be conducted in compliance with all applicable state and local laws, rules and regulations and Merchant's leases and other agreements, except as otherwise provided for in the Sale Guidelines and Approval Order.
- (e) Absent prior consent by the Merchant, Agent will not cause any non-emergency repairs or maintenance (emergency repairs are repairs necessary to preserve the security of a Store premise or to ensure customer safety) to be conducted at the Stores.
- (f) To the best of Agent's knowledge, all Additional Agent Goods are in compliance with all applicable federal, state or local product safety laws, rules and standards. All Additional Agent Goods shall be of like kind and no lesser quality to the Merchandise located in the Stores.

Section 12. Insurance.

- 12.1 Merchant's Liability Insurance. Subject to approval by the Bankruptcy Court, Merchant shall continue at its cost and expense until the Sale Termination Date, in such amounts as it currently has in effect, all of its liability insurance policies, including, but not limited to, commercial general liability, products liability, comprehensive public liability, auto liability and umbrella liability insurance, covering injuries to persons and property in, or in connection with, Merchant's operation of the Stores; and Merchant shall cause Agent to be named as an additional named insured (as its interest may appear) with respect to all such policies. Merchant shall deliver to Agent certificates evidencing such insurance setting forth the duration thereof and naming Agent as an additional named insured, in form reasonably satisfactory to Agent. All such policies shall require at least thirty (30) days' prior notice to Agent of cancellation, non-renewal or material change during the Sale Term. In the event of a claim under any such policies, Merchant shall be responsible for the payment of all deductibles, retentions or self-insured amounts thereunder, unless it is determined that liability arose by reason of the willful misconduct or grossly negligent acts or omissions of Agent, or Agent's employees, independent contractors or agents. Merchant shall not make any change in the amount of any deductibles or selfinsurance amounts prior to the Sale Termination Date without Agent's prior written consent.
- Merchant's Casualty Insurance. Subject to approval by the Bankruptcy Court, Merchant shall provide, as an Occupancy Expense, throughout the Sale Term fire, flood, theft and extended coverage casualty insurance covering the Merchandise in a total amount equal to no less than the retail value thereof. From and after the date of this Agreement until the Sale Termination Date, all such policies will also name Agent as an additional named insured or loss payee, as applicable (as its interest may appear). In the event of a loss to the Merchandise on or after the date of this Agreement, the Proceeds of such insurance attributable to the Merchandise, shall constitute Proceeds hereunder. Merchant shall deliver to Agent certificates evidencing such insurance, setting forth the duration thereof and naming the Agent as an additional insured or loss payee, as applicable, in form and substance reasonably satisfactory to Agent. All such policies shall require at least thirty (30) days' prior notice to the Agent of cancellation, non-renewal or material change during the Sale Term. Merchant shall not make any change in the amount of any deductibles or self-insurance amounts prior to the Sale Termination Date without Agent's prior written consent.

- 12.3 Agent's Insurance. Agent shall maintain at Agent's cost as an Expense hereunder throughout the Sale Term, in such amounts as it currently has in effect, commercial general liability policies covering injuries to persons and property in or in connection with Agent's agency at the Store, and shall cause Merchant to be named as an additional insured with respect to such policies. Agent shall deliver to Merchant certificates evidencing such insurance policies setting forth the duration thereof and naming Merchant as an additional insured, in form and substance reasonably satisfactory to Merchant. In the event of a claim under any such policies, Agent shall be responsible for the payment of all deductibles, retentions or self-insured amounts thereunder, unless it is determined that liability arose by reason of the willful misconduct or grossly negligent acts or omissions of Merchant or Merchant's employees, independent contractors or agents (other than Agent or Agent's employees, agents or independent contractors). Agent shall not make any change in the amount of any deductibles or self insurance amounts prior to the Sale Termination Date without Merchant's prior written consent.
- 12.4 <u>Worker's Compensation Insurance</u>. Subject to approval by the Bankruptcy Court, Merchant shall at all times during the Sale Term maintain in full force and effect workers' compensation insurance (including employer liability insurance) covering all Retained Employees in compliance with all statutory requirements.

Section 13. Indemnification.

- Merchant's Indemnification. Merchant shall indemnify and hold Agent and its 13.1 officers, directors, employees, agents, representatives, and independent contractors (collectively, "Agent Indemnified Parties") harmless from and against all claims, causes of action, demands, penalties, losses, liability, damage, or other obligations, including, without limitation, reasonable attorneys' fees and expenses, directly or indirectly asserted against, resulting from or related to: (i) Merchant's material breach of or failure to comply with any of its agreements, covenants, representations or warranties contained in any Agency Document; (ii) subject to Agent's compliance with its obligations under Section 8.3 hereof, any failure by Merchant to pay any Sales Taxes to the proper taxing authorities or to properly file with any taxing authorities any reports or documents required by applicable law to be filed in respect thereof; (iii) any failure of Merchant to pay to its employees any wages, salaries or benefits due to such employees during the Sale Term; (iv) any consumer warranty or products liability claims or other claims of any type or kind and by any party whatsoever relating to Merchandise; (v) any liability or other claims asserted by customers, any of Merchant's employees, or any other person against any Agent Indemnified Party (including, without limitation, claims by employees arising under collective bargaining agreements, worker's compensation or under the WARN Act); (vi) any harassment or any other unlawful, tortious, or otherwise actionable treatment of any customers, employees or agents of Agent by Merchant or any of its representatives; (vi) any failure of Merchant to pay to any Occupancy Expenses or Central Service Expenses during the Sale Term; and (vii) the gross negligence (including omissions) or willful misconduct of the Merchant, its officers, directors, employees, agents (other than Agent) or representatives.
- 13.2 Agent Indemnification. Agent shall indemnify and hold the Merchant and its officers, directors, employees, agents and representatives harmless from and against all claims, causes of action, demands, penalties, losses, liability, damage, or other obligations, including, without limitation, reasonable attorneys' fees and expenses, directly or indirectly asserted against, resulting from, or related to: (i) Agent's material breach of or failure to comply with any of its agreements, covenants, representations or warranties contained in any Agency Document; (ii) any claims by any party engaged by Agent as an employee or independent contractor arising out of such employment; (iii) any harassment or any other unlawful, tortious or otherwise actionable treatment of any customers, employees or agents

of the Merchant by Agent or any of its representatives; (iv) any consumer warranty, products liability claims or other claims of any type or kind and by any party whatsoever relating to Additional Agent Goods; (v) as set forth in section 8.3 above; (vi) any liability or other claims asserted by customers, any of Merchant's employees, or any other person against any Merchant Indemnified Party by reason of acts or omissions of Agent or Agent's representatives in connection with the Sale, and (vii) the gross negligence (including omissions) or willful misconduct of Agent, its officers, directors, employees, agents or representatives.

Section 14. <u>Defaults</u>. The following shall constitute "<u>Events of Default</u>" hereunder:

- (a) The Merchant or Agent shall fail to perform any material obligation hereunder if such failure remains uncured five (5) days after receipt of written notice thereof;
- (b) Any representation or warranty made by the Merchant or Agent proves untrue in any material respect as of the date made and, to the extent curable, continues uncured five (5) days after written notice to the defaulting party, save for and representations regarding Retail Price of the Merchandise or the representations in Section 11.1(o) hereof, in which case Agent's sole remedy shall be such adjustment to the Guaranteed Amount mutually acceptable to the Merchant and Agent;
- (c) The filing of a motion by any party to covert the Merchant's bankruptcy case to a case under another chapter of the Bankruptcy Code (other than chapter 11) or the filing of a motion by any party to appoint a chapter 11 trustee; or
- (d) The Sale is terminated prior to the Sale Termination Date or materially interrupted or impaired at any Store for any reason other than (i) an Event of Default by Agent, or (ii) any other material breach or action by Agent not authorized hereunder.

Upon an Event of Default, the non-defaulting party (in the case of (a) or (b) above), or Agent (in the case of (c) above) may in its discretion elect to terminate this Agreement, and any party's damages or entitlement to equitable relief on account of an Event of Default shall (in addition to the right to terminate as provided above) be determined by a court of competent jurisdiction.

- Section 15. Agent's Security Interest. In consideration of and effective upon payment by Agent of the Initial Guaranty Payment on the Payment Date and the delivery of the Letter of Credit to the Lender, as Merchant's designee, Merchant hereby grants to Agent first priority, senior security interests in and liens upon: (i) the Merchandise; (ii) to the extent of Merchant's ownership therein, the Additional Agent Goods; (iii) all Proceeds (including, without limitation, credit card Proceeds); (iv) the commission regarding the sale or other disposition of Merchant's Consignment Goods under Section 5.4 hereof; (v) the commission regarding the sale or other disposition of Owned FF&E under Section 7 hereof; (vi) any Sharing Amount, but only up to the amount of Agent's percentage share of such Sharing Amount under Section 3.2(a) hereof, and (vii) all "proceeds" (within the meaning of Section 9-102(a)(64) of the Code) of each of the foregoing (all of which are collectively referred to herein as the "Agent Collateral"), to secure the full payment and performance of all obligations of Merchant to Agent hereunder. Upon entry of the Approval Order, payment of the Initial Guaranty Payment, and delivery of the Letter of Credit to the Lender, the security interests and liens granted to the Agent hereunder shall be deemed properly perfected without the necessity of filing UCC-1 financing statements or any other documentation.
- (a) Without any further act by or on behalf of the Agent or any other party (including (without limitation) the Lender and Merchant), the Agent's security interests and liens in the Agent

Collateral created hereunder are (i) validly created, (ii) effective upon entry of the Approval Order, perfected, and (iii) senior to all other liens and security interests, provided, however, that (i) until the Merchant receives payment in full of the Guaranteed Amount, the Additional Agent Goods Fee, the Sharing Amount (if any), Expenses, the proceeds realized upon the sale of Owned FF&E (less the FF&E Commission) or Additional Guaranteed Amount, and such others amounts due to Merchant hereunder, the security interest granted to Agent hereunder shall remain junior and subordinate in all respects to the security interests of Lender in the Agent Collateral (other than the Additional Agent Goods and proceeds thereof in which Merchant has not property or other interest and Lender has no security interest or other lien), but solely to the extent and amount of the unpaid portion of the Guaranteed Amount, the Additional Agent Goods Fee, the Sharing Amount (if any), Expenses, the proceeds realized upon the sale of Owned FF&E (less the FF&E Commission) or the Additional Guaranteed Amount, and such others amountsdue to Merchant hereunder, and (ii) upon payment in full of the Guaranteed Amount, the Additional Agent Goods Fee, the Sharing Amount (if any), Expenses, the proceeds realized upon the sale of Owned FF&E (less the FF&E Commission) or the Additional Guaranteed Amount, and such others amountsdue to Merchant hereunder, any security interest or lien of the Lender in the Agent Collateral shall be junior and subordinate in all respects to the security interest and liens of Agent. Merchant shall cooperate with Agent with respect to all filings (including (without limitation) UUC-1 financing statements) and other actions to the extent reasonably requested by Agent in connection with the security interests and liens granted under this Agreement.

- (b) Merchant will not sell, grant, assign or transfer any security interest in, or permit to exist any encumbrance on, any of the Agent Collateral other than in favor of the Agent and Lender.
- (c) In the event of an occurrence of an Event of Default by the Merchant hereunder, in any jurisdiction where the enforcement of its rights hereunder is sought, the Agent shall have, in addition to all other rights and remedies, the rights and remedies of a secured party under the Code.
- (d) "<u>Code</u>" shall mean the Uniform Commercial Code as the same may be in effect from time to time in the State of Florida.

Section 16. Bidding and Auction. The bidding procedures order shall provide that, in consideration of Agent conducting its due diligence and entering into this Agreement (which may serve as a base by which other offers may be measured and is subject to higher and better offers by way of an auction process) ("Auction"); then in the event that this Agreement shall not be consummated with Agent because Merchant elects as a consequence of the Auction to accept or pursue an alternate bid or sale mechanism through the Bankruptcy Court (including, without limitation, one or more transactions for the purchase or any portion of Merchant's assets and business as a going concern or otherwise) (the "Alternate Transaction"), and provided that there is no Event of Default by Agent then pending hereunder and that each of the Break-Up Fee Conditions (hereinafter defined) are satisfied as determined by the Bankruptcy Court, Merchant shall pay Agent (on the first business day following (i) receipt of the initial guaranty payment in the case of a liquidation Alternate Transaction from the initial guaranty payment received from such Alternate Transaction and available for expenditure by the Merchant, (ii) entry of an order approving an Alternative Transaction other than a liquidation transaction from the Deposit (as defined below) received from such bidder and available for expenditure by the Merchant, or (iii) such later date as Agent may agree) an amount equal to the sum of (i) \$75,000 as a combined break-up fee and expense reimbursement to cover Agent's reasonable out-of-pocket costs and expenses associated with due diligence and professionals' fees and expenses associated with, among other things, negotiating, drafting, and obtaining approval of this Agreement and the bid procedures (the "Bid Protection"), and (ii) Agent's documented, actual out-of-pocket costs of signage and freight in an amount not to exceed \$350,000 (the

"Signage Costs"); provided, however, that (x) to participate in the Auction, a potential bidder shall be required to (1) post a deposit in an amount not less than the sum of the Bid Protection and the Signage Costs, and such other amount as the Merchant shall require (the "Deposit") and (2) irrevocably affirm in writing, if such potential bidder is the successful bidder, upon entry of the order approving the transaction with such successful bidder, the Merchant is authorized to, and the Merchant shall, use the Deposit to pay the Agent the Bid Protection and the Signage Costs from the Deposit, (y) if the Agent is not the "successful bidder" at the Auction and the successful bidder is the liquidating agent, the Agent agrees to make the signage available to the successful bidder upon receipt of the Bid Protections and Signage Costs from the Merchant. At the Auction: (i) the Bid Protections shall not be eligible to be "credit bid" by Agent; (ii) the first overbid by a competing bidder shall be an amount equal a Guaranty Percentage of not less than 28.5% (as quantified by the Merchant, acting reasonably, in the case of a going concern bid), with successive overbids in increments of not more than 0.1% absent Agent's consent; and (iii) all competing liquidation bids shall be on the same terms and conditions as this Agreement other than as to the increased Guaranty Percentage as determined by Merchant in its sole discretion. For the avoidance of all doubt, the requirements of clause (iii) of the immediately preceding sentence shall not apply to any going concern bids received by Merchant. For avoidance of doubt, Agent's right to receive payment of the Bid Protections and reimbursement of the Signage Costs as provided herein shall be the sole and exclusive remedy of Agent in the event of termination of the Agency Agreement to accept an Alternate Transaction. In the event Agent not the "successful bidder" at the auction, but is the "next highest bidder, Agent agrees to serve as the "back up" bidder and close the transaction contemplated by this Agreement, as may modified by the Agent's bidding at the Auction; provided the Sale Commencement Date occurs no later than May 7, 2015 (absent the Agent's express written consent, which may be withheld or granted in the Agent's sole and absolute discretion or modified at the Auction).

Section 17. Miscellaneous.

17.1 <u>Notices</u>. All notices and communications provided for pursuant to this Agreement shall be in writing and sent by email, by hand, by facsimile or by Federal Express or other recognized overnight delivery service, as follows:

If to the Agent: HILCO MERCHANT RESOURCES, LLC

5 Revere Drive, Suite 206 Northbrook, IL 60062 Attention: Ian S. Fredericks Tel: (847) 418-2075

Fax: (847) 897-0859

Email: <u>ifredericks@hilcotrading.com</u>

GORDON BROTHERS RETAIL PARTNERS, LLC

Prudential Tower 800 Boylston Street Boston, MA 02119 Attn: Michael Chartock Tel: 617.210.7116

Email: mchartock@gordonbrothers.com

If to the Merchant: Simply Fashion Stores, Ltd.

2110 NW 95th Avenue Miami, Florida 33172 Attn: Swapnil Shah Tel: 305.592.2712 Email: sjs@mahudi.com

With a copy to (which shall not constitute notice)

KapilaMukamal 1000 South Federal Highway, Suite 200 Fort Lauderdale, Florida 33316 Attn: Soneet R. Kapila, CRO

Tel: 954.761.1011

Email: skapila@kapilamukamal

With a copy to (which shall not constitute notice)

Berger Singerman LLP 1450 Brickell Avenue Miami, Florida 33131

Attn: Paul Steven Singerman

Tel: 305.714.4343

Email: <u>Singerman@bergersingerman.com</u> Attn: Christopher Andrew Jarvinen

Tel: 305.714.4363

Email: cjarvinen@bergersingerman.com

If to the Lender: JNS INVT, LLC

c/o Genovese Joblove & Battista, P.A. 100 Southeast Second Street, 44th Floor

Miami, Florida 33131 Attn: Paul J. Battista Tel: 305.349.2300

Email: pbattista@gjb-law.com

- 17.2 <u>Governing Law/Exclusive Jurisdiction</u>. This Agreement shall be governed by and interpreted in accordance with the laws of the State of Florida, without reference to any conflict of laws provisions thereof, except where governed by the Bankruptcy Code. Each of the parties hereto irrevocably and unconditionally submits, for itself and its properties, to the exclusive jurisdiction of the Bankruptcy Court, in any action or proceeding arising out of or relating to this Agreement.
- 17.3 <u>Amendments</u>. This Agreement, the Exhibits hereto, and the Agency Documents may not be modified except in a written instrument executed by each of the parties hereto and consented to in writing by the Lender (unless the obligations to the Lender are paid in full), which consent shall not be unreasonably withheld, conditioned or delayed.
- 17.4 <u>No Waiver</u>. No consent or waiver by any party, express or implied, to or of any breach or default by the other in the performance of its obligations hereunder shall be deemed or construed to be a consent or waiver to or of any other breach or default in the performance by such other party of the same or any other obligation of such party. Failure on the part of any party to complain of

any act or failure to act by the other party or to declare the other party in default, irrespective of how long such failure continues, shall not constitute a waiver by such party of its rights hereunder.

- 17.5 <u>Currency</u>. All reference to dollars in this Agreement and all schedules, exhibits, and ancillary documents related to this Agreement shall refer to US dollars.
- 17.6 <u>Successors and Assigns</u>. This Agreement shall inure to the benefit of and be binding upon Agent and Merchant and their respective successors and assigns, including, but not limited to, any chapter 11 or chapter 7 trustee; provided, however, that this Agreement may not be assigned by Merchant or Agent to any party without the prior written consent of the other (which consent may be granted or withheld in the applicable party's sole discretion).
- 17.7 Execution in Counterparts. This Agreement may be executed in one or more counterparts. Each such counterpart shall be deemed an original but all such counterparts together shall constitute one and the same agreement. This Agreement, to the extent signed and delivered by means of a facsimile machine, electronic mail or other electronic transmission in which the actual signature is evident, shall be treated in all manner and respects as an original agreement or instrument and shall be considered to have the same binding legal effect as if it were the original signed version thereof delivered in person. At the request of any party hereto, each other party hereto or thereto shall re-execute original forms hereof and deliver them to all other parties. No party hereto shall raise the use of a facsimile machine, electronic mail, or other electronic transmission in which the actual signature is evident to deliver a signature or the fact that any signature or agreement or instrument was transmitted or communicated through the use of a facsimile machine, electronic mail or other electronic transmission in which the actual signature is evident as a defense to the formation of a contract and each party forever waives such defense. In proving this Agreement, it shall not be necessary to produce or account for more than one such counterpart signed by the party against which enforcement is sought.
- 17.8 <u>Section Headings</u>. The headings of sections of this Agreement are inserted for convenience only and shall not be considered for the purpose of determining the meaning or legal effect of any provisions hereof.
- 17.9 <u>Wiring of Funds</u>. All amounts required to be paid by Agent or the Merchant under any provision of this Agreement shall be made by wire transfer of immediately available funds which shall be wired by Agent or Merchant, as applicable, no later as 2:00 p.m. (Eastern Time) on the date that such payment is due; provided, however, that all of the information necessary to complete the wire transfer has been received by Agent or Merchant, as applicable, by 10:00 a.m. (Eastern Time) on the date that such payment is due. In the event that the date on which any such payment is due is not a business day, then such payment shall be made by wire transfer on the next business day.
- 17.10 <u>Nature of Remedies</u>. Except to the extent expressly set forth herein, all rights, remedies, powers, privilege and adjustments under sections 3.1(b), 3.4 and 11.1(o) shall be in addition to and not in limitation of those provided elsewhere in this Agreement or by applicable law. No failure to exercise and no delay in exercising, on the part of the Agent, any right, remedy, power, privilege or adjustment hereunder, shall operate as a waiver thereof; nor shall any single or partial exercise of any right, remedy, power, privilege, or adjustment hereunder preclude any other or further exercise thereof or the exercise of any other right, remedy, power, privilege, or adjustment. For the avoidance of doubt and notwithstanding anything to the contrary in this Agreement, the rights granted to Agent pursuant to Section 16 shall be the Agent's sole and exclusive rights and remedies in the event of a termination of this Agreement by Merchant under Section 16.

17.11 <u>Entire Agreement</u>. This Agreement contains the entire agreement between the parties hereto with respect to the transactions contemplated hereby and supersedes and cancels all prior agreements, including, but not limited to, all proposals, letters of intent or representations, written or oral, with respect thereto.

IN WITNESS WHEREOF, the Agent and the Merchant hereby execute this Agreement by their duly authorized representatives as a sealed instrument as of the day and year first written above.

HILCO MERCHANT RESOURCES, LLC

By:			
Print	Name	and	– Title
GORDON	BROTHERS RETA	IL PARTNERS, I	LLC
By: Print			
Print	Name	and	Title

SIMPLY FASHION STORES, LTD.

Print Name and Title: SONEET R. KAPILA

C-R.O ,

Agreed and Accepted as to Sections

3.3, 3.4, 8.3, 8.8, 8.10, 15, 16, and 17: JNS INVT, LLC

Print Name and Title: Swapzil J. Shah, Lander

List of Exhibits

By:

Exhibit 1 - Store List.

Exhibit 3.1(b) – Merchandise Level Adjustment Schedule.

Exhibit 3.3(b) - Merchant's Designated Account.

Exhibit 3.4 – Form of Letter of Credit.

Exhibit 4.1(c) – Per Store, Per Diem Occupancy Expenses.

Exhibit 5.1(a) - Inventory Taking Instructions.

Exhibit 5.2(b) - Distribution Center

Exhibit 7.1 – Owned FF&E

Exhibit 11.1(o) - Inventory Mix

Exhibit 11.1(s) – Promotional Calendar

Exhibit 11.1(u) - Locations to be Closed

Exhibit 8.1 - Sale Guidelines

IN WITNESS WHEREOF, the Agent and the Merchant hereby execute this Agreement by their duly authorized representatives as a sealed instrument as of the day and year first written above.

	HILCO MERCHANT RESOURCES, LLC
	By: Print Name and Title: Ian Fredericks VP & Assistant General Counsel, Managing Member GORDON BROTHERS RETAIL PARTNERS, LLC
	By: Print Name and Title:
	SIMPLY FASHION STORES, LTD.
	By: Print Name and Title:
Agreed and Accepted as to Sec 3.3, 3.4, 8.3, 8.8, 8.10, 15, 16, an JNS INVT, LLC	
By:Print Name and Title:	List of Exhibits
E 1274 1 - 64 - 124	List of Lambies
Exhibit 1 – Store List. Exhibit 3.1(b) – Merchandise Level Adju Exhibit 3.3(b) – Merchant's Designated a Exhibit 3.4 – Form of Letter of Credit. Exhibit 4.1(c) – Per Store, Per Diem Occ	Account. supancy Expenses.
Exhibit 5.1(a) – Inventory Taking Instruction Exhibit 5.2(b) – Distribution Center Exhibit 7.1 – Owned FF&E Exhibit 11.1(o) – Inventory Mix	ctions.

Exhibit 11.1(s) – Promotional Calendar Exhibit 11.1(u) – Locations to be Closed

Exhibit 8.1 - Sale Guidelines

IN WITNESS WHEREOF, the Agent and the Merchant hereby execute this Agreement by their duly authorized representatives as a sealed instrument as of the day and year first written above.

HILCO MERCHANT RESOURCES, LLC

	By: Print Name and Title:
	By: Print Name and Title: Print Name and Title: SIMPLY FASHION STORES, LTD.
	By: Print Name and Title:
Agreed and Accepted as 3.3, 3.4, 8.3, 8.8, 8.10, 15, JNS INVT, LLC	
By:	
	<u>List of Exhibits</u>

Exhibit 1 – Store List.

Exhibit 3.1(b) – Merchandise Level Adjustment Schedule.

Exhibit 3.3(b) - Merchant's Designated Account.

Exhibit 3.4 – Form of Letter of Credit.

Exhibit 4.1(c) – Per Store, Per Diem Occupancy Expenses.

Exhibit 5.1(a) - Inventory Taking Instructions.

Exhibit 5.2(b) - Distribution Center

Exhibit 7.1 - Owned FF&E

Exhibit 11.1(o) - Inventory Mix

Exhibit 11.1(s) – Promotional Calendar

Exhibit 11.1(u) - Locations to be Closed

Exhibit 8.1 - Sale Guidelines

Store List

									Callina
Store #	Location Type	Name	Address	City	State	Zip	Phone	Open Date	Selling Sq. Ft.
12	SIMPLY FASHIONS	CHURCH SQUARE S/C	917 N. CAROLINE STREET	BALTIMORE	MD	21205	410/844-1426	7/18/14	2,910
14	SIMPLY FASHIONS	LEE STREET SHOPPING CTR	4523 LEE STREET	ALEXANDRIA	LA	71302	318/619-9940	05/29/98	3000
15	SIMPLY FASHIONS	INWOOD CENTRAL	8161 ANTOINE	HOUSTON	TX	77088	281/931-5450	03/15/02	3.272
16	SIMPLY FASHIONS	PEMBERTON SQUARE	3505 PEMBERTON SQUARE BLVD #43	VICKSBURG	MS	39180	601/629-9562	08/28/98	4,573
18	SIMPLY FASHIONS	GATEWAY SHOPPING CENTER	5238-14 NORWOOD AVENUE	JACKSONVILLE	FL	32208	904/768-1488	02/20/93	4532
19	SIMPLY FASHIONS	AMERICAN WAY PLAZA	4667 AMERICAN WAY	MEMPHIS	TN	38118	901/362-2545	03/19/93	4800
20	SIMPLY FASHIONS	JEFFERSON SQUARE S/C	2801 OLIVE ST STE. 8	PINE BLUFF	AR	71603	870/534-1432	05/28/93	4,672
21	DOTS	EISENHOWER CROSSING	4673 PRESIDENTIAL PKWY, SUITE M	MACON	GA	31206	478/960-3928	7/3/14	4,368
22	SIMPLY FASHIONS	FREESTANDING	69-73 W. CHELTEN AVENUE	PHILADELPHIA	PA	19144	215/435-1317	3/21/14	4,858
24	SIMPLY FASHIONS	CENTER CITY	1629 POPLAR AVE	MEMPHIS	TN	38104	901/274-9551	7/16/05	3,800
26	SIMPLY FASHIONS	TAMPA FESTIVAL CENTRE	2525 E. HILLSBOROUGH AVE. STE.101	TAMPA	FL	33610	813/237-5150	10/28/93	3,900
27	SIMPLY FASHIONS	GENTILLY S/C	3155 GENTILLY BLVD	NEW ORLEANS	LA	70122	504/286-0793	10/01/93	3,018
30	SIMPLY FASHIONS	MERIDIAN MARKETPLACE	2526 67TH AVE LOOP, #106	MERIDIAN	MS	39307	601/553-2990	5/2/08	3,067
31	DOTS	EASTGATE S/C	26200 EASTGATE BLVD	ROSEVILLE	MI	48066	586/675-0852	6/7/14	7,450
32	SIMPLY FASHIONS	FOWLERS PLAZA	1433 S. EISENHOWER PKWY	MACON	GA	31206	478/781-9672	5/11/07	4,300
33	SIMPLY FASHIONS	CHATHAM VILLAGE SQUARE	8658 S. COTTAGE GROVE AVE, UNIT #402	CHICIAGO	IL	60619	312/485-4518	1/31/15	3,060
36	SIMPLY FASHIONS	LINDELL MARKETPLACE	4167 LINDELL BLVD	ST. LOUIS	MO	63108	314/296-2106	01/30/15	3,850
37	SIMPLY FASHIONS	GENTILLY S/C	4800 CHEF MENTEUR HWY SUITE F	NEW ORLEANS	LA	70126	504/940-2255	10/05/10	3,000
39	SIMPLY FASHIONS	FREESTANDING	456 MEETING STREET	CHARLESTON	SC	29403	843/722-2526	06/30/00	3880
40	DOTS	BRICKTOWN SQUARE	6560 WEST FULLERTON AVE	CHICAGO	IL	60707	224/213-0590	1/31/15	6,000
41	DOTS	EAST FOREST PLAZA	5424 FOREST DRIVE	COLUMBIA	SC	29206	803/239-6545	1/30/15	4,723
43	SIMPLY FASHIONS	LINCOLN S/C	26150 GREENFIELD RD	OAK PARK	MI	48237	248/514-1483	7/25/14	7,800
44	SIMPLY FASHIONS	N/A	2111 HWY 82 EAST, SUITE C	GREENVILLE	MS	38710	662/334-3138	02/08/02	3,000
45	SIMPLY FASHIONS	NORTH EASTWOOD PLAZA	8939 EAST 38TH STREET	INDIANAPOLIS	IN	46226	317/897-8935	02/08/02	3,300
46	SIMPLY FASHIONS	REDFORD PLAZA	9189 TELEGRAPH RD, SPACE #260	REDFORD	MI	48239	313/255-2376	4/10/09	3,660
47	DOTS	SALEM CONSUMER SQUARE	5477 SALEM AVE	DAYTON	ОН	45426	937/823-4770	1/30/15	3,500
48	SIMPLY FASHIONS	SOUTHGATE SHOPPING CENTER	1631 GORDON HWY., SPACE 11	AUGUSTA	GA	30906	706/790-0669	05/27/94	6,570
49	SIMPLY FASHIONS	OVERBROOK PLAZA	5610 LANCASTER AVE UNIT #900 A&B	PHILADELPHIA	PA	19131	267/317-0453	11/26/14	4,370
52	DOTS	VICTORY CROSSING S/C	4010 A VICTORY BLVD	PORTSMOUTH	VA	23701	757/532-6617	1/30/15	5,250
53	SIMPLY FASHIONS	95TH & JEFFREY	2023 E. 95TH STREET	CHICAGO	IL	60617	773/933-8660	10/3/08	2,717
54	SIMPLY FASHIONS	SUPER 1 FOODS S/C	2600 WAGGONER AVE., STE. 208	SHREVEPORT	LA	71108	318/636-1354	03/27/04	12,006
55	SIMPLY FASHIONS	NORTHWOOD S/C	4445 NORTH STATE ST.	JACKSON	MS	39206	601/362-0420	04/28/04	7,135
57	SIMPLY FASHIONS	AMERICANA PLAZA	2141 AMERICANA BLVD, SUITE 110 & 111	ORLANDO	FL	32839	407/854-8336	2/20/09	2,800
58	SIMPLY FASHIONS	FAMILY DOLLAR BUILDING	1024 NORTH STATE STREET	CLARKSDALE	MS	38614	662/627-1040	8/27/03	3,000
59	SIMPLY FASHIONS	LINWOOD SQUARE S/C	3135 PROSPECT	KANSAS CITY	MO	64128	816/861-1015	7/15/94	8000
62	SIMPLY FASHIONS SIMPLY FASHIONS	MT. CLARE JUNCTION	1229 W. PRATT STREET, SUITE B 16743 TORRENCE AVE	BALTIMORE	MD IL	21233	410/244-8440	10/04/06	3,000
63		THE LANDINGS		LANSING		60438	708/418-3830	5/4/09	6,000
66 68	SIMPLY FASHIONS SIMPLY FASHIONS	N.W. JUNCTION S/C WYANDOTTE PLAZA	3188 W. NORTHSIDE DR. 7722 STATE AVE	JACKSON KANSAS CITY	MS KS	39213 66112	601/366-7365	9/25/89	2847 4.490
70	DOTS	OAKLAND POINTE PLAZA	332 N.TELEGRAPH RD	PONTIAC	MI	48341	913/302-7118 248/309-0487	4/4/14 1/30/15	4,490 8,600
				HUNTSVILLE			•		
72 75	SIMPLY FASHIONS DOTS	NORTHSIDE PLAZA MARCUS PLAZA	3018 MEMORIAL PKWY NW 624 WEST MAIN STREET	NORWICH	AL CT	35810 06360	256/859-8004 860/222-4126	2/14/09 11/14/14	4,000 3,850
75 76	DOTS	WATERTOWER PLAZA	40 WATERTOWER PLAZA	LEOMINSTER	MA	01453	774/364-2450	1/30/15	3,600
76 77	DOTS	GATEWAY MARKETPLACE	1355 W. 8 MILE RD	HIGHLAND PARK	MI	48203	248/228-7987	6/7/14	4,000
77 79	DOTS	WHITEHALL S/C	2180 MACARTHUR RD SPACE 6-A	WHITEHALL	PA	18052	484/523-0298	1/30/15	4,250
80	DOTS	ROUNDTREE PLACE	2477 ELLSWORTH RD	YPSILANTI	MI	48197	734/740-2991	1/30/15	4,230
82	SIMPLY FASHIONS	SOUTHLAND MALL	20505 S. DIXIE HWY, SUITE #829	CUTLER BAY	FL	33189	305/234-6131	5/1/14	5,373
85	SIMPLY FASHIONS	K-MART PLAZA	2107 SOUTH US HWY 1	FT PIERCE	FL	34950	561/489-8068	03/31/00	2500
0.5	5 ET LASITIONS	K III III I ENEM	2237 300111 03 11441 1	I ILINGE		34330	301/403 0000	03/31/00	2500

Store List

Store #	Location Type	Name	Address	City	State	Zip	Phone	Open Date	Selling Sq. Ft.
86	SIMPLY FASHIONS	BURLINGTON PLAZA	3753 AUSTELL RD, SUITE 160 & 170	AUSTELL	GA	30106	770/819-8150	6/26/10	3,200
91	SIMPLY FASHIONS	CHICAGO HEIGHTS HILLTOP PLAZA	1333 HILLTOP AVE, UNIT D	CHICAGO HEIGHTS	IL	60411	708/481-8730	12/12/13	3,708
92	DOTS	NEWPORT S/C	1765 MONMOUTH STREET	NEWPORT	KY	41071	859/414-4068	10/25/14	4,000
94	SIMPLY FASHIONS	PINE HILLS MARKETPLACE	5311 WEST COLONIAL DRIVE	ORLANDO	FL	32808	407/578-6079	02/12/03	3,600
95	SIMPLY FASHIONS	CAHOKIA VILLAGE S/C	1024 CAMP JACKSON ROAD	CAHOKIA	IL	62206	618/337-6106	11/11/94	2908
97	SIMPLY FASHIONS	THE SHOPS AT WHITE OAK VILLAGE	4501 SOUTH LABURNUM AVE SPACE 145	RICHMOND	VA	23231	804/222-2117	9/17/10	3,965
99	SIMPLY FASHIONS	GREENWOOD WEST	2304 HWY. 82 W.	GREENWOOD	MS	38930	662/453-7437	8/01/90	2880
100	DOTS	CHERRYDALE POINT S/C	1510 POINSETT HWY	GREENVILLE	SC	29609	864/270-3583	1/30/15	3,900
101	SIMPLY FASHIONS	NEWMARKET SOUTH S/C	105 NEW MARKET SQUARE EAST	NEWPORT NEWS	VA	23605	757/244-5041	2/10/95	4538
102	DOTS	FOOD 4 LESS S/C	6930 S. ASHLAND AVE, SUITE 3	CHICAGO	IL	60636	312/550-0866	6/10/14	3,900
104	SIMPLY FASHIONS	SOUTHLAKE MALL	1000 SOUTHLAKE CIRCLE, SUITE 2409	MORROW	GA	30260	404/615-5158	5/31/14	3,824
106	DOTS	RACEWAY PARK S/C	13053 ASHLAND AVE	CALUMET PARK	IL	60827	312/848-7624	5/31/14	4.000
107	SIMPLY FASHIONS	POINCIANA PLAZA	1225 W. 45TH STREET. SUITE #303 & #304	MAGNOLIA PARK	FL	33407	561/494-0705	2/25/2011	2,400
109	DOTS	HERITAGE PLACE S/C	12551 OLIVE BLVD	CREVE COEUR	MO	63141	314/304-4598	1/30/15	4,800
110	SIMPLY FASHIONS	POPLAR PLAZA S/C	4930 POPLAR LEVEL ROAD	LOUISVILLE	KY	40219	502/968-7435	2/28/95	5000
111	SIMPLY FASHIONS	LANGLEY SQUARE	39 WEST MERCURY BLVD	HAMPTON	VA	23669	757/723-1536	11/12/99	3440
114	SIMPLY FASHIONS	NORTH POINTE S/C	5430G NORTH TRYON STREET	CHARLOTTE	NC	28213	704/599-9777	05/17/96	3200
116	DOTS	SPENCER SQUARE	3552 SPENCER HWY	PASADENA	TX	77504	713/205-2976	1/30/15	4,000
118	DOTS	VILLAGE S/C	3604 VILLAGE COURT	GARY	IN	46408	219/742-8550	6/27/14	6,500
122	DOTS	LYNNHAVEN NORTH S/C	2720 N. MALL DR SUITE 200	VIRGINIA BEACH	VA	23452	757/266-1324	1/30/15	5,000
123	DOTS	CEDAR HILLS S/C	3606 BLANDING BLVD	JACKSONVILLE	FL	32210	904/404-6926	11/8/14	6.000
124	SIMPLY FASHIONS	ROEBUCK MARKETPLACE	9164 PARKWAY EAST	BIRMINGHAM	AL	35206	205/815-1173	8/15/08	3,000
126	SIMPLY FASHIONS	FREESTANDING	9 N 52ND ST	PHILADELPHIA	PA	19139	267/449-4409	1/30/15	6,336
127	SIMPLY FASHIONS	RIVER COMMONS	4021 BEHRMAN PLACE, SPACE 1-2	NEW ORLEANS	LA	70114	504/366-3967	05/01/04	3000
128	SIMPLY FASHIONS	EASTLAND HILL CENTER	605 GALLATIN ROAD	NASHVILLE	TN	37206	615/226-1066	06/30/00	4200
129	SIMPLY FASHIONS	WEST TOWN PLAZA	4263 WEST THIRD STREET	DAYTON	ОН	45417	937/263-6167	03/20/04	4,350
130	SIMPLY FASHIONS	GRAND BOULEVARD PLAZA	5401 S. WENTWORTH AVE., UNIT #7	CHICAGO	IL	60609	773/924-3379	5/01/95	3500
133	SIMPLY FASHIONS	CAPITOL PLAZA	2288 EAST SOUTH BLVD	MONTGOMERY	AL	36116	334/284-3433	2/28/93	3421
137	SIMPLY FASHIONS	PARKWAY PLAZA	2001 NW EVANGELIN THRUWAY, SUITE 400	LAFAYETTE	LA	70501	337/235-8980	04/04/03	3.247
138	SIMPLY FASHIONS	N/A	5963 PLANK ROAD, SUITE B	BATON ROUGE	LA	70805	225/355-1336	10/06/06	10,140
139	SIMPLY FASHIONS	DODGE PLAZA	7748 LANDOVER ROAD	LANDOVER	MD	20785	301/773-3691	11/10/95	2400
141	SIMPLY FASHIONS	FOOD WORLD PLAZA SOUTH	4200 MCFARLAND BOULEVARD	TUSCALOOSA	AL	35405	205/349-4360	01/01/92	2600
144	SIMPLY FASHIONS	FRAYSER PLAZA	2146 FRAYSER BOULEVARD	MEMPHIS	TN	38127	901/354-8886	04/10/98	3000
145	SIMPLY FASHIONS	IVES DAIRY CROSSING S/C	19925 NW 2ND AVE, BAY 1	MIAMI GARDENS	FL	33169	785/510-2707	10/3/14	3,000
148	SIMPLY FASHIONS	SHOPS AT NORTHEAST VILLAGE	14477 GRATIOT AVENUE	DETROIT	MI	48205	313/527-0282	03/01/01	6012
151	SIMPLY FASHIONS	TOWN & COUNTRY PLAZA	3300 N. PACE BLVD. #35	PENSACOLA	FL	32505	850/432-5711	01/01/92	4312
153	DOTS	SHOPPES AT GARNER	4446 FAYETTEVILLE RD	RALEIGH	NC	27603	919/368-1387	1/30/15	4.200
154	DOTS	NORTHLINE CROSSING S/C	4400 NORTH FREEWAY E 1000	HOUSTON	TX	77022	281/507-8484	1/31/15	4,200
155	SIMPLY FASHIONS	CANE RUN	3230 CRUMS LANE	LOUISVILLE	KY	40216	502/449-0883	02/23/96	3000
157	SIMPLY FASHIONS	EASTOVER S/C	5037 Indian Head Hwy	Oxon Hill	MD	20745	301/749-1435	4/26/96	3200
158	DOTS	STEELYARD COMMONS S/C	3495 STEELYARD DRIVE	CLEVELAND	OH	44109	216/212-2560	6/21/14	6,500
159	SIMPLY FASHIONS	WESTERN HILLS MALL	7201 AARON ARONOV DRIVE, STE. 36A	FAIRFIELD	AL	35064	205/925-6004	9/29/06	3,850
160	SIMPLY FASHIONS	THE SHOPPES OF LIBERTY CITY	1156 NW 54TH STREET	MIAMI	FL	33127	305/757-0797	08/24/00	3000
161	SIMPLY FASHIONS	BLUE PARKWAY TOWN CENTER	4307 EAST 50TH TERRACE, SPACE A & B	KANSAS CITY	MO	64130	816/921-2810	2/3/06	3,176
162	SIMPLY FASHIONS	GRIGGS ROAD S/C	4421 GRIGGS ROAD	HOUSTON	TX	77021	713/741-4604	2/15/03	5,000
163	SIMPLY FASHIONS	MADISON SQUARE	2585 MADISON AVENUE	MONTGOMERY	AL	36107	334/241-8995	10/24/08	4,000
164	SIMPLY FASHIONS	SELMA MALL		SELMA	AL	36703			3000
165			1391 HIGHLAND AVE., SUITE 113		MS		334/418-8884	03/13/96	
102	SIMPLY FASHIONS	FAMILY DOLLAR CENTER	3111 W. CAPITOL STREET	JACKSON	IVIS	39209	601/354-5564	02/27/04	3,297

Store List

Store #	Location Type	Name	Address	City	State	Zip	Phone	Open Date	Selling Sq. Ft.
166	SIMPLY FASHIONS	LANTANA VILLAGE SQUARE	1301 S. DIXIE HWY	LANTANA	FL	33462	561/586-1959	2/13/09	3,500
167	SIMPLY FASHIONS	AZALEA SHOPPING CENTER	2121 BEMISS ROAD	VALDOSTA	GA	31602	912/245-8040	3/15/96	2920
170	DOTS	TAYLOR COMMONS	1341 S. MAIN AVE SUITE 210	SCRANTON	PA	18504	570/209-0637	1/30/15	7,040
173	SIMPLY FASHIONS	EUTAW VILLAGE	2710 BRAGG BLVD	FAYETTEVILLE	NC	28303	910/321-9483	07/21/2000	6,500
174	SIMPLY FASHIONS	NORTHSIDE SHOPPING CENTRE	7900 NORTHWEST 27TH AVE., STE. 11B	MIAMI	FL	33147	305/691-1604	07/24/96	4,000
175	SIMPLY FASHIONS	LAMAR AIRWAYS S/C	2236 LAMAR AVENUE	MEMPHIS	TN	38114	901/454-9377	05/25/96	4033
176	DOTS	STONY ISLAND PLAZA	1627 E. 95TH ST	CHICAGO	IL	60617	312/914-1902	1/30/15	3,500
178	DOTS	WEST ALLIS	6822 W. GREENFIELD AVE. #A-114	WEST ALLIS	WI	53214	414-207-2971	03/13/15	-
180	SIMPLY FASHIONS	WEST RIDGE	3050 MARTIN LUTHER KING JR DR. SPACE F1	ATLANTA	GA	30311	404/699-7395	4/14/04	2,886
182	SIMPLY FASHIONS	LAUDERHILL MALL	1341 N.W. 40TH AVENUE	LAUDERHILL	FL	33313	954/587-2859	09/01/1989	3600
183	SIMPLY FASHIONS	SAVE A LOT CTR	1181 E MAIN STREET	COLUMBUS	ОН	43205	614/253-3779	5/21/08	3,000
184	DOTS	MODEL T PLAZA	14110 WOODWARD AVENUE	HIGHLAND PARK	MI	48203	248/915-8625	8/29/14	3,840
185	SIMPLY FASHIONS	KING SHOPPING CENTER	7057 MARTIN LUTHER KING DR	LANDOVER	MD	20785	301/773-5117	6/21/99	4,360
186	DOTS	SOUTHGATE USA S/C	21640 LIBBY ROAD	MAPLE HEIGHTS	ОН	44137	216/347-3418	1/31/15	4,400
187	DOTS	THRUWAY PLAZA	150 THRUWAY PLAZA DR	CHEEKTOWAGA	NY	14225	716/462-2133	1/30/15	4,978
188	DOTS	MIRACLE MILE S/C	4128 WILLIAM PENN HIGHWAY	MONROEVILLE	PA	15146	412/552-9383	11/17/14	4,700
189	SIMPLY FASHIONS	MARSHFIELD PLAZA	11612 S. MARSHFIELD AVE	CHICAGO	IL	60643	773/660-8840	9/18/09	3,000
190	DOTS	SOUTHERN PLAZA	4200 SOUTH EAST ST #31	INDIANAPOLIS	IN	46227	317/650-5238	1/30/15	5,003
191	DOTS	FONDREN SOUTHWEST VILLAGE	11168 FONDREN ROAD	HOUSTON	TX	77096	832/544-0371	8/29/14	5,500
192	SIMPLY FASHIONS	VILLAGE GREEN	1126 280 BYPASS	PHENIX CITY	AL	36867	334/298-1897	2/25/11	4,200
193	SIMPLY FASHIONS	N/A	3163 CLEVELAND AVENUE	COLUMBUS	ОН	43224	614/268-9112	08/15/96	4464
196	SIMPLY FASHIONS	DOLPHIN PLAZA	17235 NW 27TH AVENUE	MIAMI GARDENS	FL	33056	954/540-4395	9/12/14	3,022
197	SIMPLY FASHIONS	BOSSIER TOWN CENTER	3143 EAST TEXAS STREET	BOSSIER CITY	LA	71111	318/747-5568	10/01/89	3,568
198	SIMPLY FASHIONS	GATEWAY	3250 JACKSON AVE	MEMPHIS	TN	38122	901/458-7111	10/22/03	4,979
199	SIMPLY FASHIONS	BEL AIR S/C	8780 E. 8 MILE ROAD	DETROIT	MI	48234	313/892-1683	08/30/96	3300
200	DOTS	MARKETPLACE AT SMYRNA	801 INDUSTRIAL BLVD SUITE 140	SMYRNA	TN	37167	615/427-8693	1/31/15	5,000
201	SIMPLY FASHIONS	GALLERY AT WARREN CONNER	4905 CONNER AVE	DETROIT	MI	48215	313/824-6314	2/11/05	4,290
203	SIMPLY FASHIONS	ELLIS AVENUE PLAZA	1335 ELLIS AVENUE, #15	JACKSON	MS	39204	601/949-8448	8/1/97	3071
204	SIMPLY FASHIONS	RIVERVIEW PLAZA	264 WEST GENESSEE STREET	SAGINAW	MI	48602	989/753-1260	11/05/04	3,000
205	SIMPLY FASHIONS	CULLEN PLAZA	9426 CULLEN BLVD	HOUSTON	TX	77051-3321	713/738-8552	03/08/02	3000
207	SIMPLY FASHIONS	THE COMMONS ROUTE 20	4417 WEST 5TH AVENUE	GARY	IN	46406	219/949-6428	09/25/04	4.032
208	DOTS	JEFFERSON VILLAGE	11226 JEFFERSON AVENU	DETROIT	MI	48214	313/815-0246	11/7/14	3,600
212	SIMPLY FASHIONS	ROYAL TOWN CENTER	8900 B WEST EIGHT MILE ROAD	FERNDALE	MI	48220	248/546-6565	03/07/03	3,612
213	SIMPLY FASHIONS	COLLEGE PARK COMMONS	7940 WEST OUTER DRIVE	DETROIT	MI	48235	313/533-7030	6/5/09	3,602
215	SIMPLY FASHIONS	SOUTH SLAPPEY VILLAGE	1030 WEST GORDON AVENUE, SUITE F	ALBANY	GA	31701	229/446-9956	9/12/97	5840
216	SIMPLY FASHIONS	WALNUT HILLS PLAZA	2047 CRATER RD #104	PETERSBURG	VA	23805	804/862-8529	09/24/99	4680
218	DOTS	SHOPS AT WHITE OAK VILLAGE	4501 SOUTH LABURNUM AVE SUITE 560	RICHMOND	VA	23231	757/603-7020	1/30/15	4.552
219	DOTS	COUNTRY CLUB PLAZA	490 LINCOLN ST	WORCESTER	MA	01613	508/735-9290	1/31/15	5,355
221	SIMPLY FASHIONS	THE CROSSINGS AT HALLS FERRY	10835 OLD HALLS FERRY RD, STE.A	ST. LOUIS	MO	63136	314/867-7885	03/02/99	3000
223	DOTS	CHICAGO	741 & 743 EAST 47TH STREET	CHICAGO	IL	60653	773-717-0299	02/27/15	-
224	SIMPLY FASHIONS	SQUARE 67	2550 RED BIRD LANE, SUITE 207	DALLAS	TX	75237	214/330-9354	08/26/98	3720
225	SIMPLY FASHIONS	GREENS CROSSROADS	221 WEST GREENS ROAD	HOUSTON	TX	77067	281/872-4744	9/25/09	3,915
227	SIMPLY FASHIONS	PLAZA ON THE BLV	8035 WEST FLORISSANT AVE, SUITE F & G	JENNINGS	MO	63136	314/385-9597	11/3/06	3,200
228	DOTS	NORTHGATE S/C	339 SQUIRE ROAD	REVERE	MA	02151	617/716-9973	11/22/14	4,200
229	SIMPLY FASHIONS	JOYLAND S/C	15550 JOY ROAD	DETROIT	MI	48228	313/838-3836	2/27/98	3025
230	SIMPLY FASHIONS	SILVER MILL COURT S/C	6115 NORTH TEUTONIA AVE	MILWAUKEE	WI	53209	414/527-4252	01/31/03	4,300
231	SIMPLY FASHIONS	MARLOW HEIGHTS S/C	3923 BRANCH AVENUE	TEMPLE HILLS	MD	20748	301/316-2020	6/25/98	2979
233	SIMPLY FASHIONS	WESTPORT COMMONS S/C	3264 W. 87TH STREET	CHICAGO	IL	60652	708/710-9119	9/19/14	3.500
233	SHALL I WOLLOWS	VVESTI ONI COMMUNICIVIS 3/C	3204 W. 07 III 3 II LEI	CITICAGO	i.e.	00032	,00,710-3119	2/12/14	3,300

Store List

Store #	Location Type	Name	Address	City	State	Zip	Phone	Open Date	Selling
	••			<u> </u>				.,	Sq. Ft.
234	SIMPLY FASHIONS	TOWNE SOUTH	2525 SOUTH MONROE ST. STE. 20 & 21	TALLAHASSEE	FL	32301	850/656-7602	02/13/03	3,500
236	SIMPLY FASHIONS	LAWNDALE PLAZA	3230 WEST ROOSEVELT ROAD	CHICAGO	IL	60624	773/638-1565	11/04/98	3200
237	SIMPLY FASHIONS	BALTIMORE	2331 CLEANLEIGH DR.	BALTIMORE	MD	21234	443.826.8891	03/27/15	-
239	SIMPLY FASHIONS	SOUTH WIND PLAZA	5335 N. MILITARY TRAIL SPACE 31&32	WEST PALM BEACH	FL	33407	561/684-5399	04/10/98	3600
240	SIMPLY FASHIONS	BUCKNER COMMONS	9208 EAST RL THORNTON FREEWAY, STE.110	DALLAS	TX	75228	214/321-3733	02/27/04	4,500
241	DOTS	TINLEY PARK PLAZA	15967 SOUTH HARLEM AVE	TINLEY PARK	IL	60477	312/718-1235	1/30/15	5,000
242	SIMPLY FASHIONS	SOUTH PHILADELPHIA S/C	2426 SOUTH 23RD STREET	PHILADELPHIA	PA	19145	215/764-0260	3/1/14	2,380
246	SIMPLY FASHIONS	CITY PLAZA	3447 UNION BLVD	ST. LOUIS	MO	63115	314/381-9088	11/09/98	3000
247	SIMPLY FASHIONS	SPEEDWAY SUPERCENTER	5824 CRAWFORDSVILLE RD	INDIANAPOLIS	IN	46224	314/240-0334	11/17/06	3,600
248	SIMPLY FASHIONS	FREESTANDING	2000 E. 71ST STREET	CHICAGO	IL	60649	773/324-9770	4/16/10	4,320
249	DOTS	CROSSROADS CENTER	6777 SPENCER HWY	PASADENA	TX	77505	832/392-5687	1/30/15	3,000
252	SIMPLY FASHIONS	JEWEL OSCO S/C	117 WEST 87TH STREET	CHICAGO	IL	60620	773/874-6063	05/12/98	6562
254	DOTS	30TH STREET S/C	3034 CROMER AVE	CANTON	ОН	44709	234/804-2546	1/30/15	3,279
255	DOTS	MANCHESTER PARKADE	370 WEST MIDDLE TURNPIKE	MANCHESTER	CT	06040	860/306-5958	2/11/15	4,691
256	DOTS	SOUTHFIELD	20176 W. 8 MILE ROAD	SOUTHFIELD	MI	48075	248-945-0282	02/27/15	-
257	SIMPLY FASHIONS	Baseline Plaza	20212 W EIGHT MILE RD	SOUTHFIELD	MI	48075	248/945-0282	09/25/98	2800
258	SIMPLY FASHIONS	SHOPPES ON MAIN	3662 EAST MAIN STREET, UNIT D	COLUMBUS	ОН	43213	614/231-2324	4/22/11	3,000
259	SIMPLY FASHIONS	SOUTH DEKALB PLAZA	2774 CANDLER RD	DECATUR	GA	30034	404/212-3880	6/23/99	3000
262	SIMPLY FASHIONS	MEADOWBROOK EAST	6707 MEADOWBROOK DRIVE	FT. WORTH	TX	76112	817/457-0580	9/19/03	4,875
263	SIMPLY FASHIONS	CHARITON SQUARE	4236 SOUTH BROADWAY	ST. LOUIS	MO	63111	314/832-9969	09/19/03	2,160
264	SIMPLY FASHIONS	LAMAR CROSSING	2938 LAMAR AVE, SUITE 116	MEMPHIS	TN	38114	901/743-8466	5/19/06	4,264
265	SIMPLY FASHIONS	WYOMING VILLAGE	1242 28TH STREET SW	WYOMING	MI	49509	616/531-6051	6/20/08	4,650
266	SIMPLY FASHIONS	STATE STREET	2524 STATE STREET	EAST ST. LOUIS	IL	62205	618/875-5461	03/31/00	3000
268	SIMPLY FASHIONS	EASTLAND CENTER	18000 VERNIER RD SPACE #741	HARPER WOODS	MI	48225	313/527-1949	12/10/10	4,154
269	SIMPLY FASHIONS	ST. STEPHENS SQUARE	2312 ST. STEPHENS RD. SUITE 6	MOBILE	AL	36617	334/452-9331	06/23/00	2,977
270	SIMPLY FASHIONS	HIGHLAND LAKES CENTER	7459 WEST COLONIAL DRIVE	ORLANDO	FL	32818	407/298-5200	6/3/11	3,032
271	DOTS	CARRIAGE CROSSING MARKETPLACE	10233 EAST SHELBY DR	COLLIERVILLE	TN	38017	901/229-7596	1/30/15	4,080
272	SIMPLY FASHIONS	ERDMAN PLAZA	3933 ERMAN AVE	BALTIMORE	MD	21213	410/276-1441	3/28/97	4,000
273	DOTS	NORTHWEST CROSSING	5762 HOLLISTER STREET	HOUSTON	TX	77040	281/543-9144	1/30/15	4,000
274	SIMPLY FASHIONS	WOODLAND VILLAGE	6046-6048 WOODLAND AVENUE	PHILADELPHIA	PA	19142	215/990-8309	11/20/14	2,964
275	SIMPLY FASHIONS	CROSS COUNTRY PLAZA	3201 MACON ROAD	COLUMBUS	GA	31906	706/569-5800	05/04/02	3,906
278	SIMPLY FASHIONS	EDENS PLAZA	4105 WEST BELTLINE BLVD #50	COLUMBIA	SC	29204	803/251-2445	1/29/10	3,000
281	SIMPLY FASHIONS	WHITEHAVEN PLAZA	4275 ELVIS PRESLEY BLVD	MEMPHIS	TN	38116	901/332-2273	05/04/02	3,410
282	SIMPLY FASHIONS	RACEWAY PARK S/C	13065 S. ASHLAND AVENUE	CALUMET PARK	IL	60827	312/502-5380	10/03/14	4,000
283	SIMPLY FASHIONS	ALBANY CROSSING	244 CORDELE RD, SUITE 125	ALBANY	GA	31705	229/883-7722	5/13/11	2,800
284	SIMPLY FASHIONS	OAKLAND POINTE	362 TELEGRAPH ROAD	PONTIAC	MI	48341	248/333-3131	9/15/06	5,340
285	SIMPLY FASHIONS	ONE & OLNEY SQUARE	5675 NORTH FRONT ST #100	PHILADELPHIA	PA	19120	267/353-3188	4/18/14	4,042
287	DOTS	NORTHWEST CROSSING	130 BLACKHORSE PIKE	AUDUBON	NJ	08106	609/314-5040	1/30/15	4,037
288	SIMPLY FASHIONS	CLEVELAND AVENUE CROSSING	854 CLEVELAND AVENUE, SUITE 108	EAST POINT	GA	30344	404/780-4596	10/22/14	3,000
301	SIMPLY FASHIONS	CLOVERLEAF S/C	1814-A N. FRONTAGE ROAD	MERIDIAN	MS	39301	601/693-6801	01/01/92	7380
302	SIMPLY FASHIONS	BATESVILLE S/C	103 KEATING ROAD	BATESVILLE	MS	38606	662/563-0217	01/01/92	3000
303	SIMPLY FASHIONS	NORTHTOWN PLAZA	5510 NORTH FREEWAY	HOUSTON	TX	77076	713/697-3451	7/05/90	3835
317	SIMPLY FASHIONS	HAMMOND AIRE PLAZA	9626 AIRLINE HIGHWAY, STE. C-3	BATON ROUGE	LA	70827	225/927-6216	03/30/93	6000
327	SIMPLY FASHIONS	WALZEM PLAZA	5334 WALZEM ROAD	SAN ANTONIO	TX	78218	210/656-7141	01/01/92	4400
337	DOTS	TOWN & COUNTRY	3816 E. BROAD STREET	COLUMBUS	OH	43213	740/919-1112	8/29/14	5,887
338	DOTS	SOUTH PHILADELPHIA S/C	2300 W. PASSYUNK AVE	PHILADELPHIA	PA	19145	267/670-1143	8/1/14	4,428
343	DOTS	EASTOWN S/C	3954 LINDEN AVE	DAYTON	ОН	45432	937/307-9240	8/29/14	4,045
344	DOTS	NORTHERN LIGHTS	3487 CLEVELAND AVE	COLUMBUS	ОН	43224	614/802-7240	8/29/14	4,500
							•		

Store List

Store #	Location Type	Name	Address	City	State	Zip	Phone	Open Date	Selling Sq. Ft.
348	DOTS	CENTURY CENTER	373 MEMORIAL AVENUE	W. SPRINGFIELD	MA	01089	413/313-4745	11/8/14	3,550
360	SIMPLY FASHIONS	FIVE POINTS WEST S/C	2209 BESSEMER ROAD	BIRMINGHAM	AL	35208	205/780-4206	08/30/90	4000
365	SIMPLY FASHIONS	CROSS CREEK	5648 GEORGETOWN ROAD	INDIANAPOLIS	IN	46254	317/388-0444	09/19/03	2,400
369	SIMPLY FASHIONS	FREESTANDING	2419 MURPHY MILL RD	DOTHAN	AL	36303	334/792-7332	10/15/2010	4.030
375	SIMPLY FASHIONS	ROSE PLAZA	4025 S. BROADWAY AVENUE	TYLER	TX	75701	903/581-2778	4/30/92	2700
377	DOTS	LINCOLN S/C	26130 GREENFIELD ROAD	OAK PARK	MI	48237	313/410-4264	8/29/14	4,800
500	SIMPLY FASHIONS	ANDERSON ROAD PLAZA	2600 ANDERSON ROAD	GREENVILLE	SC	29611	864/269-1204	01/01/92	5,900
501	SIMPLY FASHIONS	METRO PLAZA	1634 EAST 63RD STREET	KANSAS CITY	MO	64110	816/363-0780	12/15/12	3,000
504	SIMPLY FASHIONS	HIGHLAND PARK PLACE	14515 SOUTH WOODWARD AVENUE	DETROIT	MI	48203	313/865-6550	08/04/05	5111
506	DOTS	LIVONIA MARKETPLACE	29508 SEVEN MILE ROAD	LIVONIA	MI	48152	248/412-3564	10/24/14	4,000
507	SIMPLY FASHIONS	EDISTO VILLAGE SHOPPING CENTER	403 RIVERSIDE DRIVE S.W.	ORANGEBURG	SC	29115	803/533-0700	07/09/05	3328
508	DOTS	WESTSHORE PLAZA	1795 E. SHERMAN BLVD	MUSKEGON	MI	49444	231/215-6635	02/13/15	6,400
509	DOTS	KLINE PLAZA	130 B KLINE VILLAGE	HARRISBURG	PA	17104	717/317-3256	11/19/14	5,200
510	DOTS	CLINTON POINTE	33816 GRATIOT AVENUE	CLINTON TOWNSHIP	MI	48035	586/348-2803	10/24/14	3,250
511	SIMPLY FASHIONS	OAK STREET SHOPPING CENTER	845 OAK STREET	ATLANTA	GA	30310	404/753-0408	3/9/1995	3420
512	DOTS	NORGATE PLAZA	7235 N. KEYSTONE AVE. SUITE H	INDIANAPOLIS	IN	46240	317/448-3811	11/8/14	4,500
513	SIMPLY FASHIONS	MLK PLAZA	1300 NORTH GRAND AVE	SAINT LOUIS	MO	63113	314-535-5758	6/13/2003	3000
514	SIMPLY FASHIONS	NORTH CHARLESTON CENTER	5900 RIVERS AVE-SUITE H	NORTH CHARLESTON	SC	29406	843-554-0730	4/1/1998	3200
515	SIMPLY FASHIONS	SUMTER SQUARE SHOPPING CENTER	1009 BROAD STREET	SUMTER	SC	29150	803/773-0972	5/28/1992	2875
516	DOTS	COTTMAN & BUSELTON S/C	2121 COTTMAN AVE	PHILADELPHIA	PA	19149	267/572-0157	1/31/15	4,426
520	SIMPLY FASHIONS	BORDEAUX SHOPPING CENTER	3200 CLARKSVILLE HIGHWAY	NASHVILLE	TN	37208	615/726-1184	5/13/1992	3000
523	SIMPLY FASHIONS	WESTLAND SHOPPING CENTER	31284 MICHIGAN AVENUE	WESTLAND	MI	48185	734/728-3634	3/29/2000	3000
524	DOTS	THE SHOPS AT STERLING PONDS	33329 VAN DYKE AVE	STERLING HEIGHTS	MI	48312	586/212-3674	10/24/14	4,803
525	SIMPLY FASHIONS	NORTHWEST PLAZA	3269 W. SIEBENTHALER AVE., UNIT 5	DAYTON	OH	45406	937/274-0096	1/28/1996	2864
526	SIMPLY FASHIONS	DORCHESTER SQUARE	4391 DORCHESTER ROAD	NORTH CHARLESTON	SC	29405	843-566-0028	08/01/05	2970
527	DOTS	CANTON CENTRE	4060 TUSCARAWAS STREET W.	CANTON	OH	44708	234/850-4204	10/3/14	5,086
528	SIMPLY FASHIONS	GEORGETOWN PLAZA SHOPPING CENTER	4941 WEST 38TH STREET	INDIANAPOLIS	IN	46254	317/328-8958	4/1/1998	4200
529	SIMPLY FASHIONS	EASTGATE SHOPPING CENTER	3134 LOUISVILLE AVENUE	MONROE	LA	71201	318/362-1206	7/31/2003	4500
531	DOTS	MIDDLESEX S/C	1310 EASTERN BLVD	BALTIMORE	MD	21221	410/999-5097	10/25/14	3,000
535	SIMPLY FASHIONS	CRICHTON PLAZA	3038 A SPRINGHILL AVE.	MOBILE	AL	36607	251/479-5811	8/24/1995	3300
537	SIMPLY FASHIONS	CHESAPEAKE CROSSING SHOPPING CENTER	1949 S. MILITARY HWY.	CHESAPEAKE	VA	23320	757/545-1863	6/27/1999	3000
538	SIMPLY FASHIONS	CROSSROADS PLAZA	1958 STONE ROSE DRIVE	ROCKY MOUNT	NC	27802	252/446-6001	5/26/1993	4000
539	SIMPLY FASHIONS	HATTIESBURG SHOPPING CENTER	5891 U.S. HIGHWAY 49	HATTIESBURG	MS	39402	601/268-3725	10/23/2000	2700
541	DOTS	LIBERTY CENTER	3728 LIBERTY STREET	ERIE	PA	16508	814/450-9296	11/25/14	5.796
543	SIMPLY FASHIONS	LEHMBERG CROSSING	993 EAST ALABAMA STREET	EAST COLUMBUS	MS	39702	662/329-3237	4/26/2001	3200
544	DOTS	SHOP CITY PLAZA	179 SHOP CITY PLAZA	SYRACUSE	NY	13206	315/412-5800	10/10/14	4,959
545	DOTS	GREAT SOUTHERN S/C	3831 SOUTH HIGH STREET	COLUMBUS	ОН	43207	614/867-2491	10/3/14	4,500
549	SIMPLY FASHIONS	FREEDOM SQUARE SHOPPING CENTER	1615 SOUTH IRBY STREET	FLORENCE	SC	29505	843-661-7847	10/26/1995	3440
554	DOTS	TOWN CENTER MARKET	2821 CHASTAIN MEADOWS PKWY NW #120	MARIETTA	GA	30066	404/858-2805	10/6/14	4.800
555	SIMPLY FASHIONS	IROQUOIS MANOR	5326 SOUTH THIRD STREET	LOUISVILLE	KY	40214	502/368-7228	9/27/1990	3360
556	SIMPLY FASHIONS	WESLEY CHAPEL SQUARE SHOPPING CENTER	2389 WESLEY CHAPEL ROAD	DECATUR	GA	30035	770/593-0905	8/02/2005	2800
558	SIMPLY FASHIONS	SHELBY PLAZA	4673 ELVIS PRESLEY BLVD.	MEMPHIS	TN	38116	901/396-7433	7/28/1993	2800
559	DOTS	AUSTINTOWN PLAZA	6000 MAHONING AVENUE #252	YOUNGSTOWN	OH	44515	330/369-6989	10/29/14	4,500
562	SIMPLY FASHIONS	NORTH POINT CENTER	3409 DICKERSON PIKE, STE 106	NASHVILLE	TN	37207	615/227-8765	5/13/2004	4000
563	DOTS	CONSUMER SQUARE WEST	3650 SOLDANO BLVD	COLUMBUS	OH	43228	516/507-0202	2/13/15	3,685
565	SIMPLY FASHIONS	BROADWAY CENTER	2043-A BROADWAY CENTER	KNOXVILLE	TN	43228 37917	865/637-7025	8/4/1994	3,685
571	SIMPLY FASHIONS	WESTWOOD SHOPPING CENTER	3709 JEWELLA AVE	SHREVEPORT	LA	71109	318/631-0970	4/19/2000	4055
575	SIMPLY FASHIONS	VILLAGE SHOPPING CENTER	3552 VILLAGE COURT	GARY	IN	46408	219/884-5180	6/14/2000	4550
3/3	SIIVIPLI FASHIUNS	VILLAGE SHOPPING CENTER	3332 VILLAGE COURT	OUVI	IIN	40406	413/004-3180	0/14/2000	4550

			Store List						
Store #	Location Type	Name	Address	City	State	Zip	Phone	Open Date	Selling Sq. Ft.
576	SIMPLY FASHIONS	SOUTHGATE SHOPPING CENTER	4439 WEST FUQUA ROAD	HOUSTON	TX	77045	713/413-3397	9/26/2000	3300
577	SIMPLY FASHIONS	HICKORY RIDGE CROSSING	3641 HICKORY HILL	MEMPHIS	TN	38115	901/566-9443	8/05/00	4000
581	SIMPLY FASHIONS	GAINSVILLE SHOPPING CENTER	1208 NORTH MAIN STREET, UNIT 1208	GAINESVILLE	FL	32608	352/264-7990	7/9/05	3000
584	SIMPLY FASHIONS	SOUTHGATE PLAZA	303-B SOUTH SLAPPEY DRIVE	ALBANY	GA	31701	229/446-2340	4/25/2002	3456
595	SIMPLY FASHIONS	PARKWAY PLAZA	1223 SILAS CREEK PARKWAY	WINSTON-SALEM	NC	27127	336/724-3676	6/24/06	5,630
599	SIMPLY FASHIONS	COURTHOUST SQUARE	100 WEST ARIZONA AVE	RUSTON	LA	71270	318/255-5955	6/17/05	3,315
617	SIMPLY FASHIONS	ASHER	6420 COLONEL GLENN ROAD, SUITE 5	LITTLE ROCK	AR	72204	501/568-7334	7/1/05	2,400
247	_							•	5,125

Merchandise Threshold Schedule

Retail Value	Adjustment Points	Adjusted Guaranty
16,000,000		27.50%
15,900,000	0.16%	27.34%
15,800,000	0.16%	27.18%
15,700,000	0.16%	27.02%
15,600,000	0.17%	26.85%
15,500,000	0.17%	26.68%
15,400,000	0.18%	26.50%
15,300,000	0.18%	26.32%
15,200,000	0.19%	26.13%
15,100,000	0.19%	25.94%
15,000,000	0.20%	25.74%

Note(s):

^{1.} Adjustments between the increments shall be on a prorata basis.

^{3.} In the event that the Retail value of the Merchandise is less than \$15,000,000, each \$100,000 (or pro rata portion thereof) increment shall decrease the Guaranty by 0.23%.

Store #	Store Name	RENT	AD VALOREM TAX	FILTER SERVICE	INSURANCE - BUSINESS	LICENSE - CITY	LICENSE- STATE & COUNTY	MAINTENANC E	SECURITY	TELEPHONE	UTILITIES	WASTE PICK UP	Total Occupancy	% Rent
12	CHURCH SQUARE S/C	129	-	-	3	1	1	2	-	-	21	-	157	
14	LEE STREET SHOPPING CTR	49	2	-	4	-	2	3	1	. 2	23	3	89	
15	INWOOD CENTRAL	-	3	-	4	-	0	4	1	. 2	17	2	32	5%
16	PEMBERTON SQUARE	55	2	-	4	0	-	4		2	16	-	83	
18	GATEWAY SHOPPING CENTER	82	6	-	4	0	1	2	1	. 2	42	-	140	
19	AMERICAN WAY PLAZA	115	1	-	4	1	1	42	1	. 2	28	3	198	
20	JEFFERSON SQUARE S/C	66	2	-	4	0	-	1	1	. 2	17	7	99	
21	EISENHOWER CROSSING	166	17	-	8	2	-	26		1	27	1	249	
22	FREESTANDING	86	-	-	3	0	-	4	1	. 2	44	8	148	
24	CENTER CITY	136	16	-	4	1	1	13	1	. 2	18	3	195	
26	TAMPA FESTIVAL CENTRE	115	21	-	10	1	0	24	1	. 2	25	8	208	
27	GENTILLY S/C	167	16	-	4	3	0	34	1	. 2	18	6	251	
30	MERIDIAN MARKETPLACE	66	23	-	7	0	-	6	1	. 2	23	3	130	
31	EASTGATE S/C	85	-	-	4	-	-	4	1	. 1	50	2	146	
32	FOWLERS PLAZA	-	2	-	4	1	-	8	1	. 2	24	3	43	3%
33	CHATHAM VILLAGE SQUARE	275	8	-	5	1	0	13	1	. 2	24	2	331	
36	LINDELL MARKETPLACE	122	8	-	5	1	0	13	1	2	24	2	178	
37	GENTILLY S/C	115	14	-	12	2	0	17	1	. 2	19	-	181	
39	FREESTANDING	79	28	-	4	2	-	8	1	. 2	41	4	168	
40	BRICKTOWN SQUARE	304	8	-	5	1	0	13	1	. 2	24	2	360	
41	EAST FOREST PLAZA	193	8	-	5	1	0	13	1	. 2	24	2	250	
43	LINCOLN S/C	-	29	-	30	1	-	32		1	36	2	131	4%
44	N/A	73	10	-	5	0	-	7	1	. 2	22	3	122	
45	NORTH EASTWOOD PLAZA	-	1	-	4	0	-	3	1	. 2	21	2	34	5%
46	REDFORD PLAZA	121	13	-	8	-	-	30	1	. 2	34	2	212	
47	SALEM CONSUMER SQUARE	-	8	-	5	1	0	13	1	. 2	24	2	56	5%
48	SOUTHGATE SHOPPING CENTER	104	1	-	4	0	-	4	1	. 2	39	2	157	
49	OVERBROOK PLAZA	122	5	-	1	1	-	3		-	31	-	163	
52	VICTORY CROSSING S/C	184	8	-	5	1	0	13	1	. 2	24	2	240	
53	95TH & JEFFREY	75	23	-	4	0	-	24	1	. 2	15	4	148	
54	SUPER 1 FOODS S/C	115	7	-	4	3	0	5	1	. 2	32	4	173	
55	NORTHWOOD S/C	137	4	-	4	0	-	4	1	. 2	38	-	190	
57	AMERICANA PLAZA	-	6	-	9	-	0	14	1	. 2	19	2	53	4%
58	FAMILY DOLLAR BUILDING	46	3	-	4	0	-	3	1	4	26	3	89	
	LINWOOD SQUARE S/C	187	1	-	4	1	-	9	1	. 2	68	-	272	
62	MT. CLARE JUNCTION	-	5	-	4	0	1	5	1	. 2	19	-	37	7%
	THE LANDINGS	190	-	-	4		-	-	1	. 2	17	3	216	
66	N.W. JUNCTION S/C	83	8	-	6	0	-	17	1	. 2	17	2	136	
68	WYANDOTTE PLAZA	98	-	-	3		-	4	1	. 2	27	3	138	
70	OAKLAND POINTE PLAZA	204	8	-	5		0		1	. 2	24	2	260	
72	NORTHSIDE PLAZA	93	1	-	4		0	4	1	. 2	33	3	141	
75	MARCUS PLAZA	36	-	-	4		-	12		-	9	0	61	
76	WATERTOWER PLAZA	203	8	-	5		0	13	1	. 2	24	2	260	
	GATEWAY MARKETPLACE	203	49	-	5		-	-	1	1	22	-	282	
79	WHITEHALL S/C	220	8		5		0		1	. 2	24	2	276	
80	ROUNDTREE PLACE	178	8		5		0		1	. 2	24	2	234	
82	SOUTHLAND MALL	121	8	-	4		1	8		1	27	-	170	
85	K-MART PLAZA	49	8	-	5		0	13	1	. 2	24	2	106	
86	BURLINGTON PLAZA	82	7	-	5		1	10	1	. 2	27	2	136	
91	CHICAGO HEIGHTS HILLTOP PLAZA	137	-	-	4	_	-	10	1	. 2	8	3	166	
92	NEWPORT S/C	56	-		3			3		1	8	-	71	
94	PINE HILLS MARKETPLACE	58	4		4		0		1	. 2	29	2	102	
95	CAHOKIA VILLAGE S/C	66	-	-	4	1	-	4	1	. 2	19	2	99	
97	THE SHOPS AT WHITE OAK VILLAGE	82	2		4	-	1	3	1	. 2	16	4	114	
99	GREENWOOD WEST	49	3		4	-		7	1	. 2	11	-	77	
100	CHERRYDALE POINT S/C	203	8	-	5		0		1	. 2	24	2	260	
101	NEWMARKET SOUTH S/C	77	6	-	4	2	-	13	1	. 2	15	2	122	

Store #	Store Name	RENT	AD VALOREM TAX	FILTER SERVICE	INSURANCE - BUSINESS	LICENSE - CITY	LICENSE- STATE & COUNTY	MAINTENANC E	SECURITY	TELEPHONE	UTILITIES	WASTE PICK UP	Total Occupancy	% Rent
102	FOOD 4 LESS S/C	253	-	-	4	-	-	4		1	16	-	278	
104	SOUTHLAKE MALL	113	-	-	4	4	-	7	-	2	20	3	151	
106	RACEWAY PARK S/C	148	49	-	3	1	-	20	1	. 1	21	3	247	
107	POINCIANA PLAZA	-	17	-	7	1	0	13	1	. 2	26	4	70	3%
109	HERITAGE PLACE S/C	206	8	-	5	1	0	13	1	. 2	24	2	262	
110	POPLAR PLAZA S/C	79	3	-	4	-	-	13	1	. 2	26	2	130	
111	LANGLEY SQUARE	100	1	-	4	1	-	3	1	. 2	16	-	128	
114	NORTH POINTE S/C	90	4	-	4	1	-	12	1	. 2	17	2	132	
116	SPENCER SQUARE	220	8	-	5	1	0	13	1	. 2	24	2	276	
118	VILLAGE S/C	-	-	-	3	1	0	5	-	1	40	3	54	8%
122	LYNNHAVEN NORTH S/C	166	8	-	5	1	0	13	1	. 2	24	2	222	
123	CEDAR HILLS S/C	93	-	-	4	5	-	4	-	-	33	0	139	
124	ROEBUCK MARKETPLACE	96	1	-	4	_	0	3	1	. 2	23	-	130	
126	FREESTANDING	165	8	-	5		0	13	1	. 2	24	2	222	
127	RIVER COMMONS	56	10	-	7	_	0	5	1	. 2	18	4	104	
128	EASTLAND HILL CENTER	99	19	-	4		1	23	1	. 2	22	4	176	
129	WEST TOWN PLAZA	47	2	-	4		-	35	1	. 2	25	2	119	
130	GRAND BOULEVARD PLAZA	149	34	-	4		-	28	1	. 2	23	4	244	
133	CAPITOL PLAZA	47	0	-	4	2	0	18	1	. 2	33	3	111	
137	PARKWAY PLAZA	38	1	-	4		0	5	1	. 2	21	3	75	
138	N/A	168	3	-	4		0	3	1	. 2	28	4	213	
139	DODGE PLAZA	104	3	-	4	-	0	2	1	. 2	15	3	135	
141	FOOD WORLD PLAZA SOUTH	70	4	-	4		0	10	1	. 2	23	3	118	
144	FRAYSER PLAZA	74	7	-	5	_	1	16	1	2	24	3	133	
145	IVES DAIRY CROSSING S/C	104	7	-	3		-	9	-	0	15	2	147	
148	SHOPS AT NORTHEAST VILLAGE	110	1	-	4			4	1	. 2	32		157	
151	TOWN & COUNTRY PLAZA	66	10	-	9		0	9	1	. 2	26	4	127	
153	SHOPPES AT GARNER	188	8	-	5		0	13	1	. 2	24	2	244	
154	NORTHLINE CROSSING S/C	267	8	-	5		0	13	1	. 2	24	2	324	
155	CANE RUN	90	2	-	4			2	1	. 2	30	2	133	
157	EASTOVER S/C	192	17	-	5		0	20	1	. 2	25	5	269	
158	STEELYARD COMMONS S/C	339	81 0	-	3		-	54	-	1	40	9	529	
159	WESTERN HILLS MALL	98	-	-	4	_	0	3		2	38	1	148	
160	THE SHOPPES OF LIBERTY CITY	152	35	-	16		U	18	1	. 2	23	-	252	
161	BLUE PARKWAY TOWN CENTER	104	14 17	-	5 13		-	26 27	1	. 2	18	-	170	
162 163	GRIGGS ROAD S/C MADISON SQUARE	181 69	7	-	7		0	10	1	. 2	24 34	4	272 135	
	SELMA MALL	54	0	-	4		0	15	1	. 2	22	4	104	
165	FAMILY DOLLAR CENTER	42	4	-	4	_	U	3	1		28	2	86	
166	LANTANA VILLAGE SOUARE	42	1		4		0	3	1	. 2	26	5	42	3%
167	AZALEA SHOPPING CENTER		4		5		0	13	1	. 2	19	4	48	3%
170	TAYLOR COMMONS	192	8		5		0	13	1	2	24	2	248	370
173	EUTAW VILLAGE	119	1		4			7	1	2	21	3	157	
174	NORTHSIDE SHOPPING CENTRE	176	20		13		0	30	1	2	29		271	
175	LAMAR AIRWAYS S/C	103	13	_	6		1	23	1	2	22	3	174	
176	STONY ISLAND PLAZA	219	8		5		0	13	1	2	24	2	276	
178	WEST ALLIS	216	8		5		0	13	1	2	24	2	272	
180	WEST RIDGE	99	12		6		-	19	1		25	2	164	
182	LAUDERHILL MALL	151	9		4		0	4	1	2	30	6	208	
183	SAVE A LOT CTR	88	-		4		-		1	2	22	-	117	
184	MODEL T PLAZA	179			3	-		26		1	16	2	230	
185	KING SHOPPING CENTER	95	25		4		0	32	1	. 2	39	1	199	
186	SOUTHGATE USA S/C	121	8		5		0	13	1	2	24	2	177	
187	THRUWAY PLAZA	227	8		5		0	13	1	2	24	2	283	
188	MIRACLE MILE S/C	237	-		2		-	5			15	4	264	
189	MARSHFIELD PLAZA	165	38		7			37	1	2	21	4	275	

Store #	Store Name	RENT	AD VALOREM TAX	FILTER SERVICE	INSURANCE - BUSINESS	LICENSE - CITY	LICENSE- STATE & COUNTY	MAINTENANC E	SECURITY	TELEPHONE	UTILITIES	WASTE PICK UP	Total Occupancy	% Rent
191	FONDREN SOUTHWEST VILLAGE	188	14		5	2	-	16		1	15	4	245	
192	VILLAGE GREEN	69	1	-	4	1	0	4	1	. 2	34	2	119	
193	N/A	86	-	-	4	-	-	22	1	. 2	29	3	147	
196	DOLPHIN PLAZA	51	10	-	6	7	1	13	-	1	22	5	117	
197	BOSSIER TOWN CENTER	127	4	-	4	2	0	17	1	. 2	25	4	186	
198	GATEWAY	50	11	-	7	1	1	17	1	. 2	30	3	120	
199	BEL AIR S/C	71	0	-	4	4	-	4	1	. 2	23	2	111	
200	MARKETPLACE AT SMYRNA	275	8	-	5	1	0	13	1	. 2	24	2	331	
201	GALLERY AT WARREN CONNER	178	13	-	4	0	-	21	1	2	32	4	255	
203	ELLIS AVENUE PLAZA	77	5	-	4	0	-	3	1	2	24	3	119	
204	RIVERVIEW PLAZA	56	0	-	4	0	-	3	1	. 2	25	1	93	
205	CULLEN PLAZA	127	15	-	9	1	-	25	1	. 2	15	1	197	
207	THE COMMONS ROUTE 20	43	1	-	4	1	-	4	1	2	34	2	91	
208	JEFFERSON VILLAGE	134	-	-	4	-	-	2	-	-	14	-	154	
212	ROYAL TOWN CENTER	137	1 2	-	4	-	-	3	1	. 2	22 20	3	172	
213 215	COLLEGE PARK COMMONS SOUTH SLAPPEY VILLAGE	95 78	10	-	4	0	-	9	1	. 2	20 29	3 2	129 138	
		107	2	-	4	2	-	6	1	. 2	29			
216 218	WALNUT HILLS PLAZA SHOPS AT WHITE OAK VILLAGE	244	8	-	4 5	1	3	13	1	. 2	24	2	150 300	
218	COUNTRY CLUB PLAZA	209	8	-	5	1	0	13	1	. 2	24	2	265	
219	THE CROSSINGS AT HALLS FERRY	133	23	-	5	1	U	21	1	. 2	14	2	205	
223	CHICAGO	258	8	-	5	1	0	13	1	. 2	24	2	314	
223	SQUARE 67	68	10	-	7	0	U	18	1	. 0	17	2	124	
225	GREENS CROSSROADS	93	3		4	1		4	1	. 2	19	2	130	
227	PLAZA ON THE BLV	155	20		4	1		32	1	2	15	-	229	
228	NORTHGATE S/C	289	35		2	5		25		_	5	0	360	
229	JOYLAND S/C	130	1		4	2		3	1	2	27	3	173	
230	SILVER MILL COURT S/C	80	0	_	4	-	0	24	1	2	41	4	157	
231	MARLOW HEIGHTS S/C	118	2	-	4	-	0	3	1	2	18	5	153	
233	WESTPORT COMMONS S/C	112			3	1		6		1	18		141	
234	TOWNE SOUTH	-	2	-	4	0	-	4	1	. 2	27	1	42	3%
236	LAWNDALE PLAZA	211	31	-	4	0	-	54	1	. 2	19	-	322	
237	BALTIMORE	187	8	-	5	1	0	13	1	. 2	24	2	243	
239	SOUTH WIND PLAZA	78	5	-	4	1	0	3	1	2	16	-	109	
240	BUCKNER COMMONS	-	15	-	5	0	-	17	1	. 2	22	-	63	3%
241	TINLEY PARK PLAZA	187	8	-	5	1	0	13	1	. 2	24	2	243	
242	SOUTH PHILADELPHIA S/C	78	6	-	3	2	-	8	1	. 2	24	6	129	
246	CITY PLAZA	151	22	-	6	1	1	53	1	. 2	17	2	255	
247	SPEEDWAY SUPERCENTER	48	1	-	4	0	-	3	1	. 2	22	2	81	
248	FREESTANDING	66	-	-	4	0	-	2	1	. 2	28	4	108	
249	CROSSROADS CENTER	132	8	-	5	1	0	13	1	. 2	24	2	188	
252	JEWEL OSCO S/C	165	43	-	4	0	-	102	1	. 2	39	7	363	
254	30TH STREET S/C	110	8	-	5	1	0	13	1	. 2	24	2	166	
255	MANCHESTER PARKADE	184	8	-	5	1	0	13	1		24	2	240	
256	SOUTHFIELD	-	8		5	1	0	13	1	. 2	24	2	56	4%
257	Baseline Plaza	102	7	-	4	0		24	1	. 2	16	-	155	
258	SHOPPES ON MAIN	88	-	-	4	-	-	5	1	. 2	19	-	118	
259	SOUTH DEKALB PLAZA	88	9	-	4	-	1	21	1	2	24	3	152	
262	MEADOWBROOK EAST	41	15	-	8	0	-	21	1	2	23	3	113	
263	CHARITON SQUARE	-	1	-	4	1	1	8	1	2	13	2	32	3%
264	LAMAR CROSSING	102	27		6	1	1	21	1	. 2	30	3	195	
265	WYOMING VILLAGE		0		4	0		-	1	. 2	28	-	35	3%
266	STATE STREET	116	7	-	5	0	-	10	1	. 2	15	2	159	
268	EASTLAND CENTER	82	1	-	4	1	-	3	-	2	40	-	132	
269	ST. STEPHENS SQUARE	82	5	-	6	2	0	11	1	. 2	25	3	136	
270	HIGHLAND LAKES CENTER	62	14	-	4	-	0	14	1	. 2	17	-	114	
271	CARRIAGE CROSSING MARKETPLACE		8	-	5	1	0	13	1	. 2	24	2	56	5%

Store #	Store Name	RENT	AD VALOREM TAX	FILTER SERVICE	INSURANCE - BUSINESS	LICENSE - CITY	LICENSE- STATE & COUNTY	MAINTENANC E	SECURITY	TELEPHONE	UTILITIES	WASTE PICK UP	Total Occupancy	% Rent
272	ERDMAN PLAZA	110	13	-	6	-	1	19	1	. 2	23	2	176	
273	NORTHWEST CROSSING	267	8	-	5	1	0	13	1	. 2	24	2	323	
274	WOODLAND VILLAGE	147	14	-	2	2	-	30		-	21	3	219	
275	CROSS COUNTRY PLAZA	118	13	-	7	2	-	3	1	. 2	29	-	175	
278	EDENS PLAZA	47	3	-	4	1	-	3	1	. 2	30	2	92	
281	WHITEHAVEN PLAZA	77	1	-	4	1	1	3	1	. 2	25	3	117	
282	RACEWAY PARK S/C	204	60	-	3	1	-	23		0	13	2	306	
283	ALBANY CROSSING	101	5	-	15	0	-	15	1	. 2	18	2	159	
284	OAKLAND POINTE	99	0	-	4	1	-	4	1	. 2	23	-	134	
285	ONE & OLNEY SQUARE	185	38	-	7	0	-	18	1	. 1	32	7	290	
287	NORTHWEST CROSSING	232	8	-	5		0	13	1	. 2	24	2	289	
288	CLEVELAND AVENUE CROSSING	110	-	-	3		-	2		1	29	-	150	
301	CLOVERLEAF S/C	108	4	-	4		-	4	1	. 2	32	3	157	
302	BATESVILLE S/C	66	8	-	5		-	9	1	. 2	16	4	110	
303	NORTHTOWN PLAZA	136	24	-	6		-	21	1	. 2	17	4	211	
317	HAMMOND AIRE PLAZA	132	9	-	11		0	31	1	. 2	32	4	223	
327	WALZEM PLAZA	57	5	-	4	0	-	13	1	. 2	23	3	108	
337	TOWN & COUNTRY	173	12	-	5	1	0	16		1	19	4	231	
338	SOUTH PHILADELPHIA S/C	147	10	-	3		-	2		1	24	5	192	
343	EASTOWN S/C	38	-	-	2		0			1	23	2	73	
344	NORTHERN LIGHTS	151		-	2		0			1	24		186	
348	CENTURY CENTER	185	8	-	5			6			14	4	224	
360	FIVE POINTS WEST S/C	159	1	-	4	2	0		1	. 2	27	2	202	
365	CROSS CREEK	69	1	-	4			13	1	. 2	17	-	107	
369	FREESTANDING	111	1	-	4	2	0	6	1	. 2	26	4	156	
375	ROSE PLAZA	86	9	-	6		-	10	1	. 2	17	3	134	
377	LINCOLN S/C	147	8	-	10	-	-	10		1	20	2	198	FOV
500 501	ANDERSON ROAD PLAZA METRO PLAZA	110	4	-	4	- 0	-	31 2	1	. 2	25 32	3 2	69 154	5%
		165	0	-	4	Ü	-	5	1	. 2		2		
504	HIGHLAND PARK PLACE	103	U	-	3		-	5	1		32	-	210	
506 507	LIVONIA MARKETPLACE EDISTO VILLAGE SHOPPING CENTER	41	13	-	5		-	6	1	1 2	16 20	-	130 89	
507	WESTSHORE PLAZA	214	13	-	5		0		1	. 2	20	2	270	
509	KLINE PLAZA	172	43	-	7		0	34			19	5	282	
510	CLINTON POINTE	69	43		3			3		1	3	2	80	
510	OAK STREET SHOPPING CENTER	88	2		4	1		3	1	2	34	2	137	
512	NORGATE PLAZA	136	-		4	1	0				13	0	153	
513	MLK PLAZA	123	25		9	2	1	28	1	2	18	-	209	
514	NORTH CHARLESTON CENTER	103	9		7	-	2		1	2	24	5	163	
515	SUMTER SQUARE SHOPPING CENTER	85	11	_	5	1	-	11	1	2	17	2	135	
516	COTTMAN & BUSELTON S/C	236	8	_	5	1	0	13	1	2	24	2	292	
520	BORDEAUX SHOPPING CENTER	87	8	_	6	1	1	19	1	2	26	3	154	
523	WESTLAND SHOPPING CENTER	74	17		5		-	11	1	2	18	3	130	
524	THE SHOPS AT STERLING PONDS	146	2		3			9	-	1	11	1	179	
525	NORTHWEST PLAZA	55		_	8			3	1	. 2	20	2	90	
526	DORCHESTER SQUARE	73	11	_	7		2		1	2	28	4	145	
527	CANTON CENTRE	227	5		3		0			0	19	-	267	
528	GEORGETOWN PLAZA SHOPPING CENTER	61	11		5		-	16	1	. 2	29	2	127	
529	EASTGATE SHOPPING CENTER	105	8		6		0		1	2	21	2	158	
531	MIDDLESEX S/C	196	10		3		2	23		1	9	5	250	
535	CRICHTON PLAZA	73	1		4	2	0	20	1	. 2	32	3	137	
537	CHESAPEAKE CROSSING SHOPPING CENTER	90	7		4	2		10	1	2	23	2	140	
538	CROSSROADS PLAZA	-	1		4	0	-	5	1	2	31	3	47	4%
539	HATTIESBURG SHOPPING CENTER	104	4		4	0	-	28	1	2	19	2	164	
541	LIBERTY CENTER	100	-		2	-	-	4		-	5	0	112	
543	LEHMBERG CROSSING	-	3	-	4	-	0	3	1	. 2	15	2	29	4%
										1			160	

Occupancy Per Diem

Store #	Store Name	RENT	AD VALOREM TAX	FILTER SERVICE	INSURANCE - BUSINESS	LICENSE - CITY	LICENSE- STATE & COUNTY	MAINTENANC E	SECURITY	TELEPHONE	UTILITIES	WASTE PICK UP	Total Occupancy	% Rent
545	GREAT SOUTHERN S/C	104	-	-	3	-	0	29	-	0	12	-	149	
549	FREEDOM SQUARE SHOPPING CENTER	103	4	-	4	2	-	4	1	2	20	2	141	
554	TOWN CENTER MARKET	169	5	-	5	-	1	11	-	1	18	2	213	
555	IROQUOIS MANOR	62	9	-	5	-	-	11	1	2	27	2	118	
556	WESLEY CHAPEL SQUARE SHOPPING CENTER	108	2	-	4	-	1	39	1	2	26	3	186	
558	SHELBY PLAZA	88	1	-	4	1	1	15	1	2	13	3	128	
559	AUSTINTOWN PLAZA	128	-	-	3	2	1	2	-	1	10	3	148	
562	NORTH POINT CENTER	132	1	-	4	1	1	4	1	2	28	3	176	
563	CONSUMER SQUARE WEST	160	8	-	5	1	0	13	1	2	24	2	216	
565	BROADWAY CENTER	89	1	-	4	1	1	-	1	2	16	2	115	
571	WESTWOOD SHOPPING CENTER	113	19	-	8	2	0	18	1	2	36	3	202	
575	VILLAGE SHOPPING CENTER	-	7	-	5	1	-	17	1	2	35	2	69	8%
576	SOUTHGATE SHOPPING CENTER	129	14	-	10	1	-	29	1	2	21	-	207	
577	HICKORY RIDGE CROSSING	58	1	-	4	1	1	3	1	2	22	3	94	
581	GAINSVILLE SHOPPING CENTER	84	13	-	4	1	-	20	1	2	22	3	150	
584	SOUTHGATE PLAZA	121	7	-	7	0	-	10	1	2	26	2	174	
595	PARKWAY PLAZA	91	17	-	10	0	-	17	1	2	24	2	164	
599	COURTHOUST SQUARE	86	2	-	4	-	0	2	1	2	12	3	111	
617	ASHER	59	3	-	4	1	-	11	1	2	18	2	100	
247	Total	27,968	2,098	-	1,217	211	51	3,150	198	382	5,803	578	41,655	-

Note:
For stores indicating percentage rent amounts, these are straight percentage rent locations; there is no base rent associated with these stores and the percentage rent amount is a straight percentage of sales with no minimum threshold or breakpoint.

Simply Fashions

Exhibit 5.2(b)

Distribution Center Merchandise

As reflected in **Simply Fashions Stores Inventory Including Cost and Retail 4-7-15.xlsx**, locations:

Location # Description

751 Dots DOTCOM

752 SF DOTCOM

- 0 Simply Warehouse
- 2 Pallet Simply Fashions
- 3 Dots Warehouse
- 6 SIMPLY FASHION TIERING WAREHOUSE

Level and Mix

Dept #	Dept Name	Class	Class Name	Sub Class	SubClass Name	Current Retail	% of Total Retail
01-51	JUNIOR	01-51-10	JUNIOR TOPS	01-51-10-10	JR WOV S/L	365,562	2.3%
01-51	JUNIOR	01-51-10	JUNIOR TOPS	01-51-10-20	JR WOV S/S	212,525	1.3%
01-51	JUNIOR	01-51-10	JUNIOR TOPS	01-51-10-30	JR WOV L/S	287,484	1.8%
01-51	JUNIOR	01-51-10	JUNIOR TOPS	01-51-10-50	JR KNI S/L	402,431	2.5%
01-51	JUNIOR	01-51-10	JUNIOR TOPS	01-51-10-60	JR KNI S/S	597,669	3.7%
01-51	JUNIOR	01-51-10	JUNIOR TOPS	01-51-10-70	JR KNI L/S	159,444	1.0%
01-51	JUNIOR	01-51-10	JUNIOR TOPS	01-51-10-80	JR 2-FOR CAMI'S	205,825	1.3%
01-51	JUNIOR	01-51-10	JUNIOR TOPS	01-51-10-82	JR 2-FOR S/S	266,602	1.7%
01-51	JUNIOR	01-51-10	JUNIOR TOPS	01-51-10-84	JR 2-FOR L/S	56,432	0.4%
01-51	JUNIOR	01-51-20	JUNIOR BOTTOMS	01-51-20-10	JR SKIRTS	154,237	1.0%
01-51	JUNIOR	01-51-20	JUNIOR BOTTOMS	01-51-20-15	JR DEN SKIRTS	45,467	0.3%
01-51	JUNIOR	01-51-20	JUNIOR BOTTOMS	01-51-20-25	JR JEANS	495,408	3.1%
01-51	JUNIOR	01-51-20	JUNIOR BOTTOMS	01-51-20-30	JR PANTS	439,848	2.7%
01-51	JUNIOR	01-51-20	JUNIOR BOTTOMS	01-51-20-40	JR CAPRIS	70,313	0.4%
01-51	JUNIOR	01-51-20	JUNIOR BOTTOMS	01-51-20-45	JR DEN CAPRIS	15,610	0.1%
01-51	JUNIOR	01-51-20	JUNIOR BOTTOMS	01-51-20-60	JR ROMPERS	65,339	0.4%
01-51	JUNIOR	01-51-20	JUNIOR BOTTOMS	01-51-20-70	JR SHORTS	196,081	1.2%
01-51	JUNIOR	01-51-20	JUNIOR BOTTOMS	01-51-20-75	JR DEN SHORTS	89,385	0.6%
01-51	JUNIOR	01-51-20	JUNIOR BOTTOMS	01-51-20-80	JR LEGGING	146,850	0.9%
01-51	JUNIOR	01-51-30	JUNIOR COORDINATES	01-51-30-10	JR WOV COOR	3,894	0.0%
01-51	JUNIOR	01-51-30	JUNIOR COORDINATES	01-51-30-35	JR DEN COOR	3,034	0.0%
01-51	JUNIOR	01-51-30	JUNIOR COORDINATES	01-51-30-50	JR KNI COOR	125,096	0.8%
01-51	JUNIOR	01-51-30	JUNIOR COORDINATES	01-51-30-55	JR ACT WEAR	174,086	1.1%
01-51	JUNIOR	01-51-40	JUNIOR OUTERWEAR	01-51-40-10	JR SWEATERS	66,019	0.4%
01-51	JUNIOR	01-51-40	JUNIOR OUTERWEAR	01-51-40-20	JR JKT	67,534	0.4%
01-51	JUNIOR	01-51-50	JUNIOR DRESSES	01-51-50-10	JR DRESSES	840,702	5.3%
01-51	JUNIOR	01-51-50	JUNIOR DRESSES	01-51-50-20	JR JUMPSUITS	44,724	0.3%
02-52	PLUS TOPS	02-52-10	PLUS TOPS	02-52-10-10	PL WOV S/L	227,074	1.4%
02-52	PLUS TOPS	02-52-10	PLUS TOPS	02-52-10-20	PL WOV S/S	238,095	1.5%
02-52	PLUS TOPS	02-52-10	PLUS TOPS	02-52-10-30	PL WOV L/S	184,972	1.2%
02-52	PLUS TOPS	02-52-10	PLUS TOPS	02-52-10-50	PL KNI S/L	333,008	2.1%
02-52	PLUS TOPS	02-52-10	PLUS TOPS	02-52-10-60	PL KNI S/S	626,283	3.9%
02-52	PLUS TOPS	02-52-10	PLUS TOPS	02-52-10-70	PL KNI L/S	126,577	0.8%
02-52	PLUS TOPS	02-52-10	PLUS TOPS	02-52-10-80	PL 2-FOR CAMI'S	199,656	1.2%
02-52	PLUS TOPS	02-52-10	PLUS TOPS	02-52-10-82	PL 2-FOR S/S	249,928	1.6%
02-52	PLUS TOPS	02-52-10	PLUS TOPS	02-52-10-84	PL 2-FOR L/S	23,408	0.1%
02-52	PLUS TOPS	02-52-30	PLUS KNIT COORDINATES	02-52-30-50	PL KNI COOR	120,831	0.8%
02-52	PLUS TOPS	02-52-30	PLUS KNIT COORDINATES	02-52-30-55	PLUS ACT WEAR	151,667	0.9%
02-52	PLUS TOPS	02-52-40	PLUS OUTERWEAR	02-52-40-10	PL SWEATERS	24,530	0.2%
02-52	PLUS TOPS	02-52-40	PLUS OUTERWEAR	02-52-40-20	PL JKTS	67,848	0.4%
02-52	PLUS TOPS	02-52-50	PLUS DRESSES	02-52-50-10	PLUS DRESSES	651,277	4.1%
02-52	PLUS TOPS	02-52-50	PLUS DRESSES	02-52-50-20	PLUS JUMPSUITS	49,374	0.3%
02-53	SUPER TOPS	02-53-10	SUPER PLUS TOPS	02-53-10-10	SP WOV S/L	48,298	0.3%
02-53	SUPER TOPS	02-53-10	SUPER PLUS TOPS	02-53-10-20	SP WOV S/S	69,083	0.4%
02-53	SUPER TOPS	02-53-10	SUPER PLUS TOPS	02-53-10-30	SP WOV L/S	61,847	0.4%
02-53	SUPER TOPS	02-53-10	SUPER PLUS TOPS	02-53-10-50	SP KNI S/L	39,373	0.2%
02-53	SUPER TOPS	02-53-10	SUPER PLUS TOPS	02-53-10-60	SP KNI S/S	199,100	1.2%
02-53	SUPER TOPS	02-53-10	SUPER PLUS TOPS	02-53-10-70	SP KNI L/S	13,976	0.1%
02-53	SUPER TOPS	02-53-10	SUPER PLUS TOPS	02-53-10-80	SP 2-FOR CAMI'S	52,483	0.3%
02-53	SUPER TOPS	02-53-10	SUPER PLUS TOPS	02-53-10-82	SP 2-FOR S/S	64,081	0.4%
02-53	SUPER TOPS	02-53-30	SUPER KNIT COORDINATES	02-53-30-50	SP KNI COOR	9,674	0.1%
02-53	SUPER TOPS	02-53-30	SUPER KNIT COORDINATES	02-53-30-55	SUPER ACT WEAR	67,490	0.4%
02-53	SUPER TOPS	02-53-40	SUPER PLUS OUTERWEAR	02-53-40-10	SP SWEATERS	4,395	0.0%
02-53	SUPER TOPS	02-53-40	SUPER PLUS OUTERWEAR	02-53-40-20	SP JKT	683	0.0%
02-53	SUPER TOPS	02-53-50	SUPER PLUS DRESSES	02-53-50-10	SP DRESS/JSUITS/RMPR	94,210	0.6%
03-52	PLUS BOTTOMS	03-52-20	PLUS BOTTOMS	03-52-20-10	PL SKIRTS	163,242	1.0%
03-52	PLUS BOTTOMS	03-52-20	PLUS BOTTOMS	03-52-20-15	PL DEN SKIRT	34,321	0.2%
03-52	PLUS BOTTOMS	03-52-20	PLUS BOTTOMS	03-52-20-25	PL JEANS	262,407	1.6%
03-52	PLUS BOTTOMS	03-52-20	PLUS BOTTOMS	03-52-20-30	PL PANTS	332,219	2.1%
03-52	PLUS BOTTOMS	03-52-20	PLUS BOTTOMS	03-52-20-40	PL CAPRIS	93,552	0.6%
03-52	PLUS BOTTOMS	03-52-20	PLUS BOTTOMS	03-52-20-45	PL DEN CAPRI	17,856	0.1%
03-52	PLUS BOTTOMS	03-52-20	PLUS BOTTOMS	03-52-20-60	PL ROMPERS	32,800	0.2%
03-52	PLUS BOTTOMS	03-52-20	PLUS BOTTOMS	03-52-20-70	PL SHORTS	225,859	1.4%
03-52	PLUS BOTTOMS	03-52-20	PLUS BOTTOMS	03-52-20-75	PL DEN SHORTS	68,151	0.4%
03-52	PLUS BOTTOMS	03-52-20	PLUS BOTTOMS	03-52-20-80	PL LEGGING	152,883	1.0%

Case 15-16885-LMI Doc 195 Filed 05/06/15 Page 96 of 281

Dept #	Dept Name	Class	Class Name	Sub Class	SubClass Name		Current Retail	% of Total Retail
03-52	PLUS BOTTOMS	03-52-30	PLUS WOVEN COORDINATES	03-52-30-10	PL WOV COOR		11,334	0.1%
03-52	PLUS BOTTOMS	03-52-30	PLUS WOVEN COORDINATES	03-52-30-35	PL DEN COOR		8,475	0.1%
03-53	SUPER BOTTOMS	03-53-20	SUPER PLUS BOTTOMS	03-53-20-10	SP SKIRTS		55,061	0.3%
03-53	SUPER BOTTOMS	03-53-20	SUPER PLUS BOTTOMS	03-53-20-15	SP DEN SKIRT		7,338	0.0%
03-53	SUPER BOTTOMS	03-53-20	SUPER PLUS BOTTOMS	03-53-20-25	SP JEANS		63,450	0.4%
03-53	SUPER BOTTOMS	03-53-20	SUPER PLUS BOTTOMS	03-53-20-30	SP PANTS		150,750	0.9%
03-53	SUPER BOTTOMS	03-53-20	SUPER PLUS BOTTOMS	03-53-20-40	SP CAPRIS		56,316	0.4%
03-53	SUPER BOTTOMS	03-53-20	SUPER PLUS BOTTOMS	03-53-20-45	SP DEN CAPRI		21,660	0.1%
03-53	SUPER BOTTOMS	03-53-20	SUPER PLUS BOTTOMS	03-53-20-70	SP SHORTS		31,815	0.2%
03-53	SUPER BOTTOMS	03-53-20	SUPER PLUS BOTTOMS	03-53-20-75	SP DEN SHORTS		13,683	0.1%
03-53	SUPER BOTTOMS	03-53-20	SUPER PLUS BOTTOMS	03-53-20-80	SP LEGGINGS		41,406	0.3%
03-53	SUPER BOTTOMS	03-53-30	SUPER WOVEN COORDINATES	03-53-30-10	SP WOV COOR		4,444	0.0%
03-53	SUPER BOTTOMS	03-53-30	SUPER WOVEN COORDINATES	03-53-30-35	SP DEN COOR		1,712	0.0%
08-59	CLOSEOUTS	08-59-10	CLOSEOUTS JUNIOR	08-59-10-99	JR CLOSEOUTS		369,150	2.3%
08-59	CLOSEOUTS	08-59-20	CLOSEOUTS PLUS	08-59-20-99	PLUS CLOSEOUTS		330,962	2.1%
07-60	ACCESSORIES	07-60-10	ACCESSORIES EARRINGS	07-60-10-01	GOLD/SILVER		211,973	1.3%
07-60	ACCESSORIES	07-60-10	ACCESSORIES EARRINGS	07-60-10-02	COLOR		235,875	1.5%
07-60	ACCESSORIES	07-60-10	ACCESSORIES EARRINGS	07-60-10-03	CLIPS		25,867	0.2%
07-60	ACCESSORIES	07-60-10	ACCESSORIES EARRINGS	07-60-10-05	TRIOS		79,081	0.5%
07-60	ACCESSORIES	07-60-12	ACCESSORIES SETS	07-60-12-01	NECK & EAR		579,970	3.6%
07-60	ACCESSORIES	07-60-12	ACCESSORIES SETS	07-60-12-03	BRACELETS SETS		-	0.0%
07-60	ACCESSORIES	07-60-13	ACCESSORIES BRACELETS	07-60-13-01	BRACELETS		210,762	1.3%
07-60	ACCESSORIES	07-60-19	NOVELTY / GIFTS	07-60-19-01	NOVELTY JEWELRY		41,368	0.3%
07-60	ACCESSORIES	07-60-19	NOVELTY / GIFTS	07-60-19-02	RINGS		41,956	0.3%
07-60	ACCESSORIES	07-60-19	NOVELTY / GIFTS	07-60-19-10	BEAUTY PRODUCTS		72,705	0.5%
07-60	ACCESSORIES	07-60-19	NOVELTY / GIFTS	07-60-19-20	ELECTRONICS		16,493	0.1%
07-60	ACCESSORIES	07-60-20	ACCESSORIES WATCHES	07-60-20-01	WATCHES		65,416	0.4%
07-60	ACCESSORIES	07-60-30	ACCESSORIES HATS	07-60-30-01	HATS		29,821	0.2%
07-60	ACCESSORIES	07-60-35	ACCESSORIES SCARVES	07-60-35-99	SCARVES		102,959	0.6%
07-60	ACCESSORIES	07-60-40	ACCESSORIES BELTS	07-60-40-02	BELTS		67,969	0.4%
07-60	ACCESSORIES	07-60-50	ACCESSORIES HANDBAGS	07-60-50-01	HANDBAGS		211,394	1.3%
07-60	ACCESSORIES	07-60-60	ACCESSORIES SUNGLASSES	07-60-60-01	SUNGLASSES		50,153	0.3%
07-60	ACCESSORIES	07-60-80	ACCESSORIES COLD WEATHER	07-60-80-01	COLD WEATHER		4,833	0.0%
07-60	ACCESSORIES	07-60-90	ACCESS SOCKS	07-60-90-01	SOCKS		72,337	0.5%
07-60	ACCESSORIES	07-60-95	ACCESSORIES MISCELLANOUS	07-60-95-01	FRAGRANCES		52,774	0.3%
07-60	ACCESSORIES	07-60-95	ACCESSORIES MISCELLANOUS	07-60-95-10	MISCELLANEOUS		14,910	0.1%
07-61	LINGERIE	07-61-01	LINGERIE JUNIOR	07-61-01-01	BRAS		131,791	0.8%
07-61	LINGERIE	07-61-01	LINGERIE JUNIOR	07-61-01-02	PANTIES		308,752	1.9%
07-61	LINGERIE	07-61-01	LINGERIE JUNIOR	07-61-01-03	1.49 PANTIES		7,727	0.0%
07-61	LINGERIE	07-61-01	LINGERIE JUNIOR	07-61-01-04	DAYWEAR		7,917	0.0%
07-61	LINGERIE	07-61-01	LINGERIE JUNIOR	07-61-01-05	SLEEPWEAR		197,349	1.2%
07-61	LINGERIE	07-61-02	LINGERIE PLUS	07-61-02-01	BRAS		-	0.0%
07-61	LINGERIE	07-61-02	LINGERIE PLUS	07-61-02-02	PANTIES		-	0.0%
07-61	LINGERIE	07-61-02	LINGERIE PLUS	07-61-02-03	2.49 PANTIES		6,867	0.0%
07-61	LINGERIE	07-61-02	LINGERIE PLUS	07-61-02-04	DAYWEAR		-	0.0%
07-61	LINGERIE	07-61-02	LINGERIE PLUS	07-61-02-05	SLEEPWEAR		_	0.0%
07-61	LINGERIE	07-61-03	LINGERIE SUPER	07-61-03-01	BRAS		_	0.0%
07-61	LINGERIE	07-61-03	LINGERIE SUPER	07-61-03-02	PANTIES		_	0.0%
07-61	LINGERIE	07-61-03	LINGERIE SUPER	07-61-03-05	SLEEPWEAR		_	0.0%
07-62	SHOES	07-62-10	SHOES	07-62-10-01	SHOES		602,004	3.8%
07-62	SHOES	07-62-10	SHOES	07-62-10-20	BOOTS		29,786	0.2%
07-62	SHOES	07-62-10	SHOES	07-62-10-25	W W BOOTS		3,368	0.0%
07-62	SHOES	07-62-10	SHOES	07-62-10-50	2 OR MORE		366,185	2.3%
07-63	ACC/CLOSEOUTS	07-63-01	ACC/CLOSEOUTS	07-63-01-10	ACCESSORY CLOSEOUTS		692	0.0%
07-63	ACC/CLOSEOUTS	07-63-01	ACC/CLOSEOUTS	07-63-01-20	LINGERIE CLOSEOUTS		296	0.0%
07-63	ACC/CLOSEOUTS	07-63-01	ACC/CLOSEOUTS	07-63-01-20	SHOE CLOSEOUTS		981	0.0%
	, 520520013	0. 00 01		3. 33 31 30	502 020020010	Total	16,000,000	100.0%

Exhibit 8.1

Sale Guidelines

SALE GUIDELINES

- A. The Sale shall be conducted so that the Stores in which sales are to occur will remain open no longer than during the normal hours of operation provided for in the respective leases for the Stores.
- B. The Sale shall be conducted in accordance with applicable state and local "Blue Laws," where applicable, so that no Sale shall be conducted on Sunday unless the Merchant had been operating such Store on a Sunday.
- C. On "shopping center" property, the Agent shall not distribute handbills, leaflets or other written materials to customers outside of any Stores' premises, unless permitted by the lease or, if distribution is customary in the "shopping center" in which such Store is located; provided that Agent may solicit customers in the Stores themselves. On "shopping center" property, the Agent shall not use any flashing lights or amplified sound to advertise the Sale or solicit customers, except as permitted under the applicable lease or agreed to by the landlord.
- D. At the conclusion of the Sale, the Agent shall vacate the Stores in broom clean condition, and shall leave the Locations in the same condition as on the Sale Commencement Date, ordinary wear and tear excepted, in accordance with Section 6 of the Agency Agreement, provided, however, that the Merchant and the Agent hereby do not undertake any greater obligation than as set forth in an applicable lease with respect to a Store. The Agent and Merchant may abandon any FF&E not sold in the Sale at the Stores at the conclusion of the Sale. Any abandoned FF&E left in a Store after a lease is rejected shall be deemed abandoned to the landlord having a right to dispose of the same as the landlord chooses without any liability whatsoever on the part of the landlord to any party and without waiver of any damage claims against the Merchant. For the avoidance of doubt, as of the Sale Termination Date, the Agent may abandon, in place and without further responsibility, any FF&E located at a Store.
- E. The Merchant and the Agent may advertise the Sale as a "going out of business," "store closing," "sale on everything," "everything must go," or similar themed sale.
- F. Agent shall be permitted to utilize display, hanging signs, and interior banners in connection with the Sale; <u>provided</u>, <u>however</u>, that such display, hanging signs, and interior banners shall be professionally produced and hung in a professional manner. The Merchant and the Agent shall not use neon or day-glo on its display, hanging signs, or interior banners. Furthermore, with respect to enclosed mall locations, no exterior signs or signs in common areas of a mall shall be used unless otherwise expressly permitted in these Sale Guidelines. In addition, the Merchant and the Agent shall be permitted to utilize exterior banners at (i) non-enclosed mall Stores and (ii) enclosed mall Stores to the extent the entrance to the applicable Store does not require entry into the enclosed mall common area; <u>provided</u>, <u>however</u>, that such banners shall be located or hung so as to make clear that the Sale is being conducted only at the affected Store, shall not be wider than the storefront of the Store, and shall not be larger than 4 feet x 40 feet. In addition, the Merchant and the Agent shall be permitted to utilize sign walkers in a safe and professional manner and in accordance with the terms of the Approval Order. Nothing contained in these Sale Guidelines shall be construed to create or impose upon the Agent any additional restrictions not contained in the applicable lease agreement.
- G. Conspicuous signs shall be posted in the cash register areas of each of the affected Stores to effect that "all sales are final."

- H. Except with respect to the hanging of exterior banners, the Agent shall not make any alterations to the storefront or exterior walls of any Stores.
- I. The Agent shall not make any alterations to interior or exterior Store lighting. No property of the landlord of a Store shall be removed or sold during the Sale. The hanging of exterior banners or in-Store signage and banners shall not constitute an alteration to a Store.
- The Agent shall keep Store premises and surrounding areas clear and orderly consistent with present practices.
- K. Subject to the provisions of the Agency Agreement the Agent shall have the right to sell all furniture, fixtures, and equipment located at the Stores and Distribution Center(s) (the "FF&E"). The Agent may advertise the sale of the FF&E in a manner consistent with these guidelines at the Stores and Distribution Center(s). The purchasers of any FF&E sold during the sale shall be permitted to remove the FF&E either through the back shipping areas at any time, or through other areas after Store business hours. For the avoidance of doubt, as of the Sale Termination Date, the Agent may abandon, in place and without further responsibility, any FF&E.
- L. The Agent shall be entitled to include Additional Agent Goods in the Sale in accordance with the terms of the Approval Order and the Agency Agreement.
- At the conclusion of the Sale at each Store, pending assumption or rejection of applicable leases, M. the landlords of the Stores shall have reasonable access to the Stores' premises as set forth in the applicable leases. The Merchant, the Agent and their agents and representatives shall continue to have exclusive and unfettered access to the Stores.
- Post-petition rents shall be paid by the Merchant as required by the Bankruptcy Code until the N. rejection or assumption and assignment of each lease. Agent shall have no responsibility therefor.
- The rights of landlords against Merchant for any damages to a Store shall be reserved in accordance with the provisions of the applicable lease.
- If and to the extent that the landlord of any Store affected hereby contends that the Agent or Merchant is in breach of or default under these Sale Guidelines, such landlord shall email or deliver written notice by overnight delivery on the Merchant's counsel and the Agent's counsel as follows:

If to the Merchant:

SIMPLY FASHION STORES, LTD. 2110 NW 95th Avenue Miami, Florida 33172 Attn: Swanpnil Shah

Tel.: (305) 592-2712 Email: sjs@mahudi.com

With a copy (which shall not constitute notice) to:

KapilaMukamal 1000 South Federal Highway, Suite 200 Fort Lauderdale, Florida 33316

Attn.: Soneet R. Kapila

Tel.: (954) 761-1011

Email: skapila@kapilamukamal.com

With a copy (which shall not constitute notice) to:

Berger Singerman LLP

1450 Brickell Avenue, Suite 1900

Miami, Florida 33131

Attn.: Paul Steven Singerman

Tel.: (305) 714-4343

Email: <u>Singerman@bergersingerman.com</u> Attn.: Christopher Andrew Jarvinen

Tel.: (305) 714-4343

Email: cjarvinen@bergersingerman.com

If to the Agent:

HILCO MERCHANT RESOURCES, LLC

5 Revere Drive, Suite 206 Northbrook, Illinois 60062 Attn.: Ian S. Fredericks

Tel.: (847) 418-2075 Fax: (847) 897-0859

Email: ifredericks@hilcotrading.com

GORDON BROTHERS RETAIL PARTNERS, LLC

Prudential Tower 800 Boylston Street

Boston, Massachussets 02119 Attn.: Michael Chartock Tel.: (617) 210-7116

Email: mchartock@gordonbrothers.com

Email. menartock@gordonoromers.com

With a copy (which shall not constitute notice) to:

Bilzin Sumberg Baena Price & Axelrod, LLP

1450 Brickell Avenue, Suite 2300

Miami, Florida 33131 Attn.: Scott L. Baena Tel.: (305) 350-2403 Email: sbaena@bilzin.com

Attn.: Jay S. Sakalo Tel.: (305) 375-6156 Email: jsakalo@bilzin.com

Exhibit B

Sale Guidelines

SALE GUIDELINES

- A. The Sale shall be conducted so that the Stores in which sales are to occur will remain open no longer than during the normal hours of operation provided for in the respective leases for the Stores.
- B. The Sale shall be conducted in accordance with applicable state and local "Blue Laws," where applicable, so that no Sale shall be conducted on Sunday unless the Merchant had been operating such Store on a Sunday.
- C. On "shopping center" property, the Agent shall not distribute handbills, leaflets or other written materials to customers outside of any Stores' premises, unless permitted by the lease or, if distribution is customary in the "shopping center" in which such Store is located; provided that Agent may solicit customers in the Stores themselves. On "shopping center" property, the Agent shall not use any flashing lights or amplified sound to advertise the Sale or solicit customers, except as permitted under the applicable lease or agreed to by the landlord.
- D. At the conclusion of the Sale, the Agent shall vacate the Stores in broom clean condition, and shall leave the Locations in the same condition as on the Sale Commencement Date, ordinary wear and tear excepted, in accordance with Section 6 of the Agency Agreement, provided, however, that the Merchant and the Agent hereby do not undertake any greater obligation than as set forth in an applicable lease with respect to a Store. The Agent and Merchant may abandon any FF&E not sold in the Sale at the Stores at the conclusion of the Sale. Any abandoned FF&E left in a Store after a lease is rejected shall be deemed abandoned to the landlord having a right to dispose of the same as the landlord chooses without any liability whatsoever on the part of the landlord to any party and without waiver of any damage claims against the Merchant. For the avoidance of doubt, as of the Sale Termination Date, the Agent may abandon, in place and without further responsibility, any FF&E located at a Store.
- E. The Merchant and the Agent may advertise the Sale as a "going out of business," "store closing," "sale on everything," "everything must go," or similar themed sale.
- F. Agent shall be permitted to utilize display, hanging signs, and interior banners in connection with the Sale; provided, however, that such display, hanging signs, and interior banners shall be professionally produced and hung in a professional manner. The Merchant and the Agent shall not use neon or day-glo on its display, hanging signs, or interior banners. Furthermore, with respect to enclosed mall locations, no exterior signs or signs in common areas of a mall shall be used unless otherwise expressly permitted in these Sale Guidelines. In addition, the Merchant and the Agent shall be permitted to utilize exterior banners at (i) non-enclosed mall Stores and (ii) enclosed mall Stores to the extent the entrance to the applicable Store does not require entry into the enclosed mall common area; provided, however, that such banners shall be located or hung so as to make clear that the Sale is being conducted only at the affected Store, shall not be wider than the storefront of the Store, and shall not be larger than 4 feet x 40 feet. In addition, the Merchant and the Agent shall be permitted to utilize sign walkers in a safe and professional manner and in accordance with the terms of the Approval Order. Nothing contained in these Sale Guidelines shall be construed to create or impose upon the Agent any additional restrictions not contained in the applicable lease agreement.
- G. Conspicuous signs shall be posted in the cash register areas of each of the affected Stores to effect that "all sales are final."
- H. Except with respect to the hanging of exterior banners, the Agent shall not make any alterations to the storefront or exterior walls of any Stores.

- I. The Agent shall not make any alterations to interior or exterior Store lighting. No property of the landlord of a Store shall be removed or sold during the Sale. The hanging of exterior banners or in-Store signage and banners shall not constitute an alteration to a Store.
- J. The Agent shall keep Store premises and surrounding areas clear and orderly consistent with present practices.
- K. Subject to the provisions of the Agency Agreement the Agent shall have the right to sell all furniture, fixtures, and equipment located at the Stores and Distribution Center(s) (the "FF&E"). The Agent may advertise the sale of the FF&E in a manner consistent with these guidelines at the Stores and Distribution Center(s). The purchasers of any FF&E sold during the sale shall be permitted to remove the FF&E either through the back shipping areas at any time, or through other areas after Store business hours. For the avoidance of doubt, as of the Sale Termination Date, the Agent may abandon, in place and without further responsibility, any FF&E.
- L. The Agent shall be entitled to include Additional Agent Goods in the Sale in accordance with the terms of the Approval Order and the Agency Agreement.
- M. At the conclusion of the Sale at each Store, pending assumption or rejection of applicable leases, the landlords of the Stores shall have reasonable access to the Stores' premises as set forth in the applicable leases. The Merchant, the Agent and their agents and representatives shall continue to have exclusive and unfettered access to the Stores.
- N. Post-petition rents shall be paid by the Merchant as required by the Bankruptcy Code until the rejection or assumption and assignment of each lease. Agent shall have no responsibility therefor.
- O. The rights of landlords against Merchant for any damages to a Store shall be reserved in accordance with the provisions of the applicable lease.
- P. If and to the extent that the landlord of any Store affected hereby contends that the Agent or Merchant is in breach of or default under these Sale Guidelines, such landlord shall email or deliver written notice by overnight delivery on the Merchant's counsel and the Agent's counsel as follows:

If to the Merchant:

SIMPLY FASHION STORES, LTD. 2110 NW 95th Avenue Miami, Florida 33172 Attn: Swanpnil Shah

Tel.: (305) 592-2712 Email: sjs@mahudi.com

With a copy (which shall not constitute notice) to:

KapilaMukamal 1000 South Federal Highway, Suite 200 Fort Lauderdale, Florida 33316

Attn.: Soneet R. Kapila Tel.: (954) 761-1011

Email: skapila@kapilamukamal.com

With a copy (which shall not constitute notice) to:

Berger Singerman LLP 1450 Brickell Avenue, Suite 1900

Miami, Florida 33131

Attn.: Paul Steven Singerman

Tel.: (305) 714-4343

Email: <u>Singerman@bergersingerman.com</u> Attn.: Christopher Andrew Jarvinen

Tel.: (305) 714-4343

Email: cjarvinen@bergersingerman.com

If to the Agent:

HILCO MERCHANT RESOURCES, LLC

5 Revere Drive, Suite 206 Northbrook, Illinois 60062 Attn.: Ian S. Fredericks Tel.: (847) 418-2075

Fax: (847) 897-0859

Email: ifredericks@hilcotrading.com

GORDON BROTHERS RETAIL PARTNERS, LLC

Prudential Tower 800 Boylston Street Boston, Massachussets 02119

Attn.: Michael Chartock Tel.: (617) 210-7116

Email: mchartock@gordonbrothers.com

With a copy (which shall not constitute notice) to:

Bilzin Sumberg Baena Price & Axelrod, LLP 1450 Brickell Avenue, Suite 2300

Miami, Florida 33131 Attn.: Scott L. Baena Tel.: (305) 350-2403 Email: sbaena@bilzin.com Attn.: Jay S. Sakalo

Tel.: (305) 375-6156 Email: <u>jsakalo@bilzin.com</u>

Exhibit C

Sale Order Notice Parties

In re Adinath Corp. Case No (15-16885) Creditor Matrix

Name	Address 1	Address 2	City	State	PostalCode	Country
Aronov Realty Management, Inc.	C/O Holland & Knight LLP	Attn: Eric B. Funt, Esq.	Miami	FL	33131	
1 MODEL MANAGEMENT, LLC	42 BOND STREET	2ND FLOOR	NEW YORK	NY	10012	
1002-1040 CAMP JACKSON						
RECEIVERSHIP	C/O FORESITE REALTY MGMT	9525 BRYN MAWR AVE.	ROSEMONT	IL	60018	
1179 E. MAIN STREET						
REDEVELOPMENT LLC	1051 EAST MAIN STREET		COLUMBUS	ОН	43205	
1179 E. MAIN STREET						
REDEVELOPMENT, LLC	1000 E. Main Street		Columbus	ОН	43205	
1996 PARKWAY PLAZA						
PARTNERSHIP, LLC	C/O BEAU BOX PROPERTY MGMT	PO BOX 66865	BATON ROUGE	LA	70896	
1ST CAPITAL CONTRACTORS	21484 KINGDOM EDGE DRIVE		KINGWOOD	TX	77339	
2525 EAST HILLSBOROUGH AVENUE,						
LLC	Scott Hall, SVP	c/o UrbanAmerica, L.P.,	New York	NY	10004	
3045 EAST TEXAS STREET, LLC	C/O SALE COUNCIL	P.O. BOX 6212	BOSSIER CITY	LA	71171	
3045 EAST TEXAS STREET, LLC	C/O SALE COUNCIL	P O Box 6212	Bossier City	LA	71171	
3330 WEST 183RD STREET	C/O MID-AMERICA ASSET MGMT					
HOLDINGS LLC	INC.	ONE PARKVIEW PLZ	OAKBROOK TERRACE	IL	60181	
	C/O THALHEIMER, ATTN: CC-1905-					
452 LLC	16-02	P.O. BOX 5160	GLEN ALLEN	VA	23058-5160	
452 LLC	c/o THALHIMER	ATTN: Commercial Accounting	Richmond	VA	23218-0702	
456 MEETING STREET, LLC	1505 JULIET DRIVE		ALLEN	TX	75013	
456 MEETING STREET, LLC as						
successor to DOROTHY Y. ENGLE and						
RALPH S. ENGLE	Dorothy Y. Engle & Ralph S. Engle	211 Ashley Avenue	Charleston	SC	29403	
5335 NORTH MILITARY, LLC	C/O SYMPATICO REAL ESTATE. LLC	222 LAKEVIEW AVE., PH5	WEST PALM BEACH	FL	33401	
	C/O METRO COMMERCIAL MGMT					
5610 ASSOCIATES, LLC	SERVICES	303 FELLOWSHIP RD	MT LAUREL	NJ	08054	
5610 Associates, LLC	OVERBROOK PLAZA/Store No49	T/A Overbrook Plaza	Blue Bell	PA	19422-2134	
5610 ASSOCIATES, LLC	303 FELLOWSHIP RD # 202		MOUNT LAUREL	NJ	08055	
5610 ASSOCIATES, LLC	OVERBROOK PLAZA	C/O METRO COMMERCIAL MGMT	MOUNT LAUREL	NJ	08054	
5610 ASSOCIATES, LLC T/A						
OVERBROOK PLAZA	Overbrook Plaza	5610 Lancaster Avenue	Philadelphia	PA	19131	
	C/O VICTORY REAL ESTATE					
59 WEST INVESTORS LLC	INVESTMENT LLC	240 BROOKSTONE CENTRE PKWY	COLUMBUS	GA	31904	

In re Adinath Corp. Case No (15-16885) Creditor Matrix

Name	Address 1	Address 2	City	State	PostalCode	Country
Name	Address 1	Address 2	city	State	PostalCode	Country
6000 WOODLAND PARTNERS, LP	C/O DANELLA REALTY & MGMT CO	PO BOX 1017	BLUE BELL	PA	19422	
0000 WOODLAND FARTNERS, EF	C/O DANCELA REALTT & WIGHT CO	c/o Daniella Realty & Management	BLOL BLLL	FA	13422	
6000 Woodland Partners, LP	WOODLAND VILLAGE/Store No274	Co.	Blue Bell	PA	19422	
oooo woodiana rartiicis, Ei	c/o Danella Realty & Management		Dide Dell	17	13422	
6000 WOODLAND PARTNERS, LP	Co	P.O. Box 1017	Blue Bell	PA	19422	
OCCUPATION TO THE TOTAL OF THE	JACOB KERKER, MAX MALEH,	1.0. BOX 1017	Dide bell	171	13422	
63 PEACHTREE STREET LLC	DAVID MALEH,	AND ALBERT ANTEBI	ATLANTA	GA	30339	
7 LIONS	1312 S BOYLE AVE, UNIT A	7 IND TEDERIT THE PERSON NAMED IN THE PERSON N	LOS ANGELES	CA	90023	
805 REALTY CORP.	C/O VISION REAL ESTATE	5544 FRANKLIN RD, STE 200	NASHVILLE	TN	37220	
9111 JENSON. LLC	C/O COMMERCIAL PROP MGMT	5615 RICHMOND AVE, STE 275	HOUSTON	TX	77057	
A & A AIR CONDITIONING	PO BOX 12084	3013 111011110112 7112 273	ALEXANDRIA	LA	71315-2084	
A & K SERVICES	PO BOX 64633		FAYETTEVILLE	NC	28306	
A LOCKSMITH SHOP, INC	2200 BROADWAY		RIVIERA BEACH	FL	33404	
A RAINBOW OF CARPETS	5044 A COOSAW CREEK BLVD.		N.CHARLESTON	SC	29420	
711111111111111111111111111111111111111	SCTTA COCCATA CALLER SETS.		111011111111111111111111111111111111111	55	23 .20	
A Roland Kimbrell Investments, LLC	CLOVERLEAF S/C/Store No301	P.O. Box 3007	Meridian	MS	39302	
A TO Z STATEWIDE PLUMBING, INC.	2215 S.W. 58TH TERRACE		HOLLYWOOD	FL	33023	
A&A AIR CONDITIONING	PO BOX 12084		ALEXANDRIA	LA	71315	
A. ROLAND KIMBRELL INVESTMENTS						
LLC	P.O. BOX 3007		MERIDIAN	MS	39302	
A. ROLAND KIMBRELL						
INVESTMENTS, LLC	P O BOX 3007		Meridian	MS	39302	
A1 ABC LOCK & SAFE	5218 S. SHEPHERD		HOUSTON	TX	77098	
A-1 PLUMBING HEATING & SEWER	4470 BALLARD ROAD		FOREST PARK	GA	30297	
A-24 HOUR DOOR AMERICA	TDI	4116 BENT ROAD	KODAK	TN	37764	
AAA GLASS CO., INC	930 ROBESON STREET		FAYETTEVILLE	NC	28305	
AAA LOCK & KEY SERVICE, INC	4013 BASELINE ROAD		LITTLE ROCK	AR	72209	
AAA SAFE & LOCK COMPANY, INC	722 SOUTH HIGHLAND		MEMPHIS	TN	38111	
AAA SAFE CO.	13242 HWY 17		OAK GROVE	LA	71263	
ABACUS IT SOLUTIONS, LLC	2189 PARKWAY LAKE DR.		BIRMINGHAM	AL	35244	
ABC FIRE EQUIPMENT, INC.	P.O. BOX 205		NORTHPORT	AL	35476	
ABCO LAND DEVELOPMENT CORP.	C/O CREST COMMERCIAL REALTY	122 NORTH JEFFERSON ST.	DAYTON	ОН	45402	

In re Adinath Corp. Case No (15-16885) Creditor Matrix

Name	Address 1	Address 2	City	State	PostalCode	Country
ABCO LAND DEVELOPMENT CORP.	122 North Jefferson Street		Dayton	ОН	45402	
ABERCROMBIE, NYKISHA	REDACT		REDACT	REDACT	REDACT	
ABLE ELEVATOR SERVICES, INC	6600 YORK ROAD	SUITE 105	BALTIMORE	MD	21212	
ABLE PLUMBING & CONTRACTING,						
INC.	24410 JOHN R		HAZEL PARK	MI	48030-1114	
ABRAHA, RIM	REDACT		REDACT	REDACT	REDACT	
ABRAM, AUSHIJZANTE	REDACT		REDACT	REDACT	REDACT	
ABSOLUTE ANGEL	1407 BROADWAY	SUITE 2510	NEW YORK	NY	10018	
ABSOLUTE GLASS & MIRROR	208 EASTGATE DR.		WHITEHOUSE	TX	75791	
ABULIZZ, HEBA	REDACT		REDACT	REDACT	REDACT	
AC GLASS COMPANY	11112 WASHINGTON HIGHWAY		GLEN ALLEN	VA	23059	
Acadia Realty Trust	Attn: Dustin P. Branch, Esq	C/O Katten Muchin Rosenman LLP	Los Angeles	CA	90067-3012	
ACCENT GLASS, INC	4031 SOUTH BLVD	·	CHARLOTTE	NC	28209	
ACCI	2496 VALLEYDALE RD		BIRMINGHAM	AL	35244	
ACCUPRINT	2023 FIRST AVENUE NORTH		BIRMINGHAM	AL	35203	
ACCURATE LOCK & KEY	PO BOX 1473		SUMTER	SC	29151	
ACH Pine Bluff, LLC	350 Pine Street	Suite 800	Beaumont	TX	77701	
	C/O ALBANESE CORMIER					
ACH PINE BLUFF, LLC.	HOLDINGS, LLC	350 PINE STREET	BEAUMONT	TX	77701	
ACHA, GENEVIEVE	REDACT		REDACT	REDACT	REDACT	
ACME LOCK & KEY	P.O. BOX 610763		BIRMINGHAM	AL	35261-0763	
ACME LOCK COMPANY	PO BOX 850165		NEW ORLEANS	LA	70185	
ACTION LOCKSMITH SERVICE, LLC	207 MADISON AVENUE		MONTGOMERY	AL	36104-3625	
ACTIVE FOOTWEAR, INC.	10 W 33RD ST, STE 1111		NEW YORK	NY	10001	
ACTIVE USA, INC	1807 E 48TH PL		LOS ANGELES	CA	90058	
ADAMS & COMPANY REAL EST LLC.	411 5TH AVE, 9TH FLOOR		NEW YORK	NY	10016	
ADAMS & COMPANY REAL ESTATE,						
LLC	ACCT #05310400	PO BOX 95000-5385	PHILADELPHIA	PA	19195-5385	
ADAMS COUNTY	TAX COLLECTOR	P.O. BOX 1128	NATCHEZ	MS	39121	
ADAMS ELECTRICAL SYSTEMS LLC	P.O. BOX 290153		COLUMBIA	SC	29229	
ADAMS, BRITTANY	REDACT		REDACT	REDACT	REDACT	
ADAMS, DESTINY	REDACT	REDACT	REDACT	REDACT	REDACT	
ADAMS, HASAN	REDACT		REDACT	REDACT	REDACT	
ADAMS, KARRI	REDACT	REDACT	REDACT	REDACT	REDACT	

Name	Address 1	Address 2	City	State	PostalCode	Country
ADAMS, KEJUANNA	REDACT		REDACT	REDACT	REDACT	
ADAMS, MONIQUE	REDACT		REDACT	REDACT	REDACT	
ADDISON, ALKISHA	REDACT		REDACT	REDACT	REDACT	
Adinath Corp.	2110 N.W. 95th Avenue		Miami	FL	33172	
ADRIONNE WHOOPER	1421 RANDALL DRIVE		MEMPHIS	TN	38116	
ADVANCE FINANCIAL CORP	PO BOX 720477		ATLANTA	GA	30358-2477	
ADVANCE PLUMBING & HEATING,						
INC.	616 FRANKLIN STREET		GRETNA	LA	70056	
ADVANCED AIR & HEATING, INC	2316 TRUMAN AVENUE		PENSACOLA	FL	32505	
ADVANCED DISPOSAL NATIONAL						
ACCTS	P.O. BOX 6003		PONTE VEDRA BEACH	FL	32004-6003	
ADVANCED ELECTRIC COMPANY, INC	359 CERNY AVENUE		EUREKA	МО	63025-2128	
ADVANCED MECHANICAL SERVICES	PO BOX 68		FAIRBORN	ОН	45324	
ADVANCED SERVICES	1208 NORTH KURZWEIL ROAD		RAYMORE	MO	64083	
ADVANTAGE SERVICE COMPANY	1797 N. CYPRESS STREET		N. LITTLE ROCK	AR	72114	
AES (GARNISHMENT)	C/O GC SERVICES, LP	PO BOX 32500	COLUMBUS	ОН	43232	
AFFORDABLE ELECTRIC, INC. (IL)	33 DEVONSHIRE DR.	1.0.200.32300	OAK BROOK	IL	60523	
AFLAC	REMITTANCE PROCESSING	1932 WYNNTON ROAD	COLUMBUS	GA	31999	
AFRAM, JOANNE	REDACT	1332 11 11111 011 1107 12	REDACT	REDACT	REDACT	
AGAIN TRADING	P.O. Box 1036		Charlotte	NC	28201-1036	
AGEE, RYAN	REDACT		REDACT	REDACT	REDACT	
	ACCOUNTS RECEIVABLE					
AGENCY GERARD, INC.	DEPARTMENT	1554 PAOLI PIKE #273	WEST CHESTER	PA	19380-6123	
AGGRESSIVE APPAREL	1412 BROADWAY, 7TH FLR		NEW YORK	NY	10018	
AGUILAR, GABRIELA	REDACT		REDACT	REDACT	REDACT	
AGUILERA, TERI	REDACT		REDACT	REDACT	REDACT	
AIDENBAUM SCHLOFF & BLOOM						
PLLC	6960 ORCHARD LAKE ROAD	SUITE 250	WEST BLOOMFIELD	МІ	48322	
AIG BENEFIT SOLUTIONS	P.O. BOX 30066		TAMPA	FL	33630-3066	
AIR CONDITIONING SERVICE, INC.	P.O. BOX 111089	414 MCNALLY DRIVE	NASHVILLE	TN	37222-1089	
AIR TECHNICAL SERVICES, INC	2856 CELA ROAD		MEMPHIS	TN	38128	1
AISHA WARREN	18247 PIERRE DRIVE		CLINTON TOWNSHIP	MI	48038	1
AKINS, DIAMOND	REDACT		REDACT	REDACT	REDACT	1
ALABAMA CHILD SUPPORT PYMT						1
CTR	PO BOX 244015		MONTGOMERY	AL	36124	

Name	Address 1	Address 2	City	State	PostalCode	Country
ALABAMA COOLING & HEATING	6111 TOWHEE DR.		PINSON	AL	35126	
ALABAMA DEPARTMENT OF REV						
(PPT)	BUSINESS PRIVILEGE TAX UNIT	PO BOX 327320	MONTGOMERY	AL	36132-7320	
Alabama Department of Revenue	Sales and Use Tax Division	PO Box 327790	Montgomery	AL	36132-7790	
-						
Alabama Department of Revenue	Gordon Persons Bldng	50 North Ripley Street	Montgomery	AL	36104	
ALABAMA DEPT. OF REVENUE	SALES AND USE TAX DIVISION	P.O. BOX 327900	MONTGOMERY	AL	36132	
ALABAMA LOCK & KEY CO., INC	PO BOX 1056		BIRMINGHAM	AL	35201	
ALABAMA POWER	600 18th Street N		BIRMINGHAM	AL	35203	
ALABAMA POWER	PO BOX 242		BIRMINGHAM	AL	35292	
ALABAMA POWER (SIGN)	PO BOX 242		BIRMINGHAM	AL	35292	
ALABAMA RETAIL ASSOCIATION	PO BOX 240669		MONTGOMERY	AL	36124	
ALABAMA STATE BAR- LICENSE	PO BOX 671		MONTGOMERY	AL	36101	
		Michele L. Lieberman - County				
Alachua County	Attn: Consumer Protection Agency	Attorney	Gainesville	FL	32601	
ALAGASCO	20 South 20th St		BIRMINGHAM	AL	35295	
ALAGASCO	PO BOX 2224		BIRMINGHAM	AL	35246-0022	
AL-AMM, LOUANA	REDACT		REDACT	REDACT	REDACT	
Alan Betten, Esquire	CHURCH SQUARE S/C/Store No12	600 Washington Avenue, Suite 300	Towson	MD	21204	
ALANIS, PRISCILLA	REDACT		REDACT	REDACT	REDACT	
ALANIZ, CRYSTAL	REDACT		REDACT	REDACT	REDACT	
ALBANY CROSSINGS, LLC	101 STELLA STREET		METAIRIE	LA	70005	
ALBANY DOOR AND HARDWARE	204 POPLAR STREET		ALBANY	GA	31707	
ALBANY ELECTRIC	P.O. BOX 5228		ALBANY	GA	31706	
ALBANY LOCK & SAFE CO.	702 N. SLAPPEY BLVD		ALBANY	GA	31701	
ALBANY WATER GAS & LIGHT						
COMMISSION	207 PINE AVE		ALBANY	GA	31701	
ALBANY WATER GAS & LIGHT						
COMMISSION	PO BOX 1788		ALBANY	GA	31702-1788	
ALBANY/DOUGHERTY COUNTY TAX						
DEPT	PO BOX 447	SUITE 150	ALBANY	GA	31702-0447	
Albert Chehebar	c/o Trendset	1407 Broadway	New York	NY	10018	
ALBERT, ANQUANETTE	REDACT		REDACT	REDACT	REDACT	
ALBERT, BOB	REDACT		REDACT	REDACT	REDACT	
ALBINO, DAVID	REDACT		REDACT	REDACT	REDACT	
ALDINE ISD TAX OFFICE	P.O. BOX 203989		HOUSTON	TX	77216-3989	

Page 5 of 176

Name	Address 1	Address 2	City	State	PostalCode	Country
ALEC, LLC- ELECTRICAL						
CONTRACTORS	1919 NORTH FLANNERY ROAD		BATON ROUGE	LA	70815	
ALEXANDER, BRANDY	REDACT		REDACT	REDACT	REDACT	
ALEXANDER, BRITTNEY	REDACT		REDACT	REDACT	REDACT	
ALEXANDER, KIRSTEN	REDACT		REDACT	REDACT	REDACT	
ALEXANDER, SHARRELLE	REDACT		REDACT	REDACT	REDACT	
ALEXANDER, SHERRY	REDACT		REDACT	REDACT	REDACT	
ALEXIS GANTT	2901 PEBBLE CREEK PKWY		BIRMINGHAM	AL	35214	
ALFARO, JOANA	REDACT		REDACT	REDACT	REDACT	
ALGARIN, AMANDA	REDACT		REDACT	REDACT	REDACT	
ALHASNAWY, ZAHRA	REDACT		REDACT	REDACT	REDACT	
ALIEF ISD TAX OFFICE	P.O. BOX 368		ALIEF	TX	77411	
ALISON R. TRAMMELL	175 WOODWARD STREET		BESSEMER	AL	35020	
ALL AMERICAN GLASS & MIRROR	5296 OLD HIGHWAY 11	SUITE 6	HATTIESBURG	MS	39402	
ALL CITY ELECTRIC (FOX LAKE)	PO BOX 638		FOX LAKE	IL	60020	
ALL COUNTY LOCK & KEY INC	5790 RODMAN STREET	SUITE # 4	HOLLYWOOD	FL	33023	
ALL FOR THE OFFICE	1612 MARALYN DRIVE		BIRMINGHAM	AL	35235	
ALL PRO GLASS & MIRROR	1517 KUEBEL STREET		HARAHAN	LA	70123	
ALL PRO PLUMBING, INC (MS)	534 FOWLER DRIVE		COLUMBUS	MS	39702	
ALL SAFE	SAFE & LOCK, INC.	1141 E BLUE HERRON BLVD	RIVIERA BEACH	FL	33404	
ALL SEASONS HEATING & AIR, INC	800 E. 10TH STREET		KANSAS CITY	MO	64106	
All Seasons Heating & Cooling	Attn.: Bruce Olds	800 E. 10th Street	Kansas City	Mo	64106	
ALL STATE FIRE EQUIPMENT	P.O. BOX 1963		INDEPENDENCE	MO	64055-1963	
ALLARD, MARY	REDACT		REDACT	REDACT	REDACT	
ALLEN ELECTRICAL CONRTACTORS						
INC.	2400 BOWMAN AVENUE		LOUISVILLE	KY	40217	
ALLEN, AMIN	REDACT		REDACT	REDACT	REDACT	
ALLEN, BETTYE	REDACT		REDACT	REDACT	REDACT	
ALLEN, DONNY	REDACT		REDACT	REDACT	REDACT	
ALLEN, LACRETA	REDACT		REDACT	REDACT	REDACT	
ALLEN, LINDA	REDACT	REDACT	REDACT	REDACT	REDACT	
ALLEN, PAIGE	REDACT		REDACT	REDACT	REDACT	
ALLEN, RASHEEK	REDACT		REDACT	REDACT	REDACT	
ALLEN, ROSEMARY	REDACT		REDACT	REDACT	REDACT	
ALLEN, SANDRA	REDACT		REDACT	REDACT	REDACT	
ALLEN, SHELEETA	REDACT		REDACT	REDACT	REDACT	

Name	Address 1	Address 2	City	State	PostalCode	Country
ALL-PHASE ELECTRIC COMPANY, INC	PO BOX 773		MILLEDGEVILLE	GA	31059	
ALL-SOUTH SUBCONTRACTORS	PO BOX 101657		BIRMINGHAM	AL	35210	
ALL-STAR ELECTRICAL OF RALEIGH,						
INC.	PO BOX 1175		GARNER	NC	27529	
ALLTECH MECHANICAL, INC	232 S. PRAIRIE AVENUE		ARLINGTON HEIGHTS	IL	60005	
ALMOND, LAQUESHA	REDACT		REDACT	REDACT	REDACT	
ALPHA TRAFFIC SERVICE (RETAINER)	C/O ALPHA TRAFFIC SERVICES	PO BOX 293	MT. OLIVE	AL	35117	
ALPHA TRAFFIC SERVICES	PO BOX 293		MT. OLIVE	AL	35117	
ALPINE MECHANICAL SERVICES	1112 OLD BETHLEHEM PIKE		SELLERSVILLE	PA	18960	
ALSTON, ASIA	REDACT		REDACT	REDACT	REDACT	
ALSTON, TERESA	REDACT		REDACT	REDACT	REDACT	
ALTA MESA LP	C/O ISENBERG MANAGEMENT	400 SOUTH ZANG, SUITE 1220	DALLAS	TX	75208	
ALTIMA LIGHTING, INC	8361 CURRENCY DRIVE		RIVIERA BEACH	FL	33404	
ALVAREZ, MARIA	REDACT		REDACT	REDACT	REDACT	
AMANDA MITCHELL	1616 SPANGLER RD NE		CANTON	ОН	44714	
AMARO, DERRICK	REDACT		REDACT	REDACT	REDACT	
AMBIANCE APPAREL	930 TOWNE AVE		LOS ANGELES	CA	90021	
AMBROSE, CASSANDRA	REDACT		REDACT	REDACT	REDACT	
AMCB PERRING, LLC	P.O. BOX 8000	DEPT. #456	BUFFALO	NY	14267	
AMCB Perring, LLC	BALTIMORE/Store No237	c/o Acadia Realty Trust	White Plains	NY	10605	
AMCB Perring, LLC	BALTIMORE/Store No237	c/o Acadia Realty Trust	White Plains	NY	10605	
AMEREN ILLINIOS	1901 Chouteau Ave		ST LOUIS	MO	63103	
AMEREN ILLINIOS	P.O. BOX 88034		CHICAGO	IL	60680-1034	
AMEREN ILLINOIS	PO BOX 88034		CHICAGO	IL	60680-1034	
AMEREN MISSOURI	1132 LOCUST ST.		ST LOUIS	MO	63101	
AMEREN MISSOURI	PO BOX 88068		CHICAGO	IL	60680-1068	
AMERICAN APPAREL GLOBAL CORP.	209 W. 38TH ST.		NY	NY	10018	
AMERICAN BOTTOMS	1 AMERICAN BOTTOM ROAD		E. ST. LOUIS	IL	62201	
AMERICAN BOTTOMS	PO BOX 459		E. ST. LOUIS	IL	62202-0459	
AMERICAN DIVERSIFIED						
DEVELOPMENTS &	NEWPORT ASSOC.	PO BOX 399	NEWPORT	KY	41072	
AMERICAN DREAM	2320 E. OLYMPIC BLVD		LOS ANGELES	CA	90021	
AMERICAN ELEC POWER	1 RIVERSIDE PLAZA		COLUMBUS	ОН	43215-2372	
AMERICAN ELECTRIC OF VIRGINIA	412 BEHL COURT		VIRGINIA BEACH	VA	23464	

Name	Address 1	Address 2	City	State	PostalCode	Country
AMERICAN ELECTRIC POWER	P.O. BOX 24002		CANTON	ОН	44701-4002	,
AMERICAN ELECTRIC POWER	P.O. BOX 24418		CANTON	ОН	44701-4418	
AMERICAN EXCHANGE	1407 BROADWAY	STE 2100	NEW YORK	NY	10018	
AMERICAN EXPRESS (TX)	PO BOX 650448		DALLAS	TX	75265-0448	
AMERICAN FIRE PROTECTION, INC	5525 EASTCLIFF INDUSTRIAL LOOP		BIRMINGHAM	AL	35210	
AMERICAN FUSION	796 E 14TH PLACE		LOS ANGELES	CA	90021	
AMERICAN INTIMATES	366 5TH AVE	STE 1011	NEW YORK	NY	10001	
AMERICAN MECHANICAL GROUP	5729 WESTBOURNE AVENUE		COLUMBUS	ОН	43213	
AMERICAN SECURITY & LOCK INC	608 W JOHNSON STREET	SUITE 15	RALEIGH	NC	27603	
AMERICAN THREADS	796 E. 14TH PLACE		LOS ANGELES	CA	90021	
AMERICAN UNITED LIFE	PO BOX 6123		INDIANAPOLIS	IN	46206-6123	
AMERICAN WATER & ENERGY						
SAVERS	4431 NORTH DIXIE HIGHWAY		BOCA RATON	FL	33431	
	Realnet Investments Management					
	LLC, Its duly authorized agent By:	C/O Director of Property				
AMERICAN WAY OWNER, LLC	Mitch Seitzinger, Manager	Management	Memphis	TN	38124-1990	
AMERICANA PLAZA, L.P.	9816 S. Military Trail	Suite C-2.3	Boynton Beach	FL	33436	
	C/O GULFSTREAM PROP MGMT,					
AMERICANA PLAZA, LP	LLC	9804 S. MILITARY TRAIL	BOYNTON BEACH	FL	33436	
AMERICASH LOANS, LLC	C/O CHECKBOOK LOAN DEPT	P.O. BOX 184	DES PLAINES	IL	60016	
AMERIS S. WALLACE	1200 CHAPEL RIDGE DR	#1221	GARDENDALE	AL	35071	
AMIEE LYNN	PO BOX 75359		CHICAGO	IL	60675-5359	
AMMONS, SHANNON	REDACT		REDACT	REDACT	REDACT	
AMS BILLING SERVICES	6915 15th ST E #204		SARASOTA	FL	34243	
AMS BILLING SERVICES	PO BOX 1047		TALLEVAST	FL	34270-1047	
ANCHOR INNOVATIVE, LLC	2412 2ND AVENUE NORTH # 6		BIRMINGHAM	AL	35203	
ANCRUM, ALPHONSO	REDACT		REDACT	REDACT	REDACT	
ANCRUM, CANDICE	REDACT		REDACT	REDACT	REDACT	
ANCRUM, LOLITA	REDACT		REDACT	REDACT	REDACT	
AND APPAREL	16001 S SAN PEDRO ST, #B		GARDENA	CA	90248	
ANDERSON, BAKEISHA	REDACT		REDACT	REDACT	REDACT	
ANDERSON, BRANDY	REDACT		REDACT	REDACT	REDACT	
ANDERSON, BRIDGIT	REDACT	REDACT	REDACT	REDACT	REDACT	
ANDERSON, COLBY	REDACT		REDACT	REDACT	REDACT	
ANDERSON, FELECIA	REDACT		REDACT	REDACT	REDACT	
ANDERSON, LATRICIA	REDACT		REDACT	REDACT	REDACT	

Page 8 of 176

Name	Address 1	Address 2	City	State	PostalCode	Country
ANDERSON, LOIS	REDACT	REDACT	REDACT	REDACT	REDACT	
ANDERSON, RODNEIKA	REDACT		REDACT	REDACT	REDACT	
ANDRE, SINDY	REDACT		REDACT	REDACT	REDACT	
ANDREA HOWARD	1209 BENTON APT 1-S		KANSAS CITY	МО	64127	
ANDREW ASCH & ASSOCIATES	110 EAST NINTH STREET	SUITE A-669	LOS ANGELES	CA	90079	
ANDREWS, RACHAEL	REDACT		REDACT	REDACT	REDACT	
ANGELA FASHION	800 EAST 12TH STREET # 102		LOS ANGELES	CA	90021	
ANGIER, BRIANA	REDACT		REDACT	REDACT	REDACT	
ANNETTE CROWLEY	119 #6 SCHOOLHOUSE RD		CHARLTON	MA	01507	
ANNISTON INVESTORS, LLC	1825 MAIN ST.		WESTON	FL	33326	
ANTHONY, KIAMBER	REDACT		REDACT	REDACT	REDACT	
ANTHONY, TIONDRA	REDACT		REDACT	REDACT	REDACT	
ANTHONY, WILLANTE	REDACT	REDACT	REDACT	REDACT	REDACT	
ANTHONY, WILLONTE	REDACT		REDACT	REDACT	REDACT	
ANTOINETTE JONES	6115 NORTH TEUTONIA AVENUE		MILWAUKEE	WI	53209	
ANYTIME ELECTRIC, INC	PO BOX 1687		CRAWFORDVILLE	FL	32326	
APEX BILLING SOLUTIONS	901 FRONT AVENUE SUITE 208		COLUMBUS	GA	31901	
APEX BILLING SOLUTIONS	PO BOX 84077		COLUMBUS	GA	31908-4077	
A-PINK USA	946 E 29TH ST		LOS ANGELES	CA	90011	
A-PLUS SERVICES, INC	PO BOX 53666		LAFAYETTE	LA	70505	
APONTE, RAYDA	REDACT		REDACT	REDACT	REDACT	
APPAREL DEALS	1155 S BOYLE AVE		LOS ANGELES	CA	90023	
APPLIN, MARIANNE	REDACT		REDACT	REDACT	REDACT	
ARBOR PLACE II, LLC	PO BOX 74883		CLEVELAND	ОН	44194-4883	
ARC INVESTMENTS, LLC	4044 OLD GENTILLY RD		NEW ORLEANS	LA	70126-4813	
ARCADIA LAKES PROPERTIES, INC.	C/O REAGAN REALTY GROUP, INC.	PO BOX 1297	GREENVILLE	SC	29602	
ARCENEAUX, ANETRIA	REDACT	REDACT	REDACT	REDACT	REDACT	
ARDIS, CAROLYN	REDACT		REDACT	REDACT	REDACT	
ARENDALL LAW FIRM, INC.	2018 MORRIS AVE		BIRMINGHAM	AL	35203	
ARIAS, SARA	REDACT		REDACT	REDACT	REDACT	
ARIEL BELVEDERE OWNER, LLC	BOX 538383		ATLANTA	GA	30353-8383	
ARIHANT INTERNATIONAL	2110 NW 95TH AVENUE		MIAMI	FL	33172	
Arihant International, LLC	2110 N.W. 95th Avenue		Miami	FL	33172	
,	6417 TRI-COUNTY PKWAY, SUITE					
ARION PERFUME AND BEAUTY, INC.	100		SCHERTZ	TX	78154	
ARK LA TEX FIRE SYSTEMS, LLC	PO BOX 78381		SHREVEPORT	LA	71137	
ARKAN JONNA	4036 TELEGRAPH RD.	SUITE 201	BLOOMFIELD HILLS	MI	48302	

Page 9 of 176

Name	Address 1	Address 2	City	State	PostalCode	Country
Arkansas Department of Finance and						
Administration	1509 West 7th Street		Little Rock	AR	72201	
Arkansas Dept of Finance and						
Administration	1509 West 7th Street		Little Rock	AR	72201	
ARKANSAS DEPT. OF FINANCE	SALES & USE TAX	1509 West 7th Street	LITTLE ROCK	AR	72203-8123	
ARKANSAS SECRETARY OF STATE	1401 W. CAPITAL	SUITE 250	LITTLE ROCK	AR	72201-1094	
ARMOR DOOR AND KEY, LLC	4014 CHOUTEAU		ST. LOUIS	МО	63110	
ARMOR LOCKSMITH, INC.	447 BRETT RIDGE ROAD		LAUREL	MS	39443	
ARMSTRONG ELECTRIC CO., INC.	SERVICE DEPARTMENT	4920 CHANEY STREET	PENSACOLA	FL	32503	
ARNEMAN, PATRICIA	REDACT		REDACT	REDACT	REDACT	
Aronov Realty Management, Inc.	C/O Holland & Knight LLP	Attn: Joaquin J. Alemany, Esq.	Miami	FL	33131	
ARONOV-BERLEF	DBA WESTERN HILLS MALL	P.O. BOX 23500	MONTGOMERY	AL	36123-5000	
ARROW EXTERMINATORS, INC.	P.O. BOX 48896		ATHENS	GA	30604-8896	
ARTCO GLOBAL GROUP	4902 CARLISLE PIKE #348		MECHANICSBURG	PA	17050	
ARTIST, JASPER	REDACT		REDACT	REDACT	REDACT	
ARTIST, JASPER	REDACT		REDACT	REDACT	REDACT	
ARVEST BANK	PO Box 1670		Lowell	AR	72745	
Ascension Parish Sale and Use Tax						
Authority	PO Box 1718		Gonzales	LA	70707	
ASHLEY BLUE/CSG APPAREL	57 WEST 38TH ST		NEW YORK	NY	10018	
Asiana Capital	P.O. Box 76327		Los Angeles	CA	90076	
ASKINS, MARY	REDACT	REDACT	REDACT	REDACT	REDACT	
ASSOCIATE ELECTRIC & BUILDING						
SERVICE, INC.	1222 W. 6TH STREET		LAUREL	MS	39440	
ASSURANCE CREDIT CORPORATION	6214 US HIGHWAY 49		HATTIESBURG	MS	39401	
AT&T	60663 SBC DRIVE		CHICAGO	IL	60663	
AT&T	85 ANNEX		ATLANTA	GA	30385	
AT&T	PO BOX 105262		ATLANTA	GA	30348-5262	
AT&T	PO BOX 5080		CAROL STREAM	IL	60197-5080	
ATLANTA HEATING & AIR				_		
CONDITIONING, INC.	5401 GLENCASTLE WAY		SUWANEE	GA	30024	
ATLANTIC HEATING & COOLING						
SERV.INC.	553 CENTRAL DRIVE		VIRGINIA BEACH	VA	23454	
ATLAS DOOR REPAIR (IL)	23900 W. INDUSTRIAL DRIVE S	SUITE # 1	PLAINFIELD	IL	60585	
ATMOS ENERGY	295 North Hubbards Lane		LOUISVILLE	KY	40207	
ATMOS ENERGY	PO BOX 790311		ST LOUIS	МО	63179-0311	
ATTERBERRY, ASHAWNTE	REDACT		REDACT	REDACT	REDACT	

Name	Address 1	Address 2	City	State	PostalCode	Country
	Attn.: Dewayne Ross, Vice					
AUDIO ELECTRONICS, INC	President of Operations	2501 WHITE AVENUE	NASHVILLE	TN	37204	
AUDUBON VENTURES LLC	C/O WOLFSON GROUP, INC.	120 W. Germantown Pike	Plymouth Meeting	PA	19462	
AUDUBON VENTURES, LLC	C/O WOLFSON GROUP, INC.	120 W. GERMANTOWN PIKE	PLYMOUTH MEETING	PA	19462	
AUGUSTA LICENSE & INSPECTION	PO BOX 9270		AUGUSTA	GA	30916-9270	
AUGUSTA UTILITIES DEPARTMENT	535 TELFAIR ST.		AUGUSTA	GA	30901	
AUGUSTA UTILITIES DEPARTMENT	PO BOX 1457		AUGUSTA	GA	30903-1457	
		c/o Developers Diversified Realty				
Austell, GA Center LLC	BURLINGTON PLAZA/Store No86	Corporation	Beachwood	ОН	44122	
		c/o Developers Diversified Realty				
Austell, GA Center LLC	BURLINGTON PLAZA/Store No86	Corporation	Beachwood	ОН	44122	
	c/o Developers Diversified Realty					
AUSTELL, GA CENTER LLC	Corporation	3300 Enterprise Parkway	Beachwood	ОН	44122	
AUSTELL, GA CENTER LLC	3870 MEDICAL PARK DRIVE		AUSTELL	GA	30106	
AUSTELL, GA CENTER LLC	PO BOX 228042		BEACHWOOD	ОН	44122	
AUSTELL, GA CENTER, LLC	DEPT 107196-21014-39503	PO BOX 228042	BEACHWOOD	ОН	44122	
AUSTIN, NATASHA	REDACT		REDACT	REDACT	REDACT	
AUSTIN, NATASHA	REDACT		REDACT	REDACT	REDACT	
AUSTINTOWN PLAZA LTD.	3729 UNION STREET		MINERAL RIDGE	ОН	44440	
AUSTINTOWN PLAZA, LTD.	3729 Union Street, PO Box 523		Mineral Ridge	ОН	44440	
AV DENIM	1407 BROADWAY		NEW YORK	NY	10018	
AVENT, SHAWANA	REDACT		REDACT	REDACT	REDACT	
AVERITT EXPRESS, INC.	P.O. BOX 3145		COOKEVILLE	TN	38502-3145	
AVERY DENNISON	15178 COLLECTIONS CENTER DRIVE		CHICAGO	IL	60693	
AVERY, RACHEL	REDACT		REDACT	REDACT	REDACT	
AVERY, TIFFANY	REDACT		REDACT	REDACT	REDACT	
	2232 CORPORATE SQUARE					
B & G PLUMBING CO.	BOULEVARD		JACKSONVILLE	FL	32216	
B & I MANAGEMENT	3729 UNION ST.		MINERAL RIDGE	ОН	44440	
B & I MANAGEMENT	A DIVISION OF P & S EQUITIES, INC.	PO BOX 523	MINERAL RIDGE	ОН	44440	
B & I MGMT COMPANY	3729 UNION STREET		MINERAL RIDGE	ОН	44440	
B C EXTINGUISHER, LLC	123 LAKEVIEW CIRCLE		CROPWELL	AL	35054	
B&Y FASHION	2416 E 16TH STREET		LOS ANGELES	CA	90021	

Page 11 of 176

Name	Address 1	Address 2	City	State	PostalCode	Country
B. Riley & Co., LLC	Steve Reiner	Managing Director	Los Angeles	CA	90025	
B.R. LAWNDALE PLAZA, LLC	DEPT LA 23084		PASADENA	CA	91185-3084	
B.R. LAWNDALE PLAZA, LLC	DEPT. LA 23084		Pasadena	CA	91185-3084	
BACCUS, SHAKYEAL	REDACT		REDACT	REDACT	REDACT	
BACON, LEVANTE	REDACT		REDACT	REDACT	REDACT	
BACON, SHACORRIUS	REDACT	REDACT	REDACT	REDACT	REDACT	
BAGLEY, ASHLEY	REDACT		REDACT	REDACT	REDACT	
BAGLEY, CAROLYN	REDACT		REDACT	REDACT	REDACT	
BAI CONSUMER SQUARE WEST LLC	P.O. BOX 713548		Cincinatti	ОН	45271-3548	
BAILEY, BRIANA	REDACT		REDACT	REDACT	REDACT	
BAILEY, CRYSTAL	REDACT		REDACT	REDACT	REDACT	
BAILEY, JAMEESHA	REDACT		REDACT	REDACT	REDACT	
BAILEY, SHAQUELLA	REDACT		REDACT	REDACT	REDACT	
BAILEY, SUSAN	REDACT		REDACT	REDACT	REDACT	
BAILEY, SYLVIA	REDACT		REDACT	REDACT	REDACT	
BAILEY, TERRY	REDACT		REDACT	REDACT	REDACT	
BAIN, CHARLES	REDACT		REDACT	REDACT	REDACT	
BAINED-BOLLING, ROBERT	REDACT		REDACT	REDACT	REDACT	
BAKER & BAKER	ATTN: MANAGING PARTNER	PO BOX 12397	COLUMBIA	SC	29211	
BAKER & BAKER REAL ESTATE						
DEVELOPERS, LLC	Baker & Baker	Managing Partner	Columbia	SC	29211-2397	
BAKER, ALIZE	REDACT		REDACT	REDACT	REDACT	
BAKER, ALIZE'	REDACT		REDACT	REDACT	REDACT	
BAKER, BRIANA	REDACT		REDACT	REDACT	REDACT	
BAKER, CHRISTAIN	REDACT		REDACT	REDACT	REDACT	
BAKER, CHRISTIAN	REDACT		REDACT	REDACT	REDACT	
BAKER, DONELSON, BEARMAN,						
CALDWELL & BERKOWITZ, PC	WELLS FARGO TOWER	420 20TH STREET NORTH	BIRMINGHAM	AL	35203	<u> </u>
BAKER, ERRIN	REDACT		REDACT	REDACT	REDACT	
BAKER, MARIO	REDACT		REDACT	REDACT	REDACT	
BAKER, YANCY	REDACT	REDACT	REDACT	REDACT	REDACT	
BALDWIN COUNTY	TAX OFFICE	121 N. WILKINSON ST., STE 112	MILLEDGEVILLE	GA	31061-3376	
BALDWIN, ARNITA	REDACT		REDACT	REDACT	REDACT	
BALLET GROUP INC.	3 EMPIRE BLVD.		SOUTH HACKENSACK	NJ	07606	
BALTIMORE CITY - CIRCUIT COURT	100 N. CALVERT STREET	ROOM 628	BALTIMORE	MD	21202	

Name	Address 1	Address 2	City	State	PostalCode	Country
			•			,
BALTIMORE CITY, DIR OF FINANCE	FIRE PREVENTION BUREAU	410 E. LEXINGTON STREET	BALTIMORE	MD	21202	
	OFFICE OF THE ZONING					
BALTIMORE HOUSING	ADMINISTRATION	417 E. FAYETTE STREET, ROOM 147	BALTIMORE	MD	21202	
BAMA FIRE PROTECTION, LLC	P.O. BOX 876		BIRMINGHAM	AL	35201	
BAMBOO JEANS	1407 BROADWAY, STE 2015		NEW YORK	NY	10018	
BANCORP SOUTH	One Mississippi Plaza		Tupelo	MS	38801	
BANK OF AMERICA (FL)	PO Box 15284		Wilmington	DE	19850	
BANK OF AMERICA (GA)	PO Box 15284		Wilmington	DE	19850	
BANK OF AMERICA (MO)	PO Box 15284		Wilmington	DE	19850	
BANK OF AMERICA (SC)	PO Box 15284		Wilmington	DE	19850	
BANK OF AMERICA (TX)	PO Box 15284		Wilmington	DE	19850	
BANK OF NORTH GEORGIA	3106 HWY 278 NW		COVINGTON	GA	30014	
BANKFINANCIAL	6415 W. 95th Street		Chicago	IL	60415-2600	
BANKS, LAWANDA	REDACT		REDACT	REDACT	REDACT	
BANKS, REGINALD	REDACT		REDACT	REDACT	REDACT	
BAPTIST MEMORIAL HOSPITAL	C/O DAVE EDWARDS	P.O. BOX 458	PARIS	TN	38242	
BARKSDALE, SHADAE	REDACT		REDACT	REDACT	REDACT	
BARNES LIMOUSINE CO.	99-10 24TH AVENUE		EAST ELMHURST	NY	11369	
BARNES, BRIANA	REDACT		REDACT	REDACT	REDACT	
BARNES, ERICA	REDACT		REDACT	REDACT	REDACT	
BARNETT, MELISSA	REDACT		REDACT	REDACT	REDACT	
BARNETT, ROSS	REDACT		REDACT	REDACT	REDACT	
BARNWELL, MELISSA	REDACT		REDACT	REDACT	REDACT	
BARON DISTRIBUTORS INC.	1428 ESSEX STREET		LOS ANGELES	CA	90021	
BARRETT, DIANN	REDACT		REDACT	REDACT	REDACT	
BARRETT, LEE	REDACT		REDACT	REDACT	REDACT	
BARRETT, MARY	REDACT		REDACT	REDACT	REDACT	
BARRETT, VICTOR	REDACT		REDACT	REDACT	REDACT	
BARRINO, PETERILL	REDACT		REDACT	REDACT	REDACT	
BARRY, CRIAG K.	REDACT		REDACT	REDACT	REDACT	
Barstein, Mark	3428 Oak Canyon Circle		Birmingham	AL	35243	
BASSO, OLIVIA	REDACT		REDACT	REDACT	REDACT	
BATES, VASSKYNEE	REDACT		REDACT	REDACT	REDACT	
BATSON, KAREN	REDACT	REDACT	REDACT	REDACT	REDACT	
BATTLES, BRENYELLE	REDACT		REDACT	REDACT	REDACT	
BATY, II, ROBERT	REDACT		REDACT	REDACT	REDACT	
BB&T	PO Box 200		Wilson	NC	27894-0200	

Page 13 of 176

Name	Address 1	Address 2	City	State	PostalCode	Country
BC WOOD INVESTMENT FUND I, LLC						
D/B/A IROQUOIS MANOR SHOPPING						
CENTER	1020 Industrial Road	Suite 40	Lexington	KY	40505	
BDB WESLEY CHAPEL	ATTN: TABITHA ROBINSON	7301 PARKWAY DRIVE	HANOVER	MD	21076	
BDB WESLEY CHAPEL, LLC	ATTN: DAWN BURKOWSKI	7301 Parkway Drive	Hanover	MD	21076	
BEAUTY 21 COSMETICS	2021 S ARCHIBALD AVE		ONTARIO	CA	91761	
BEAUTY GROUP	10 W 33RD		NEW YORK	NY	10001	
BEAVER PLUMBING CO.	P.O. BOX 470115		ST. LOUIS	MO	63147	
BEAVER, LAQUISHA	REDACT		REDACT	REDACT	REDACT	
BECKLER PLUMBING COMPANY	4119 EDITH AVENUE		CINCINNATI	ОН	45227	
BECTON, APRIL	REDACT		REDACT	REDACT	REDACT	
BECTON, DEMONT	REDACT		REDACT	REDACT	REDACT	
BEE'S KEYS COMPLETE SECURITY						
SERVICE	7711 INWOOD ROAD		DALLAS	TX	75209	
BEL AIR 8 MILE LLC	C/O MITCHELL ROSEN	25550 GRAND RIVER	REDFORD	MI	48240	
BEL AIR REALTY LLC	150 GREAT NECK ROAD	SUITE 304	GREAT NECK	NY	11021	
BEL-AIR 8 MILE, LLC	25550 Grand River Avenue		Redford	MI	48240	
BEL-AIR REALTY,LLC	150 GREAT NECK RD.		GREAT NECK	NY	11021	
BEL-AIR REALTY,LLC	150 GREAT NECK RD., Ste. 304		GREAT NECK	NY	11021	
BELCHER, KIMBERLY	REDACT		REDACT	REDACT	REDACT	
BELCHER, NATAYAH	REDACT		REDACT	REDACT	REDACT	
BELCHER, NATAYAH	REDACT		REDACT	REDACT	REDACT	
BELIN, NICOLE	REDACT		REDACT	REDACT	REDACT	
BELL, ALEXIS	REDACT	REDACT	REDACT	REDACT	REDACT	
BELL, SONYA	REDACT		REDACT	REDACT	REDACT	
BELL, TRINA	REDACT		REDACT	REDACT	REDACT	
BELLA D	525 7TH AVE		NEW YORK	NY	10018	
BELLE MEADE PARTNERS LLC	C/O CORPORATE REALTY INC.	1750 CLEARVIEW PKWY	MELARIE	LA	70001	
BELLE MEADE PARTNERS, LLC	C/O CORPORATE REALTY, INC	1750 CLEARVIEW PKWY	METAIRIE	LA	70001	
BELLE PROMINADE SHOPS I, LLC	240 BROOKSTONE CENTRE PKWY		COLUMBUS	GA	31904	
BELLERIVE, ALEXANDRIA	REDACT		REDACT	REDACT	REDACT	
BELLO, VLADIMILKA	REDACT		REDACT	REDACT	REDACT	
BELTON, CHARITY	REDACT		REDACT	REDACT	REDACT	
BELTWAY AUTO & PLATE GLASS, INC	3827 PENN BELT PLACE		FORESTVILLE	MD	20747	
Belz Investco GP	GATEWAY/Store No198	c/o Belz Enterprises	Memphis	TN	38103	
BELZ INVESTCO GP	ATTN: A/R	P O Box 3661	Memphis	TN	38173-0661	

Page 14 of 176

Name	Address 1	Address 2	City	State	PostalCode	Country
BELZ INVESTCO GROUP	ATTN: A/R	P.O. BOX 3661	MEMPHIS	TN	38173-0661	
BEN ELIAS INDUSTRIES	1400 BROADWAY		NEW YORK	NY	10018	
BENA J	1458 SAN PEDRO ST		LOS ANGELES	CA	90079	
BENEFIELD, CHRISTINA	REDACT	REDACT	REDACT	REDACT	REDACT	
BENJAMIN, PRISCILLA	REDACT	REDACT	REDACT	REDACT	REDACT	
BENNETT, BESSIE	REDACT		REDACT	REDACT	REDACT	
BENNETT, DARYL	REDACT		REDACT	REDACT	REDACT	
BENNETT, KIMBERLY	REDACT		REDACT	REDACT	REDACT	
BENNETT, MILDRED	REDACT		REDACT	REDACT	REDACT	
BENNETT, SANISHA	REDACT		REDACT	REDACT	REDACT	
BENNETT'S FIRE PROTECTION	PO BOX 335		LAKE PARK	GA	31636	
BENNS, MERCEDES	REDACT		REDACT	REDACT	REDACT	
BENNY SWAN ELECTRICAL SERVICE	PO BOX 5345		TYLER	TX	75712	
BENSON, ASHLEY	REDACT		REDACT	REDACT	REDACT	
BENTON & CENTENO, LLP	2019 THIRD AVENUE N		BIRMINGHAM	AL	35203	
BENTON CHARTER TOWNSHIP	TREASURER	1725 TERRITORIAL RD STE B	BENTON HARBOR	MI	49022	
BENTON, ANGELA	REDACT		REDACT	REDACT	REDACT	
BENTON, SAMANTHA	REDACT	REDACT	REDACT	REDACT	REDACT	
BERG, ROBERT	REDACT		REDACT	REDACT	REDACT	
BERKOWITZ POLLACK & BRANT	CERTIFIED PUBLIC ACCOUNTANTS	200 S. BISCAYNE BLVD.,6TH FL	MIAMI	FL	33131-5351	
BERKSHIRE LIFE INSURANCE CO	PO BOX 644776	200 S. BISCATNE BEVD.,OTTTE	PITTSBURG	PA	15264-4776	1
BERNADETTE EDWARDS	1710 2ND AVENUE N.	APT.#415	BIRMINGHAM	AL	35203	
BERRY, DAWN	REDACT	A11.#415	REDACT	REDACT	REDACT	
BERRY'S LOCK SERVICE, LLC	PO BOX 18694		RAYTOWN	MO	64133	
BESSEMER UTILITIES	1600 1ST AVE NORTH		BESSEMER	AL	35021	
BEST SECURITY INDUSTRIES INC	755 NW 17TH AVENUE	SUITE 101	DELRAY BEACH	FL	33445	1
BESTFIT INTERNATIONAL	900 CANADA COURT		CITY OF INDUSTRY	CA	91748	1
BETANCOURT, ANGELICA	REDACT		REDACT	REDACT	REDACT	1
BETHEA, BOBBY	REDACT		REDACT	REDACT	REDACT	
BETO, JEROME	REDACT		REDACT	REDACT	REDACT	
BETSIL, KAREE	REDACT		REDACT	REDACT	REDACT	
BEVERIDGE, TAMARA	REDACT		REDACT	REDACT	REDACT	
BEXAR COUNTY	TAX ASSESSOR-COLLECTOR	P.O. BOX 839950	SAN ANTONIO	TX	78283-3950	
BG NORTH CHARLESTON LLC	DEPT: 104797-20609-6968	P.O. BOX 931563	CLEVELAND	ОН	44193	
	c/o Developers Diversified Realty					
BG NORTH CHARLESTON, LLC	Corporation	3300 Enterprise Pkwy	Beachwood	ОН	44122	

Page 15 of 176

Name	Address 1	Address 2	City	State	PostalCode	Country
BG THRUWAY, LLC	C/O DDR CORP.	ATTN:EXECUTIVE VP LEASING	BEACHWOOD	ОН	44122	
BG THRUWAY, LLC	c/o DDR Corp.	3300 Enterprise Parkway	Beachwood	ОН	44122	
BGE	110 West Fayette St		BALTIMORE	MD	21201	
BGE	P.O. BOX 13070		PHILADELPHIA	PA	19101-3070	
BH WORLDWIDE LLC	302 5TH AVE, 14TH FLR		NEW YORK	NY	10001	
Bhavana Shah	2110 N.W. 95th Avenue		Miami	FL	33172	
Bhavana Shah Revocable Trust	2110 N.W. 95th Avenue		Miami	FL	33172	
BIAXIDENT	1410 S MAIN ST		LOS ANGELES	CA	90015	
Bibb County	Attn: Consumer Protection Agency	Judd Drake - County Attorney	Macon	GA	31201	
BIBB/MACON TAX COMMISSIONER	P.O. BOX 4724		MACON	GA	31208-4724	
BIJOU INTERNATIONAL	42 WEST 39TH ST		NEW YORK	NY	10018	
BIPCO, INC (BHAM INDUSTRIAL						
PRODUCTS)	9064 HELENA ROAD		PELHAM	AL	35124	
BIRDSONG, SHONARAH	REDACT		REDACT	REDACT	REDACT	
BIRMINGHAM DOOR TO DOOR						
SHUTTLE	281 SEQUOYAH ROAD		PELL CITY	AL	35128	
BIRMINGHAM RUBBER STAMP	P.O. BOX 11083		BIRMINGHAM	AL	35202	
BIRMINGHAM TROPHY	4807 5TH AVENUE SOUTH		BIRMINGHAM	AL	35222-2222	
BISHOP, BARRY	REDACT		REDACT	REDACT	REDACT	
BISHOP, DEBORAH	REDACT		REDACT	REDACT	REDACT	
BLACK, CORETTA	REDACT		REDACT	REDACT	REDACT	
BLACK, LYNDSAY	REDACT		REDACT	REDACT	REDACT	
BLACK, VINCENZA	REDACT	REDACT	REDACT	REDACT	REDACT	
BLACK, VINCENZA MIA	REDACT		REDACT	REDACT	REDACT	
BLACKMON, ASHLEY	REDACT	REDACT	REDACT	REDACT	REDACT	
BLACKWELL, MICHELLE	REDACT		REDACT	REDACT	REDACT	
BLACKWELL, SHAVANNA	REDACT	REDACT	REDACT	REDACT	REDACT	
BLAIR SERVICES, INC.	325 HICKORY AVENUE		HARAHAN	LA	70123	
BLAKE, SOPHIA	REDACT		REDACT	REDACT	REDACT	
BLANK ROME LLP	ONE LOGAN SQUARE		PHILADELPHIA	PA	19103-6993	
BLASSINGAME, WAKEENA	REDACT	REDACT	REDACT	REDACT	REDACT	
BLOCK AND COMPANY	DEPARTMENT #10293	PO BOX 87618	CHICAGO	IL	60680-0618	
BLOMBERG, DIANA	REDACT	REDACT	REDACT	REDACT	REDACT	
BLOSSOM	18120 ROWLAND ST		CITY OF INDUSTRY	CA	91748	
BLUE CROSS BLUE SHIELD (PYMT						
PROCESSING)	ATTN: PAYMENT PROCESSING	PO BOX 360387	BIRMINGHAM	AL	35236-0387	

Page 16 of 176

Name	Address 1	Address 2	City	State	PostalCode	Country
BLUE FAITH	246 WEST 38TH STREET	SUITE #703	NEW YORK	NY	10018	
BLUE GALAXY	2369 E. 51ST STREET		VERNON	CA	90058	
BLUE STAR CLOTHING CO.	1370 BROADWAY		NEW YORK	NY	10018	
BLUNT, BRIA	REDACT		REDACT	REDACT	REDACT	
BLUNT, IESHA	REDACT	REDACT	REDACT	REDACT	REDACT	
BLUNT, TERRELL	REDACT		REDACT	REDACT	REDACT	
BLYTHER, JUANITA	REDACT	REDACT	REDACT	REDACT	REDACT	
BMGM CO/HONEYCOMB KIDZ	1385 BROADWAY		NEW YORK	NY	10018	
BOARD OF WATER COMMISSIONERS	735 RANDOLPH		DETROIT	MI	48226	
BOARD OF WATER COMMISSIONERS	CITY OF DETROIT	PO BOX 32711	DETROIT	МІ	48232-0711	
BOARD OF WATER COMMISSIONERS	PO BOX 32711		DETROIT	МІ	48232-0711	
BOB BATY	1623 WILTON STREET		CHARLESTON	SC	29412	
BOLDINAP, GERALD	REDACT		REDACT	REDACT	REDACT	
BOLIVAR COUNTY	TAX COLLECTOR	PO BOX 248	CLEVELAND	MS	38732	
BOLLING, JULIE	REDACT		REDACT	REDACT	REDACT	
BOLTON, ZAKIEA	REDACT		REDACT	REDACT	REDACT	
BOMBAY COURIER INT AIR EXP	250 WEST 40TH STREET	8TH FLOOR	NEW YORK	NY	10018-1511	
	CONSUMER SQUARE WEST/Store					
Bon Aviv Investments	No563	Attn: Amit Barnoon	Englewood Cliffs	NJ	07632-0000	
BONDE, DANALLE	REDACT		REDACT	REDACT	REDACT	
BONHOMME, SIDNEY	REDACT		REDACT	REDACT	REDACT	
BONILLA, ASHLEY	REDACT		REDACT	REDACT	REDACT	
BONILLA, LUZ	REDACT		REDACT	REDACT	REDACT	
BONNER, JENNIFER	REDACT		REDACT	REDACT	REDACT	
BOOG, CASSANDRA	REDACT		REDACT	REDACT	REDACT	
BOOKER, JASMINE	REDACT		REDACT	REDACT	REDACT	
BOOM-BOOM	1526 E. WASHINGTON BLVD		LOS ANGELES	CA	90021	
Borough of Audubon	Attn: Dave Taraschi	606 West Nicholson Road	Audubon	NJ	08106	
BOROUGH OF AUDUBON	OFFICE OF THE FIRE MARSHALL	606 W. NICHOLSON ROAD	AUDUBON	NJ	08106	
BORUM, ANDREA	REDACT		REDACT	REDACT	REDACT	
BOSSIER CITY	SALES TAX DEPT.	P.O.BOX 71313	BOSSIER CITY	LA	71171-1313	
Bossier City-Parish	Sales Tax Division	PO Box 71313	Bossier City	LA	71171-1313	
BOSSIER PARISH/PROP TAX	SHERIFF/TAX COLLECTOR	P.O. BOX 20905	SHREVEPORT	LA	71120-0905	
BOST, LATRICE	REDACT		REDACT	REDACT	REDACT	
BOTHWELL, CAROL	REDACT		REDACT	REDACT	REDACT	

Page 17 of 176

Name	Address 1	Address 2	City	State	PostalCode	Country
BOULEVARD	1375 E. 6TH ST.	UNIT 5	LOS ANGELES	CA	90021	
BOURLAND, WALL & WENZEL P.C.	301 COMMERCE STREET	SUITE 1500	FORT WORTH	TX	76102	
BOWE, YUSHANDA	REDACT		REDACT	REDACT	REDACT	
BOWIE, STEPHEN	REDACT		REDACT	REDACT	REDACT	
BOWSER, CRYSTAL	REDACT		REDACT	REDACT	REDACT	
BOX RESELLERS (DUFFY BOX)	800 WHARTON DRIVE		ATLANTA	GA	30336	
BOYD, CHELSEA	REDACT	REDACT	REDACT	REDACT	REDACT	
BOYD, MARKO	REDACT		REDACT	REDACT	REDACT	
BOYD, TAMMY	REDACT		REDACT	REDACT	REDACT	
BOYKIN, KANDETRIA	REDACT		REDACT	REDACT	REDACT	
BOZIER, TOSHIBA	REDACT		REDACT	REDACT	REDACT	
BRADFORD, JASMINA	REDACT		REDACT	REDACT	REDACT	
BRADLEY ARANT BOULT CUMMINGS	PO BOX 830709		BIRMINGHAM	AL	35283-0709	
BRADLEY PLUMBING & HEATING,						
INC	PO BOX 210007		MONTGOMERY	AL	36121-0007	
BRADLEY, CAROL	REDACT		REDACT	REDACT	REDACT	
BRADLEY, SHAYNNA	REDACT		REDACT	REDACT	REDACT	
BRADLEY, TONY	REDACT		REDACT	REDACT	REDACT	
BRAESWOOD DEVELOPMENT LLC.	4545 BISSONNET STREET	SUITE 100	BELLAIRE	TX	77401	
Branch Banking	P.O.Box 890011		Charlotte	NC	28289-0011	
BRANCH BANKING AND TRUST						
COMPANY	C/O FICKLING & CO.	ATTN: D. LAURENS	MACON	GA	31201	
BRANDON, GLORIA	REDACT		REDACT	REDACT	REDACT	
BRANTLEY, RACHEL	REDACT		REDACT	REDACT	REDACT	
BRAXTON BRITT PLUMBING SER.INC.	P.O. BOX 7018		WILSON	NC	27895	
BRAXTON, AL' ZHANA	REDACT		REDACT	REDACT	REDACT	
BRAXTON, AL'ZHANA	REDACT		REDACT	REDACT	REDACT	
BRAZIER, KENNETH	REDACT	REDACT	REDACT	REDACT	REDACT	
BRAZIL, JAKARIOUS	REDACT		REDACT	REDACT	REDACT	
BRC RUSSELLVILLE, LLC	C/O BLUE RIDGE CAPITAL, LLC	ONE BUCKHEAD PLZ	ATLANTA	GA	30327	
BRC RUSSELLVILLE, LLC	c/o Blue Ridge Capital, LLC	One Buckhead Plaza	Atlanta	GA	30305	
BRE DDR BR OAK WHITE VA, LLC	c/o DDR Corp.	3300 Enterprise Parkway	Beachwood	ОН	44122	
BRE DDR BR WHITE OAK, LLC	DEPT 342570 25325 57898	PO BOX 535769	ATLANTA	GA	30353-5769	
BRE DDR BR WHITE OAK, LLC	DEPT 353732-25325-00060037	P.O. BOX 535769	ATLANTA	GA	30353-5769	

Name	Address 1	Address 2	City	State	PostalCode	Country
BRE RETAIL RESIDUAL OWNER 6, LLC	C/O BRIXMOR PROPERTY GROUP	420 LEXINGTON AVE	NEW YORK	NY	10170	
BRE Retail Residual Owner 6, LLC	30TH STREET S/C/Store No254	c/o Brixmor Property Group	New York	NY	10170	
BRE Retail Residual Owner 6, LLC	30TH STREET S/C/Store No254	c/o Brixmor Property Group	Cronshohocken	PA	19428	
BRE RETAIL RESIDUAL OWNER 6, LLC	C/O BRIXMOR PROPERTY GROUP	P O Box 645349	Cincinatti	ОН	45264-5349	
	C/O Bibby Financial Services (CA)					
BREAKAWAY FASHIONS	Inc.	File#51042	Los Angeles	CA	90074-1042	
BRENDA LITKE	114 NORTH AVE		WEST SENECA	NY	14224	
BRENDA WALKERLY	3496 CEDARBROOK		UNIVERSITY HEIGHTS	ОН	44118	
BREWER, TARA	REDACT		REDACT	REDACT	REDACT	
BRICKTOWN SQUARE, LLC	C/O BONNIE MGMT	8430 W. BRYN MAWR AVENUE	CHICAGO	IL	60631	
BRICKTOWN SQUARE, LLC	c/o Bonnie Management	8430 W. Bryn Mawr	Chicago	IL	60631	
BRIDGES, VALERIE	REDACT	REDACT	REDACT	REDACT	REDACT	
BRIGGS, MARK	REDACT		REDACT	REDACT	REDACT	
	3420 RICHARD ARRINGTON JR BLVI					
BRIGHT FUTURE ELECTRIC, LLC	NORTH		BIRMINGHAM	AL	35234	
BRIGHT PAR 3 ASSOCIATES, LP	C/O FLETCHER BRIGHT CO.	537 MARKET ST, SUITE 400	CHATTANOOGA	TN	37402	
BRIGHT WAY INTERNATIONAL	18333 GALE AVE		CITY OF INDUSTRY	CA	91748	
BRIGHT-MEYERS ALBANY ASSOC., LP	C/O FLETCHER BRIGHT CO.	537 MARKET ST., STE 400	CHATTANOOGA	TN	37402	
BRIGHT-MEYERS ALBANY						
ASSOCIATES, LP	C/O FLETCHER BRIGHT CO.	537 Market Street	Chattanooga	TN	37402	
BRIGHT-MEYERS KENNASAW						
ASSOCIATES LP	C/O FLETCHER BRIGHT CO.	537 MARKET STREET, SUITE 400	CHATTANOOGA	TN	37402	
BRIGHT-MEYERS KENNESAW						
ASSOCIATES, LP	C/O FLETCHER BRIGHT CO.	537 Market Street	CHATTANOOGA	TN	37402	
BRIMMER, AMBER	REDACT		REDACT	REDACT	REDACT	
BRISBON, ROSIE	REDACT		REDACT	REDACT	REDACT	
BRISSETT, MARVA	REDACT	REDACT	REDACT	REDACT	REDACT	
BRISTOW, CHIQUITA	REDACT		REDACT	REDACT	REDACT	
BRIXMOR GA APOLLO II SUB, LLC	LEASE ID 5218009	PO BOX 74623	CLEVELAND	ОН	44194	
BRIXMOR GA PARKWAY PLAZA LP	C/O BRIXMOR PROPERTY GROUP	PO BOX 74644	CLEVELAND	ОН	44194	
BRIXMOR GA PARKWAY PLAZA, LP	c/o Brixmor Property Group	420 Lexington Avenue	New York	NY	10170	
BRIXMOR GA ROUNDTREE PLACE LLC	c/o Brixmor Property Group	420 Lexington Avenue	New York	NY	10170	

Page 19 of 176

Name	Address 1	Address 2	City	State	PostalCode	Country
BRIXMOR GA ROUNDTREE PLACE,		ATTN: OFFICE OF GENERAL				
LLC.	C/O BRIXMOR PROPERTY GROUP	COUNSEL	NEW YORK	NY	10170	
BRIXMOR HOLDIINGS 1 SPE, LLC	c/o Brixmor Property Group	420 Lexington Avenue	New York	NY	10170	
BRIXMOR HOLDINGS 1 SPE, LLC	C/O BRIXMOR PROPERTY GROUP	PO BOX 713439	CINCINNATI	ОН	45271-3439	
	C/O BRE RETAIL RESIDUAL OWNER I					
BRIXMOR HOLDINGS 8 SPE, LLC	LLC	PO BOX 713538	CINCINNATI	ОН	45271-3538	
Brixmor Property Group	C/O Holland & Knight LLP	Attn: Joaquin J. Alemany, Esq.	Miami	FL	33131	
Brixmor Property Group	C/O Holland & Knight LLP	Attn: Eric B. Funt, Esq.	Miami	FL	33131	
BRIXMOR SPE 3 LLC	C/O BRIXMOR PROPERTY GROUP	P.O. BOX 713547	CINCINNATI	ОН	45271-3547	
BRIXMOR SPE 4 LLC	c/o Brixmor Property Group	420 Lexington Avenue	New York	NY	10170	
BRIXMOR SPE 4, LLC.	C/O BRIXMOR PROPERTY GROUP	P.O. BOX 30906	NEW YORK	NY	10087-0906	
BRIXMOR SPE 5 LLC	c/o Brixmor Property Group	P O Box 713538	Cincinatti	ОН	45271-3538	
DDIVATOR CDE E 11.0	C/O DDIVATOR PRODERTY CROUD	DO DOV 743530	CINICININIATI	011	45274 2520	
BRIXMOR SPE 5, LLC BRIXMOR STOCKBRIDGE VILLAGE.	C/O BRIXMOR PROPERTY GROUP	PO BOX 713538	CINCINNATI	ОН	45271-3538	
	C/O CENTRO NP HOLDINGS 1 SPE,	D O DOV 533337	CHARLOTTE	NG	20200 2227	
LLC	LLC	P.O. BOX 533337	CHARLOTTE	NC	28290-3337	
BRIXMOR/IA SPENCER SQUARE, LLC	C/O BRIXMORE PROPERTY GROUP	PO BOX 713439	CINCINNATI	он	45271-3439	
BRIAINION, IA SPENCER SQUARE, LLC	C/O BRIXIVIORE PROPERTY GROUP	FO BOX 713439	CINCINNATI	Оп	43271-3439	
BRIXMOR/IA SPENCER SQUARE, LLC	c/o Brixmor Property Group	420 Lexington Avenue	New York	NY	10170	
BRIXMOR/IA TINLEY PARK PLAZA,	cy o Brixmor Property Group	420 Eckington / Wende	IVEW TOTK	141	10170	
LLC	c/o Brixmor Property Group	420 Lexington Avenue	New York	NY	10170	
BRIXMORE/IA TINLEY PARK PLAZA.	, o =	g				
LLC	C/O BRIXMOR PROPERTY GRP	ATTN:OFF GENERAL COUNSEL	NEW YORK	NY	10170	
BRN JACKSON LLC successor in						
Interest to Ellis Avenue Properties,	C/O CPI INVESTMENT ADVISORS,					
LLC	LLC	P O Box 12486	Jackson	MS	39236	
	C/O CPI INVESTMENT ADVISORS,					
BRN JACKSON, LLC	LLC	P.O. BOX 12486	JACKSON	MS	39236	
BROADWAY CENTER PARTNERSHIP	P.O. BOX 22013		KNOXVILLE	TN	37933	
BROADWAY CENTER PARTNERSHIP	P O BOX 22013		KNOXVILLE	TN	37933	
BROADWAY, TY'QUEESHA	REDACT		REDACT	REDACT	REDACT	
BROCKS, LIDEYA	REDACT		REDACT	REDACT	REDACT	

Name	Address 1	Address 2	City	State	PostalCode	Country
BRONSON	1136 S FLOWER		LOS ANGELES	CA	90015	
BROOKS, AARON	REDACT		REDACT	REDACT	REDACT	
BROOKS, DE'LACEANA	REDACT		REDACT	REDACT	REDACT	
BROOKS, DE'LANCEANA	REDACT		REDACT	REDACT	REDACT	
BROOKS, DEVERAN	REDACT		REDACT	REDACT	REDACT	
BROOKS, MARVA	REDACT		REDACT	REDACT	REDACT	
BROOKS, MARVA	REDACT		REDACT	REDACT	REDACT	
BROOKS, SHANEE	REDACT		REDACT	REDACT	REDACT	
BROOKS, SHATWYLA	REDACT		REDACT	REDACT	REDACT	
BROWARD COUNTY PUBLIC WORKS						
DEPT	115 S. ANDREWS AVE., SUITE A600		FORT LAUDERDALE	FL	33301	
BROWARD COUNTY PUBLIC WORKS						
DEPT	PO BOX 669300		POMPANO BEACH	FL	33066-9300	
BROWARD COUNTY REVENUE						
COLLECTOR	115 S. ANDREWS AVE. RM A-100		FT. LAUDERDALE	FL	33301	
BROWARD COUNTY WWS (USE						
EUW170)	PO BOX 669300		POMPANO BEACH	FL	33066-9300	
BROWN, ALESIA	REDACT	REDACT	REDACT	REDACT	REDACT	
BROWN, ALEXIS	REDACT	REDACT	REDACT	REDACT	REDACT	
BROWN, ANGELA	REDACT		REDACT	REDACT	REDACT	
BROWN, ARIELL	REDACT		REDACT	REDACT	REDACT	
BROWN, BRITTANY	REDACT		REDACT	REDACT	REDACT	
BROWN, CLANESIA	REDACT		REDACT	REDACT	REDACT	
BROWN, DAILYA	REDACT		REDACT	REDACT	REDACT	
BROWN, DANIELLE	REDACT		REDACT	REDACT	REDACT	
BROWN, DASHAE	REDACT		REDACT	REDACT	REDACT	
BROWN, JACKIE	REDACT		REDACT	REDACT	REDACT	
BROWN, JESSICA	REDACT		REDACT	REDACT	REDACT	
BROWN, JR., KENNETH	REDACT		REDACT	REDACT	REDACT	
BROWN, KENDASIA	REDACT		REDACT	REDACT	REDACT	
BROWN, KIMBERLY	REDACT		REDACT	REDACT	REDACT	
BROWN, KIMBERLY	REDACT		REDACT	REDACT	REDACT	
BROWN, LACHANNAN	REDACT	REDACT	REDACT	REDACT	REDACT	
BROWN, MARKEEBADA	REDACT		REDACT	REDACT	REDACT	
BROWN, MARKELA	REDACT		REDACT	REDACT	REDACT	
BROWN, ROBIE	REDACT		REDACT	REDACT	REDACT	
BROWN, ROMEKIA	REDACT		REDACT	REDACT	REDACT	
BROWN, SHAYLA	REDACT		REDACT	REDACT	REDACT	

Page 21 of 176

Name	Address 1	Address 2	City	State	PostalCode	Country
BROWN, TIMOTHY	REDACT		REDACT	REDACT	REDACT	
BROWN-ANDERSON, KYIRHEIK	REDACT	REDACT	REDACT	REDACT	REDACT	
BROWN-SMITH, MELISSA	REDACT		REDACT	REDACT	REDACT	
Bruce Gendelman Co., Inc.	Attn: Carolyn Rabbitt	500 W Brown Deer Road	Milwaukee	WI	53217	
BRUNO, JAMES	REDACT		REDACT	REDACT	REDACT	
BRYANT, CAMERON	REDACT		REDACT	REDACT	REDACT	
BRYANT, KAREN	REDACT		REDACT	REDACT	REDACT	
BRYANT, MARCIA	REDACT	REDACT	REDACT	REDACT	REDACT	
BRYANT, SADE'	REDACT		REDACT	REDACT	REDACT	
BRYANT, SHANIQUEA	REDACT		REDACT	REDACT	REDACT	
BRYD, ROBERT	REDACT		REDACT	REDACT	REDACT	
BRYM, NAIMATOU	REDACT		REDACT	REDACT	REDACT	
BSG 95TH & JEFFERY, LLC	c/o Terraco, Inc.	3201 Old Glenview Road	Wilmette	IL	60091	
BSG 95TH & JEFFREY, LLC	C/O TERRACO INC	3201 OLD GLENVIEW RD	WILMETTE	IL	60091	
BUCHANA, AUBLES	REDACT		REDACT	REDACT	REDACT	
BUCHANAN ACQUISITIONS, L.P.	c/o Sitt Asset Management, LLC	One Penn Plaza	New York	NY	10119	
BUCHANAN ACQUISITIONS, LP.	C/O SITT ASSET MGMT, LLC	ONE PENN PLAZA SUITE 3430	NEW YORK	NY	10017	
BUCHANAN, CUNARD	REDACT		REDACT	REDACT	REDACT	
BUCHANAN, PAMELA	REDACT		REDACT	REDACT	REDACT	
BUCHANAN, WILDON	REDACT		REDACT	REDACT	REDACT	
BUCKLEY, BRIANA	REDACT		REDACT	REDACT	REDACT	
BUCKLEY, NAKIA	REDACT		REDACT	REDACT	REDACT	
BUCKNER JUBILEE INVESTMENTS						
GROUP, LLC.	P.O. BOX 205069		DALLAS	TX	75320-5069	
BUCKNER SHOPPING CENTER, LP	3100 Araphoe Avenue	Suite 501	Boulder	СО	80303	
Bucks County	Attn: David Heckler	Bucks County Justice Center	Doylestown	PA	18901	
BUFFALO ROCK COMPANY	PO BOX 2247		BIRMINGHAM	AL	35201	
BUFORD CLAIRMONT CO. LTD	3715 NORTHSIDE PKWY NW	SUITE 4-210	ATLANTA	GA	30327	
BUFORD, ALFONZO	REDACT		REDACT	REDACT	REDACT	
BUFORD, CAMERON	REDACT		REDACT	REDACT	REDACT	
BULDOWSKI, REBECCA	REDACT		REDACT	REDACT	REDACT	
BULLOCK, BRYANT	REDACT		REDACT	REDACT	REDACT	
BUNCH, ALTAVIA	REDACT	REDACT	REDACT	REDACT	REDACT	
BUNN, JILL	REDACT		REDACT	REDACT	REDACT	
BURGESS, TERRENCE	REDACT		REDACT	REDACT	REDACT	
BURGESS, TIFFANY	REDACT		REDACT	REDACT	REDACT	
BURGOS, SHIRLENE	REDACT	REDACT	REDACT	REDACT	REDACT	

Name	Address 1	Address 2	City	State	PostalCode	Country
BURKS, AUDORIA	REDACT		REDACT	REDACT	REDACT	
BURL REID	PO BOX 191		BATESVILLE	MS	38606	
BURLEY CHURCH ET AL	C/O FINMARC MANAGEMENT, INC.	7200 WISCONSIN AVE.	BETHESDA	MD	20814	
BURNETT, HATTIE	REDACT		REDACT	REDACT	REDACT	
BURNETT, IRMA	REDACT		REDACT	REDACT	REDACT	
BURNS, MAHOGANY	REDACT		REDACT	REDACT	REDACT	
BURNS, MARTHA	REDACT		REDACT	REDACT	REDACT	
BURNS, SHELBI	REDACT		REDACT	REDACT	REDACT	
BURTON, CRYSTAL	REDACT		REDACT	REDACT	REDACT	
BURTON, FELICE	REDACT		REDACT	REDACT	REDACT	
BUSH ENTERPRISE	1312 GOLDWIRE STREET SW		BIRMINGHAM	AL	35211	
BUSH, HILTON	REDACT	REDACT	REDACT	REDACT	REDACT	
BUSH, TINA	REDACT	REDACT	REDACT	REDACT	REDACT	
BUSH, WAYNE	REDACT		REDACT	REDACT	REDACT	
BUSINESSCARD CREDIT CENTER	DONNA YOUNGBLOOD	P.O.BOX 830667	BIRMINGHAM	AL	35283	
BUSINESSCARD CREDIT CENTER	JACKIE HARRISON	P.O.BOX 830667	BIRMINGHAM	AL	35283	
BUSSEY, QUANITA	REDACT		REDACT	REDACT	REDACT	
BUSTLETON PARTNERS	C/O KIMCO REALTY CORP.	3333 NEW HYDE PARK ROAD	NEW HYDE PARK	NY	11042	
BUSTLETON PARTNERS	C/O KIMCO REALTY CORP	3333 New Hyde Park Road	New Hyde Park	NY	11042	
BUTLER, DELON	REDACT		REDACT	REDACT	REDACT	
BUTLER, JACQUELINE	REDACT		REDACT	REDACT	REDACT	
BUTLER-HODGE, VELINDA	REDACT		REDACT	REDACT	REDACT	
BUZZ JEANS USA	1407 BROADWAY, STE 1901		NEW YORK	NY	10018	
BYANT, AISHA	REDACT		REDACT	REDACT	REDACT	
BYNUM, ANITRA	REDACT		REDACT	REDACT	REDACT	
BYRD, ALICIA	REDACT	REDACT	REDACT	REDACT	REDACT	
BYRD, CONSTANCE	REDACT		REDACT	REDACT	REDACT	
BYRD'S CLEANING	513 EXPLORER CIRCLE		DOTHAN	AL	36301	
C. DAVID COTTINGHAM	STANDING CHAPTER 13 TRUSTEE	PO DRAWER 020588	TUSCALOOSA	AL	35402	
CA NEW PLAN FIXED RATE						
PARTNERSHIP LP	C/O CENTRO NP, LLC REIT 19 CNP	PO BOX 841530	DALLAS	TX	75284-1530	
CA NEW PLAN FIXED RATE						
PARTNERSHIP, LP	c/o Brixmor Property Group	420 Lexington Avenue	New York	NY	10170	
CA NEW PLAN FIXED RATE						
PARTNERSHIP, LP	c/o Centro Properties Group	420 Lexington Avenue	New York	NY	10170	
CABALLERO, LUZ	REDACT		REDACT	REDACT	REDACT	
CABRERA, VANESA	REDACT		REDACT	REDACT	REDACT	

Page 23 of 176

Name	Address 1	Address 2	City	State	PostalCode	Country
CACIA	1407 BROADWAY, 14TH FLR		NEW YORK	NY	10018	
CADDO PARISH	TAX COLLECTOR	PO BOX 20905	SHREVEPORT	LA	71120-0905	
CADDO- SHREVEPORT	SALES & USE TAX	PO BOX 104	SHREVEPORT	LA	71161	
Caddo-Shreveport Sales and Use Tax						
Commission	PO Box 104		Shreveport	LA	71161	
CAGE, LENOR	REDACT		REDACT	REDACT	REDACT	
CAGE, LEONA	REDACT	REDACT	REDACT	REDACT	REDACT	
Cahokia Village	Attn: Drake Jackson	103 Main Street	Cahokia	IL	62206	
CAINE, DENITA	REDACT	REDACT	REDACT	REDACT	REDACT	
CALA PRODUCTS	3121 S MAIN ST		LOS ANGELES	CA	90007	
Calcasieu Parish Sales and Use Tax						
Dept	PO Box 2050		Lake Charles	LA	70602-2050	
CALDWELL, KRISTINA	REDACT		REDACT	REDACT	REDACT	
CALDWELL, TIFFANY	REDACT		REDACT	REDACT	REDACT	
CALHOUN COUNTY	1702 NOBLE STREET	SUITE 107	ANNISTON	AL	36201	
CALHOUN, ANGIE	REDACT		REDACT	REDACT	REDACT	
CALIFORNIA HUB	8230 INDUSTRY AVE		PICO RIVERA	CA	90660	
CALIFORNIA OFF PRICE	796 E 14TH PLACE		LOS ANGELES	CA	90021	
CALLADONATO, REBECCA	REDACT		REDACT	REDACT	REDACT	
CALLAHAN, LANISE	REDACT		REDACT	REDACT	REDACT	
CALVILLO, JAMIE	REDACT		REDACT	REDACT	REDACT	
CAM ELECTRICAL SERVICES, INC.	C/O CHRIS MILLER	3412 CORNWALL ROAD	BALTIMORE	MD	21222	
CAMPBELL, ANDREA	REDACT		REDACT	REDACT	REDACT	
CAMPBELL, JERITA	REDACT		REDACT	REDACT	REDACT	
CAMPBELL, MALIKA	REDACT		REDACT	REDACT	REDACT	
CAMPBELL, SHALENA	REDACT		REDACT	REDACT	REDACT	
CANADA, RONNIE	REDACT		REDACT	REDACT	REDACT	
CANDACE JOHNSON	1700 ARTHUR STREET		GARY	IN	46404	
CANE RUN MALL, LLC	C/O G J REALTY	49 WEST 37TH ST	NEW YORK	NY	10018-6257	
CANE RUN MALL, LLC	c/o Hogan Development Co.	420 West Liberty Street	Louisville	KY	40202	
CANNON, DEBRENNIS	REDACT		REDACT	REDACT	REDACT	
CANTATA	726 E. 12TH STREET #102		LOS ANGELES	CA	90021	
CANTON CENTRE DEVELOPMENT,						
LLC	370 SEVENTH AVE.	SUITE 1600	NEW YORK	NY	10001-3903	
CANTON CENTRE DEVELOPMENT,						
LLC	370 Seventh Avenue	Suite 1600	New York	NY	10001-3903	
CANTON CITY UTILITIES	626 30TH STREET NW		CANTON	ОН	44709	
CAPITAL FURNITURE CO, INC.	ATTY J BARBOUR	PO BOX 199	RIDGELAND	MS	39158	

Page 24 of 176

Name	Address 1	Address 2	City	State	PostalCode	Country
CAPITAL ONE	PO Box 4199		Houston	TX	77210-4199	
CAPITAL ONE BANK (USA) NA						
(GARNISHMENT)	C/O WEBER & OLCESE, PLC	3250 W. BIG BEAVER ROAD	TROY	MI	48084	
CAPITAL ONE BANK (USA), NA	PO BOX 60599		CITY OF INDUSTRY	CA	91716-0599	
CAPITAL PLAZA ASSOCIATES, LLC	3038 Woodley Road		Montgomery	AL	36146	
CAPITOL PLAZA ASSOCIATES, LLC	678 REISTERSTOWN RD		PIKESVILLE	MD	21208	
		c/o Moore Property Manangement				
Capitol Plaza Associates, LLC	CAPITOL PLAZA/Store No133	LLC	Montgomery	AL	36102	
CAPPADONNA ELECTRIAL						
CONTRACTORS, INC.	12755 COGBURN		SAN ANTONIO	TX	78249	
CAPREF SMYRNA LLC	PO BOX 74696		CLEVELAND	ОН	44194-4696	
CAPREF SMYRNA LLC	8343 Douglas Avenue	Suite 300	Dallas	TX	75225	
CAR MAX	2696 JOHN HAWKINS PARKWAY		HOOVER	AL	35244	
CARABALLO, ALYSHA	REDACT		REDACT	REDACT	REDACT	
CARABALLO, CATHERINE	REDACT		REDACT	REDACT	REDACT	
CARABALLO, CATHERINE	REDACT		REDACT	REDACT	REDACT	
CARDINAL PARTNERS LLC	C/O Daniels Realty Group, Inc.	8710 Manchester Road	Brentwood	MO	63144	
CARGO, ERICA	REDACT	REDACT	REDACT	REDACT	REDACT	
CARL REED	318 MACEDONIA STATION DRIVE		POWDER SPRINGS	GA	30127	
CARLISLE, SHERRY	REDACT		REDACT	REDACT	REDACT	
CARL'S PLUMBING SERVICES INC.	1756 S. 56TH ST.		MILWAUKEE	WI	53214	
CARMAX	2696 JOHN HAWKINS PKWY		HOOVER	AL	35244	
CARMICHAEL, TIFFANY	REDACT		REDACT	REDACT	REDACT	
CARNEY, TYLER	REDACT		REDACT	REDACT	REDACT	
CAROL FOR EVA GRAHAM, INC.	366 FIFTH AVENUE		NEW YORK	NY	10001	
CAROLE WREN	30-30 47TH AVENUE	5TH FLOOR	LONG ISLAND CITY	NY	11101	
CAROLINA ENTERPRISES	P.O. BOX 13559		FLORENCE	SC	29504	
CAROLINA ENTERPRISES, INC.	P O BOX 13559		FLORENCE	SC	29504	
CAROLL GAUTIER	1588 SALERNO CIRCLE		WESTON	FL	33327	
CAROLYN CARGAL	2593 CAMELOT WAY		GREENWOOD	IN	46143	
CARPENTER, LISA	REDACT		REDACT	REDACT	REDACT	
CARPET TECH, INC	PO BOX 667		CORDOVA	TN	38088	
CARR, DUANE	REDACT		REDACT	REDACT	REDACT	
CARR, LASHONDA	REDACT		REDACT	REDACT	REDACT	
CARR, MAE	REDACT		REDACT	REDACT	REDACT	
CARR, TISHA	REDACT		REDACT	REDACT	REDACT	
CARRACEDO, JR., FELIX	REDACT		REDACT	REDACT	REDACT	

Page 25 of 176

Name	Address 1	Address 2	City	State	PostalCode	Country
CARRASQUILLO, ALEX	REDACT	REDACT	REDACT	REDACT	REDACT	
CARRIAGE CROSSING MARKET						
PLACE, LLC	2660 EAST CHASE LANE	SUITE100	MONTGOMERY	AL	36117	
CARRIAGE CROSSING MARKET						
PLACE, LLC	2660 EastChase Lane	Suite 100	Montgomery	AL	36117	
CARRICK, ANTOINETTE	REDACT		REDACT	REDACT	REDACT	
CARRIER, SAVANA	REDACT		REDACT	REDACT	REDACT	
CARRINI - LADIES	145 TALMADGE RD.		EDISON	NJ	08817	
CARRION, CHRISTOPHER	REDACT		REDACT	REDACT	REDACT	
CARSON, KEMO	REDACT		REDACT	REDACT	REDACT	
CARSON, TARRON	REDACT	REDACT	REDACT	REDACT	REDACT	
CARTER, BREANNA	REDACT		REDACT	REDACT	REDACT	
CARTER, ELAINE	REDACT	REDACT	REDACT	REDACT	REDACT	
CARTER, LAUREEN	REDACT	REDACT	REDACT	REDACT	REDACT	
CARTER, LAUREEN	REDACT		REDACT	REDACT	REDACT	
CARTER, NASHYIAH	REDACT		REDACT	REDACT	REDACT	
CARTER, ROY	REDACT		REDACT	REDACT	REDACT	
CARTER, TABATHA	REDACT		REDACT	REDACT	REDACT	
CARTER, THELMA	REDACT		REDACT	REDACT	REDACT	
CARTER, THERESA	REDACT	REDACT	REDACT	REDACT	REDACT	
CARTWRIGHT, MARTINEZ	REDACT		REDACT	REDACT	REDACT	
CARVER, BELL	REDACT		REDACT	REDACT	REDACT	
CASE, DOREEN	REDACT		REDACT	REDACT	REDACT	
CASH, INC.	402 WILKINS WISE ROAD	SUITE 21	COLUMBUS	MS	39705	
CASSEY, TAMEKA	REDACT		REDACT	REDACT	REDACT	
CASSIDY GLASS	2283 Mt. Olive Rd.		Gardendale	AL	35071	
CASTLEPOINT RISK MANAGEMENT	PO BOX 905930		CHARLOTTE	NC	28290-5930	
CATHEY, TIMIKA	REDACT		REDACT	REDACT	REDACT	
CATO, CASSANDRA	REDACT		REDACT	REDACT	REDACT	
CAUDLE, AQUARIUS	REDACT		REDACT	REDACT	REDACT	
CAVE, EDNA	REDACT		REDACT	REDACT	REDACT	
CAVE, EDNA	REDACT		REDACT	REDACT	REDACT	
CB CHICAGO PARTNERS, LTD	LOCKBOX # 9947	PO BOX 8500	PHILADELPHIA	PA	19178-9947	
CB&T (Columbus Bank &						
Trust)(Community Bank&Trust)	PO Box 120		Columbus	GA	31902	
	c/o Victory Real Estate Investments,					
CEDAR HILLS CONSOLIDATED, LLC	LLC	240 Brookstone Centre Parkway	Columbus	GA	31904	

Page 26 of 176

Name	Address 1	Address 2	City	State	PostalCode	Country
CEDAR SOUTH PHILADELPHIA 1 LL	C/O CEDAR SHOPPING CENTERS IN,	44 SOUTH BAYLES AVE.	PORT WASHINGTON	NY	11050	
	C/O CEDAR SHOPPING CENTERS					
CEDAR SOUTH PHILADELPHIA 1 LL	INC.	44 SOUTH BAYLES AVENUE	PORT WASHINGTON	NY	11050	
	SOUTH PHILADELPHIA S/C/Store					
Cedar South Philadelphia I, LLC	No242	c/o Cedar Realty Trust, Inc.	Port Washington	NY	11050	
	SOUTH PHILADELPHIA S/C/Store					
Cedar South Philadelphia I, LLC	No242	c/o Cedar Realty Trust, Inc.	Port Washington	NY	11050	
CEDAR-SOUTH PHILADELPHIA I, LLC	C/O CEDAR REALTY TRUST, INC.	44 SOUTH BAYLES AVE #304	PORT WASHINGTON	NY	11050	
CEDAR-SOUTH PHILADELPHIA I, LLC	C/O CEDAR REALTY TRUST INC.	44 SOUTH BAYLES AVE,	PORT WASHINGTON	NY	11050	
CEDAR-SOUTH PHILADELPHIA I, LLC	C/O Cedar Realty Trust, Inc.	44 South Bayles Avenue	Port Washington	NY	11050	
CEDAR-SOUTH PHILADELPHIA I, LLC	Cedar Realty Trust, Inc.	44 South Bayles Avenue	Port Washington	NY	11050	
CEDRIKA LASTER	322 PARADISE LAKE LANE		HOOVER	AL	35244	
CELEBRITY PINK	1708 GAGE ROAD		MONTEBELLO	CA	90640	
CENTERPOINT ENERGY	525 Milam St 9th Floor		Shreveport	LA	71151	
CENTERPOINT ENERGY	PO BOX 4583		HOUSTON	TX	77210-4583	
CENTERPOINT ENERGY	PO BOX 4981		HOUSTON	TX	77210-4981	
CENTRAL CASHERING-TAMPA (FALSE						
ALARMS)	CITY OF TAMPA	2105 N. NEBRASKA AVENUE	TAMPA	FL	33602	
CENTRAL COLLECTION AGENCY	DIVISION OF TAXATION	205 W. SAINT CLAIR AVENUE	CLEVELAND	ОН	44113-1503	
CENTRAL PAPER CO., INC	PO BOX 12807		BIRMINGHAM	AL	35202-2807	
Centrecorp Management Services	Attn.: Jeff J. Procell, CPM, CSM,					
LLP	RPA	4400 A North Freeway Suite 900	Houston	TX	77022	
CENTRO BRADLEY SPE 3 LLC	c/o Centro Properties Group	420 Lexington Avenue	New York	NY	10170	
CENTRO BRADLEY SPE 3, LLC	C/O CENTRO PROP. GROUP	23676 NETWORK PLACE	CHICAGO	IL	60673-1240	
CENTRO GA APOLLO II SUB, LLC	LEASE ID 5218009	PO BOX 281116	ATLANTA	GA	30384-1116	
Centro NP Holdings 12 SPE, LLC	NORTHTOWN PLAZA/Store No303	420 Lexington Avenue, 7th Floor	New York	NY	10170	
CENTRO NP HOLDINGS 12 SPE, LLC	420 LEXINGTON AVENUE		NEW YORK	NY	10170	
CENTRO NP HOLDINGS 12 SPE, LLC	PO BOX 74205		CLEVELAND	ОН	44194	
	C/O BRE RETAIL RESIDUAL OWNER					
CENTRO NP HOLDINGS 8 SPE, LLC	1 LLC	PO BOX 71358	CINCINNATI	ОН	45271-3538	

Name	Address 1	Address 2	City	State	PostalCode	Country
CENTRO NP RESIDUAL SHOPS	C/O BRE RETAIL RESIDUAL OWNER I					
RIVERDALE LLC	LLC	P O BOX 713530	CINCINNATI	ОН	45271-3530	
CENTURY CENTER, LLC	73 State Street	Suite 209	Springfield	MA	1103	
CENTURY CENTER, LLC.	73 STATE STREET	SUITE 209	SPRINGFIELD	MA	01103	
CENTURY ELEVATOR, INC	6801 MARLBORO PIKE		FORESTVILLE	MD	20747	
CHAMBERS, LATRICE	REDACT		REDACT	REDACT	REDACT	
CHAMBLEY, LINDA	REDACT		REDACT	REDACT	REDACT	
CHAMPION SAFE & LOCK SERVICE	PO BOX 11436		CHARLOTTE	NC	28220	
CHAMPION, BEVERLY	REDACT		REDACT	REDACT	REDACT	
Chan Yong Jeong, Esq.	Counsel for Urban	Jeong & Likens, L.C.	Los Angeles	CA	90017	
CHANCEY & REYNOLDS, INC	614 VAN STREET		KNOXVILLE	TN	37921	
CHANEY, AVONI	REDACT		REDACT	REDACT	REDACT	
CHANEY, LILLIAN	REDACT		REDACT	REDACT	REDACT	
CHANEY, LILY	REDACT		REDACT	REDACT	REDACT	
CHAPMAN, CHANNON	REDACT		REDACT	REDACT	REDACT	
CHAPMAN, SHARVELLE	REDACT		REDACT	REDACT	REDACT	
CHAPMAN, THELMA	REDACT		REDACT	REDACT	REDACT	
CHAPTER 13 OFFICE	PO BOX 3613		DURHAM	NC	27702-3613	
CHAPTER 13 TRUSTEE	MIDDLE DISTRICT-ATHENS &					
(ATHEN&MAC)	MACON	PO BOX 102043	ATLANTA	GA	30368-2043	
CHARLES, SANDRA	REDACT		REDACT	REDACT	REDACT	
CHARLES, SIKERRA	REDACT	REDACT	REDACT	REDACT	REDACT	
CHARLESTON COUNTY (SOLID						
WASTE)	P.O. BOX 100166		COLUMBIA	SC	29202-3166	
CHARLESTON COUNTY (BUS LIC- SC)	4045 BRIDGE VIEW DRIVE		NORTH CHARLESTON	sc	29405-7464	
CHARLESTON COUNTY TREASURER	PO BOX 100242		COLUMBIA	sc	29202-3242	
CHARLESTON WATER SYSTEM	103 Saint Philip Street		CHARLESTON	SC	29403	
CHARLESTON WATER SYSTEM	PO BOX 568		CHARLESTON	SC	29402-0568	
CHARLIE CALDWELL JR	CITY MARSHALL	1244 TEXAS AVENUE	SHREVEPORT	LA	71101	
CHARLOTTE ALARM MGMT						
SERVICES	PO BOX 602486		CHARLOTTE	NC	28260-5486	
CHARLOTTE/MECKLENBERG COUNTY	TAX COLLECTOR	PO BOX 1400	CHARLOTTE	NC	28201-1400	
CHARTER ONE	PO Box 7000	ROP450	Providence	RI	02940	
Charter Township of Clinton	Atnn:Kim Meltzer	40700 Romeo Plank Rd	Clinton Township	МІ	48038	

Name	Address 1	Address 2	City	State	PostalCode	Country
			,			,
CHARTER TOWNSHIP OF REDFORD	TREASURER	PO BOX 39401	REDFORD	MI	48239	
CHARTER TOWNSHIP OF YPSILANTI	7200 S. HURON RIVER DRIVE		YPSILANTI	MI	48197	
CHASE CARD SERVICES	CARDMEMBER SERVICE	PO BOX 94014	PALATINE	IL	60094-4014	
CHASE, CHRISTINE	REDACT		REDACT	REDACT	REDACT	
CHATEAU INT'L INC.	188 WHITMAN AVE.		EDISON	NJ	08817	
CHATHAM COUNTY	TAX COMMISSIONER	P.O. BOX 9827	SAVANNAH	GA	31412	
CHATMAN, DA'SHAWN	REDACT		REDACT	REDACT	REDACT	
CHATTANOOGA GAS	2207 Olan Mills Drive		Chattanooga	TN	37421	
CHATTANOOGA GAS	PO BOX 59004		KNOXVILLE	TN	37950-9004	
CHEAPER LAWN CARE & SERVICES,						
INC.	19283 COUNTY ROAD 532		GREENWOOD	MS	38930	
CHEATHAM, ANTOINE	REDACT		REDACT	REDACT	REDACT	
CHERISH	2460 EAST 12TH ST		LOS ANGELES	CA	90021	
CHEROKEE ELECTIRC CO. INC	1747 SKYLINE ROAD		MERIDIAN	MS	39301	
CHERRY, CYNTHIA	REDACT		REDACT	REDACT	REDACT	
CHERRY, MORGAN	REDACT		REDACT	REDACT	REDACT	
CHERRY, RACHEL	REDACT		REDACT	REDACT	REDACT	
Chesapeake City	Attn: Consumer Protection Agency	Jan L. Procter - City Attorney	Chesapeake	VA	23322	
CHESAPEAKE CROSSING S/C C/O	THALHEIMER MGMT CO	PO BOX 702	RICHMOND	VA	23218-0702	
CHICAGO FLOOR SYSTEMS CORP	1029 N THROOP		CHICAGO	IL	60642	
CHICAGO METRO ASSOCIATES LP	C/O SPATZ CENTERS LLC	14 N. PEORIA ST. 3F	CHICAGO	IL	60607	
	C/O AT PROPERTIES, PROPERTY					
CHICAGO TITLE LAND TRUST CO.	MGMT	212 E. OHIO ST	CHICAGO	IL	60611	
CHIEF FIRE PROTECTION COMPANY	689 WHITEHALL ST SW		ATLANTA	GA	30310	
CHILDRES, MARIE	REDACT	REDACT	REDACT	REDACT	REDACT	
CHILDS, OLIVIA	REDACT		REDACT	REDACT	REDACT	
CHILDS, TAYLOR	REDACT	REDACT	REDACT	REDACT	REDACT	
CHOICE TEMPS, INC.	P.O. BOX 75343		CHICAGO	IL	60675-5343	
CHRISTIAN, D ANGELA	REDACT		REDACT	REDACT	REDACT	
CHRISTIAN, NICHOLE	REDACT		REDACT	REDACT	REDACT	
CHRISTIAN, NIKOLE	REDACT		REDACT	REDACT	REDACT	
C-III ASSET MANAGEMENT LLC	5557 CANAL BLVD		NEW ORLEANS	LA	70124	
C-III ASSET MGMT LLC	C/O FORESITE REALTY MGMT, LLC	9525 BRYN MAWR AVE	ROSEMONT	IL	60018	

Name	Address 1	Address 2	City	State	PostalCode	Country
CIII, SS-CSMS07C5, CAHOKIA			,			,
VILLAGE	C/O THE DESCO GROUP, INC.	25 N. BRENTWOOD BLVD.	ST.LOUIS	мо	63105	
Cincinnati Inc. Tax Bureau	Income Tax Office	805 Central Ave.	Cincinnati	ОН	45202	
Cincinnati Insurance Company						
(CPP/CPA 1082327) (ECP 0316780)						
(CAP 5242001) (ECP 0319340)	P.O. Box 145496		Cincinnati	ОН	45250-5496	
Cincinnati Insurance Company						
(CPP/CPA 1082327) (ECP 0316780)						
(CAP 5242001) (ECP 0319340)	6200 S. Gilmore Road		Fairfield	ОН	45014-5141	
CIRCLE CITY PEST CONTROL	5747 N POST ROAD		INDIANAPOLIS	IN	46216	
CIRCUIT COURT OF JACKSON						
COUNTY	PO BOX 219569		KANSAS CITY	MO	64121	
CIRCUIT COURT OF MONTGOMERY	MONTGOMERY COUNTY					
COUNTY	COURTHOUSE	251 S. LAWRENCE STREET	MONTGOMERY	AL	36104	
CIT	Attn.: Tina Moore	134 Wooding Avenue	Danville	VA	24541	
	Attn.: Gerard Imbriano, Vice					
CIT	President	Commercial Services	New York	NY	10036	
CIT	Attn.: Kevin Ritter	Commercial Services	New York	NY	10036	
CIT	P.O. Box 1036		Charlotte	NC	28301-1036	
CITATION COLLECTION SERVICES	PO BOX 80239		INDIANAPOLIS	IN	45280	
	Attn: STEPHEN PATTON -					
Cith of Chicago	CORPORATION COUNSEL	121 NORTH LASALLE STREET	Chicago	IL	60602	
CITIZENS ENERGY GROUP	2020 North Meridian St.		INDIANAPOLIS	IN	46202	
CITIZENS ENERGY GROUP	PO BOX 7056		INDIANAPOLIS	IN	46207-7056	
CITIZENS NATIONAL BANK - MS	PO Box 911		Meridian	MS	39302	
City Attorney	Augusta Law Department	520 Greene St.	Augusta	GA	30901	
City Attorney	600 E 4th St		Charlotte	NC	28202	
City Attorney	626 State Street		Erie	PA	16501	
City Attorney	Peter Bade	City Hall	Flint	MI	48502	
City Attorney	Nicolle Shalley	P.O. Box 490	Gainesville	FL	32627-0490	
City Attorney	206 S. Main Street		Greenville	SC	29601	
City Attorney	251 E. Ohio St. Suite, 160		Indianapolis	IN	46204	
City Attorney	Tami Alexander	City of Jacksonville	Jacksonville	FL	62650	
City Attorney	28th Floor, City Hall	414 E 12th St.	Kansas City	MO	64106	
City Attorney	Terry A. Marecki	33000 Civic Center Drive	Livonia	MI	48154	
City Attorney	125 N. Main St. Room 336		Memphis	TN	38103	

Name	Address 1	Address 2	City	State	PostalCode	Country
City Attorney	One Public Square, Suite 204	P. O. Box 196300	Nashville	TN	37219-6300	
City Attorney	2400 Washington Avenue	9th Floor	Newport News	VA	23607	
City Attorney	James M. Messer	222 W. Main St.	Pensacola	FL	32502	
City Attorney	Terri Scott	505 Travis St	Shreveport	LA	71101	
City Attorney	Office of the City Attorney	2201 University Blvd 35401	Tuscaloosa	AL	35403-2089	
City Attorney	PO Box 2511		Winston-Salem	NC	27102	
City Attorney or City Manager	700 Poplar St.		Macon	GA	31201	
City Attorney or City Manager	205 Lawrence Street		Marietta	GA	30060	
CITY FASHION EXPRESS, INC	PO BOX 894829		LOS ANGELES	CA	90189	
City of Manchester	Attn: Ryan Barry, Town Attorney	Town Hall, 2nd Floor	Manchester	ст	06040	
City of ALBANY, GA	Attn: C. Nathan Davis, City Attorney	222 Pine Avenue	Albany	GA	31701	
CITY OF ALEXANDRIA	Mr. Chuck Johnson- City Attorney	915 Third Street	Alexandria	LA	71301	
CITY OF ALEXANDRIA	TAX COLLECTOR/BUSINESS OFFICE	PO BOX 71	ALEXANDRIA	LA	71309-0071	
CITY OF ALEXANDRIA- UTILITIES	625 MURRAY ST		ALEXANDRIA	LA	71301	
CITY OF ALEXANDRIA- UTILITIES	PO BOX 1925		LAKE CHARLES	LA	70602-1925	
CITY OF ANNISTON	P.O. BOX 2168		ANNISTON	AL	36202-2168	
City of ATLANTA	Attn: Cathy Hampton	Department of Law	Atlanta	GA	30303	
CITY OF ATLANTA	GENERAL BUSINESS LICNESE	PO BOX 932053	ATLANTA	GA	31193-2053	
City of AUSTELL	Attn: Carolyn Duncan, City Clerk:	5000 Austell-Powder Springs Road	Austell	GA	30106	
City of BALTIMORE	Attn: George Nilson, City Solicitor	Baltimore City Law Department	Baltimore	MD	21202	
CITY OF BALTIMORE	COLLECTION DIVISION	200 HOLIDAY STREET	BALTIMORE	MD	21202	
CITY OF BALTIMORE	250 CITY HALL		BALTIMORE	MD	21202	
CITY OF BALTIMORE	REVENUE COLLECTIONS	P.O. Box 17535	BALTIMORE	MD	21297-1535	
City of Batesville	Attn: Richard Phillips	103 College St	Batesville	MS	38606-2197	
CITY OF BATESVILLE	PO BOX 689	Ü	BATESVILLE	MS	38606	
CITY OF BATESVILLE GAS & WATER	106 COLLEGE		BATESVILLE	MS	38606	
CITY OF BATESVILLE GAS & WATER	PO BOX 689		BATESVILLE	MS	38606-0689	
CITY OF BATON ROUGE	BRPD- ALARM ENFORCEMENT DIV.	P.O. BOX 2406	BATON ROUGE	LA	70821	

Name	Address 1	Address 2	City	State	PostalCode	Country
CITY OF BATON ROUGE ALARM						
ENFO	ATTN: ALARM ENFORCEMENT	P.O. BOX 2406	BATON BOUGE	LA	70821	
City of Baton Rouge-Parish of East						
Baton Rouge	Dept of Finance	PO Box 2590	Baton Rouge	LA	70821-2590	
CITY OF BESSEMER	CLERK AND TREASURER	1806 THIRD AVENUE NORTH	BESSEMER	AL	35020	
	Thomas Bentley , Acting City					
City of Birmingham	Attorney	Legal Department	Birmingham	AL	35203	
CITY OF BIRMINGHAM	REVENUE DIVISION	PO BOX 830638	BIRMINGHAM	AL	35283-0638	
City of Bossier	Attn: City Clerk	620 Benton Road	Bossier City	LA	71171-5337	
CITY OF BOSSIER CITY	DEPT. OF FINANCE	P.O.BOX 5399	BOSSIER CITY	LA	71171-5399	
	Attn: Elissa Levan and Karen Ruff,					
City of Bowie	City Attorney	C/O Funk & Bolton, P.A.	Baltimore	MD	21201	
	Attn: Mary Ann Coogan -	, i				
City of Camillus	Supervisor	4600 W. Genesee Street	Syracuse	NY	13219	
•	Attn: Joseph Martuccio - Law					
City of Canton	Director	Canton City Hall	Canton	ОН	44702	
•	Attn: Scoot A. Liberman - Municipal	·				
City of Centerville	Attorney	RE: City of Centerville	Dayton	ОН	45402	
	Attn: Charlton deSaussure, Jr					
City of Charleston	Corporation Counsel	80 Broad Street	Charleston	sc	29401-0304	
CITY OF CHARLESTON SC	PO BOX 22009		CHARLESTON	SC	29413-2009	
CITY OF CHATTANOOGA	TREASURER	P.O. BOX 191	CHATTANOOGA	TN	37401-0191	
	Attn: KEVIN G. SCHENK, Town					
City of CHEEKTOWAGA	Attorney	3301 Broadway	Cheektowaga	NY	14227	
		-				
City of Chesapeake	Attn: Jan L. Proctor - City Attorney	306 Cedar Road	Chesapeake	VA	23322	
CITY OF CHESAPEAKE TREASURER	P.O. BOX 16495		CHESAPEAKE	VA	23328-6495	
CITY OF CHESAPEAKE, VA	COMMISSIONER OF THE REVENUE	P.O.BOX 15285	CHESAPEAKE	VA	23328-5285	
CITY OF CHICAGO - BLDGS	DEPT OF REVENUE	8108 INNOVATION WAY	CHICAGO	IL	60682-0081	
	DEPT. OF FINANCE, COLLECTIONS					
CITY OF CHICAGO - LASALLE	UNIT	CITY HALL, ROOM 107A	CHICAGO	IL	60602	
CITY OF CHICAGO- DEPT OF FINANCE	COLLECTIONS UNIT	CITY HALL, ROOM 107A	CHICAGO	IL	60602	
CITY OF CHICAGO DEPT. OF						
BUILDINGS	DEPARTMENT OF REVENUE	8108 INNOVATION WAY	CHICAGO	IL	60682-0081	
City of Chicago Heights	Attn: Lori Wilcox – City Clerk	1601 Chicago Road	Chicago Heights	IL	60411	
CITY OF CHICAGO HEIGHTS	CITY CLERK	1601 CHICAGO RD.	CHICAGO HEIGHTS	IL	60411	

Page 32 of 176

Name	Address 1	Address 2	City	State	PostalCode	Country
City of Clarks Summit	Attn: Patrick M. Rogan - Solicitor	RE: City of CLARKS SUMMIT	Scranton	PA	18508	
	Attn: Curtis D. Boschert, City					
City of Clarksdale	Attorney	121 Sunflower Avenue	Clarksdale	MS	38614	
CITY OF CLARKSDALE, MS.	PO BOX 940		CLARKSDALE	MS	38614	
City of Cleveland	Attn: Consiumer Protection Agency	Jeanne M. Schmotzer, MA, CMC	Cleveland	ОН	44115	
CITY OF CLEVELAND	TAX COLLECTOR	P.O. BOX 1439	CLEVELAND	MS	38732	
City of Columbia	Teresa Wilson	P.O. Box 147	Columbia	SC	29217	
CITY OF COLUMBIA	1737 Main Street		COLUMBIA	SC	29201	
CITY OF COLUMBIA	PO BOX 7997		COLUMBIA	SC	29202-7997	
CITY OF COLUMBIA, SC	LICENSE DIVISION	PO BOX 7997	COLUMBIA	SC	29202	
City of Columbus	Attn: Clifton Fay - City Attorney	100 10th Street	Columbus	GA	31901	
	Attn: JEFFREY J. TURNAGE - City					
City of Columbus	Attorney	523 Main St.	Columbus	MS	39701	
City of Columbus	Richard C Pfeiffer, Jr	77 North Front Street	Columbus	ОН	43125	
City of Columbus	Income Tax Division	PO Box 182158	Columbus	ОН	43218-2158	
	ATTN: J. Ross Dyer, County					
City of CORDOVA	Attorney	160 N. Main Suite 950	Memphis	TN	38103	
CITY OF COVINGTON	P.O. BOX 1527		COVINGTON	GA	30015	
CITY OF COVINGTON	2194 Emory St NW		Covington	GA	30014	
	ATTN: Mark Perkins, City					
City of CREVE COEUR	Administrator	300 N New Ballas Road	Creve Coeur	МО	63141	
City of Dallas	Warren M.S. Ernst	1500 Marilla St #7DN	Dallas	TX	75201	
CITY OF DALLAS (ALARM PERMITS)	SECURITY ALARMS	PO BOX 139076	DALLAS	TX	75313-9076	
CITY OF DALLAS SPECIAL						
COLLECTIONS	CITY HALL, 2D SOUTH		DALLAS	TX	75277	
CITY OF DALLAS- WATER UTILITIES	1500 Marilla Street #2DS	Dallas Water Utilities Department	DALLAS	TX	75201	
CITY OF DALLAS- WATER UTILITIES	CITY HALL, 2D SOUTH		DALLAS	TX	75277	
	Attn: Ridley S. Nimmo, II - City					
City of Davison	Attorney	C/O Plunkett Cooney, P.C.	Flint	MI	48502	
City of DAYTON	ATTN: Warren Price - City Manager	101 W. Third Street	DAYTON	ОН	45402	
City of Dayton	Income Tax Division	PO Box 2806	Dayton	ОН	45401-2806	
CITY OF DAYTON	101 W. THIRD ST		DAYTON	ОН	45402	

Page 33 of 176

Name	Address 1	Address 2	City	State	PostalCode	Country
CITY OF DAYTON	P.O. BOX 740575		CINCINNATI	ОН	45263-0575	
CITY OF DAYTONA BEACH (SIGN						
RENEWALS)	PO BOX 2451		DAYTONA BEACH	FL	32115-2451	
CITY OF DEARBORN	DEPT. 3102	PO BOX 30516	LANSING	MI	48909-8016	
City of DECATUR	ATTN: Peggy Merriss - City Manager	509 N. McDonough St	Decatur	GA	30031	
City of DETROIT	ATTN: Janice M. Winfrey - City Clerk	2 Woodward Ave Suite 200	DETROIT	MI	48226	
	DEPT OF ADMINISTRATIVE	COLEMAN A. YOUNG MUNICIPAL				
CITY OF DETROIT	HEARINGS	CTR	DETROIT	MI	48226	
CITY OF DETROIT	TREASURER-FIRE MARSHALL DIV	1301 THIRD STREET	DETROIT	MI	48226	
CITY OF DETROIT	DEPT # 268301	PO BOX 55000	DETROIT	MI	48255-2683	
	BLDGS & SAFETY ENGINEERING	402 COLEMAN A. YOUNG				
CITY OF DETROIT- BLDGS & SAFETY	DEPT	MUNICIPAL CTR	DETROIT	MI	48226	
CITY OF DETROIT- TREASURER	INCOME TAX	PO BOX 67000	DETROIT	MI	48267-1319	
	ATTN: Michael K. West City					
City of DOTHAN	Manager	126 N. Saint Andrews Street	DOTHAN	AL	36303	
CITY OF DOTHAN	LICENSE DIVISION	PO BOX 2128	DOTHAN	AL	36302	
CITY OF DOTHAN- UTILITIES	200 Kilgore Dr		DOTHAN	AL	36301	
CITY OF DOTHAN- UTILITIES	PO BOX 6728		DOTHAN	AL	36302	
CITY OF DURHAM	101 CITY HALL PLAZA		DURHAM	NC	27701	
CITY OF EAST POINT	BUSINESS LICENSE DIVISION	1526 E. FORREST AVE	EAST POINT	GA	30344	
CITY OF EAST POINT	2791 EAST POINT STREET		EAST POINT	GA	30344-3239	
City of East St. Louis	ATTN: Dorene Hoosman, City Clerk	301 River Park Dr.	East St. Louis	IL	62201	
CITY OF EAST ST. LOUIS IL.	DEPT OF REGULATORY AFFAIRS	301 RIVER PARK DR.	EAST ST.LOUIS	IL	62201	
CITY OF EAST ST. LOUIS- SEWER						
DEPT	301 RIVER PARK DR.		ST. LOUIS	IL	62201	
CITY OF EAST ST. LOUIS- SEWER						
DEPT	PO BOX 472165		TULSA	OK	74147-2165	
CITY OF FAIRBURN	56 MALONE STREET		FAIRBURN	GA	30213	
CITY OF FAIRFIELD	4701 GARY AVE.	P.O. DRAWER 437	FAIRFIELD	AL	35064	
CITY OF FAIRFIELD, AL	OCCUPATIONAL TAX ADMIN	PO DRAWER 437	FAIRFIELD	AL	35064	
	ATTN: Karen M. McDonald - City					
City of Fayetteville	Attorney	433 Hay Street	Fayetteville	NC	28301	
CITY OF FAYETTEVILLE	FINANCE DEPT	PO DRAWER D	FAYETTEVILLE	NC	28302	
CITY OF FERGUSON	110 CHURCH STREET		FERGUSON	МО	63135	
CITY OF FLINT	PO BOX 2056		FLINT	MI	48501	

Page 34 of 176

Name	Address 1	Address 2	City	State	PostalCode	Country
CITY OF FLINT, TREASURER	ATTN:WITHHOLDING SECTION	PO BOX 529	EATON RAPIDS	MI	48827-0529	
	Attn: James W. Peterson, Jr City	C/O Clarke, Johnson, Peterson &				
City of Florence	Attorney	McLean P.A.	Florence	sc	29501	
CITY OF FLORENCE- UTILITY DIV	1440 MCCURDY ROAD		FLORENCE	SC	29506	
CITY OF FLORENCE- UTILITY DIV	UTILITY FINANCE DIVISION	PO Box 602756	CHARLOTTE	NC	28260-2756	
CITY OF FLORENCE WATER	CITY CENTER	324 W. EVANS STREET	FLORENCE	SC	29501	
CITY OF FLORENCE, SC	BUSINESS LICENSE OFFICE	324 WEST EVANS ST	FLORENCE	SC	29501	
CITY OF FOREST PARK	TAX DEPT.	P.O. BOX 69	FOREST PARK	GA	30298	
	Attn: Robert V. Schwerer - City					
City of Fort Pierce	Attorney	100 N. U.S. Highway 1	Fort Pierce	FL	34950	
City of Fort Worth	Attn: Sarah J. Fullenwider	1000 Throckmorton St.	Fort Worth	TX	76102	
CITY OF FORT WORTH FIRE DEPT	FIRE DEPT REVENUE GROUP	PO BOX 17026	FORT WORTH	TX	76102	
CITY OF FORT WORTH- WATER DEPT	908 MONROE ST.		FORT WORTH	TX	76102	
CITY OF FORT WORTH- WATER DEPT	PO BOX 961003		FORT WORTH	TX	76161-0003	
	PLANNING AND DEVELOPMENT					
CITY OF FORT WORTH, TEXAS	DEPARTMENT ALARM UNIT	1000 THROCKMORTON	FORT WORTH	TEXAS	76102-6311	
CITY OF FT. PIERCE	OFFICE OF CITY CLERK	P.O.BOX 1480	FT. PIERCE	FL	34954	
CITY OF GAINESVILLE	OCC TAX DIVISION	P.O. BOX 490	GAINESVILLE	FL	32602	
CITY OF GARY IN	401 BROADWAY, SUITE 100		GARY	IN	46402	
	ATTN: ANDREW ALEXANDER III -					
City of Greenville	City Attorney	C/O Lake Tindall, LLP	Greenville	MS	38701	
CITY OF GREENVILLE MS.	TAX COLLECTOR	P.O. BOX 9	GREENVILLE	MS	38701	
CITY OF GREENVILLE MS.	TAX COLLECTOR	P.O. BOX 897	GREENVILLE	MS	38702-0897	
CITY OF GREENVILLE- WATER DEPT	127 SOUTH POPULAR		GREENVILLE	sc	38701	
CITY OF GREENVILLE- WATER DEPT	PO BOX 897		GREENVILLE	MS	38702-0897	
	ATTN: Don Brock, Jr., Esq City					
City of Greenwood	Attorney	308 Fulton St.	Greenwood	MS	38930	
CITY OF GREENWOOD, MS	CITY TAX COLLECTOR	PO BOX 907	GREENWOOD	MS	38935	
	ATTN: Michael J. Stebbins - City					
City of Gulf Breeze	Attorney	504 North Baylon Street	Pensacola	FL	32502	
, , , , , , , , , , , , , , , , , , , ,	ATTN: Vanessa T. Valldejuli - City	,				
City of Hampton	Attorney	22 Lincoln St.	Hampton	VA	23669	
CITY OF HAMPTON	TREASURER'S OFFICE	PO BOX 638	HAMPTON	VA	23669-0638	
CITY OF HAMPTON	PO BOX 636		HAMPTON	VA	23669	

Page 35 of 176

Name	Address 1	Address 2	City	State	PostalCode	Country
CITY OF HAMTRAMCK INCOME TAX						
DEPT	3401 EVALINE		HAMTRAMCK	MI	48212	
City of Harper Woods	Attn: J. Russell LaBarge,	19617 Harper	Harper Woods	MI	48225	
CITY OF HARPER WOODS	19617 HARPER AVENUE		HARPER WOODS	MI	48225-2095	
City of Harrisburg	Attn: City Attorney	10 North Second Street	Harrisburg	PA	17101-1678	
CITY OF HARRISBURG	TAX & ENFORCEMENT OFFICE	10 N. 2ND ST, SUITE 305-A	HARRISBURG	PA	17101	
	Charles E Lawrence Jr - City	C/O Charles Lawrence Attorney At				
CITY OF HATTIESBURG	Attorney	Law®	Hattiesburg	MS	39403-1624	
CITY OF HATTIESBURG	CITY TAX DEPT	PO BOX 1898	HATTIESBURG	MS	39403-1898	
CITY OF HATTIESBURG	200 FORREST ST		HATTIESBURG	MS	39401	
CITY OF HATTIESBURG- WATER	PO BOX 1897		HATTIESBURG	MS	39403-1897	
CITY OF HAZEL PARK	111 E. NINE MILE ROAD		HAZEL PARK	MI	48030	
City of Highland Park	Attn: Brenda Green	12050 Woodward Ave	Highland Park	MI	48203-3578	
CITY OF HIGHLAND PARK	12050 WOODWARD		HIGHLAND PARK	MI	48203	
CITY OF HIGHLAND PARK	3401 EVALINE		HAMTRAMCK	MI	48212	
City of Houston	David M. Feldman	City of Houston	Houston	TX	77251	
CITY OF HOUSTON	P.O. BOX 1560		HOUSTON	TX	77251	
CITY OF HOUSTON	900 Bagby, 4th Floor		HOUSTON	TX	77002	
CITY OF HOUSTON (FIRE DEPT)	FIRE DEPT PERMITS	PO BOX 3625	HOUSTON	TX	77253	
CITY OF HOUSTON- ARA ALARM						
ADMIN	ARA ALARM ADMINISTRATOR	PO BOX 203887	HOUSTON	TX	77216-3887	
CITY OF HOUSTON- SIGN						
ADMINISTRATION	PO BOX 2688		HOUSTON	TX	77252-2688	
CITY OF HOUSTON- SOLID WASTE	SOLID WASTE MGMT DEPT	PO BOX 1562	HOUSTON	TX	77251	
CITY OF HOUSTON- WATER DEPT	4200 LEELAND ST		HOUSTON	TX	77023	
CITY OF HOUSTON- WATER DEPT	PO BOX 1560		HOUSTON	TX	77251	
City of Huntsville	Peter Joffrion	City Hall, 6th Floor	Huntsville	AL	35801	
CITY OF HUNTSVILLE	LICENSE DIV.	PO BOX 308	HUNTSVILLE	AL	35804	
	OFFICE OF FINANCE AND					
CITY OF INDIANAPOLIS	MANAGEMENT	200 E. WASHINGTON STREET	INDIANAPOLIS	IN	46204	
	OFFICE OF FINANCE AND	200 E WASHINGTON STREET,				
CITY OF INDIANAPOLIS	MANAGEMENT	ROOM 2260	INDIANAPOLIS	IN	46204	
CITY OF IRONDALE	DEPT. OF FINANCE & REVENUE	PO BOX 100188	IRONDALE	AL	35210	
CITY OF IDOUDALS AWATED SHOTE	5445 D54 COM DD11/5		10010115		25240	
CITY OF IRONDALE- WATER SYSTEM	5415 BEACON DRIVE		IRONDALE	AL	35210	
CITY OF IRONDALE- WATER SYSTEM	P.O. BOX 580091		CHARLOTTE	NC	28258-0091	

Name	Address 1	Address 2	City	State	PostalCode	Country
	ATTN: Monica D. Joiner - City					
City of Jackson	Attorney	455 E Capitol St.	Jackson	MS	39207	
CITY OF JACKSON- PYMT	·	·				
PROCESSING	219 PRESIDENT ST		JACKSON	MS	39205	
CITY OF JACKSON- PYMT						
PROCESSING	PO BOX 1595		JACKSON	MS	39215-1595	
CITY OF JACKSON, MS	LICENSE DIVISION	PO BOX 22708	JACKSON	MS	39225-2708	
CITY OF JACKSONVILLE, FLORIDA	231 E FORSYTH STREET	ROOM 141	JACKSONVILLE	FL	32202	
City of Jennings	Cheryl Balke, City Clerk	Attn: City Attorney	Jennings	МО	63136	
CITY OF JENNINGS	2120 HORD AVENUE		JENNINGS	МО	63136	
	Attn: Doug Bach - County					
City of Kansas City	Administrator	701 N 7th Street	Kansas City	KS	66101	
CITY OF KANSAS CITY, MO	REVENUE DIVISION	PO BOX 840101	KANSAS CITY	МО	64184-0101	
CITY OF KENTWOOD- TAX	PO BOX 88018		CHICAGO	IL	60680-1018	
		C/O Smith Moore Leatherwood,				
City of Knightdale	Attn: Clyde Holt III, Town Attorney	LLP, Suite 2800	Raleigh	NC	27601	
City of Knoxville	Charles Swanson	Room 699, City County Building	Knoxville	TN	37902	
CITY OF KNOXVILLE, TN	REVENUE OFFICE	P.O. BOX 15001	KNOXVILLE	TN	37901-5001	
City of Lafayette	Attn: Michael D. Hebert	705 West University Avenue	Lafayette	LA	70506	
CITY OF LAGRANGE	PO BOX 430	,	LAGRANGE	GA	30241	
	Attn: Christy Goddeau - City	C/O Torcivia, Donlon, Goddeau &				
City of Lake Worth	Attorney	Ansay, P.A.	West Palm Beach	FL	33407	
·	·	,				
City of Lansing	Attn: Donna Kooyenga, Village Clerk	3141 Ridge Rd.	Lansing	IL	60438	
CITY OF LANSING	TREASURER	PO BOX 40712	LANSING	MI	48901-7912	
	Attn: Charles Faranda - City					
City of Lauderhill	Manager	5581 W. Oakland Park Blvd.	Lauderhill	FL	33313	
CITY OF LAUDERHILL- UTILITY	2000 CITY HALL DR		LAUDERHILL	FL	33313	
CITY OF LAUDERHILL- UTILITY	PO BOX 31511		TAMPA	FL	33631-3511	
CITY OF LAUDERHILL, FL	5581 W OAKLAND PARK BLVD		LAUDERHILL	FL	33313	
		C/O HORTMAN HARLOW BASSI				
City of Laurel	Attn: Diedra Bassi, City Attorney	ROBINSON & MCDANIEL PLLC	Laurel	MS	39441	
CITY OF LAUREL CITY CLERKS OFFICE	CITY CLERKS OFFICE	PO BOX 647	LAUREL	MS	39441	
CITY OF LAUREL PUBLIC UTILITY	2504 Moose Dr		Laurel	MS	39440	
CITY OF LAUREL PUBLIC UTILITY	PO BOX 647		LAUREL	MS	39441-0647	
City of Lebanon	Attn: Mark S. Yurick - City Attorney	C/O Taft Stettinius & Hollister LLP	Columbus	ОН	43215-4213	

Page 37 of 176

Name	Address 1	Address 2	City	State	PostalCode	Country
CITY OF LEOMINSTER	BUILDING DEPARTMENT	25 WEST STREET	LEOMINSTER	MA	01453	
		500 West Markham Street, Suite				
City of Little Rock	Thomas M. Carpenter	310	Little Rock	AR	72201	
-	DEPT OF PLANNING &					
CITY OF LITTLE ROCK (DEVELOPMENT	723 WEST MARKHAM STREET	LITTLE ROCK	AR	72201-1334	
CITY OF LITTLE ROCK (REVENUE)	REVENUE DIVISION	500 W. MARKHAM ST, RM 100	LITTLE ROCK	AR	72201	
CITY OF LIVONIA	INSPECTION DEPT	33000 CIVIC CENTER DRIVE	LIVONIA	MI	48154	
	Attn: Chip Fletcher, County					
City of Lutz	Attorney	C/O: Hillsborough County	Tampa	FL	33602	
CITY OF MACON	CITY CLERK'S OFFICE	PO BOX 247	MACON	GA	31202-0247	
		City of Maple Heights Law				
City of Maple Heights	Attn: John J. Montello, Law Director	Department	Maple Heights	ОН	44137	
CITY OF MAPLE HEIGHTS	5353 LEE ROAD	·	MAPLE HEIGHTS	ОН	44137	
CITY OF MARKHAM	16313 KEDZIE PARKWAY		MARKHAM	IL	60428	
CITY OF MARKHAM, IL	16313 S. KEDZIE PARKWAY		MARKHAM	IL	60428	
CITY OF MEMPHIS	TREASURER	PO BOX 185	MEMPHIS	TN	38101-0185	
	Pam McInnis - Clerk and Hugh					
CITY OF MERIDIAN	Smith	City Hall	Meridian	MS	39302-1430	
CITY OF MERIDIAN	PO BOX 1430	·	MERIDIAN	MS	39302	
CITY OF MERIDIAN	301 23RD AVENUE		MERIDIAN	MS	39302-1430	
CITY OF MERIDIAN	PO BOX 231		MERIDIAN	MS	39302-0231	
	Attn: Victoria Méndez, City					
City of Miami	Attorney	Office of the City Attorney	Miami	FL	33130	
CITY OF MIAMI (ALARM UNIT)	ALARM UNIT	444 SW 2ND AVE	MIAMI	FL	33130	
CITY OF MIAMI (ATL)	P.O. BOX 105206		ATLANTA	GA	30348-5206	
	Attn: Sonja Knighton Dickens, City					
City of Miami Gardens	Attorney	18605 NW 27th Avenue	Miami Gardens	FL	33056	
CITY OF MILLEDGEVILLE	119 E. HANCOCK ST	PO BOX 1900	MILLEDGEVILLE	GA	31059-1900	
CITY OF MILLEDGEVILLE	119 E. HANCOCK ST		MILLEDGEVILLE	GA	31061	
CITY OF MILLEDGEVILLE	PO BOX 1900		MILLEDGEVILLE	GA	31059-1900	
City of Milwaukee	Attn: Grant F. Langley, City Attorney	Frank P. Zeidler Municipal Building	Milwaukee	wı	53202	
CITY OF MILWAUKEE	CITY TREASURER	PO BOX 78776	MILWAUKEE	WI	53278-0776	
CITY OF MILWAUKEE	LICENSE DIVISION	200 E. WELLS STREET, ROOM 105	MILWAUKEE	WI	53202	
CITY OF MILWAUKEE	MILWAUKEE DEV CENTER	809 N. BROADWAY	MILWAUKEE	WI	53202-3617	
-		-				
CITY OF MILWAUKEE HEALTH DEPT	DISEASE CONTROL &	ENVIRONMENTAL HEALTH	MILWAUKEE	wı	53202	
City of Mobile	Ricardo A. Woods	205 Government Street	Mobile	AL	36644	

Page 38 of 176

Name	Address 1	Address 2	City	State	PostalCode	Country
CITY OF MOBILE	REVENUE DEPT. #1530	P.O. BOX 11407	BIRMINGHAM	AL	35246-1530	
CITY OF MOBILE- SALES TAX	DEPT# 1519	PO BOX 11407	BIRMINGHAM	AL	35246-1519	
	Attn: Nanci S. Summersgill, City					
City of Monroe	Attorney	400 Lea Joyner Memorial Expwy	Monroe	LA	71201-7509	
CITY OF MONROE	TAX AND REVENUE DEPT.	P.O. BOX 123	MONROE	LA	71210	
CITY OF MONROE	300 WEST CROWELL STREET		MONROE	NC	28112	
CITY OF MONROE	UTILITY OPERATIONS DIVISION	209 South 12th Street	MONROE	LA	71202	
City of Monroeville	Gregory Erosenko	2700 Monroeville Blvd.	Monroeville	PA	15146	
City of Montgomery	Kim O. Fehl	103 North Perry St.	Montgomery	AL	36104	
CITY OF MONTGOMERY	DEPT OF FINANCE- SALES TAX	PO BOX 830469	BIRMINGHAM	AL	35283-0469	
CITY OF MONTGOMERY	DEPT. OF FINANCE-LICENSE	P.O.BOX 830469	BIRMINGHAM	AL	35283-0469	
City of Morrow	Attn: Viet Tran, City Clerk	1500 Morrow Rd.	Morrow	GA	30260	
CITY OF MORROW	1500 MORROW RD		MORROW	GA	30260	
CITY OF MUSKEGON	Frank Peterson - City Manager	933 Terrace St.	Muskegon	MI	49440	
CITY OF NATCHEZ	124 SOUTH PEARL ST		NATCHEZ	MS	39121	
	ATTN: Sharonda Williams - City					
City of New Orleans	Attorney	1300 Perdido St.	New Orleans	LA	70112	
City of New Orleans	1300 Perdido St.		New Orleans	LA	70112	
CITY OF NEW ORLEANS	1300 PERDIDO STREET	RM.7E05	NEW ORLEANS	LA	70112	
CITY OF NEW ORLEANS	BUREAU OF THE TREASURY	1300 PERDIDO ST. RM 1W15	NEW ORLEANS	LA	70112	
CITY OF NEW ORLEANS	BUREAU OF REVENUE	PO BOX 61840	NEW ORLEANS	LA	70161-1840	
CITY OF NEW ORLEANS (FIRE						
PREVENTION)	FIRE PREVENTION BILLING	317 DECATUR STREET	NEW ORLEANS	LA	70130	
·	Attn: Thomas J. Fromme, City					
City of Newport	Manager	998 Monmouth Street	Newport	KY	41071	
CITY OF NEWPORT	P.O. BOX 1090		NEWPORT	KY	41071-1090	
CITY OF NEWPORT NEWS, VA						
(TREASURER)	MARTY G. EUBANK, TREASURER	PO BOX 975	NEWPORT NEWS	VA	23607-0975	
	Attn: Donna Lisa Dowhite-Owens,					
City of Niagra Falls	СРМ	City Administrator and	Niagara Falls	NY	14302-0069	
CITY OF NORFOLK	P.O. BOX 2260		NORFOLK	VA	23514	
CITY OF NORFOLK (PP TAX)	TREASURER	PO BOX 3215	NORFOLK	VA	23514-3215	
City of North Charleston	Attn: Brady Hair, City Attorney	City Hall	North Charleston	SC	29406	
CITY OF NORTH MIAMI BEACH	CITY OF NORTH MIAMI BEACH	17011 NE 19TH AVENUE	NORTH MIAMI BEACH	FL	33162	
CITY OF NORTH MIAMI BEACH	P.O. BOX 600427		NORTH MIAMI BEACH	FL	33160-0427	
CITY OF NORTHWOODS	4600 OAKRIDGE BLVD.		NORTHWOODS	МО	63121	
City of Norwich	Attn: Betsy M. Barrett - City Clerk	City Hall, Room 215	Norwich	СТ	06360	

Page 39 of 176

Name	Address 1	Address 2	City	State	PostalCode	Country
CITY OF NORWICH	ATTN: TOWN CLERK	100 BROADWAY	NORWICH	СТ	06360	
CITY OF OAK PARK	Erik Tungate - City Manager	City of Oak Park	Oak Park	MI	48237	
CITY OF OAK PARK	14000 OAK PARK BLVD.		OAK PARK	MI	48237	
City of Orangeburg	Attn: James Walsh - City Attorney	949 Middleton Street	Orangeburg	SC	29115	
CITY OF ORANGEBURG	PO BOX 1183		ORANGEBURG	SC	29116-1183	
	ATTN: Mayanne Downs - City					
City of Orlando	Attorney	400 S. Orange Avenue	Orlando	FL	32802	
CITY OF PASADENA	P.O. BOX 672	-	PASADENA	TX	77501	
CITY OF PETERSBURG	TREASURER-LICENSE	135 N. UNION STREET	PETERSBURG	VA	23803	
	Attn: Wallace B. Hunter - City					
City of Phenix City	Manager	601 12th Street	Phenix City	AL	36867	
CITY OF PHENIX CITY	601 12TH STREET		PHENIX CITY	AL	36867	
City of Philadelphia	Attn: Shelley R. Smith - City Solicitor	1515 Arch Street	Philadelphia	PA	19102	
CITY OF PHILADELPHIA	DEPT. OF LICENSES & INSPECTIONS	1401 JOHN F. KENNEDY BLVD.	PHILADELPHIA	PA	19102	
CITY OF PHILADELPHIA	1401 JOHN F. KENNEDY BLVD		PHILADELPHIA	PA	19107	
CITY OF PHILADELPHIA	P.O. BOX 41496		PHILADELPHIA	PA	19101-1496	
CITY OF PHILADELPHIA CODE VIOL	DEPARTMENT OF FINANCE	P.O. BOX 56318	PHILADELPHIA	PA	19130-6318	
	Attn: Althea Hadden-Scott, City					
City of Pine Bluff	Attorney	200 East 8th Ave.	Pine Bluff	AR	71601	
CITY OF PINE BLUFF	CITY COLLECTOR	200 E. 8TH. AVE.	PINE BLUFF	AR	71601	
CITY OF PONTIAC (BUS LIC)	DEPT OF BUILDING & PLANNING	47450 WOODWARD AVENUE	PONTIAC	MI	48341	
CITY OF PONTIAC (DETROIT W/H)	WITHHOLDING PMTS	PO BOX 530	EATON RAPIDS	MI	48827-0530	
CITY OF PONTIAC (PROP TAX)	DEPT 77639	PO BOX 77000	DETROIT	MI	48277-2000	
	Attn: George M. Wilson, City					
City of Portsmouth	Attorney	801 Crawford St.	Portsmouth	VA	23704	
CITY OF PORTSMOUTH	COMMISSIONER OF REVENUE	801 CRAWFORD ST.	PORTSMOUTH	VA	23704	
City of Raleigh	Thomas A McCormick	1 Exchange Plaza #1020	Raleigh	NC	27601	
CITY OF RALEIGH	222 W. HARGETT ST		RALEIGH	NC	27601	
CITY OF RALEIGH	PO BOX 71081		CHARLOTTE	NC	28272-1081	
CITY OF RALEIGH, NC	PRIVILEGE LICENSE TAX	PO BOX 590	RALEIGH	NC	27602	
City of Redford	Attn: Garth Christie, City Clerk	15145 Beech Daly	Redford	MI	48239	
City of Revere	Attn: Ashley E. Melnik, City Clerk	281 Broadway	Revere	MA	02151	
CITY OF REVERE	281 BROADWAY		REVERE	MA	02151	

Page 40 of 176

Name	Address 1	Address 2	City	State	PostalCode	Country
City of Richmond	Allen L. Jackson	900 E. Broad St.	Richmond	VA	23219	
CITY OF RICHMOND	900 E. BROAD ST., SUITE 300		RICHMOND	VA	23219	
CITY OF RICHMOND	DEPARTMENT OF PUBLIC UTILITIES	P.O. Box 26060	RICHMOND	VA	23274-0001	
CITY OF RICHMOND COLLECTNS DIV	RM 102, CITY HALL	P.O. BOX 26505	RICHMOND	VA	23261	
City of Richmond Heights	Attn: R. Todd Hunt, Director of Law	C/O Walter & Haverfield LLP	Cleveland	ОН	44114	
CITY OF RICHMOND- PUBLIC						
UTILITIES	900 E BROAD ST RM 110		RICHMOND	VA	23219	
CITY OF RICHMOND- PUBLIC						
UTILITIES	PO BOX 26060		RICHMOND	VA	23274-0001	
CITY OF RIVERDALE	971 WILSON RD.		RIVERDALE	GA	30082	
CITY OF RIVERDALE	6690 CHURCH ST		RIVERDALE	GA	30274-2333	
City of Rocky Mount	Attn: Pamela O. Casey	331 S. Franklin St.	Rocky Mount	NC	27802	
CITY OF ROCKY MOUNT	PO BOX 1180		ROCKY MOUNT	NC	27802-1180	
CITY OF ROCKY MOUNT	331 S. FRANKLIN ST.		ROCKY MOUNT	NC	27802	
CITY OF ROOSEVELT PARK	900 OAK RIDGE RD		ROOSEVELT PARK	MI	49441	
CITY OF ROSEVILLE	Scott Adkins - City Manager	29777 Gratiot Avenue	ROSEVILLE	MI	48066	
CITY OF ROSEVILLE	29777 GRATIOT	P.O. BOX 290	ROSEVILLE	MI	48066	
CITY OF ROSEVILLE WATER						
DEPARTMENT	29777 GRATIOT AVENUE		ROSEVILLE	MI	48066	
CITY OF ROSEVILLE WATER						
DEPARTMENT	29777 GRATIOT AVENUE	P.O. BOX 290	ROSEVILLE	MI	48066	
	Attn: Raymond C. Liggins - Village					
City of Royal Palm Beach	Manager	1050 Royal Palm Beach Blvd.	West Palm Beach	FL	33401	
City of Ruston	Attn: City Attorney	401 N. Trenton	Ruston	LA	71270	
CITY OF RUSTON	701 E. TENNESSEE AVE		RUSTON	LA	71270	
CITY OF RUSTON	PO BOX 307		RUSTON	LA	71273	
CITY OF SAGINAW	TREASURER'S OFFICE	1315 S WASHINGTON AVENUE	SAGINAW	MI	48601	
CITY OF SAGINAW (W/H)	TREASURER	CITY HALL	SAGINAW	MI	48601	
City of San Antonio	Robert F. Greenblum	100 Military Plaza	San Antonio	TX	78283	
CITY OF SAVANNAH	REVENUE DEPARTMENT	PO BOX 1228	SAVANNAH	GA	31402-1228	
CITY OF SAVANNAH- UTILITY						
SERVICES	132 E. BROUGHTON ST		SAVANNAH	GA	31401	
CITY OF SAVANNAH- UTILITY						
SERVICES	PO BOX 1968		SAVANNAH	GA	31402-1968	

Name	Address 1	Address 2	City	State	PostalCode	Country
City of Scranton	Attn: Jason A. Shrive, City Solicitor	340 North Washington Ave	Scranton	PA	18503	
	Attn: Chip Fletcher, County					
City of Seffner	Attorney	601 E. Kennedy Blvd.	Tampa	FL	33602	
CITY OF SELMA	PO BOX 450		SELMA	AL	36702-0450	
City of Selma City Attorney's	Attn: Jimmy L Nunn	222 Broad Street	Selma	AL	36702	
CITY OF SHREVEPORT	DEPT OF WATER & SEWERAGE	505 Travis St.	SHREVEPORT	LA	71153-0065	
CITY OF SHREVEPORT	DEPT OF WATER & SEWERAGE	PO Box 30065	SHREVEPORT	LA	71153-0065	
CITY OF SHREVEPORT (LIC 30168)	P.O. BOX 30168		SHREVEPORT	LA	71130-0168	
CITY OF SHREVEPORT(PP TAX)	REVENUE DIVISION	P.O.BOX 30040	SHREVEPORT	LA	71130-0040	
	ATTN: Thomas G. Lobe - Law					
City of Solon	Director	The Rockefeller Building, Ste. 1300	Cleveland	ОН	44113	
,		46th Judicial District Court Building				
CITY OF SOUTHFIELD	Sue Ward-Witkowski, City Attorney	– 1st Floor	Southfield	MI	48037	
CITY OF SOUTHFIELD	TREASURER	PO BOX 369	SOUTHFIELD	MI	48037-0369	
	46TH JUDICIAL DISTRICT COURT					
CITY OF SOUTHFIELD- WATER DEPT	BUILDING – 1ST FLOOR	26000 EVERGREEN ROAD	SOUTHFIELD	MI	40837	
CITY OF SOUTHFIELD- WATER DEPT	PO BOX 33835		DETROIT	MI	48232-0835	
CITY OF SPARTANBURG	PO BOX 1749		SPARTANBURG	sc	29304	
	ATTN: Donna Edmundson - City					
City of Spring	Attorney	City of Houston Legal Department	Houston	TX	77002	
City of St. Louis	Attn: City Attorney	1114 Market , Room 401	St Louis	МО	63101	
City of St. Louis	Attn: Jennifer M. Joyce	1114 Market St.	St. Louis	МО	63101	
CITY OF ST. LOUIS	COLLECTOR OF REVENUE	412 CITY HALL- EARNINGS TAX	ST. LOUIS	МО	63103	
CITY OF ST. LOUIS	COLLECTOR OF REVENUE	412 CITY HALL- P/R EXPENSE TAX	ST. LOUIS	мо	63103	
CITY OF ST. LOUIS- PP	COLLECTOR OF REVENUE	PO BOX 66877	ST. LOUIS	МО	63166-6877	
CITY OF ST. LOUIS(LICENSE)	LICENSE COLLECTOR	PO BOX 78158	ST. LOUIS	МО	63178-8158	
, ,						
City of Sterling Heights	Attn: Jeff Bahorski, City Attorney	12900 Hall Road	Sterling Heights	MI	48313	
,	Attn: MARK D. VANDERPOOL, CITY		Ü			
City of Sterling Heights	MANAGER	40555 Utica Road	Sterling Heights	MI	48311-8009	
CITY OF STERLING HEIGHTS	OFFICE OF BUILDING SERVICES	40555 UTICA ROAD	STERLING HEIGHTS	MI	48313	
CITY OF SUMTER	Deron L. McCormick - City Manager	PO Box 1449	Sumter	sc	29150	
CITY OF SUMTER	LICENSE DIVISION	PO BOX 1449	SUMTER	SC	29151	
CITY OF SUMTER	21 N. MAIN STREET		SUMTER	SC	29150	

Page 42 of 176

Name	Address 1	Address 2	City	State	PostalCode	Country
CITY OF SUMTER	PO BOX 310		SUMTER	SC	29151-0310	
City of Syracuse	Attn: John P. Copanas - City Clerk	231 City Hall	Syracuse	NY	13202	
CITY OF TALLADEGA	REVENUE DEPT.	PO BOX 498	TALLADEGA	AL	35161	
CITY OF TALLAHASSEE	City Attorney Lewis E. Shelley	300 South Adams Street, 2nd Floor	Tallahassee	FL	32301	
CITY OF TALLAHASSEE	REVENUE DIVISION BOX A4	300 S. ADAMS STREET	TALLAHASSEE	FL	32301	
CITY OF TALLAHASSEE	435 N. MACOMB STREET		TALLAHASSEE	FL	32301	
CITY OF TALLAHASSEE	435 N. MACOMB STREET, Relay Box		TALLAHASSEE	FL	32301	
CITY OF TAMPA	BUSINESS TAX DIVISION	P.O. BOX 2200	TAMPA	FL	33601-2200	
City of TAYLORS	Attn: Mark W. Tollison, Attorney	301 University Ridge,	Greenville	SC	29601	
CITY OF THOMASVILLE	PO BOX 1397		THOMASVILLE	GA	31799	
City of Tinley Park	Attn: Patrick E. Rea, Village Clerk	16250 S. Oak Park Ave.	Tinley Park	IL	60477	
CITY OF TOLEDO	TAX COMMISSIONER	1 GOVERNMENT CENTER, STE 2070	TOLEDO	ОН	43604-2280	
CITY OF TUSCALOOSA	REVENUE DEPARTMENT	PO BOX 2089	TUSCALOOSA	AL	35403-2089	
CITY OF TUSCALOOSA WATER DEPT	2201 UNIVERSITY BLVD		Tuscaloosa	AL	35401	
CITY OF TUSCALOOSA WATER DEPT	DEPT # 2533	PO BOX 2153	BIRMINGHAM	AL	35287-2533	
City of Tyler	Deborah Pullum	P.O. Box 2039	Tyler	TX	75710	
CITY OF TYLER (ALARM)	ALARM COORDINATOR	711 W GERGUSON ST	TYLER	TX	75702	
CITY OF TYLER- WATER UTILITIES	511 W LOCUST ST		TYLER	TX	75702	
CITY OF TYLER- WATER UTILITIES	PO BOX 336		TYLER	TX	75710-0336	
CITY OF UNION CITY	5047 UNION ST		UNION CITY	GA	30291-1497	
CITY OF UNION CITY	5047 UNION STREET		UNION CITY	GA	30291-1497	
City of Valdosta	Attn: Teresa S. Bolden, City Clerk	216 E Central Ave	Valdosta	GA	31603	
CITY OF VALDOSTA	BUSINESS LICENSE	P.O. BOX 1125	VALDOSTA	GA	31603-1125	
CITY OF VALDOSTA	216 E. CENTRAL AVE.		VALDOSTA	GA	331603	
CITY OF VALDOSTA	PO BOX 1125		VALDOSTA	GA	31603-1125	
City of Vicksburg	Attn: Walter Osborne, City Clerk	City Clerk	Vicksburg	MS	39180	
CITY OF VICKSBURG	TAX COLLECTOR	PO BOX 150	VICKSBURG	MS	39181	
CITY OF VIRGINIA BEACH	2401 COURTHOUSE DRIVE		VIRGINIA BEACH	VA	23456	

Page 43 of 176

Name	Address 1	Address 2	City	State	PostalCode	Country
CITY OF WARNER ROBBINS	202 N. Davis Drive	PO Box 717	Warner Robins	GA	31093-3348	•
CITY OF WARNER ROBBINS	700 WATSON BLVD		WARNER ROBINS	GA	31093	
CITY OF WARNER ROBINS	TAX OFFICE	202 N DAVIS DR PMB718	WARNER ROBINS	GA	31093	
CITY OF WARNER ROBINS- UTILITY	202 N DAVIS DR PMB 717		WARNER ROBINS	GA	31093-3348	
	DIV OF BUILDING & SAFETY					
CITY OF WARREN	ENGINEERING	ONE CITY SQUARE SUITE 305	WARREN	MI	48093	
	Attn: Thomas A. Cloud, City					
City of Wauchula	Attorney	301 East Pine Street	Orlando	FL	32801	
City of West Allis	Attn: Scott Post, City Attorney	Attorney's Office	West Allis	WI	53214	
CITY OF WEST ALLIS	7525 W. GREENFIELD AVENUE		WEST ALLIS	WI	53214	
CITY OF WEST ALLIS	7524 WEST GREENFIELD AVENUE		WEST ALLIS	WI	53214-4648	
CITY OF WEST MEMPHIS	BOX 1728		WEST MEMPHIS	AR	72303	
City of West Palm Beach	Attn: City Attorney	401 Clematis Street, Fifth Floor	West Palm Beach	FL	33401	
CITY OF WEST PALM BEACH- ALARM						
RENEWALS	POLICE ALARM SERVICES	PO BOX 22088	TAMPA	FL	33622	
CITY OF WEST PALM BEACH FIRE						
RESCUE	PO BOX 3366		WEST PALM BEACH	FL	33402	
CITY OF WEST PALM BEACH-						
LICENSING	LICENSING DEPARTMENT	P.O. BOX 31627	TAMPA	FL	33631-3627	
CITY OF WEST PALM BEACH-FALSE						
ALARM	FALSE BURGLAR ALARM	P.O. BOX 3366	WEST PALM BEACH	FL	33402	
	Attn: William Reichelt, City	J.Edward Christian Municipal Office				
City of West Springfield	Attorney	Building	West Springfield	MA	01089	
CITY OF WESTLAND	James G. Fausone City Attorney	City Clerk Department	Westland	MI	48185	
CITY OF WESTLAND	PO BOX 85040		WESTLAND	MI	48185	
City of Whitehall	Attn: Michael Shannon	360 S. Yearling Road	Whitehall	ОН	43213	
CITY OF WHITEHALL	INCOME TAX DIVISION	360 S YEARLING ROAD	WHITEHALL	ОН	43213	
CITY OF WILSON	PRIVILEGE LICENSE DEPT	PO BOX 10	WILSON	NC	27894	
CITY OF WILSON- COLLECTION						
DIVISION	112 GOLDSBORO ST E		WILSON	NC	27893	
CITY OF WILSON- COLLECTION						
DIVISION	PO BOX 2407		WILSON	NC	27894-2407	
City of Winston-Salem	Attn: Angela Carmon, City Attorney	101 N. Main Street	Winston-Salem	NC	27101	
CITY OF WINSTON-SALEM	PO BOX 2756		WINSTON-SALEM	NC	27102-2756	
CITY OF WINSTON-SALEM REVENUE						
DIV	101 N. MAIN STREET		WINSTON-SALEM	NC	27101	

Page 44 of 176

Name	Address 1	Address 2	City	State	PostalCode	Country
CITY OF WINSTON-SALEM REVENUE						,
DIV	PO BOX 580055		CHARLOTTE	NC	28258-0055	
City of Worcester	Attn: City Clerk	455 Main Street	Worcester	MA	01608	
CITY OF WORCESTER	CITY HALL, ROOM 206	455 MAIN STREET	WORCESTER	MA	01608	
City of WYOMING	Attn: Curtis L. Holt, City Manager	1155 28th St SW	Wyoming	MI	49509	
CITY OF WYOMING	CITY CLERK'S OFFICE	1155 28TH STREET SW	WYOMING	MI	49509-0905	
CITY OF WYOMING	1155 28TH ST SW		WYOMING	MI	49509	
CITY OF WYOMING	PO BOX 630422		CINCINNATI	ОН	45263-0422	
City of Youngstown	Attn: Martin Hume, Law Director	26 South Phelps Street	Youngstown	ОН	44503	
City of Ypsilanti	Attn: John M. Barr, City Attorney	105 Pearl Street	Ypsilanti	MI	48197	
CITY PAPER COMPANY	PO BOX 1968		BIRMINGHAM	AL	35201	
CITY TREASURER	Kevin Brown	135 North Union Street	Petersburg	VA	23803	
	John T. Atkinson, City Treasurer and					
CITY TREASURER	Mark D. Stiles- City Attorney	2401 Courthouse Dr.	Virginia Beach	VA	23456	
CITY TREASURER, PETERSBURG	135 N. UNION ST		PETERSBURG	VA	23803	
CITY TREASURER, PETERSBURG	PO BOX 1271		PETERSBURG	VA	23804	
CITY WEST APPAREL	1385 BROADWAY, SUITE 1504		NEW YORK	NY	10018	
CLAIBORNE, ASHLEY	REDACT		REDACT	REDACT	REDACT	
CLARK, DOMINIQUE	REDACT		REDACT	REDACT	REDACT	
CLARK, FRANKIE	REDACT		REDACT	REDACT	REDACT	
CLARK, JAMES	REDACT	REDACT	REDACT	REDACT	REDACT	
CLARK, JEANNE	REDACT		REDACT	REDACT	REDACT	
CLARK, LUCY	REDACT		REDACT	REDACT	REDACT	
CLARK, PEGGY	REDACT		REDACT	REDACT	REDACT	
CLARK-HANNA, LATASHA	REDACT		REDACT	REDACT	REDACT	
CLARKSDALE PUBLIC UTILITIES	416 3RD ST		CLARKSDALE	MS	38614	
CLARKSDALE PUBLIC UTILITIES	PO BOX 70		CLARKSDALE	MS	38614-0070	
CLAVEL, ANGEL	REDACT		REDACT	REDACT	REDACT	
CLAY, CHARLISIA	REDACT		REDACT	REDACT	REDACT	
CLAY, SHEILA	REDACT		REDACT	REDACT	REDACT	
CLAY, VONNE	REDACT		REDACT	REDACT	REDACT	
CLAYBOURNE, TAIRRA	REDACT		REDACT	REDACT	REDACT	
CLAYTON COUNTY BOARD OF						
COMMISSIONERS	121 S. MCDONOUGH STREET	ANNEX 2	JONESBORO	GA	30236	

Page 45 of 176

Name	Address 1	Address 2	City	State	PostalCode	Country
CLAYTON COUNTY FIRE &						
EMERGENCY SERVICES	P.O. BOX 742874		RIVERDALE	GA	30274	
CLAYTON COUNTY TAX	ADMINISTRATION ANNEX 3, 2ND					
COMMISSIONER	FLOOR	121 S. MCDONOUGH ST.	JONESBORO	GA	30236-3694	
CLAYTON, CHAUNCY	REDACT		REDACT	REDACT	REDACT	
CLEAR CHANNEL BROADCASTING						
INC.	5529 COLLECTIONS CENTER DRIVE		CHICAGO	IL	60693	
CLEAVES, RITA	REDACT		REDACT	REDACT	REDACT	
CLEMENT COMMUNICATIONS, INC	PO BOX 6017		CAROL STREAM	IL	60197-6017	
CLEMMONS, LISA	REDACT		REDACT	REDACT	REDACT	
CLEMONS, CHARITY	REDACT		REDACT	REDACT	REDACT	
CLEMONS, RONDRIC	REDACT	REDACT	REDACT	REDACT	REDACT	
CLEMONS, SHAQUIEL	REDACT		REDACT	REDACT	REDACT	
CLEO APPAREL	777 10TH STE, STE113		LOS ANGELES	CA	90021	
CLEO APPRAEL	1015 S. CROCKER ST.	#P1, P-2	LOS ANGELES	CA	90021	
CLERK OF CIRCUIT COURT	2425 NIMMA PKWY.	BUILDING 10, 3RD FLOOR	VIRGINIA BEACH	VA	23452	
CLERK OF COURT	RICHMOND CIRCUIT COURT	400 N. NINTH STREET	RICHMOND	VA	23219	
CLERK OF THE CIRCUIT COURT	PRINCE GEORGE'S CO-LICENSE	14735 MAIN STREET	UPPER MARLBORO	MD	20772-9987	
CLEVELAND AVENUE CROSSING, LLC	106 E. 8th Avenue		Rome	GA	30161	
CLEVELAND AVENUE CROSSING, LLC.	106 E. 8TH AVENUE		ROME	GA	30161	
CLEVELAND PORT CITY GLASS &						
PAINT	PO BOX 419		CLEVELAND	MS	38732	
CLEVELAND, TERRELL	REDACT		REDACT	REDACT	REDACT	
CLIATT, DE-ANNA	REDACT	REDACT	REDACT	REDACT	REDACT	
CLICK MODEL MANAGEMENT, INC.	129 WEST 27TH STREET	12TH FLOOR	NEW YORK	NY	10001	
CLIMATE CONTROL (TX)	813 WOODLAND HILLS DRIVE		TYLER	TX	75701	
CLINTON TOWNSHIP TREASURER	40700 ROMEO PLANK RD		CHARTER TWP OF CLINTON	МІ	48038	
CLINTON TOWNSHIP TREASURER	WATER & SEWER DIVISION	PO BOX 553160	DETROIT	МІ	48255-3160	
CLOTHINGOLOGY INC	214 W 39TH ST, 5TH FLR		NEW YORK	NY	10018	
CLOUD, BRYANNA	REDACT		REDACT	REDACT	REDACT	
CLOVER LOCKSMITHS	PO BOX 577		TINLEY PARK	IL	60477	
CMI AIR CONDITIONING &	-					1
ELECTRICAL	202 OLD DIXIE HIGHWAY		LAKE PARK	FL	33403	
CMQ FLOOR COVERING	2524 FORD ROAD		BRISTOL	PA	19007-6802	1

Page 46 of 176

Name	Address 1	Address 2	City	State	PostalCode	Country
CMS Mechanical Services, LLC	Nextech	445 West Drive, Ste. 101	Melbourne	FL	32904	
CMS MECHANICAL SERVICES, LLC	NEXTECH	445 WEST DRIVE	MELBOURNE	FL	32904	
COAHOMA COUNTY	TAX ASSESSOR COLLECTOR	P.O. BOX 219	CLARKSDALE	MS	38614	
COALFIRE SYSTEMS, INC	361 CENTENNIAL PARKWAY	SUITE 150	LOUISVILLE	СО	80027	
COAST SAFE & LOCK CO., INC.	P.O. BOX 66257		MOBILE	AL	36606	
COAST TO COAST LOGISTICS, INC.	1411 LANTANA DRIVE		BIRMINGHAM	AL	35226	
COBB COUNTY	BUSINESS LICENSE DIVISION	PO BOX 649	MARIETTA	GA	30061	
COBB COUNTY TAX COMMISSIONER	PO BOX 100127		MARIETTA	GA	30061-7027	
COBB EMC	1000 EMC PARKWAY		MARIETTA	GA	30060	
COBB EMC	P.O. BOX 369		MARIETTA	GA	30061-0369	
COBB PLUMBING CO., INC	2286 PODESTA COVE		MEMPHIS	TN	38134	
COBB, CLAYDOUS	REDACT		REDACT	REDACT	REDACT	
COBB, JOHNATHON	REDACT		REDACT	REDACT	REDACT	
COBB, JONATHAN	REDACT		REDACT	REDACT	REDACT	
COBB, SCOTTIE	REDACT		REDACT	REDACT	REDACT	
COBBINS, VANESSA	REDACT		REDACT	REDACT	REDACT	
COBHAM, VENESSA	REDACT		REDACT	REDACT	REDACT	
COCKRELL, CYNTHIA	REDACT		REDACT	REDACT	REDACT	
CODY, JAMARCUS	REDACT		REDACT	REDACT	REDACT	
CODY, NIESHA	REDACT		REDACT	REDACT	REDACT	
COEFIELD, JARVELL	REDACT		REDACT	REDACT	REDACT	
COKER ELECTRIC PINE BUFF	6712 DOLLARWAY ROAD		PINE BLUFF	AR	71602	
COLBERT, CAITLIN	REDACT	REDACT	REDACT	REDACT	REDACT	
COLE MT RICHMOND VA, LLC	C/O COLE REAL ESTATE INVESTMENTS	2325 EAST CAMELBACK RD	PHOENIX	AZ	85016	
COLE MT RICHMOND VA, LLC	c/o Cole Real Estate Investments	2325 East Camelback Road	Phoenix	AZ	85016	
COLE, DANIELLE	REDACT	REDACT	REDACT	REDACT	REDACT	
COLE, JONELL	REDACT		REDACT	REDACT	REDACT	
COLE, KAREN	REDACT		REDACT	REDACT	REDACT	
COLEMAN, ALEXANDRIA	REDACT		REDACT	REDACT	REDACT	
COLEMAN, SHANNON	REDACT		REDACT	REDACT	REDACT	
COLEMAN, SHARHONDA	REDACT		REDACT	REDACT	REDACT	
COLEMAN, STEPHEN	REDACT		REDACT	REDACT	REDACT	
COLEMAN, WILHELMINA	REDACT		REDACT	REDACT	REDACT	
COLLIER, CIARA	REDACT		REDACT	REDACT	REDACT	

Page 47 of 176

Name	Address 1	Address 2	City	State	PostalCode	Country
COLLINS, ASHLEY	REDACT		REDACT	REDACT	REDACT	
COLLINS, DOLLIE	REDACT		REDACT	REDACT	REDACT	
COLLINS, JOACHIM	REDACT		REDACT	REDACT	REDACT	
COLLINS, JOSEPH	REDACT		REDACT	REDACT	REDACT	
COLLINS, KAIRRA	REDACT	REDACT	REDACT	REDACT	REDACT	
COLLINS, MELISSA	REDACT	REDACT	REDACT	REDACT	REDACT	
COLLINS, NANCY	REDACT		REDACT	REDACT	REDACT	
COLLINS, TERRESA	REDACT	REDACT	REDACT	REDACT	REDACT	
COLOMBINO HEADWEAR	61 WILLETT STREET	BLDG J- 2ND FLOOR	PASSAIC	NJ	07055	
COLONIAL LIFE (PROCESSING						
CENTER)	PROCESSING CENTER	PO BOX 1365	COLUMBIA	SC	29202-1365	
COLONIAL WEBB CONTRACTORS	2820 ACKLEY AVENUE		RICHMOND	VA	23228	
COLQUITT, TRACEY	REDACT	REDACT	REDACT	REDACT	REDACT	
COLUMBIA GAS OF MASSACHUSETTS	2025 ROOSEVELT AVE		SPRINGFIELD	MA	01104-0000	
COLUMBIA GAS OF MASSACHUSETTS	P.O. BOX 742514		CINCINNATI	ОН	45274-2514	
COLUMBIA GAS OF OHIO	290 W NATIONWIDE BLVD #114		COLUMBUS	ОН	43215	
COLUMBIA GAS OF OHIO	P.O. BOX 742510		CINCINNATI	ОН	45274-2510	
COLUMBIA GAS OF VIRGINIA	1809 COYOTE DR		CHESTER	VA	23836	
COLUMBIA GAS OF VIRGINIA	PO BOX 742529		CINCINNATI	ОН	45274-2529	
COLUMBUS- CITY TREASURER	90 WEST BROAD ST		COLUMBUS	ОН	43215	
COLUMBUS- CITY TREASURER	DEPARTMENT OF PUBLIC UTILITIES	910 DUBLIN ROAD	COLUMBUS	ОН	43215	
COLUMBUS- CITY TREASURER	POWER SEWER & WATER SERVICES		COLUMBUS	ОН	43218-2882	
COLUMBUS- CITY TREASURER	POWER SEWER & WATER SERVICES	PO BOX 182882	COLUMBUS	ОН	43218-2882	
COLUMBUS- CITY TREASURER	SEWER & WATER SERVICES	PO BOX 182882	COLUMBUS	ОН	43218-2882	
COLUMBUS CONSOLIDATED						
GOVERNMENT	REVENUE DIVISION	PO BOX 1397	COLUMBUS	GA	31902-1397	
COLUMBUS GAS	290 W Nationwide Blvd		Columbus	ОН	43215	
COLUMBUS GAS	P.O. BOX 742510		CINCINNATI	ОН	45274-2510	
COLUMBUS LIGHT & WATER DEPT	420 4TH AVE S.		COLUMBUS	MS	39701	
COLUMBUS LIGHT & WATER DEPT	PO BOX 22806		JACKSON	MS	39225-2806	
COLUMBUS SIGN CO.	P.O. BOX 09746		COLUMBUS OH		43209-9746	
COLUMBUS WATER WORKS	1421 VETERANS PKWY		COLUMBUS	GA	31901	
COLUMBUS WATER WORKS	PO BOX 1600		COLUMBUS	GA	31902-1600	

Page 48 of 176

Name	Address 1	Address 2	City	State	PostalCode	Country
COMED	3 LINCOLN CENTER		OAKBROOK TERRACE	IL	60181	
COMED	P.O. BOX 6111		CAROL STREAM	IL	60197-6111	
COMERICA BANK	PO Box 75000		Detroit	MI	48275-8158	
COMFORT TECH	2511 WARWICK CT.		BAY CITY	MI	48706	
COMMANDER SIGNS, INC	201 50TH STREET		FAIRFIELD	AL	35064	
COMMERCIAL DOOR & GATE	4153 SW 47TH AVENUE	BAY # 102	DAVIE	FL	33314	
COMMONFIELDS OF CAHOKIA	PUBLIC WATER DISTRICT	2525 MOUSETTE LANE	CAHOKIA	IL	62206	
COMMONWEALTH COMMERCIAL						
PARTNERS, INC.	LANGLEY SQUARE	PO BOX 71150	RICHMOND	VA	23255	
Commonwealth of Kentucky	501 High Street		Frankfort	KY	40601	
Commonwealth of Massachusetts	Dept of Revenue	PO Box 7010	Boston	MA	02204	
COMMONWEALTH OF						
PENNSYLVANIA	DEPT. OF STATE	P.O. BOX 8722	HARRISBURG	PA	17105-8722	
COMMUNITY DEV CORP OF KANSAS						
CITY	C/O UNIVERSAL MGMT INC	PO BOX 172035	KANSAS CITY	KS	66117	
COMMUNITY DEVELOPMENT CORP.						
OF KANSAS CITY	Donald Maxwell	2420 East Linwood Boulevard	Kansas City	MI	64109	
COMPANION LIFE	P.O. BOX 100102		COLUMBIA	SC	29202-3102	
COMPASS BANK	PO BOX 830927		BIRMINGHAM	AL	35283-0927	
COMPBENEFITS (MO)	PO BOX 219051		KANSAS CITY	MO	64121-9051	
COMPTON, JAMESE	REDACT		REDACT	REDACT	REDACT	
Comptroller of Maryland	Revenue Administration Division	Taxpayer Service Section	Annapolis	MD	21411-0001	
COMPUTER ELECTRIC, INC.	P.O. BOX 660488		MIAMI	FL	33266	
COMSUMERS ENERGY	P.O. BOX 740309		CINCINNATI	ОН	45274-0309	
CONLEY, ALLISON	REDACT		REDACT	REDACT	REDACT	
Connecticut Connecticut Department						
of Revenue Services	25 Sigourney Street		Hartford	СТ	6106	
CONNECTICUT SECRETARY OF STATE	P.O. BOX 150470		HARTFORD	СТ	06115	
CONNECTION 18	107 NEW YORK AVE		JERSEY CITY	NJ	07307-1769	
CONNIE LAWSON	SECRETARY OF STATE	302 W. WASHINGTON STREET	INDIANAPOLIS	IN	46204	
CONNOR, IEISHA	REDACT		REDACT	REDACT	REDACT	
CONNORS, ULMEATER	REDACT	REDACT	REDACT	REDACT	REDACT	
CONSTANCE L. MILLER	501 E. ARIZONA AVE.		RUSTON	LA	71270	

Name	Address 1	Address 2	City	State	PostalCode	Country
CONSTANGY, BROOKS & SMITH, LLC.	P.O. BOX 102476		ATLANTA	GA	30368-0476	
CONSTRUCTION CODE						
ENFORCEMENT	6465 MULLINS STATION		MEMPHIS	TN	38134	
CONSUMERS ENERGY	1 ENERGY PLAZA DR		JACKSON	MI	49201	
CONSUMERS ENERGY	P.O. BOX 740309		CINCINNATI	OH	45274-0309	
	C/O BAI CONSUMERS SQUARE					
CONSUMERS SQUARE WEST LLC	WEST LLC	PO BOX 713548	CINCINNATI	ОН	45271-3548	
CONTAINER SERVICE CORP	PO BOX 70		RINGGOLD	GA	30736	
COOK, KELLEY	REDACT	REDACT	REDACT	REDACT	REDACT	
COOK, NYOKA	REDACT		REDACT	REDACT	REDACT	
COOKS, UNSER	REDACT		REDACT	REDACT	REDACT	
COOKS, WILLIE	REDACT		REDACT	REDACT	REDACT	
COOLRITE, INC	AIR CONDITIONING & HEATING	425 HULMEVILLE AVENUE	PENNDEL	PA	19047	
COOPER, AALIYAH	REDACT		REDACT	REDACT	REDACT	
COOPER, CARLOS	REDACT		REDACT	REDACT	REDACT	
COOPER, DENISE	REDACT		REDACT	REDACT	REDACT	
COOPER, PATRICIA	REDACT		REDACT	REDACT	REDACT	
COOPER, PATRICIA	REDACT		REDACT	REDACT	REDACT	
COOPER, SHERRI	REDACT		REDACT	REDACT	REDACT	
COOPER, THALIA	REDACT		REDACT	REDACT	REDACT	
COPELAND, KEZIAH	REDACT		REDACT	REDACT	REDACT	
COPPEDGE, DAPHINE	REDACT		REDACT	REDACT	REDACT	
CORBIE, MARIKA	REDACT	REDACT	REDACT	REDACT	REDACT	
CORDERO, MONIQUE	REDACT	REDACT	REDACT	REDACT	REDACT	
CORNELL, JASON	REDACT		REDACT	REDACT	REDACT	
CORNELL, LATOYA	REDACT		REDACT	REDACT	REDACT	
CORNELL, THERESA	REDACT		REDACT	REDACT	REDACT	
CORPORATION DIVISION	P.O. BOX 30054		LANSING	MI	48909	
CORREA, DANIGSHA	REDACT		REDACT	REDACT	REDACT	
CORTEZ, SCOTT	REDACT		REDACT	REDACT	REDACT	
COTTEN, TATIANNA	REDACT		REDACT	REDACT	REDACT	
COTTLE, CHERYL	REDACT		REDACT	REDACT	REDACT	
COTTO, ANTONYO	REDACT		REDACT	REDACT	REDACT	
COTTON EXPRESS CORP	WINGS MFG. CORP	15 WILKINSON AVE	JERSEY CITY	NJ	07305	
COUNTRY CLUB PLAZA ASSOCIATES,						
LLC	C/O JSK MGMT CO, LLC	PO BOX 880169	BOCA RATON	FL	33488	

Name	Address 1	Address 2	City	State	PostalCode	Country
COUNTRY CLUB PLAZA ASSOCIATES,						
LLC	C/O JSK MGMT CO, LLC	P O Box 880169	Boca Raton	FL	33488	
County Attorney	Mike O'Connell	Jefferson Hall of Justice	Louisville	KY	40202	
County Attorney	Chip Fletcher	601 E. Kennedy Blvd.	Tampa	FL	33602	
County Clerk	Susan Kaltenbach	111 South Michigan Ave.	Saginaw	MI	48602	
COUNTY OF HENRICO, VA.	DEPT OF FINANCE	PO BOX 90790	HENRICO	VA	23228-0790	
County of Hillsdale	Attn: Marney Kast	Hillsdale County Courthouse	Hillsdale	MI	49242	
COURAGE CLOTHING	1407 BROADWAY		NEW YORK	NY	10018	
COURTNEY & CAMP	P.O. BOX 529		JACKSON	MS	39205	
COVENTRY II DDR FAIRPLAIN, LLC	C/O DEPT 314434-20002-34887	PO BOX 774554	CHICAGO	IL	60677-4005	
COVINGTON ELECTRIC SERVICE	PO BOX 720105		BYRAM	MS	39272	
COX, JACOB	REDACT		REDACT	REDACT	REDACT	
COX, JAMES L.	REDACT		REDACT	REDACT	REDACT	
COX, JASMINE	REDACT		REDACT	REDACT	REDACT	
COX, NETOYA	REDACT		REDACT	REDACT	REDACT	
COYETT, NIEYA	REDACT		REDACT	REDACT	REDACT	
CPS ENERGY	145 NAVARRO ST		SAN ANTONIO	TX	78205-2934	
CPS ENERGY	PO BOX 2678		SAN ANTONIO	TX	78289-0001	
CRAFT, LEEASHA	REDACT		REDACT	REDACT	REDACT	
CRAFTSMAN PRINTING, INC	120 CITATION COURT		BIRMINGHAM	AL	35209	
CRAIG WATERS	203 GADD RD		CHATTANOOGA	TN	37415	
CRANFILL, JOHN	REDACT		REDACT	REDACT	REDACT	
CRAWFORD, ELIZABETH	REDACT		REDACT	REDACT	REDACT	
CRAWFORD, KIMBERLY	REDACT	REDACT	REDACT	REDACT	REDACT	
CRAWFORD, RITA	REDACT		REDACT	REDACT	REDACT	
CRAWLEY, DASHA	REDACT		REDACT	REDACT	REDACT	
CRAWLEY, MALIK	REDACT		REDACT	REDACT	REDACT	
CRAYTON, MICHEL'LE	REDACT		REDACT	REDACT	REDACT	
CRAYTON, MICHEL'LE	REDACT		REDACT	REDACT	REDACT	
CRAZY COOL BUYS	36148 ECORSE ROAD		ROMULUS	MI	48174	
CREATIVE FILING SYSTEMS, INC.	1350 HOLTON LANE		TAKOMA PARK	MD	20912	
CREATIVE LOOKS	1400 BROADWAY ROOM 1201		NEW YORK	NY	10018	
CREDIT ACCEPTANCE CORP	C/O WEBER & OLCESE, PLC	3250 W. BIG BEAVER RD.	TROY	MI	48084	
CRESPO, NELSON	REDACT		REDACT	REDACT	REDACT	
CREST COMMERCIAL REALTY	122 N. JEFFERSON STREET		DAYTON	OH	45402	
CREWS, GEORGE	REDACT		REDACT	REDACT	REDACT	
CRICHTON MOBILE ASSOCIATES	C/O ABC PROPERTIES	PO BOX 315	EMERSON	NJ	07630	

Name	Address 1	Address 2	City	State	PostalCode	Country
CRICHTON MOBILE ASSOCIATES, LLC	C/O ABC PROPERTIES	P O BOX 315	EMERSON	NJ	07630	
CRITTENDEN CO. AR	TAX COLLECTOR	250 PINE ST., STE 2	MARION	AR	72364	
CRITTENDEN, LAMONNA	REDACT		REDACT	REDACT	REDACT	
CROFT, AJANE	REDACT		REDACT	REDACT	REDACT	
CRONE & MCEVOY, PLC	5583 Murray Ave	#120	Memphis	TN	38119	
CROSBY ELECTRIC COMPANY, INC	PO BOX 240368		MONTGOMERY	AL	36124	
CROSS COUNTRY PLAZA, LLC	C/O CORO REALTY ADVISORS, LLC	3715 NORTHSIDE PKWY	ATLANTA	GA	30327	
CROSS CREEK PLAZA, INC	238 N MCPHERSON CHURCH RD		FAYETTEVILLE	NC	28303	
CROSSING SHOPPING CENTER, LP	c/o Priority Properties	1045 South Wood Mills Road	Town & Country	МО	63017	
CROSSINGS SHOPPING CENTER, LP	C/O KEY BANK	PO BOX 944509	CLEVELAND	ОН	44194-4509	
,	C/O JOHNSON, PRICE & SPRINKLE,					
CROSSROADS PLAZA, LLC	PA	79 WOODFIN PLACE	ASHEVILLE	NC	28801	
Crossroads Plaza, LLC	CROSSROADS PLAZA/Store No538	PO Box 3476	Gaithersburg	MD	20885	
CROSSROADS PLAZA, LLC	ATTN: Michael D. Green	P.O. Box 3476	Gaithersburg	MD	20885	
CROWLEY, ANNETTE	REDACT		REDACT	REDACT	REDACT	
CROWN COMMERCIAL RE & DVLP,						
LLC	903 S BUTTERNUT CIRCLE		FRANKFORT	IL	60423	
CROWN COMMERCIAL REAL ESTATE						
& DEVELOPMENT, LLC	Musa P. Tadros, Mg Mem	903 S. Butternut Circle	Frankfort	IL	60423	
CROWN OF HEARTS	2731 S. ALAMEDA ST		LOS ANGELES	CA	90058	
CRP II - TAMPA FESTIVAL, LLC	C/O FORGE PROPERTY MGMT	102 W. WHITING STREET	TAMPA	FL	33602	
CRYSTAL K	800 MCGARRY ST	5TH FLOOR	LOS ANGELES	CA	90021	
CSFB 2004-C5 RETAIL 8955, LLC	ATTN: NORTH EASTWOOD S/C	600 WASHINGTON AVE.	ST LOUIS	MO	63101	
CSS	1314 PARK BLVD		CAMDEN	NJ	08103	
CUFF, ELIZABETH	REDACT		REDACT	REDACT	REDACT	
CULBERT, HERRALDO	REDACT		REDACT	REDACT	REDACT	
CULBREATH, BARBARA	REDACT	REDACT	REDACT	REDACT	REDACT	
CUMBERLAND CO TAX COLLECTOR	P.O. BOX 1070		CHARLOTTE	NC	28201-1070	
CUMMINGS, ANNIE	REDACT		REDACT	REDACT	REDACT	
CUMMINGS, SHETARA	REDACT		REDACT	REDACT	REDACT	
CUNNINGHAM, APRYLE	REDACT		REDACT	REDACT	REDACT	

Page 52 of 176

Name	Address 1	Address 2	City	State	PostalCode	Country
CUNNINGHAM, CHERRISH	REDACT		REDACT	REDACT	REDACT	
CUNNINGHAM, VICTORIA	REDACT		REDACT	REDACT	REDACT	
CURELL, JOCELYN	REDACT		REDACT	REDACT	REDACT	
CURRY, CAPRISHA	REDACT	REDACT	REDACT	REDACT	REDACT	
CURRY, LARRY	REDACT	REDACT	REDACT	REDACT	REDACT	
CURTIS C REDING	CHAPTER 13 TRISTEE	P.O. BOX 613108	MEMPHIS	TN	38101-3108	
CURTIS, MARSHA	REDACT		REDACT	REDACT	REDACT	
CUSTOM MECHANICAL, INC	1414 10TH STREET		LAKE PARK	FL	33403	
Cutler Bay	Attn: Mitchell A. Bierman	10720 Caribbean Boulevard	Cutler Bay	FL	33189	
D K ENTERPRISES, LLC	C/O MOORE PROPERTY MGMT, LLC	DO BOY 1429	MONTGOMERY	AL	36102	
D.C. TREASURER	SALES TAX	PO BOX 470	WASHINGTON	DC	20044	
D.C. TREASURER	OFFICE OF TAX AND REVENUE	P.O.BOX 96019	WASHINGTON	DC	20090-6019	+
D.C. TREASURER	DEPT OF CONSUMER &	REGULATORY AFFAIRS	WASHINGTON	DC	20090-0019	
D.C. INLASONEN	DEL I OI CONSOIVIEN &	NEGOLATORT ALTAINS	WASHINGTON		20030	
D.C. TREASURER DEPT OF CONSUM-	DEPT OF CONSUMER &	REGULATORY AFFAIRS	WASHINGTON	DC	20090	
DAIRYANN DONES	715 EASTERN MANOR LANE		BIRMINGHAM	AL	35215	
DALEY, OLIVIA	REDACT		REDACT	REDACT	REDACT	
Dallas County	Attn: Consumer Protection Agency	Susan Hawk, District Attorney	Dallas	TX	75207	
DALLAS COUNTY CLEK OF COURT	P.O. BOX 1148	,	SELMA	AL	36702	
DALLAS COUNTY CLERK	509 MAIN STREET	SUITE 200	DALLAS	TX	75202	
DALLAS COUNTY TAX COLLECTOR	P.O.BOX 987		SELMA	AL	36702-0987	
DALLAS COUNTY, TX	TAX COLLECTOR	P.O. BOX 139066	DALLAS	TX	75313-9066	
DALTON, BEVERLY	REDACT		REDACT	REDACT	REDACT	
DALTON, ELANTRA	REDACT		REDACT	REDACT	REDACT	
DALTON, SHAMYRA	REDACT		REDACT	REDACT	REDACT	
DAN GREEN	820 MAIN CREEKS ROAD		CHESAPEAKE	VA	23320	
DANDRIDGE, SHEENA	REDACT		REDACT	REDACT	REDACT	
DANIEL, DANIELA	REDACT	REDACT	REDACT	REDACT	REDACT	
DANIEL, LISA	REDACT		REDACT	REDACT	REDACT	
DANIELS, ASHTON	REDACT	REDACT	REDACT	REDACT	REDACT	
DANIELS, CHRISTOPHER D.	REDACT		REDACT	REDACT	REDACT	
DANIELS, MINNIE	REDACT		REDACT	REDACT	REDACT	
DANIEL-WILLIAMS, TYEISHA	REDACT		REDACT	REDACT	REDACT	
DARBY, LATOYA	REDACT	REDACT	REDACT	REDACT	REDACT	
DAS, CAROLINE	REDACT	REDACT	REDACT	REDACT	REDACT	
DAVENPORT, BRITTANY	REDACT		REDACT	REDACT	REDACT	

Page 53 of 176

Name	Address 1	Address 2	City	State	PostalCode	Country
DAVID EDWARDS ATTY AT LAW	PO BOX 458		PARIS	TN	38242	
DAVID G. PEAKE	CHAPTER 13 TRUSTEE	PO BOX 2158	MEMPHIS	TN	38101-2158	
DAVID RUSKIN (CHAPTER 13						
TRUSTEE)	CHAPTER 13 STANDING TRUSTEE	1593 RELIABLE PARKWAY	CHICAGO	IL	60686-0015	
DAVID TENNEY	604 RUNNING BROOK ROAD		HOOVER	AL	35226	
DAVID'S PLACE	OFF PRICE CLOTHING COMPANY	1155 S BOYLE AVE	LOS ANGELES	CA	90023	
DAVID'S WHOLESALE	7395 W 20TH AVE		HIALEAH	FL	33014	
DAVILA ELECTRIC CO.	1842 BANDERA RD		SAN ANTONIO	TX	78228-3801	
DAVIS, AMBER	REDACT		REDACT	REDACT	REDACT	
DAVIS, ANDRE	REDACT		REDACT	REDACT	REDACT	
DAVIS, AVERII	REDACT		REDACT	REDACT	REDACT	
DAVIS, BRITTANY	REDACT	REDACT	REDACT	REDACT	REDACT	
DAVIS, BRITTANY	REDACT		REDACT	REDACT	REDACT	
DAVIS, CEDRINA	REDACT		REDACT	REDACT	REDACT	
DAVIS, CENTRELL	REDACT		REDACT	REDACT	REDACT	
DAVIS, CHYNNA	REDACT		REDACT	REDACT	REDACT	
DAVIS, DEWALT	REDACT		REDACT	REDACT	REDACT	
DAVIS, EARL	REDACT	REDACT	REDACT	REDACT	REDACT	
DAVIS, ERIE'L	REDACT	REDACT	REDACT	REDACT	REDACT	
DAVIS, GEORGIANA	REDACT		REDACT	REDACT	REDACT	
DAVIS, III, WILLIAM	REDACT		REDACT	REDACT	REDACT	
DAVIS, JACOBIA	REDACT	REDACT	REDACT	REDACT	REDACT	
DAVIS, JACQUELINE	REDACT		REDACT	REDACT	REDACT	
DAVIS, LANETTE	REDACT		REDACT	REDACT	REDACT	
DAVIS, LARRY	REDACT		REDACT	REDACT	REDACT	
DAVIS, MICHELLE	REDACT	REDACT	REDACT	REDACT	REDACT	
DAVIS, REGINA	REDACT	REDACT	REDACT	REDACT	REDACT	
DAVIS, SARA	REDACT		REDACT	REDACT	REDACT	
DAVIS, SHERIDA	REDACT		REDACT	REDACT	REDACT	
DAVIS, STEDMON	REDACT		REDACT	REDACT	REDACT	
DAVIS, TENAYA	REDACT		REDACT	REDACT	REDACT	
DAVIS, TRACY	REDACT		REDACT	REDACT	REDACT	
DAVIS, VICTORIA	REDACT		REDACT	REDACT	REDACT	
DAVIS, WALT	REDACT	REDACT	REDACT	REDACT	REDACT	
DAVIS, WALTER	REDACT		REDACT	REDACT	REDACT	
DAVIS, WILLONDA	REDACT		REDACT	REDACT	REDACT	
DAVIS, WYONNA	REDACT	REDACT	REDACT	REDACT	REDACT	

Name	Address 1	Address 2	City	State	PostalCode	Country
DAWSON, ALEXIS	REDACT		REDACT	REDACT	REDACT	
DAWSON, ALEXUS	REDACT		REDACT	REDACT	REDACT	
DAWSON, BRENDA	REDACT		REDACT	REDACT	REDACT	
DAWSON, SAMANTHA	REDACT		REDACT	REDACT	REDACT	
DAWSON, SHANTA	REDACT		REDACT	REDACT	REDACT	
DAYTON POWER & LIGHT COMPANY	PO BOX 740598		CINCINNATI	ОН	45274-0598	
DAYTON POWER AND LIGHT						
COMPANY	1065 WOODMAN DR.		DAYTON	OH	45432	
DC CHILD SUPPORT CLEARING						
HOUSE	PO BOX 37868		WASHINGTON	DC	20013-7868	
DC CHILD SUPPORT CLEARINGHOUSE	PO BOX 37868		WASHINGTON	DC	20013-7898	
DC DEPT OF EMPLOYMENT SERVICES	PO BOX 96664		WASHINGTON	DC	20090-6664	
DC Office of Tax and Revenue	1101 4th Street SW	Suite 270 West	Washington	DC	20024	
DCPTJPS 312 LAGRANGE GA, LLC	6528 N. 63rd Place		Paradise Valley	AZ	85253	
DCR MANAGEMENT, LLC	609 Rosedale Avenue		Roselle	IL	60172	
DCR MGMT LLC	609 ROSEDALE AVE.		ROSELLE	IL	60172	
DCTJPS 312 LAGRANGE GA, LLC	PO BOX 535587		ATLANTA	GA	30353-5587	
DDR Corp.	C/O Kelley Drye & Warren	Attn: Robert L. LeHane, Esq.	New York	NY	10178	
DDRTC EISENHOWER CROSSING, LLC	ACCT #342878-30403-57863	PO BOX 534420	ATLANTA	GA	30353-4420	
DDRTC EISENHOWER CROSSING, LLC.	3300 ENTERPRISE PL		BEACHWOOD	ОН	44122	
DDRTC EISENHOWER CROSSING, LLC.	P.O. BOX 534420		ATLANTA	GA	30353-4420	
DE COLORES/DE DESIGN	950 E. 12TH ST.	SUITE #112	LOS ANGELES	CA	90021	
DE LEON, EDITH	REDACT	REDACT	REDACT	REDACT	REDACT	
DEALS 4 ALL SEASONS	1620 E 22ND ST		LOS ANGELES	CA	90011	
DEALS WHOLESALE	ONE PACIFIC ST		WOOD RIDGE	NJ	07075	
DEAN, DAVEDA	REDACT		REDACT	REDACT	REDACT	
DEAN, FREDERICK	REDACT		REDACT	REDACT	REDACT	
DEAN, HARVEY	REDACT		REDACT	REDACT	REDACT	
DEAN, KENNETH	REDACT		REDACT	REDACT	REDACT	
DEAN, SHARON	REDACT		REDACT	REDACT	REDACT	
DEAR, TREVON	REDACT		REDACT	REDACT	REDACT	
DEAR, TREVON	REDACT	REDACT	REDACT	REDACT	REDACT	

Name	Address 1	Address 2	City	State	PostalCode	Country
Deaton's Mechanical	1435 Brookville Way	Suite J	Indianapolis	IN	46239	
DEATON'S MECHANICAL CO.	1435 BROOKVILLE WAY	SUITE J	INDIANAPOLIS	IN	46239	
DEBEL, JANIKA	REDACT		REDACT	REDACT	REDACT	
DEBORAH LAMMI	1052 JONES HILL RD		ASHBY	MA	01431	
	C/O PROVOSTY, SADLER,					
DEBRA A. JORDAN	DELAUNAY, FIORENZA & SOBEL	934 THIRD STREET	ALEXANDRIA	LA	71309	
DECATUR LOCK & KEY SHOP, INC.	380 WEST PONCE DE LEON AVE		DECATUR	GA	30030	
DEDICATED PERSONNEL, INC	DEPT. #1488	P.O. BOX 11407	BIRMINGHAM	AL	35246-1488	
DEGREE, J'COYIA	REDACT		REDACT	REDACT	REDACT	
		Robert James, Dekalb County				
Dekalb County	Attn: Consumer Protection Agency	Attorney	Decatur	GA	30030	
DEKALB COUNTY	LICENSE & INTERNAL AUDIT	PO BOX 100020	DECATUR	GA	30031-7020	
	FALSE ALARM REDUCTION					
DEKALB COUNTY FARP	PROGRAM	P.O. BOX 116569	ATLANTA	GA	30368-6569	
DEKALB COUNTY TAX						
COMMISSIONER	COLLECTIONS DIVISION-PP	PO BOX 100004	DECATUR	GA	30031-7004	
DEKALB COUNTY, GEORGIA	178 SAMS STREET		DECATUR	GA	30030	
	TREASURY & ACCOUNTING					
DEKALB COUNTY, GEORGIA	SERVICES, PO BOX 1027		DECATUR	GA	30031-1027	
DEKALB COUNTY, GEORGIA	TREASURY & ACCTING SERVICES	1300 Commerce Drive	DECATUR	GA	30031-1027	
DELCHAMPS PLAZA ASSOCIATES	C/O CLK MULTIFAMILY MGMT LLC	5545 MURRAY AVE	MEMPHIS	TN	38119	
DELIRIOUS APPAREL	1942 E. 46TH ST.		VERNON	CA	90058	
DELL MARKETING, LP	C/O DELL USA, LP	PO BOX 534118	ATLANTA	GA	30353-4118	
DELLIHUE, MARCETTIA	REDACT		REDACT	REDACT	REDACT	
DELUXE BUSINESS FORMS &						
SOLUTIONS	DELUXE FOR BUSINESS	PO BOX 742572	CINCINNATI	ОН	45274-2572	
DEMERY, TAMARA	REDACT		REDACT	REDACT	REDACT	
DENHAM, TAWANY	REDACT		REDACT	REDACT	REDACT	
DENISE CANTOR	8469 BUCKHURST RD.		CORDOVA	TN	38016	
DENNIE, AMBER	REDACT		REDACT	REDACT	REDACT	
DENNIE, MARIAH	REDACT		REDACT	REDACT	REDACT	
DENNIS, SHEKERIA	REDACT		REDACT	REDACT	REDACT	
DENNIS, TALISA	REDACT	REDACT	REDACT	REDACT	REDACT	
DENNIS, TOIA	REDACT		REDACT	REDACT	REDACT	
DEPARTMENT OF ASSESSMENTS &	ROOM 801-301 W. PRESTON					
TAXATION	STREET		BALTIMORE	MD	21201	

Name	Address 1	Address 2	City	State	PostalCode	Country
DEPARTMENT OF FINANCE			,			,
COLLECTIONS UNIT	CITY HALL, ROOM 107A	121 NORTH LASELLE STREET	CHICAGO	IL	60602	
DEPARTMENT OF FINANCIAL	·					
INSTITUTIONS	PO BOX 7846		MADISON	WI	53707-7846	
DEPARTMENT OF PUBLIC SAFETY -						
LICENSE SECTION	750 PIEDMONT ROAD	SOUTH ENTRANCE	COLUMBUS	ОН	43224	
DEPARTMENT OF REVENUE						
(GARNMT)	PO BOX 1033		JACKSON	MS	39215-1033	
DEPARTMENT OF STATE	BUREAU OF CORPORATION	P.O. BOX 8722	HARRISBURG	PA	17105-8722	
DEPARTMENT OF STATE						
(FICTITIOUS)	FICTITIOUS NAME REGISTRATION	P.O. BOX 1300	TALLAHASSEE	FL	32302-1300	
DEPECHE MODE NY/HIGH						
ACCESSORIES INC	72-68TH STREET		GUTTENBERG	NJ	07093	
Dependable Building Services	1125 Tower Road		Schaumberg	IL	60173	
DEPENDABLE BUILDING SERVICES	1125 TOWER ROAD		SCHAUMBURG	IL	60173	
DEPT OF CONSUMER &						
REGULATORY AFFAIRS	CORPORATIONS DIVISION	PO BOX 92300	WASHINGTON	DC	20090	
DEPT OF PUBLIC UTILITIES	1016 RUSSELL ST		ORANGEBURG	SC	29115	
DEPT OF PUBLIC UTILITIES	PO BOX 1057		ORANGEBURG	SC	29116-1057	
DEPT. OF FINANCIAL INSTITUTIONS	TRADEMARK SECTION	P.O. BOX 7847	MADISON	WI	53707-7847	
DESIDERIO, YASMIN	REDACT		REDACT	REDACT	REDACT	
DESIRE SPORTSWEAR INC.	455 BARREL AVENUE		CARLSTADT	NJ	07072	
DESTINY BUTTS	2054 KNOLLWOOD PLACE		BIRMINGHAM	AL	35242	
DEWEESE PAYROLL ADVANCE, LLC	404 HWY 45 A S		WEST POINT	MS	39773	
DEX IMAGING	P.O. BOX 16244		MOBILE	AL	36616	
Dex Imaging, Inc.	1045 Downtowner Boulevard		Mobile	AL	36609	
DGPOM MASTER TENANT LLC	C/O REDICO MGMT	ONE TOWNE SQUARE	SOUTHFIELD	MI	48203	
DGPOM MASTER TENANT LLC	c/o REDICO Management	One Towne Square	Southfield	MI	48076	
DGPOM MASTER TENANT LLC	ONE TOWNE SQUARE		SOUTHFIELD	MI	48076	
DGPOM MASTER TENANT LLC	ONE TOWNE SQUARE, SUITE 1600		SOUTHFIELD	MI	48076	
DIAZ, ALEXANDRA	REDACT		REDACT	REDACT	REDACT	
DICKEY, KENISHA	REDACT		REDACT	REDACT	REDACT	
DILLON, TASHIEKA	REDACT	REDACT	REDACT	REDACT	REDACT	

Name	Address 1	Address 2	City	State	PostalCode	Country
	FALSE ALARM REDUCTION					
DIR OF FINAN CITY OF BALTIMORE	PROGRAM	P.O. BOX 17535	BALTIMORE	MD	21297-1535	
DIRECTOR OF FINANCE BALTIMORE						
(ALARM PERMITS)	FIRE PREVENTION BUREAU	410 E. LEXINGTON STREET	BALTIMORE	MD	21202	
DISTRICT COURT OF BARBOUR						
COUNTY	303 E BROAD STREET	ROOM 201	EUFAULA	AL	36207	
DISTRICT COURT OF JEFFERSON	716 RICHARD ARRINGTON JR BLVD					
COUNTY	N	ROOM 500	BIRMINGHAM	AL	35203	
DISTRICT COURT OF MONTGOMERY	P.O. BOX 1667		MONTGOMERY	AL	36102-1667	
DIVERSIFIED COLLECTION SERVICES,						
INC	PO BOX 979103		ST. LOUIS	MO	63197-9000	
DIVERSITY MEDICAL						
TRANSPORTATION, LLC	ATTN: AHMED ELMI	2700 CROSSROADS PLAZA DR	COLUMBUS	ОН	43219	
DIVERSITY MEDICAL						
TRANSPORTATION, LLC	ATTN: AHMED ELMI	2700 Crossroads Plaza Drive	Columbus	ОН	43219	
DIVISION OF CHILD SUPPORT	FAMILY SUPPORT REGISTRY	PO BOX 1800	CARROLLTON	GA	30112-1800	
DIXIE ELECTRIC POWER	1863 MS-184		LAUREL	MS	39443	
DIXIE ELECTRIC POWER	PO BOX 88		LAUREL	MS	39441-0088	
DIXIE GLASSHOPPERS, INC	2706 E. PONCE DE LEON AVENUE		DECATUR	GA	30030	
DIXIE STAFFING SERVICES	c/o IVIS INVESTMENTS	P.O. BOX 102268	ATLANTA	GA	30368-2268	
DIXION, SHAUN	REDACT		REDACT	REDACT	REDACT	
DIXON, CHRISTAL	REDACT	_	REDACT	REDACT	REDACT	
DIXON, KEYATTA	REDACT		REDACT	REDACT	REDACT	
DIZZY LIZZY	7526 TYRONE AVENUE		VAN NUYS	CA	91402	
	c/o Moore Property Management,					
DK ENTERPRISES, LLC	LLC	POB 1429	Montgomery	AL	36102	
DNA	747 E 10TH ST	STE 103	LOS ANGELES	CA	90021	
DO-ALL GLASS	8720 S. COMMERCIAL		CHICAGO	IL	60617	
DOBBINS, CHAKA	REDACT	_	REDACT	REDACT	REDACT	
DOBBINS, TYREE	REDACT		REDACT	REDACT	REDACT	
DOCMANN PRINTING & ASSOC., INC.	4889 EAST 345TH STREET		WILLOUGHBY	ОН	44094	
Dodge Plaza Limited Partnership	DODGE PLAZA/Store No139	Attn: Elliot Yadin	Landover	MD	20785	
DODGE PLAZA LIMITED						
PARTNERSHIP	2252 Brightseat Road		Landover	MD	20785	
DODGE PLAZA LTD PARTNERSHIP	2252 BRIGHTSEAT ROAD		LANDOVER	MD	20785	

Name	Address 1	Address 2	City	State	PostalCode	Country
DOLCE VITA INTIMATES, LLC	1000 1ST ST		HARIOSN	NJ	07029	
DOLPHIN PLAZA PW LP	14565 NW 26TH Avenue		Opa Locka	FL	33054	
DOLPHIN PLAZA PW, LP	14565 NW 26TH AVENUE		OPA LOCKA	FL	33054	
DOLPHIN, SHEVERN	REDACT		REDACT	REDACT	REDACT	
DOMBROSKI, ASHLEY	REDACT		REDACT	REDACT	REDACT	
DOMINION	120 TREDEGAR ST		RICHMOND	VA	23219	
DOMINION	PO BOX 26543		RICHMOND	VA	23290-0001	
DOMINION EAST OHIO	P.O. BOX 26785		RICHMOND	VA	23261-6785	
DOMINION VIRGINIA POWER	PO BOX 26543		RICHMOND	VA	23290-0001	
DOMIT INVESTMENT GROUP	201 Vulcan Road	Suite 106	Birmingham	AL	35209	
DOMIT INVESTMENT GROUP, LLC	201 VULCAN RD., STE 106		BIRMINGHAM	AL	35209	
DONALDSON PROPERTIES, LTD	PO BOX 1842		SAN ANTONIO	TX	78297-1842	
DONALDSON PROPERTIES, LTD.	PO BOX 1842		SAN ANTONIO	TX	78297	
DONALDSON PROPERTIES, LTD.	70 NE Loop 410	Suite 185	San Antonio	TX	78216	
DONIGER BURROUGHS	603 Rose Avenue		Venice	CA	90291	
DONNA HAMMILL	P.O. BOX 1275		GRIFFIN	GA	30224	
DONNIE HAMMONDS PLUMBING,						
LLC.	705 W. SHAWNEE DRIVE		MONTGOMERY	AL	36107	
DON'S PLUMBING, INC	526 NE 33 STREET		OAKLAND PARK	FL	33334	
DORCHESTER SQUARE ASSOC, LP	C/O DEWBERRY CAPITAL	1545 PEACHTREE STREET, STE 250	ATLANTA	GA	30309	
DORCHESTER SQUARE ASSOCIATES,						
LP	C/O DEWBERRY CAPITAL	1545 PEACHTREE STREET	ATLANTA	GA	30309	
DORGIN, LINDA	REDACT		REDACT	REDACT	REDACT	
DOROTHY HORN	685 COLFAX		BENTON HARBOR	MI	49022	
DORSETT, RODDERICK	REDACT		REDACT	REDACT	REDACT	
DORSEY, MECIE	REDACT		REDACT	REDACT	REDACT	
DORSEY, NAYA	REDACT		REDACT	REDACT	REDACT	
DORSEY, SHAKAYLA	REDACT	REDACT	REDACT	REDACT	REDACT	
DOUBLE TAKE	68 34TH STREET		BROOKLYN	NY	11232	
	HILLSBOROUGH COUNTY TAX					
DOUG BELDEN, TAX COLLECTOR	COLLECTOR	P.O. BOX 30009	TAMPA	FL	33672-2930	
DOUGHERTY COUNTY	TAX DEPARTMENT	PO BOX 1827	ALBANY	GA	31702-8127	
DOUGLAS COUNTY	TAX COMMISSIONER	P.O. BOX 1177	DOUGLASVILLE	GA	30133-1177	
DOWDLE, TAMIKA	REDACT		REDACT	REDACT	REDACT	
DOWNS SAFE & LOCK CO.	1846 CHESHIRE BRIDGE ROAD		ATLANTA	GA	30324-4945	
DOYLE ROGERS FAMILY LP	P.O. DRAWER A		BATESVILLE	AR	72503	

Name	Address 1	Address 2	City	State	PostalCode	Country
DOYLE ROGERS FAMILY REALTY L.P.	P O DRAWER A		Batesville	AR	72503	
Doyle Rogers Family Realty Limited						
Partnership	ASHER/Store No617	111 Center Street Suite 1510	Little Rock	AR	72201	
DP&L	1065 Woodman Drive		Dayton	ОН	45432	
DP&L	P.O. BOX 740598		CINCINNATI	OH	45274-0598	
DRAINE, OLIVIA	REDACT		REDACT	REDACT	REDACT	
DRAPER AND KRAMER, INC.	107 ADELL PLACE		ELMHURST	IL	60126	
DRAPER, WILLIAM	REDACT		REDACT	REDACT	REDACT	
DRAYTON, TERESA	REDACT		REDACT	REDACT	REDACT	
DREAMWEAR INC.	183 MADISON AVENUE	SUITE 1001	NEW YORK	NY	10016	
DRIZ CONNECTION	1121 S LOS ANGELES ST		LOS ANGELES	CA	90015	
DRUMMER, JASMINE	REDACT		REDACT	REDACT	REDACT	
DTE ENERGY	4141 S BROAD ST		PHILIDELPHIA	PA	19112	
DTE ENERGY	PO BOX 740786		CINCINNATI	ОН	45274-0786	
DUFFY BOX & RECYCLING, INC.	2931 FRANCIS SCOTT KEY		TANETTOWN	MD	21787	
DUKE ENERGY	526 SOUTH CHURCH STREET		CHARLOTTE	NC	28202	
DUKE ENERGY	PO BOX 1004		CHARLOTTE	NC	28201-1004	
DUKE ENERGY	PO BOX 1326		CHARLOTTE	NC	28201-1326	
DUKE ENERGY	PO BOX 1327		CHARLOTTE	NC	28201-1327	
DUKE ENERGY	PO BOX 70516		CHARLOTTE	NC	28272-0516	
DUKE ENERGY PROGRESS	410 S WILMINGTON ST		RALEIGH	NC	27601	
DUKE ENERGY PROGRESS	P.O. BOX 1003		CHARLOTTE	NC	28201-1003	
DUKES, DENVER	REDACT		REDACT	REDACT	REDACT	
DUNCAN, MICHELE	REDACT	REDACT	REDACT	REDACT	REDACT	
DUNCAN, SADE	REDACT	REDACT	REDACT	REDACT	REDACT	
DUNHAM, BERNICE	REDACT		REDACT	REDACT	REDACT	
DUNN, COURTNEI	REDACT		REDACT	REDACT	REDACT	
DUNN, SHAQUETTA	REDACT		REDACT	REDACT	REDACT	
DUNN, SHAQUETTA	REDACT		REDACT	REDACT	REDACT	
DUNSON, KENDRA	REDACT		REDACT	REDACT	REDACT	
DUPREE, BASIA	REDACT		REDACT	REDACT	REDACT	
DUQUESNE LIGHT	411 7TH AVE STE 3		PITTSBURGH	PA	15219	
DUQUESNE LIGHT	PAYMENT PROCESSING CENTER		PITTSBURGH	PA	15230-0010	
	PAYMENT PROCESSING CENTER				101010	
DUQUESNE LIGHT	P.O. Box 10		PITTSBURGH	PA	15230-0010	
DURAN, GINA	REDACT		REDACT	REDACT	REDACT	
DURANT, ERIC	REDACT		REDACT	REDACT	REDACT	

Page 60 of 176

Name	Address 1	Address 2	City	State	PostalCode	Country
DURANT, TREMAINE	REDACT		REDACT	REDACT	REDACT	
DURGA, LLC	11320 CHESTER ROAD		CINCINNATI	OH	45246	
	NORTH EASTWOOD PLAZA/Store					
Durga, LLC	No45	11320 Chester Road	Cincinnati	ОН	45246	
DURHAM COUNTY	TAX COLLECTOR	P.O.BOX 3397	DURHAM	NC	27702	
DURR, CANDACE	REDACT		REDACT	REDACT	REDACT	
DUSTY, HARRELL	REDACT		REDACT	REDACT	REDACT	
DUVAL COUNTY	TAX COLLECTOR	231 E. FORSYTH ST., ROOM 130	JACKSONVILLE	FL	32202	
	38TH FLOOR, ACCOUNTS					
DYKEMA GOSSETT PLLC	RECEIVABLE	400 RENAISSANCE CENTER	DETROIT	MI	48243	
DYNA-CRAFT CONTRACTING, INC	10762 S. PROSPECT AVENUE		CHICAGO	IL	60643	
DYNAMIC EXPRESS INC	2501 71ST STREET		NORTH BERGEN	NJ	07047	
DYNASTY FASHION INC.	1706 MAPLE AVENUE		LOS ANGELES	CA	90015	
E & T AUTO BODY	548 STATE STREET		CALUMET CITY	IL	60409	
E. BATON ROUGE PARISH	DEPT. OF FINANCE-REVENUE	P.O. BOX 2590	BATON ROUGE	LA	70821	
E. EUGENE HASTINGS	CHAPTER 13 TRUSTEE	PO BOX 270	MEMPHIS	TN	38101-0270	
E.L.F.	JA COSMETICS US INC	45 MAYHILL STREET	SADDLE BROOK	NJ	07663	
EAGLE ELECTRIC	573 HWY 61 NORTH		NATCHEZ	MS	39120	
EAGLE'S GLASS & MIRROR, INC	610-1 N. LANE AVE		JACKSONVILLE	FL	32254	
EARNESTINE TAYLOR	608 NORTHSIDE		WARNER ROBINS	GA	31093	
EASLEY, GARRY	REDACT		REDACT	REDACT	REDACT	
EASLEY, SAM	REDACT		REDACT	REDACT	REDACT	
EAST BATON ROUGE PARISH	SHERIFF/TAX COLLECTOR	P.O. BOX 91285	BATON ROUGE	LA	70821-9285	
EAST FASHION	844 6TH AVE, SUITE 609		NEW YORK	NY	10001	
EAST FOREST PLAZA I, LLC	c/o Wilson Kibler	111 Laurel Street	Columbia	SC	29101	
EAST LION	110 E. 9TH STREET	#A 1037	LOS ANGELES	CA	90079	
EAST RIVER MZL, LLC	C/O WINSLOW PROPERTY MGMT	80 HAYDEN AVE.	LEXINGTON	MA	02421	
EASTER, MICHAEL	REDACT		REDACT	REDACT	REDACT	
EASTERLING, LEE	REDACT		REDACT	REDACT	REDACT	
EASTGATE MERCHANTS ASSOC	PO BOX 4169		MONROE	LA	71201	
EASTGATE SHOPPING CENTER	30665 NORTHWESTERN HWY	SUITE 200	FARMINGTON HILLS	MI	48334	
EASTGATE SHOPPING CENTER	30665 Northwester	Suite 200	Farmington Hills	MI	48334	
EASTLAKE MANAGEMENT &						
DEVELOPMENT CORPORATION	2850 S. Michigan Avenue		Chicago	IL	60616	
EASTLAKE MGMT & DEVELOPMENT	COMMERCIAL PROPERTY	1				1
CORP	MANAGER	2850 S MICHIGAN AVENUE	CHICAGO	IL	60616	
Eastland Center	Attn.: Jean Tallent	18000 Vernier Boulevard	Harper Woods	MI	48225	

Page 61 of 176

Name	Address 1	Address 2	City	State	PostalCode	Country
EASTMAN FIRE PROTECTION			•			·
COMPANY, LLC	1450 SOUTER BLVD		TROY	MI	48083	
EASTOVER PLAZA IMPROVEMENTS,						
LLC	C/O DLC MGMT CORP	PO BOX 5122	WHITE PLAINS	NY	10602	
EASTOVER PLAZA IMPROVEMENTS,						
LLC	c/o DLC Management Corporation	580 White Plains Road	Tarrytown	NY	10591	
EASTRIDGE COMMONS REALTY, LLC	150 GREAT NECK ROAD	SUITE 304	GREAT NECK	NY	11021	
EATON, JAMIECE	REDACT		REDACT	REDACT	REDACT	
ECHOLS & SONS, INC	2084 N JIM MILLER ROAD	SUITE 105	DALLAS	TX	75217	
ECONOCO CORPORATION						
(SELLUTIONS)	P.O. BOX 29		KICKSVILLE	NY	11802	
EDDINGS, GERALD	REDACT	REDACT	REDACT	REDACT	REDACT	
EDELEN, LATACHA	REDACT		REDACT	REDACT	REDACT	
EDGE ELECTRIC, INC	2201 N. 39TH AVENUE		HOLLYWOOD	FL	33021	
EDISTO VILLAGE - ORANGEBURG, LLC		85-A MILL STREET, STE 100	ROSWELL	GA	30075	
	EDISTO VILLAGE SHOPPING					
Edisto Village - Orangeburg, LLC	CENTER/Store No507	85 Mill Street, Building A, Suite 100	Roswell	GA	30075	
EDISTO VILLAGE - ORANGEBURG, LLC	'	85-A Mill Street	Roswell	GA	30075	
Edisto Village-Orangeburg, LLC	900 Nations Bank Plaza	1901 Main St.	Columbia	SC	29202	
EDM IMPORTS	33 35TH ST, 5TH FLR		BROOKLYN	NY	11232	
EDWARD, MATHIAS	REDACT		REDACT	REDACT	REDACT	
EDWARDS ELECTRICAL &						
MECHANICAL	ML 505	PO BOX 145400	CINCINNATI	ОН	45250-5400	
EDWARDS, AUNDRAKE	REDACT		REDACT	REDACT	REDACT	
EDWARDS, JUDY	REDACT		REDACT	REDACT	REDACT	
EDWARDS, KATHY	REDACT	REDACT	REDACT	REDACT	REDACT	
EDWARDS, KEVIN	REDACT		REDACT	REDACT	REDACT	
EDWARDS, RICHARD	REDACT		REDACT	REDACT	REDACT	
EDWARDS, ROLAND	REDACT		REDACT	REDACT	REDACT	
EFW - ESTES FORWARDING						
WORLDWIDE	P.O.BOX 26206		RICHMOND	VA	23260	
EGBEBU, ROSLYN	REDACT	REDACT	REDACT	REDACT	REDACT	
EHL	501 EAST 89 ST		BROOKLYN	NY	11236	
EKB TEXTILES	777 EAST WASHINGTON BLVD		LOS ANGELES	CA	90021	
ELECTRIC WIRING SERVICE	P.O. BOX 770635		MEMPHIS	TN	38177	

Page 62 of 176

Name	Address 1	Address 2	City	State	PostalCode	Country
ELECTRICAL HANDYMAN SERVICES	7046-B W. HILLSBOROUGH AVE		TAMPA	FL	33634	
ELEGANT ESSENCE	47-61 PEARSON PLACE, 2FL		LONG ISLAND CITY	NY	11101	
ELIGAH, DEMETRICE	REDACT		REDACT	REDACT	REDACT	
ELIGAH, TACQUIRA	REDACT		REDACT	REDACT	REDACT	
ELLIS, JOHN	REDACT		REDACT	REDACT	REDACT	
ELLIS, MAURNECE	REDACT		REDACT	REDACT	REDACT	
ELLIS, TINA	REDACT	REDACT	REDACT	REDACT	REDACT	
ELWOOD STAFFING	PO BOX 1024		COLUMBUS	IN	47202	
EM COLUMBUS II, LLC	2740 B EASTLAND MALL		COLUMBUS	ОН	43232	
EMD GROUP	1410 BROADWAY		NEW YORK	NY	10018	
EMERALD COAST UTILITIES	9255 Sturdevant St		PENSACOLA	FL	32514	
EMERALD COAST UTILITIES	PO BOX 18870		PENSACOLA	FL	32523-8870	
EMPLOYMENT SCREENING SERVICES		PO BOX 830520	BIRMINGHAM	AL	35283	
ENERGY SHERLOCK	40692 MANOR HOUSE ROAD		LEESBURG	VA	20175	
ENGLEBERT, AUSTIN	REDACT		REDACT	REDACT	REDACT	
ENGLISH, CECELIA	REDACT		REDACT	REDACT	REDACT	
ENGLISH, DEANDRE	REDACT		REDACT	REDACT	REDACT	
ENT GLOBAL	1430 S. GRIFFITH AVE		LOS ANGELES	CA	90021	
ENTERGY	5755 CHOCTAW DR		BATON ROUGE	LA	70806	
ENTERGY	PO BOX 8101		BATON ROUGE	LA	70891-8101	
ENTERGY	PO BOX 8103		BATON ROUGE	LA	70891-8103	
ENTERGY	PO BOX 8105		BATON ROUGE	LA	70891-8105	
ENTERGY	PO BOX 8106		BATON ROUGE	LA	70891-8106	
ENTERGY	PO BOX 8108		BATON ROUGE	LA	70891-8108	
ENTERPRISE ASSOCIATES	PO BOX 822187		PHILADELPHIA	PA	19182-2187	
ENTERPRISE ASSOCIATES, GP	888 Bestgate Road	Suite 411	Annapolis	MD	21401	
ENVY CLOTHING	1915 VIOLET STREET		LOS ANGELES	CA	90021	
EPB- REMITTANCE PROCESSING	10 West M.L. King Blvd		CHATTANOOGA	TN	37402	
EPB- REMITTANCE PROCESSING	PO BOX 182254		CHATTANOOGA	TN	37422-7254	
EPICOR RETAIL SOLUTIONS CORP	C/O T60167U	PO BOX 66512	CHICAGO	IL	60666-0512	
Epicor Software Retail Solutions	Attn.: Martin Geary	2800 Trans Canada Highway	Point-Claire		QC H9R 1B1	CANADA
EPOCH APPAREL	1407 BROADWAY, #613		NEW YORK	NY	10018	
EPPS, MAGGIE	REDACT		REDACT	REDACT	REDACT	
EPPS, STEPHANIE	REDACT		REDACT	REDACT	REDACT	

Name	Address 1	Address 2	City	State	PostalCode	Country
EQUAL OPPORTUNITY TEMPORARY						
STAFFING AGENCY, INC.	613 14TH STREET	P.O. BOX 629	BESSEMER	AL	35020	
EQUITY DEVELOPMENT CORP	PO BOX 21446		WACO	TX	76702-1446	
EQUITY DEVELOPMENT						
CORPORATION	P. O. Box 519		Segoville	TX	75159	
EQUITY ONE (FL	410 PARK AVE #1220		NEW YORK	NY	10022	
	CITY NATIONAL BANK, PO Box 01-					
EQUITY ONE (FL	9170		MIAMI	FL	33101-9170	
EQUITY ONE (FL PORTFOLIO) INC	CITY NATIONAL BANK	PO BOX 01-9170	MIAMI	FL	33101-9170	
EQUITY ONE (FLORIDA PORTFOLIO),						
INC.	1600 N.E. Miami Gardens Drive		N. Miami Beach	FL	33179	
EQUITY ONE (LA PORTFOLIO) LLC	BANK OF AMERICA	PO BOX 404716	ATLANTA	GA	30384-4716	
ERIC BAIRD	678 LEE ROAD 450		PHENIX CITY	AL	36870	
ERIE WATER WORKS	240 W 12TH ST		ERIE	PA	16501	
ERIE WATER WORKS	P.O. BOX 6179		ERIE	PA	16512-6179	
ERIN RODGERS	1024 CLINTON STREET	APT. 305	HOBOKEN	NJ	07030	
ERMAK, EKATERINA	REDACT		REDACT	REDACT	REDACT	
ERVIN, RODERICK	REDACT	REDACT	REDACT	REDACT	REDACT	
,						
Escambia County	Attn: Consumer Protection Agency	County Attorney	Pensacola	FL	32502	
ESCAMBIA COUNTY	TAX COLLECTOR	P.O. BOX 1312	PENSACOLA	FL	35291	
ESCOBAR, ARACELY	REDACT		REDACT	REDACT	REDACT	
ESL / ELMER & SON LOCKSMITHS,						
INC.	ESL CONTRACTORS HARDWARE	P.O. BOX 5	STEGER	IL	60475	
ESPINOZA, ENEDELIA	REDACT		REDACT	REDACT	REDACT	
ESQUIRE FOOTWEAR	385 5TH AVE		NEW YORK	NY	10016	
ESSENTIALS NEW YORK	148 W 37TH ST, 10TH FLR		NEW YORK	NY	10018	
ESTES EXPRESS LINES	PO BOX 25612		RICHMOND	VA	23260-5612	
ESTI COUTURE	1407 BROADWAY, STE 2507		NEW YORK	NY	10018	
ESTRIDGE, TAKAI	REDACT		REDACT	REDACT	REDACT	
EUTAW SHOPPING CENTER	C/O STOUT PROPERTIES	P.O. BOX 87108	FAYETTEVILLE	NC	28304	
EUTAW SHOPPING CENTER, INC.	P.O. Box 35005		Fayetteville	NC	28303	
EVANS WHITFIELD, SHARMAINE	REDACT		REDACT	REDACT	REDACT	
EVANS, CHAELA	REDACT		REDACT	REDACT	REDACT	
EVANS, CIERRA	REDACT		REDACT	REDACT	REDACT	
EVANS, DAPHNE	REDACT		REDACT	REDACT	REDACT	
EVANS, DERRICK	REDACT		REDACT	REDACT	REDACT	

Page 64 of 176

Name	Address 1	Address 2	City	State	PostalCode	Country
EVANS, THERESA	REDACT		REDACT	REDACT	REDACT	
EVELYN ORSBORN	3309 COACH HOUSE COURT		CONYERS	GA	30013	
EVERGREEN PLAZA ASSOCIATES	C/O TPG MGMT, INC	9730 SOUTH WESTERN AVENUE	EVERGREEN PARK	IL	60805	
		9730 SOUTH WESTERN AVE, SUITE				
EVERGREEN PLAZA ASSOCIATES I, LP	C/O TPG MANAGEMENT INC.	418	EVERGREEN PARK	IL	60805	
EVERSOURCE	300 Cadwell Dr		SPRINGFIELD	MA	01104-174	
EVERSOURCE	P.O. BOX 650032		DALLAS	TX	75265-0032	
EVERSOURCE	P.O. BOX 650851		DALLAS	TX	75265-0851	
EVERSOURCE	PO BOX 660369		DALLAS	TX	75266-0369	
EWING, FELICIA	REDACT		REDACT	REDACT	REDACT	
EWING, FELICIA	REDACT	REDACT	REDACT	REDACT	REDACT	
EXACT TARGET, INC	DEPT CH 17808		PALATINE	IL	60055-7808	
EXCEL HANDBAGS, CO., INC.	3651 NW 81ST STREET		MIAMI	FL	33147	
EXCELL BRANDS	3 INDEPENDENCE WAY	STE 114	PRINCETON	NJ	08540	
EXIST INC.	1650 NW 23RD AVE		FT. LAUDERDALE	FL	33311	
EXPRESS GLASS	3315 JACKSON ST.		ALEXANDRIA	LA	71301	
EXPRESS SERVICES, INC	P.O. BOX 535434		ATLANTA	GA	30353-5434	
EXTERM PEST CONTROL	3941 S. 71ST STREET		WILMAUKEE	WI	53220-2301	
EXTRA SPORTSWEAR INC	115 KENNEDY DR		SAYREVILLE	NJ	08872	
EZ APPAREL	148 West 37th	10th Floor	NEW YORK	NY	10018	
EZELL, AMINAH	REDACT		REDACT	REDACT	REDACT	
EZELL, JAVAIR	REDACT		REDACT	REDACT	REDACT	
EZELL, JEROME	REDACT		REDACT	REDACT	REDACT	
EZELL, YOLANDA	REDACT		REDACT	REDACT	REDACT	
F.E.S. FIRE & SECURITY, LLC	P.O. BOX 127		MIAMITOWN	ОН	45041-0127	
FAHNDRICH, JR., DONALD	REDACT		REDACT	REDACT	REDACT	
FAILLE, LORY	REDACT		REDACT	REDACT	REDACT	
FALSE ALARM REDUCTION (MD)	1300 MERCANTILE LANE	SUITE 100K	LARGO	MD	20774	
FALSE ALARM REDUCTION (FL)	P.O. BOX 5489		GAINESVILLE	FL	32627-5489	
FAME FASHION HOUSE	214 WEST 39TH ST		NEW YORK	NY	10018	
FAMILY SUPPORT PAYMENT CENTER	P.O. BOX 109001		JEFFERSON CITY	мо	65110-9001	
FANIEL. ALISHA	REDACT		REDACT	REDACT	REDACT	
FANIEL, MARQUIST	REDACT		REDACT	REDACT	REDACT	
FARBOD MIRMASHHOURI	65-11 108TH STREET	APT#1B	QUEENS, FOREST HILLS	NY	11375	
FARR, JANEA	REDACT	REDACT	REDACT	REDACT	REDACT	
FARWELL, KAITLYN	REDACT	-	REDACT	REDACT	REDACT	
FASHION FORWARD NY	PO BOX 1279		NEW YORK	NY	10156	

Page 65 of 176

Name	Address 1	Address 2	City	State	PostalCode	Country
FASHION MOUNTAIN	240 W 37TH ST		NEW YORK	NY	10018	
FASHION ROAD	1407 BROADWAY, RM 1811		NEW YORK	NY	10018	
FAST TURN WHOLESALE	17709 E VALLEY BLVD		CITY OF INDUSTRY	CA	91174	
FAYETTEVILLE FIRE DEPT	433 HAY STREET		FAYETTEVILLE	NC	28301-5537	
FEARON, TAHAUNEY	REDACT	REDACT	REDACT	REDACT	REDACT	
FEATHER BRANCH	2922 S MAIN ST		LOS ANGELES	CA	90007	
FEATHERSON, ANIKA	REDACT		REDACT	REDACT	REDACT	
Federal Insurance Company (8225-						
3074)	15 Mountainview Rd		Warren	NJ	07059-6711	
FEDERICO, JOY	REDACT		REDACT	REDACT	REDACT	
FEDEX (TX)	P.O. BOX 660481		DALLAS	TX	75266-0481	
FEFE, VALERIE	REDACT		REDACT	REDACT	REDACT	
FELDERHUNTER, DARLENE	REDACT		REDACT	REDACT	REDACT	
FELTON, JACOB	REDACT	REDACT	REDACT	REDACT	REDACT	
FELTON, MICHAEL	REDACT		REDACT	REDACT	REDACT	
FERNANDO, MARIELLE	REDACT		REDACT	REDACT	REDACT	
FERRELL, DESTYNE	REDACT		REDACT	REDACT	REDACT	
FESIK, WILLIAM	REDACT		REDACT	REDACT	REDACT	
FICKLIN, SHANNON	REDACT		REDACT	REDACT	REDACT	
FIELD A/C & HEATING, LLC	P.O. BOX 265		LYON	MS	38645	
FIELDS, DANIEL	REDACT		REDACT	REDACT	REDACT	
FIELDS, JACQUELINE	REDACT		REDACT	REDACT	REDACT	
FIELDS, SR., EDGAR	REDACT		REDACT	REDACT	REDACT	
FIGUEROA, MONICA	REDACT		REDACT	REDACT	REDACT	
Finance One	File 749208		Los Angeles	CA	90074-9208	
FINE WEAR FASHIONS	1501 BROADWAY	12TH FLOOR	NEW YORK	NY	10036	
FINELINE TECHNOLOGIES, INC	PO BOX 921933		NORCROSS	GA	30010	
FINESSE NOVELTY CORP	2 CHANNEL DRIVE	SUITE 200, 2ND FLOOR	PORT WASHINGTON	NY	11050	
FINFROCK, KRYSTAL	REDACT	REDACT	REDACT	REDACT	REDACT	
Fireman's Fund Insurance Company						
(SHX 00057970113)	777 San Marin Drive		Novato	CA	94998	
FIRST AMERICAN BANK	1650 Louis Ave		Elk Grove Village	IL	60007	
FIRST CHOICE PERSONNEL, LLC	P.O. BOX 155		GADSDEN	AL	35902-0155	
	D33Central Bank Operations -					
FIRST CITIZENS BANK - NC	DAC02	PO Box 27131	Raleigh	NC	27611-7131	
FIRST CITIZENS BANK - SC	1230 Main Street	PO Box 29	Columbia	SC	29202-0029	
FIRST COMMERCIAL REALTY	27600 NORTHWESTERN HIGHWAY		SOUTHFIELD	MI	48034	

Page 66 of 176

Name	Address 1	Address 2	City	State	PostalCode	Country
FIRST MERIT BANK, N.A.	295 FirstMerit Circle		Akron	ОН	44307-2359	
FIRST MIDWEST BANK	PO Box 580		Joliet	IL	60434-0580	
FIRST SECURITY BANK	130 US 51		Batesville	MS	38606	
FIRST STATE BANK AND TRUST CO.	PO Box 4810		Valdosta	GA	31604-4810	
FIRST TENNESSEE	PO Box 84		Memphis	TN	38101	
FISHER FIRE EXTINGUISHER SERVICE	PO BOX 3364		JACKSON	MS	39207	
FISHER, D'EBONY	REDACT		REDACT	REDACT	REDACT	
FISHER, HOPE	REDACT	REDACT	REDACT	REDACT	REDACT	
FISHER, SEAN	REDACT		REDACT	REDACT	REDACT	
FITZ MECHANICAL	6311 HARFORD ROAD		BALTIMORE	MD	21214	
FITZGERALD, TYANA	REDACT	REDACT	REDACT	REDACT	REDACT	
FITZPATRICK, MAURICE	REDACT		REDACT	REDACT	REDACT	
FLAGG, ANTHONY	REDACT		REDACT	REDACT	REDACT	
FLAGG, ANTONIQUE	REDACT		REDACT	REDACT	REDACT	
FLAGSTAR BANK	5151 Corporate Drive	MS E-203-1	Troy	MI	48098	
FLAMINGO PLUMBING & BACKFLOW						
SERVICES, LLC	2781 VISTA PARKWAY	SUITE K10	WEST PALM BEACH	FL	33411	
FLANAGAN, KRISTIN	REDACT		REDACT	REDACT	REDACT	
FLAUNT MODEL MANAGEMENT,						
INC.	35 WEST 35TH STREET		NEW YORK	NY	10001	
FLENTROY, BEVERLY	REDACT	REDACT	REDACT	REDACT	REDACT	
FLETCHER BRIGHT	C/O LURIE & ASSOCIATES, LTD	2650 THOUSAND OAKS BLVD	MEMPHIS	TN	38118	
	HICKORY RIDGE CROSSING/Store					
Fletcher Bright	No577	c/o Lurie & Associates, LLC	Memphis	TN	38118	
FLETCHER BRIGHT C/O LURIE &						
ASSOCIATES, LLC	C/O LURIE & ASSOC. LTD.	2650 Thousand Oaks Blvd.	MEMPHIS	TN	38118	
FLETCHER, SUMER	REDACT	REDACT	REDACT	REDACT	REDACT	
FLINT EMC - SEDC	494 PERRY PKWY		FORSYTH	GA	31029	
FLINT EMC - SEDC	PO BOX 530812		ATLANTA	GA	30353-0812	
FLINT EMC - SEDC	PO BOX 530812		ATLANTA	GA	30353-0812	
FLINT EMC - SEDC	98 Carl Vinson Parkway	P.O. Box 7089	Warner Robins	GA	31095	
FLINT ENERGIES	3 S. Macon Street	P.O. Box 308	Reynolds	GA	31076-0308	
FLOORS UNLIMITED OF ALEXANDRIA	1925 GUS KAPLAN DRIVE		ALEXANDRIA	LA	71301	
FLORENCE COUNTY TREASUER	PO BOX 100501		FLORENCE	SC	29501	
FLORES, NICOLE	REDACT		REDACT	REDACT	REDACT	

Name	Address 1	Address 2	City	State	PostalCode	Country
FLORES, VALANTE	REDACT		REDACT	REDACT	REDACT	
Florida Department of Revenue	5050 W. Tennesse St		Tallahassee	FL	32399-0100	
Florida Department of Revenue	5050 West Tennessee Street		Tallahassee	FL	32399-0100	
FLORIDA DEPT OF STATE	DIVISION OF CORPORATION	PO BOX 6327	TALLAHASSEE	FL	32314	
FLORIDA DISBURSEMENT UNIT	P.O. BOX 8500		TALLAHASSEE	FL	32314-8500	
FLORIDA ELECTRICAL COMPANY	901 WEST GOVERNMENT STREET		PENSACOLA	FL	32501	
FLORIDA PLUMBING WORKS INC	632 N. RIDGEWOOD AVENUE		DAYTONA BEACH	FL	32114	
FLORIDA U.C. FUND	FLORIDA DEPT OF REVENUE	5050 W. TENNESSEE ST.	TALLAHASSEE	FL	32399-0110	
FLORISSANT GLASS COMPANY	PO BOX 1104		FLORISSANT	MO	63031	
FLOWERS, TIFFANI	REDACT		REDACT	REDACT	REDACT	
FLOYD, PERLICIA	REDACT		REDACT	REDACT	REDACT	
FLOYD, TACHAE	REDACT		REDACT	REDACT	REDACT	
FLOYD, TACHAE	REDACT		REDACT	REDACT	REDACT	
FLY, JANICE	REDACT		REDACT	REDACT	REDACT	
FOGLE, A'TIYANA	REDACT		REDACT	REDACT	REDACT	
FOGLE, A'TIYANA	REDACT		REDACT	REDACT	REDACT	
FOLMAR & ASSOCIATES	P.O. BOX 16765		MOBILE	AL	36616	
FONSECA, DANIEL	REDACT		REDACT	REDACT	REDACT	
FONTENOT, DORIS	REDACT		REDACT	REDACT	REDACT	
FORD, SKYLAR	REDACT		REDACT	REDACT	REDACT	
FOREMAN, DARRYL	REDACT		REDACT	REDACT	REDACT	
FOREVER 17	1407 BROADWAY, RM 1610		NEW YORK	NY	10018	
FOREVER LINK	18738 SAN JOSE AVE		CITY OF INDUSTRY	CA	91748	
FORREST COUNTY, MS	TAX COLLECTOR	P.O. BOX 1689	HATTIESBURG	MS	39403	
Forsyth County	Attn: Consumer Protection Agency	County Attorney	Winston-Salem	NC	27101	
		.,,			-	
FORSYTH COUNTY TAX COLLECTOR	P.O. BOX 82		WINSTON-SALEM	NC	27102-0082	
FORT PIERCE UTILITIES AUTHORITY	206 S. 6TH ST		FORT PIERCE	FL	34950	
FORT PIERCE UTILITIES AUTHORITY	PO BOX 13929		FORT PIERCE	FL	34979-3929	
FORT PIERCE UTILITIES AUTHORITY	PO BOX 13929		FORT PIERCE	FL	34979-3929	
FORT PIERCE UTILITIES AUTHORITY	P.O. Box 3191		FORT PIERCE	FL	34948	
FORTE', STEFANIE	REDACT		REDACT	REDACT	REDACT	
FORTUNE DYNAMIC INC.	21923 FERRERO PARKWAY		CITY OF INDUSTRY	CA	91789	

Page 68 of 176

Name	Address 1	Address 2	City	State	PostalCode	Country
FORTUNE, SANDRA	REDACT		REDACT	REDACT	REDACT	
FOSTER, II, DANTONIO	REDACT		REDACT	REDACT	REDACT	
FOSTER, WILLIAM	REDACT		REDACT	REDACT	REDACT	
FOUNTAIN PARK STATION, INC	PO BOX 645164		CINCINNATI	ОН	45264-5164	
FOUNTAIN, LOVE	REDACT	REDACT	REDACT	REDACT	REDACT	
FOUNTAIN, UNIQUE	REDACT		REDACT	REDACT	REDACT	
FOUST, CARLA	REDACT		REDACT	REDACT	REDACT	
FOWLERS HOLDINGS LLLP	Donald P. Fowler, GP	410 N. Peters Road	Knoxville	TN	37922	
FOX, DESIREE	REDACT		REDACT	REDACT	REDACT	
FOX, DESIREE	REDACT		REDACT	REDACT	REDACT	
FOX, JONESE	REDACT		REDACT	REDACT	REDACT	
FOX, MARILYN	REDACT		REDACT	REDACT	REDACT	
FOX, RAQUEL	REDACT		REDACT	REDACT	REDACT	
FPL	700 UNIVERSE BLVD		JUNO BEACH	FL	33408	
FPL	GENERAL MAIL FACILITY		MIAMI	FL	33188-0001	
FRANKLIN COLLECTION SERVICE Franklin County	C/O MITCHELL, MCNUTT & SAMS Attn: Consumer Protection Division	PO BOX 466 Franklin County Attorney Ron O'Brien	TUPELO	MS	38802 43215	
FRANKLIN, FALLON	REDACT	o briefi	REDACT	REDACT	REDACT	
FRANKLIN, LOREAL	REDACT		REDACT	REDACT	REDACT	
FRANKLIN, THERESA	REDACT		REDACT	REDACT	REDACT	
FRANSISCO, LETICIA	REDACT		REDACT	REDACT	REDACT	
FRASIER, TYESHIA	REDACT		REDACT	REDACT	REDACT	
FRAZIER, LATINA	REDACT		REDACT	REDACT	REDACT	
FRECKLES/MAX IMPORTS	1407 BROADWAY, STE 1903		NEW YORK	NY	10018	
FREE WILL	3220 MINES AVE		LOS ANGELES	CA	90023	
FREEMAN, ANDRETTA	REDACT		REDACT	REDACT	REDACT	
FREEMAN, ANNETTE	REDACT		REDACT	REDACT	REDACT	
FREEMAN, CRYSTAL	REDACT		REDACT	REDACT	REDACT	
FREEMAN, JEANETTE	REDACT		REDACT	REDACT	REDACT	
FREEMAN, ROBERT	REDACT		REDACT	REDACT	REDACT	
FREEMAN, RODNEY	REDACT		REDACT	REDACT	REDACT	
FRIESON, SUPETRIUS	REDACT	_	REDACT	REDACT	REDACT	
FRISON, S'CHALANA	REDACT	REDACT	REDACT	REDACT	REDACT	
FRUITPORT TOWNSHIP	6543 AIRLINE ROAD		FRUITPORT	MI	49415	
FRUITPORT TOWNSHIP	5865 AIRLINE ROAD		FRUITPORT	MI	49415	
FT PIERCE UTILITY	206 6th Street		Fort Pierce	FL	34950	

Page 69 of 176

Name	Address 1	Address 2	City	State	PostalCode	Country
FTC Commercial	P.O.Box 51228		Los Angeles	CA	90051-5528	
FUENTES, RANDI	REDACT		REDACT	REDACT	REDACT	
FUENTES, RUTH	REDACT		REDACT	REDACT	REDACT	
FULLER, THERESA	REDACT		REDACT	REDACT	REDACT	
FULTON COUNTY FIRE RESCUE	5440 FULTON INDUSTRIAL BLVD		ATLANTA	GA	30336	
FULTON COUNTY TAX						
COMMISSIONER	PO BOX 105052		ATLANTA	GA	30348-5052	
FULTON, STEPHON	REDACT		REDACT	REDACT	REDACT	
FUNB 2000-C2 TORRENCE RETAIL	c/o Joseph Freed and Associates,					
LLC, successor to KFS LANDINGS, LLC	LLC	33 South State Street	Chicago	IL	60603	
FUNB 2000-C2 TORRENCE RETAIL,						
LLC	C/O EDGEMARK ASSET MGMT, LLC	2215 YORK RD.	OAK BROOK	IL	60561	
FUNDERBURK, ANRIONIONO	REDACT		REDACT	REDACT	REDACT	
FUNDERBURK, TEAIRA	REDACT		REDACT	REDACT	REDACT	
FUNEZ, ROMINA	REDACT		REDACT	REDACT	REDACT	
FUQUA, AUNTAY	REDACT		REDACT	REDACT	REDACT	
FUSCHILLO, DIANA	REDACT		REDACT	REDACT	REDACT	
FUSCHILLO, LAURA	REDACT		REDACT	REDACT	REDACT	
FUTRELL, JENNIFER	REDACT		REDACT	REDACT	REDACT	
FUTURE FAMILY LLLP	3100 Araphoe Avenue	Suite 501	Boulder	СО	80303	
FUTURE FAMILY LLLP	18631 N. 19 Avenue		Phoenix	AZ	85027	
FUTURE FAMILY LLP STORE NO. 261	18631 N. 19TH AVENUE		PHOENIX	AZ	85027	
FW IL-RIVERVIEW PLAZA, LLC	1211 W. 22ND STREET	SUITE 300	OAK BROOK	IL	60523	
G & S ELECTRICAL SERVICES, LLC	P.O. BOX 18432		MEMPHIS	TN	38181-0432	
G & S OFF PRICE, INC.	1424 S. MAIN ST.		LOS ANGELES	CA	90015	
G & S SERVICE COMPANY, INC	3101 FLORAL PARK ROAD		BRANDYWINE	MD	20613-9115	
G&S Service Company	3101 Floral Park Road		Brandywine	MD	20613-9115	
GAINES, PAGE	REDACT		REDACT	REDACT	REDACT	
GAINES, VALEEKA	REDACT		REDACT	REDACT	REDACT	
GAINESVILLE ELECTRIC, INC	2386 NE 18TH TERRACE	SUITE A	GAINESVILLE	FL	32609-4417	
·						
GAINESVILLE REGIONAL UTILITIES	301 SE 4th Ave		Gainesville	FL	32601	
GAINESVILLE REGIONAL UTILITIES	P.O. BOX 147051		GAINESVILLE	FL	32614-7051	
	C/O RMC KONOVER PROPERTY					
GAINESVILLE S. C. COMPANY, LTD.	TRUST	8902 N Dale Mabry Hwy, SUITE 200	TAMPA	FL	33614	

Name	Address 1	Address 2	City	State	PostalCode	Country
GAINSVILLE FIRE RESCUE	RISK REDUCTION BUREAU	1025 NE 13TH STREET	GAINESVILLE	FL	32601	
	C/O RMC KONOVER PROPERTY					
GAINSVILLE S/C COMPANY	TRUST	8902 N DALE MABRY HWY	TAMPA	FL	33614	
GAITAN, STEPHANIE	REDACT		REDACT	REDACT	REDACT	
GALES, TERRANCE	REDACT		REDACT	REDACT	REDACT	
GALILEO APOLLO IV SUB LLC	C/O GALILEO	P.O. BOX 74644	CLEVELAND	ОН	44194	
GALILEO PARKWAY PLAZA LP, LLC	C/O CENTRO PROPERTIES GROUP	PO BOX 74644	CLEVELAND	ОН	44194	
GALVAN LOPEZ, JENIFFER	REDACT	REDACT	REDACT	REDACT	REDACT	
GALVIN, JANELIA	REDACT		REDACT	REDACT	REDACT	
GAMBLE, ANTHONY	REDACT		REDACT	REDACT	REDACT	
GAMBLE, BRENAE	REDACT	REDACT	REDACT	REDACT	REDACT	
GANLEY CHEVROLET	13865 BROOK PARK ROAD		BROOK PARK	ОН	44142	
GANT, SANTANGULA	REDACT		REDACT	REDACT	REDACT	
GANTT, BENGELA	REDACT		REDACT	REDACT	REDACT	
GARAGE DOORS OF INDIANAPOLIS	5041 WEST 96TH STREET		INDIANAPOLIS	IN	46268	
GARCIA, BIANCA	REDACT		REDACT	REDACT	REDACT	
GARCIA, EZRA	REDACT		REDACT	REDACT	REDACT	
GARCIA, GUADALUPE	REDACT		REDACT	REDACT	REDACT	
GARCIA, JASON	REDACT		REDACT	REDACT	REDACT	
GARCIA, MIGUEL	REDACT		REDACT	REDACT	REDACT	
GARCIA, MONSERRATE	REDACT		REDACT	REDACT	REDACT	
GARCIA, RAFAELA	REDACT		REDACT	REDACT	REDACT	
GARCIA, RAFAELA	REDACT		REDACT	REDACT	REDACT	
GARCIA, SHIRLENE	REDACT	REDACT	REDACT	REDACT	REDACT	
GARDNER, BRIAN	REDACT		REDACT	REDACT	REDACT	
GARDNER, JAMESHA	REDACT		REDACT	REDACT	REDACT	
GARNER, JULIETTE	REDACT	REDACT	REDACT	REDACT	REDACT	
GARNER, KEISHA	REDACT		REDACT	REDACT	REDACT	
GARNER, KELLYE	REDACT		REDACT	REDACT	REDACT	
GARNER, NAGIA	REDACT		REDACT	REDACT	REDACT	
GARRETT & TULLY, P.C.	225 S. LAKE AVE	SUITE 1400	PASADENA	CA	91101	
GARRETT, BRITTANY	REDACT		REDACT	REDACT	REDACT	
GARY SANITARY DISTRICT	3600 West 3rd Avenue		Gary	IN	46406	
GARY SANITARY DISTRICT	PO BOX 388		GARY	IN	46402	
GARY SANITATION DISTRICT	3600 West 3rd Avenue		Gary	IN	46406	
GARY SANITATION DISTRICT	P.O. BOX 388		GARY	IN	46402	

Page 71 of 176

Name	Address 1	Address 2	City	State	PostalCode	Country
GARY, JASON	REDACT		REDACT	REDACT	REDACT	
GARY, SHEILA	REDACT		REDACT	REDACT	REDACT	
GAS SOUTH, LLC	3625 Cumberland Boulevard #1500		Atlanta	GA	30339	
GAS SOUTH, LLC	P.O. BOX 530552		ATLANTA	GA	30353-0552	
GASSO HOLDING CO.	ATTN: MR. NAJAH GASSO	20320 W EIGHT MILE ROAD	SOUTHFIELD	MI	48075	
GASSO HOLDING COMPANY, LLC	Attn: Mr. Najah Gasso	20320 W Eight Mile Rd	Southfield	MI	48075	
GASTON, JAQEIA	REDACT		REDACT	REDACT	REDACT	
GATE CITY GLASS CO., INC.	1102 TROOST AVENUE		KANSAS CITY	МО	64106-3005	
GATES, BRYAN	REDACT		REDACT	REDACT	REDACT	
GATES, TAMAKIA	REDACT		REDACT	REDACT	REDACT	
GATEWAY ARTHUR, INC	PO BOX 57021		NEWARK	NJ	07101-7021	
GATEWAY CENTER ECONOMIC DEVELOPMENT PARTNERSHIP, LTD.	5184 Norwood Avenue		Jacksonville	FL	32208	
GATEWAY RETAIL CENTER, LLC	C/O TERRANOVA CORP. ATTN:A/R	801 ARTHUR GODFREY RD.	MIAMI BEACH	FL	33140	
GATOR HILLCREST PARTNERS, LLLP	1595 NE 163RD STREET		NORTH MIAMI	FL	33162	
Gator Linton Partners, Ltd.	1595E 163rd Street		North Miami Beach	FL	33162	
GATSON, JIMMY	REDACT		REDACT	REDACT	REDACT	
GAUTIER, CAROLL	REDACT		REDACT	REDACT	REDACT	
GAYLES, ANGELA	REDACT		REDACT	REDACT	REDACT	
GAYTAN, CAROL	REDACT		REDACT	REDACT	REDACT	
GAZE U.S.A. INC.	16656 S MATEO ST		LOS ANGELES	CA	90021	
GBGC HATTIESBURG RETAIL, LLC and HATTIESBURG HIGHWAY 49, LLC	9907 E Bell Road	Suite 110	Scottsdale	AZ	85260	
GC SERVICES	P.O. BOX 32500		COLUMBUS	OH	43232-0500	
GEE, TIANA	REDACT		REDACT	REDACT	REDACT	ļ
GEE, TIANA	REDACT		REDACT	REDACT	REDACT	
GENERAL SESSION COURT CLERK	P.O. BOX 3824		MEMPHIS	TN	38173	
GENERAL SESSIONS COURT CLERK GENTILE, KATHY	PO BOX 196304 REDACT		NASHVILLE REDACT	TN REDACT	37219-6304 REDACT	
GENTILLY, LLC	Edward Badough, Jr.	c/o Mr. & Mrs. Badough, Jr.	New Braunfels	TX	78131	
GENTILLY,L.L.C.	P.O. BOX 311240	,	NEW BRAUNFELS	TX.	78131-1240	

Name	Address 1	Address 2	City	State	PostalCode	Country
			,			,
GEORGE W. STEVENSON, TRUSTEE	5350 POPLAR AVENUE	SUITE 500	MEMPHIS	TN	38119-3697	
GEORGE, ROSELYNN	REDACT		REDACT	REDACT	REDACT	
GEORGE, VINCENT	REDACT		REDACT	REDACT	REDACT	
GEORGETOWN PLAZA, LP	C/O THE BROADBENT CO.	117 EAST WASHINGTON STREET	INDIANAPOLIS	IN	46204	
	GEORGETOWN PLAZA SHOPPING					
Georgetown Plaza, LP	CENTER/Store No528	117 E. Washington Street, Suite 300	Indianapolis	IN	46204	
GEORGETOWN PLAZA, LP	117 East Washington Street	Suite 300	Indianapolis	IN	46204	
Georgia Department of Revenue	Taxpayer Services Division	PO Box 105499	Atlanta	GA	30348-5499	
Georgia Department of Revenue	1800 Century Center Blvd., N.E.		Atlanta	GA	30345	
GEORGIA DEPARTMENT OF						
REVENUE	PROCESSING CENTER	PO BOX 740320	ATLANTA	GA	30374-0320	
GEORGIA DEPARTMENT OF						
REVENUE	PROCESSING CENTER	PO BOX105499	ATLANTA	GA	30348-5499	
GEORGIA DEPT OF LABOR	PO BOX 740234		ATLANTA	GA	30374-0234	
GEORGIA GLASS CONTRACTORS, INC	6487 HWY 85		RIVERDALE	GA	30274	
GEORGIA NATURAL GAS	817 W Peachtree St NW		Atlanta	GA	30308	
GEORGIA NATURAL GAS	PO BOX 105445		ATLANTA	GA	30348-5445	
GEORGIA POWER	1607 Walton Way		Atlanta	GA	30396	
GEORGIA POWER	96 ANNEX		ATLANTA	GA	30396-0001	
GERALD FORMAN, ESQ.	Attorney at Law	113 W. Bayridge Drive	Weston	FL	33326	
GERARDO M. BORRIOS	5811 BONNIE VIEW ROAD	APT 1142	DALLAS	TX	75241	
GERBER-WITHERS WHITEHAVEN	C/O THE GERBER REALTY COMPANY	1355 B LYNNFIELD ROAD	MEMPHIS	TN	38119	
GEROSKY, JEFF	REDACT		REDACT	REDACT	REDACT	
GIBBS, DOMINIQUE	REDACT		REDACT	REDACT	REDACT	
GIBBS, KYRAH	REDACT		REDACT	REDACT	REDACT	
GIBBS, RACHEL	REDACT		REDACT	REDACT	REDACT	
GIBSON, BRIDGETTE	REDACT		REDACT	REDACT	REDACT	
GIBSON, CHRISTIE	REDACT		REDACT	REDACT	REDACT	
GIBSON, CORDELIA	REDACT		REDACT	REDACT	REDACT	
GIBSON, JULIAN	REDACT		REDACT	REDACT	REDACT	
GIBSON, MATHEW	REDACT		REDACT	REDACT	REDACT	
GIDDENS, TROY	REDACT		REDACT	REDACT	REDACT	
GILBERT, DELORIS	REDACT		REDACT	REDACT	REDACT	
GILBERT, ERICA	REDACT		REDACT	REDACT	REDACT	
GILBERT, EURANETTE	REDACT		REDACT	REDACT	REDACT	

Page 73 of 176

Name	Address 1	Address 2	City	State	PostalCode	Country
GILBERT, SHQUANA	REDACT		REDACT	REDACT	REDACT	
GILLIAM, DEJONNEA	REDACT		REDACT	REDACT	REDACT	
GINA GROUP INC.	10 WEST 33RD STREET	3RD FLOOR	NEW YORK	NY	10001	
GIVENS, STEVANNA	REDACT		REDACT	REDACT	REDACT	
GLA GARNER SHOP II, LLC	c/o WRS, Inc.	550 Long Point Road	Mt. Pleasant	SC	29464	
GLADE'S PLAZA ENTERPRISES, INC	200 SOUTH MAIN ST		BELLE GLADE	FL	33430	
GLASS WORKS PLUS, LLC	5084-A PINSON VALLEY PKWY		BIRMINGHAM	AL	35215	
GLASSCO, JAMECO	REDACT		REDACT	REDACT	REDACT	
GLENN'S AUTOMATED EQUIPMENT	775 MILLER DRIVE		PELL CITY	AL	35125	
GLENWOOD CROSS COUNTRY CO.						
LLC	9525 BIRKDALE CROSSING DR	SUITE 200	HUNTERSVILLE	NC	28078	
GLENWOOD CROSS COUNTRY						
COMPANY LLC	16740 Birkdale Commons	Suite 306	Huntersville	NC	28078-4460	
GLOBAL GOLD	1410 BROADWAY		NEW YORK	NY	10018	
GLOVER, REINALDO	REDACT		REDACT	REDACT	REDACT	
GMA ACESSORIES	3 EMPIRE BLVD.		SOUTH HACKENSACK	NJ	07606	
GO GO APPAREL	1400 BROADWAY, RM#514		NEW YORK	NY	10018	
GOBERN, KHADIJAH	REDACT		REDACT	REDACT	REDACT	
GOINS, LESLIE	REDACT		REDACT	REDACT	REDACT	
GOLD MECH, INC	PO BOX 3927		AUGUSTA	GA	30914-3927	
GOLDEN TOUCH	1410 BROADWAY		NEW YORK	NY	10018	
GOLDEN WEST	16750 CHESTNUT ST		CITY OF INDUSTRY	CA	91748	
GOLDMONT REALTY CORP- SC	1360 EAST 14TH STREET	SUITE 101	BROOKLYN	NY	11230	
GOLDSTONE HOSIERY	10 W. 33RD. ST. #1006		NEW YORK	NY	10001	
GOMEZ, EDITH	REDACT		REDACT	REDACT	REDACT	
GOMEZ, SANDRA	REDACT		REDACT	REDACT	REDACT	
GONCI, BAILEY	REDACT		REDACT	REDACT	REDACT	
GONZALEZ, FERNANDO	REDACT		REDACT	REDACT	REDACT	
GONZALEZ, KIARRA	REDACT		REDACT	REDACT	REDACT	
GONZALEZ, SEBASTIAN	REDACT		REDACT	REDACT	REDACT	
GOOD STUFF APPAREL	216 N OAKHURST		BEVERLY HILLS	CA	90210	
GOODEN, BRIANA	REDACT		REDACT	REDACT	REDACT	
GOODEN, CRAIG	REDACT		REDACT	REDACT	REDACT	
GOODEN, SHANITA	REDACT		REDACT	REDACT	REDACT	
Goodman Factors	P.O. Box 29647		Dallas	TX	75229-9647	
GOODMAN GLASS CO.	3325 WEST MERCURY BLVD		HAMPTON	VA	23666	

Page 74 of 176

Name	Address 1	Address 2	City	State	PostalCode	Country
GOODSON, JOSHUA	REDACT		REDACT	REDACT	REDACT	
GOODWIN, ANDREW	REDACT		REDACT	REDACT	REDACT	
GOODWIN, ISMAEL	REDACT		REDACT	REDACT	REDACT	
GOODWIN, JOYAH	REDACT		REDACT	REDACT	REDACT	
GOOSBY, ZIPPORAH	REDACT		REDACT	REDACT	REDACT	
Gordon Brothers Retail Partners, LLC	Attn.: Michael Chartock	Prudential Tower	Boston	MA	02119-0000	
GORDON, FELISHA	REDACT		REDACT	REDACT	REDACT	
GORDON, GARFIELD	REDACT		REDACT	REDACT	REDACT	
GORDON, KALLA	REDACT	REDACT	REDACT	REDACT	REDACT	
GORDON, NICOLE	REDACT	REDACT	REDACT	REDACT	REDACT	
GORDON, TAMMY	REDACT		REDACT	REDACT	REDACT	
GOSHA, ERIC	REDACT		REDACT	REDACT	REDACT	
GOSS, TANYA	REDACT		REDACT	REDACT	REDACT	
GOULIONE, KATHLEEN	REDACT		REDACT	REDACT	REDACT	
GRABLE, LYNETTE	REDACT		REDACT	REDACT	REDACT	
GRACE, SHANEKA	REDACT	REDACT	REDACT	REDACT	REDACT	
GRACE, TEQUANDRA	REDACT		REDACT	REDACT	REDACT	
GRAHAM, ANGELA	REDACT		REDACT	REDACT	REDACT	
GRAHAM, DAWN	REDACT		REDACT	REDACT	REDACT	
GRAHAM, LATOYA	REDACT		REDACT	REDACT	REDACT	
GRAINGER	DEPT. 833492044		PALATINE	IL	60038-0001	
GRANGER, BRETTNY	REDACT		REDACT	REDACT	REDACT	
GRANT, II, MELVIN	REDACT		REDACT	REDACT	REDACT	
GRANVILLE, YAKERIA	REDACT		REDACT	REDACT	REDACT	
GRAVES, JEREMY	REDACT		REDACT	REDACT	REDACT	
GRAVES, MARILYN	REDACT		REDACT	REDACT	REDACT	
GRAY, IESHA	REDACT	REDACT	REDACT	REDACT	REDACT	
GRAY, LAQUEASHA	REDACT		REDACT	REDACT	REDACT	
GRAY, MAURICE	REDACT		REDACT	REDACT	REDACT	
GRAY, QUENTERRIUS	REDACT		REDACT	REDACT	REDACT	
GRAY, SHANTANEYA	REDACT		REDACT	REDACT	REDACT	
GRAYSON CONSTRUCTION, LLC.	1401 MONTGOMERY HIGHWAY	SUITE 19	BIRMINGHAM	AL	35216	
Great American Insurance Group						
(IMP 2-04-51-19-04)	301 E Fourth Street		Cincinnati	ОН	45202	
GREAT LAKES						
ELECTRICALMAINTENANCE, INC.	3817 HAYES ST.		GARY	IN	46408-2026	

Name	Address 1	Address 2	City	State	PostalCode	Country
Great Lakes Reinsurance (UK) PLC						
(E15NV03970)	Plantation Place	30 Fenchurch Street	London		EC3M 3AJ	United Kingdom
GREATER CINCINNATI						
WATERWORKS	4747 SPRING GROVE AVENUE		CINCINNATI	ОН	45232	
GREATER CINCINNATI						
WATERWORKS	PO BOX 742505		CINCINNATI	ОН	45274-2505	
GREATWIDE AMERICAN TRANS						
FREIGHT	PO BOX 405828		ATLANTA	GA	30384-5828	
GRECCO, JOY	REDACT		REDACT	REDACT	REDACT	
GREEN, ARLETTE	REDACT		REDACT	REDACT	REDACT	
GREEN, CLARA	REDACT		REDACT	REDACT	REDACT	
GREEN, LEONARD	REDACT		REDACT	REDACT	REDACT	
GREEN, MICHELLE	REDACT		REDACT	REDACT	REDACT	
GREEN, RICKESHA	REDACT		REDACT	REDACT	REDACT	
GREEN, SHAKESHA	REDACT		REDACT	REDACT	REDACT	
GREEN, TAJSHEENA	REDACT		REDACT	REDACT	REDACT	
GREENE & COOPER, LLP	615 COLONIAL PARK DRIVE	SUITE 104	ROSWELL	GA	30075	
GREENHIVES, JAMNISHA	REDACT	REDACT	REDACT	REDACT	REDACT	
Greenville County	Attn: Consumer Protection Agency GREENVILLE COUNTY TAX	Mark W. Tollison, County Attorney	Greenville	sc	29601	
GREENVILLE COUNTY SC	COLLECTOR	DEPARTMENT 390	COLUMBIA	sc	29202-3221	
GREENVILLE WATER	407 W Broad St		GREENVILLE	SC	29601	
GREENVILLE WATER	PO BOX 687		GREENVILLE	SC	29602-0687	
Greenwood Station LLC	GREENWOOD WEST/Store No99	c/o Phillip Edison & Co.	Cincinnati	ОН	45242	
GREENWOOD STATION, LLC	PO BOX 203137		DALLAS	TX	75320-3137	
GREENWOOD STATION, LLC	c/o Phillips Edison & Co.	11690 Grooms Road	Cincinnati	ОН	45242	
GREENWOOD UTILITIES	101 Wright Place		Greenwood	MS	38930	
GREENWOOD UTILITIES	PO BOX 866		GREENWOOD	MS	38935-0866	
GREGG, KENDRA	REDACT		REDACT	REDACT	REDACT	
GREGORY PEST SOLUTIONS	P.O. BOX 6713		GREENVILLE	SC	29606-6713	
GRENADA LOANS, INC	237 S. MAIN STREET		GRENADA	MS	38901	
GRIESHABER, JANET	REDACT		REDACT	REDACT	REDACT	
GRIFFIN ELECTRIC	P.O. BOX 1545		GREER	SC	29652	
GRIFFIN, DONNA	REDACT		REDACT	REDACT	REDACT	
GRIFFIN, LATANYA	REDACT		REDACT	REDACT	REDACT	
GRIFFIN, SHAWN	REDACT		REDACT	REDACT	REDACT	
GRIFFIN, TONY	REDACT		REDACT	REDACT	REDACT	

Page 76 of 176

Name	Address 1	Address 2	City	State	PostalCode	Country
GRIGGS, KEIARA	REDACT	REDACT	REDACT	REDACT	REDACT	
GROOMES, CHANITA	REDACT		REDACT	REDACT	REDACT	
GROSS, LARISSIA	REDACT		REDACT	REDACT	REDACT	
GROSS, TAYVIN	REDACT		REDACT	REDACT	REDACT	
GROTE, TRACEY	REDACT	REDACT	REDACT	REDACT	REDACT	
GRUBBS, JASMINE	REDACT	REDACT	REDACT	REDACT	REDACT	
GS II OXFORD COMMONS LLC	DEPT #314434-20338-35216	PO BOX 92320	CLEVELAND	ОН	44193	
GUARANTY BANK	PO Box 240200		Milwaukee	WI	53224-9010	
GUARANTY LOAN & REAL ESTATE	P.O. BOX 2208		WEST MEMPHIS	AR	72303-2208	
GUARDIAN LIFE (824480)	PO BOX 824480		PHILADELPHIA	PA	19182-4480	
GUERRA, SARAH	REDACT	REDACT	REDACT	REDACT	REDACT	
GUERRERO, MIAH	REDACT	REDACT	REDACT	REDACT	REDACT	
GUERRERO, MIAH	REDACT	REDACT	REDACT	REDACT	REDACT	
GULF INLAND, LLC	PO BOX 6058		MONROE	LA	71211	
GULF INLAND, LLC	P O BOX 6058		MONROE	LA	71211	
GULF POWER CO.	9105 Bergenline Avenue		Birmingham	AL	07047	
GULF POWER CO.	P.O. BOX 830660		BIRMINGHAM	AL	35283-0660	
GURECKI, JILL	REDACT		REDACT	REDACT	REDACT	
GUTIERREZ, ELVIA	REDACT		REDACT	REDACT	REDACT	
GUY, NICHELLE	REDACT		REDACT	REDACT	REDACT	
GUYTON, STEPHANIE	REDACT		REDACT	REDACT	REDACT	
GUZMAN, SANDRA	REDACT		REDACT	REDACT	REDACT	
GW&T, LLC	FIRST COMMERCIAL	27600 NWESTERN HWY, STE 200	SOUTHFIELD	MI	48034	
GW&T, LLC	27600 Northwestern Highway	Suite 200	Southfield	MI	48034	
H G HILL REALTY CO.	3011 ARMORY DRIVE	SUITE 130	NASHVILLE	TN	37204	
H.G. HILL REALTY COMPANY, LLC	3011 Armory Drive	Suite 130	Nashville	TN	37204-3721	
HAAS, JOHNATHAN	REDACT		REDACT	REDACT	REDACT	
HADLEY, MIKENA	REDACT		REDACT	REDACT	REDACT	
HAGGINS, STEPHEN	REDACT	REDACT	REDACT	REDACT	REDACT	
HAIRSTON, ANDRE	REDACT		REDACT	REDACT	REDACT	
HAITHCOCK, ADA	REDACT		REDACT	REDACT	REDACT	
HAL SYSTEMS, INC	PO BOX 1015		NEWNAN	GA	30264-1015	
HALL, ANGELA	REDACT		REDACT	REDACT	REDACT	
HALL, BRITTANY	REDACT		REDACT	REDACT	REDACT	
HALL, CHARLIE	REDACT		REDACT	REDACT	REDACT	
HALL, CHRISTINA	REDACT		REDACT	REDACT	REDACT	
HALL, ERICA	REDACT		REDACT	REDACT	REDACT	

Name	Address 1	Address 2	City	State	PostalCode	Country
HALL, JOHN	REDACT		REDACT	REDACT	REDACT	
HALL, JOSHUA	REDACT		REDACT	REDACT	REDACT	
HALL, KIMBERLY	REDACT		REDACT	REDACT	REDACT	
HALL, LATAMARA	REDACT		REDACT	REDACT	REDACT	
HALL, LEKETA	REDACT		REDACT	REDACT	REDACT	
HALL, MEA	REDACT		REDACT	REDACT	REDACT	
HALL, MICHAEL	REDACT		REDACT	REDACT	REDACT	
HALL, MICHELLE	REDACT		REDACT	REDACT	REDACT	
HALL, SHEMIKA	REDACT		REDACT	REDACT	REDACT	
HALLWOOD PLAZA, INC.	3371 ORCHARD LAKE RD		KEEGO HARBOR	MI	48320	
HALLWOOD PLAZA, INC.	20411 W. 12 Mile Road	Suite 201	Southfield	MI	48076	
HALPERN ENTERPRISES INC.	5269 BUFORD HWY		ATLANTA	GA	30340	
HALPERN ENTERPRISES, INC	5269 BUFORD HIGHWAY		ATLANTA	GA	30340	
HALPERN ENTERPRISES, INC	5269 BUFORD HWY		ATLANTA	GA	30340	
HALPERN ENTERPRISES,INC.	5269 BUFORD HIGHWAY		ATLANTA	GA	30340	
HAM, JACQUETTA	REDACT		REDACT	REDACT	REDACT	
HAMILTON CO MUNIC CIVIL COURT	1000 MAIN ST ROOM 115	DO DOV 44047	CINCINNATI	ОН	45202	
HAMILTON COUNTY, TN	TRUSTEE	PO BOX 11047	CHATTANOOGA	TN	37401	
HAMILTON LOCK & SAFE, INC.	12217 NW 8TH PLACE		NEWBERRY	FL	32669	
HAMILTON, CYNTHIA	REDACT		REDACT	REDACT	REDACT	
HAMILTON, JEFFERY	REDACT		REDACT	REDACT	REDACT	
HAMILTON, JULIUS	REDACT		REDACT	REDACT	REDACT	
HAMILTON, LAKEISHA	REDACT		REDACT	REDACT	REDACT	
HAMILTON, SHARI	REDACT		REDACT	REDACT	REDACT	
HAMMILL, CHUCK	REDACT	REDACT	REDACT	REDACT	REDACT	
HAMMILL, DONNA	REDACT		REDACT	REDACT	REDACT	
HAMMOND CITY COURT	5925 CALUMET AVENUE		HAMMOND	IN	46320	
HAMMONDS, KIMBERLY	REDACT		REDACT	REDACT	REDACT	
Hampton City	Attn: Consumer Protection Agency	Vanessa T. Valldejuli - City Attorney	Hampton	VA	23669	
HAMPTON, BRANDON	REDACT	in the state of th	REDACT	REDACT	REDACT	1
HAMPTON, JAMES	REDACT		REDACT	REDACT	REDACT	1
HANCOCK BANK	PO Box 4019		Gulfport	MS	39502	1
HAND, LATRICA	REDACT		REDACT	REDACT	REDACT	1
HANDY ELECTRICIAN	881 MEMORIAL DRIVE	(UNIT308)	ATLANTA	GA	30316	1
HANDY, CHRISTOPHER	REDACT	/	REDACT	REDACT	REDACT	1
HANGERS UNLIMITED	5000 W. ELECTRIC AVENUE		WEST MILWAUKEE	WI	53219	1

Page 78 of 176

Name	Address 1	Address 2	City	State	PostalCode	Country
HANNAH SLAMEN	5225 7TH AVE S		BIRMINGHAM	AL	35212	
HANNER, CASSANDRA	REDACT		REDACT	REDACT	REDACT	
HANS JOHNSON PLUMBING & AIR,						
LLC	204 S.W. 16 COURT		FT LAUDERDALE	FL	33315	
HANSCO	PO BOX 75171		HOUSTON	TX	77234-5171	
HARDEN, CHI'KEYERRA	REDACT	REDACT	REDACT	REDACT	REDACT	
HARDISON, BYRAN	REDACT		REDACT	REDACT	REDACT	
HARDY, ANTONIO	REDACT		REDACT	REDACT	REDACT	
HARDY, TAJ'SHEENA	REDACT		REDACT	REDACT	REDACT	
HARDY, YVETTE	REDACT		REDACT	REDACT	REDACT	
HARMAN, BRANDON	REDACT		REDACT	REDACT	REDACT	
HARMON, GARIA	REDACT		REDACT	REDACT	REDACT	
HAROLD J BERKLEY, JR.	CHAPTER 13 TRUSTEE	P.O. BOX 321454	FLOWOOD	MS	39232	
HAROLD WHITTEN	113 MANGROVE DRIVE		ALABASTER	AL	35007	
HARPER, LESLIE	REDACT	REDACT	REDACT	REDACT	REDACT	
HARPER, NIJA	REDACT		REDACT	REDACT	REDACT	
HARPER, TAQUANA	REDACT		REDACT	REDACT	REDACT	
HARRELL, DAYGIAQUE	REDACT		REDACT	REDACT	REDACT	
HARRIS BANK	PO Box 4320		Carol Stream	IL	60197-4320	
Harris County	Attn: Consuimer Protection Agency	Vince Ryan, County Attorney	Houston	TX	77002	
HARRIS COUNTY	TAX ASSESSOR	P.O.BOX 4622	HOUSTON	TX	77210-4622	
HARRIS COUNTY ALARM DETAIL	P.O. BOX 4049		HOUSTON	TX	77210-4049	
HARRIS COUNTY MUD #119	TAX ASSESSOR/COLLECTOR	P.O. BOX 925928	HOUSTON	TX	77292-5928	
HARRIS, CHRISTINA	REDACT		REDACT	REDACT	REDACT	
HARRIS, CORTEZ	REDACT	REDACT	REDACT	REDACT	REDACT	
HARRIS, DAVARTHUR	REDACT		REDACT	REDACT	REDACT	
HARRIS, ESSENCE	REDACT		REDACT	REDACT	REDACT	
HARRIS, JR., JERRY	REDACT		REDACT	REDACT	REDACT	
HARRIS, KAISSHA	REDACT		REDACT	REDACT	REDACT	
HARRIS, KIM	REDACT		REDACT	REDACT	REDACT	
HARRIS, KIMBERLY	REDACT		REDACT	REDACT	REDACT	
HARRIS, LUTETIE	REDACT		REDACT	REDACT	REDACT	
HARRIS, MARY	REDACT		REDACT	REDACT	REDACT	
HARRIS, RACHEL	REDACT		REDACT	REDACT	REDACT	
HARRIS, SANTANA	REDACT		REDACT	REDACT	REDACT	
HARRIS, SHEILA	REDACT		REDACT	REDACT	REDACT	
HARRIS, TIFFANY	REDACT	REDACT	REDACT	REDACT	REDACT	

Page 79 of 176

Name	Address 1	Address 2	City	State	PostalCode	Country
HARRIS-JOHNSON, ANTOINETTE	REDACT		REDACT	REDACT	REDACT	
HARRISON, BARBARA	REDACT		REDACT	REDACT	REDACT	
HARRISON, BEVERLY	REDACT	REDACT	REDACT	REDACT	REDACT	
HARRISON, JEREMY	REDACT		REDACT	REDACT	REDACT	
HARRISON, OLIVER	REDACT	REDACT	REDACT	REDACT	REDACT	
HARRISON, SASHA	REDACT		REDACT	REDACT	REDACT	
HART, DAMION	REDACT		REDACT	REDACT	REDACT	
HART, TERENCE	REDACT		REDACT	REDACT	REDACT	
HARTLEY, CARLA	REDACT	REDACT	REDACT	REDACT	REDACT	
HARTMAN INCOME	2909 Hillcroft	Suite 420	Houston	TX	77057	
HARTMAN INCOME	C/O HARTMAN MGMT	2909 Hillcroft St.	HOUSTON	TX	77057	
HARTMAN INCOME REIT PROPERTY						
HOLDING	C/O HARTMAN MGMT	2909 HILLCROFT STREET	HOUSTON	TX	77057	
HARVEY, BOBBY	REDACT	REDACT	REDACT	REDACT	REDACT	
HARVEY, JANICE	REDACT		REDACT	REDACT	REDACT	
HARVEY, JESSYCA	REDACT	REDACT	REDACT	REDACT	REDACT	
HARVEY-NEELY, RHONDA	REDACT		REDACT	REDACT	REDACT	
HARWELL, HELEN	REDACT		REDACT	REDACT	REDACT	
HASLAM, DONNA	REDACT		REDACT	REDACT	REDACT	
HATCHER, BRITTANY	REDACT		REDACT	REDACT	REDACT	
HATTEN, SYDNEY	REDACT		REDACT	REDACT	REDACT	
HAUCK HOLDING NEW ORLEANS, LLC	7720 Montgomery Road	1st Floor	Cincinnati	ОН	45236	
HAWKINS, ALISHIA	REDACT		REDACT	REDACT	REDACT	
HAWKINS, MARKITA	REDACT		REDACT	REDACT	REDACT	
HAWKINS, SARAH	REDACT	REDACT	REDACT	REDACT	REDACT	
HAWKINS, TASHA	REDACT		REDACT	REDACT	REDACT	
HAWKINS, THADDAEUSS	REDACT		REDACT	REDACT	REDACT	
HAYES, MAURICIA	REDACT		REDACT	REDACT	REDACT	
HAYES, VICTORIA	REDACT		REDACT	REDACT	REDACT	
HAYS, T'KEYAH	REDACT		REDACT	REDACT	REDACT	
HAZAN IMPORT CORP.	P.O. BOX 1279		NEW YORK	NY	10056	
HAZLEY, ROBERTA	REDACT		REDACT	REDACT	REDACT	
HB LOGISTICS	RTS FINANCIAL SERVICE	PO BOX 840267	DALLAS	TX	75284-0267	
HB LOGISTICS, LLC	PO BOX 667		FULTONDALE	AL	35068	
HD NEWMARKET, LLC	ATTN: SHANE MCGROARTY	5310 S. ALSTON AVE	DURHAM	NC	27713	
	Harvey Lindsay Commercial Real					
HD NEWMARKET, LLC	Estate	701 Town Center Drive	Newport News	VA	23606	

Page 80 of 176

Name	Address 1	Address 2	City	State	PostalCode	Country
HEARD, JASMYNE	REDACT		REDACT	REDACT	REDACT	
HEARTS N HIPS	2424 E 26TH		VERNON	CA	90058	
HEDGESPETH, DYNISHA	REDACT		REDACT	REDACT	REDACT	
HELLAMS, RODNEY	REDACT		REDACT	REDACT	REDACT	
HELMI, SAHAR	REDACT		REDACT	REDACT	REDACT	
HEMINGWAY, DIASIA	REDACT	REDACT	REDACT	REDACT	REDACT	
HEMPHILL, DAKHARIA	REDACT		REDACT	REDACT	REDACT	
HENDERSON, BRITTANY	REDACT		REDACT	REDACT	REDACT	
HENDERSON, KARLTON	REDACT		REDACT	REDACT	REDACT	
HENDERSON, REGINA	REDACT		REDACT	REDACT	REDACT	
HENERY, TAYLOR	REDACT		REDACT	REDACT	REDACT	
HENGEHOLD, AMANDA	REDACT		REDACT	REDACT	REDACT	
HENRICH, LYNN	REDACT		REDACT	REDACT	REDACT	
HENRY, AMANDA	REDACT		REDACT	REDACT	REDACT	
HENRY, INDIA	REDACT		REDACT	REDACT	REDACT	
HENRY, TYTIANA	REDACT		REDACT	REDACT	REDACT	
HENRY'S LOCK & KEY SERVICE	PO BOX 82135		TAMPA	FL	33682-2135	
HENTHORN, JUSTIN	REDACT		REDACT	REDACT	REDACT	
HENTHORN, ROBERT	REDACT		REDACT	REDACT	REDACT	
HERBERT BARRETT, JR.	2492 PEEKSKILL DRIVE		COLUMBUS	ОН	43219	
HERITAGE GMC-BUICK	930 EAST I-30	P.O. BOX 6	ROCKWALL	TX	75087	
HERITAGE LOGISTICS LLC	P.O. BOX 125		SYLACAUGA	AL	35150	
HERMITAGE PLACE LP	P.O. BOX 648		CONCORD	NC	28026-0648	
HERNANDEZ, ALEX	REDACT		REDACT	REDACT	REDACT	
HERNANDEZ, ANGIE	REDACT		REDACT	REDACT	REDACT	
HERNANDEZ, CAROL	REDACT		REDACT	REDACT	REDACT	
HERNANDEZ, ELIZABETH	REDACT	REDACT	REDACT	REDACT	REDACT	
HERNANDEZ, JORGE	REDACT		REDACT	REDACT	REDACT	
HERTZ CAR SALES	450 WILLIAM F. MCCLELLAN HWY		E.BOSTON	MA	02128	
HFDB, LLC	C/O CHARLES WAYNE PROPERTIES	444 SEABREEZE BLVD, STE 1000	DAYTONA BEACH	FL	32118-3953	
HICKMAN, HELENA	REDACT		REDACT	REDACT	REDACT	
HICKS, ALESIA	REDACT		REDACT	REDACT	REDACT	
HICKS, LATIFAH	REDACT		REDACT	REDACT	REDACT	
HICKS, LISA	REDACT		REDACT	REDACT	REDACT	
HICKS, REMEKIA	REDACT	REDACT	REDACT	REDACT	REDACT	
HICKSON, LATONYA	REDACT	REDACT	REDACT	REDACT	REDACT	

Page 81 of 176

Name	Address 1	Address 2	City	State	PostalCode	Country
HICKSON, TYLISHA	REDACT		REDACT	REDACT	REDACT	
HIGDON PAPER & PACKAGING	P.O. BOX 1865		PELHAM	AL	35124	
HIGGINS, ASHLEY	REDACT		REDACT	REDACT	REDACT	
HIGHLAND LAKES CENTER LLC	PO BOX 643342		PITTSBURGH	PA	15264-3342	
	C/O AC COMMERCIAL PROPERTY					
HIGHLAND LAKES CENTER, LLC	MGMT	1350 Orange Avenue,	Winter park	FL	32789	
	C/O AC COMMERCIAL PROPERTY					
HIGHLAND LAKES PROPERTY, LLC	MGMT	1350 ORANGE AVE	WINTER PARK	FL	32789	
HIGHWAY JEANS	1407 BROADWAY	SUITE 1405	NEW YORK	NY	10018	
Hilco Merchant Resources, LLC	Attn.: Ian S. Fredericks	5 Revere Drive, Suite 206	Northbrook	IL	60062	
HILL, ANGENITA	REDACT		REDACT	REDACT	REDACT	
HILL, ANNIE	REDACT		REDACT	REDACT	REDACT	
HILL, BILLY	REDACT		REDACT	REDACT	REDACT	
HILL, GREGORY	REDACT		REDACT	REDACT	REDACT	
HILL, GRE'SHAWN	REDACT	REDACT	REDACT	REDACT	REDACT	
HILL, JENNIFER	REDACT		REDACT	REDACT	REDACT	
HILL, JOHN KINNARD	REDACT		REDACT	REDACT	REDACT	
HILL, KARAH	REDACT		REDACT	REDACT	REDACT	
HILL, REGINA	REDACT		REDACT	REDACT	REDACT	
HILLIARD, MARILYN	REDACT		REDACT	REDACT	REDACT	
Hillsborough County	Attn: Consumer Protection Agency	Chip Fletcher - County Attorney	Tampa	FL	33602	
HILLSBOROUGH COUNTY	TAX COLLECTOR	PO BOX 172920	TAMPA	FL	33672-0920	
HILLSBOROUGH COUNTY TAX						
COLLECTOR	PO BOX 30009		TAMPA	FL	33630-0009	
Hinds County	Attn: Consumer Protection Agency	Sherri M. Flowers, County Attorney	Jackson	MS	39201	
HINDS COUNTY TAX COLLECTOR	P.O. BOX 1727		JACKSON	MS	39215-1727	
HINSHAW & CULBERTSON	8142 SOLUTIONS CENTER DR.		CHICAGO	IL	60677-8001	
HINTON, DEANNA	REDACT		REDACT	REDACT	REDACT	
HIRERIGHT SOLUTIONS, INC.	P.O. BOX 847783		DALLAS	TX	75284-7783	
HODGE, DARIUS	REDACT		REDACT	REDACT	REDACT	
HODGE, DONNALYN	REDACT	REDACT	REDACT	REDACT	REDACT	
HODGES, PATRICIA	REDACT		REDACT	REDACT	REDACT	
HOFFMAN & HARPST CO., INC.	642 E. WOODRUFF AVE.		TOLEDO	ОН	43604	
HOGAN, PETER	REDACT		REDACT	REDACT	REDACT	
HOGROE, LADACHE	REDACT		REDACT	REDACT	REDACT	
HOGUE, DEBRA	REDACT		REDACT	REDACT	REDACT	

Page 82 of 176

Name	Address 1	Address 2	City	State	PostalCode	Country
HOLDER, VANESSA	REDACT	REDACT	REDACT	REDACT	REDACT	
HOLLEY, MONTESHA	REDACT		REDACT	REDACT	REDACT	
HOLLINS, ANTWON	REDACT		REDACT	REDACT	REDACT	
HOLLINS, SHANNON	REDACT	REDACT	REDACT	REDACT	REDACT	
HOLLOWELL, CHARITY	REDACT		REDACT	REDACT	REDACT	
HOLMAN, TARA	REDACT		REDACT	REDACT	REDACT	
HOLMES, MARY	REDACT		REDACT	REDACT	REDACT	
HOLSEY, APRIL	REDACT	REDACT	REDACT	REDACT	REDACT	
HOLSEY, AUDRIANNA	REDACT		REDACT	REDACT	REDACT	
	CORPORATE INTERNSHIP					
HOLY FAMILY CRISTO REY	PROGRAM	PO BOX 8070 ENSLEY	BIRMINGHAM	AL	35218	
HOME DEPOT U.S.A. INC	PO BOX 7247-7491		PHILADELPHIA	PA	19170-7491	
HOME DEPOT U.S.A., INC.	Jennifer M. Evans, Atty	2455 Paces Ferry Road, N.W.	Atlanta	GA	30339	
HONG KONG NORTHWEST						
INVESTMENT LLC	9720 TOWN PARK DR., #180		HOUSTON	TX	77036	
HOOD, KADEL	REDACT		REDACT	REDACT	REDACT	
HOOD, KRISTI	REDACT		REDACT	REDACT	REDACT	
HOOD, STARNESHA	REDACT		REDACT	REDACT	REDACT	
HOOKER, MICHELLE	REDACT		REDACT	REDACT	REDACT	
HOOVER ELECTRIC, INC	2123 W. 8 MILE ROAD		DETROIT	MI	48203	
HOPE, TYFFANI	REDACT		REDACT	REDACT	REDACT	
HOPKINS, LISA	REDACT		REDACT	REDACT	REDACT	
HOPSON, MELISSA	REDACT		REDACT	REDACT	REDACT	
HORNE, LATONIA	REDACT		REDACT	REDACT	REDACT	
HORNSBY, JACKIE	REDACT		REDACT	REDACT	REDACT	
HORVATH, JOSH	REDACT		REDACT	REDACT	REDACT	
HOSIERY NETWORK	10 W. 33RD ST.	SUITE 1209	NEW YORK	NY	10001	
HOT CHOCOLATE, INC.	P.O. BOX 111702		LOS ANGELES	CA	90011	
HOT GINGER	801 E. 7TH ST.		LOS ANGELES	CA	90021	
HOT SHOT	1407 BROADWAY, RM 2018		NEW YORK	NY	10018	
HOUSTON COUNTY, AL	REVENUE COMMISSIONER	PO BOX 6406	DOTHAN	AL	36302-6406	
HOUSTON COUNTY, GA	TAX COMMISSIONER	P.O. BOX 7799	WARNER ROBINS	GA	31095-7799	
HOUSTON R.E. INCOME PROPERTIES		1450 West Sam Houston Parkway				
XVII, LTD.	C/O Hartman Management, Inc.	North	Houston	TX	77043	
HOUSTON, DENISE	REDACT		REDACT	REDACT	REDACT	
HOUSTON, KENYATTA	REDACT		REDACT	REDACT	REDACT	
HOWARD, ELLEN	REDACT		REDACT	REDACT	REDACT	
HOWARD, GREGORY	REDACT		REDACT	REDACT	REDACT	

Page 83 of 176

Name	Address 1	Address 2	City	State	PostalCode	Country
HOWARD, JESTINA R.	REDACT		REDACT	REDACT	REDACT	
HOWARD, KATIA	REDACT		REDACT	REDACT	REDACT	
HOWARD, KIERRA	REDACT		REDACT	REDACT	REDACT	
HOWARD, LATOYA	REDACT		REDACT	REDACT	REDACT	
HOWARD, ROSA	REDACT		REDACT	REDACT	REDACT	
HOWARD, TEAISHA	REDACT		REDACT	REDACT	REDACT	
HOWELL, SHAYTICE	REDACT		REDACT	REDACT	REDACT	
HOYE, DESTINEE	REDACT		REDACT	REDACT	REDACT	
HP II, LLC	C/O FIRST COMMERCIAL	27600 NORTHWESTERN HWY	SOUTHFIELD	MI	48034	
HP II, LLC	C/O FIRST COMMERCIAL REALTY	27600 Northwestern Hwy	Southfeld	MI	48034	
HRI REALTY SERVICES, INC	ATTN: RAY SPADOFORA	909 POYDRAS STREET	NEW ORLEANS	LA	70112	
HRSD	1434 Air Rail Avenue		Virginia Beach	VA	23455	
HRSD	PO BOX 71092		CHARLOTTE	NC	28272-1092	
	C/O HT SKYLINE SEVEN REAL					
HT RETAIL, LLC	ESTATE, LLC	5825 GLENRIDGE DR	ATLANTA	GA	30328	
	C/O SKYLINE SEVEN REAL ESTATE,					
HT RETAIL, LLC	LLC	5825 Glenridge Dr. Bld. 1	Atlanta	GA	30328	
HUBBARD, ESTHER	REDACT	REDACT	REDACT	REDACT	REDACT	
HUBBARD, NATASHA	REDACT		REDACT	REDACT	REDACT	
HUDGENS, TYEISHA	REDACT		REDACT	REDACT	REDACT	
HUDSON, DIONNA	REDACT		REDACT	REDACT	REDACT	
HUDSON, QIANA	REDACT		REDACT	REDACT	REDACT	
HUDSON, REGINA	REDACT		REDACT	REDACT	REDACT	
HUDSON-WALLACE, DEISJA	REDACT	REDACT	REDACT	REDACT	REDACT	
HUELL, CHARLES	REDACT		REDACT	REDACT	REDACT	
HUEY, GERA	REDACT		REDACT	REDACT	REDACT	
HUGGINS, KASIL	REDACT		REDACT	REDACT	REDACT	
HUGHES, BRIANCA	REDACT		REDACT	REDACT	REDACT	
HUGHES, CHIEKO	REDACT		REDACT	REDACT	REDACT	
HUGHES, CLAUDETTE	REDACT		REDACT	REDACT	REDACT	
HUMPHREY, CONTASHIA	REDACT		REDACT	REDACT	REDACT	
HUMPHREY, CORTASHIA	REDACT		REDACT	REDACT	REDACT	
HUNT, AUSTIN	REDACT		REDACT	REDACT	REDACT	
HUNT, IDA	REDACT		REDACT	REDACT	REDACT	
HUNTER, SHAKERAH	REDACT		REDACT	REDACT	REDACT	
HUNTER, SHAKERAH	REDACT		REDACT	REDACT	REDACT	
HUNTINGTON BANK	41 South High Street		Columbus	ОН	43287	

Name	Address 1	Address 2	City	State	PostalCode	Country
			,			
HUNTSVILLE GLASS COMPANY, INC.	2201 HOLMES AVE. NW		HUNTSVILLE	AL	35816	
HUNTSVILLE UTILITIES	112 Spragins Street		Huntsville	AL	35895	
HURLEY BOARD OF HOSPITAL MGS.	1038 S GRAND TRAVERSE		FLINT	MI	48502	
HURT, TAKIYA	REDACT		REDACT	REDACT	REDACT	
HUTCHINSON, JENNIFER	REDACT		REDACT	REDACT	REDACT	
HYATT LEGAL PLANS, INC	PO BOX 714893		COLUMBUS	ОН	43271-4893	
HYDRANT GROUP	1407 BROADWAY		NEW YORK	NY	10018	
HYMER, KATHRYN	REDACT		REDACT	REDACT	REDACT	
HYNES, JARED	REDACT		REDACT	REDACT	REDACT	
HYNES, KRISTINA	REDACT	REDACT	REDACT	REDACT	REDACT	
HYP HATS	10 E 34TH STREET, 6TH FLOOR		NEW YORK	NY	10016	
I LOVE ACCESSORIES	10 W 33RD	RM 1206	NEW YORK	NY	10001	
I SHALOM / NY ACCESSORY GROUP	411 5TH AVE		NEW YORK	NY	10016	
IBARRA, GRISELDA	REDACT		REDACT	REDACT	REDACT	
IBARRA, GRISELDA	REDACT		REDACT	REDACT	REDACT	
IBERIA BANK	PO Box 7299		Little Rock	AR	72217-7299	
IBERIA BANK	PO BOX 13740		NEW IBERIA	LA	70562-3740	
IberiaBank, a Louisiana banking						
corporation	Attn.: Loan Operations	1101 East Admiral Doyle Dr	New Iberia	LA	70562-2440	
Iberville Parish Sales Tax Department	PO Box 355		Plaquemine	LA	70765-0335	
IJ MODE INC	6043 MALBURG WAY		VERNON	CA	90058	
IL DEPT OF EMPLOYMENT SECURITY	PO BOX 19299		SPRINGFIELD	IL	62794-9299	
ILLINOIS AMERICAN WATER	300 North Water Works Drive		Belleville	IL	62223	
ILLINOIS AMERICAN WATER	PO BOX 94551		PALATINE	IL	60094-4551	
Illinois Department of Revenue	Sales Tax Processing Division	PO Box 19014	Springfield	IL	62764-9014	
Illinois Department of Revenue	PO BOX 1040		GALESBURG	IL	61402-1040	
ILLINOIS DEPT OF REVENUE	SALES TAX PROCESSING DIVISION	P.O. BOX 19014	SPRINGFIELD	IL	62794-9014	1
ILLINOIS SECRETARY OF STATE	DEPT OF BUSINESS SERVICES	LIMITED LIABILITY DIVISION	SPRINGFIELD	IL	62756	
IMAGE TECHNOLOGY SOLUTIONS	5 BROWN CIRCLE		ALABASTER	AL	35007	
IMAGE WORKS	100 CENTURY PARK SOUTH	SUITE 104	BIRMINGHAM	AL	35226	
ImageWorks	Attn.: Rick Roth	100 Century Park South Suite 104	Birmingham	AL	35228	1

Name	Address 1	Address 2	City	State	PostalCode	Country
IMPULSE OFF PRICE APPAREL	1120 E. PICO BLVD.		LOS ANGELES	CA	90021	
IN TOUCH FOOTWEAR, INC.	1100 HATCHER AVE, STE C		CITY OF INDUSTRY	CA	91748	
INCORRUPTIBLE LOVE	1827 CATLIN STREET		FULLERTON	CA	92833	
INDEPENDENCE BUSINESS SUPPLY	4550 HINCKLEY PARKWAY		CLEVELAND	ОН	44109	
INDIAN TRAIL SQUARE, LLC	321 HENRY STREET		LEXINGTON	KY	40508	
INDIANA AMERICAN WATER	555 East County Line Road	Suite 201	Greenwood	IN	46143	
INDIANA AMERICAN WATER	PO BOX 94551		PALATINE	IL	60094-4551	
Indiana Department of Revenue	PO Box 7218		Indianapolis	IN	46207-7218	
Indiana Department of Revenue	1025 Widener Ln		South Bend	IN	46614	
INDIANA DEPT OF REVENUE	SALES TAX EARLY FILER	P.O. BOX 6074	INDIANAPOLIS	IN	46206-6074	
INDIANA DEPT OF WORKFORCE DEV	PO BOX 847		INDIANAPOLIS	IN	46206-0847	
INDIANA PARTNERS, LLC	C/O PAM AM EQUITIES INC	ATTN: D.IWANIER	NEW YORK	NY	10022	
	THE COMMONS ROUTE 20/Store					
Indiana Partners, LLC	No207	4369 West 5th Avenue	Gary	IN	46406	
	C/O PAM AM EQUITIES INC ATTN:					
INDIANA PARTNERS, LLC	D.IWANIER	18 E. 50th St.,	New York	NY	10022	
INDUSTRIAL FIRE PROTECTION, INC.	P.O. BOX 8831		COLUMBUS	MS	39705	
INDUSTRIAL STAFFING OF	INDUSTRIAL STAFFING OF					
ALABAMA(PORTER)	ALABAMA	ATTENTION: NADIA, INC.	BIRMINGHAM	AL	35202	
INDYA JOHNSON	4131 EAGLE CREST DR.		BIRMINGHAM	AL	35242	
INLAND REISTERTOWN SPE I, LLC	NW 6137	PO BOX 1450	MINNEAPOLIS	MN	55485-6137	
INLAND WESTERN MILWAUKEE						
MIDTOWN LLC	NW6141 (LEASE #19364)	PO BOX 1450	MINNEAPOLIS	MN	55485-6141	
INNOMARK COMMUNICATIONS	P.O. BOX 636542		CINCINNATI	ОН	45263-6542	
INTEGRITY SERVICES USA, INC	12241 S. DIXIE HWY		MIAMI	FL	33156	
Internal Revenue Service	Centralized Insolvency Operations	P.O. Box 7346	Philadelphia	PA	19101-7345	
Internal Revenue Service	2670 Market Street	Mail Stop 5-Q30, 133	Philadelphia	PA	19104-5016	
Internal Revenue Service	Special Procedures - Insolvency	7850 SW 6th Court	Plantation	FL	33324	
INTERNATIONAL CONCEPTS	1376 SCOUT TRACE		HOOVER	AL	35244	
INTIMATECO, LLC	149 MADISON AVE, STE 300		NEW YORK	NY	10016	
INVERRARY FALLS, LLC	ID# T000026	431 FAIRWAY DR	DEERFIELD BEACH	FL	33441	
INVESTMENTS FOR NORTH						
LOUISIANA	211 W. Alabama Avenue		Ruston	LA	71270	
IPL	1230 West Morris Street		INDIANAPOLIS	IN	46221	

Name	Address 1	Address 2	City	State	PostalCode	Country
IPL	PO BOX 110		INDIANAPOLIS	IN	46206-0110	
IRBY, MALISSA	REDACT		REDACT	REDACT	REDACT	
IRBY, VINEHSA	REDACT		REDACT	REDACT	REDACT	
IRIS DESIGN	800 MCGARY ST, #201		LOS ANGELES	CA	90021	
IRIZARRY-MILUS, BIANCA	REDACT	REDACT	REDACT	REDACT	REDACT	
IRONDALE DEVELOPERS, INC	C/O BARSTEIN & ASSOCIATES, INC	PO BOX 13096	BIRMINGHAM	AL	35202	
Irondale Developers, Inc.	P.O. Box 13096		Birmingham	AL	35202-3096	
IROQUOIS LEGACY CENTER, LLC	7950 SOLUTION CENTER		CHICAGO	IL	60677-7009	
IROQUOIS MANOR, LLC	321 HENRY STREET		LEXINGTON	KY	40508	
Isaac Chehebar	c/o Trendset	1407 Broadway	New York	NY	10018	
ISAAC IMPORTS	1407 BROADWAY		NEW YORK	NY	10018	
ISABELLA'S JEWELRY	547 MORELAND RD		HUNTINGDON VALLEY	PA	19006	
IT SAVVY, LLC	PO BOX 3296		GLEN ELLYN	IL	60138	
ITALIAN FABRICS, INC.	4768 S ALAMEDA ST		VERNON	CA	90058	
ITZEL LLAMAS	4213 SANDUSKY		KANSAS CITY	KS	66102	
IVERSON, JEREMY	REDACT		REDACT	REDACT	REDACT	
IVERSON, MELISSA	REDACT		REDACT	REDACT	REDACT	
IVEY MECHANICAL COMPANY, LLC	P.O. BOX 610		KOSCIUSKO REDACT	MS REDACT	39090 REDACT	
,			-			
IVORY, CYNTHIA IVORY, JOSHUA	REDACT REDACT		REDACT REDACT	REDACT REDACT	REDACT REDACT	
IVORY, SHANNON	REDACT		REDACT	REDACT	REDACT	
J & S LOCKSMITH	5816 YADKIN ROAD		FAYETTEVILLE	NC	28303	
J & S PLUMBING, INC.	370 BOND STREET		ELK GROVE VILLAGE	IL	60007	
JELINC.	3210 S DODGE BLVD	SUITE #5	TUSCON	AZ	85713	
J&L FOOTWEAR	1285 BIXBY DRIVE	JOHE #J	CITY OF INDUSTRY	CA	91745	
J. T. Milligan	VILLAGE GREEN/Store No192	3007 Bear Point Dr	Panama City	FL	32408	
J.D. ESCO. INC	3107 25TH AVE.	UNIT A	TUSCALOOSA	AL	35401	
J.P. WHOLESALE	10138 E GANEY	ONITA	ELMONTE	CA	91733	
J.T. MILLIGAN	3007 BEAR POINT DR.		PANAMA CITY	FL	32408	
J.T. MILLIGAN	3007 Bear Point Drive		Panama City	FL	32408	
JACARANDA PLAZA PARTNERS, LLC	C/O RAM REALTY SERVICES	4801 PGA BLVD	PALM BEACH GARDENS	FL	33418	
Jack Chehebar	c/o Trendset	1407 Broadway	New York	NY	10018	
JACK WARD & SONS PLUMBING CO	809 MERIDIAN STREET	2.000.00,	NASHVILLE	TN	37207	

Page 87 of 176

Name	Address 1	Address 2	City	State	PostalCode	Country
JACKIE DAVIS	2101 RICHLAND AVE., APT 9		METARIE	LA	70001	
JACKIE HARRISON	Sage Telecom	P.O. Box 79051	Phoenix	AZ	85062-9051	
JACKSON CORPORATION	330 5TH AVENUE	11th FLOOR	NEW YORK	NY	10001	
JACKSON COUNTY COLLECTOR	PO BOX 219747		KANSAS CITY	MO	64121-9747	
JACKSON, CHRISTINA	REDACT	REDACT	REDACT	REDACT	REDACT	
JACKSON, DAWN	REDACT		REDACT	REDACT	REDACT	
JACKSON, DEBORAH	REDACT		REDACT	REDACT	REDACT	
JACKSON, DORIS	REDACT		REDACT	REDACT	REDACT	
JACKSON, EBONY	REDACT		REDACT	REDACT	REDACT	
JACKSON, ELISE	REDACT	REDACT	REDACT	REDACT	REDACT	
JACKSON, JAKIRIA	REDACT		REDACT	REDACT	REDACT	
JACKSON, JAQUNNA	REDACT		REDACT	REDACT	REDACT	
JACKSON, KARTIER	REDACT		REDACT	REDACT	REDACT	
JACKSON, KIMBERLY	REDACT		REDACT	REDACT	REDACT	
JACKSON, KIMBERLY	REDACT		REDACT	REDACT	REDACT	
JACKSON, LAVONDA	REDACT		REDACT	REDACT	REDACT	
JACKSON, LISSA	REDACT		REDACT	REDACT	REDACT	
JACKSON, MARTEDRIA	REDACT		REDACT	REDACT	REDACT	
JACKSON, MILTON	REDACT		REDACT	REDACT	REDACT	
JACKSON, RASHEEDA	REDACT	REDACT	REDACT	REDACT	REDACT	
JACKSON, ROCHELLE	REDACT		REDACT	REDACT	REDACT	
JACKSON-MOORE, KRISTINE	REDACT		REDACT	REDACT	REDACT	
JACOX, ISSAC	REDACT		REDACT	REDACT	REDACT	
JACQUELINE DAVIS	3047 O'NEAL LANE		BATON ROUGE	LA	70816	
JACQUELINE SEARCY	13024 FOX HAVEN DRIVE		FLORISSANT	MO	63033	
JAG AIR MECHANICAL, INC	11878 WILES ROAD		CORAL SPRINGS	FL	33076	
JAG DBA PENN AIR MECHANICAL	11878 WILES ROAD		CORAL SPRINGS	FL	33076	
JAGUAR/LAND ROVER/PORSCHE OF						
CHATTANOOGA	7648 LEE HIGHWAY		CHATTANOOGA	TN	37421	
JAKE LABAR & SON EXTERMINATING						
CO., INC.	PO BOX 8394		ROCKY MOUNT	NC	27804	
JAKE MARSHALL SERVICE, INC	PO BOX 4324		CHATTANOOGA	TN	37405	
JAMERSON, JASMINE	REDACT		REDACT	REDACT	REDACT	
JAMES, BRODRIC	REDACT	REDACT	REDACT	REDACT	REDACT	
JAMES, DONTAYSHA	REDACT		REDACT	REDACT	REDACT	
JAMES, JESSICA	REDACT		REDACT	REDACT	REDACT	
JAMES, JHONISHA	REDACT		REDACT	REDACT	REDACT	
JAMES, SLIGLER	REDACT		REDACT	REDACT	REDACT	

Page 88 of 176

Name	Address 1	Address 2	City	State	PostalCode	Country
JAMES, WESLEY	REDACT		REDACT	REDACT	REDACT	
JAMISON, NIKKI	REDACT		REDACT	REDACT	REDACT	
JANAF SHOPPING CENTER, LLC	JANAF OFFICE BUILDING	SUITE 520	NORFOLK	VA	23502	
JANPAK, INC	1140 JORDAN RD		HUNTSVILLE	AL	35811	
JASON MICHAEL KATZ	30665 NORTHWESTERN HWY	STE 202	FARMINGTON HILLS	MI	48334	
JASON, GARY	REDACT		REDACT	REDACT	REDACT	
JAXIS	1407 BROADWAY, RM 2300		NEW YORK	NY	10018	
JAY DAY INC	1400 BROADWAY		NEW YORK	NY	10018	
JBH LIMITED COMPANY LTD.	Noble Management Company	5821-C Lake Worth Road	Greenacres	FL	33407	
JBH Limited Company, LTD	POINCIANA PLAZA/Store No107	5821-C Lake Worth Road	Greenacres	FL	33407	
JCS APPAREL GROUP	1407 BROADWAY	SUITE 202	NEW YORK	NY	10018	
JDRF - BIRMINGHAM	600 BEACON PARKWAY W	SUITE 860	BIRMINGHAM	AL	35209	
JEA	43 W Church St		JACKSONVILLE	FL	32202	
JEA	P.O. BOX 45047		JACKSONVILLE	FL	32232-5047	
JEAN, CALVIN	REDACT		REDACT	REDACT	REDACT	
JEANS JACKET CLOTHING	115 E PICO BLVD		LOS ANGELES	CA	90015	
JEFFERIES, MARISSA	REDACT		REDACT	REDACT	REDACT	
JEFFERIES, MOLLY	REDACT		REDACT	REDACT	REDACT	
JEFFERSON BANK & TRUST	2301 Market Street		St. Louis	MO	63103	
JEFFERSON CO. DEPT OF REVENUE	EDUCATION SALES TAX	PO BOX 830710	BIRMINGHAM	AL	35283-0710	
Jefferson County	Attn: Consumer Protection Agency	Mike O'Connell - County Attorney	Louisville	KY	40202	
Jefferson County	Attn: Tony Petelos	716 Richard Arrington Jr. Blvd.	North Birmingham	AL	35203	
JEFFERSON COUNTY (SLS TX)	DEPT OF REVENUE	PO BOX 830710	BIRMINGHAM	AL	35283-0710	
JEFFERSON COUNTY DEPT OF						
REVENUE	PO BOX 12207		BIRMINGHAM	AL	35202-2207	
JEFFERSON COUNTY, AL	TAX COLLECTOR	ROOM 160 COURTHOUSE	BIRMINGHAM	AL	35203	
JEFFERSON COUNTY, AR	TAX COLLECTOR	PO DRAWER A	PINE BLUFF	AR	71611-6320	
JEFFERSON COUNTY, KY SHERIFF	P.O. BOX 34570		LOUISVILLE	KY	40232-4570	
JEFFERSON PARISH	SHERIFF/TAX COLLECTOR	PO BOX 248	GRETNA	LA	70054-0248	
JEFFERSON PARISH DEPT OF WATER	1221 Elmwood Park Blvd	Suite 103	Jefferson	LA	70123	
JEFFERSON PARISH DEPT OF WATER	PO BOX 10007		JEFFERSON	LA	70181-0007	
Jefferson Parish Sheriff's Office	Bureau of Revenue and Taxation	Sales Tax Division	Gretna	LA	70054	
JEFFERSON PARISH-PP	PROPERTY TAX DIVISION	P.O. BOX 130	GRETNA	LA	70054-0130	

Page 89 of 176

Name	Address 1	Address 2	City	State	PostalCode	Country
JEFFERSON, ASHLEY	REDACT		REDACT	REDACT	REDACT	
JEFFERSON, UNIQUE	REDACT		REDACT	REDACT	REDACT	
JEFFREYS, TWANDA	REDACT		REDACT	REDACT	REDACT	
JEM INTERNATIONAL	1 EAST 33RD STREET	11TH FLOOR	NEW YORK	NY	10016	
JENKINS, ALEXUS	REDACT	REDACT	REDACT	REDACT	REDACT	
JENKINS, CONSTANCE	REDACT		REDACT	REDACT	REDACT	
JENKINS, ISAIAH	REDACT		REDACT	REDACT	REDACT	
JENKINS, JANAYE	REDACT		REDACT	REDACT	REDACT	
JENKINS, MARLANA	REDACT	REDACT	REDACT	REDACT	REDACT	
JENKINS, MELVA	REDACT		REDACT	REDACT	REDACT	
JENKINS, RONNIE	REDACT		REDACT	REDACT	REDACT	
JENKINS, SHANEEN	REDACT		REDACT	REDACT	REDACT	
JEPTHA R. BARBOUR, IV	PO BOX 199		RIDGLEAND	MS	39158-0199	
JERRIETT, ANGELA	REDACT		REDACT	REDACT	REDACT	
JERRY'S & KWALL'S, INC.	dba JERRY'S A/C & HEATING	P.O. BOX 6925	FLORENCE	SC	29502	
JESCO/ALLIANCE FOOTWARE CORP.	37 WEST 37TH ST.,	SUITE 301	NEW YORK	NY	10118	
JESUS, DARIO DE	REDACT		REDACT	REDACT	REDACT	
JESUS, MARCO DE	REDACT		REDACT	REDACT	REDACT	
JEWEL FOOD STORES, INC	C/O SUPERVALUE REF# PA1142647	PO BOX 958844	ST. LOUIS	мо	63195-8844	
JEWEL FOOD STORES, INC.	C/O SUPERVALU	P.O. BOX 958844	St. Louis	MO	63195	
	For Piggly Wiggly c/o Daniella				-	
J-H-J-, INC.	Satawa	5151 Plank Road	Baton Rouge	LA	70805	
JHONS, ANDRIETTA	REDACT		REDACT	REDACT	REDACT	
JIMENEZ, CESAR	REDACT		REDACT	REDACT	REDACT	
JIMENEZ, MARLENE	REDACT		REDACT	REDACT	REDACT	
JIMMY MEADE	196 WOODCREEK LANE		PELL CITY	AL	35128	
JIMMY'S GLASS SERVICE, INC	8801 KINGSTON RD.		SHREVEPORT	LOUISIANA	71118	
JMB CRAFTS	ATTN: JERRY BOYD	95 BOYD ROAD	EMPIRE	AL	35063	
	C/O COMMERCIAL BROKERS PROP					
JNPR, LLC	MGMT	3331 SEVERN AVENUE	METAIRIE	LA	70002	
	c/o Commercial Brokers Property					
JNPR, LLC	Management LLC	3331 Severn Avenue	Metairie	LA	70002	
JNS FASHIONS, LLC	2110 NW 95TH AVE.		MIAMI	FL	33172	
	c/o GENOVESE JOBLOVE &					
JNS INVT, LLC	BATTISTA, P.A.	Attn: Paul J. Battista, Esq.	Miami	FL	33131	

Name	Address 1	Address 2	City	State	PostalCode	Country
	c/o GENOVESE JOBLOVE &					
JNS INVT, LLC	BATTISTA, P.A.	Heather L. Harmon, Esq.	Miami	FL	33131	
JNS INVT, LLC, a Florida limited						
liability company	2110 N.W. 95th Avenue		Miami	FL	33172	
JOAL TRADING	10 WEST 33RD STREET	8TH FLOOR	NEW YORK	NY	10001	
JOE FIDA'S AUTO PLATE SERVICE	160 PITTSBURGH STREET		UNIONTOWN	PA	15401	
JOHN C MCALEER III	STANDING TRUSTEE	PO BOX 1779	MEMPHIS	TN	38101-1779	
	DALLAS COUNTY TAX ASSESSOR-					
JOHN R. AMES	COL	P.O. BOX 139033	DALLAS	TX	75313-9033	
JOHN RANDOLPH MEDICAL CENTER	P.O. BOX 13620		RICHMOND	VA	23226	
JOHN S SIMPSON	C/O SIMPSON LAW FIRM	P.O. BOX 1410	RIDGELAND	MS	39158-1410	
JOHN, SHANE	REDACT		REDACT	REDACT	REDACT	
JOHNIGAN, MAYA	REDACT		REDACT	REDACT	REDACT	
JOHNS, GREG	REDACT	REDACT	REDACT	REDACT	REDACT	
JOHNSON III, CLARENCE	REDACT		REDACT	REDACT	REDACT	
JOHNSON, ASHLEY	REDACT		REDACT	REDACT	REDACT	
JOHNSON, BLAIR	REDACT		REDACT	REDACT	REDACT	
JOHNSON, BRANDON	REDACT		REDACT	REDACT	REDACT	
JOHNSON, CALVIN	REDACT		REDACT	REDACT	REDACT	
JOHNSON, CANDACE	REDACT		REDACT	REDACT	REDACT	
JOHNSON, COURTNEY	REDACT		REDACT	REDACT	REDACT	
JOHNSON, DARAK	REDACT		REDACT	REDACT	REDACT	
JOHNSON, DOMINIQUE	REDACT	REDACT	REDACT	REDACT	REDACT	
JOHNSON, ELAINE	REDACT		REDACT	REDACT	REDACT	
JOHNSON, FELICIA	REDACT		REDACT	REDACT	REDACT	
JOHNSON, HONDRAY	REDACT		REDACT	REDACT	REDACT	
JOHNSON, JABARRE	REDACT		REDACT	REDACT	REDACT	
JOHNSON, JAMIE	REDACT	REDACT	REDACT	REDACT	REDACT	
JOHNSON, JAMIE	REDACT		REDACT	REDACT	REDACT	
JOHNSON, JESSICA	REDACT		REDACT	REDACT	REDACT	
JOHNSON, JOSEPH	REDACT		REDACT	REDACT	REDACT	
JOHNSON, JOYAN	REDACT		REDACT	REDACT	REDACT	
JOHNSON, JUAN	REDACT	REDACT	REDACT	REDACT	REDACT	
JOHNSON, KIRK	REDACT		REDACT	REDACT	REDACT	
JOHNSON, KRISTINA	REDACT		REDACT	REDACT	REDACT	
JOHNSON, KRISTINA	REDACT	REDACT	REDACT	REDACT	REDACT	
JOHNSON, LATIA	REDACT		REDACT	REDACT	REDACT	

Page 91 of 176

Name	Address 1	Address 2	City	State	PostalCode	Country
JOHNSON, MARGARET	REDACT		REDACT	REDACT	REDACT	
JOHNSON, MARQUITA	REDACT		REDACT	REDACT	REDACT	
JOHNSON, MICHELLE	REDACT		REDACT	REDACT	REDACT	
JOHNSON, MICHELLE	REDACT		REDACT	REDACT	REDACT	
JOHNSON, RAYMOND	REDACT		REDACT	REDACT	REDACT	
JOHNSON, REGINA	REDACT		REDACT	REDACT	REDACT	
JOHNSON, SANDRA	REDACT		REDACT	REDACT	REDACT	
JOHNSON, SANDRA	REDACT		REDACT	REDACT	REDACT	
JOHNSON, THOMAS	REDACT		REDACT	REDACT	REDACT	
JOHNSON, TYLER	REDACT	REDACT	REDACT	REDACT	REDACT	
JOLIE INTIMATE	183 MADISON AVE. SUITE 1001		NEW YORK	NY	10016	
JON PIC CREATIVE AGENCY	317 S. MAIN ST. #1		EL DORADO	KS	67042	
JONATHAN K APPAREL	2636 S. MAIN ST		LOS ANGELES	CA	90007	
Jones County	Attn: Consumer Protection Agency	Frank Childs County Attorney	Macon	GA	31202-0898	
JONES COUNTY	TAX COLLECTOR	P.O. BOX 511	LAUREL	MS	39441	
JONES, ALICIA	REDACT		REDACT	REDACT	REDACT	
JONES, ALLAHEA	REDACT	REDACT	REDACT	REDACT	REDACT	
JONES, CHRISTELL	REDACT		REDACT	REDACT	REDACT	
JONES, CLEO	REDACT		REDACT	REDACT	REDACT	
JONES, DIAMOND	REDACT		REDACT	REDACT	REDACT	
JONES, DORIS	REDACT		REDACT	REDACT	REDACT	
JONES, EBONI	REDACT		REDACT	REDACT	REDACT	
JONES, EEAN	REDACT		REDACT	REDACT	REDACT	
JONES, FOSTER, JOHNSTON &						
STUBBS, PA	PO BOX 3475		WEST PALM BEACH	FL	33402-3475	
JONES, GEORGE	REDACT		REDACT	REDACT	REDACT	
JONES, JAHRESE	REDACT		REDACT	REDACT	REDACT	
JONES, JESSICA	REDACT		REDACT	REDACT	REDACT	
JONES, JESSICA	REDACT	_	REDACT	REDACT	REDACT	
JONES, JOSEPH	REDACT		REDACT	REDACT	REDACT	
JONES, JOSHUA	REDACT		REDACT	REDACT	REDACT	
JONES, KEVIN	REDACT		REDACT	REDACT	REDACT	
JONES, KYNETRA	REDACT		REDACT	REDACT	REDACT	
JONES, LA'CHASTITY	REDACT		REDACT	REDACT	REDACT	
JONES, MURRAY	REDACT		REDACT	REDACT	REDACT	
JONES, RENEE	REDACT		REDACT	REDACT	REDACT	
JONES, SHARON	REDACT	REDACT	REDACT	REDACT	REDACT	

Page 92 of 176

Name	Address 1	Address 2	City	State	PostalCode	Country
JONES, SHEENA	REDACT		REDACT	REDACT	REDACT	
JONES, TAMEKA	REDACT		REDACT	REDACT	REDACT	
JONES, TARIKIA	REDACT		REDACT	REDACT	REDACT	
JONES, TYRALL	REDACT		REDACT	REDACT	REDACT	
JONES, YAZMEEN	REDACT		REDACT	REDACT	REDACT	
JORDAN, AMANDA	REDACT		REDACT	REDACT	REDACT	
JORDAN, CHARITA	REDACT		REDACT	REDACT	REDACT	
JORDAN, DOMINICK	REDACT		REDACT	REDACT	REDACT	
JORDAN, LATONIA	REDACT		REDACT	REDACT	REDACT	
JORDAN, RAVEN	REDACT		REDACT	REDACT	REDACT	
JORDAN, SHEILA	REDACT		REDACT	REDACT	REDACT	
JORDAN, STACEY	REDACT		REDACT	REDACT	REDACT	
JORDAN, VANESSA	REDACT		REDACT	REDACT	REDACT	
Joseph Chehebar	c/o Trendset	1407 Broadway	New York	NY	10018	
JOULEVETTE, ELASTACIA	REDACT	,	REDACT	REDACT	REDACT	
JOY GRECCO	228 CHIPPER ROAD		WATERBURY	СТ	06704	
	C/O SIDNEY A GORDON REAL					
JOYLAND SHOPPING CENTER, INC.	ESTATE LLC	P.O. BOX 193	TALLMAN	NY	10982-0193	
	SIDNEY A GORDON REAL ESTATE					
JOYLAND SHOPPING CENTER, INC.	LLC	P O Box 193	Tallman	NY	10982-0193	
JP MORGAN CHASE	PO Box 659754		San Antonio	TX	78265-9754	
JP MORGAN CHASE - INDIANA	PO Box 659754		San Antonio	TX	78265-9754	
JP MORGAN CHASE - LOUISIANA	PO Box 659754		San Antonio	TX	78265-9754	
JP MORGAN CHASE - MICHIGAN	PO Box 659754		San Antonio	TX	78265-9754	
JP MORGAN CHASE - TEXAS	PO Box 659754		San Antonio	TX	78265-9754	
JP ORIGINALS	19101 E. WALNUT DR. N.		CITY OF INDUSTRY	CA	91748	
JSK	237 W 32 ST		LOS ANGELES	CA	90007	
JSMN SHOPPING CENTER, LLC	280 IDAHO STREET		PARAMUS	NJ	07652	
JT / RIVERSIDE ELECTRIC	1410 QUEEN PALM DRIVE		EDGEWATER	FL	32132	
JUDGE OF PROBATE (COLUMBIANA)	SHELBY COUNTY	PO BOX 825	COLUMBIANA	AL	35051	
JULIA	3187 S. ALAMEDA		LOS ANGELES	CA	90058	
JUST ENERGY	5251 Westheimer Road	Suite 1000	Houston	TX	77056	
JUST ENERGY	P.O. BOX 650518		DALLAS	TX	75265-0518	
JUST ONE LLC	1400 BROADWAY	18TH FLOOR	NEW YORK	NY	10018	
JUSTE, ESTHER	REDACT		REDACT	REDACT	REDACT	
КТОО	1152 CROCKER ST		LOS ANGELES	CA	90021	
KAIGLER, KARLETTA	REDACT		REDACT	REDACT	REDACT	

Page 93 of 176

Name	Address 1	Address 2	City	State	PostalCode	Country
KAISER, ALBERTHA	REDACT		REDACT	REDACT	REDACT	
KANE PLUMBING CO., INC.	403 E. LABURNUM AVE.		RICHMOND	VA	23222	
KANSAS CITY BOARD OF PUBLIC						
UTILITIES	540 Minnesota Avenue		Kansas City	KS	66101	
KANSAS CITY BOARD OF PUBLIC						
UTILITIES	P.O.BOX 219661		KANSAS CITY	МО	64121-9661	
KANSAS CITY PUBLIC UTILITIES	PO BOX 219661		KANSAS CITY	MO	64121-9661	
Kansas Department of Revenue	915 SW Harrison St #300		Topeka	KS	66612	
KANSAS GAS SERVICE	7421 W. 129th		Overland Park	KS	66213	
KANSAS GAS SERVICE	PO BOX 219046		KANSAS CITY	МО	64121-9046	
KANSAS SALES TAX	KANSAS DEPT. OF REVENUE	915 S.W. HARRISON STREET	ТОРЕКА	KS	66625-2007	
KANSAS SECRETARY OF STATE	MEMORIAL HALL/1ST FLOOR	120 SW 10TH AVENUE	TOPEKA	KS	66612-1594	
KAPASI GLASS MART	398 WEST MAIN STREET		SPARTANBURG	SC	29301	
KAREN WARREN	565 LONE RIDER CT		ROCKWALL	TX	75087	
KARRAN, MOHAN	REDACT		REDACT	REDACT	REDACT	
KASTNER ELECTRIC, INC	PO BOX 1394		MADISONVILLE	LA	70447	
KATHY COULTER, CLERK		501 14TH STREET	PHENIX CITY	AL	36867	
KATIE FORD	157 ELDER STREET		IRONDALE	AL	35210	
KAY FASHION	2416 E 16TH ST		LOS ANGELES	CA	90021	
KBF 54TH, LLC	C/O PK PARTNERS, LLC	3610 RIVER CROSSING PKWY	INDIANAPOLIS	IN	46240	
KC WATER SERVICES	4800 E. 63rd Street		Kansas City	МО	64130	
KC WATER SERVICES	PO BOX 807045		KANSAS CITY	MO	64180-7045	
KCMO WATER SERVICES DEPT	4800 E. 63rd Street		Kansas City	MO	64130	
KCMO WATER SERVICES DEPT	PO BOX 807045		KANSAS CITY	MO	64180-7045	
KCP&L	1200 Main St.		Kansas City	MO	64105	
KCP&L	PO BOX 219330		KANSAS CITY	MO	64121-9330	
KEARNEY, D'ZHANE	REDACT		REDACT	REDACT	REDACT	
KEENPAC, LLC	25 MAIN STREET	2ND FLOOR	GOSHEN	NY	10924	
KEITH ROGERS	807 TENNESSEE AVE		BESSEMER	AL	35020	
KELEX	1385 BROADWAY		NEW YORK	NY	10018	
KELLAM, TRACY	REDACT		REDACT	REDACT	REDACT	
KELLEY, ASHLEY	REDACT	-	REDACT	REDACT	REDACT	
KELLEY, LINDA	REDACT		REDACT	REDACT	REDACT	
KELLUM, ETHEL	REDACT		REDACT	REDACT	REDACT	
KELLUM, RUBY	REDACT		REDACT	REDACT	REDACT	
KELLY, PATRICIA	REDACT		REDACT	REDACT	REDACT	
KELLY, RAYLYNN	REDACT		REDACT	REDACT	REDACT	

Page 94 of 176

Name	Address 1	Address 2	City	State	PostalCode	Country
KELLY, ROBERT	REDACT		REDACT	REDACT	REDACT	
KEN ROSE ASSOCIATES	502 HAMBURG TURNPIKE	SUITE 203	WAYNE	NJ	07470	
KENDRICK, MARAS	REDACT		REDACT	REDACT	REDACT	
KENMART REALTY, LLC	16300 NW 48TH AVENUE		MIAMI	FL	33014	
Kennedy Mech	P.O. Box 1214		Walker	LA	70785-1214	
KENNEDY MECHANICAL SERVICES	PO BOX 1214		WALKER	LA	70785-1214	
KENNETH J. KELSON DBA FLORIDA						
ELECTRIC	901 WEST GOVERNMENT STREET		PENSACOLA	FL	32502	
KENS' MILESTONE MECHANICAL,						
INC.	P.O. BOX 40544		HOUSTON	TX	77240	
Kentucky Revenue Cabinet	501 High Street		Frankfort	KY	40620	
KENTUCKY STATE TREASURER	KENTUCKY DEPARTMENT OF REVENUE		FRANKFORT	ку	40620	
KENTUCKY STATE TREASURER	OFFICE OF SECRETARY OF STATE	P.O. BOX 718	FRANKFORT	KY	40602-0718	
	KENTUCKY DEPARTMENT OF					
KENTUCKY STATE TREASURER	REVENUE	P.O. BOX 718	FRANKFORT	KY	40620	
	1050 US Highway 127 South, Suite					
Kentucky State Treasury	100		Frankfort	KY	40601	
KENYADA, DUNEY	REDACT		REDACT	REDACT	REDACT	
KENYATTA SCOTT	117 WESTCHESTER DRIVE		MONROE	LA	71203	
KERRI BUNN	552 MATHEWS ROAD		WEST BLOCTON	AL	35184	
KERRY SMITH	2737 Barnhill Court		Xenia	ОН	45385	
KEVIN S. REDDEN	258 OLEANDER ROAD		LIVINGSTON	TX	77351	
KEY, JAMESHA	REDACT		REDACT	REDACT	REDACT	
KEY, JENNIFER	REDACT	REDACT	REDACT	REDACT	REDACT	
KEYES, BREANNA	REDACT		REDACT	REDACT	REDACT	
KHEAA	PO BOX 4869		FRANKFORT	KY	40604-4869	
KHEM, HOEUN	REDACT		REDACT	REDACT	REDACT	
KHURSHIDOVA, ZULEYHA	REDACT		REDACT	REDACT	REDACT	
KIKE, SURITA	REDACT		REDACT	REDACT	REDACT	
KILLINS, CERESE	REDACT		REDACT	REDACT	REDACT	
Kimbro Mechanical	1877 Air Lane Drive		Nashville	TN	37210	
KIMBRO MECHANICAL, LLC	1877 AIR LANE DRIVE		NASHVILLE	TN	37210	
KIMCO BATON ROUGE 1183, LLC	SLAB1183/LSIMP6/00	PO BOX 6203	HICKSVILLE	NY	11802-6203	
·		P O BOX 6203 DEPT				
KIMCO BATON ROUGE 1183, LLC	SLAB1183/LSIMP6/00	CODE:SLAB1183/	Hicksville	NY	11802-6203	
KIMCO BATON ROUGE 1183, LLC	9600 Airline Hwy		Baton Rouge	LA	70815	

Name	Address 1	Address 2	City	State	PostalCode	Country
KIMCO BATON ROUGE 1183, LLC	P.O. BOX 6203		HICKSVILLE	NY	11802-6203	
	IVES DAIRY CROSSING S/C/Store					
Kimco Of North Miami Inc.	No145	c/o Kimco Realty Corporation	Charlotte	NC	28287	
KIMCO OF NORTH MIAMI, INC.	C/O KIMCO REALTY CORP.	ATTN: DIR OF SPECIALTY LEASING	CHARLOTTE	NC	28287	
KIMCO OF NORTH MIAMI, INC.	Kimco Realty Corporation	6060 Piedmont Row Drive South	Charlotte	NC	28287	
	C/O KIMIJO PROPERTIES DEPT #					
KIMIJO PROPERTIES LLC	197	PO BOX 528	COLUMBIA	SC	29202	
	C/O KIMIJO PROPERTIES, DEPT					
KIMIJO PROPERTIES, LLC	#197	P O BOX 528	Columbia	SC	29202	
KIM'S ELECTRIC, INC	13619 NORTH MAIN STREET		JACKSONVILLE	FL	32218	
	KING SHOPPING CENTER/Store	c/o Harvey Property Management				
King Associates Limited Partnership	No185	Company, Inc	Bethesda	MD	20814	
KING ASSOCIATES LIMITED	c/o Harvey Property Management					
PARTNERSHIP	Company	6931 Arlington Road	Bethesda	MD	20814	
	C/O HARVEY PROPERTY MGMT CO					
KING ASSOCIATES, LP	INC	6931 ARLINGTON RD, SUITE 500	BETHESDA	MD	20814	
KING ASSOCIATES, LP	% HARVEY PROPERTY MGMT. CO.	6931 ARLINGTON ROAD, SUITE 500	BETHESDA	MD	20814	
KING, ARIELLE	REDACT		REDACT	REDACT	REDACT	
KING, ARIELLE	REDACT		REDACT	REDACT	REDACT	
KING, CAMILLE	REDACT		REDACT	REDACT	REDACT	
KING, CASSANDRA	REDACT		REDACT	REDACT	REDACT	
KING, CLAUDETTE	REDACT		REDACT	REDACT	REDACT	
KING, CRYSTAL	REDACT		REDACT	REDACT	REDACT	
KING, CYNTRICIA	REDACT		REDACT	REDACT	REDACT	
KING, DANIELLE	REDACT		REDACT	REDACT	REDACT	
KING, DANIELLE	REDACT		REDACT	REDACT	REDACT	
KING, JASMINE	REDACT		REDACT	REDACT	REDACT	
KING, LATIGRA	REDACT		REDACT	REDACT	REDACT	
KING, TENISHA	REDACT		REDACT	REDACT	REDACT	
KING, TROY	REDACT		REDACT	REDACT	REDACT	
KIRKPATRICK, NATALIE	REDACT		REDACT	REDACT	REDACT	
KIRSCHNER, KAYLA	REDACT		REDACT	REDACT	REDACT	
KITCHEN, QUANISHA	REDACT		REDACT	REDACT	REDACT	
KIZER, JR., PRINCE	REDACT	REDACT	REDACT	REDACT	REDACT	
KLEIN BROS SAFE & LOCK	1101 WEST BROADWAY		LOUISVILLE	KY	40203	

Page 96 of 176

Name	Address 1	Address 2	City	State	PostalCode	Country
KLEIN ISD	7200 SPRING - CYPRESS RD.		KLEIN	TX	77379-3299	
KNAPIK, ANISSA	REDACT		REDACT	REDACT	REDACT	
KNAPIK, PAUL	REDACT		REDACT	REDACT	REDACT	
KNAPIK, RYAN	REDACT		REDACT	REDACT	REDACT	
KNIGHT TRANSPORTATION SERVICE						
INC.	P.O. BOX 29897		PHOENIX	AZ	85038-9897	
KNIGHT, CHASMINE	REDACT	REDACT	REDACT	REDACT	REDACT	
KNIGHT, KEENAN	REDACT		REDACT	REDACT	REDACT	
KNOX CO GENERAL SESSION	PO BOX 379		KNOXVILLE	TN	37901	
KNOX COUNTY	P.O. BOX 70		KNOXVILLE	TN	37901	
KNOX FIRE EXTINGUISHER COMPANY	1201 UNIVERSITY AVE		KNOXVILLE	TN	37921	
KNOX, DEBRA	REDACT		REDACT	REDACT	REDACT	
KNUCKLES, MARY	REDACT		REDACT	REDACT	REDACT	
KONICA MINOLTA BUSINESS					_	
SOLUTIONS	DEPT AT 952823		ATLANTA	GA	31192-2823	
Konica Minolta Business Solutions						
U.S.A., Inc.	100 Williams Drive		Ramsey	NJ	07448	
KONICA MINOLTA PREMIER FINANCE	PO BOX 740423		ATLANTA	GA	30374-0423	
KOP KLINE PLAZA, LLC	C/O BRIXMOR PROPERTY GROUP	#4038057	CINCINNATI	ОН	45271.3538	
	C/O BRIXMOR PROPERTY GRP					
KOP KLINE PLAZA, LLC	#4038057	P O BOX 645346	Cincinatti	ОН	45264-5346	
KOTERA, JOSEPH	REDACT		REDACT	REDACT	REDACT	
KPDISTRIBUTIONS	5231 SIMPSON FERRY RD		MECHANICSBURG	PA	17050	
KRG GAINESVILLE, LLC	5220 PAYSHERE CIRCLE		CHICAGO	IL	60674	
KRISTEN COLLINS	38 HART STREET #4		BROOKLYN	NY	11206	
	C/O BIBBY INTERNATIONAL TRADE					
KRISTINO	FINANCE INC.	600 TOWN PARK LANE	KENNESAW	GA	30144	
KRT PROPERTY HOLDINGS LLC	c/o Brixmor Property Group	420 Lexington Avenue	New York	NY	10170	
KRT PROPERTY HOLDINGS, LLC.	C/O BRIXMOR PROPERTY GROUP	P.O. BOX 30875	NEW YORK	NY	10087-0875	
KRUGER, BRANDY	REDACT		REDACT	REDACT	REDACT	
KRUSH NY	1407 BROADWAY, STE 1704		NEW YORK	NY	10018	
KSK VISUAL INGENUITY	32300 AURORA ROAD		CLEVELAND	ОН	44139-2820	
KUB	PO BOX 59017		KNOXVILLE	TN	37950-9017	
KURZWEIL, ROBYN	REDACT	İ	REDACT	REDACT	REDACT	

Name	Address 1	Address 2	City	State	PostalCode	Country
KUYKENDALL, QUZANNA E.	REDACT		REDACT	REDACT	REDACT	
KWIK TICKET, INC	4101 GLENWOOD ROAD		BROOKLYN	NY	11210	
L & J MANCHESTER II, LLC	360 BLOOMFIELD AVE		WINDSOR	СТ	06095	
L & L ASSOCIATES	17650 WEST 12 MILE ROAD		SOUTHFIELD	MI	48076	
L & R HANDY MAN	1731 E. LAWRENCE STREET		TYLER	TX	75702	
L G ACCESSORIES	6 LINCOLN AVE.		DIX HILLS	NY	11746	
L J R MGMT, LLC	32710 FRANKLIN RD		FRANKLIN	MI	48025	
L&J MANCHESTER II, LLC	360 Bloomfield Avenue		Windsor	СТ	6095	
L. KIANOFF & ASSOCIATES, INC	1128 22ND STREET SOUTH		BIRMINGHAM	AL	35205-2803	
LA APPAREL	209 W 38TH ST, STE 1010		NEW YORK	NY	10018	
LA FASHION HUB, INC	1620-A S. LOS ANGELES STREET		LOS ANGELES	CA	90015	
LA JEANS	111 EAST PICO BLVD		LOS ANGELES	CA	90015	
LA MAIN CONNECTION	2011 E 27TH STREET		VERNON	CA	90058	
LA TEEZ	1801 E. 41ST PL. #4		LOS ANGELES	CA	90058	
LA TOUJOURS JEUNE	737 STANDFORD AVE, STE B		LOS ANGELES	CA	90021	
LABELS INTER-GLOBAL, INC.	109 WEST 38TH STREET	SUITE 701	NEW YORK	NY	10018	
LABONTE, VANESSA	REDACT	REDACT	REDACT	REDACT	REDACT	
LACEY, JACQUESA	REDACT		REDACT	REDACT	REDACT	
LACLEDE GAS COMPANY	720 Olive Street		Saint Louis	MO	63101	
LACLEDE GAS COMPANY	DRAWER 2		ST. LOUIS	MO	63171	
LAE, DAVID MELGAR	REDACT		REDACT	REDACT	REDACT	
LAFAYETTE CONSOLIDATED GOVMT	PO BOX 4024		LAFAYETTE	LA	70502	
LAFAYETTE CONSOLIDATED GOVT	PERSONAL PROPERTY TAX	PO BOX 4024	LAFAYETTE	LA	70502	
Lafayette Parish	PO Box 3883		Lafayette	LA	70502-3883	
LAFAYETTE PARISH TAX COLLECTOR	P.O. BOX 52667		LAFAYETTE	LA	70505	
LAFAYETTE, KIMBERLY	REDACT	REDACT	REDACT	REDACT	REDACT	
LAKE COUNTY, IN	TREASURER	2293 N. MAIN STREET	CROWN POINT	IN	46307-1896	
LAKE MEADOWS ASSOCIATES	C/O DRAPER & KRAMER	ATTN: DEE WELLS	ELMHURST	IL	60126	
LAKESHORE VILLAGE	C/O M & P SHOPPING CENTERS	5025-M WINTERS CHAPEL RD	ATLANTA	GA	30360	
LAKITA TRIPLETT	19930 BIRWOOD		DETROIT	MI	48221	
LALONDE, MARY	REDACT		REDACT	REDACT	REDACT	
LAMAR AIRWAYS, LLC	ATTN: MICHAEL E. GOLDSTEIN	999 SOUTH SHADY GROVE RD	MEMPHIS	TN	38120	
Lamar Airways, LLC	LAMAR AIRWAYS S/C/Store No175	Attn: Michael E. Goldstein	Memphis	TN	38119	
LAMAR AIRWAYS, LLC	6060 Poplar Avenue	Suite 450 P	Memphis	TN	38119	

Page 98 of 176

Name	Address 1	Address 2	City	State	PostalCode	Country
LAMAR CROSSING MAIDEN, LLC	C/O TIME EQUITIES	55 FIFTH AVENUE	NEW YORK	NY	10003	
LAMAR CROSSING SHOPPING CTR						
EQUITIES LLC	M & T BANK	ATTN JOAN STAPLEY # 1037894	BUFFALO	NY	14203	
LAMAR SOUTH, LLC	137 Jordan Drive		Chattanooga	TN	37421	
LAMMI, DEBORAH	REDACT		REDACT	REDACT	REDACT	
LAMMI, JOHN	REDACT		REDACT	REDACT	REDACT	
LAND, S'KERNEY	REDACT		REDACT	REDACT	REDACT	
LANDING VENTURES	605 W. 47TH STREET	SUITE 200	KANSAS CITY	MO	64112	
LANDRUM, VINCENT	REDACT	REDACT	REDACT	REDACT	REDACT	
LANE, JEREMY	REDACT		REDACT	REDACT	REDACT	
LANG, DIONTE	REDACT		REDACT	REDACT	REDACT	
LANGHONE, JAMES	REDACT		REDACT	REDACT	REDACT	
Langley Square Associates, LLC	LANGLEY SQUARE/Store No111	c/o The Katsias Company	Virginia Beach	VA	23451	
LANGLEY SQUARE ASSOCIATES, LLC	c/o The Katsias Company	2901 S. Lynnhaven Road	Virginia Beach	VA	23452	
LANGSTON, CATHY	REDACT		REDACT	REDACT	REDACT	
LANGSTON, KATHY	REDACT		REDACT	REDACT	REDACT	
LANY GROUP LLC	1407 BROADWAY	ROOM 301	NEW YORK	NY	10018	
LARGO/BORDEAUX ASSOCIATES LTD.	C/O MITCHELL COMPANY	P.O. BOX 160306	MOBILE	AL	36616-1306	
LARS, KEENA	REDACT		REDACT	REDACT	REDACT	
LASETER, MAXINE	REDACT	REDACT	REDACT	REDACT	REDACT	
LASSITER, ELVA	REDACT		REDACT	REDACT	REDACT	
LATASHA BLAKE	P.O. BOX 133		TRUSSVILLE	AL	35173	
LATHAM, DANIELLE	REDACT	REDACT	REDACT	REDACT	REDACT	
LaTonya Simms	c/o Ronald A. Marron, Esq.	LAW OFFICES OF RONALD A. MARRON	San Diego	CA	92103	
20.0.170 0	9,0 11011011 111011 011,1 2541	LAW OFFICES OF DOUGLAS J.	Sun Biogo	G/ 1	32103	
LaTonya Simms	c/o Douglas J. Campion, Esq.	CAMPION	San Diego	CA	92127	
Ed l'Ollyd Sillinis	c, o Bougius s. cumpion, Esq.	Law Offices of Douglas J. Campion,	Sun Diego	C/ C	32127	
LaTonya Simms	c/o Douglas J. Campion, Esq.	APC	San Diego	CA	92108	
LATTER & BLUM PROPERTY MGMT	-,		22 2.050	- J.	32200	
INC	5557 CANAL BLVD		NEW ORLEANS	LA	70124	
LAUDERDALE COUNTY	TAX COLLECTOR	P.O. BOX 5205	MERIDIAN	MS	39302-5205	
				1	32222 2200	
LAUDERHILL MALL INVESTMENT, LLC	1267 N.W. 40TH AVE		LAUDERHILL	FL	33313	
LAVIGNE, LARAY	REDACT		REDACT	REDACT	REDACT	

Name	Address 1	Address 2	City	State	PostalCode	Country
LAW ENFORCEMENT PUBLICATIONS	2348 WEST A.J. HWY # 173		MORRISTOWN	TN	37814	
LAWNDALE PLAZA LTD PARTNERS	3500 W ROOSEVELT ROAD		CHICAGO	IL	60624	
LAWRENCE, SHEREKA	REDACT		REDACT	REDACT	REDACT	
LAWSON, LATANYA	REDACT		REDACT	REDACT	REDACT	
LAWSON, ZAKIYYIAH	REDACT		REDACT	REDACT	REDACT	
LBUBS 2002-C2 RED BIRD LLC						
(SQUARE 67 LIMITED PARTNERSHIP)	2550 West Red Bird Lane		Dallas	TX	75237	
LBUBS 2002-C2 RED BIRD, LLC	C/O WOODMONT COMPANY	2100 W. 7TH STREET	FT. WORTH	TX	76107	
LBUBS 2002-C2 RED BIRD, LLC	C/O WOODMONT COMPANY	2100 W. 7TH STREET	FT WORTH	TX	76107	
LBUBS 2007-C1 COMPLEX 2740, LLC	C/O THE WOODMONT COMPANY	2100 W. 7TH ST	FT WORTH	TX	76107	
LBUBS 2007-C1 CONNER RETAIL LLC	C/O FRIEDMAN MGMT CO.	34974 WEST 12 MILE RD.	FARMINGTON HILLS	MI	48331	
LBUBS 2007-C1 CONNER RETAIL, LLC	C/O FRIEDMAN MGMT. CO.	Erica Bruce, Property Manager	Farmington Hills	MI	48331	
LC APPAREL INC	1370 BROADWAY	SUITE 1212	NEW YORK	NY	10018	
LEACH, KATELYN	REDACT		REDACT	REDACT	REDACT	
LEACH, SHAWNA	REDACT		REDACT	REDACT	REDACT	
LEALI, TENAI	REDACT		REDACT	REDACT	REDACT	
LEBRON-ARROYO, JOSE L.	REDACT		REDACT	REDACT	REDACT	
LEDEE, APRIL	REDACT	REDACT	REDACT	REDACT	REDACT	
LEE, SAVANNAH	REDACT		REDACT	REDACT	REDACT	
LEE, SEBRENA	REDACT		REDACT	REDACT	REDACT	
LEE, YIKIMA	REDACT		REDACT	REDACT	REDACT	
LEEDS WATER	8651 Thornton Avenue		Leeds	AL	35094	
LEFLORE COUNTY	TAX COLLECTOR	P.O. BOX 1349	GREENWOOD	MS	38935	
LEG APPAREL	65 RAILROAD AVE	STE# 209	RIDGEFIELD	NJ	07657	
LEGACY WYANDOTTE, LLC	C/O LEGACY ASSET MGMT, LLC	4717 CENTRAL STREET	KANSAS CITY	МО	64112	
LEGACY WYANDOTTE, LLC	4717 Central Street		Kansas City	МО	64112	
LEGEND APPAREL LLC	1942 E 46TH ST		VERNON	CA	90058	
LEHMATH, CARRIE-BETH	REDACT		REDACT	REDACT	REDACT	
LEHMBERG CROSSING PARTNERS,						
LLC	C/O SOUTHEAST PROPERTIES, LLC	100 CALUMET GARDENS	MADISON	MS	39110	
LEHMBERG CROSSING PARTNERS,	,					
LLC	C/O Southeast Properties, LLC	100 Calumet Gardens	Madison	MS	39110	

Name	Address 1	Address 2	City	State	PostalCode	Country
LEHR MIDDLEBROOKS & VREELAND,						
PC	PO BOX 11945		BIRMINGHAM	AL	35202-1945	
		Herbert W.A. Thiele, County				
Leon County	Attn: Consumer Protection Agency	Attorney	Tallahassee	FL	32301	
LEONARD, LATIA	REDACT		REDACT	REDACT	REDACT	
LETIZIA, SANDRA	REDACT		REDACT	REDACT	REDACT	
LEVERT, SIMONE	REDACT	REDACT	REDACT	REDACT	REDACT	
LEVIN PROPERTIES, LP	C/O LEVIN MANAGEMENT CORP.	PO BOX 326	PLAINFIELD	NJ	07061-0326	
LEVIN PROPERTIES, LP	C/O LEVIN MANAGEMENT CORP.	P O BOX 326	PLAINFIELD	NJ	07061-0326	
LEVITTOWN, LP	C/O DLC Management Corporation	ATTN: General Counsel	Tarrytown	NY	10591	
LEWIS ELECTRIC 1 INC	399 WAREHOUSE RD.		OAK RIDGE	TN	37830	
LEWIS, DONYELL	REDACT		REDACT	REDACT	REDACT	
LEWIS, EBONEE	REDACT		REDACT	REDACT	REDACT	
LEWIS, GERDELL	REDACT		REDACT	REDACT	REDACT	
LEWIS, KELAUNA	REDACT		REDACT	REDACT	REDACT	
LEWIS, KRAIG	REDACT		REDACT	REDACT	REDACT	
LEWIS, SIRRANVILLE	REDACT		REDACT	REDACT	REDACT	
LEWIS, VIRGINIA	REDACT	REDACT	REDACT	REDACT	REDACT	
LG EDGEBROOK, LLC	c/o Leon Capital Group	2311 Cedar Springs	Dallas	TX	75201	
LG&E	820 West Broadway		Louisville	KY	40202	
LG&E	PO BOX 9001960		LOUISVILLE	KY	40290-1960	
LIBERTY BANK	PO Box 60131		New Orleans	LA	70160-0131	
Liberty Mutual Insurance Company						
(TH7651290289015)	175 Berkeley St		Boston	MA	02116-0000	
LIBERTY POWER	25901 NETWORK PLACE		CHICAGO	IL	60673-1259	
LIBERTY POWER HOLDINGS, LLC	25901 NETWORK PLACE		CHICAGO	IL	60673-1259	
LIBERTY UTILITIES ARKANSAS	75 REMITTANCE DRIVE		CHICAGO	IL	60675-6602	
LIBERTY UTILITIES ARKANSAS	75 REMITTANCE DRIVE, Ste. 6602		CHICAGO	IL	60675-6602	
LIMELIGHT	921 CANADA COURT		CITY OF INDUSTRY	CA	91748	
LINCOLN CENTER	100 N. Pond Drive	Suite F	Walled Lake	MI	48390	
LINCOLN PARISH	P.O. BOX 863		RUSTON	LA	71273-0863	
LINCOLN PARISH	SHERIFF / TAX COLLECTOR	P.O. BOX 2070	RUSTON	LA	71273-2070	
LINCOLN PARISH POLICE JURY	PO BOX 979		RUSTON	LA	71273-0979	

Name	Address 1	Address 2	City	State	PostalCode	Country
Lincoln Parish Sales & Use Tax						,
Commission	PO Box 863		Ruston	LA	71273-0863	
LINCOLN SHOPPING CENTER	100 N. POND DRIVE	SUITE F	WALLED LAKE	MI	48390	
LINCOLN SHOPPING CENTER	100 N. POND DR.	SUITE F	WALLED LAKE	MI	48390	
LINCOLN SHOPPING CENTER	a Division of Lincoln Center	100 N. Pond Drive	Walled Lake	MI	48390	
LINDA B. GORE	PO BOX 205		GADSDEN	AL	35902	
LINDA KELLEY	92 CR 173		CORINTH	MS	38834	
LINDELL MARKET PLACE, L.P.	c/o Staenberg Group, Inc.	2127 Innerbelt Business Center Drive	St. Louis	МО	63114	
LINDELL MARKET PLACE, LP	C/O STAENBERG GROUP, INC.	2127 INNERBELT BUSINESS CT DR	ST LOUIS	МО	63114	
LITKE, BRENDA	REDACT		REDACT	REDACT	REDACT	
LITTLE, DINISTY	REDACT		REDACT	REDACT	REDACT	
LITTLE, NAPORCSHA	REDACT		REDACT	REDACT	REDACT	
LIVONIA PHOENIX DEVELOPER						
PARCEL LLC	38500 Woodward Avenue	Suite 200	Bloomfield Hills	MI	48304	
LIVONIA PHOENIX DEVELOPER						
PARCEL, LLC	C/O LORMAX STERN DVLP CO.	38500 WOODWARD AVE.	BLOOMFIELD HILLS	MI	48304	
LIVONIA PHOENIZ PARCEL, LLC.	38500 WOODWARD AVENUE		BLOOMFIELD HILLS	MI	48304	
	38500 WOODWARD AVENUE, SUITE					
LIVONIA PHOENIZ PARCEL, LLC.	200		BLOOMFIELD HILLS	MI	48304	
LJ ACCESSORIES	140 CANDACE DR.		MAITLAND	FL	32751	
LM TIC I LLC, LM TIC II LLC, LM TIC III		c/o New Lauderhill Mall				
LLC and LM TIC IV LLC	LAUDERHILL MALL/Store No182	Management LLC	Lauderhill	FL	33313	
LM TIC I LLC, LM TIC II LLC, LM TIC III	c/o New Lauderderhill Mall					
LLC, and LM TIC IV LLC	Management, LLC	1267 NW 40th Avenue	Lauderhill	FL	33313	
LN 2000, LLC	C/O NAJCO, INC	56 EAST ANDREWS DRIVE	ATLANTA	GA	30305	
LOCKE BARKLEY, CHPTR 13 TRUSTEE	P.O. BOX 1859		MEMPHIS	TN	38101-1859	
LOCKHART, JALESSA	REDACT		REDACT	REDACT	REDACT	
LOCKHART, KANDISS	REDACT		REDACT	REDACT	REDACT	
LOCKHART, KANDISS	REDACT		REDACT	REDACT	REDACT	
LOCKHART, MARY	REDACT		REDACT	REDACT	REDACT	
LOCKOUT EXPRESS	228 W. LINCOLN HWY	#181	SCHERERVILLE	IN	46375	
LOGAN, CORRETTA	REDACT		REDACT	REDACT	REDACT	
LOGAN, JABRIRIL	REDACT		REDACT	REDACT	REDACT	
LOGGINS, JOI	REDACT		REDACT	REDACT	REDACT	
LOGGINS, JOI	REDACT		REDACT	REDACT	REDACT	

Name	Address 1	Address 2	City	State	PostalCode	Country
LOKSAFE SERVICES & DOOR						
HARDWARE, INC	1219 GRIMMETT DR.		SHREVEPORT	LA	71107	
LONDON, NOLA	REDACT		REDACT	REDACT	REDACT	
LONGANECKER, TRACI	REDACT	REDACT	REDACT	REDACT	REDACT	
LONGS, ANDREEL	REDACT		REDACT	REDACT	REDACT	
LOPEZ, ALYSSA	REDACT	REDACT	REDACT	REDACT	REDACT	
LOPEZ, ANJELIKA	REDACT		REDACT	REDACT	REDACT	
LOPEZ-YANEZ, ASHLEY	REDACT		REDACT	REDACT	REDACT	
LOR CORPORATION	6350 RUCKER RD, SUITE 101		INDIANAPOLIS	IN	46220	
LOR Corporation	CROSS CREEK/Store No365	6350 Rucker Rd, Suite 101	Indianapolis	IN	46220	
LOR CORPORATION	6350 Rucker Road	Suite 101	INDIANAPOLIS	IN	46220	
LOREDO, LUISA	REDACT	REDACT	REDACT	REDACT	REDACT	
LORENZI, ELIANA	REDACT		REDACT	REDACT	REDACT	
LOU ANN PINDER	11717 14TH STREET	APT E-6	TAMPA	FL	33612	
LOUANA AL-AMM	3111 HOKENDAUQUA ST		WHITEHALL	PA	18052	
LOUISE PARIS LTD	1407 BROADWAY,SUITE 1405		NEW YORK	NY	10018	
Louisiana Department of Revenue	PO Box 201		Baton Rouge	LA	70821-0201	
Louisiana Department of Revenue	900 Murray Street		Alexandria	LA	71301-7610	
LOUISIANA DEPT OF AGRICULTURE	DIV. OF WEIGHTS & MEASURES	P.O. BOX 91081	BATON ROUGE	LA	70821-9081	
LOUISIANA DEPT OF REVENUE	PO BOX 3138		BATON ROUGE	LA	70821-3138	
LOUISVILLE / JEFFERSON COUNTY	REVENUE COMMISSION	PO BOX 35410	LOUISVILLE	KY	40232-5410	
LOUISVILLE FARP	633 WEST JEFFERSON ST.		LOUISVILLE	KY	40202	
LOUISVILLE METRO REVENUE						
COMMISSION	PO BOX 35410		LOUISVILLE	KY	40232-5410	
LOUISVILLE WATER COMPANY	550 South Third Street		Louisville	KY	40202	
LOUISVILLE WATER COMPANY	PO BOX 32460		LOUISVILLE	KY	40232-2460	
LOVELY 153	2840 E 11TH ST		LOS ANGELES	CA	90023	
LOVETT, SAMBRIA	REDACT		REDACT	REDACT	REDACT	
LOWER IMPROVEMENTS LLC	C/O DLC MGMT CORP	PO BOX 5965	HICKSVILLE	NY	11802-5965	
LOWNDES COUNTY	TAX COLLECTOR	P.O. BOX 1077	COLUMBUS	MS	39703	
LOWNDES COUNTY,GA.	P.O.BOX 1409		VALDOSTA	GA	31603	
LOZA, LOURDES	REDACT		REDACT	REDACT	REDACT	
LOZADA, YOLANDA	REDACT		REDACT	REDACT	REDACT	
LR MEDIA GROUP, INC.	430 E. 162ND STREET, #432		SOUTH HOLLAND	IL	60473	

Name	Address 1	Address 2	City	State	PostalCode	Country
LR TRADING	1385 BRAODWAY	ROOM 2109	NEW YORK	NY	10018	
LSREF2 CLOVER PROPERTY 18, LLC	UNIVERSITY MALL	PO BOX 864857	ORLANDO	FL	32886-4857	
LUCAS, BRANDON	REDACT		REDACT	REDACT	REDACT	
LUCAS, JABARRE	REDACT		REDACT	REDACT	REDACT	
LUCAS, MICHELLE	REDACT		REDACT	REDACT	REDACT	
LUCENT PRODUCT INC	5515 DANIELS ST		CHINO	CA	91710	
LUNDY, ANGELA	REDACT		REDACT	REDACT	REDACT	
LUS-LAFAYETTE UTILITIES SYSTEM	705 W. University Ave		Lafayette	LA	70506	
LUS-LAFAYETTE UTILITIES SYSTEM	PO BOX 4024		LAFAYETTE	LA	70502-4024	
LUV ME MORE CORP.	525 7TH AVE, RM 712		NEW YORK	NY	10018	
LUX	362 FIFTH AVE	SUITE 601	NEW YORK	NY	10001	
LYNCH, SHIREKA	REDACT		REDACT	REDACT	REDACT	
LYNES, DWAYNE	REDACT		REDACT	REDACT	REDACT	
LYNNHAVEN SPE, LLC and BARTON						
LYNNHAVEN, LLC	Bell Partners, Inc.	300 N. Greene Street	Greensboro	NC	27401	
LYONS, CRYSTAL	REDACT		REDACT	REDACT	REDACT	
LYONS, DOUGHTY & VELDHUIS, PC	PO BOX 1269		MT. LAUREL	NJ	08054	
LYONS, KEYONTHA	REDACT		REDACT	REDACT	REDACT	
M & M WEST TN INVESTMENT LP	C/O M & M DEVELOPMENT CO. INC	11235 WEST POINT DRIVE, STE 1	KNOXVILLE	TN	37934	
M & P HOLDINGS, LLC	5025-M WINTERS CHAPEL RD		ATLANTA	GA	30360	
M&A CLOTHING/KAKTUS						
SPORTSWEAR	1407 BROADWAY		NEW YORK	NY	10018	
M&A IMPORTS	7050 NEW HORIZONS BLVD		NORTH AMITYVILLE	NY	11701	
M&M West Tennessee Investments						
LP	SHELBY PLAZA/Store No558	11235 West Point Drive Ste 1	Knoxville	TN	37934	
M&M WEST TENNESSEE						
INVESTMENTS, LP	C/O M & M DVLP. CO., INC.	11235 West Pointe Dr.	Knoxville	TN	37934	
M&T BANK	101 South Salina Street		Syracuse	NY	13202	
M.D. OUTDOOR, LLC	P.O. Box 1197		Laurel	MS	39441	
M.I.S.S. SPORTSWEAR	1384 BROADWAY		NEW YORK	NY	10018	
MABIN, MARICE	REDACT		REDACT	REDACT	REDACT	
MAC PAPERS	PO BOX 930513		ATLANTA	GA	31193-0513	
MACIAS, ELENA	REDACT		REDACT	REDACT	REDACT	
MACK, EVAN	REDACT		REDACT	REDACT	REDACT	
MACK, LASHERIA	REDACT	REDACT	REDACT	REDACT	REDACT	

Name	Address 1	Address 2	City	State	PostalCode	Country
MACKEY, JUSTIN	REDACT		REDACT	REDACT	REDACT	,
MACON, ALEXANDRIA	REDACT		REDACT	REDACT	REDACT	
MACON/BIBB CO.	682 CHERRY STREET		MACON	GA	31201	
,						
MADISON AVENUE COMMONS, LLC	ATTN: MANNY MALEKAN	48 E. OLD COUNTRY RD	MINEOLA	NY	11501	
MADISON AVENUE COMMONS, LLC	ATTN: MANNY MALEKAN	48 E OLD COUNTY RD	MINEOLA	NY	11501	
MADISON AVENUE COMMONS, LLC	48 E. Old Country Road	Suite 203	Mineola	NY	11501	
MADISON BRANDS, INC.	1407 BROADWAY, STE 1204		NEW YORK	NY	10018	
Madison County	Attn: Consumer Protection Agency	Jeff Rich - County Attorney	Huntsville	AL	35801	
MADISON COUNTY, AL	SALES TAX DIVISION	MADISON COUNTY COURTHOUSE	HUNTSVILLE	AL	35801-4820	
MADISON COUNTY, AL	TAX COLLECTOR	100 NORTHSIDE SQUARE	HUNTSVILLE	AL	35801-4820	
MADISON SQUARE ASSOC., LTD	C/O CBL & ASSOC.	LEASE ID LSIMPFA0	CLEVELAND	ОН	44194-4343	
MADISON SQUARE ASSOCIATES,						
LTD.	CBL & Associates Management, Inc.	Attn: Joe Black	Huntsville	AL	35806	
MADISON, AMBER	REDACT		REDACT	REDACT	REDACT	
MADISON, CHEMEKIA	REDACT		REDACT	REDACT	REDACT	
MAGIC STITCHES, LLC	3131 4TH AVENUE SOUTH		BIRMINGHAM	AL	35233	
MAGISTRATE COURT OF CHATHAM						
COUNTY	133 MONTGOMERY STREET	ROOM 303	SAVANNAH	GA	31401	
MAGISTRATE COURT OF						
DOUGHERTY CO	225 PINE AVENUE	RM 308	ALBANY	GA	31701	
MAGISTRATE CT OF BIBB CO	CLERKS OFFICE ROOM 101	BIBB CO COURTHOUSE	MACON	GA	31201	
MAGLAS, CARLO	REDACT		REDACT	REDACT	REDACT	
Mahavir International, Inc.	2110 N.W. 95th Avenue		Miami	FL	33172	
MAHOGANY CHICAGO 47, LLC	C/O SKILKEN	4270 MORSE RD	COLUMBUS	ОН	43230	
MAHOGANY CHICAGO 47, LLC	4270 Morse Road		Columbus	ОН	43230	
MAHONE, ARETHA	REDACT		REDACT	REDACT	REDACT	
MAHONING COUNTY	50 WESTCHESTER DR	ROOM 201	YOUNGSTOWN	ОН	44515	
Mahudi International Corp.	2110 N.W. 95th Avenue		Miami	FL	33172	
MAHUDI INT'L (APPAREL)	2110-12 NW 95TH AVENUE		MIAMI	FL	33172	
MAIL FINANCE	ATTN: CASH APPLICATIONS	478 WHEELERS FARMS ROAD	MILFORD	СТ	06461	
MAIN STREET LOUNGEWEAR INC	10 EAST 33RD STREET	10TH FLOOR	NEW YORK	NY	10016	
MAJURIE, LEQUANDA	REDACT		REDACT	REDACT	REDACT	
MAKAE, HEATHER	REDACT		REDACT	REDACT	REDACT	

Name	Address 1	Address 2	City	State	PostalCode	Country
MALLET, KENDALL	REDACT		REDACT	REDACT	REDACT	
MALLORY, CLORESSA	REDACT		REDACT	REDACT	REDACT	
MALLOY, AMY	REDACT		REDACT	REDACT	REDACT	
MALONE, BERITA	REDACT	REDACT	REDACT	REDACT	REDACT	
MALONE, BRITTNEE	REDACT		REDACT	REDACT	REDACT	
MALONE, MARVIN	REDACT		REDACT	REDACT	REDACT	
MAM, CAMILLE	REDACT		REDACT	REDACT	REDACT	
MANHATTAN VILLAGE ,LLC	P.O. BOX 252451		WEST BLOOMFIELD	MI	48325-2451	
MANN, REGINALD	REDACT	REDACT	REDACT	REDACT	REDACT	
MANNING, KAREON	REDACT		REDACT	REDACT	REDACT	
MANNING, ROZELL	REDACT		REDACT	REDACT	REDACT	
MANSON, CLEOPATRA	REDACT		REDACT	REDACT	REDACT	
MANSON, NATHAN	REDACT		REDACT	REDACT	REDACT	
MANSON, NATHAN	REDACT		REDACT	REDACT	REDACT	
MANTLE, TIFFANY	REDACT		REDACT	REDACT	REDACT	
MANUEL, HOPE	REDACT	REDACT	REDACT	REDACT	REDACT	
MANY, MONICA	REDACT		REDACT	REDACT	REDACT	
MAPLE DRIVE PARTNERS, LLLP	3151 Maple Drive, N.E.		Atlanta	GA	30305	
MAPLE DRIVE PARTNERS, LP	ATTN: TOM PALMER	3151 MAPLE DRIVE, N.E.	ATLANTA	GA	30305	
MARDIS, PHILLIP	REDACT		REDACT	REDACT	REDACT	
MARIA HERNANDEZ	224 SHEFFIELD CT		HOOVER	AL	35226	
MARILYN HILLIARD	27565 FRANKLIN RD, #202		DETROIT	MI	48034	
MARION COUNTY SMALL CLAIMS						
CRT	PIKE TOWNSHIP DIVISION	5665 LAFAYETTE RD, STE B	INDIANAPOLIS	IN	46254	
MARION COUNTY TREASURER	PO BOX 6145		INDIANAPOLIS	IN	46206-6145	
Mark Barstein	P.O. Box 13096		Birmingham	AL	35202	
MADE COLLADE DARTNERS	C/O CLINIDICE COMMA DEVI DMT CO	PO BOX 160544	MOBILE	A.1	36616	
MARKS SQUARE PARTNERS	C/O SUNRISE COMM DEVLPMT CO REDACT	PO BOX 160544	REDACT	AL REDACT	REDACT	-
MARLER, KEEFER	2120 L STREET NW	CLUTE 000		DC	20037	-
MARLOW HEIGHTS S/C LTD. MARLOW HEIGHTS SHOPPING	ZIZU L STREET INW	SUITE 800	WASHINGTON	DC	20037	
CENTER LP	2120 L Street NW, Suite 800		Mashington	DC	20037	
CENTER LP	2120 L Street NW, Suite 800		Washington	DC	20037	
MARQ PACKAGING SYSTEMS, INC.	P.O. BOX 9063		YAKIMA	WA	98909	
MARSH, CHANELL	REDACT		REDACT	REDACT	REDACT	
MARSH, PAMELA	REDACT		REDACT	REDACT	REDACT	
MARSH, SHERLENE	REDACT		REDACT	REDACT	REDACT	
MARSHALL, DESHADA	REDACT		REDACT	REDACT	REDACT	

Name	Address 1	Address 2	City	State	PostalCode	Country
MARSHALL, MARQUETTA	REDACT		REDACT	REDACT	REDACT	
MARSHALL, RICHAARD	REDACT	REDACT	REDACT	REDACT	REDACT	
MARTIN, ALIYAH	REDACT	REDACT	REDACT	REDACT	REDACT	
MARTIN, DEBRA	REDACT		REDACT	REDACT	REDACT	
MARTIN, HARLEQUIN	REDACT		REDACT	REDACT	REDACT	
MARTIN, JASMINE	REDACT		REDACT	REDACT	REDACT	
MARTIN, LYNSI	REDACT		REDACT	REDACT	REDACT	
MARTIN, MANDY	REDACT		REDACT	REDACT	REDACT	
MARTIN, TAMARA	REDACT		REDACT	REDACT	REDACT	
MARTIN, TONYA	REDACT		REDACT	REDACT	REDACT	
MARTIN, VANESSA	REDACT		REDACT	REDACT	REDACT	
MARTIN, WILLIE	REDACT		REDACT	REDACT	REDACT	
MARTINEZ, DESIREE	REDACT		REDACT	REDACT	REDACT	
MARY JANE DESIGNS	2376 60TH STREET	ATTN: ACCOUNTS RECEIVABLE	BROOKLYN	NY	11204	
MARYLAND COMPTROLLER OF						
TREASURY	SALES & USE TAX	110 CARROLL ST.	ANNAPOLIS	MD	21411-0001	
	C/O ASHKENAZY ACQUISITION					
MARYLAND CROSSING REALTY, LLC	CORP.	150 EAST 58TH ST.	NEW YORK	NY	10155	
MARYLAND DEPT OF ASSESSMENTS	PERSONAL PROPERTY DIVISION	301 WEST PRESTON ST. RM 801	BALTIMORE	MD	21201-2395	
Maryland State Comptroller	80 Calvert Street	PO Box 466	Annapolis	MD	21404-0466	
MASHATT, ASHLEY	REDACT		REDACT	REDACT	REDACT	
MASON, ANTHONY	REDACT		REDACT	REDACT	REDACT	
MASON, CAROLYN	REDACT		REDACT	REDACT	REDACT	
MASON, KIMBERLY	REDACT		REDACT	REDACT	REDACT	
MASON, PAULA	REDACT		REDACT	REDACT	REDACT	
MASON-NORWOOD, SHANTINA	REDACT		REDACT	REDACT	REDACT	
Massachusetts Department of						
Revenue	436 Dwight Street		Springfield	MA	01103	
MASTERS MECHANICAL CORP	815-E BLUE CRAB ROAD		NEWPORT NEWS	VA	23606	
MATAHARI	8335 DEVONSHIRE RD.		MONTREAL	QUEBEC	H4P 2L1	
MATANKY REALTY GROUP						
(SIGNAGE)	200 N. LASALLE STREET # 2350		CHICAGO	IL	60601-1044	
MATEO, NOELIA	REDACT		REDACT	REDACT	REDACT	
MATHEWS, KIERA	REDACT		REDACT	REDACT	REDACT	
MATTHEWS, DEMARCUS	REDACT		REDACT	REDACT	REDACT	
MATTHEWS, JANELL	REDACT		REDACT	REDACT	REDACT	
MAVERICK APPAREL	1407 BROADWAY, STE 3501		NEW YORK	NY	10018	

Name	Address 1	Address 2	City	State	PostalCode	Country
MAXIE, BRANDON	REDACT		REDACT	REDACT	REDACT	
MAXIE, MICHAEL	REDACT		REDACT	REDACT	REDACT	
MAXIMOTO, JOCELYN	REDACT		REDACT	REDACT	REDACT	
MAXWELL, KENYA	REDACT		REDACT	REDACT	REDACT	
MAXWELL, NEKYIA	REDACT		REDACT	REDACT	REDACT	
MAYA APPAREL	15007 BURIN AVE		LAWNDALE	CA	90260	
MAYERS, DAPHANA	REDACT	REDACT	REDACT	REDACT	REDACT	
MAYFIELD, LAURA	REDACT		REDACT	REDACT	REDACT	
MAZE, LAVERNE	REDACT		REDACT	REDACT	REDACT	
MAZUMA CREDIT UNION	9300 Troost Ave.		Kansas City	МО	64131-3000	
MBD PROPERTIES, INC. (ANDERSON						
PLAZA, LLC and 827 BROADWAY, LLC	P.O. BOX 3749		Greenville	sc	29608	
MC GLASS COMPANY	5915 DE MOSS		HOUSTON	TX	77081	
MCBH CHURCH SQUARE, LLC	C/O MCB REAL ESTATE, LLC	2328 WEST JOPPA ROAD	LUTHERVILLE	MD	21093	
MCBH Church Square, LLC	CHURCH SQUARE S/C/Store No12	c/o MCB Real Estate, LLC	Baltimore	MD	21218	
MCBH CHURCH SQUARE, LLC	c/o MCB Real Estate LLC	2701 North Charles Street	Baltimore	MD	21218	
MCBOUNDS, JEANETTE	REDACT		REDACT	REDACT	REDACT	
MCBRIDE, MICHAEL	REDACT		REDACT	REDACT	REDACT	
MCBRIDE, TAMIKA	REDACT	REDACT	REDACT	REDACT	REDACT	
MCCAIN, JASMINE	REDACT		REDACT	REDACT	REDACT	
MCCALEBB, SHONEKA	REDACT		REDACT	REDACT	REDACT	
MCCALL, DONNA	REDACT	REDACT	REDACT	REDACT	REDACT	
MCCANDLESS, ERICA	REDACT		REDACT	REDACT	REDACT	
MCCANTS, JACKIE	REDACT	REDACT	REDACT	REDACT	REDACT	
MCCASKEL, JOHNNIE	REDACT		REDACT	REDACT	REDACT	
MCCAULEY, TOSHA	REDACT		REDACT	REDACT	REDACT	
MCCLAIN, LATAVIA	REDACT		REDACT	REDACT	REDACT	
MCCLAIN, MICHELLE	REDACT		REDACT	REDACT	REDACT	
MCCLENDON, CASSANDRA	REDACT		REDACT	REDACT	REDACT	
MCCLEOD, MARTENIQUE	REDACT		REDACT	REDACT	REDACT	
MCCLEOD, SARAH	REDACT		REDACT	REDACT	REDACT	
MCCONNICO, TANAJAH	REDACT	REDACT	REDACT	REDACT	REDACT	
MCCORMICK, BARBARA	REDACT		REDACT	REDACT	REDACT	
MCCOY, DEMORRIS	REDACT		REDACT	REDACT	REDACT	
MCCRANEY, SHAWNITA	REDACT		REDACT	REDACT	REDACT	
MCCULLERS, AMENDU	REDACT		REDACT	REDACT	REDACT	

Name	Address 1	Address 2	City	State	PostalCode	Country
MCCULLOUGH, BRIA	REDACT		REDACT	REDACT	REDACT	
MCCULLUM, CHAMIA	REDACT		REDACT	REDACT	REDACT	
MCCULLUM, DALILA	REDACT		REDACT	REDACT	REDACT	
MCDADE, LINZZY	REDACT		REDACT	REDACT	REDACT	
MCDANIEL REFRESHMENTS						
SOLUTIONS	2145 ALTON ROAD		IRONDALE	AL	35210	
MCDANIEL, ARMAZER	REDACT		REDACT	REDACT	REDACT	
MCDONALD, MICHAEL	REDACT	REDACT	REDACT	REDACT	REDACT	
MCDUFFEY, TYSON	REDACT	REDACT	REDACT	REDACT	REDACT	
MCFADDEN, DANIEL	REDACT	REDACT	REDACT	REDACT	REDACT	
MCGEE, BRENDA	REDACT		REDACT	REDACT	REDACT	
MCGEE, MONICA	REDACT		REDACT	REDACT	REDACT	
MCGEE, RONNIE	REDACT		REDACT	REDACT	REDACT	
MCGRADY, PRINCESS	REDACT		REDACT	REDACT	REDACT	
MCGRADY, PRINCESS	REDACT		REDACT	REDACT	REDACT	
MCGRIFF, NYESHA	REDACT		REDACT	REDACT	REDACT	
MCGRIFF, SEIBELS & WILLIAMS INC. MCGUEEN, CHERYL	DRAWER #456 REDACT	P.O. BOX 11407	BIRMINGHAM REDACT	AL REDACT	35246-0001 REDACT	
MCGUIRE, CODY	REDACT		REDACT	REDACT	REDACT	
MCGUIRE, LAKITA	REDACT		REDACT	REDACT	REDACT	
MCGUIRE, LAKITA	REDACT		REDACT	REDACT	REDACT	
MCGUIRE, WENDY	REDACT		REDACT	REDACT	REDACT	
MCHAHON, JARAY	REDACT		REDACT	REDACT	REDACT	
MCINTYER, GRACE	REDACT		REDACT	REDACT	REDACT	
MCJ IMPROVEMENTS, LLC	C/O DLC MGMT CORP	PO BOX 5122	WHITE PLAINS	NY	10602	
MCJ IMPROVEMENTS, LLC	c/o DLC Management Corporation	580 White Plains Road	Tarrytown	NY	10591	
MCKENNA, LONG & ALDRIDGE, LLP	ATTORNEYS AT LAW	PO BOX 116573	ATLANTA	GA	30368	
MCKENZIE, ROGER	REDACT		REDACT	REDACT	REDACT	
MCKENZIE, SHAMANEKA	REDACT	REDACT	REDACT	REDACT	REDACT	
MCKINNEY, BREAUNA	REDACT		REDACT	REDACT	REDACT	
MCKINNON, KENNETH	REDACT		REDACT	REDACT	REDACT	
MCLAIN, TIA	REDACT		REDACT	REDACT	REDACT	
MCLAREN, MAMIE	REDACT		REDACT	REDACT	REDACT	
MCLAURIN, JEQUNTEY	REDACT		REDACT	REDACT	REDACT	
MCMAHON, CARLETTA	REDACT		REDACT	REDACT	REDACT	

Page 109 of 176

Name	Address 1	Address 2	City	State	PostalCode	Country
MCMILLER, LISA	REDACT	REDACT	REDACT	REDACT	REDACT	
MCNAIR, YDENIA	REDACT		REDACT	REDACT	REDACT	
MCNEAL, CEDRIC	REDACT		REDACT	REDACT	REDACT	
MCNEALEY, MICHELLE	REDACT	REDACT	REDACT	REDACT	REDACT	
MCNEIL, DOMINIQUE	REDACT		REDACT	REDACT	REDACT	
MCPHERSON, ALPHONSO	REDACT		REDACT	REDACT	REDACT	
MCPHERSON, DARAYEL	REDACT		REDACT	REDACT	REDACT	
MCPHILLIPS LAW FIRM PA	P.O. BOX 143406		CORAL GABLES	FL	33114	
MCREAL, SHAHNA	REDACT		REDACT	REDACT	REDACT	
MCREYNOLDS, BLAKE	REDACT		REDACT	REDACT	REDACT	
MCTIER, LATRIA	REDACT		REDACT	REDACT	REDACT	
MCWILLIAMS, IVORY	REDACT		REDACT	REDACT	REDACT	
MD OUTDOOR, LLC	PO BOX 1197		LAUREL	MS	39441	
MDR/SOUTH SLAPPEY LLC	C/O JP PROPERTIES, INC	135 PARKSIDE CLOSE	ALPHARETTA	GA	30022-1437	
MDR/SOUTH SLAPPEY, LLC	C/O JP PROPERTIES, INC	135 Parkside Close	Alpharetta	GA	30022-1437	
MEA BRANDS	145 TALMADGE RD		EDISON	NJ	08817	
		4200 NORTHSIDE PKWY, BLDG 10,				
MEADOWBROOK CROSSING LLC	C/O REDD REALTY SERVICES	STE 101	ATLANTA	GA	30327	
MEADOWBROOK CROSSING LLC	C/O REDD REALTY SERVICES	4200 NORTHSIDE PKWY, BLDG 10	ATLANTA	GA	30327	
MEADOWBROOK SHOPPING CENTER		,				
LIMITED	C/O ISENBERG MGMT ASSOC	400 S. Zang Blvd.	Dallas	TX	75208	
MEADOWBROOK SHOPPING CENTER						
LTD	C/O ISENBERG MGMT ASSOC	400 S. ZANG BLVD, SUITE 1220	DALLAS	TX	75208	
MECHANICAL SYSTEMS & SERVICE	464 INDUSTRIAL PARKWAY		WEST MONROE	LA	71291	
MECKLENBURG COUNTY	TAX COLLECTOR	PO BOX 71063	CHARLOTTE	NC NC	28272-1063	
MEDINA, JOSHUA	REDACT	PO BOX 71063	REDACT	REDACT	REDACT	
MEDLEY, NAQUANDA	REDACT		REDACT	REDACT	REDACT	
MEEKS, CHERICE	REDACT		REDACT	REDACT	REDACT	
MEETU MAGIC INC.	4800 WESTSIDE AVE.		NORTH BERGEN	NJ	07047	
MEGAN A. WILLIAMS	PO BOX 26393		BIRMINGHAM	AL	35260	
MEGAPATH	DEPT 0324	PO BOX 120324	DALLAS	TX	75312-0324	
WEGAFATTI	DLF 1 0324	FO BOX 120324	DALLAS	17	73312-0324	
MegaPath	6800 Koll Center Parkway, Suite 200		Pleasonton	CA	94566	
MegaPath	6800 Koll Center Parkway	Suite 200	Pleasonton	CA	94566	
MEGAWEAR, INC.	1407 BROADWAY, STE. 3505		NEW YORK	NY	10018	
MELENDEZ, KARINA	REDACT		REDACT	REDACT	REDACT	

Name	Address 1	Address 2	City	State	PostalCode	Country
MELENDREZ, KARINA	REDACT		REDACT	REDACT	REDACT	
MELISSA BRY	P.O. Box 685-Midtown Statiom		NEW YORK	NY	10018-0685	
MELISSA IVERSON	9 RIDGE RD		DUDLEY	MA	01571	
MELNEK, AUSTIN	REDACT		REDACT	REDACT	REDACT	
MELVIN S. ROOS AND CO., INC.	P.O. BOX 44689		ATLANTA	GA	30336-5689	
MENDOZA, MAYELY	REDACT	REDACT	REDACT	REDACT	REDACT	
MERAM PROPERTIES, LLC	260 E. BROWN STREET	SUITE 200	BIRMINGHAM	MI	48009	
MERCADO, OMAYRA	REDACT	REDACT	REDACT	REDACT	REDACT	
Merchant Business Credit Inc.	1441 Broadway, 22nd Floor		New York	NY	10018	
		6961 PEACHTREE IND BLVD, STE				
MERCHANT SQUARE S/C, LLC	C/O SAFEWAY GROUP, INC	101	NORCROSS	GA	30092	
MERCHANT SQUARE SHOPPING		6961 Peachtree Industrial				
CENTER, LLC	c/o Safeway Group, Inc.	Boulevard	Norcross	GA	30092	
MERIDIAN CENTER GROUP, LLC	C/O R. ALEX SMITH	PO DRAWER 3308	MERIDIAN	MS	39303	
	MERIDIAN MARKETPLACE/Store					
Meridian Center Group, LLC	No30	601 Union Street	Seattle	WA	98101	
MERIDIAN CENTER GROUP, LLC						
SOUTHERN CUSTOM MALLS, INC.	Michael Hatmaker	471 Harbor Drive South	Indian Rocks Beach	FL	33785	
MERIUS, AUSTIN	REDACT		REDACT	REDACT	REDACT	
MERRITT, PATRICIA	REDACT		REDACT	REDACT	REDACT	
MERRITT, PATRICIA	REDACT		REDACT	REDACT	REDACT	
METAYER, SHEENA	REDACT		REDACT	REDACT	REDACT	
METCALF, ALVIN	REDACT		REDACT	REDACT	REDACT	
METLIFE (PA)	PO BOX 8500-3895		PHILADELPHIA	PA	19178-3895	
METOYER, ASHLEY	REDACT	REDACT	REDACT	REDACT	REDACT	
METRO 22 INC	1400 BROADWAY		NEW YORK	NY	10018	
	MICHAEL F FREEMAN -					
METRO ALARM OFFICE	ADMINISTRATOR	PO BOX 178	MEMPHIS	TN	38101-0178	
METRO DOOR (GREAT RIVER)	3500 SUNRISE HWY	BUILDING 100,STE 210	GREAT RIVER	NY	11739	
METRO FIRE & SAFETY, INC	130 WOLFPACK LANE		DURHAM	NC	27704	
METRO GLASS & DOOR SERVICES	734 PARK MANOR DRIVE		ORLANDO	FL	32825	
METRO HEATING & AIR						
CONDITIONING	717 PHILLIPS AVENUE		TOLEDO	ОН	43612	
METRO LOCK & SECURITY	6000 OLD COLLINSVILLE ROAD		FAIRVIEW HEIGHTS	IL	62208	
METRO MECHANICAL SERVICES	851 NORTH PINE HILL ROAD		BIRMINGHAM	AL	35217	
METRO PLAZA, LLC	1801 EAST 63RD STREET		KANSAS CITY	МО	64130	
METRO PLAZA, LLC	1801 E. 63rd Street		Kansas City	МО	64130	

Name	Address 1	Address 2	City	State	PostalCode	Country
METRO ST. LOUIS SEWER DISTRICT	2350 Market Street		St. Louis	МО	63103-2555	
METRO ST. LOUIS SEWER DISTRICT	PO BOX 437		ST. LOUIS	MO	63166-0437	
METRO WATER SERVICES	1700 3rd Ave N		Nashville	TN	37208	
METRO WATER SERVICES	PO BOX 305225		NASHVILLE	TN	37230-5225	
METROPOLITAN GOVERNMENT	ALARM REGISTRATION	PO BOX 196321	NASHVILLE	TN	37219-6321	
Metropolitan Government of						
Nashville	Director of Law Saul Solomon	Department of Law	Nashville	TN	37219-6300	
METROPOLITAN ST. LOUIS SEWER						
DISTRICT	2350 Market Street		St. Louis	MO	63103-2555	
METROPOLITAN ST. LOUIS SEWER						
DISTRICT	P.O. BOX 437		ST. LOUIS	MO	63166-0437	
METROPOLITAN TRUSTEE	P.O. BOX 305012		NASHVILLE	TN	37230-5012	
MEWBORN, BERNARD	REDACT		REDACT	REDACT	REDACT	
MEYERLAND GLASS & MIRROR CO	12922 MURPHY ROAD		STAFFORD	TX	77477	
MEZZANINE	1329 E 16th St.		LOS ANGELES	CA	90021	
MIA YOUNG	614 EXECUTIVE CENTER DRIVE	#107	WEST PALM BEACH	FL	33401	
MIAMI-DADE COUNTY TAX						
COLLECTOR	TAX COLLECTOR	PO BOX 13701	MIAMI	FL	33101-3701	
MIAMI-DADE FIRE RESCUE DEPT.	FINANCE BUREAU	9300 NW 41ST STREET	MIAMI	FL	33178-2424	
		11500 NW 25TH STREET - 2ND				
MIAMI-DADE POLICE DEPARTMENT	FALSE ALARM ENFORCEMENT UNIT	FLOOR	MIAMI	FL	33172	
MIAMI-DADE WATER & SEWER DEPT	3071 SW 38th Avenue		Miami	FL	33146	
MIAMI-DADE WATER & SEWER DEPT	PO BOX 026055		MIAMI	FL	33102-6055	
MICHCON GAS (1ST USA VISA)	4747 S Broad St		Philadelphia	PA	19112	
MICHELLE TRUETT	51 LEONARD ROAD		LAGRANGE	GA	30241	
Michigan Department of Revenue	PO Box 30324		Lansing	MI	48909-7824	
	430 W Allegan St		Lansing	MI	48933	
Michigan Dept of Treasury	PO Box 30324		Lansing	MI	48909-7824	
MICHIGAN ESC	P.O. BOX 33598		DETROIT	MI	48232-5598	
MICHIGAN GUARANTY AGENCY	P.O. BOX 7074		INDIANAPOLIS	IN	46207-7074	

Name	Address 1	Address 2	City	State	PostalCode	Country
						,
MICHIGAN ST DISBURSEMENT UNIT	PO BOX 30350		LANSING	MI	48909-7850	
MICHIGAN TOWNSHIP SERVICES						
MUSKEGON	5855 AIRLINE		FRUITPORT	MI	49415	
MICKEALS, IMANI	REDACT		REDACT	REDACT	REDACT	
MICKELSON, RAVEN	REDACT		REDACT	REDACT	REDACT	
MICKLE, STARREASHA	REDACT		REDACT	REDACT	REDACT	
MICKLER, CHARLENE	REDACT		REDACT	REDACT	REDACT	
MID MLK IMPROVEMENTS, LLC	ACCT#7527978054	PO BOX 2074	HICKSVILLE	NY	11802	
MID THRUST IMPORTS	830 E 14TH PLACE		LOS ANGELES	CA	90021	
MID-AMERICA ASSET MGMT., INC.	AS RECEIVER FOR GRENOBLE	SQUARE S/C	OAKBROOK TERRACE	IL	60181	
MIDDLE TENNESSEE ELECTRIC	555 New Salem Highway		Murfreesboro	TN	37129	
MIDDLE TENNESSEE ELECTRIC	P.O.BOX 608		MURFREESBORO	TN	37133-0608	
	C/O MARYLAND FINANCIAL					
MIDDLESEX ASSOCIATES LLP	INVESTORS, INC.	2800 Quarry Lake Drive	Baltimore	MD	21209	
	C/O MD FINANCIAL INVESTORS,					
MIDDLESEX ASSOCIATES, LLP	INC.	2800 QUARRY LAKE DRIVE, STE 340	BALTIMORE	MD	21209	
MIDLAND REALTY LLC	10689 N. Pennsylvania St.	Suite 100	INDIANAPOLIS	IN	46280	
MIDLAND REALTY, LLC	10689 N PENNSYLVANIA ST	SUITE 100	INDIANAPOLIS	IN	46280	
MIKE GREENE CAMP FUND	C/O TAMARACK CORPS	6735 TELEGRAPH RD	BLOOMFIELD	MI	48301	
Milberg Factors	99 Park Ave.		New York	NY	10016	
MILES, DEVONTE	REDACT		REDACT	REDACT	REDACT	
MILES, DYONISIA	REDACT		REDACT	REDACT	REDACT	
MILES, RYAN	REDACT		REDACT	REDACT	REDACT	
MILES, TANYANA	REDACT		REDACT	REDACT	REDACT	
MILLEN, BOBBY D.	REDACT		REDACT	REDACT	REDACT	
MILLER, CARICE	REDACT		REDACT	REDACT	REDACT	
MILLER, CONSTANCE	REDACT		REDACT	REDACT	REDACT	
MILLER, RUSSELL	REDACT		REDACT	REDACT	REDACT	
MILLER, VANESSA	REDACT		REDACT	REDACT	REDACT	
MILLS, CYLYNTHIA	REDACT		REDACT	REDACT	REDACT	
MILLSAP, JEVETTE	REDACT		REDACT	REDACT	REDACT	
MILTON, TIFFANY	REDACT		REDACT	REDACT	REDACT	
MIMS, ALEXIA	REDACT		REDACT	REDACT	REDACT	
MINDINGALL, BRENDA	REDACT		REDACT	REDACT	REDACT	
MINGGIA, MARK	REDACT		REDACT	REDACT	REDACT	
MINOR, JANISE	REDACT		REDACT	REDACT	REDACT	

Name	Address 1	Address 2	City	State	PostalCode	Country
MIRAGE	1410 BROADWAY, #505		NEW YORK	NY	10018	
MISS POPULAR	1800-C ATLANTIC AVE		VIRGINIA BEACH	VA	23451	
Mississippi Dept of Revenue	PO Box 1033		Jackson	MS	38215-1033	
MISSISSIPPI POWER	2992 W Beach Blvd		Gulfport	MS	39501	
MISSISSIPPI POWER	PO BOX 245		BIRMINGHAM	AL	35201-0245	
MISSISSIPPI STATE TAX COMM	SALES TAX DIVISION	PO BOX 23075	JACKSON	MS	39225-3075	
Mississippi Tax Commission	P.O. BOX 22808		JACKSON	MS	39225-2808	
MISSISSIPPI VALLEY GAS	1701 N. STATE ST.		JACKSON	MS	39210	
MISSISSIPPI VALLEY GAS	2119 Dalton St		Tupelo	MS	38801	
MISSOURI AMERICAN WATER	727 Craig Road		St. Louis	MO	63141	
MISSOURI AMERICAN WATER	PO BOX 94551		PALATINE	IL	60094-4551	
Missouri Dept of Revenue	Taxation Division	PO Box 840	Jefferson City	MO	65105-0840	
MISSOURI DEPT OF REVENUE	WITHHOLDING TAX	PO BOX 999	JEFFERSON CITY	MO	65108-0999	
MISSOURI DEPT OF REVENUE	SALES & USE TAX	PO BOX 840	JEFFERSON CITY	МО	65105-0840	
MISSOURI GAS ENERGY	3420 Broadway St		Kansas City	МО	64111	
MISSOURI GAS ENERGY	PO BOX 219255		KANSAS CITY	MO	64121-9255	
Missouri Missouri Department of	Harry S Truman State Office					
Revenue	Building	301 West High Street	Jefferson City,	МО	65101	
MISSOURI, SHELANDRA	REDACT		REDACT	REDACT	REDACT	
MITCHELL, AMANDA	REDACT		REDACT	REDACT	REDACT	
MITCHELL, CHERIE	REDACT	REDACT	REDACT	REDACT	REDACT	
MITCHELL, MONIQUE	REDACT	REDACT	REDACT	REDACT	REDACT	
MITCHELL, TRENIKA	REDACT		REDACT	REDACT	REDACT	
MIXON, ALEXIS	REDACT		REDACT	REDACT	REDACT	
MLG&W	245 South Main		Memphis	TN	38103	
MLG&W	PO BOX 388		MEMPHIS	TN	38145-0388	
MLK PLAZA, LP	MLK PLAZA	1045 S. WOODS MILL RD, STE 1	TOWN & COUNTRY	МО	63017	
MLK PLAZA, LP	MLK PLAZA	1045 S. Woods Mills Road	Town & Country	МО	63017	
MMG COLONIAL HOLDINGS, LLC	C/O HORIZON PROPERTIES	7785 NW 146TH ST	MIAMI LAKES	FL	33016	
MMI, LLC	1724 Rudder Industrial Park Dr.		Fennton	МО	63026	
	1724 RUDDER INDUSTRIAL PARK					
MMI, LLC	DRIVE		FENTON	МО	63026	
MO DEPT OF HIGHER EDUCATION	C/O ASA	PO BOX 414533	BOSTON	MA	02241-4533	
MOBILE AREA WATER & SEWER	207 North Catherine Street		Mobile	AL	36604	
MOBILE AREA WATER & SEWER	PO BOX 830130		BIRMINGHAM	AL	35283-0130	
MOBILE COUNTY	REVENUE COMMISSIONER	P.O. BOX 1169	MOBILE	AL	36633-1169	
MOBILE COUNTY- SALES TAX	DEPT# 1524	PO BOX 11407	BIRMINGHAM	AL	35246-1524	
MOBILE GAS	2828 DAUPHIN STREET		MOBILE	AL	36606	

Page 114 of 176

Name	Address 1	Address 2	City	State	PostalCode	Country
MOBILE GAS	PO BOX 105001		ATLANTA	GA	30348-5001	
MODEL T PLAZA, LLC	C/O RODNEY GAPPY	24435 HALSTED RD	FARMINGTON HILLS	MI	48334	
MODEL T PLAZA, LLC	c/o Rodney Gappy	24435 Halsted Road	Farmington Hills	MI	48334	
MOD'LAND WOMEN'S APPAREL	1410 BROADWAY, STE 1901		NEW YORK	NY	10018	
MOLBROUGH, PHILIP	REDACT		REDACT	REDACT	REDACT	
MOLICA, JEAN	REDACT		REDACT	REDACT	REDACT	
MOLTON, ALLEN & WILLIAMS	ATTN: ELIZABETH ENLOE	1000 URBAN CENTER DRIVE	BIRMINGHAM	AL	35242	
MOMENTUM	P.O. BOX 830525		BIRMINGHAM	AL	35210-2096	
MONARCH AT NORTH CHARLESTON,						
LLC	P.O. BOX 930498		ATLANTA	GA	31193-0498	
MONROE, LA'NIK	REDACT		REDACT	REDACT	REDACT	
MONROEVILLE MUNICIPAL						
AUTHORITY	219 SPEELMAN LANE		MONROEVILLE	PA	15146	
MONTGOMERY COUNTY	TAX COLLECTOR	P.O. BOX 1667	MONTGOMERY	AL	36102-1667	
MONTGOMERY COUNTY						
ENVIRONMENTAL SERVICES	1850 Spaulding Road		Kettering	ОН	45432	
MONTGOMERY COUNTY	, ,					
ENVIRONMENTAL SERVICES	PO BOX 742598		CINCINNATI	ОН	45274-2598	
MONTGOMERY TOWNE CENTER						
STATION, INC.	1870 PAYSHERE CIRCLE		CHICAGO	IL	60674	
MONTGOMERY WATER & SEWER	1850 Spaulding Road		Kettering	ОН	45432	
MONTGOMERY WATER & SEWER	PO BOX 1670		MONTGOMERY	AL	36102-1670	
MONTGOMERY WATER WORKS	1850 Spaulding Road		Kettering	ОН	45432	
MONTGOMERY WATER WORKS	P.O. BOX 1670		MONTGOMERY	AL	36102-1670	
MONTGOMERY, CHAD	REDACT		REDACT	REDACT	REDACT	
MONTGOMERY, JAY	REDACT		REDACT	REDACT	REDACT	
MONTGOMERY, MUARION	REDACT		REDACT	REDACT	REDACT	
MONTIEL, JOSEPH	REDACT		REDACT	REDACT	REDACT	
MONTIEL, VANESSA	REDACT		REDACT	REDACT	REDACT	
MOODY PLUMBING, INC.	4100 NW 120 AVE		CORAL SPRINGS	FL	33065	
MOODY, LASHONA	REDACT	REDACT	REDACT	REDACT	REDACT	
MOORE, COURTNEY	REDACT		REDACT	REDACT	REDACT	
MOORE, DAVU	REDACT		REDACT	REDACT	REDACT	
MOORE, JAZZMAN	REDACT	REDACT	REDACT	REDACT	REDACT	
MOORE, KEE'ANA	REDACT		REDACT	REDACT	REDACT	
MOORE, MILTON	REDACT		REDACT	REDACT	REDACT	
MOORE, RAYMOND	REDACT		REDACT	REDACT	REDACT	
MOORE, TY	REDACT		REDACT	REDACT	REDACT	

Name	Address 1	Address 2	City	State	PostalCode	Country
MOORE, VANESSA	REDACT		REDACT	REDACT	REDACT	
MOORE, YASHICA	REDACT		REDACT	REDACT	REDACT	
MOOREHEAD, BRITTANY	REDACT		REDACT	REDACT	REDACT	
MORALES, BRENDA	REDACT		REDACT	REDACT	REDACT	
MORESON CONFERENCING, INC	1116 20TH STREET S. #101		BIRMINGHAM	AL	35205	
MORGAN, GABRIEL	REDACT		REDACT	REDACT	REDACT	
MORGAN, KARI	REDACT		REDACT	REDACT	REDACT	
MORGAN, MICHAEL	REDACT	REDACT	REDACT	REDACT	REDACT	
MORGAN, ZEAON	REDACT		REDACT	REDACT	REDACT	
MORHON, KEVIN	REDACT		REDACT	REDACT	REDACT	
MORMAN, ROLISHA	REDACT		REDACT	REDACT	REDACT	
Morris Satnick Ft. Pierce Associates,						
LLC	K-MART PLAZA/Store No85	3333 New Hyde Park Road Ste 100	New Hyde Park	NY	11042-0020	
		3333 New Hyde Park Road, P.O. Box				
MORRIS SATNICK FT. PIERCE, LLC	formery KIMCO	5020	New Hyde Park	NY	11042-0020	
MORRIS, CRYSTAL	REDACT		REDACT	REDACT	REDACT	
MORRIS, CRYSTAL	REDACT		REDACT	REDACT	REDACT	
MORRIS, GLENN	REDACT		REDACT	REDACT	REDACT	
MORRIS, MELANIE	REDACT		REDACT	REDACT	REDACT	
MORRIS, NENE	REDACT		REDACT	REDACT	REDACT	
MORRIS, NEYANNA	REDACT		REDACT	REDACT	REDACT	
MORRIS, TIKESHIA	REDACT		REDACT	REDACT	REDACT	
MORRIS, TIKESHIA	REDACT		REDACT	REDACT	REDACT	
MORRIS, UNIQUE	REDACT		REDACT	REDACT	REDACT	
MORRIS/SATNICK FT PIERCE ASSOC,						
LLC	ATTN: ACCOUNTING DEPT	350 VETERANS BLVD	RUTHERFORD	NJ	07070	
MORRIS-RILEY, LATOSHA	REDACT		REDACT	REDACT	REDACT	
MORST, TIFFANI	REDACT		REDACT	REDACT	REDACT	
MOSKOS, JR., JERRY	REDACT	REDACT	REDACT	REDACT	REDACT	
MOSS APPAREL GROUP	830 SOUTH HILL STREET	SUITE 100	LOS ANGELES	CA	90014	
MOSS, CEDRICK	REDACT		REDACT	REDACT	REDACT	
MOSS, DEXTER	REDACT		REDACT	REDACT	REDACT	
MOTION INDUSTRIES	PO BOX 404130		ATLANTA	GA	30384	
MOTIVE	807 E 12TH ST, STE 224		LOS ANGELES	CA	90021	
MOTIVIC INC.	236 5TH. AVE. 6TH. FLOOR		NEW YORK	NY	10001	
MOUNTAIN BROOK PORTABLE						
DETAIL	4563 LITTLE RIVER ROAD		BIRMINGHAM	AL	35213-2317	
MOUNTAIN VIEW STATION	NW 601233	PO BOX 1450	MINNEAPOLIS	MN	55485-6012	

Name	Address 1	Address 2	City	State	PostalCode	Country
MR NOAH	LEONARD A. FEINBERG	P.O. BOX 8533	BENSALEM	PA	19020-8533	
MS EMPLOYMENT SECURITY						
COMMISSION	PO BOX 22781		JACKSON	MS	39225-2781	
MSA MODELS	570 7TH AVENUE	SUITE 1000	NEW YORK	NY	10018	
MUELLER, BRITTANY	REDACT		REDACT	REDACT	REDACT	
MUHAMMAD, MECCA	REDACT		REDACT	REDACT	REDACT	
MUHAMMAD, MECCA	REDACT		REDACT	REDACT	REDACT	
MUHAMMED, JABRILA	REDACT		REDACT	REDACT	REDACT	
MULERO, CHELSEY	REDACT		REDACT	REDACT	REDACT	
MULLINS BUILDING PRODUCTS INC	5631-C CLIFFORD CIRCLE		BIRMINGHAM	AL	35210	
MUNICIPALITY OF MONROEVILLE	BUSINESS TAX OFFICE	2700 MONROEVILLE BLVD	MONROEVILLE	PA	15146-2388	
MUNIVE, JESSICA	REDACT		REDACT	REDACT	REDACT	
MURRAY, ANDREA	REDACT		REDACT	REDACT	REDACT	
MURRAY, ANDREA	REDACT		REDACT	REDACT	REDACT	
MURRAY, DANISHA	REDACT		REDACT	REDACT	REDACT	
MURRAY, DANISHA	REDACT		REDACT	REDACT	REDACT	
Muscogee County	Attn: Consumer Protection Agency	Clifton Fay, City Attorney	Columbus	GA	31901	
MUSCOGEE COUNTY	TAX COMMISSIONER	PO BOX 1441	COLUMBUS	GA	31902	
MX ENERGY	100 Constellation Way		Baltimore	MD	21202	
MY OFFICE PRODUCTS(BUSINESS	·					
PROD)	PO BOX 306003		NASHVILLE	TN	37230-6003	
MYA PROPERTIES, LLC	C/O I.M.C. PROPERTIES MGMT, INC.	696 NE 125TH ST	NORTH MIAMI	FL	33161	
MYERS, COURTNEY	REDACT		REDACT	REDACT	REDACT	
MYERS, FLORENCE	REDACT	REDACT	REDACT	REDACT	REDACT	
MYERS, LORETTA	REDACT		REDACT	REDACT	REDACT	
MYRIAD TRADING INC.	1495 S. Hudson Ave		ONTARIO	CA	91761	
MYSTIQUE APPAREL	1410 BROADWAY, SUITE 3103		NEW YORK	NY	10018	
MYTH CLOTHING CO, INC	525 7TH AVE, RM 1010		NEW YORK	NY	10018	
N G & G FACILITY SERVICES INT'L	P.O. BOX 845147		BOSTON	MA	02284-5147	
N STAR GAS	300 Cadwell Dr.		Springfield	MA,	01104-0000	
N.Y. INVASION INC.	132 WEST 36TH STREET		NEW YORK	NY	10018	
NADEAU, CHANTAL	REDACT		REDACT	REDACT	REDACT	
NADLER & ASSOCIATES PC	2871 ACTON ROAD	SUITE 101-A	BIRMINGHAM	AL	35243	
NAGY, BARBARA	REDACT		REDACT	REDACT	REDACT	
NAHANCO	276 WATER STREET	P.O. BOX 818	NORTH BENNINGTON	VT	05257	

Name	Address 1	Address 2	City	State	PostalCode	Country
NALEVANKO, KIM	REDACT		REDACT	REDACT	REDACT	
NANA/MANGO USA	5620 1st AVE (1ST FLOOR)		BROOKLYN	NY	11220	
NANCY J. WHALEY TRUSTEE	SUITE 120	303 PEACHTREE CENTER AVE	ATLANTA	GA	30303	
NASH COUNTY TAX COLLECTOR	PO BOX 1070		CHARLOTTE	NC	28201-1070	
NASH, MATOIA	REDACT	REDACT	REDACT	REDACT	REDACT	
NASHUA CORP	P.O. BOX 802035		CHICAGO	IL	60680-2035	
NASHVILLE ELECTRIC SERVICE	1214 CHURCH STREET		NASHVILLE	TN	37246-0003	
NATALIE KIRKPATRICK	1114 GARRETT AVE		WEST DEPTFORD	NJ	08096	
National Association of Attorneys						
General	Karen Cordry	2030 M St NW 8th Floor	Washington	DC	20036	
NATIONAL CITIES CORPORATION	3794 PARK AVENUE		MEMPHIS	TN	38111	
NATIONAL FUEL GAS	99 River Dr		Garfield	NJ	07026-0000	
NATIONAL FUEL GAS	P.O. BOX 371835		PITTSBURGH	PA	15250-7835	
NATIONAL GRID	40 Sylvan Rd		Waltham	MA	02451-0000	
NATIONAL GRID	P.O. BOX 11735		NEWARK	NJ	07101-4735	
NATIONAL GRID	P.O. BOX 11737		NEWARK	NJ	07101-4737	
NATIONAL GRID	P.O. BOX 11742		NEWARK	NJ	07101-4742	
NATIONAL QUICK CASH #458	8202 S. STONY ISLAND AVE.		CHICAGO	IL	60617	
NATIONAL QUIK CASH	1451 SIBLEY BLVD		CALUMET CITY	IL	60409	
NATIONAL SHOPPIING PLAZAS, INC.	200 West Madison Street	Suite 4200	Chicago	IL	60606-3465	
NATIONAL SHOPPING PLAZAS, INC.	200 W MADISON STREET	SUITE 4200	CHICAGO	IL	60606-3465	
NATIVE INTIMATES	148 MADISON AVE	SUITE 3	NEW YORK	NY	10016	
NAVARRO DE VAZQUEZ, MICAELA	REDACT		REDACT	REDACT	REDACT	
NAVARRO, ZULEIKA	REDACT	REDACT	REDACT	REDACT	REDACT	
NAVI/ACCETECH CO	3007 BANDINI BLVD		VERNON	CA	90058	
NC DEPARTMENT OF REVENUE	NCDOR	P.O. BOX 25000	RALEIGH	NC	27640-0530	
NC DEPT OF REVENUE (NCDOR)	PO BOX 25000		RALEIGH	NC	27640-0150	
NC EMPLOYMENT SECURITY COMM	P.O. Box 25903		RALEIGH	NC	27611-5903	
NEAL, KRYSTLE	REDACT		REDACT	REDACT	REDACT	
NEAL, LATANYA	REDACT		REDACT	REDACT	REDACT	
NEAL, TARNELL	REDACT		REDACT	REDACT	REDACT	
NEESO JEANS	1407 BROADWAY		NEW YORK	NY	10018	
NELSON WOODS, TAMARA	REDACT		REDACT	REDACT	REDACT	
NELSON, BRYANT	REDACT		REDACT	REDACT	REDACT	

Name	Address 1	Address 2	City	State	PostalCode	Country
NES JEWELRY	20TH WEST 33RD ST		NEW YORK	NY	10001	
NESMITH, EBONE	REDACT		REDACT	REDACT	REDACT	
NEW ALBERTSONS INC.	C/O NEW ALBERTSON'S INC	#62572	ST LOUIS	МО	63195-6679	
NEW CLASSIC	47-22 37TH STREET		LONG ISLAND CITY	NY	11101	
New Commercial Capital Inc.	File 749269		Los Angeles	CA	90074-9269	
NEW EASTLAND MALL DEVELOPER	14451 COLLECTIONS CENTER DRIVE		CHICAGO	IL	60693	
NEW EASTLAND MALL DEVELOPERS,						
LLC	ATTN: MALL MANAGEMENT	18000 VERNIER ROAD	HARPER WOODS	MI	48225	
NEW GOLDEN TRADING	1385 BROADWAY		NEW YORK	NY	10018	
NEW JERSEY AMERICAN WATER	1025 Laurel Oak Road		Voorhees	NJ	08043-0000	
NEW JERSEY AMERICAN WATER	BOX 371331		PITTSBURGH	PA	15250-7331	
New Jersey Division of Taxation	Bankruptcy Section,	P.O. Box 245	Trenton	NJ	08695-0245	
NEW LIFE ACCESSORIES	19 W 34TH ST, STE 806		NEW YORK	NY	10001	
NEW ORLEANS (RIVER COMMONS)	C/O RIVERCREST REALTY					
DDP, LLC	ASSOCIATES, LLC	8816 SIX FORKS ROAD	RALEIGH	NC	27615	
	C/O BRE RETAIL RESIDUAL OWNER					
NEW PLAN VICTORY SQUARE, LLC	1LLC	PO BOX 713530	CINCINNATI	ОН	45271-3530	
New York Department of Taxation &						
Finance	Bankruptcy Section, P.O. Box 5300		Albany	NY	12205-0300	
NEW YORK ELEGANCE	180 MADISON AVE, STE 1502		NEW YORK	NY	10016	
New York State Department of						
Taxation and Finance	TCC/Account Services Section	W A Harriman Campus	Albany	NY	12227	
New York State Income Tax Dept.	PO Box 4131		Binghamton	NY	13902	
NEWELL, GEORGINA	REDACT	REDACT	REDACT	REDACT	REDACT	
Newport Company	Attn.: Jaimie Niemczura	Real Estate Manager	Newport	KY	41072	
Newport Company	25700 Science Park Drive., #370		Beachwood	ОН	44122	
NEWPORT NEWS WATERWORKS	2400 Washington Ave.	9th Floor	Newport News	VA	23607	
NEWPORT NEWS WATERWORKS	PO BOX 979		NEWPORT NEWS	VA	23607-0979	
NEWPORT PROPERTY VENTURES,						
LTD	3211 PONCE DE LEON BLVD	SUITE 202	CORAL GABLES	FL	33134	
NEWTON COUNTY	1113 USHER ST.		COVINGTON	GA	30014	
NEWTON, KEA	REDACT	REDACT	REDACT	REDACT	REDACT	
NEXGEN PACKAGING, LLC	ATTN: SANDI JUREK	1130 WINDHAM PARKWAY	ROMEOVILLE	IL	60446	
NEXT MANAGEMENT, LLC	15 WATTS STREET		NEW YORK	NY	10013	
NG 199 KEATING ROAD, LLC	C/O NIGHTINGALE REALTY, LLC	1430 BROADWAY, SUITE 1605	NEW YORK	NY	10018	
NG 199 Keating Road, LLC	BATESVILLE S/C/Store No302	c/o Nightingale Realty, LLC	New York	NY	10018	

Name	Address 1	Address 2	City	State	PostalCode	Country
NG 199 KEATING ROAD, LLC	C/O NIGHTINGALE REALTY, LLC	1430 Broadway	New York	NY	10018	
NG PALM AIRE MARKETPLACE LLC	C/O NIGHTINGALE PROPERTIES, LLC	1430 BROADWAY	NEW YORK	NY	10018	
NG&G FACILITY SERVICES INT'L	PO BOX 845147		BOSTON	MA	02284-5147	
NGOSIA, NECHESSA	REDACT		REDACT	REDACT	REDACT	
NICHOLAS J. NACLERIO	353 55TH STREET		BROOKLYN	NY	11220	
NICHOLAS, JASMINE	REDACT		REDACT	REDACT	REDACT	
NICHOLAS, NIKITA	REDACT		REDACT	REDACT	REDACT	
NICHOLS, ALLONDA	REDACT		REDACT	REDACT	REDACT	
NICHOLSON, JOSH	REDACT	REDACT	REDACT	REDACT	REDACT	
NICHOLSON, SOPHIA	REDACT		REDACT	REDACT	REDACT	
NICOR	P.O. BOX 5407		CAROL STREAM	IL	60197-5407	
NICOR GAS	1844 Ferry Rd		Naperville	IL	60563-9662	
NICOR GAS	P.O. BOX 5407		CAROL STREAM	IL	60197-5407	
NIMROD, DIMITRI	REDACT		REDACT	REDACT	REDACT	
NIPSCO	801 E. 86th Avenue		Merrillville	IN	46410	
NIPSCO	PO BOX 13007		MERRILLVILLE	IN	46411-3007	
NIVINS, LINETTE	REDACT		REDACT	REDACT	REDACT	
NOBLE MGMT. CO.	4280 PROFESSIONAL CTR DR	SUITE 100	PALM BEACH GARDENS	FL	33410	
NOLES-FRYE REALTY, INC.	P.O. BOX 13148		ALEXANDRIA	LA	71315	
NONN'S FLOORING INC DES						
SHOWPLACE	7550 GRABER ROAD		MIDDLETON	WI	53562	
NORFOLK DEPT OF PUBLIC HEALTH	830 SOUTHAMPTON AVE	ATTN: 7020/21	NORFOLK	VA	23510	
NORRIS, OLEVIA	REDACT	711111. 7020/21	REDACT	REDACT	REDACT	
NORTH ALABAMA GLASS COMPANY,	REDICT		REDITET	KEDACI	KEBACI	
INC.	625 SECOND AVE SE		DECATUR	AL	35601	
NORTH AMERICAN INDUSTRIES	456 JOHNSON AVE		BROOKLYN	NY	11237	
North Carolina Department of	13030111130117112		SHOOKETH.		11237	
Revenue	P.O. Box 25000		Raleigh	NC	27640-0640	
North Carolina Department of						
Revenue	501 N Wilmington St		Raleigh	NC	27604	
North Carolina Dept of Revenue	PO Box 25000		Raleigh	NC	27640	
NORTH CAROLINA DEPT OF			3			
REVENUE	PO BOX 25000		RALEIGH	NC	27640-0645	
NORTH OAKS PLAZA GROUP, INC	23 NORTH OAKS PLAZA	SUITE 216	ST. LOUIS	МО	63121	
NORTHERN KENTUCKY WATER						
DISTRICT	2835 Crescent Springs Road		Erlanger	KY	41018	

Name	Address 1	Address 2	City	State	PostalCode	Country
NORTHERN KENTUCKY WATER						
DISTRICT	PO BOX 188190		ERLANGER	KY	41018-8190	
NORTHERN LIGHTS IMPROVEMENTS,						
LLC	C/O DLC MGMT CORP	PO BOX 5122	WHITE PLAINS	NY	10602	
NORTHERN LIGHTS IMPROVEMENTS,						
LLC	C/O DLC MGMT CORP	P O BOX 944342	Cleveland	ОН	44194	
NORTHERN LIGHTS IMPROVMENTS,						
LLC	580 White Plains Road		Tarrytown	NY	10591	
NORTHERN LIGHTS IMPROVMENTS,						
LLC	C/O DLC MANAGEMENT CORP	PO BOX 944342	CLEVELAND	ОН	44194	
	C/O ALLEN ASSOCIATES					
NORTHGATE SHOPPING CENTER LP	PROPERTIES, INC	P O Box 590249	Newton Centre	MA	02459	
	C/O ALLEN ASSOCIATES					
NORTHGATE SHOPPING CENTER, LLC	PROPERTIES, INC.	P.O. BOX 590249	NEWTON CENTRE	MA	02459	
NORTHLAND CENTER MICHIGAN, LLC	PO BOX 824829		PHILADELPHIA	PA	19182-4829	
NORTHLAND CENTER MICHIGAN, LLC	IN RECEIVERSHIP	PO BOX 7379	BLOOMFIELD HILLS	MI	48302-7379	
		21500 NORTHWESTERN HWY,				
NORTHLAND CENTER MICHIGAN, LLC		#1212	SOUTHFIELD	MI	48075	
	C/O CENTRE CORP MGMT					
NORTHLINE COMMONS LLC	SERVICES, LLLP	3829 W. SPRING CREEK PKWY	PLANO	TX	75023	
NORTHLINE COMMONS, LLC	3829 W. Spring Creek Parkway	Suite 112	Plano	TX	75023	
NORTHPOINT RETAIL PARTNERSHIP	C/O NDG Properties, Inc.	4117 Hillsboro Pike	Nashville	TN	37215	
NORTH POINT RETAIL BARTHERSHIP	6/0 ND 6 DD 6D5D7/55 NJ 6	4447 LWL CDODO DD CTT 007			27245	
NORTHPOINT RETAIL PARTNERSHIP	C/O NDG PROPERTIES INC.	4117 HILLSBORO RD, STE 207	NASHVILLE	TN	37215	
NOTE OF THE BASE O	6/0 NDC DD0D5D7/50	4447.1111.60000.0115			27245	
NORTHPOINT RETAIL PARTNERSHIP	C/O NDG PROPERTIES	4117 HILLSBORO PIKE	NASHVILLE	TN	37215	
NORTHPOINT RETAIL PARTNERSHIP	C/O NDG PROPERTIES	4117 HILLSBORO PIKE, STE 207	NASHVILLE	TN	37215	
NORTHSIDE CENTRE, LLC	696 NE 125TH STREET	TIT THEESDONG FIRE, STE 207	NORTH MIAMI	FL	33161	1
NORTHSIDE CENTRE, LLC	7900 NW 27TH AVENUE	+	MIAMI	FL	33147	
NORTHSIDE CENTRE, LLC	7900 NW 27TH AVENUE, #E-234	1	MIAMI	FL	33147	
NORTHSIDE CENTRE, LLC, as a	7.500 NVV 2/111 AVLINOL, #E-234	1	IVIIAIVII	I L	33147	
successor in interest to 7900 N.W.						
Avenue, LLC	696 NE 125th Street		North Miami	FL	33161	
NORTHWEST JUNCTION LP	Pacific Bay Investments, Inc.	Nancy M. Lane, CCIM	Jackson	MS	39216	-
NOVIUMES! JONCHON TA	racinc bay investments, inc.	INATICY IVI. Latte, CCIIVI	Jackson	IVIS	39210	l

Name	Address 1	Address 2	City	State	PostalCode	Country
NORTHWEST JUNCTION, LP	C/O NANCY M. LANE, CCIM	1855 LAKELAND DRIVE, SUITE E-21	JACKSON	MS	39216-4954	
NORTHWEST JUNCTION, LP	C/O NANCY M. LANE, CCIM	1855 LAKELAND DRIVE SUITE E-21	JACKSON	MS	39216	
NORTHWEST PLAZA DEVELOPERS,						
LLC	6725 W. CENTRAL AVE., UNIT U		TOLEDO	ОН	43617	
NORTHWEST PLAZA DEVELOPERS,						
LLC	6725 W. CENTRAL AVENUE	UNIT U	TOLEDO	ОН	43617	
NORWICH PUBLIC UTILITIES	173 N Main St		Norwich	СТ	06360-0000	
NORWICH PUBLIC UTILITIES	P.O. BOX 1087		NORWICH	СТ	06360-1087	
	c/o Konover Commercial					
NORWICH REALTY ASSOCIATES, LLC	Corporation	342 North Main Street	West Hartford	СТ	06117	
NORWICH REALTY ASSOCIATES, LLC	C/O KONOVER COMMERICAL CORP	342 NORTH MAIN ST.	WEST HARTFORD	СТ	06117	
NORWOOD, DARLENE	REDACT		REDACT	REDACT	REDACT	
NORWOOD, JAMES	REDACT		REDACT	REDACT	REDACT	
NORWOOD, SHERRI	REDACT		REDACT	REDACT	REDACT	
NORWOOD, YOLANDA	REDACT		REDACT	REDACT	REDACT	
NOVAK, CHRISTOPHER	REDACT		REDACT	REDACT	REDACT	
NRAI (NATIONAL REGISTERED						
AGENTS)	PO BOX 12432		NEWARK	NJ	07101-3532	
NRG-HAMPSHIRE HILLS, LLC	5401-A South Blvd		Charlotte	NC	28217	
NRG-HAMPSHIRE, LLC	624 TYVOLA RD	SUITE 103-172	CHARLOTTE	NC	28217	
NRTA	60 SHAKER ROAD EAST		EAST LONGMEADOW	MA	01028	
NUNEMAKER, NICHOLE	REDACT		REDACT	REDACT	REDACT	
NUNEZ, ELVIA	REDACT		REDACT	REDACT	REDACT	
NW KENTWOOD TOWNE CENTER,						
LLC	C/O FREIDMAN MGMT. CO.	34975 W. 12 MILE RD.	FARMINGTON HILLS	MI	48331	
NY JEFFERSON LLC	c/o Islam Yaldo	31000 Northwestern Highway	Farmington Hills	MI	48334	
NY JEFFERSON LLC	C/O ISAM YALDO	31000 NORTHWESTERN HWY	FARMINGTON HILLS	MI	48334	
NYS DEPT OF STATE	ONE COMMERCE PLAZA	99 WASHINGTON AVENUE	ALBANY	NY	12231	
NYS INCOME TAX DEPT	PROCESSING UNIT	PO BOX 4111	BINGHAMTON	NY	13902-4111	
NYSEG	18 Link Drive		Binghamton	NY	13904	
NYSEG	P.O. BOX 847812		BOSTON	MA	02284-7812	
NYSIF DISABILITY BENEFITS	P.O. BOX 5239		NEW YORK	NY	10008-5239	
OAK PARK WATER	14000 OAK PARK BOULEVARD		OAK PARK	MI	48237	
OAK REALTY PARTNERS, INC	5975 S. QUEBEC ST.	SUITE 141	GREENWOOD VILLA	со	80111	
OAKLAND COUNTY TREASURER	1200 N. TELEGRAPH RD		PONTIAC	MI	48341-0479	

Name	Address 1	Address 2	City	State	PostalCode	Country
OAKLAND POINTE PARTNERS LLC	29800 MIDDLEBELT RD	SUITE 200	FARMINGTON HILLS	MI	48334	
OAKLAND POINTE PARTNERS, LLC	31150 Northwestern Highway	Suite 200	Farmington Hills	MI	48334	
OAKWORTH CAPITAL BANK	2100A Southbridge Pkwy	St 445	Birmingham	AL	35209	
OBEID, LISA	REDACT		REDACT	REDACT	REDACT	
OBJEX FASHION	820 4TH AVENUE		BROOKLYN	NY	11232	
OCCASION	8 HERBERT AVENUE		PORT WASHINGTON	NY	11050	
ODOM, BRENDA	REDACT		REDACT	REDACT	REDACT	
	C/O NAI BRANNEN GODDARD LLC,					
ODYSSEY (II) DP XIX, LLC	AS RECEIVER	P O BOX 101739	ATLANTA	GA	30392-1739	
1						
ODYSSEY (II) DP XIX, LLC	C/O NAI BRANNEN GODDARD LLC,	AS RECEIVER	ATLANTA	GA	30392-1739	
OFFICE EFFICIENCIES (INDIA)	127, DAMJI SHAMJI INDUSTRIAL					
PVT.LIMITED	ESTATE,	L.B.S. ROAD, VIKHROLI (W),	MUMBAI		400 083	India
OFFICE OF CIRCUIT CLERK	P.O. BOX 16994		CLAYTON	MO	63105-6994	
OFFICE OF THE CHAPTER 13						
TRUSTEE	KEITH L. RUCINSKI, TRUSTEE	3600 MOMENTUM PLACE	CHICAGO	IL	60689-5336	
OFFICE OF THE CHAPTER 13						
TRUSTEE - AUGUSTA	P.O. BOX 102173		ATLANTA	GA	30368-2173	
OFFICE OF UNEMPLOYMENT OF						
COMPENSATION TAX SERVICES	LABOR & INDUSTRY BUILDING	P.O. BOX 68568	HARRISBURG	PA	17106-8568	
OFFICE, ASHLEY	REDACT		REDACT	REDACT	REDACT	
OHARA, WONNERFAY	REDACT		REDACT	REDACT	REDACT	
OHIO CSPC	PO BOX 182394		COLUMBUS	ОН	43218	
Ohio Department of Taxation	P.O.Box 530		Columbus	ОН	43216-0530	
Ohio Dept of Taxation	PO Box 182215		Columbus	ОН	43218	
OHIO EDISON	76 South Main Street		Akron	Ohio	44308	
OHIO EDISON	P.O. BOX 3687		AKRON	ОН	44309-3687	
OHIO SECRETARY OF STATE	P.O. BOX 670		COLUMBUS	ОН	43216	
OHIO TREASURER OF STATE	BMV	PO BOX 183089	COLUMBUS	ОН	43218-3089	
OLD DOMINION FREIGHT LINE	PO BOX 198475		ATLANTA	GA	30384-8475	
OLEM SHOE CORP.	P.O. BOX 013168		MIAMI	FL	33101-3168	
	c/o Blaxberg, Grayson & Kukoff &					
Olem Shoe Corporation	Twombly, P.A.	Ian J. Kukoff	Miami	FL	33131-1506	
-	·				ĺ	
OLIVER CREEK HOLDINGS, LLLP	C/O MOORE PROPERTY MGMT, LLC	PO BOX 1429	MONTGOMERY	AL	36102	
OLIVER CREEK SPE, LLC	PO BOX 235021		MONTGOMERY	AL	36123-5021	
OLIVER, KENDALE	REDACT		REDACT	REDACT	REDACT	

Name	Address 1	Address 2	City	State	PostalCode	Country
OLIVERAS, MARYBEL	REDACT		REDACT	REDACT	REDACT	
OLIVIA MILLER	1 W 34TH ST		NEW YORK	NY	10001	
OMAHA FIXTURE INTERNATIONAL	P.O. BOX 30097		OMAHA	NE	68103	
ONDUSKO, MATTHEW	REDACT		REDACT	REDACT	REDACT	
ONE STEP UP	A/R DEPT	1412 BROADWAY, 3RD FLOOR	NEW YORK	NY	10018	
ONE TEN WEST FORTIETH	C/O Adams and Company Real					
ASSOCIATES	Estate	411 Fifth Avenue	New York	New York	10016	
ONEAL, LASHONDA	REDACT	REDACT	REDACT	REDACT	REDACT	
ONEPOINT TECHNOLOGIES, INC	380 Trickum Road Northeast		Woodstock	GA	30188	
ONEPOINT TECHNOLOGIES, INC	PO BOX 1849		WOODSTOCK	GA	30188-1369	
Onondaga County	Attn: Consumer Protection Agency	Gordon J. Cuffy - County Attorney	Syracuse	NY	13202	
ONYX DESIGN GROUP	65 MEMORIAL RD, STE 320		WEST HARTFORD	СТ	06107	
OPUS MEDIA PRODUCTIONS LLC	3724 WOODVALE ROAD		BIRMINGHAM	AL	35223	
ORANGE BLOSSOM ASSOCIATE						
STORE NO. 57	181 S. FRANKLING AVENUE	SUITE 302	VALLEY STREAM	NY	11581	
ORANGE BLOSSOM ASSOCIATES	181 S. Franklin Avenue	Suite 302	Valley Stream	NY	11581	
ORANGE BLOSSOM ASSOCIATES	181 S. FRANKLIN AVENUE	SUITE302	VALLEY STREAM	NY	11581	
Orange County	Attn: Consumer Protection Agency	Jeffrey J. Newton - County Attorney	Orlando	FL	32802-1393	
ORANGE COUNTY TAX COLLECTOR	P.O. BOX 545100		ORLANDO	FL	32854-5100	
0.0.0.02.000.00.000.000.000.000.000.000	110130031313	Ms. D'Anne Haydel - County	011211120		5205 : 5200	
Orangeburg County	Attn: Consumer Protection Agency	Attorney	Orangeburg	sc	29115	
- tanget and tane	, and the second second second	,				
ORANGEBURG COUNTY TREASURE	PO BOX 9000		ORANGEBURG	sc	29116	
ORION SYSTEMS	2512 EAST 71ST STREET	SUITE B	TULSA	OK	74136	
ORLANDO UTILITIES COMMISSION	100 West Anderson Street		Orlando	FL	32801	
ORLANDO UTILITIES COMMISSION	P.O. BOX 4901		ORLANDO	FL	32802-4901	
ORLY	350 FIFTH AVE, ROOM 6721		NEW YORK	NY	10118	
ORTIZ, JESENIA	REDACT		REDACT	REDACT	REDACT	+
ORTIZ, PATRICIA	REDACT		REDACT	REDACT	REDACT	+
OSBEY, LATASHIA	REDACT		REDACT	REDACT	REDACT	+
OSTIGUY, CLAYTON	REDACT		REDACT	REDACT	REDACT	+
OTTO, LINDA	REDACT	1	REDACT	REDACT	REDACT	
OTTO, LINDA	ILDACI	<u> </u>	ILEDACI	NEDACI	NEDACI	1

Name	Address 1	Address 2	City	State	PostalCode	Country
			•			,
Ouachita Parish	Taxation and Revenue Department	PO Box 123	Monroe	LA	71210	
OUACHITA PARISH	SHERIFF/TAX COLLECTOR	P.O. BOX 1803	MONROE	LA	71210-1803	
OUC- ORLANDO UTILITIES						
COMMISSION	100 W. Anderson St.		Orlando	FL	32801	
OUC- ORLANDO UTILITIES						
COMMISSION	PO BOX 31329		TAMPA	FL	33631-3329	
OUTER DRIVE & SOUTHFIELD, LLC	C/O RODNEY GAPPY	24435 Halsted Road	Farmington Hills	MI	48335	
OUTER DRIVE & SOUTHFIELD, LLC	C/O RODNEY GAPPY	24435 HALSTED RD	FARMINGTON HILLS	MI	48335	
OUTLAW, KALYA	REDACT		REDACT	REDACT	REDACT	
OVERHEAD DOOR CO. OF ATLANTA	PO BOX 14107		ATLANTA	GA	30324	
OVERHEAD DOOR COMPANY OF						
NASHVILLE	PO BOX 241		MADISON	TN	37116-0241	
OVERHEAD DOOR OF MEMPHIS, INC.	2080 ELVIS PRESLEY BLVD.		MEMPHIS	TN	38106	
OWENS, ANTONIO	REDACT		REDACT	REDACT	REDACT	
OWENS, ERICKA	REDACT		REDACT	REDACT	REDACT	
OWENS, SHARON	REDACT		REDACT	REDACT	REDACT	
OWNEY, KENYADA	REDACT		REDACT	REDACT	REDACT	
PACHECO, DELIA	REDACT		REDACT	REDACT	REDACT	
PACHECO, DELIA	REDACT		REDACT	REDACT	REDACT	
PACIFIC LINK	1223 CORONET DR	SUITE 3	DALTON	GA	30720	
PADRO, ANDREA	REDACT		REDACT	REDACT	REDACT	
PAETEC - WINDSTREAM	PO BOX 1283		BUFFALO	NY	14240-1283	
PAETEC-WINDSTREAM	4001 Rodney Parham Rd.		Little Rock	AR	72212	
PAETEC-WINDSTREAM	P.O. BOX 9001013		LOUISVILLE	KY	40290-1013	
PAINT & BODY EXPERTS	1842 2ND STREET		SLIDELL	LA	70458	
Palm Beach County	Attn: Consumer Protection Agency	Philip Mugavero, County Attorney	West Palm Beach	FL	33401	
PALM BEACH COUNTY	TAX COLLECTOR	PO BOX 3353	WEST PALM BEACH	FL	33402-3353	
PALM BEACH COUNTY	P.O. BOX 3353		WEST PALM BEACH	FL	33402	
PALM BEACH COUNTY SHERIFF'S						
OFFICE	ACCOUNTING DEPT/ALARM UNIT	PO BOX 24681	WEST PALM BEACH	FL	33416-4681	
PAMELA JACKSON	4182 N. 56TH STREET		MILWAUKEE	WI	53216	
PAN OCEANIC	15 W 37TH		NEW YORK	NY	10018	
PANNELL, CHIQUIA	REDACT		REDACT	REDACT	REDACT	
PANOLA COUNTY	TAX COLLECTOR	151 PUBLIC SQUARE, STE C	BATESVILLE	MS	38606	

Name	Address 1	Address 2	City	State	PostalCode	Country
PANTIES PLUS, INC.	320 5th AVENUE	2ND FLOOR	NEW YORK	NY	10001	
PARANET CORPORATION	3675 CRESTWOOD PARKWAY	SUITE 350	DULUTH	GA	30096	
Parish Attorney	Mary E. Roper	222 Saint Louis St.	Baton Rouge	LA	70821	
Parish of Rapides Sales and Use Tax						
Department	PO Box 671		Alexandria	LA	71309	
Parish of Richland	Richland Parish Tax Commission	PO Box 688	Rayville	LA	71269	
Parish of Terrebonne	Sales & Use Dept	PO Box 670	Houma	LA	70361	
PARISI, GABRIELLA	REDACT		REDACT	REDACT	REDACT	
PARK AVENUE SPORTSWEAR LTD.	820 4TH AVE		BROOKLYN	NY	11232	
	C/O PRINCIPLE COMMERCIAL REAL					
PARK CENTER PARTNERSHIP	ESTATE CO.	15455 CONWAY ROAD	CHESTERFIELD	MO	63107	
PARKER, LANAIR	REDACT		REDACT	REDACT	REDACT	
PARKER, SHIETHA	REDACT		REDACT	REDACT	REDACT	
PARKER, TIFFANY	REDACT		REDACT	REDACT	REDACT	
PARKS, ALAIJAH	REDACT		REDACT	REDACT	REDACT	
PARKSIDE SHOPPING CENTER, LP	PO BOX 32591		BALTIMORE	MD	21282-2591	
Parkway Plaza, LLC	PARKWAY PLAZA/Store No137	Attn: Gene Wakefield	Broussard	LA	70518	
PARKWAY PLAZA, LLC	P.O. Box 98		Broussard	LA	70518	
PARRISH III, HENRY	REDACT		REDACT	REDACT	REDACT	
Pasadena City Attorney	1211 Southmore		Pasadena	TX	77502	
PASS, WALTER	REDACT		REDACT	REDACT	REDACT	
	HENLEY, LOTTERHOS, & HENLEY,					
PAT HENLEY	PLLC	P.O. BOX 389	JACKSON	MS	39205-0389	
PATEL, NANCY	REDACT		REDACT	REDACT	REDACT	
PATRICK RAGON ELECTRICAL						
SERVICES	2510 FAIRFAX ROAD		INDIANAPOLIS	IN	46227	
PATRIOT FIRE PROTECTION, INC.	PO BOX 402		LOCUST GROVE	GA	30248	
PATTERSON, BRITTANY	REDACT		REDACT	REDACT	REDACT	
PATTERSON, DONNA	REDACT		REDACT	REDACT	REDACT	
PATTERSON, JO ANN	REDACT	REDACT	REDACT	REDACT	REDACT	
PATTERSON, TISHAWNA	REDACT		REDACT	REDACT	REDACT	
PATTERSON, TYNNETTA	REDACT		REDACT	REDACT	REDACT	
PATTERSON, TYNNETTA	REDACT		REDACT	REDACT	REDACT	
PATTON'S INC.	P.O. BOX 63128		CHARLOTTE	NC	28263-3128	
PAUCIELLO, SHAWNA	REDACT		REDACT	REDACT	REDACT	
PAUCIELLO, SHAWNA	REDACT		REDACT	REDACT	REDACT	

Name	Address 1	Address 2	City	State	PostalCode	Country
PAUL BUTERA	C/O BUTERA PROPERTIES	1590 W. ALGONQUIN #223	HOFFMAN ESTATES	IL	60195	
Paul D. Harrill, Esq.	Counsel for Edisto)	McNair Law Firm, P.A.	Columbia	SC	29211	
PAUL, ALEXANDRA	REDACT		REDACT	REDACT	REDACT	
PAUL, CHRIS	REDACT		REDACT	REDACT	REDACT	
PAULINE BUCKINGHAM	569 PARKHILL DRIVE	APT.19	FAIRLAWN	ОН	44333	
PAVANA	P.O. BOX 237191		NEW YORK	NY	10023	
PAYNE, ALYCE	REDACT	REDACT	REDACT	REDACT	REDACT	
PAYNE, COLLAND	REDACT		REDACT	REDACT	REDACT	
PAYNE, JASMYNE	REDACT		REDACT	REDACT	REDACT	
PAYNE, WILLIE	REDACT		REDACT	REDACT	REDACT	
PAYTON, MINNIE	REDACT		REDACT	REDACT	REDACT	
PC SYSTEMS BY TKE	ATTN: TIMOTHY K. EWING	13735 HILLSIDE AVE	THURMONT	MD	21788	
PEARSON, CARLA	REDACT	REDACT	REDACT	REDACT	REDACT	
PEARSON, TAMEKA	REDACT		REDACT	REDACT	REDACT	
PECO	2301 Market Street		Philadelphia	PA	19101	
PECO	PAYMENT PROCESSING	P.O. Box 37629	PHILADELPHIA	PA	19101-0629	
PECO - PAYMENT PROCESSING	PAYMENT PROCESSING	P.O. Box 37629	PHILADELPHIA	PA	19101-0629	
PEIFER SAFE & LOCK, LLC	5287 KNIGHT ARNOLD ROAD		MEMPHIS	TN	38118	
PENDLETON, ARIEL	REDACT		REDACT	REDACT	REDACT	
PENDLETON, AVA	REDACT		REDACT	REDACT	REDACT	
PENDLEY, ALISON	REDACT		REDACT	REDACT	REDACT	
PENELEC	76 South Main Street		Akron	ОН	44308	
PENELEC	P.O. BOX 3687		AKRON	ОН	44309-3687	
PENNINGTON, CHERYL	REDACT		REDACT	REDACT	REDACT	
Pennsylvania Department of						
Revenue	Office of Chief Counsel	PO Box 281061	Harrisburg	PA	17128	
	Bureau of Business Trust Fund					
Pennsylvania Dept of Revenue	Taxes	PO Box 280901	Harrisburg	PA	17128-0901	
PEOPLES GAS	130 E RANDOLPH DR		CHICAGO	IL	60601	
PEOPLES GAS	P.O. BOX 19100		GREEN BAY	WI	54307-9100	
PEOPLES NATURAL GAS COMPANY	375 North Shore Drive	Suite 600	Pittsburgh	PA	15212	
PEOPLES NATURAL GAS COMPANY	P.O. BOX 644760		PITTSBURGH	PA	15264-4760	
PEOPLE'S UNITED BANK	624 West Main Street		Norwich	СТ	06360	
PEPCO	5100 Harding Highway		Mays Landing	NJ	08330-0000	
PEPCO	PO BOX 13608		PHILADELPHIA	PA	19101-3608	
PERCY, HOLLY	REDACT		REDACT	REDACT	REDACT	

Name	Address 1	Address 2	City	State	PostalCode	Country
PERDUE, POLLY	REDACT		REDACT	REDACT	REDACT	
PERFECT FORMS, INC.	2035 CORTE DEL NOGAL	SUITE 165	CARLSBAD	CA	92011	
PERFECT IMAGE ACCESORIES LLC	10 WEST 33RD STREET	SUITE 1117	NEW YORK	NY	10001	
PERFORMANT RECOVERY, INC	PO BOX 9055		PLEASANTON	CA	94566-9055	
PERKINS METRO GLASS	5295 I-55 NORTH	SUITE C	JACKSON	MS	39206	
PERKINS, ANTONIO	REDACT		REDACT	REDACT	REDACT	
PERKINS, LAKEESHA	REDACT		REDACT	REDACT	REDACT	
PERKO, LLC	150 WEST SLAUSON		LOS ANGELES	CA	90003	
PERRY, BRITTANY	REDACT		REDACT	REDACT	REDACT	
PERRY, DOROTHY	REDACT		REDACT	REDACT	REDACT	
PERRY, GWAN	REDACT		REDACT	REDACT	REDACT	
PERRY, JUANITA	REDACT		REDACT	REDACT	REDACT	
PERRY, LISA	REDACT		REDACT	REDACT	REDACT	
PERRY, RODERICK	REDACT		REDACT	REDACT	REDACT	
PERRY, RODERICK	REDACT		REDACT	REDACT	REDACT	
PERSAUD, SATKHRAM	REDACT		REDACT	REDACT	REDACT	
PETE, KIMBERLY	REDACT	REDACT	REDACT	REDACT	REDACT	
PETERSBURG GENERAL DISTRICT						
COURT	CIVIL DIVISION	35 EAST TABB STREET	PETERSBURG	VA	23803	
PETERSON, ANITA	REDACT		REDACT	REDACT	REDACT	
PETERSON, KEEFE	REDACT		REDACT	REDACT	REDACT	
PETERSON, LATOYA	REDACT		REDACT	REDACT	REDACT	
PETERSON, TYRON	REDACT	REDACT	REDACT	REDACT	REDACT	
PETE'S PRINT	1615 MONTGOMERY HIGHWAY	SUITE 110	HOOVER	AL	35216	
PETTWAY, LAKIA	REDACT		REDACT	REDACT	REDACT	
PETTWAY, SHAWANNA	REDACT		REDACT	REDACT	REDACT	
PETTY CASH / JUDY EDWARDS	22361 TRACE LANE		MCCALLA	AL	35111	
PHAM, HEATHER	REDACT		REDACT	REDACT	REDACT	
PHELPS, JENKINS, GIBSON &						
FOWLER	ATTORNEYS AT LAW	P.O. BOX 020848	TUSCALOOSA	AL	35402-0848	
PHENIX CITY UTILITIES	1119 Broad St		Phenix City	AL	36867	
PHENIX CITY UTILITIES	PO BOX 760		PHENIX CITY	AL	36868-0760	
PHILADELPHIA GAS WORKS	800 W. Montgomery Avenue		Philadelphia	PA	19122	
PHILADELPHIA GAS WORKS	P.O. BOX 11700		NEWARK	NJ	07101-4700	
PHILADELPHIA REVENUE						
DEPARTMENT	1401 JOHN F. KENNEDY BLVD.	MUNICIPAL SERVICES BUILDING	PHILADELPHIA	PA	19102-1697	
Phillip Edison & Co.	C/O Holland & Knight LLP	Attn: Joaquin J. Alemany, Esq.	Miami	FL	33131	
Phillip Edison & Co.	C/O Holland & Knight LLP	Attn: Eric B. Funt, Esq.	Miami	FL	33131	

Name	Address 1	Address 2	City	State	PostalCode	Country
PHILLIPS, ASHLEY	REDACT		REDACT	REDACT	REDACT	
PHILLIPS, JAILA	REDACT		REDACT	REDACT	REDACT	
PHILLIPS, KRISTEN	REDACT		REDACT	REDACT	REDACT	
PHILLIPS, MALIK	REDACT	REDACT	REDACT	REDACT	REDACT	
PICKED FRESH LLC	1385 BROADWAY, 16TH FL		NEW YORK	NY	10018	
PIEDMONT FIRE PROTECTION, INC	8130 LARKHAVEN ROAD		CHARLOTTE	NC	28216	
PIEDMONT NATIONAL CORP	1570 JEAN STREET		MONTGOMERY	AL	36107	
PIEDMONT NATURAL GAS	4720 Pidmont Row Drive		Charlotte	NC	28210	
PIEDMONT NATURAL GAS	PO BOX 660920		DALLAS	TX	75266-0920	
PIEDMONT SERVICE GROUP	533 BYRON STREET	SUITE 1	CHESAPEAKE	VA	23320	
PIERCE, RHONDA	REDACT		REDACT	REDACT	REDACT	
PIETRALCZYK, KRIS	REDACT		REDACT	REDACT	REDACT	
PINCKNEY, PENNY	REDACT		REDACT	REDACT	REDACT	
PINDER, BERNARD	REDACT	REDACT	REDACT	REDACT	REDACT	
PINDER, BERNARD	REDACT		REDACT	REDACT	REDACT	
PINDER, LOU ANN	REDACT	REDACT	REDACT	REDACT	REDACT	
	PINE HILLS MARKETPLACE/Store					
Pine Hill. CRP LLC	No94	4801 PGA Blvd.	Palm Beach Gardens	FL	33418	
PINE HILLS CRP LLC	Keith L. Cummings, VP	4801 PGA Boulevard	Palm Beach Gardens	FL	33418	
	C/O BERKADIA COMMERCIAL MTG,					
PINE HILLS CRP, LLC	LLC.	P.O. BOX 535704	ATLANTA	GA	30353-5704	
PINER'S LOCK & SAFE	3861 HENDERICKS AVENUE		JACKSONVILLE	FL	32207	
PINET, HELLEN	REDACT		REDACT	REDACT	REDACT	
PINK APPLE INC	95 SEAVIEW BLVD, UNIT 104		PORT WASHINGTON	NY	11050	
PIONEER CREDIT RECOVERY, INC	ATTN: WAGE GARNISHMENT DEPT	PO BOX 158	ARCADE	NY	14009	
PIPP MOBILE STORAGE SYSTEMS,						
INC.	P.O. BOX 674812		DETROIT	MI	48267-4812	
PITNEY BOWES - PURCHASE POWER	PO BOX 371874		PITTSBURGH	PA	15250-7874	
PITNEY BOWES GLOBAL FINANCIAL	PO BOX 371887		PITTSBURGH	PA	15250-7887	
PITTMAN, JESSICA	REDACT		REDACT	REDACT	REDACT	
PIXELGLUE	P.O. BOX 841034		HOLLYWOOD	FL	33084	
PK INTERNATIONAL, LLC	6547 DEVONHURST DR		ST. LOUIS	MO	63139	
	3333 New Hyde Park Road, P.O. Box					
PL CHERRYDALE POINT, LLC	5020	Suite 100	New Hyde Park	NY	11042-0020	<u> </u>

Name	Address 1	Address 2	City	State	PostalCode	Country
PL CHERRYDALE POINT, LLC.	PO BOX 5020		NEW HYDE PARK	NY	11042-0020	
PLAINS CAPITAL BANK	PO Box 271		Lubbock	TX	79408	
PLANK PLAZA INVESTORS, LLC	ATTN: HOWARD ARNBERG	1825 MAIN ST	WESTON	FL	33326	
Plaquemines Parish Sales Tax						
Division	Suite 301A 8056 Highway 23		Belle Chasse	LA	70037	
PLATINUM PLUS FOR BUSINESS	P.O. BOX 15469		WILMINGTON	DE	19886-5469	
PLATINUM PLUS FOR BUSINESS	PO BOX 15469		WILMINGTON	DE	19886-5469	
PLAYA DEL REY SUNGLASSES	2890 AMWILER RD.		ATLANTA	GA	30360	
Pleune Service Co.	750 Himes St SE		Grand Rapids,	MI	49548	
PLEUNE SERVICE COMPANY, INC	750 HIMES STREET SE		GRAND RAPIDS	MI	49548	
PLOTT, LANEY	REDACT		REDACT	REDACT	REDACT	
PLS LOGISTICS SERVICES	5119 RELIABLE PARKWAY		CHICAGO	IL	60686-0051	
PNC	PO Box 609		Pittsburgh	PA	15230-9738	
PNC BANK	One Financial Parkway	Locator Z1-Yb42-0201	Kalamazoo	MI	49009-9738	
PNC BANK	PO Box 609		Pittsburgh	PA	15230-9738	
POETRY	2111 S. LONG BEACH AVE.		LOS ANGELES	CA	90058	
POLK, ALEXIS	REDACT		REDACT	REDACT	REDACT	
	BLUE PARKWAY TOWN					
Polsinelli Shalton & Welte, P.C.	CENTER/Store No161	Attention: Mark Bryant, Esquire	Kansas City	МО	641120	
POOF	1466 BROADWAY		NEW YORK	NY	10036	
POOLE, JONATHAN	REDACT		REDACT	REDACT	REDACT	
POPE, JASMOND	REDACT		REDACT	REDACT	REDACT	
POPLAR PLAZA, LLC	3332 NE 190 Street	#1010	Aventura	FL	33180	
POPLAR PLAZA, LLC	ATTN: URI SEGEV	3332 NE 190 ST., #1010	AVENTURA	FL	33180	
POPOVICH, GINA	REDACT		REDACT	REDACT	REDACT	
PORTAGE CENTER LLC STORE NO.						
286	207 WEST 25TH STREET	SUITE 602	NEW YORK	NY	10001	
PORTAGE CENTER, LLC	207 West 25th Street	Suite 602	New York	NY	10001	
PORTER, ANTONIO	REDACT		REDACT	REDACT	REDACT	
PORTER, BENYETTA	REDACT		REDACT	REDACT	REDACT	
PORTER, FREDERICK	REDACT		REDACT	REDACT	REDACT	
PORTER, LISA	REDACT		REDACT	REDACT	REDACT	
PORTER, MICHAEL	REDACT		REDACT	REDACT	REDACT	
PORTER, SHERI	REDACT		REDACT	REDACT	REDACT	
PORTER-LOVE, CIARA	REDACT		REDACT	REDACT	REDACT	
PORTIS, CHRISTI	REDACT		REDACT	REDACT	REDACT	
PORTSMOUTH ASSOCIATES, LLC	c/o Bowen Construction Company	1775 Woodstock Road	Roswell	GA	30075	

Name	Address 1	Address 2	City	State	PostalCode	Country
PORTSMOUTH CIRCUIT COURT	P.O. BOX 1217		PORTSMOUTH	VA	23705-1217	
POSTAL BOX DELIVERY SERVICE	PO BOX 22		BIRMINGHAM	AL	35201	
POULLARD, WILLIE	REDACT		REDACT	REDACT	REDACT	
POUNCEY, TOUWANNA	REDACT		REDACT	REDACT	REDACT	
POWELL, GWENDOLYN	REDACT		REDACT	REDACT	REDACT	
POWER ELECTRIC SERVICES,						
INCORPORATED	4714 E. HILLSBOROUGH AVENUE		TAMPA	FL	33610	
PPL ELECTRIC UTILITIES	2 NORTH 9TH STREET CPC-GENN1		ALLENTOWN	PA	18101-1175	
PPL ELECTRIC UTILITIES	827 Hausman Road		Allentown	PA	18104-9392	
PRASS, TILLETHA	REDACT		REDACT	REDACT	REDACT	
PREFERRED - HALSTED, LLC	C/O ZIFKIN REALTY MGMT, LLC	600 W. JACKSON BLVD.	CHICAGO	IL	60661	
PREFERRED FRAGRANCE	PO BOX 638566	OGG VV. SACKSON BEVB.	CINCINNATI	ОН	45263-8566	
PRICE, CRESHONNA	REDACT	REDACT	REDACT	REDACT	REDACT	
PRICE, KEION	REDACT	REDACI	REDACT	REDACT	REDACT	
PRICE, SHELIA	REDACT		REDACT	REDACT	REDACT	
Primary	2110 NW 95th Ave.		Miami	FL	33172	
PRIMESTOR 119, LLC	c/o UCR Asset Services	8080 Park Lane	Dallas	TX	75231	
PRIMESTOR 119, LLC	C/O UCR ASSET SERVICES	MARSHFIELD PLZ	DALLAS	TX	75373-1517	
PRIMESTOR 119, LLC	PO BOX 731517		DALLAS	TX	75373-1517	
Prince George's County	Attn: M. Andree Green	County Administration Building	Upper Marlboro	MD	20772-305	
PRINCE GEORGE'S COUNTY	PO BOX 17578	,	BALTIMORE	MD	21297-1578	
PRINCE, TEQUILA	REDACT	REDACT	REDACT	REDACT	REDACT	
PRINTERS, BIANCA	REDACT		REDACT	REDACT	REDACT	
PROFESSIONAL HEAT & AIR						
CONDITIONING	1321 DAHL ROAD		LOUISVILLE	KY	40213	
Professional Heating & Cooling	1321 Dahl Road		Louisville	KY	40213	
PROFESSIONAL PLUMBING	107 BALDWIN DRIVE		ALBANY	GA	31707	
PROFILE ENERGY SERVICES, LLC	P.O. BOX 635356		CINCINNATI	ОН	45263-5356	
PROFILE INDUSTRIES	1407 BW		NEW YORK	NY	10016	
PROJECT 28 CLOTHING, LLC	525 7TH AVE,STE 1201		NEW YORK	NY	10018	
PROPHET, BILLY	REDACT		REDACT	REDACT	REDACT	
PROPHET, MARGARET	REDACT		REDACT	REDACT	REDACT	
PROPHETE, RANDY	REDACT		REDACT	REDACT	REDACT	
PROTECTIVE LIFE	INSURANCE COMPANY	PO BOX 2224	BIRMINGHAM	AL	35246-0030	
PRUETT ELECTRICAL SERVICE, INC	PO BOX 2574		WEST MONROE	LA	71294	
PSE&G CO	80 Park Plaza		Newark	NJ	07101-0000	
PSE&G CO	P.O. BOX 14444		NEW BRUNSWICK	NJ	08906-4444	

Name	Address 1	Address 2	City	State	PostalCode	Country
PSI - PAPER SYSTEMS						
INCORPORATED	LBX#774403	4403 SOLUTIONS CENTER	CHICAGO	IL	60677-4004	
PT HALO	2731 S ALAMEDA ST		LOS ANGELES	CA	90058	
PUBLIC ASSISTANCE COLLECTION						
UNIT	PO BOX 8938		MADISON	WI	53708-8938	
PUBLIC WORKS COMMISSION	955 Old Wilmington Rd		Fayetteville	NC	28301	
PUBLIC WORKS COMMISSION	PO BOX 7000		FAYETTEVILLE	NC	28302-7000	
PUGH, FLORENCE	REDACT	REDACT	REDACT	REDACT	REDACT	
PUGH, TAMIA	REDACT		REDACT	REDACT	REDACT	
PULASKI COUNTY, AR	TREASURER	PO BOX 8101	LITTLE ROCK	AR	72203-8101	
PURE MECHANICAL, LLC	535 S. 93RD STREET		MILWAUKEE	WI	53214	
PUSHCHAK, VALERIE	REDACT		REDACT	REDACT	REDACT	
PUZZLES ENTERPRISES, INC.	3022 S GRAND AVENUE		LOS ANGELES	CA	90007	
PYE BARKER FIRE & SAFETY, INC	PO BOX 69		ROSWELL	GA	30077-0069	
PZ MIRACLE LIMITED PARTNERSHIP	Zamias Services, Inc.	300 Market Street	Johnstown	PA	15901	
PZ MIRACLE LP	C/O ZAMIAS SERVICES, INC.	300 MARKET STREET	JOHNSTOWN	PA	15901	
Q 4 DESIGN/SAMSUNG	20 WEST 33RD ST		NEW YORK	NY	10018	
QC LENDERS- HARVEY # 2	PO BOX 250		GILBERTS	IL	60136	
QUALITY, INC	6200 HUNTLEY ROAD	SUITE D	COLUMBUS	ОН	43229	
QUENOMOEN, ANTHONY	REDACT		REDACT	REDACT	REDACT	
QUENOMOEN, JILL	REDACT		REDACT	REDACT	REDACT	
QUEZADA, YARELI	REDACT		REDACT	REDACT	REDACT	
QUINN, TRACY	REDACT	REDACT	REDACT	REDACT	REDACT	
QUINN, TRACY	REDACT		REDACT	REDACT	REDACT	
QUINTANA, XIOMARA	REDACT		REDACT	REDACT	REDACT	
R & W RIVERVIEW LTD.	NW 5579	PO BOX 1450	MINNEAPOLIS	MN	55485-5579	
R&W RIVERVIEW, LIMITED	NW 5579	P O BOX 1450	Minneapolis	MN	55485-5579	
	Rod Noles					
R.A. MONSUR, Agent	c/o Noles Frye Prop Mgmt.	P. O. Box 13148	Alexandria	LA	71315	
R.G.M. INC	20232 LIVERNOIS	SUITE 1	DETROIT	MI	48221	
R.H.K. JOINT VENTURE	6267 Sturbridge Court		Sarasota	FL	34238	
RACEWAY LAND DEVELOPMENT LLC	P O BOX 0077		Chicago	IL	60690-0077	
RACEWAY LAND DEVELOPMENT LLC	PO BOX 0077		CHICAGO	IL	60690-0077	
RAINBOW ACCESSORIES	313 w 37th st 5th floor		NEW YORK	NY	10018	
RAINES, BROOKLYN	REDACT		REDACT	REDACT	REDACT	

Name	Address 1	Address 2	City	State	PostalCode	Country
RAINEY, BOBBI	REDACT		REDACT	REDACT	REDACT	
RAINEY, VICKI	REDACT		REDACT	REDACT	REDACT	
RALEIGH RETAIL NO. 2, LLC	C/O WATERS, INC.	301 S. McDowell Street	Charlotte	NC	28204	
RALEIGH RETAIL NO. 2, LLC	C/O WATERS INC.	301 S. MCDOWELL ST. STE 210	CHARLOTTE	NC	28204	
RAMCO- GERSHENSON PROPERTIES						
LP	FAIRLANE MEADOWS	P.O. BOX 350018	BOSTON	MA	02241-0518	
RAMCO-GERSHENSON PROPERTIES						
LP	WEST ALLIS TOWNE CENTRE	PO BOX 350018	BOSTON	MA	02241-0518	
RAMCO-GERSHENSON PROPERTIES,						
L.P.	31500 Northwestern Highway	Suite 300	Farmington Hills	MI	48334	
RAMCO-GERSHENSON PROPERTIES,						
LP	31500 NORTHWESTERN HYW	STE 300	FARMINGTON HILLS	MI	48334	
RAMCO-GERSHENSON PROPERTIES,						
LP	CLINTON POINT MASTER OCCUP	ID 00009153	BOSTON	MA	02241-0518	
RAMIREZ, CHLOE	REDACT		REDACT	REDACT	REDACT	
RAMIREZ, MARIA	REDACT		REDACT	REDACT	REDACT	
RAMOS, LOURDES	REDACT	REDACT	REDACT	REDACT	REDACT	
RAMOS, LOURDES	REDACT		REDACT	REDACT	REDACT	
RAMOS, TANGANIKA	REDACT		REDACT	REDACT	REDACT	
RAMSEY, AVIA	REDACT		REDACT	REDACT	REDACT	
RAMSEY, STACEY	REDACT		REDACT	REDACT	REDACT	
RANCHO NORTHWEST CROSSINGS						
LLC	C/O THE WOODMONT COMPANY	2100 W. 7TH ST.	FT WORTH	TX	76107	
RANCHO NORTHWEST CROSSINGS,						
LLC	c/o The Montwood Company	2100 W. 7th Street	Fort Worth	TX	76107	
RAND PROPERTIES, LLC	28860 Southfiled Road	Suite 262	Southfield	MI	48076	
RAND PROPERTIES, LLC	24255 W 13 MILE ROAD	SUITE 220	BINGHAM FARMS	MI	48025	
RANDALL, SHAWNDRIA	REDACT		REDACT	REDACT	REDACT	
RANDLE, DESIREE	REDACT		REDACT	REDACT	REDACT	
RANDSTAD WORK SOLUTIONS	PO BOX 105046		ATLANTA	GA	30348-5046	
RAPIDES PARISH	P.O. BOX 1590		ALEXANDRIA	LA	71309	
RAPIDES PARISH	PO BOX 671		ALEXANDRIA	LA	71309-0671	
RASCOE, LISA	REDACT		REDACT	REDACT	REDACT	
RATCLIFF LAW OFFICE	PO BOX 1453		MADISON	MS	39130	
RATTERREE HEAT AND AIR	7715 HARWELL		FORT WORTH	TX	76108	
RAUDALES, ESTHER	REDACT		REDACT	REDACT	REDACT	
RAWLS, SHAWN	REDACT	REDACT	REDACT	REDACT	REDACT	
RAY, LUVASIA	REDACT		REDACT	REDACT	REDACT	

Name	Address 1	Address 2	City	State	PostalCode	Country
RAY, TIMARIUS	REDACT		REDACT	REDACT	REDACT	
RCG-UNION CITY, LLC	C/O RCG VENTURES I, LLC	PO BOX 53483	ATLANTA	GA	30355	
REAL ESTATE ASSET PURCHASE CO.	C/O EQUITY INVESTMENT GROUP	111 E. WAYNE STREET, STE 500	FORT WAYNE	IN	46802	
REAL FLOORS, INC	5237 HALLS MILL ROAD	BUILDING G	MOBILE	AL	36619	
REAL PEOPLE MODELS AND TALENT	KIDDIN' AROUND OF ALABAMA INC		BIRMINGHAM	AL	35205	
REALNET AMERICAN WAY	C/O MAKOWSKY RINGEL					
ACQUISTIONS LLC	GREENGERG LLC	PO BOX 241990	MEMPHIS	TN	38124-1990	
REAVES, GABRIELLE	REDACT		REDACT	REDACT	REDACT	
REBECCA CALLADONATO	315 ROSEMARY AVE		AMBLER	PA	19002	
RED CIRCLE FOOTWEAR	17900 AJAX CIR		CITY OF INDUSTRY	CA	91748	
RED RIVER BANK	PO Box 12550		Alexander	LA	71315-2550	
REDD, SYDNEY	REDACT		REDACT	REDACT	REDACT	
REDFORD TOWNSHIP- WATER DEPT REED, CARL	12200 BEECH DALY ROAD REDACT		REDFORD REDACT	MI REDACT	48239 REDACT	
REESE, DOBREANNA	REDACT		REDACT	REDACT	REDACT	
REGAL INVESTMENTS	P.O. BOX 45261		LITTLE ROCK	AR	72214	
REGENCY LIGHTING	REGENCY ENTERPRISES INC.	P.O. BOX 205325	DALLAS	TX	75320-5325	
REGINA DAVIS	300 HOLLY HILLS ROAD	APT.226	COLUMBUS	MS	39705	
REGISTER OF DEEDS	WAKE COUNTY	P.O. BOX 1897	RALEIGH	NC	27602	
REGISTER'S HEATING & COOLING	NOEL REGISTER	180 MT ZION CHURCH ROAD	MAUK	GA	31058	
REID, BRIANNA	REDACT		REDACT	REDACT	REDACT	
REID, PHYLLIS	REDACT	REDACT	REDACT	REDACT	REDACT	
REID, PHYLLIS	REDACT		REDACT	REDACT	REDACT	
REID, SHEREE	REDACT		REDACT	REDACT	REDACT	
REILEY, JUDITH	REDACT		REDACT	REDACT	REDACT	
RELIABLE AUTO FINANCE	P.O. BOX 9700		GRAND RAPIDS	MI	49509	
RELIABLE GLASS & MAINTENANCE	20232 LIVERNOIS AVENUE		DETROIT	МІ	48221	
RELIANT	1000 Main St		Houston	TX	77002	
RELIANT	PO BOX 650475		DALLAS	TX	75265-0475	
RENDLEMAN, RUGAYA	REDACT		REDACT	REDACT	REDACT	
RENOVATION REALTY INVESTMENT	C/O RICHTER REALTY & INVSTMT					
SERVICES, INC.	INC.	18650 W CORPORATE DR, STE 103	BROOKFIELD	WI	53045	
REPUBLIC FINANCE	103-1 WOODLAND ROAD		BATESVILLE	MS	38606	

Name	Address 1	Address 2	City	State	PostalCode	Country
RETAIL MAINTENANCE						
MANAGEMENT, INC.	3 EAST DEER PARK ROAD		DIX HILLS	NY	11746	
RETRANS	DEPT 1886	PO BOX 2153	BIRMINGHAM	AL	35287-1886	
Revenue Discovery Systems	PO Box 830725		Birmingham	AL	35283-0725	
REVENUE DISCOVERY SYSTEMS	SALES TAX DIVISION	P.O. BOX 830725	BIRMINGHAM	AL	35283-0725	
REVENUE DISCOVERY SYSTEMS	PO BOX 830725		BIRMINGHAM	AL	35283	
REVERE CONTROL SYSTEMS, INC	2240 ROCKY RIDGE ROAD		BIRMINGHAM	AL	35216	
REYES, FARIS	REDACT		REDACT	REDACT	REDACT	
REYES, JULIAN	REDACT		REDACT	REDACT	REDACT	
REYNOLDS, ULANDRA	REDACT	REDACT	REDACT	REDACT	REDACT	
RHK JOINT VENTURE	C/O CHARLES P. HOOVER	6267 STURBRIDGE COURT	SARASOTA	FL	34238	
Rhode Island Tax Assessor	State of Rhode Island	Division of Taxation	Providence	RI	02908	
RI - DIVISION OF TAXATION	EMPLOYER TAX SECTION	ONE CAPITAL HILL, STE 36	PROVIDENCE	RI	02908-5829	
RICE, SHEQUIA	REDACT		REDACT	REDACT	REDACT	
RICH COLOR	600 E. WASHINGTON BLVD. #203		LOS ANGELES	CA	90015	
RICH, HEATHER	REDACT		REDACT	REDACT	REDACT	
RICHARD C. DANIELSEN	DANIELSEN PLUMBING	P.O. BOX 90445	COLUMBIA	SC	29290-1445	
RICHARDSON, ADRIENNE	REDACT		REDACT	REDACT	REDACT	
RICHARDSON, ASHLEY	REDACT	REDACT	REDACT	REDACT	REDACT	
RICHARDSON, JAZMIN	REDACT		REDACT	REDACT	REDACT	
RICHEY, YAHNAE	REDACT	REDACT	REDACT	REDACT	REDACT	
RICHIE WILLIAMSON PHOTOGRAPHY	130 WEST 130TH STREET		NEW YORK	NY	10027-2002	
RICHLAND AIR CONDITIONING						
SERVICE	740 US HWY 49 SOUTH		RICHLAND	MS	39218	
RICHLAND COUNTY	TREASURER	PO BOX 8028	COLUMBIA	SC	29202-8028	
Richmond County	Attn: Consumer Protection Agency	Richmond County Attorney	Augusta	GA	30911	
RICHMOND COUNTY TAX						
COMMISSIONER	530 GREENE ST	ROOM 117, MUNICIPAL BLDG	AUGUSTA	GA	30911	
RICHMOND, KENDRICK	REDACT	REDACT	REDACT	REDACT	REDACT	
RICHMOND, WAYNE	REDACT		REDACT	REDACT	REDACT	
RICHMONDHAVEN, MAYIM	REDACT		REDACT	REDACT	REDACT	
RICHMOND'HAVEN, MAYIM	REDACT		REDACT	REDACT	REDACT	
RICKS, TYLESTA	REDACT		REDACT	REDACT	REDACT	
RIDGEWAY, LISA	REDACT	REDACT	REDACT	REDACT	REDACT	
RILEY, MAYA	REDACT		REDACT	REDACT	REDACT	
RIPLAY	18856 E SAN JOSE AVE		CITY OF INDUSTRY	CA	91748	

Page 135 of 176

Name	Address 1	Address 2	City	State	PostalCode	Country
RITE AID CORPORATION - 84385	P.O. BOX 360321		PITTSBURG	PA	15250-6321	
RITEWAY SERVICE COMPANY	9010 BROADWAY		SAN ANTONIO	TX	78217-6117	
RIVERA, ANDREINA	REDACT		REDACT	REDACT	REDACT	
RIVERA, LIZA	REDACT		REDACT	REDACT	REDACT	
RIVERA, THERESE	REDACT		REDACT	REDACT	REDACT	
RIVERA, THERESE	REDACT	REDACT	REDACT	REDACT	REDACT	
RIVERS, BETH	REDACT		REDACT	REDACT	REDACT	
RIVERS, BRANDON	REDACT		REDACT	REDACT	REDACT	
RIVERS, NAKESHA	REDACT	REDACT	REDACT	REDACT	REDACT	
RIVIERA APPAREL GROUP	1407 BROADWAY, STE 2010		NEW YORK	NY	10018	
RL BB-GA, LLC	C/O FICKLING MGMT SERVICES, LLC	PO BOX 310	MACON	GA	31202-0310	
RMS PROPERTIES IX, LLC	1491 WEST SCHAUMBURG RD.		SCHAUMBURG	IL	60194	
ROACH, TERESA	REDACT		REDACT	REDACT	REDACT	
ROAN, JOSHUA	REDACT		REDACT	REDACT	REDACT	
ROBB, MEGAN	REDACT		REDACT	REDACT	REDACT	
ROBERSON, TYRONE	REDACT		REDACT	REDACT	REDACT	
ROBERTS, BRENDA	REDACT		REDACT	REDACT	REDACT	
ROBERTS, DONQUAVIA	REDACT		REDACT	REDACT	REDACT	
ROBERTS, TRAVONNE	REDACT		REDACT	REDACT	REDACT	
ROBERTS, TREVOR	REDACT		REDACT	REDACT	REDACT	
ROBERTS, TYESHA	REDACT		REDACT	REDACT	REDACT	
ROBINSON, CASSANDRA	REDACT		REDACT	REDACT	REDACT	
ROBINSON, DOMINIQUE	REDACT		REDACT	REDACT	REDACT	
ROBINSON, GLYNDA	REDACT		REDACT	REDACT	REDACT	
ROBINSON, JAZZALYN	REDACT		REDACT	REDACT	REDACT	
ROBINSON, JENNIFER	REDACT		REDACT	REDACT	REDACT	
ROBINSON, KIIMBERLY	REDACT		REDACT	REDACT	REDACT	
ROBINSON, LASAUNDRA	REDACT		REDACT	REDACT	REDACT	
ROBINSON, MARQUIS	REDACT		REDACT	REDACT	REDACT	
ROBINSON, MARTINIQUE	REDACT		REDACT	REDACT	REDACT	
ROBINSON, SABRINA	REDACT		REDACT	REDACT	REDACT	
ROBINSON, SHIKEIDA	REDACT		REDACT	REDACT	REDACT	
ROBINSON, TAMIKA	REDACT	REDACT	REDACT	REDACT	REDACT	
ROBINSON, TRESHA	REDACT	REDACT	REDACT	REDACT	REDACT	
ROBYN KURZWEIL	490 S. KEYSER AVENUE		TAYLOR	PA	18517	
ROCHELLE LISTON	101 CHESTNUT ST	CPO 1003	BEREA	KY	40404	

Name	Address 1	Address 2	City	State	PostalCode	Country
ROCHELLE, CIERRA	REDACT		REDACT	REDACT	REDACT	
RODDY, MEGAN	REDACT		REDACT	REDACT	REDACT	
RODGERS, QUANITA	REDACT		REDACT	REDACT	REDACT	
Rodney Barstein	P.O. Box 13096		Birmingham	AL	35202	
RODRIGUEZ, IAN	REDACT		REDACT	REDACT	REDACT	
RODRIGUEZ, LEDDIERE	REDACT		REDACT	REDACT	REDACT	
RODRIGUEZ, STEPHANIE	REDACT		REDACT	REDACT	REDACT	
RODRIGUEZ, TORREY	REDACT	REDACT	REDACT	REDACT	REDACT	
RODRIGUEZ, VICTORIA	REDACT		REDACT	REDACT	REDACT	
ROEBUCK MARKETPLACE						
ASSOCIATES, LLC	324 DATURA STREET	SUITE 102	WEST PALM BEACH	FL	33401	
	ROEBUCK MARKETPLACE/Store	c/o American Commercial Realty				
Roebuck Newco, LLC	No124	Corp	West Palm Beach	FL	33401	
ROEBUCK NEWCO, LLC	324 Datura Street	Suite 102	West Palm Beach	FL	33401	
ROEBUCK NEWCO, LLC	PO BOX 534723		ATLANTA	GA	30353-4723	
ROGERS, DEBRA	REDACT		REDACT	REDACT	REDACT	
ROGERS, RAY SHAWN	REDACT		REDACT	REDACT	REDACT	
ROLLER, LANIKA	REDACT		REDACT	REDACT	REDACT	
ROLLING, DIAMOND	REDACT		REDACT	REDACT	REDACT	
ROLLING, DIAMOND	REDACT		REDACT	REDACT	REDACT	
ROLLINS, PORSCHE	REDACT		REDACT	REDACT	REDACT	
ROMAN FASHION	1524 EAST 15TH STREET		LOS ANGELES	CA	90021	
ROMAN, JESSICA	REDACT		REDACT	REDACT	REDACT	
ROMERO, JAYDELIN	REDACT	REDACT	REDACT	REDACT	REDACT	
ROMINE, CORIENA	REDACT		REDACT	REDACT	REDACT	
RONNIE GOLDSTEIN	390 5TH AVENUE	ROOM 810	NEW YORK	NY	10018	
ROOSTH FARM RENTAL, LLC	P.O. BOX 8300		Tyler	TX	75711-8300	
ROOSTH FARM RENTAL, LLC	PO BOX 8300		TYLER	TX	75711-8300	
ROOSTH RENTALS	PO BOX 8300		TYLER	TX	75711-8300	
ROPER, CHRISTY	REDACT		REDACT	REDACT	REDACT	
ROQUEMORE'S KEY & SAFE	1004 SOUTH 8TH ST.		LANETT	AL	36863	
ROSALES, EDNA	REDACT		REDACT	REDACT	REDACT	
ROSALES, WHITNEY	REDACT		REDACT	REDACT	REDACT	
ROSARI, RUTH	REDACT	REDACT	REDACT	REDACT	REDACT	
ROSE GEORGE	640 SULLIVAN LANE		UNIVERSITY	IL	60484	
ROSE, CHRIS	REDACT	REDACT	REDACT	REDACT	REDACT	
ROSEE FASHION	1407 BROADWAY, STE 2021		NEW YORK	NY	10018	
ROSEMOND, IDAISHA	REDACT		REDACT	REDACT	REDACT	

Name	Address 1	Address 2	City	State	PostalCode	Country
Rosenthal & Rosenthal Inc.	P.O. Box 88926		Chicago	IL	60695-1926	
ROSEY TOMATO	555 SEACAUCUS RD		SEACAUCUS	NJ	07094	
ROSOKOW, GARDENIA	REDACT		REDACT	REDACT	REDACT	
ROSS, ALEISHA	REDACT		REDACT	REDACT	REDACT	
ROSS, DONNA	REDACT		REDACT	REDACT	REDACT	
ROSY LOVES	110 9TH ST	STE#B849	LOS ANGELES	CA	90079	
ROTO-ROOTER PLUMBERS-BATON						
ROUGE	P.O. BOX 64966		BATON ROUGE	LA	70896	
ROTO-ROOTER SERVICES (CHICAGO)	5672 COLLECTION CENTER DR.		CHICAGO	IL	60693	
ROTO-ROOTER SERVICES CO (IL)	5672 COLLECTION CENTER DRIVE		CHICAGO	IL	60693	
ROWDEN, DORIS	REDACT		REDACT	REDACT	REDACT	
ROY A GEORGE	640 SULLIVAN LN		UNIVERSITY PARK	IL	60466	
ROY, PICEAN	REDACT		REDACT	REDACT	REDACT	
ROYAL OAK TOWNSHIP	21131 GARDEN LANE	SUITE 205	FERNDALE	MI	48220	
ROYAL TIME	32-A W 30TH STREET		NEW YORK	NY	10001	
ROYAL TOWN CENTER, LLC	C/O FIRST COMMERICAL	27600 NORTHWESTERN HWY,	SOUTHFIELD	MI	48034	
ROYAL TOWN CENTER, LLC	29500 Telegraph Road	Sutie 110	Southfield	MI	48034	
ROYAL TOWN CENTER, LLC	27600 NORTHWESTERN HIGHWAY		SOUTHFIELD	MI	48034	
	27600 NORTHWESTERN HIGHWAY,					
ROYAL TOWN CENTER, LLC	SUITE 200		SOUTHFIELD	MI	48034	
RPAI US MGMT, LLC	LOCK BOX NW 6141	(LEASE#19364)	MINNEAPOLIS	MN	55485-6141	
RPAI US MGMT., LLC	NW 6137	PO BOX 1450	MINNEAPOLIS	MN	55485-6137	
RSHOWS, LLC	PO BOX 1207		PURVIS	MS	39475	
RUCKER, ANGELINA	REDACT	REDACT	REDACT	REDACT	REDACT	
RUCKER, BOBBY	REDACT		REDACT	REDACT	REDACT	
RUCKER, CHRIS	REDACT		REDACT	REDACT	REDACT	
RUDOLPH, ANIESHA	REDACT		REDACT	REDACT	REDACT	
RUFFIN, DANIELLE	REDACT		REDACT	REDACT	REDACT	
RUNNELS, TRAVIS	REDACT		REDACT	REDACT	REDACT	
RUNNERS INTERNATIONALE	333 CHABANEL WEST	STE 504	MONTREAL	QUEBEC	H2N 2E7	
	Attn: Chip Fletcher, County					
Ruskin, Hillsborough County	Attorney	800 Twiggs Street	Tampa	FL	33602	
RUSSELL C. SIMON	CHAPTER 13 TRUSTEE	PO BOX 1898	MEMPHIS	TN	38101-1898	
RUSSELL COUNTY	P.O. BOX 669		PHENIX CITY	AL	36868	
RUSSELL, ALYCIA	REDACT		REDACT	REDACT	REDACT	

Name	Address 1	Address 2	City	State	PostalCode	Country
RUSSELL, SAMANTHA	REDACT	REDACT	REDACT	REDACT	REDACT	
RUTHERFORD COUNTY	BUSINESS TAX	319 NORTH MAPLE STREET	MURFREESBURO	TN	37130	
RYAN, COLIN	REDACT		REDACT	REDACT	REDACT	
S & L MANAGEMENT, LLC	6547 DEVONHURST DR		ST. LOUIS	MO	63129	
S4U	2633 NEVIN AVE		LOS ANGELES	CA	90011	
SAFE-KEY SECURITY	PO BOX 265249		DAYTONA BEACH	FL	32126	
SAGE SOFTWARE, INC	14855 COLLECTION CENTER DRIVE		CHICAGO	IL	60693	
SAIA MOTOR FREIGHT LINE, LLC	P.O. BOX 730532		DALLAS	TX	75373-0532	
SAIGON WASH & RAG COMPANY	114 Bokoshe Circle		Loudon	TN	37774	
SALAMANCA, AMANDA	REDACT		REDACT	REDACT	REDACT	
SALDONA, DORA	REDACT		REDACT	REDACT	REDACT	
	c/o Moonbeam Capital					
SALEM CONSUMER SQUARE OH LLC	Investments, LLC	2100 Pleasant Hill Road	Duluth	GA	30096	
SALGADO, DAISY	REDACT		REDACT	REDACT	REDACT	
SAM'S CLUB	3900 Grants Mill Rd.		Irondale	AL	35210	
SANCHEZ, ALFREDO	REDACT		REDACT	REDACT	REDACT	
SANCHEZ, JULISSA	REDACT		REDACT	REDACT	REDACT	
SANDERS AIR AND HEAT	611 -22ND AVE SOUTH		MERIDIAN	MS	39301	
SANDERS, CARLA	REDACT		REDACT	REDACT	REDACT	
SANDERS, DIANE	REDACT		REDACT	REDACT	REDACT	
SANDERS, JASMINE	REDACT		REDACT	REDACT	REDACT	
SANDERS, KANISHA	REDACT		REDACT	REDACT	REDACT	
SANDERS, SHALEA	REDACT		REDACT	REDACT	REDACT	
SANDERS, SIDNE	REDACT		REDACT	REDACT	REDACT	
SANITATION DISTRICT NO.1	1045 Eaton Dr.		Ft. Wright	KY	41017	
SANITATION DISTRICT NO.1	P.O. BOX 12112		COVINGTON	KY	41012-0112	
SANSONE PLAZA ON THE						
BOULEVARD	ATTN: TIMOTHY S. SANSONE	120 S. Central Avenue	Clayton	МО	63105	
SANSONE PLAZA ON THE						
BOULEVARD LLC	ATTN: TIMOTHY S. SANSONE	120 S CENTRAL AVENUE	CLAYTON	МО	63105	
SANTADER BANK	Santander Way	R11 EPV 02 23	East Providence	RI	2915	
SAPD ALARMS OFFICE	315 S. SANTA ROSA STREET		SAN ANTONIO	TX	78207	
SARAMAX	1372 BROADWAY	7TH FLR	NEW YORK	NY	10018	
SAROJ INTERNATIONAL	2615 S SANTA FE AVE		VERNON	CA	90058	
SASHA	460/A MAIN AVENUE		WALLINGTON	NJ	07057	
SATIN REAL ESTATE, LC	P.O. BOX 941		ST. CHARLES	MO.	63302-0941	
SATKHRAM PERSAUD	1407 BROADWAY SUITE 1414	1	NEW YORK CITY	NY	10018	

Name	Address 1	Address 2	City	State	PostalCode	Country
SAUER, SALLY	REDACT		REDACT	REDACT	REDACT	
SAUL HOLDINGS LIMITED						
PARTNERSHIP	P O Box 64288		Baltimore	MD	21264-4288	
SAUL HOLDINGS, LP	PO BOX 64288		BALTIMORE	MD	21264-4288	
SAUL SUBSIDIARY I, LP	PO BOX 64812		BALTIMORE	MD	21264-4812	
SAUNDERS, BRALYNN	REDACT		REDACT	REDACT	REDACT	
SAUNDERS, MARIYA	REDACT		REDACT	REDACT	REDACT	
SAVANTE APPAREL	1400 BROADWAY, ROOM 1207		NEW YORK	NY	10018	
SAVICE, MORRILYN	REDACT		REDACT	REDACT	REDACT	
SAY WHAT/AMERICAN ATTITUDE	1639 CENTRE ST.		RIDGEWOOD	NY	11385	
SB Capital Group, LLC	Attn.: Robert Raskin	1010 Northern Bvd.	Great Neck	NY	11021	
SBS FINANCIAL SERVICES, INC	1818 W BELMONT AVENUE	SUITE 203	CHICAGO	IL	60657	
SC DEPARTMENT OF REVENUE	CORPORATION TENTATIVE		COLUMBIA	sc	29214-0006	
SC Dept of Revenue	PO Box 125		Columbia	sc	29214	
SC DEPT OF REVENUE	PO BOX 125		COLUMBIA	sc	29214-0215	
SC DEPT OF REVENUE						
(GARNISHMENT)	DEPT 04/W/08		FLORENCE	sc	29502	
SC DPT OF EMPLOYMENT &						
WORKFORCE	ATTN: BPC COLLECTIONS UNIT	PO BOX 2644	COLUMBIA	sc	29202	
SC EMPLOYMENT SECURITY COMM	PO BOX 7103		COLUMBIA	sc	29202	
SCALES, LAQUITA	REDACT		REDACT	REDACT	REDACT	
SCANA ENERGY	3344 Peachtree Rd		Atlanta	GA	30326	
SCANA ENERGY	PO BOX 100157		COLUMBIA	SC	29202-3157	
SCAPLEN, AMBER	REDACT		REDACT	REDACT	REDACT	
SCAPLEN, AMBER	REDACT	REDACT	REDACT	REDACT	REDACT	
SCARLET, SHAQUILLE	REDACT		REDACT	REDACT	REDACT	
SCC	CLERKS OFFICE	P.O. BOX 1197	RICHMOND	VA	23218	
SCE&G	1487 Jefferies Highway		Walterboro	Sc	29488	
SCE&G	PO BOX 100255		COLUMBIA	SC	29202-3255	
SCHEFFLER, KILEY	REDACT		REDACT	REDACT	REDACT	
SCHER, CORI	REDACT		REDACT	REDACT	REDACT	
SCHICKEL ENTERPRISES, LTD						
PRTNRSHP	11601 PLEASANT RIDGE RD	SUITE 300	LITTLE ROCK	AR	72212-2241	
SCHILLECI MADISON, LLC	1761 Platt Place		Montgomery	AL	36124-2942	
SCHILLECI MADISON, LLC	C/O JMS REALTY, LLC	PO BOX 242942	MONTGOMERY	AL	36124-2942	
SCHULER, JADA	REDACT		REDACT	REDACT	REDACT	
SCOTT C'S AUTO OUTLET	524 MILEGROUND ROAD		MORGANTOWN	WV	26505	

Name	Address 1	Address 2	City	State	PostalCode	Country
SCOTT ELECTRIC (TX)	PO BOX 1884		ROANOKE	TX	76262	
SCOTT ELECTRIC, LLC (OH)	6013 BAHIAMAR ROAD		TOLEDO	ОН	43611	
SCOTT HOLCOMBE	3041 CANTERBURY RD.		BIRMINGHAM	AL	35223	
SCOTT SERVICES COMPANY	5636 CLIFFORD CIRCLE		BIRMINGHAM	AL	35210	
SCOTT, ANGELICA	REDACT		REDACT	REDACT	REDACT	
SCOTT, CASSANDRA	REDACT		REDACT	REDACT	REDACT	
SCOTT, CHRISTINA	REDACT		REDACT	REDACT	REDACT	
SCOTT, DAMEIKA	REDACT		REDACT	REDACT	REDACT	
SCOTT, DWIGHT	REDACT		REDACT	REDACT	REDACT	
SCOTT, GERTRUDE	REDACT		REDACT	REDACT	REDACT	
SCOTT, KECIA	REDACT		REDACT	REDACT	REDACT	
SCOTT, SHARPE	REDACT		REDACT	REDACT	REDACT	
SCOTT-WILLIAMS, LAVONDA	REDACT		REDACT	REDACT	REDACT	
SCOTTY GRAY FLOORING	5722 CARRINGTON LAKE PKWY		TRUSSVILLE	AL	35173	
SCRUGGS, DESMOND	REDACT		REDACT	REDACT	REDACT	
SE&M	P.O. BOX 7159		ROCKY MOUNT	NC	27804	
SEAGRAVE, DOMINIQUE	REDACT		REDACT	REDACT	REDACT	
SEAGULL SELECT	4005 ROYAL DRIVE	SUITE 400	KENNESAW	GA	30144	
SEALEY, ADREANN	REDACT	REDACT	REDACT	REDACT	REDACT	
SEAWAY NATIONAL BANK	645 East 87th St.		Chicago	IL	60619	
SECREATRY OF STATE (MO)	CORPORATION DIVISION	P.O. BOX 778	JEFFERSON CITY	MO	65102	
SECRETARY OF STATE (TX)	FILINGS- CORPORATION SECTION	P.O. BOX 13697	AUSTIN	TX	78711-3697	
	BUSINESS & COMMERCIAL					
SECRETARY OF STATE (AR)	SERVICES	SERVICES DIVISION	LITTLE ROCK	AR	72203-8014	
SECRETARY OF STATE (IL)	DEPT OF BUSINESS SERVICES	501 S. SECOND STREET	SPRINGFIELD	IL	62756	
SECRETARY OF STATE (SC)	1205 PENDLETON STREET	SUITE 525	COLUMBIA	SC	29201	
SECRETARY OF STATE (MS)	PO BOX 136		JACKSON	MS	39205-0136	
SECRETARY OF STATE (NC)	CORPORATIONS DIVISION	PO BOX 29622	RALEIGH	NC	27626-0622	
SECRETARY OF STATE (OH)	PO BOX 1329		COLUMBUS	ОН	43216	
SECRETARY OF STATE BUSINESS						
SERVICE	P.O. BOX 5616		MONTGOMERY	AL	36103-5616	
SECURE DESTRUCTION SERVICE	P.O. BOX 18574		HUNTSVILLE	AL	35804-8574	
SECURITY FINANCE	3801 NAMEOKI ROAD	STE. 15	GRANITE CITY	IL	62040	
SEESE, PA	1 E. BROWARD BLVD	SUITE 700	FT LAUDERDALE	FL	33301	
SELECT CLOTHING	1725 S MAPLE		LOS ANGELES	CA	90015	
SELLERS RICHARDSON HOLMAN &						
WEST LLP	216 SUMMIT BOULEVARD	SUITE 300	BIRMINGHAM	AL	35243	

Name	Address 1	Address 2	City	State	PostalCode	Country
SELLERS, DEDREA	REDACT		REDACT	REDACT	REDACT	
SELMA COMMUNITY BUILDERS, INC	C/O SELMA MALL, ANTHONY	ALBERIGI, REG MGR	SELMA	AL	36703	
SELMA COMMUNITY BUILDERS, INC.	c/o Aronov Realty Management	1391 E. Highland Avenue	Selma	AL	36703	
SELMA WATER WORKS	1600 Selma Avenue		Selma	AL	36702	
SELMA WATER WORKS	PO BOX 326		SELMA	AL	36702-0326	
SENSATIONAL COLLECTIONS	1407 BROADWAY		NEW YORK CITY	NY	10018	
SENSUAL, INC.	183 MADISON AVE, STE 401		NEW YORK	NY	10016	
SENTRY	PO BOX 885		HILLBURN	NY	10931	
SENTRY TECHNOLOGY CORP	1881 LAKELAND AVE		RONKONKOMA	NY	11779-7416	
SERVICE MASTER PROPERTY						
RESTORATION	4105 HICKORY HILL ROAD #101		MEMPHIS	TN	38115	
SERVIN, GLADYS	REDACT		REDACT	REDACT	REDACT	
SEUELL, CHRIS	REDACT		REDACT	REDACT	REDACT	
SEVEN APPAREL GROUP	347 5TH AVENUE	SUITE 201	NEW YORK	NY	10016	
SEVEN GRAND ASSOCIATES, LLC	2451 E. BASELINE ROAD	SUITE 300	GILBERT	AZ	85234	
SEVEN STAR ENTERPRISES	9009 SUNDANCE LANE		SHREVEPORT	LA	71106	
SEWELL, TANTALINE	REDACT		REDACT	REDACT	REDACT	
SEWER & WATER BOARD OF NEW						
ORLEANS	625 SAINT JOSEPH STREET		NEW ORLEANS	LA	70165-6501	
SG FOOTWEAR	3 UNIVERSITY PLAZA	SUITE 400	HACKENSACK	NJ	07601	
SG2, LLC	C/O MALLORY PROPERTIES	445 W. OKLAHOMA AVENUE	MILWAUKEE	WI	53207	
SGP - SIGNS GRAPHICS PRINTING,						
INC	PO BOX 720477		ATLANTA	GA	30358-2477	
SGR APPAREL	1385 BROADWAY, RM 910		NEW YORK	NY	10018	
SHACKELFORD, EBONIE	REDACT	REDACT	REDACT	REDACT	REDACT	
SHAG OF MISSISSIPPI, INC.	Donnie Hendrix	P.O. Box 1207	Purvis	MS	39475	
SHAG OF MISSISSIPPI, INC.	P.O. BOX 1207		PURVIS	MS	39475	
Shah, Bhavana	2110 N.W. 95th Avenue		Miami	FL	33172	
Shah, Shail	2110 N.W. 95th Avenue		Miami	FL	33172	
Shah, Swapnil	2110 N.W. 95th Avenue		Miami	FL	33172	
Shail Shah Revocable Trust	2110 N.W. 95th Avenue		Miami	FL	33172	
SHAIL SHAH TRUST	C/O SWAPNIL SHAH	2110 NW 95TH AVE	MIAMI	FL	33172	
SHALOM INTL CORP	1050 AMBOY AVENUE		PERTH AMBOY	NJ	08861	
SHANNON SQUARE, LLC	181 S. FRANKLIN AVE.	SUITE 302	VALLEY STREAM	NY	11581	
SHANNON, HALETHER	REDACT		REDACT	REDACT	REDACT	
SHARIF, AASIYA	REDACT		REDACT	REDACT	REDACT	

Name	Address 1	Address 2	City	State	PostalCode	Country
SHARON MARION	12725 ORLEY DRIVE		FLORESSANT	MO	63033	
SHARPE, DE'MARCUS	REDACT		REDACT	REDACT	REDACT	
SHAW, DEBBIE	REDACT		REDACT	REDACT	REDACT	
SHELBY ASSOCIATES	888 BRICKELL AVE SUITE 100		MIAMI	FL	33131	
SHELBY COUNTY TRUSTEE	PO BOX 2751		MEMPHIS	TN	38101-2751	
SHELBY COUNTY, TN	BUSINESS TAX	150 WASHINGTON AVE.	MEMPHIS	TN	38103	
SHELTON, CAMERON	REDACT		REDACT	REDACT	REDACT	
SHELTON, ELLEN	REDACT		REDACT	REDACT	REDACT	
SHELTON, JONYA	REDACT	REDACT	REDACT	REDACT	REDACT	
SHEMIKA HALL	20120 HANNA		DETROIT	MI	48203	
SHEPARD, BRIGITTE	REDACT	REDACT	REDACT	REDACT	REDACT	
SHEPHERD, BOBBIE	REDACT		REDACT	REDACT	REDACT	
SHEPHERD, TERRYN	REDACT		REDACT	REDACT	REDACT	
SHERRILL COMPANY, LLC	PO BOX 7122		COLUMBUS	GA	31908-7122	
SHERROD, SHANNON	REDACT		REDACT	REDACT	REDACT	
SHERWIN WILLIAMS (AL)	1905 CRESTWOOD BLVD		BIRMINGHAM	AL	35210-2033	
SHI INTERNATIONAL CORPORATION	P.O. BOX 952121 2821 S OLIVE STREET		DALLAS LOS ANGELES	TX CA	75395-2121 90007	
SHILAT CLOTHING	55 WEST BEAVER CREEK RD	UNIT 23		ONTARIO	L4B 1K5	
SHINE MARK ACCESSORIES		UNII 23	RICHMOND HILL		91789	
SHOE REPUBLIC	21301 FERRERO PKWY		WALNUT	CA	10001	
SHOP CITY PW/LB, LLC	314 Fifth Avenue C/O PLATINUM PROPERTY MGMT		New York	NY	10001	
SHOPPES OF LIBERTY CITY, LLC	INC	11098 BISCAYNE BLVD	MIAMI	FL	33161	
SHOPS ON BLUE PARKWAY, LLC						
(SWOPE COMMUNITY BUILDERS)	4001 Blue Parkway		Kansas City	MO	64130	
SHOPS ON BLUE PARKWAY. LLC	C/O COLLIERS INTERNATIONAL	4520 MAIN STREET SUITE 1000	KANSAS CITY	MO	64111	
SHOPS ON MAIN ASSOCIATES, LLC	678 Reisterstown Road		Baltimore	MD	21208	
SHOPS ON MAIN STREET LLC	678 REISTERSTOWN RD		BALTIMORE	MD	21208	
SHORELINE	100 Lauman Lane		Hicksville Lane	NY	11801	
SHORTER, GARY	REDACT		REDACT	REDACT	REDACT	
SHORTER, LATOYA	REDACT	REDACT	REDACT	REDACT	REDACT	
SHORTER, LEMOND	REDACT		REDACT	REDACT	REDACT	
SHOWS DEVELOPMENT, INC.	PO BOX 1358		PURVIS	MS	39475	
SHREVEPORT WESTWOOD						
INVESTORS, LLC	Attn.: Alan Light, Manager	9907 E Bell Road	Scottsdale	AZ	85260	

Name	Address 1	Address 2	City	State	PostalCode	Country
SHREVEPORT WESTWOOD						
INVESTORS, LLC	9907 E. BELL ROAD, SUITE 110		SCOTTSDALE	AZ	85260	
SHUNTAIN MILLER	3139 KELLING ST		HOUSTON	TX	77045	
SIFUENTES, CARMEN	REDACT		REDACT	REDACT	REDACT	
SIGMON, DANA	REDACT		REDACT	REDACT	REDACT	
SIGN GRAPHICS PRINTING, INC.	PO BOX 720477		ATLANTA	GA	30358-2477	
SILLMAN, KACEY	REDACT		REDACT	REDACT	REDACT	
SILVER HILL MANAGEMENT CO., LLC	P.O. BOX 170440		MILWAUKEE	WI	53217	
SIEVER THEE IVIANAGENIENT CO., EEC	F.O. BOX 170440		WILWAOKEE	VVI	33217	
SILVER MILL MANAGEMENT CO., LLC	7177 N. Port Washington Road,	Suite 100	Glendale	WI	53217	
SILVER TEXTILES, INC.	2101 SOUTH FLOWER ST		LOS ANGELES	CA	90007	
SIMIEN, ANGELA	REDACT		REDACT	REDACT	REDACT	
SIMIEN, PHILLIP	REDACT		REDACT	REDACT	REDACT	
SIMMONS FIRST	PO Box 7009		Pine Bluff	AR	71611-7009	
SIMMONS, CHARMAINE	REDACT		REDACT	REDACT	REDACT	
SIMMONS, GABRIELLE	REDACT		REDACT	REDACT	REDACT	
SIMMONS, MARY	REDACT		REDACT	REDACT	REDACT	
SIMMONS, QUIXARI	REDACT		REDACT	REDACT	REDACT	
SIMMS, MARKEISHA	REDACT		REDACT	REDACT	REDACT	
SIMON, JMONTEA	REDACT		REDACT	REDACT	REDACT	
SIMON, J'MONTEA	REDACT		REDACT	REDACT	REDACT	
SIMPKINS, COLBY	REDACT		REDACT	REDACT	REDACT	
SIMPSON, CHYENNE	REDACT	REDACT	REDACT	REDACT	REDACT	
SIMPSON, KENNDRICK	REDACT		REDACT	REDACT	REDACT	
SIMS, ANGELINA	REDACT	REDACT	REDACT	REDACT	REDACT	
SINDRELLA DISTRIBUTORS, LLC	1407 BROADWAY, STE 2011		NEW YORK	NY	10018	
SINGLETARY, TYREE	REDACT		REDACT	REDACT	REDACT	
SINGLETON, LARRY	REDACT		REDACT	REDACT	REDACT	
SINGLETON, ZHANE'	REDACT		REDACT	REDACT	REDACT	
Sirranville Lewis	1475 Tomahawk Road		Birmingham	AL	35214	
SK INTERNATIONAL	1385 BROADWAY		NEW YORK	NY	10018	
SKINDEEP	545 WEST 111TH STREET 4K		NEW YORK	NY	10025-1962	
SKIVER, BROOKLYN	REDACT		REDACT	REDACT	REDACT	
SKYSE, KEOSHA	REDACT		REDACT	REDACT	REDACT	
SLACK, TENISHA	REDACT		REDACT	REDACT	REDACT	
SLATER, KAREMA	REDACT		REDACT	REDACT	REDACT	
SLAUGHTER, MICHELLE	REDACT		REDACT	REDACT	REDACT	

Name	Address 1	Address 2	City	State	PostalCode	Country
SMALL, RONITA	REDACT		REDACT	REDACT	REDACT	
SMALLS, SANTANA	REDACT		REDACT	REDACT	REDACT	
SMITH COUNTY	TAX COLLECTOR	P.O. BOX 2011	TYLER	TX	75710-2011	
SMITH, ANDREA	REDACT		REDACT	REDACT	REDACT	
SMITH, ANGELIA	REDACT		REDACT	REDACT	REDACT	
SMITH, BRANDON	REDACT		REDACT	REDACT	REDACT	
SMITH, CHARLOTTE	REDACT		REDACT	REDACT	REDACT	
SMITH, DEASA	REDACT		REDACT	REDACT	REDACT	
SMITH, DEBRA	REDACT	REDACT	REDACT	REDACT	REDACT	
SMITH, DENODRE	REDACT		REDACT	REDACT	REDACT	
SMITH, DEVIN	REDACT		REDACT	REDACT	REDACT	
SMITH, DOMINIQUE	REDACT	REDACT	REDACT	REDACT	REDACT	
SMITH, JAROD	REDACT		REDACT	REDACT	REDACT	
SMITH, JR., KEVIN O.	REDACT	REDACT	REDACT	REDACT	REDACT	
SMITH, KYRA	REDACT		REDACT	REDACT	REDACT	
SMITH, LAKESHA	REDACT		REDACT	REDACT	REDACT	
SMITH, LATOYA	REDACT		REDACT	REDACT	REDACT	
SMITH, MARK	REDACT		REDACT	REDACT	REDACT	
SMITH, MARQUITA	REDACT		REDACT	REDACT	REDACT	
SMITH, MONIQUE	REDACT		REDACT	REDACT	REDACT	
SMITH, NUHKISHAA	REDACT		REDACT	REDACT	REDACT	
SMITH, REGAL	REDACT		REDACT	REDACT	REDACT	
SMITH, SADIE	REDACT		REDACT	REDACT	REDACT	
SMITH, SEDRIA	REDACT		REDACT	REDACT	REDACT	
SMITH, SHAMANE	REDACT		REDACT	REDACT	REDACT	
SMITH, SHAVON	REDACT	REDACT	REDACT	REDACT	REDACT	
SMITH, SHELBY	REDACT		REDACT	REDACT	REDACT	
SMITH, SHEMIKA	REDACT		REDACT	REDACT	REDACT	
SMITH, TRENA	REDACT		REDACT	REDACT	REDACT	
SMITH, TWANA	REDACT		REDACT	REDACT	REDACT	
SMITH, VERONICA	REDACT		REDACT	REDACT	REDACT	
SMOOT, ANGELA	REDACT		REDACT	REDACT	REDACT	
SMOTHERS, ALEXIS	REDACT		REDACT	REDACT	REDACT	
SMOTHERS, TANNER	REDACT		REDACT	REDACT	REDACT	
SMYRNA UTILITIES	2800 King Street Southeast		Smyrna	GA	30080	
SMYRNA UTILITIES	P.O. BOX 290009		NASHVILLE	TN	37229-0009	
SNAPPING SHOALS EMC	14750 Brown Bridge Road		Covington	GA	30016	

Name	Address 1	Address 2	City	State	PostalCode	Country
SNAPPING SHOALS EMC	PO BOX 73		COVINGTON	GA	30015-0073	,
SNEED, KATHY	REDACT	REDACT	REDACT	REDACT	REDACT	
SNIDER, CRYSTAL	REDACT		REDACT	REDACT	REDACT	
SNOW, ALEXANDRIA	REDACT	REDACT	REDACT	REDACT	REDACT	
SNOWDEN, KIANNA	REDACT	REDACT	REDACT	REDACT	REDACT	
SNUGWALLET COMPANY	110 VISTA CENTER DR		FOREST	VA	24551	
SOB	1504 S. MAIN ST.		LOS ANGELES	CA	90015	
SOHO APPAREL LTD	525 7TH AVE		NEW YORK	NY	10018	
SOLAR CONTRACT CARPET	12227 BEECH DALY ROAD		REDFORD	MI	48239	
SOLARWINDS	PO BOX 730720		DALLAS	TX	75373-0720	
SOLOFF PROPERTIES LTD-						
MILLEDGEVILLE	C/O SOLOFF PROPERTIES INC	427 EAST 5TH ST.	CHATTANOOGA	TN	37403	
SOLOMAN, JAZZMON	REDACT		REDACT	REDACT	REDACT	
SOLUTION ORIENTED, LLC	1421 PAVILLON DRIVE		BIRMINGHAM	AL	35226-2235	
SOSA, ANA	REDACT		REDACT	REDACT	REDACT	
South Carolina Department of						
Revenue	PO Box 12265		Columbia	SC	29211	
SOUTH CAROLINA TAX COMMISSION	SALES TAX DIVISION	PO BOX 125	COLUMBIA	sc	29214	
SOUTH END GLASS & MIRROR, INC	5559 NATIONAL TURNPIKE		LOUISVILLE	ку	40214-3725	
SOUTH FULTON RETAIL						
DEVELOPMENT LLC	1946 MONROE DR NE		ATLANTA	GA	30324	
SOUTHARD, LINDA	REDACT		REDACT	REDACT	REDACT	
SOUTHEASTERN FREIGHT LINES, INC.						
(SEFL)	P.O. BOX 1691		COLUMBIA	sc	29202	
SOUTHEASTERN INVESTMENT						
GROUP, LLC	PO BOX 235021		MONTGOMERY	AL	36123-5021	
SOUTHERN AIR AND ELECTRIC	6060 HIGHWAY 165		BALL	LA	71405	
SOUTHERN COMPANY						
(CONSOLIDATED)	30 Ivan Allen Jr. Blvd. NW		Atlanta	GA	30308	
SOUTHERN COMPANY						
(CONSOLIDATED)	PO BOX 105090		ATLANTA	GA	30348-5090	
Southern Development of	FAMILY DOLLAR CENTER/Store					
Mississippi	No165	PO BOX 1207	Purvis	MS	39475	
SOUTHERN DEVELOPMENT OF						
MISSISSIPPI	40 Deep South Lane , P.O. Box 1207		Purvis	MS	39475	

Name	Address 1	Address 2	City	State	PostalCode	Country
SOUTHERN DEVELOPMENT OF						,
MISSISSIPPI	P.O. Box 1207		Purvis	MS	39475	
SOUTHERN DEVELOPMENT OF MS	PO BOX 1207		PURVIS	MS	39475	
SOUTHERN DEVELOPMENT OF MS	P.O. BOX 1207		PURVIS	MS	39475	
SOUTHERN PLAZA, LLC	1 East Oak Hill Drive	Suite 302	Westmont	IL	60559	
SOUTHERN SAFETY SUPPLY CO.	2135 BROADWAY		MACON	GA	31206	
SOUTHFIELD POLICE DEPT	PO BOX 2055		SOUTHFIELD	MI	48037-2055	
SOUTHGATE AUGUSTA EQUITIES,	Meybohm Commercial Properties,					
LLC	LLC (Broker)	55 Fifth Avenue	New York	NY	10003-4398	
SOUTHGATE AUGUSTA EQUITIES.	C/O DIAL, DUNLAP & EDWARDS,					
LLC	LLC	PO BOX 1504	COLUMBIA	SC	29202	
SOUTHGATE AUGUSTA SHOPPING						
CTR GRP, LLC	M&T BANK-ONE M&T PLAZA, 7TH	FL CT ATTN: J. STAPLEY	BUFFALO	NY	14203	
SOUTHGATE CENTER ASSOCIATES,						
LLC	20950 Libby Road		Maple Hts.	ОН	44137	
	·					
SOUTHLAND MALL PROPERTIES LLC	P O BOX 730613		DALLAS	TX	75373-0613	
	c/o Gumberg Asset Management					
SOUTHLAND MALL PROPERTIES, LLC	Corp. Agent	3200 N. Federal Highway	Ft. Lauderdale	FL	33306	
SOUTHLAND MALL PROPERTIES, LLC	PO BOX 865061		ORLANDO	FL	32886-5061	
SOUTHLAND MALL SHOPPING						
CENTER LLC	P.O. BOX 934160		ATLANTA	GA	31193-4160	
SOUTHPOINTE PLAZA LLC	8816 SIX FORKS RD	SUITE 201	RALEIGH	NC	27615	
SOUTHTOWN GLASS COMPANY	11219 HICKMAN MILLS DRIVE		KANSAS CITY	MO	64134-4204	
SOUTHWESTERN ELECTRIC POWER	428 Travis St		Shreveport	LA	71101	
SOUTHWESTERN ELECTRIC POWER	PO BOX 24422		CANTON	ОН	44701-4422	
SOUTHWIND ELECTRIC, INC.	3001 EASTERN BOULEVARD		MIDDLE RIVER	MD	21221	
SPALDING SOFTWARE, INC.	P.O. BOX 921188		NORCOSS	GA	30010-1188	
SPANN, CHRISTY	REDACT		REDACT	REDACT	REDACT	
SPARKS, ANTHONY	REDACT		REDACT	REDACT	REDACT	
SPARKS, SARAH	REDACT		REDACT	REDACT	REDACT	
SPARKS, TONY	REDACT		REDACT	REDACT	REDACT	

Name	Address 1	Address 2	City	State	PostalCode	Country
SPARTANBURG COUNTY TREASURER	PO BOX 100260		COLUMBIA	sc	29202-3260	
SPECIALIZED LOGISTICS, LLC	PO BOX 815		PINSON	AL	35126	
SPECIALTY ROLL PRODUCTS, INC.	601 25TH AVE	PO BOX 5374	MERIDIAN	MS	39301	
SPECIALTY STORE SERVICES	454 JARVIS		DES PLAINS	IL	60018	
SPEEDWAY WATERWORKS	1450 North Lynhurst Drive		Indianapolis	IN	46224	
SPEEDWAY WATERWORKS	PO BOX 6485		INDIANAPOLIS	IN	46206	
SPEER, DENZEL	REDACT		REDACT	REDACT	REDACT	
SPENCER, ARIANA	REDACT		REDACT	REDACT	REDACT	
SPENCER, CLAUDIA	REDACT		REDACT	REDACT	REDACT	
SPENCER, CLAUDIA	REDACT		REDACT	REDACT	REDACT	
SPENCER, EBONY	REDACT		REDACT	REDACT	REDACT	
SPENCER, LATANYA	REDACT		REDACT	REDACT	REDACT	
SPENCER, SIERRA	REDACT		REDACT	REDACT	REDACT	
SPIVEY, AMBER	REDACT		REDACT	REDACT	REDACT	
SPOON JEANS	1407 BROADWAY		NEW YORK	NY	10018	
SPORTSWEAR GROUP, LLC	1407 BROADWAY	SUITE 2805	NEW YORK	NY	10018	
SPRATLEY-COOPER, DEBORAH	REDACT		REDACT	REDACT	REDACT	
SPRIGGS, LEON	REDACT		REDACT	REDACT	REDACT	
SPRING IMPORT	1407 BROADWAY, STE 715		NEW YORK	NY	10018	
SPRINGER EQUIPMENT CO.	PO BOX 100274		IRONDALE	AL	35210	
SPRINGFIELD ELECTRIC	1306 POPLAR		GREENWOOD	MS	38930	
SPRINKLER MATIC	4740 DAVIE ROAD		DAVIE	FL	33314	
SPRINT (CAROL STREAM, IL)	PO BOX 4191		CAROL STREAM	IL	60197-4191	
SPRINT (LONDON, KY)(DM						
LAPTOPS)	PO BOX 8077		LONDON	KY	40742	
SPYLER & CO.	10 WEST 33RD STREET	SUITE 240	NEW YORK	NY	10001	
SQUARE 67 SHOPPING CENTER	ISENBERG MANAGEMENT ASSOC.	400 S.ZANG BLVD., SUITE 1220	DALLAS	TX	75208	
SS NESBITT & CO., INC	3500 BLUE LAKE DRIVE	SUITE 120	BIRMINGHAM	AL	35243	
St. Landry Parish	Sales and Use Tax Dept	PO Box 1210	Opelousas	LA	70571-1210	
ST. LOUIS COUNTY	7900 CARONDOLET AVENUE		CLAYTON	МО	63105	
ST. LOUIS COUNTY	COLLECTOR OF REVENUE	P.O.BOX 11491	ST. LOUIS	MO	63105-0291	
ST. LUCIE COUNTY TAX COLLECTOR	PO BOX 308		FORT PIERCE	FL	34954-0308	
St. Martin Parish	Sales & Use Tax Dept.	PO Box 1000	Breaux Bridge	LA	70517	
ST. STEPHENS SQUARE PARTNERS	C/O HEAD COMPANIES	P O BOX 230	Point Clear	AL	36564	

Page 148 of 176

Name	Address 1	Address 2	City	State	PostalCode	Country
ST. STEPHEN'S SQUARE						
PARTNERSHIP	C/O HEAD COMPANIES	P.O. BOX 230	POINT CLEAR	AL	36564	
ST. VINCENT'S WELLNESS SERVICES	ATTN: KIMBERLY RIDER	2800 UNIVERSITY BLVD	BIRMINGHAM	AL	35233	
STACY, ANTOINETTE	REDACT		REDACT	REDACT	REDACT	
STAFFMARK	ATTN: U.S. BANK	P.O. BOX 952386	ST. LOUIS	MO	63195	
STALCUP, COURTNEY	REDACT		REDACT	REDACT	REDACT	
STALEY, PEGGY	REDACT		REDACT	REDACT	REDACT	
STANDARD INSURANCE CO (3358)	PO BOX 3358		PORTLAND	OR	97208-3358	
STANDRING, MARYLYNN	REDACT		REDACT	REDACT	REDACT	
STANKIEWICZ, KRISTA	REDACT		REDACT	REDACT	REDACT	
Stanley Convergent Security						
Solutions, Inc.	101 Airview Lane		Alabaster	IL	35007	
Stanley Convergent Security						
Solutions, Inc.	9998 Crosspoint Boulevard	Suite 200	Indianapolis	IN	46256	1
Stanley Convergent Security						
Solutions, Inc.	55 Shurman Boulevard	Suite 900	Naperville	IL	60563	
STANLEY SECURITY	SOLUTIONS INC.	DEPT CH 14202	PALATINE	IL	60055-4202	
STANLEY, JENESHIA	REDACT		REDACT	REDACT	REDACT	
STAPLES, ANGELA	REDACT		REDACT	REDACT	REDACT	
STAPLES, INC.	DEPT. ATL	PO BOX 405386	ATLANTA	GA	30384-5386	
STARLIGHT ACCESSORIES	10 WEST 33 STREET	SUITE 800	NEW YORK	NY	10001	
STARWEAR, INC.	1384 BROADWAY	SUITE 1606	NEW YORK	NY	10018	
	COMPTROLLER OF PUBLIC					
STATE COMPTROLLER - TEXAS	ACCOUNTS	PO BOX 149348	AUSTIN	TX	78714-9348	
STATE CORPORATION COMMISSION	P.O. BOX 1197		RICHMOND	VA	23218-1197	1
				_		
STATE COURT OF FULTON COUNTY	185 CENTRAL AVE, SW		ATLANTA	GA	30303	1
STATE DEPT OF ASSESSMENT &				_		
TAXATION	301 W. PRESTON ST, ROOM 801		BALTIMORE	MD	21201	
STATE DISBURSEMENT UNIT	P.O. BOX 5400		CAROL STREAM	IL	60197-5400	
STATE INSURANCE FUND	DISABILITY BENEFITS	PO BOX 5261	BINGHAMTON	NY	13902-5261	
STATE LIFE INSURANCE (CA)	LONG TERM ADMIN OFFICE	PO BOX 4243	WOODLAND HILLS	CA	91365-4243	
STATE OF AL DEPT. OF REVENUE	COLLECTION SERVICES DIVISION	P.O. BOX 327820	MONTGOMERY	AL	36132-7820	
				_		
State of Alabama Attorney General	Attention Bankruptcy Dept	P.O. Box 300152	Montgomery	AL	36130-0152	

Name	Address 1	Address 2	C:h.	State	Dostal Code	Carrature
Name	Address 1	Address 2	City	State	PostalCode	Country
State of Arkansas Attorney General	Attention Bankruptcy Dept	323 Center St.	Little Rock	AR	72201-2610	
State of Connecticut	Dept of Revenue Services	PO Box 2937	Hartford	CT	06104-2937	
State of connecticut	COMMISSIONER OF REVENUE	1 0 Box 2557	Hartiora	Ci	00104 2557	
STATE OF CONNECTICUT	SERVICES	P.O. BOX 2937	HARTFORD	ст	06104-2937	
State of Connecticut Attorney						
General	Attention Bankruptcy Dept	55 Elm St.	Hartford	СТ	06106	
State of Florida	Department of Revenue	5050 W. Tennessee Street	Tallahassee	FL	32399-0100	
STATE OF FLORIDA	DISBURSEMENT UNIT	PO BOX 8500	TALLAHASSEE	FL	32314-8500	
State of Florida Attorney General	Attention Bankruptcy Dept	The Capitol, PL 01	Tallahassee	FL	32399-1050	
State of Georgia Attorney General	Attention Bankruptcy Dept	40 Capital Square, SW	Atlanta	GA	30334-1300	
State of Illinois Attorney General	Attention Bankruptcy Dept	100 West Randolph Street	Chicago	IL	60601	
State of Indiana Attangua Consul	Attactics Basissonton Bast	Indiana Causana at Cantan Cauth	to die a e e e lie		46204	
State of Indiana Attorney General State of Kansas	Attention Bankruptcy Dept Department of Revenue	Indiana Government Center South 915 SW Harrison St	Indianapolis Topeka	IN KS	46204 66625	
State of Karisas	Department of Revenue	915 SW Harrison St	торека	K2	00025	
State of Kansas Attorney General	Attention Bankruptcy Dept	120 SW 10th Ave., 2nd Floor	Topeka	KS	66612-1597	
State of Kalisas Attorney General	Attention Bankruptcy Dept	120 3W 10th Ave., 2nd Floor	Торека	N3	00012-1397	
State of Kentucky Attorney General	Attention Bankruptcy Dept	700 Capitol Avenue, Suite 118	Frankfort	ΚY	40601	
state of Rentaerry Filterine, General	Accounted Burna aproy Bept	700 capitor/tremae, saite 110	T G T M G T C		10001	
State of Louisiana Attorney General	Attention Bankruptcy Dept	P.O. Box 94095	Baton Rouge	LA	70804-4095	
STATE OF MARYLAND	PERSONAL PROPERTY DIVISION	PO BOX 17052	BALTIMAORE	MD	21297-1052	
State of Maryland Attorney General	Attention Bankruptcy Dept	200 St. Paul Place	Baltimore	MD	21202-2202	
State of Massachusetts Attorney						
General	Attention Bankruptcy Dept	One Ashburton Place	Boston	MA	02108-1698	
STATE OF MICHIGAN	7064 CROWNER DRIVE		LANSING	MI	48980-0001	
STATE OF MICHIGAN (CORP DIV)	CORPORATION DIVISION	PO BOX 30054	LANSING	MI	48909	
		G. Mennen Williams Building, 7th				
State of Michigan Attorney General	Attention Bankruptcy Dept	Floor	Lansing	MI	48909-0212	
	l		1			
State of Mississippi Attorney General	Attention Bankruptcy Dept	Walter Sillers Building	Jackson	MS	39201	
	l				65400	
State of Missouri Attorney General	Attention Bankruptcy Dept	Supreme Court Building	Jefferson City	MO	65102	

Name	Address 1	Address 2	City	State	PostalCode	Country
State of New Jersey Attorney						
General	Attention Bankruptcy Dept	RJ Hughes Justice Complex	Trenton	NJ	08625-0080	
State of New York Attorney General	Attention Bankruptcy Dept	The Capitol	Albany	NY	12224-0341	
State of North Carolina Attorney						
General	Attention Bankruptcy Dept	9001 Mail Service Center	Raleigh	NC	27699-9001	
State of Ohio Attorney General	Attention Bankruptcy Dept	30 E. Broad St., 14th Floor	Columbus	ОН	43215	
State of Pennsylvania Attorney						
General	Attention Bankruptcy Dept	Strawberry Square	Harrisburg	PA	17120	
State of South Carolina Attorney						
General	Attention Bankruptcy Dept	P.O. Box 11549	Columbia	SC	29211-1549	
State of Tennessee Attorney General	Attention Bankruptcy Dept	P.O. Box 20207	Nashville	TN	37202-0207	
State of Texas Attorney General	Attention Bankruptcy Dept	Capitol Station	Austin	TX	78711-2548	
State of Texas Accorney deficial	recention bank aptcy bept	Capitor Station	/ tustiii	1/4	75711 2546	
State of Virginia Attorney General	Attention Bankruptcy Dept	900 East Main Street	Richmond	VA	23219	
State of Wisconsin Attorney General	Attention Bankruptcy Dept	Wisconsin Department of Justice	Madison	WI	53707-7857	
STATE STREET SHOPPING CENTER,	Attention Bankruptcy Dept	Wisconsin Department of Justice	Iviauison	VVI	33707-7837	
LLC	C/O PRIORITY PROPERTIES, LLC	1045 South Woods Mill Rd.	Town & Country	мо	63017	
STATE STREET SHOPPING CENTER,	C/OTRIORITT FROM ERTIES, EEC	1043 South Woods Will No.	Town & Country	IVIO	03017	
LLC	C/O PRIORITY PROPERTIES, LLC	1045 SOUTH WOODS MILL RD	TOWN & COUNTRY	мо	63017	
STATE TAX COMMISSION	OFFICE OF REVENUE	PO BOX 23050	JACKSON	MS	39225-3050	
STATON, MELBA	REDACT	1 6 BOX 23030	REDACT	REDACT	REDACT	
STEELE, NANETTE	REDACT	REDACT	REDACT	REDACT	REDACT	
STEELE, TORRIE	REDACT	REDACT	REDACT	REDACT	REDACT	
STEELYARD COMMONS, LLC	The Offices at Legacy Village	225333 Cedar Road	Lyndhurst	OH	44124	
STEELYARD COMMONS, LLC	3447 Steelyard Commons	22555 CCddi Nodd	Cleveland	ОН	44109	
STEELYARD COMMONS, LLC	P.O. BOX 635401		CINCINNATI	ОН	45263-5401	
STEELYARD COMMONS, LLC	C/O FIRST INTERSTATE	25333 CEDAR RD, STE 300	LYNDHURST	ОН	44124	
Stefanie R. Shelley, Esq.	Counsel for Gator	Tobin & Reyes, P.A.	Boca Raton	FL	33432	
STELLAR STAFFING, LLC.	P.O. BOX 202056		DALLAS	TX	75320-2056	
STEPHENS, TAMERRA	REDACT		REDACT	REDACT	REDACT	
STEPHENSON, JUSTIN	REDACT		REDACT	REDACT	REDACT	
STEPP, KIMBOLYN	REDACT		REDACT	REDACT	REDACT	
STERBENZ, DANIEL	REDACT	REDACT	REDACT	REDACT	REDACT	
Sterling Factors	P.O. Box 742-Midtwn Stn		New York	NY	10018	
STERLING FACTORS	PO BOX 742 Wildtwir 5til	MIDTOWN STATION	NEW YORK	NY	10018	

Name	Address 1	Address 2	City	State	PostalCode	Country
STERLING PONDS, LLC	C/O GALLO COMPANIES	6303 26 Mile Rd,	Washington Township	MI	48094	
STERLING PONDS, LLC	C/O GALLO COMPANIES	6303 26 MILE RD.	WASHINGTON TOWNSHIP	MI	48094	
STERNIS, DELFINE	REDACT		REDACT	REDACT	REDACT	
STEVEN E. CROOK & ASSOCIATES	P O BOX 50028		NASHVILLE	TN	37205	
STEVEN E. CROOKS & ASSOC.	P.O. BOX 50028		NASHVILLE	TN	37205	
STEWART, ANDREA	REDACT		REDACT	REDACT	REDACT	
STEWART, ANGELA	REDACT		REDACT	REDACT	REDACT	
STEWART, CHARMAINE	REDACT		REDACT	REDACT	REDACT	
STIGLER, ASHLEY	REDACT		REDACT	REDACT	REDACT	
STILES, LAFREDA	REDACT	REDACT	REDACT	REDACT	REDACT	
STILES, TRISCIA	REDACT	REDACT	REDACT	REDACT	REDACT	
STINSON, DI'ANE	REDACT		REDACT	REDACT	REDACT	
STOCKWELL GREETING	7115 W N AVE., #190		AOK PARK	IL	60302	
STOKES & CLINTON, PC	P.O. BOX 991801		MOBILE	AL	36691	
STOKES, JEANETTE	REDACT		REDACT	REDACT	REDACT	
STOKES, QUEATIN	REDACT		REDACT	REDACT	REDACT	
STONE MOUNTAIN	10 W. 33RD STREET	SUITE 728	NEW YORK	NY	10001	
STONE, TIFFANY	REDACT		REDACT	REDACT	REDACT	
	C/O FOREST CITY COMMERCIAL					
STONECREST MALL LLC	MGMT INC	PO BOX 72219	CLEVELAND	ОН	44192-0219	
STONEMONT PARTNERSHIP	1100 SPRING STREET N.W.	SUITE 550	ATLANTA	GA	30309-2848	
STONY ISLAND, LLC	Principal Real Estate Investors, LLC	801 Grand Avenue	Des Moines	IA	50392-1370	
STRAWDER, SHERRY	REDACT	REDACT	REDACT	REDACT	REDACT	
STREAMLINE NATIONWIDE TAX						
SOLUTIONS	1600 REDBUD BLVD.	SUITE 301	MCKINNEY	TX	75069	
STREET DENIM	91-15 139TH ST		JAMAICA	NY	11435	
STREETER, SWAYNE	REDACT		REDACT	REDACT	REDACT	
STRICKLEN, CHARLLOTE	REDACT		REDACT	REDACT	REDACT	
STROMAN, MARK	REDACT		REDACT	REDACT	REDACT	
STROMAN, MESHADAE	REDACT		REDACT	REDACT	REDACT	
STROMAN, TONY	REDACT		REDACT	REDACT	REDACT	
STROUD, MONICA	REDACT		REDACT	REDACT	REDACT	
STRUGGS, BREANNA	REDACT		REDACT	REDACT	REDACT	
STUDIO 8	214 W 39TH STREET	SUITE 1000	NEW YORK	NY	10018	
STUDIO IMPORTS	2252 HAYES STREET		HOLLYWOOD	FL	33020	
STUDIO NAZAR	499 7TH AVENUE		NEW YORK	NY	10018	
STURDIVANT, BIONCA	REDACT	REDACT	REDACT	REDACT	REDACT	

Name	Address 1	Address 2	City	State	PostalCode	Country
STURDIVANT, LAWASTANA	REDACT		REDACT	REDACT	REDACT	
STYLES YOU LOVE	1007 W GROVE AVE		ORANGE	CA	92865	
SULLIVAN, TOSHIBA	REDACT		REDACT	REDACT	REDACT	
SUMMER RIO CORPORATION	17448 RAILROAD ST.		CITY OF INDUSTRY	CA	91748	
SUMMIT REALTY GROUP	470 MCCORMICK BLVD.		COLUMBUS	ОН	43213	
SUMNER CO GEN SESSIONS	PO BOX 549		GALLATIN	TN	37066	
SUMTER COUNTY FAMILY COURT	108 N. MAGNOLIA STREET		SUMTER	SC	29150-4900	
SUMTER COUNTY TREASURER	PO BOX 1775		SUMTER	SC	29151-1775	
SUMTER NATIONAL LP	P.O. BOX 1837		COLUMBIA	SC	29202-1837	
SUMTER NATIONAL, LP	C/O CB Richard Ellis, Columbia	P.O.Box 1837	Columbia	SC	29202	
SUMTER SQUARE SOUTH CAROLINA						
LP	C/O CBRE COLUMBIA	PO BOX 1837	COLUMBIA	SC	29202-1837	
SUN LIFE ASSURANCE COMPANY	PO BOX 4655	SORT 2270	CAROL STREAM	IL	60197-4655	
SUN NEON SIGN & ELECTRIC CO.	4 SADDLE LANE		CHERRY HILL	NJ	08002	
SUNFLOWER IMPORTS	412 W PICO BLVD		LOS ANGELES	CA	90015	
SUNNY DAY	1407 BROADWAY, RM 1419		NEW YORK	NY	10018	
SUNRISE, INC	5615 S. BEECH DALY		DEARBORN HEIGHTS	MI	48125	
Sunrise, Inc.	5615 S. Beech Daly		Dearborn Heights	MI	48125	
SUNTRUST BANK	PO Box 305183		Nashville	TN	37230-5183	
SUNTRUST BANK, AUGUSTA	PO Box 305183		Nashville	TN	37230-5183	
SUPER TRADER	389 5TH AVE	6TH FLR, STE 602	NEW YORK	NY	10016	
SUPER, LLC	LEASE ID 1496160	PO BOX 74234	CLEVELAND	ОН	44194-4234	
	DBA CENTRO NP HOLDINGS 12 SPE,					
SUPER, LLC	LLC	PO BOX 74205	CLEVELAND	ОН	44914-4205	
SUPERIOR TRACK INT'L, INC.	1385 BROADWAY, STE.1207		NEW YORK	NY	10018	
SURE-FIT SECURITY	8213 FENTON STREET		SILVER SPRING	MD	20910	
SUREN, MARYBELL	REDACT		REDACT	REDACT	REDACT	
SURGICK, LUTHER	REDACT		REDACT	REDACT	REDACT	
SURITA, LILJUANA	REDACT		REDACT	REDACT	REDACT	
SURITA, MARCUS	REDACT		REDACT	REDACT	REDACT	
SURREY FONDREN INVESTORS, LLC	C/O CITIBANK	P O BOX 10136	Uniondale	NY	11555	
SURREY FONDREN INVESTORS, LLC	C/O CITIBANK	P.O. BOX 10136	UNIONDALE	NY	11555	
SUSO 2 MITCHELLVILLE LP	C/O SLATE US OPPORTUNITY	HOLDING LP	CLEVELAND	ОН	44194-4498	
SUSTAINABLE SOLUTIONS GROUP	DEPT # 40299	PO BOX 740209	ATLANTA	GA	30374-0209	
SUTTON, TIERSY	REDACT		REDACT	REDACT	REDACT	

Name	Address 1	Address 2	City	State	PostalCode	Country
Swapnil Shah Revocable Trust	2110 N.W. 95th Avenue		Miami	FL	33172	
SWAPNIL SHAH TRUST	C/O SWAPNIL SHAH	2110 NW 95TH AVE	MIAMI	FL	33172	
SWEAT T'S AND MORE INC.	219-1 36TH ST.		BROOKLYN	NY	11232	
	BLUE PARKWAY TOWN					
Swope Community Builders	CENTER/Store No161	Attn: Chuck Gatson	Kansas City	МО	84130	
SYKES, JZATIA	REDACT		REDACT	REDACT	REDACT	
SYKES, REBA	REDACT		REDACT	REDACT	REDACT	
SYLVIA FORD BROWN	CHAPTER 13 TRUSTEE	P.O. BOX 1924	MEMPHIS	TN	38101-1924	
SYMETRA LIFE INSURANCE	PO BOX 440		ASHLAND	WI	54806	
SYMPATICO REAL ESTATE, LLC	222 LAKEVIEW AVE	PH5	WEST PALM BEACH	FL	33401	
T & K ELECTRIC	12518 ANMAR DRIVE		CLEVELAND	TX	77328	
T C P INWOOD PARTNERS, LP	C/O TCP REALTY SERVICES, LLC	500 N. AKARD ST.	DALLAS	TX	75201	
TABANI RALEIGH TN, LLC	PO BOX 504156		ST. LOUIS	МО	63150-4156	
TALLADEGA COUNTY	REVENUE COMMISSIONER	P.O. BOX 1119	TALLADEGA	AL	35161	
TALLADEGA DEVELOPMENT LIMITED	C/O BARSTEIN & ASSOCIATES, INC	PO BOX 13096	BIRMINGHAM	AL	35202-3096	
TALLAHATCHIE VALLEY EPA	250 Power Dr.		Batesville	MS	38606	
TALLAHATCHIE VALLEY EPA	PO BOX 513		BATESVILLE	MS	38606-0513	
TALX UC EXPRESS (EQUIFAX)	4076 PAYSPHERE CIRLCE		CHICAGO	IL	60674-4076	
TAMMY GORDON	904 FAYETTE AVE		BROWNSVILLE	PA	15417	
Tangipahoa Parish School System						
Sales Tax Div	PO Box 159		Amitie	LA	70422-0159	
TANIKA HARRELL	1320 8TH PLACE		PLEASANT GROVE	AL	35127	
TARRANT COUNTY	TAX ASSESSOR-COLLECTOR	PO BOX 961018	FT WORTH	TX	76161-0018	
TARRANT, DONIEL	REDACT		REDACT	REDACT	REDACT	
TASHA APPAREL	6901 MCKINLEY AVENUE		LOS ANGELES	CA	90001	
TATANGO, INC.	2211 ELLIOTT AVENUE	SUITE 200	SEATTLE	WA	98121	
TATE, KENYATTA	REDACT		REDACT	REDACT	REDACT	
TATE, KENYATTA	REDACT		REDACT	REDACT	REDACT	
TAUTGES, JELISHA	REDACT		REDACT	REDACT	REDACT	
TAXBREAK- NEON WORKFORCE						
TECHNOLOGIES	PO BOX 403		GADSDEN	AL	35902	
TAYLOR AIR CONDITIONING & HTG	PO BOX 2		CLEVELAND	MS	38732	
Taylor Associates Limited						
Partnership	TAYLOR COMMONS/Store No170	c/o Zamlas Services Inc.	Johnstown	PA	15901	
TAYLOR ASSOCIATES LIMITED		,				
PARTNERSHIP	Zamias Services, Inc.	300 Market Street	Johnstown	PA	15901	

Name	Address 1	Address 2	City	State	PostalCode	Country
TAYLOR ASSOCIATES, LP	C/O ZAMIAS SERVICES, INC.	300 MARKET ST	JOHNSTOWN	PA	15901	
TAYLOR WHITTEN	113 MANGROVE DRIVE		ALABASTER	AL	35007	
TAYLOR, AMELIA	REDACT	REDACT	REDACT	REDACT	REDACT	
TAYLOR, DELORES	REDACT		REDACT	REDACT	REDACT	
TAYLOR, DERRICK	REDACT		REDACT	REDACT	REDACT	
TAYLOR, DUANE	REDACT	REDACT	REDACT	REDACT	REDACT	
TAYLOR, EARNESTINE	REDACT		REDACT	REDACT	REDACT	
TAYLOR, HEATHER	REDACT		REDACT	REDACT	REDACT	
TAYLOR, HENRY	REDACT		REDACT	REDACT	REDACT	
TAYLOR, JR., DUANE	REDACT		REDACT	REDACT	REDACT	
TAYLOR, LAKEISHA	REDACT	REDACT	REDACT	REDACT	REDACT	
TAYLOR, SUSAN	REDACT		REDACT	REDACT	REDACT	
TAYLOR, TORRY	REDACT		REDACT	REDACT	REDACT	
TCCI BROAD STREET, LLC	C/O CASTO	P O BOX 1450	Columbus	ОН	43216	
TCCI BROAD STREET, LLC	C/O CASTO	PO BOX 1450	COLUMBUS	ОН	43216	
TCF NATIONAL BANK	800 Burr Ridge Pkway		Burr Ridge	IL	60527	
TCP Inwood Partners, L.P	INWOOD CENTRAL/Store No15	c/o TCP Realty Services, LLC	Dallas	TX	75201	
TCP INWOOD PARTNERS, LP	8200 Brookriver Dr. (2002 Lease) 1845 Woodall Rodgers Freeway	Suite N-610	Dallas	тх	75247	
TCP INWOOD PARTNERS, LP	(9/8/06 ltr)	Suite 1030	Houston	TX	75201	
TDH - TENNESSEE DOOR &						
HARDWARE, INC	3103 NORBROOK DRIVE		MEMPHIS	TN	38116	
TDX TECH	PO BOX 1575	#177	MINNEAPOLIS	MN	55480-1575	
TEAGUE, ASHLEY	REDACT	REDACT	REDACT	REDACT	REDACT	
TECHERA, SARENITY	REDACT		REDACT	REDACT	REDACT	
Technology Insurance Company						
(TWC3475001)	59 Maiden Lane		New York	NY	10038	
TECO - TAMPA ELECTRIC	2200 E Sligh Ave		Tampa	FL	33610	
TECO - TAMPA ELECTRIC	PO BOX 31318		TAMPA	FL	33631-3318	
TEJERA, KIARA	REDACT		REDACT	REDACT	REDACT	
TEKLINKS, INC	PO BOX 830674	MSC # 703	BIRMINGHAM	AL	35283	
TELECHECK SERVICES, INC	PO BOX 60028		CITY OF INDUSTRY	CA	91716-0028	
TELERIK, INC.	201 JONES ROAD	1ST FLOOR	WALTHAM	MA	02451	
TEL-EX SHOPPING CENTER, LLC	C/O FOX REALTY, LLC	2150-B FRANKLIN ROAD	BLOOMFIELD	MI	48302	
TELLIS, MELISHA	REDACT		REDACT	REDACT	REDACT	
TELLIS, TEQUEALA	REDACT		REDACT	REDACT	REDACT	
TELLMAN, KATHRYN	REDACT		REDACT	REDACT	REDACT	

Name	Address 1	Address 2	City	State	PostalCode	Country
TEMPLE, EBONI	REDACT		REDACT	REDACT	REDACT	
TEMPLE, ROBERTA	REDACT		REDACT	REDACT	REDACT	
		c/o HAS Commercial Realty				
Tenalok Partners, LTD	FRAYSER PLAZA/Store No144	Services, LLC	Houston	TX	77024	
	C/O HSA COMMERCIAL REALTY					
TENALOK PARTNERS, LTD	SERVICES LLC	701 N POST OAK RD, SUITE 515	HOUSTON	TX	77024	
	c/o HSA Commercial Realty					
TENALOK PARTNERS, LTD.	Services, LLC	701 N. Post Oak Road	Houston	TX	77024	
TENNER, CHRISTINA	REDACT		REDACT	REDACT	REDACT	
TENNESSE, LAVENNA	REDACT		REDACT	REDACT	REDACT	
	ANDREW JACKSON STATE OFFICE					
TENNESSEE DEPT. OF REVENUE	BL	500 DEADERICK ST.	NASHVILLE	TN	37242-0482	
	500 Deaderick Street, Andrew					
Tennessee Department of Revenue	Jackson Building		Nashville	TN	37242	
TENUTA-NOWACKI, DEBORAH	REDACT		REDACT	REDACT	REDACT	
TERESA ROACH	5040 APPALACHIAN DRIVE		GOODLETSVILLE	TN	37072	
TERRY, ANTWANETTE	REDACT		REDACT	REDACT	REDACT	
TERRY, DAVID	REDACT		REDACT	REDACT	REDACT	
TERRY, TIAH	REDACT		REDACT	REDACT	REDACT	
TESS SPORTSWEAR	9500 RUE MEILLEUR. STE 400		MONTREAL	QUEBEC	H2N 2B7	
TEX, BUFFY	REDACT		REDACT	REDACT	REDACT	
Texas Department of Revenue	Capitol Station	P.O. Box 13528	Austin	TX	78711-3528	
THATCH, SEAN	REDACT		REDACT	REDACT	REDACT	
THATCH, SEAN	REDACT		REDACT	REDACT	REDACT	
THAT'S MY GIRL	1407 BROADWAY		NEW YORK	NY	10018	
THE BIRMINGHAM NEWS	PO BOX 9001017		LOUISVILLE	KY	40290-1017	
THE CENTRE AT FORESTVILLE LLC	3393 DONNELL DR		FORESTVILLE	MD	20747	
THE CHECK CASHING STORE	6340 NW 5TH WAY		FT LAUDERDALE	FL	33309	
THE CIT GROUP/COMMERCIAL SVCS,						
INC.	201 SOUTH TYRON ST, STE 300		CHARLOTTE	NC	28202	
THE CITY OF BOSTON	BOX 55809		BOSTON	MA	02205	
The City of East Point	Attn: Corliss Lawson	2777 East Point Street	East Point	GA	30344	
The City of Ferndale, Michigan	Attn: Marne McGrath	300 East Nine Mile Road	Ferndale	MI	48220	
	Attn: Robert V. Schwerer - City					
The City of Fort Pierce	Attorney	100 N. U.S. Highway 1	Fort Pierce	FL	34950	
The City of Gary, Indiana	Attn: Gregory Thomas	401 Broadway, Suite 101 B	Gary	IN	46402	
The City of Leominster	Attn: Lynn Bouchard, City Clerk	25 West Street	Leominster	MA	01453	
The City of Pontiac	Attn: Sherikia L. Hawkins	47450 Woodward Avenue	Pontiac	MI	48342	

Name	Address 1	Address 2	City	State	PostalCode	Country
THE COMMONWEALTH OF			,			,
MASSACHUSETTS	ONE ASHBURTON PLACE		BOSTON	MA	02108-1512	
THE ERDLAND COMPANY	C/O REGIONAL MGMT INC.	11 E Fayette Street	Baltimore	MD	21202	
THE ERDLAND COMPANY	C/O REGIONAL MGMT INC.	11 E. FAYETTE ST.	BALTIMORE	MD	21202	
THE FRUITPORT PARCEL R DEV., LP	211 NORTH STADIUM	SUITE 201	COLUMBIA	МО	65203	
THE GERBER REALTY COMPANY	1779 Kirby Parkway	Suite 5A	Memphis	TN	38138	
The Gerber Realty Family Company	WHITEHAVEN PLAZA/Store No281	1779 Kirby Parkway, Suite 5A	Memphis	TN	38138	
THE HOFFMAN & HARPST CO., INC.	642 E. WOODRUFF AVENUE		TOLEDO	ОН	43604	
The Horton Group, Inc.	10320 Orland Parkway		Orland Park	IL	60467	
THE ILLUMINATING COMPANY	6896 Miller Rd		Brecksville	ОН	44141	
THE ILLUMINATING COMPANY	P.O. BOX 3687		AKRON	ОН	44309-3687	
THE OVERHEAD DOOR COMPANY OF						
MEMPHIS, INC.	2080 ELVIS PRESLEY BLVD.		MEMPHIS	TN	38106	
	THE SHOPPES OF LIBERTY	c/o Platinum Property				
The Shoppes of Liberty City, LLC	CITY/Store No160	Management, Inc.	Ft. Lauderdale	FL	33319-4970	
THE SHOPPES OF LIBERTY CITY, LLC	11098 Biscayne Boulevard	Suite 103	Miami	FL	33161	
THF FRUITPORT PARCEL R. DEV., LP	211 North Stadium	Suite 201	Columbia	MI	65203	
THIRKILL, THOMAS	REDACT		REDACT	REDACT	REDACT	
THOMAS COUNTY TAX						
COMMISSIONER	P.O. BOX 2175		THOMASVILLE	GA	31799	
THOMAS, ALADE	REDACT		REDACT	REDACT	REDACT	
THOMAS, ALLEN	REDACT		REDACT	REDACT	REDACT	
THOMAS, BRANDON	REDACT		REDACT	REDACT	REDACT	
THOMAS, BRENDA	REDACT		REDACT	REDACT	REDACT	
THOMAS, DEBRA	REDACT		REDACT	REDACT	REDACT	
THOMAS, ELOUISE	REDACT	REDACT	REDACT	REDACT	REDACT	
THOMAS, IRIS	REDACT		REDACT	REDACT	REDACT	
THOMAS, IRIS	REDACT		REDACT	REDACT	REDACT	
THOMAS, KIA	REDACT		REDACT	REDACT	REDACT	
THOMAS, LAKEISHA	REDACT		REDACT	REDACT	REDACT	
THOMAS, LATOYA	REDACT		REDACT	REDACT	REDACT	
THOMAS, LEON	REDACT		REDACT	REDACT	REDACT	
THOMAS, MARCUS	REDACT	REDACT	REDACT	REDACT	REDACT	

Name	Address 1	Address 2	City	State	PostalCode	Country
THOMAS, MARVIE	REDACT		REDACT	REDACT	REDACT	
THOMAS, MURDIES	REDACT		REDACT	REDACT	REDACT	
THOMAS, MYRTIECE	REDACT		REDACT	REDACT	REDACT	
THOMAS, PRISCILLA	REDACT		REDACT	REDACT	REDACT	
THOMAS, ROCHELLE	REDACT		REDACT	REDACT	REDACT	
THOMAS, RODMAN	REDACT		REDACT	REDACT	REDACT	
THOMAS, SANTRICA	REDACT		REDACT	REDACT	REDACT	
THOMAS, SANTRICA	REDACT		REDACT	REDACT	REDACT	
THOMAS, SHAMARDE	REDACT		REDACT	REDACT	REDACT	
THOMAS, SHANTIA	REDACT		REDACT	REDACT	REDACT	
THOMAS, SHONA	REDACT		REDACT	REDACT	REDACT	
THOMAS, SIERRA	REDACT		REDACT	REDACT	REDACT	
THOMAS, TEISHA	REDACT		REDACT	REDACT	REDACT	
THOMAS, VICTORIA	REDACT		REDACT	REDACT	REDACT	
THOMPSON, ANGEL	REDACT	REDACT	REDACT	REDACT	REDACT	
THOMPSON, DARRYL	REDACT		REDACT	REDACT	REDACT	
THOMPSON, KEYOSHEA	REDACT	REDACT	REDACT	REDACT	REDACT	
THOMPSON, PARIS	REDACT		REDACT	REDACT	REDACT	
THOMPSON, REBECCA	REDACT		REDACT	REDACT	REDACT	
THOMPSON, SHANTELL	REDACT		REDACT	REDACT	REDACT	
THOMPSON, TORONADO	REDACT		REDACT	REDACT	REDACT	
THORNTON, EBONY	REDACT		REDACT	REDACT	REDACT	
THORPE SHEET METAL, INC.	P.O. BOX 1447		WASHINGTON	MS	39190	
THORPE, VALENCIA	REDACT		REDACT	REDACT	REDACT	
THORPE-SETTLES, VALENCIA	REDACT		REDACT	REDACT	REDACT	
THURMAN, SIERRA	REDACT		REDACT	REDACT	REDACT	
THURMOND, BOOKER T.	REDACT		REDACT	REDACT	REDACT	
TILLMAN, KIM	REDACT		REDACT	REDACT	REDACT	
TIME WORLD WATHCES	2600 NORTHAVEN RD, STE 100		DALLAS	TX	75229	
TIMING FASHION	2809 S SANTA FE AVE		VERNON	CA	90058	
TIMMONS, MARKETTA	REDACT		REDACT	REDACT	REDACT	
TIMMONS, MYISHA	REDACT		REDACT	REDACT	REDACT	
TIMS, DEBORAH	REDACT		REDACT	REDACT	REDACT	
TINDLEY, SHAKIR	REDACT		REDACT	REDACT	REDACT	
TINLEY PARK ASSOCIATES	c/o Spatz Centers	14 N. Peoria Street	Chicago	IL	60607	
TIPTON, HEATHER	REDACT		REDACT	REDACT	REDACT	
TITANIUM FABRICS	6001 W SLAUSON		COMMERCE	CA	90040	

Name	Address 1	Address 2	City	State	PostalCode	Country
TITANIUM MARKETING INC.	MURRAY HILL STATION	P.O. BOX 1851	NEW YORK	NY	10156-1851	
TLE DBA NATIONAL PAYDAY LOAN	PO BOX 332		TALMADGE	CA	95481	
TN CHILD SUPPORT	PO BOX 305200		NASHVILLE	TN		
TN DEPT OF EMPLOYMENT						
SECURITY	EMPLOYER ACCOUNTS	PO BOX 101	NASHVILLE	TN	37202-0101	
TNCI	PO BOX 981038		BOSTON	MA	02298-1038	
TNCI	2 Charlesgate West		Boston	MA	02215-0000	
TODAY APPAREL	1407 BROADWAY, RM 1107		NEW YORK	NY	10018	
TODD WALL PLUMBING LLC.	30605 WESTWOOD		MADISON HEIGHTS	MI	48071	
TOKTAM REZAEI	11132 76TH AVENUE	APT. 5I	FOREST HILLS	NY	11375	
TOLBERT, DEANTINIQUE	REDACT		REDACT	REDACT	REDACT	
TOLEDO EDISON	76 S Main St		Akron	ОН	44308	
TOLEDO EDISON	PO BOX 3687		AKRON	ОН	44309-3687	
TOM DREXLER PLUMBING	2232 BARDSTOWN RD		LOUISVILLE	KY	40205	
TOM VAUGHN	STANDING TRUSTEE	P.O. BOX 588	MEMPHIS	TN	38101-0588	
TOMBAUGH, BRITTANY	REDACT		REDACT	REDACT	REDACT	
TOMBAUGH, KELLY	REDACT		REDACT	REDACT	REDACT	
TOMBAUGH, SUSAN	REDACT		REDACT	REDACT	REDACT	
TOMEO, DANA	REDACT		REDACT	REDACT	REDACT	
TOMPKINS-CLARK, YVONNE	REDACT		REDACT	REDACT	REDACT	
TOOMBS, ZENOVIA	REDACT		REDACT	REDACT	REDACT	
TOP FASHION OF NY	1407 BROADWAY, ROOM 2112		NEW YORK	NY	10018	
TOP THIS	511 E 87TH PLACE		LOS ANGELES	CA	90003	
TORAN, TIA	REDACT		REDACT	REDACT	REDACT	
TORRES PASTOR, ONYXOMAR	REDACT		REDACT	REDACT	REDACT	
TORRES, KARINA	REDACT		REDACT	REDACT	REDACT	
TORRES, LAUBRITA	REDACT		REDACT	REDACT	REDACT	
TORRES, MARIA	REDACT		REDACT	REDACT	REDACT	
TORRES, OMAYRA	REDACT	REDACT	REDACT	REDACT	REDACT	
TORRES, SONIA	REDACT	REDACT	REDACT	REDACT	REDACT	
Torus National Insurance Company						
(51785C152ASP)	5th Floor	88 Leadenhall Street	London		EC3A 3BP	United Kingdom
TOTAL MEDIA, INC	2 NORTH CORPORATE DRIVE		RIVERDALE	NJ	07457	
TOUCH BLUE	110 EAST 9TH STREET	# B-880	LOS ANGELES	CA	90079	
TOURE-DAVIS, NATHALIE	REDACT		REDACT	REDACT	REDACT	
TOWE, JAMEL	REDACT	REDACT	REDACT	REDACT	REDACT	
TOWER IMPROVEMENTS, LLC	C/O DLC MANAGEMENT CORP	PO BOX 5122	WHITEPLAINS	NY	10602	

Name	Address 1	Address 2	City	State	PostalCode	Country
TOWER PLAZA ASSOCIATES, LLC	1 BELMONT AVENUE	GSB BLDG, SUITE 503	BALA CYNWYN	PA	19004	
TOWN CENTERS LIMITED						
PARTNERSHIP	c/o Mansfield Square Mgmt. Corp.	910 East Broad Street	Columbus	ОН	43205	
TOWN CENTERS, LP	4270 MORSE ROAD		COLUMBUS	ОН	43230	
TOWN CLERK TOWN OF						
MANCHESTER	P.O. BOX 191		MANCHESTER	СТ	06045-0191	
TOWN OF CHEEKTOWAGA	275 ALEXANDER AVENUE		CHEEKTOWAGA	NY	14211	
TOWN OF COLLIERVILLE	Lynn Carmack	500 Poplar View Parkway	Collierville	TN	38017	
TOWN OF COLLIERVILLE	500 POPLAR VIEW PARKWAY	·	COLLIERVILLE	TN	38017	
TOWN OF CUTLER BAY	10720 CARIBBEAN BLVD		CUTLER BAY	FL	33189	
Town of Lantana	Attn: Deborah Manzo	500 Greynolds Circle	Lantana	FL	33462	
TOWN OF LANTANA	500 GREYNOLDS CIRCLE		LANTANA	FL	33462	
TOWN OF LANTANA	318 S. DIXIE HIGHWAY		LANTANA	FL	33462	
TOWN OF LANTANA	500 GREYNOLDS CIRCLE		LANTANA	FL	33462-4544	
		c/o Corbett, White, Davis & Ashton,				
Town of MANGONIA PARK	ATTN: KEITH W. DAVIS, ESQ.	P.A.	Lantana	FL	33462	
TOWN OF MANGONIA PARK	1755 EAST TIFFANY DR.		MANGONIA PARK	FL	33407	
TOWN OF MANGONIA PARK-						
UTILITIES	1755 E TIFFANY DRIVE		MANGONIA PARK	FL	33407	
		C/O Jordan Price Wall Gray Jones &				
Town of Morrisville	Attn: Frank Gray, Town Attorney	Carlton, PLLC	Raleigh	NC	27605	
TOWN OF SALINA	DEPT OF PLANNING & DEV	201 SCHOOL ROAD	LIVERPOOL	NY	13088	
Town of Smyrna	Attn: Jeff Peach	315 South Lowry Street,	Smyrna	TN	37167	
TOWN OF SMYRNA	BUSINESS TAX	315 SOUTH LOWRY STREET	SMYRNA	TN	37167	
TOWN OF SPEEDWAY	ATTN: ALARM PERMIT	1450 N. LYNHURST DRIVE	SPEEDWAY	IN	46224	
TOWN OF WEST SPRINGFIELD	TOWN CLERK'S OFFICE	26 CENTRAL STREET, SUITE 8	WEST SPRINGFIELD	MA	01089	
TOWNE SOUTH REALTY LLC	C/O TALCOR	1018 Thomasville Road	Tallahassee	FL	32303	
TOWNE SOUTH REALTY, LLC	C/O TALCOR	1018 THOMASVILLE ROAD	TALLAHASSEE	FL	32303	
TOWNSEND, CYNTHIA	REDACT		REDACT	REDACT	REDACT	
TOWNSEND, KYLE	REDACT		REDACT	REDACT	REDACT	
TOYOTA MATERIALS HANDLING	PO BOX 100575		IRONDALE	AL	35210	
TRACEY COLQUITT	3109 RAINBOW FOREST CIRCLE	APT.I	DECATUR	GA	30034	
TRACY DAVIS	412 GARDEN HOMES CIRCLE		MONTGOMERY	AL	36116	
TRADEWINDS CLOTHING INC.	4301 GLENWOOD RD		BROOKLYN	NY	11210	
TRAMMELL, MARTINA	REDACT	REDACT	REDACT	REDACT	REDACT	
TRAMMER, SHREE	REDACT		REDACT	REDACT	REDACT	
•						
TRANSPORTATION IMPACT BILLING	10000 COLLEGE BLVD.	STE 235	OVERLAND PARK	KS	66210	

Name	Address 1	Address 2	City	State	PostalCode	Country
						,
TRAVIS A. HULSEY, DIRECTOR	DEPT OF REVENUE / FLEET STAFF	PO BOX 11088	BIRMINGHAM	AL	35202-1088	
TRAVIS, ELYSE	REDACT	REDACT	REDACT	REDACT	REDACT	
TRAVIS, TAMYIA	REDACT		REDACT	REDACT	REDACT	
TRC STAFFING SERVICES	P.O. BOX 468389		ATLANTA	GA	31146-8389	
TREASURER- CITY OF SAGINAW	1315 S. WASHINGTON		SAGINAW	MI	48601	
	ACCTG. OFFICE - DEPT. OF LABOR &					
TREASURER OF VIRGINIA	IND.	MAIN STREET CENTRE	RICHMOND	VA	23219-2430	
TREASURER OF VIRGINIA						
(MERRIFIELD)	STATE CORPORATION COMMISSION	P.O. BOX 7613	MERRIFIELD	VA	22116-7613	
TREASURER, STATE OF NEW JERSEY	NJ DEPARTMENT OF TREASURY	P.O. BOX 308	TRENTON	NJ	08646-0308	
TRENDSET ORIGINALS, LLC	1407 BROADWAY	5TH FLOOR	NEW YORK	NY	10018	
TRENDY ZONE	237 W 31ST ST	SHITEOOK	LOS ANGELES	CA	30007	
TREY CHILDS' ELECTRICAL SERVICES,	1889 LAKE WASHINGTON ROAD		LOS ANGLES	CA	30007	
INC	EAST		HOLLANDALE	MS	38748	
TRI CITY HOME SERVICES, LLC	2607 WOODRUFF ROAD	SUITE E 415	SIMPSONVILLE	SC	29681	
TRI CITY NATIONAL BANK	PO Box 44017		West Allis	WI	53214	
TRI-COASTAL	P.O.BOX 2348	GRAND CENTRAL STATION	NEW YORK CITY	NY	10164-0442	
TRILLIUM FACILITY SOLUTIONS	P.O. BOX 98		NEW BALTIMORE	MI	48047	
TRINITY CONTRACTORS	561 SIMMONS DRIVE		TRUSSVILLE	AL	35173	
TRIPLETT, TIFFANY	REDACT		REDACT	REDACT	REDACT	
TRIPP, ADDIE	REDACT		REDACT	REDACT	REDACT	
TRISCIA STILES	1365 ARMAND DR		MEMPHIS	TN	38103	
TRI-STATE AUTOMATIC DOORS, INC	4022 HWY 49 SOUTH		FLORENCE	MS	39073	
TRIUMPH HOSIERY CORP.	2750 N. 29TH AVENUE	SUITE 202	HOLLYWOOD	FL	33020	
TROUP COUNTY	TAX COMMISSIONER	100 RIDLEY AVENUE	LAGRANGE	GA	30240-2724	
TRUE LOVE ACCESSORIES	10 W 33RD, STE 715		NEW YORK	NY	10001	
TRUPIANO, VICTORIA	REDACT	REDACT	REDACT	REDACT	REDACT	
TRUSS, DESHUNTA	REDACT	REDACT	REDACT	REDACT	REDACT	
TRUSTMARK	PO Box 291		Jackson	MS	39205	
TRUSTMARK NATIONAL BANK	PO Box 291		Jackson	MS	39205	
TUCKER, BILAL	REDACT	REDACT	REDACT	REDACT	REDACT	
TUCKER, BRIAN	REDACT		REDACT	REDACT	REDACT	
TUCKER, JOSHALYN	REDACT		REDACT	REDACT	REDACT	
TUCKER, MISTY	REDACT		REDACT	REDACT	REDACT	
TUCKER, ZANE	REDACT		REDACT	REDACT	REDACT	

Page 161 of 176

Address 1	Address 2	City	State	PostalCode	Country
500 WEST GRANT ROAD		TUSCON	AZ	85705	
578 MAIN STREET		TULLYTOWN	PA	19007	
REDACT		REDACT	REDACT	REDACT	
REDACT		REDACT	REDACT	REDACT	
270 W 38TH ST, 19TH FLR		NEW YORK	NY	10018	
REDACT		REDACT	REDACT	REDACT	
REDACT	REDACT	REDACT	REDACT	REDACT	
REDACT		REDACT	REDACT	REDACT	
REDACT		REDACT	REDACT	REDACT	
REDACT		REDACT	REDACT	REDACT	
ROOM 124, COURTHOUSE	714 GREENSBORO AVE	TUSCALOOSA	AL	35401-1891	
PO BOX 20738		TUSCALOOSA	AL	35402	
P.O. BOX 1612		DES MOINES	IA	50306-1612	
1340 DISCOVERY INDUSTRIAL					
COURT		MABLETON	GA	30126	
26 WEST 36TH STREET		NEW YORK	NY	10018	
P.O. BOX 659791		SAN ANTONIO	TX	78265-9791	
REDACT		REDACT	REDACT	REDACT	
REDACT		REDACT	REDACT	REDACT	
REDACT		REDACT	REDACT	REDACT	
REDACT	REDACT	REDACT	REDACT	REDACT	
1310 Madrid Street		Marshall	MN	56258	
1310 Madrid Street		Marshall	MN	56258	
90 MYRTLE STREET		CRANFORD	NJ	07016-3235	
NATIONAL PAYMENT CENTER	PO BOX 105081	ATLANTA	GA	30348-5081	
NCIF					1
				_	
	351 24TH ST N				
					1
		-			1
	Saite 300				+
	500 WEST GRANT ROAD 578 MAIN STREET REDACT ROOM 124, COURTHOUSE PO BOX 20738 P.O. BOX 1612 1340 DISCOVERY INDUSTRIAL COURT 26 WEST 36TH STREET P.O. BOX 659791 REDACT REDA	500 WEST GRANT ROAD 578 MAIN STREET REDACT REDACT 270 W 38TH ST, 19TH FLR REDACT REDACT REDACT REDACT REDACT REDACT REDACT REDACT REDACT ROOM 124, COURTHOUSE 714 GREENSBORO AVE PO BOX 20738 P.O. BOX 1612 1340 DISCOVERY INDUSTRIAL COURT 26 WEST 36TH STREET P.O. BOX 659791 REDACT 3110 Madrid Street PO BOX 105081 NCIF PO BOX 790363 2403 W. MAIN STREET 1423 S MAPLE AVE BUSINESS MAIL ENTRY 351 24TH ST N Attn: Dustin P. Branch, Esq. Katten Muchin Rosenman LLP 2525 North 12th Street Suite 360	500 WEST GRANT ROAD 578 MAIN STREET TULLYTOWN REDACT REDAC	500 WEST GRANT ROAD 578 MAIN STREET TULLYTOWN PA REDACT RE	SOO WEST GRANT ROAD

Name	Address 1	Address 2	City	State	PostalCode	Country
			,			
UGI UTILITIES, INC.	2525 North 12th Street	Suite 360	Reading	PA	19605	
UGI UTILITIES, INC.	P.O. BOX 15523		WILMINGTON	DE	19886-5523	
UNCENSORED	1407 BROADWAY	SUITE 919	NEW YORK	NY	10018	
UNDERWOOD, KIMBERLY	REDACT		REDACT	REDACT	REDACT	
UNEEDA GLASS COMPANY	PO BOX 1023		COLUMBUS	GA	31902	
UNIFIED GOVERNMENT TREASURY						
WYANDOTTE CO.	PO BOX 175013		KANSAS CITY	KS	66117-5013	
UNIFIED GOVT-KANSAS CITY,KS.(LIC)	UNIFIED GOV'T-BUSINESS LIC	4601 STATE AVE., STE 87	KANSAS CITY	KS	66102	
UNIFOUR FIRE & SAFETY	P.O. BOX 9489		HICKORY	NC	28603	
UNISHIPPERS BHM	3337 NORTH HULLEN STREET	SUITE 300	METAIRIE	LA	70002	
UNITED CARPET OF ILLINOIS	1375 EAST IRVING PARK ROAD		ITASCA	IL	60143	
UNITED CARPET COMPANY, INC.	5434 EAGLE INDUSTRIAL COURT		HAZELWOOD	MO	63042	
UNITED GLASS, LLC	2252 GREENWOOD STREET	SUITE B	KENNER	LA	70062	
UNITED MISSISSIPPI BANK	PO Box 670		Natchez	MS	39121-0670	
UNITED PLYWOODS & LUMBER, INC	PO BOX 1088		BIRMINGHAM	AL	35201	
UNITED STATES TREASURY	DEPT. OF TRESURY	INTERNAL REVENUE SERVICE	CINCINNATI	ОН	45999-0009	
UNITED STATES TREASURY	ACS SUPPORT - STOP 813G	PO BOX 145566	CINCINNATI	ОН	45250-5566	
		801 TOM MARTIN DRIVE, RM 137C				
UNITED STATES TREASURY	INTERNAL REVENUE SERVICE	3	BIRMINGHAM	AL	35211	
UNIVERSAL FASHION OUTLET	1500 MAIN STREET		LOS ANGELES	CA	90015	
UPPERMAN, ALJEANAE	REDACT		REDACT	REDACT	REDACT	
UPS (PA) JUDY	PO BOX 7247-0244		PHILADELPHIA	PA	19170-0001	
UPS (TONY)	P.O. BOX 7247-0244		PHILADELPHIA	PA	19170-0001	
UPS FREIGHT	PO BOX 533238		CHARLOTTE	NC	28290-3238	
UPTON, BRITTNEY	REDACT		REDACT	REDACT	REDACT	
UPTON, SHANNON	REDACT	REDACT	REDACT	REDACT	REDACT	
UPTON, SHANNON	REDACT		REDACT	REDACT	REDACT	
URBAN PARTNERSHIP BANK	PO Box 19260		Chicago	IL	60619-0260	
Urban Textile, Inc.	3700 S. Santa Fe Avenue		Vernon	CA	90058	
US BANK	PO Box 1800		Saint Paul	MN	55101-0800	
US Bank Equipment Finance	1310 Madrid Street	Suite 101	Marshall	MN	56258	
US BANK EQUIPMENT FINANCE	PO BOX 790448		ST LOUIS	MO	63179-0448	
US DEPT OF TREASURY	DEBT MGT SVCS	P.O. BOX 979101	ST. LOUIS	МО	63197-9000	
US POSTAL SERVICE	Business Mail Entry	351 24th St. N	Birmingham	AL	35201-0395	

Name	Address 1	Address 2	City	State	PostalCode	Country
US TREASURY INTERNAL REV SERV	801 TOM MARTIN DRIVE	RM 137-C3	BIRMINGHAM	AL	35211-6426	
USA SAFE & LOCK	26803 HARPER AVENUE		ST. CLAIR SHORES	MI	48081	
		1000 URBAN CENTER DRIVE, SUITE				
USI	ATTN: TANYA COUNTS	400	BIRMINGHAM	AL	35242	
	C/O THE DESCO GRP, INC. ATTN:					
USR-DESCO CITY PLAZA, LLC	PROP MGR	25 N. Brentwood Blvd.	St. Louis	MO	63105	
USR-DESCO CITY PLAZA, LLC	C/O THE DESCO GRP, INC	ATTN: PROP MGR	ST. LOUIS	MO	63105	
UTILITY BILLING SERVICES	221 E Capitol Ave		Little Rock	AR	72202	
UTILITY BILLING SERVICES	PO BOX 8100		LITTLE ROCK	AR	72203-8100	
UTILITY OPERATIONS DIVISION	CITY OF MONROE, PO BOX 1743		MONROE	LA	71210-1743	
VALENTINE, DESHONNA	REDACT		REDACT	REDACT	REDACT	
VALENZUELA, JULIA	REDACT		REDACT	REDACT	REDACT	
VALIA SPORTSWEAR LTD.	9500 MEILLEUR	SUITE 400	MONTREAL	QC	H2N 2B7	CANADA
VALLERY, A'DASHA	REDACT		REDACT	REDACT	REDACT	
VALLEY PLUMBING COMPANY	P.O. BOX 1368		PHENIX CITY	AL	36868	
VALUTRACK TECHNOLOGY						
SOLUTIONS	133 MAIN STREET SUITE 300		MEDWAY	MA	02053	
VANBUREN, NAPORCSHA	REDACT		REDACT	REDACT	REDACT	
VANESSA DOUGHTY	10 ASHBURY WOODS DRIVE	APT#312	HUNTSVILLE	AL	35824	
VANESSA JOHNSON	2200 E. ROSE AVE. #9		DES MOINES	IA	50320	
VANZANT, KARISHA	REDACT		REDACT	REDACT	REDACT	
VARGAS, ROCIO	REDACT		REDACT	REDACT	REDACT	
VARGAS, TIFFANY	REDACT		REDACT	REDACT	REDACT	
VAST, MALCOLM	REDACT	REDACT	REDACT	REDACT	REDACT	
VAULT SPORTSWEAR, INC.	1407 BROADWAY, STE. 1810		NEW YORK	NY	10018	
VAZQUEZ, FABIOLA	REDACT		REDACT	REDACT	REDACT	
VCG SOUTHLAKE MALL, LLC	1000 Southlake Mall		Morrow	GA	30260	
VCG SOUTHLAKE MALL, LLC	1000 Southlake Cir		Morrow	GA	30260	
VCG SOUTHLAKE MALL, LLC	VCG SOUTHLAKE MALL, LLC	P.O. Box 538624	ATLANTA	GA	30353-8624	
VCG-SOUTHLAKE MALL, LLC	PO BOX 538624		ATLANTA	GA	30353-8624	
VEAL, DIANN	REDACT		REDACT	REDACT	REDACT	
VECTREN	211 Northwest Riverside Drive		Evansville	IN	47708	
VECTREN	P.O. BOX 6262		INDIANAPOLIS	IN	46206-6262	
VECTREN ENERGY DELIVERY	P.O. BOX 6262		INDIANAPOLIS	IN	46206	
VECTREN ENERGY DELIVERY	211 Northwest Riverside Drive		Evansville	IN	47708	
VECTREN ENERGY DELIVERY	PO BOX 6262		INDIANAPOLIS	IN	46206-6262	
VEGA, XIOMARA	REDACT	REDACT	REDACT	REDACT	REDACT	

Page 164 of 176

Name	Address 1	Address 2	City	State	PostalCode	Country
VELASQUEZ, JOEL	REDACT	REDACT	REDACT	REDACT	REDACT	
VELAZQUEZ, REINALDO	REDACT		REDACT	REDACT	REDACT	
VENTRUM-COVINGTON, INC.	CB Richard Ellis Asset Svcs.	3230 Peachtree Corners Circle	Norcross	GA	30092	
	C/O PNC BANK ATTN: NINA					
VENTRUM-COVINGTON, INC.	SHIROKOVA	ACCT 5325217737	ATLANTA	GA	30305	
VERDUE CROSSROADS, LLC	P.O. Box 980714		Houston	TX	77098	
VERDUE CROSSROADS, LLC	22329 GOSLING		SPRING	TX	77389	
Verizon Wireless	One Verizon Place		Alpharetta	GA	30004	
Verizon Wireless	P.O. BOX 660108		Dallas	TX	75266-0108	
VERIZON WIRELESS (judy)	PO BOX 660108		DALLAS	TX	75266-0108	
VIA, CYNTHIA	REDACT		REDACT	REDACT	REDACT	
VIBES SPORTSWEAR/TEE DEES INC.	1407 BROADWAY	STE 1419	NEW YORK	NY	10018	
VICK, BREYONNA	REDACT		REDACT	REDACT	REDACT	
VICKERS, ALEX	REDACT		REDACT	REDACT	REDACT	
VICKERS, HERMAN	REDACT		REDACT	REDACT	REDACT	
VICKSBURG INCOM	C/O THE VICKSBURG MALL	3505 Pemberton Square Blvd.	VICKSBURG	MS	39180	
		c/o CBL & Associates				
Vicksburg Income Properties, LLC	PEMBERTON SQUARE/Store No16	Manangement, Inc.	Chattanooga	TN	37421-6000	
VICKSBURG INCOME PROPERTIES,						
LLC	C/O THE VICKSBURG MALL	3505 PEMBERTON SQUARE BLVD	VICKSBURG	MS	39180	
VICKSBURG MALL ASSOCIATES, LTD.	CBL & ASSOCIATES MANAGEMENT, INC.	CBL Center	Chattanooga	TN	37421-6000	
VIDAL DAVIS	P.O. BOX 614		NATCHEZ	MS	39121	
VILLAGE HILLS SHOPPING CENTER,						
LLC	C/O URMISH PATEL	PO BOX 43	PELHAM	AL	35124	
VILLAGE JEANS	2421 MCDONALD AVENUE		BROOKLYN	NY	11223	
VILLAGE OF BERKELEY	5819 ELECTRIC AVE.		BERKELEY	IL	60163	
VILLAGE OF CAHOKIA	VILLAGE CLERK	103 MAIN STREET	CAHOKIA	IL	62206	
VILLAGE OF CAHOKIA- SEWER DEPT	201 WEST 4TH STREET		CAHOKIA	IL	62206	
Village of Calumet Park	The Honorable Mayor Denson	12409 South Throop,	Calumet Park	IL	60827	
VILLAGE OF CALUMET PARK	12409 SOUTH TROUP		CALUMET PARK	IL	60827	
VILLAGE OF HAZEL CREST	2903 WEST 175TH STREET		HAZEL CREST	IL	60429-1706	
VILLAGE OF OLYMPIA FIELDS	20040 GOVERNORS HIGHWAY		OLYMPIA FIELDS	IL	60461	
VILLAGE OF TINLEY PARK	16250 S. OAK PARK AVENUE		TINLEY PARK	IL	60477	

Name	Address 1	Address 2	City	State	PostalCode	Country
	C/O RIVERCREST REALTY					
VILLAGE REALTY, LLC	ASSOCIATES, LLC	8816 SIX FORKS ROAD	RALEIGH	NC	27615	
VILLAGE REALTY, LLC	Rivercrest Realty Associates, LLC	Attn:Denise Wagner	Raleigh	NC	27615	
VILLALBA, GENESIS	REDACT		REDACT	REDACT	REDACT	
VILLALBA, GENESIS	REDACT		REDACT	REDACT	REDACT	
VILLALBA, MICHELLE	REDACT	REDACT	REDACT	REDACT	REDACT	
VINZIANT, VERONICA	REDACT		REDACT	REDACT	REDACT	
VIOLA GUNTER	176 LOUIS SCOTT COURT		EASTOVER	SC	29044	
VIP JEANS	1407 BROADWAY, STE 2801		NEW YORK	NY	10018	
VIR INTERNATIONAL	2110 NW 95TH AVE		MIAMI	FL	33172	
VIR International, LLC	2110 N.W. 95th Avenue		Miami	FL	33172	
VIR INTERNATIONAL, LLC	2110 NW 95TH AVENUE		MIAMI	FL	33172	
Virginia Beach City	Attn: Consumer Protection Agency	Mark D. Stiles, City Attorney	Virginia Beach	VA	23456	
VIRGINIA NATURAL GAS	544 S. Independence Blvd.		Virginia Beach	VA	23452	
VIRGINIA NATURAL GAS	PO BOX 5409		CAROL STREAM	IL	60197-5409	
VIRGINIA NATURAL GAS	PO BOX 70840		CHARLOTTE	NC	28272-0840	
VIRGINIA PARK COMMUNITY						
INVESTMENT ASSOC INC	C/O FOURMIDABLE	32500 TELEGRAPH RD.	BINGHAM FARMS	MI	48025	
	Virginia Department of Taxation					
Virginia Department of Revenue	Office of Customer Service	P.O. Box 1115	Richmond	VA	23218-1115	
VISION APPAREL	1407 BROADWAY	LOBBY FLOOR	NEW YORK	NY	10018	
VISION ELECTRICAL CONTRACTORS,						
LLC	59 WEST LINFIELD TRAPPE ROAD		LIMERICK	PA	19468	
VOIVEDICH, DEBORAH	REDACT		REDACT	REDACT	REDACT	
VOLUSIA COUNTY - REVENUE						
DIVISION	TAX PROCESSING CENTER	PO BOX 23237	TAMPA	FL	33623-2237	
VPI	1099 wall street west	STE 391	Lyndhurst	NJ	07071	
VU DAI VAN	290 B SGT PRENTISS DRIVE		NATCHEZ	MS	39120	
WADE, CYNTHIA	REDACT	REDACT	REDACT	REDACT	REDACT	
WADE, JAYLON	REDACT		REDACT	REDACT	REDACT	
WADE, LASHAWN	REDACT		REDACT	REDACT	REDACT	
Wake County	Attn: Consumer Protection Agency	Scott W. Warren - County Attorney	Raleigh	NC	27601	
WAKE COUNTY DEPT OF REVENUE	P.O. BOX 580084		CHARLOTTE	NC	28296-0084	

Name	Address 1	Address 2	City	State	PostalCode	Country
WALDON, RICHARD	REDACT		REDACT	REDACT	REDACT	
WALKER, ANGEL	REDACT		REDACT	REDACT	REDACT	
WALKER, ASHLEY	REDACT		REDACT	REDACT	REDACT	
WALKER, ASHLEY	REDACT		REDACT	REDACT	REDACT	
WALKER, BRANDON	REDACT		REDACT	REDACT	REDACT	
WALKER, BRENDAN	REDACT	REDACT	REDACT	REDACT	REDACT	
WALKER, BRIANNA	REDACT		REDACT	REDACT	REDACT	
WALKER, CHARLIE	REDACT	REDACT	REDACT	REDACT	REDACT	
WALKER, DANAKA	REDACT		REDACT	REDACT	REDACT	
WALKER, DONALD	REDACT		REDACT	REDACT	REDACT	
WALKER, JOHN	REDACT	REDACT	REDACT	REDACT	REDACT	
WALKER, KEN	REDACT		REDACT	REDACT	REDACT	
WALKER, KENNETH	REDACT	REDACT	REDACT	REDACT	REDACT	
WALKER, MARCUS	REDACT		REDACT	REDACT	REDACT	
WALKER, MARCUS	REDACT		REDACT	REDACT	REDACT	
WALKER, MARCUS	REDACT		REDACT	REDACT	REDACT	
WALKER, TONY	REDACT		REDACT	REDACT	REDACT	
WALKER, VANESSA	REDACT		REDACT	REDACT	REDACT	
WALKER, VANESSA	REDACT		REDACT	REDACT	REDACT	
WALKERLY, AUTUMN	REDACT	REDACT	REDACT	REDACT	REDACT	
WALKERLY, BRENDA	REDACT		REDACT	REDACT	REDACT	
WALKERLY, DYLAN	REDACT		REDACT	REDACT	REDACT	
WALKER-WHITE	5728 SHAKESPEARE ROAD	P.O. BOX 1518	COLUMBIA	SC	29202	
WALLACE, ANISHA	REDACT		REDACT	REDACT	REDACT	
WALLACE, CAROLYN	REDACT		REDACT	REDACT	REDACT	
WALLACE, PAIGE	REDACT		REDACT	REDACT	REDACT	
WALLER, PENNY	REDACT	REDACT	REDACT	REDACT	REDACT	
		2529 Virginia Beach Boulevard,				
Walnut Hill Plaza Associates, LLC	WALNUT HILLS PLAZA/Store No216	Suite 200	Virginia Beach	VA	23452	
WALNUT HILL PLAZA ASSOCIATES,						
LLC	2529 Virginia Beach Blvd.	Suite 200	Virginia Beach	VA	23452	
WALNUT HILL PLAZA ASSOCIATES,						
LLC	2529 VIRGINIA BEACH BLVD	SUITE 200	VIRGINIA BEACH	VA	23452	
WALTON EMC NATURAL GAS	842 U.S. Hwy 78		Monroe	GA	30655	
WALTON EMC NATURAL GAS	PO BOX 1347		MONROE	GA	30655-1347	
WALTON MANAGEMENT SERVICES	3321 DORIS AVENUE		OCEAN	NJ	07712	
WARD, ANTONEIA	REDACT		REDACT	REDACT	REDACT	

Page 167 of 176

Name	Address 1	Address 2	City	State	PostalCode	Country
WARD, ANTONEIA	REDACT		REDACT	REDACT	REDACT	
WARD, JANET	REDACT		REDACT	REDACT	REDACT	
WARD, JANICE	REDACT		REDACT	REDACT	REDACT	
WARD, KORY	REDACT		REDACT	REDACT	REDACT	
WARD, LAROSHA	REDACT	REDACT	REDACT	REDACT	REDACT	
WARD, TREASURE	REDACT		REDACT	REDACT	REDACT	
WARE, DEMETRIS	REDACT	REDACT	REDACT	REDACT	REDACT	
WARNER, DALPHINE	REDACT		REDACT	REDACT	REDACT	
WARNOCK, RANDALL	REDACT		REDACT	REDACT	REDACT	
WARREN CONNER LLC	C/O KEY BANK BOX 712845		CINCINNATI	ОН	45271-2845	
WARREN CONNER, LLC	C/O THOR EQUITIES. LLC	P.O. BOX 712845	CINCINNATI	ОН	45271-2554	
WARREN COUNTY TAX COLLECTOR	P.O.BOX 351		VICKSBURG	MS	39181	
WARREN, AISHA	REDACT	REDACT	REDACT	REDACT	REDACT	
WARREN, GLENN	REDACT		REDACT	REDACT	REDACT	
WARREN, KAREN	REDACT		REDACT	REDACT	REDACT	
WARREN, STEVEN	REDACT		REDACT	REDACT	REDACT	
WASHINGTON GAS	101 Constitution Avenue, NW		Washington	DC	20080	
WASHINGTON GAS	PO BOX 37747		PHILADELPHIA	PA	19101-5047	
Washington Parish Sales Tax Dept	PO Drawer 508		Franklinton	LA	70438	
WASHINGTON SUBURAN	SANITARY COMMISSION	14501 SWEITZER LANE	LAUREL	MD	20707-5901	
WASHINGTON, CHRYSTAL	REDACT		REDACT	REDACT	REDACT	
WASHINGTON, DAKNETRA	REDACT		REDACT	REDACT	REDACT	
WASHINGTON, DC (FINANCE &						
REVENUE)	DEPT OF FINANCE & REVENUE	PO BOX 7792	WASHINGTON	DC	20044	
WASHINGTON, DEBORA	REDACT	REDACT	REDACT	REDACT	REDACT	
WASHINGTON, DEJA	REDACT		REDACT	REDACT	REDACT	
WASHINGTON, DEREK	REDACT		REDACT	REDACT	REDACT	
WASHINGTON, PRECIOUS	REDACT		REDACT	REDACT	REDACT	
WASHINGTON, TIAJUANA	REDACT		REDACT	REDACT	REDACT	
WASTE MANAGEMENT	24516 NETWORK PLACE		CHICAGO	IL	60673-1245	
WATER WORKS PLUMBING						
SERVICES, LLC	476 S. CREST ROAD		CHATTANOOGA	TN	37404	
WATER, GAS & LIGHT COMMISSION	207 Pine Ave		Albany	GA	31701	
WATER, GAS & LIGHT COMMISSION	PO BOX 1788		ALBANY	GA	31702-1788	

Name	Address 1	Address 2	City	State	PostalCode	Country
WATKINS, DEIRDRE	REDACT	REDACT	REDACT	REDACT	REDACT	
WATKINS, WILBORN	REDACT		REDACT	REDACT	REDACT	
WATSON ELECTRICAL						
CONSTRUCTION	PO BOX 3105		WILSON	NC	27895	
WATSON, ALONZO	REDACT		REDACT	REDACT	REDACT	
WATSON, BARBARA	REDACT	REDACT	REDACT	REDACT	REDACT	
WATSON, LANIECE	REDACT	REDACT	REDACT	REDACT	REDACT	
WAYS, DAJUAN	REDACT		REDACT	REDACT	REDACT	
WBBMT 2004-C10 Retail 31260, LLC	34975 W. Twelve Mile Road, #100		Farmington Hills	МІ	48331	
WBCMT 2004-C10 RETAIL 31260, LLC	C/O FRIEDMAN MGMT COMPANY	34975 W. TWELVE MILE RD.	FARMINGTON HILLS	MI	48331	
WE ENERGIES	231 W. Michigan		Milwaukee	WI	53203	
WE ENERGIES	PO BOX 90001		MILWAUKEE	WI	53290-0001	
WEAKLEY, CYNTHIA	REDACT		REDACT	REDACT	REDACT	
WEATHERS ELECTRIC, INC.	P.O. BOX 9275	521 7TH AVE. SOUTH	COLUMBUS	MS	39705-9275	
WEBB, FELICIA	REDACT		REDACT	REDACT	REDACT	
WEBB, JORDAN	REDACT	REDACT	REDACT	REDACT	REDACT	
WEBROOT, INC	385 INTERLOCKEN CRESCENT	SUITE 800	BROOMFIELD	СО	80021	
Webster Parish Sales & Use Tax						
Commission	PO Box 357		Minden	LA	71058-0357	
Webster Parish Sales and Use Tax						
Commission	PO Box 357		Minden	LA	71058-0357	
WEBSTER, DELANDO	REDACT		REDACT	REDACT	REDACT	
WEBSTER, DONAVAN	REDACT		REDACT	REDACT	REDACT	
WEBSTER, DONAVAN	REDACT		REDACT	REDACT	REDACT	
WEBSTER, KIMBERLY	REDACT		REDACT	REDACT	REDACT	
Weingarten Realty Investors	C/O Kelley Drye & Warren LLP	Attn: Robert L. LeHane, Esq.	New York	NY	10178	
	Jan W. Odom, Associate General					
WEINGARTEN REALTY INVESTORS	Counsel	P.O. Box 924133	Houston	TX	77292-4133	
WEINGARTEN REALTY INVESTORS	ATTN: JAN W. ODOM, ASSO.					
STORE NO. 23	GENERAL COUNSEL	P.O. BOX 924133	HOUSTON	TX	77292-4133	
WELBOURNE ELECTRICAL SERVICES	5951 ARBOR STREET		HYATTSVILLE	MD	20781	
WELCH, JOSEPTH	REDACT		REDACT	REDACT	REDACT	
WELCH, JR., JOSEPTH	REDACT		REDACT	REDACT	REDACT	
WELCH, THOMONDA	REDACT		REDACT	REDACT	REDACT	
WELDON, RUSSELL	REDACT		REDACT	REDACT	REDACT	

Name	Address 1	Address 2	City	State	PostalCode	Country
WELLROX	200 MADISON AVENUE	SUITE 2225	NEW YORK	NY	10016	
WELLS FARGO	PO Box 63020		San Francisco	CA	94163	
WELLS FARGO BANK	PO Box 6995		Portland	OR	97228-6995	
Wells Fargo Bank, N.A., as Trustee	CAHOKIA VILLAGE	c/o C-III Asset Management LLC	Irving	TX	75039	
		c/o Foresite Realty Management,				
Wells Fargo Bank, N.A., as Trustee	CAHOKIA VILLAGE	LLC	Rosemont	IL	60018	
WELLS FARGO BANK. N.A. as Trustee						
for the registered holders of Credit						
Suisse First Boston Mortgage						
Securities Corp,, Commercial						
Mortgage Pass-Through Certificates,	c/o C-III Asset Management, LLC					
Series 2007-C5	f/k/a Centerline Servicing, Inc.	5221 N. O'Connor Blvd.	Irving	TX	75039	
		Factoring Credit Senior Relations				
Wells Fargo Capital Finance	John McConnell	Manager	New York	NY	10017	
Wells Fargo Trade Capital Service	P.O. Box 842674		Boston	MA	02284-2674	
WELLSPRING SOFTWARE	445 SOVEREIGN CT		MANCHESTER	МО	63011	
WENTE, DANNY	REDACT		REDACT	REDACT	REDACT	
WERNER, BRANDI	REDACT	REDACT	REDACT	REDACT	REDACT	
WESLEY B. JONES ELECTRICAL	PO BOX 822585		VICKSBURG	MS	39182	
West Baton Rouge Parish Dept of						
Revenue	PO Box 86		Port Allen	LA	70767	
WEST COAST JOBBERS	1508 S MAIN ST		LOS ANGELES	CA	90015	
West Feliciana Parish School Board						
Sales Tax Collector	PO Box 1910		St. Francisville	LA	70775	
West Ridge, LLC	WEST RIDGE/Store No180	c/o Halpern Enterpirses, Inc.	Atlanta	GA	30340	
WEST RIDGE, LLC (GLENN S.						
CARACAPPA)	c/o Halpern Enterprises	5269 Buford Highway	Atlanta	GA	30340	
WEST, WANDA	REDACT		REDACT	REDACT	REDACT	
WESTBROOK, DAZJAH	REDACT		REDACT	REDACT	REDACT	
		c/o Aronov Realty Management,				
Western Hills, LLC	WESTERN HILLS MALL/Store No159	Inc.	Montgomery	AL	36116-1781	
		c/o Aronov Realty Management,				
Western Hills, LLC	WESTERN HILLS MALL/Store No159	Inc.	Montgomery	AL	36123-5000	
WESTERN HILLS, LLC and BERLEF						
ASSOCIATES, LLP	Aronov Realty Management, Inc.	P O BOX 23500	Montgomery	AL	36123-5000	

Name	Address 1	Address 2	City	State	PostalCode	Country
WESTERN TRADING COMPANY	1509 RIO VISTA AVE.		LOS ANGELES	CA	90023	
WESTON & MCELVAIN, LLP	27520 HAWTHORNE BLVD	SUITE 180	ROLLING HILLS ESTATES	CA	90274	
WESTPORT ASSOCIATES LTD.						
PARTNERSHIP	618 W. Fulton Street		Chicago	IL	60661	
WESTWOOD FOOTWEAR	18995 E RAILROAD ST		CITY OF INDUSTRY	CA	91748	
WESTWOOD VILLAGE, LP	PO BOX 3239		WILSON	NC	27895	
WET PAINT KNITWEAR	850 METROPOLITAN AVE		BROOKLYN	NY	11211	
WHARTON ACQUISITIONS	C/O Wharton Group	8 Industrial Way East	Eatontown	NJ	7724	
WHARTON REALTY GRP, BLDG-ICS	·	·				
OLNEY LLC	8 INDUSTRIAL WAY EAST	2ND FLOOR	EATONTOWN	NJ	07724	
WHATELY, RITA	REDACT		REDACT	REDACT	REDACT	
WHEELER, AIREONNA	REDACT	REDACT	REDACT	REDACT	REDACT	
WHEELER, RASHONA	REDACT		REDACT	REDACT	REDACT	
WHEELER, RONALD	REDACT		REDACT	REDACT	REDACT	
WHEELER, RONALD	REDACT		REDACT	REDACT	REDACT	
WHEELER, VICTORIA	REDACT		REDACT	REDACT	REDACT	
WHEELINGS, KAYETTA	REDACT		REDACT	REDACT	REDACT	
WHISPERS FASHIONS	1407 BROADWAY, #1809		NEW YORK	NY	10018	
WHITE LINE	5620 1ST AVENUE	4TH FLOOR	BROOKLYN	NY	11220	
WHITE, ANTWUAN	REDACT		REDACT	REDACT	REDACT	
WHITE, DANISHA	REDACT		REDACT	REDACT	REDACT	
WHITE, DEJA	REDACT		REDACT	REDACT	REDACT	
WHITE, DENA	REDACT		REDACT	REDACT	REDACT	
WHITE, KAYLA	REDACT		REDACT	REDACT	REDACT	
WHITE, LAPORSHIA	REDACT		REDACT	REDACT	REDACT	
WHITE, RENEE	REDACT		REDACT	REDACT	REDACT	
WHITE, ROY	REDACT		REDACT	REDACT	REDACT	
WHITE, SR., BRUCE A.	REDACT	REDACT	REDACT	REDACT	REDACT	
WHITEHALL TOWNSHIP	3221 MACARTHUR ROAD		WHITEHALL	PA	18052	
WHITEHALL TOWNSHIP AUTHORITY	Joseph M. McMahon III - Manager	1901 Schadt Avenue	Whitehall	PA	18052-3728	
				<u> </u>		
WHITEHALL TOWNSHIP AUTHORITY	1901 SCHADT AVENUE		WHITEHALL	PA	18052-3728	
WHITFIELD, LINDA	REDACT		REDACT	REDACT	REDACT	
WHITLEY, DEONKA	REDACT		REDACT	REDACT	REDACT	
WHITLOCK, SHANIKA	REDACT		REDACT	REDACT	REDACT	
WHITNER, ALYSSA	REDACT		REDACT	REDACT	REDACT	
WHITTEN, HAROLD	REDACT		REDACT	REDACT	REDACT	

Name	Address 1	Address 2	City	State	PostalCode	Country
WHITTIER, AYANNA	REDACT		REDACT	REDACT	REDACT	
WHITTLE, JAMES	REDACT		REDACT	REDACT	REDACT	
	C/O WHEELER REAL ESTATE	NORTH, 2529 VIRGINIA BEACH				
WHLR-TAMPA FESTIVAL, LLC	LLC,RIVERSEDGE	BLVD.	VIRGINIA BEACH	VA	23452	
WIGGINS, LIONELL	REDACT		REDACT	REDACT	REDACT	
WILBANKS, MALLORY	REDACT		REDACT	REDACT	REDACT	
WILBERTON, SHANICE	REDACT		REDACT	REDACT	REDACT	
WILBERTON, SHANICE	REDACT		REDACT	REDACT	REDACT	
WILBORN, TERRICA	REDACT		REDACT	REDACT	REDACT	
WILBUR'S PLUMBING, HEATING, AIR						
CONDITIONING, & GENERATORS	P.O. BOX 737		TRUSSVILLE	AL	35173	
WILCO CAPITAL, INC (INNOVATIVE						
STAFF)	PO BOX 130493		BIRMINGHAM	AL	35213	
WILD ROSE APPAREL	740 EAST PICO BLVD. #113		LOS ANGELES	CA	90021	
WILDE, IDELL	REDACT		REDACT	REDACT	REDACT	
WILEY, DESMOND	REDACT		REDACT	REDACT	REDACT	
WILHITE, JASMINE	REDACT		REDACT	REDACT	REDACT	
WILIAMS, SHANTE	REDACT		REDACT	REDACT	REDACT	
WILKINSON CROSSING, LLC	C/O HUNTER & ASSOCIATES	127 W HARGETT ST	RALEIGH	NC	27601	
WILKINSON, KIERA	REDACT	REDACT	REDACT	REDACT	REDACT	
WILKINSON, KIERA	REDACT		REDACT	REDACT	REDACT	
WILLIAM, CARVIN	REDACT		REDACT	REDACT	REDACT	
WILLIAMS, ALEXIA	REDACT		REDACT	REDACT	REDACT	
WILLIAMS, ALICIA	REDACT		REDACT	REDACT	REDACT	
WILLIAMS, ARNEETRA	REDACT		REDACT	REDACT	REDACT	
WILLIAMS, BERNICE	REDACT	REDACT	REDACT	REDACT	REDACT	
WILLIAMS, BEVERLY	REDACT		REDACT	REDACT	REDACT	
WILLIAMS, BLAIR	REDACT		REDACT	REDACT	REDACT	
WILLIAMS, DOROTHY	REDACT		REDACT	REDACT	REDACT	
WILLIAMS, FATIAMA	REDACT		REDACT	REDACT	REDACT	
WILLIAMS, GERMAINE	REDACT		REDACT	REDACT	REDACT	
WILLIAMS, JAMAL	REDACT		REDACT	REDACT	REDACT	
WILLIAMS, JAMIE	REDACT		REDACT	REDACT	REDACT	
WILLIAMS, JUANICIA	REDACT		REDACT	REDACT	REDACT	
WILLIAMS, JULIET	REDACT		REDACT	REDACT	REDACT	
WILLIAMS, KALIL	REDACT		REDACT	REDACT	REDACT	
WILLIAMS, KEITH	REDACT		REDACT	REDACT	REDACT	

Name	Address 1	Address 2	City	State	PostalCode	Country
WILLIAMS, KIMBERLY	REDACT		REDACT	REDACT	REDACT	
WILLIAMS, KIMBERLY	REDACT		REDACT	REDACT	REDACT	
WILLIAMS, KIMBERLY	REDACT	REDACT	REDACT	REDACT	REDACT	
WILLIAMS, KYLA	REDACT		REDACT	REDACT	REDACT	
WILLIAMS, LOLA	REDACT	REDACT	REDACT	REDACT	REDACT	
WILLIAMS, LORI	REDACT		REDACT	REDACT	REDACT	
WILLIAMS, MARSHA	REDACT		REDACT	REDACT	REDACT	
WILLIAMS, MONISHA	REDACT		REDACT	REDACT	REDACT	
WILLIAMS, RHONDA	REDACT		REDACT	REDACT	REDACT	
WILLIAMS, ROBERT	REDACT		REDACT	REDACT	REDACT	
WILLIAMS, ROBERT	REDACT	REDACT	REDACT	REDACT	REDACT	
WILLIAMS, ROBERT L.	REDACT		REDACT	REDACT	REDACT	
WILLIAMS, SADE	REDACT		REDACT	REDACT	REDACT	
WILLIAMS, SHANDA	REDACT		REDACT	REDACT	REDACT	
WILLIAMS, SHAWN	REDACT		REDACT	REDACT	REDACT	
WILLIAMS, SHEREE	REDACT		REDACT	REDACT	REDACT	
WILLIAMS, TAMEKA	REDACT		REDACT	REDACT	REDACT	
WILLIAMS, TELECIA	REDACT		REDACT	REDACT	REDACT	
WILLIAMS, TEQUILA	REDACT	REDACT	REDACT	REDACT	REDACT	
WILLIAMS, TEQUISA	REDACT		REDACT	REDACT	REDACT	
WILLIAMS, TEREA	REDACT	REDACT	REDACT	REDACT	REDACT	
WILLIAMS, TERRY	REDACT		REDACT	REDACT	REDACT	
WILLIAMS, ULANDA	REDACT		REDACT	REDACT	REDACT	
WILLIAMS, VALERIE	REDACT		REDACT	REDACT	REDACT	
WILLIAMS, YOLANDA	REDACT		REDACT	REDACT	REDACT	
WILLIE POULLARD	3105 63RD PLACE		CHEVERLY	MD	20785	
WILLIE SNIPES	2646 TENNYSON DRIVE		WYOMING	MI	49509	
WILLIS, CARLAN	REDACT		REDACT	REDACT	REDACT	
WILLIS, DANESHA	REDACT		REDACT	REDACT	REDACT	
WILLIS, DOMINIQUE	REDACT		REDACT	REDACT	REDACT	
WILLIS, JANAE	REDACT		REDACT	REDACT	REDACT	
WILSON CO. TAX COLLECTOR	P.O. BOX 580328		CHARLOTTE	NC	28258-0328	
WILSON COUNTY TAX						
ADMINISTRATOR	PO BOX 1162		WILSON	NC	27894-1162	
WILSON, ANDRE	REDACT		REDACT	REDACT	REDACT	
WILSON, CHANON	REDACT		REDACT	REDACT	REDACT	
WILSON, CHARISMA	REDACT		REDACT	REDACT	REDACT	

Name	Address 1	Address 2	City	State	PostalCode	Country
WILSON, CHRISTOPHER	REDACT		REDACT	REDACT	REDACT	
WILSON, DANA	REDACT		REDACT	REDACT	REDACT	
WILSON, DANASHA	REDACT		REDACT	REDACT	REDACT	
WILSON, DOMONIQUE	REDACT		REDACT	REDACT	REDACT	
WILSON, ISAIH	REDACT	REDACT	REDACT	REDACT	REDACT	
WILSON, IYESHA	REDACT		REDACT	REDACT	REDACT	
WILSON, JOYCLYNN	REDACT		REDACT	REDACT	REDACT	
WILSON, LINDA	REDACT		REDACT	REDACT	REDACT	
WILSON, MARIA	REDACT		REDACT	REDACT	REDACT	
WILSON, MELANIE	REDACT		REDACT	REDACT	REDACT	
WILSON, PATRICE	REDACT	REDACT	REDACT	REDACT	REDACT	
WILSON, SONDRA	REDACT		REDACT	REDACT	REDACT	
WILSON, TIARA	REDACT		REDACT	REDACT	REDACT	
WILSON, VICTORIA	REDACT		REDACT	REDACT	REDACT	
WINBROOK MANAGEMENT, LLC	370 Seventh Avenue	Suite 1600	New York	NY	10001-3903	
WINDHAM, JASMINE	REDACT	REDACT	REDACT	REDACT	REDACT	
WINDOM, SAMANTHA	REDACT		REDACT	REDACT	REDACT	
WINN, CAMERON	REDACT		REDACT	REDACT	REDACT	
WISCONSIN DEPT. OF REV (930208)	P.O. BOX 930208		MILWAUKEE	WI	53293	
Wisconsin Department of Revenue	Customer Service Bureau	P.O. Box 8949	Madison	wı	53708-8949	
WISE, EBONY	REDACT		REDACT	REDACT	REDACT	
WISEMAN, TIFFINY	REDACT		REDACT	REDACT	REDACT	
WITHWORTH, CANDELARIA	REDACT		REDACT	REDACT	REDACT	
W-M DEVELOPMENT CO, LLC	10689 N. PENNSYLVANIA STREET	SUITE 100	INDIANAPOLIS	IN	46280	
W-M DEVELOPMENT COMPANY LLC	10689 N Pennsylvania St.	Suite 100	Indianapolis	IN	46280	
WOMACK, TARON	REDACT		REDACT	REDACT	REDACT	
WOODALL'S TOTAL COMFORT						
SYSTEMS	3608 HIGHWAY 90		MARIANNA	FL	32446	
WOODFOREST BANK	PO Box 7889		The Woodlands	TX	77387	-
WOODS, ANGELA	REDACT	1	REDACT	REDACT	REDACT	ļ
WOODS, CANDICE	REDACT	REDACT	REDACT	REDACT	REDACT	1
WOODS, DIAAMOND	REDACT		REDACT	REDACT	REDACT	
WOOTEN, DEBRANN	REDACT		REDACT	REDACT	REDACT	
WOOTEN, NORRIS	REDACT	REDACT	REDACT	REDACT	REDACT	

Page 174 of 176

Name	Address 1	Address 2	City	State	PostalCode	Country
		Joseph D. Early, Jr., County District				,
Worcester County	Attn: Consumer Protection Agency	Attorney	Worcester	MA	01608	
WORDS, SHAWAN	REDACT	·	REDACT	REDACT	REDACT	
WORLD FINANCE CORP OF GA	COLLECTIONS DEPT	PO BOX 71847	ALBANY	GA	31708	
WORSHMAN, AARION	REDACT		REDACT	REDACT	REDACT	
WR9000	1407 BROADWAY, STE 448		NEW YORK	NY	10018	
WRI-AEW Lone Star Retail Portfolio,	SOUTHGATE SHOPPING					
LLC	CENTER/Store No576	P.O. Box 679003	Dallas	TX	75267	
WRI-AEW LONE STAR RETAIL						
PORTFOLIO, LLC	P.O. Box 203783		Houston	TX	77216-3783	
WRI-AEW LONE STAR RETAIL						
PORTFOLIO, LLC	LOCATION #0031-537 LSIMPFS02	PO BOX 301531	DALLAS	TX	75303-1531	
WRI-AEW LONE STAR RETAIL						
PORTFOLIO, LLC	LOCATION #0088-537 LSIMPFS01	PO BOX 301531	DALLAS	TX	75303-1531	
WRI-AEW LONE STAR RETAIL						
PORTFOLIO, LLC	LOCATION #0056-537	PO BOX 301531	DALLAS	TX	75303-1531	
WRIGHT, ANTOINETTE	REDACT		REDACT	REDACT	REDACT	
WRIGHT, CORINTHIA	REDACT		REDACT	REDACT	REDACT	
WRIGHT, DASHAY	REDACT		REDACT	REDACT	REDACT	
WRIGHT, DORATHEIA	REDACT		REDACT	REDACT	REDACT	
WRIGHT, WAYNE	REDACT		REDACT	REDACT	REDACT	
WRIGHT/HURD PROPERTIES, LLC	3621 Arcadia Drive		Tuscaloosa	AL	35404	
WRIGHT/HURD PROPERTIES, LLC	1418 GREENSBORO AVE	SUITE 5	TUSCALOOSA	AL	35401	
WRIGHT'S ELECTRICAL						
CONTRACTING CO	PO BOX 21953		CHARLESTON	SC	29413	
WRI-TC INTERNATIONAL DRIVE						
VALUE CENTER, LLC	P.O. Box 924133		Houston	TX	77292-4133	
WRI-TC, INTERNATIONAL DRIVE						
VALUE CENTER, LLC	ATTN: GENERAL COUNSEL	P.O. BOX 924133	HOUSTON	TX	77292-4133	
WYDMAN, ROBERT	REDACT		REDACT	REDACT	REDACT	
WYOMING MALL, LTD	700 MALL DRIVE		PORTAGE	MI	49024	
WYOMING MALL, LTD.	700 Mall Drive		Portage	MI	49024	
XTREME CABLES	32 BRUNSWICK AVE		EDISON	NY	08817	
XTREME TIME	20 WEST 33RD ST	6TH FLOOR	NEW YORK	NY	10001	
Y & W Enterprises Company	NORTHSIDE PLAZA/Store No72	712 Highway 72 East	Huntsville	AL	35801	
Y & W ENTERPRISES COMPANY	712 Highway 72 East		Huntsville	AL	35801	
Y & W ENTERPRISES, CO	712 HWY 72 EAST		HUNTSVILLE	AL	35811	
Y.M. ACCESSORIES INC	1261 BROADWAY	SUITE 408	NEW YORK	NY	10001	

Name	Address 1	Address 2	City	State	PostalCode	Country
YASINSKY, CHRISTOPHER	REDACT		REDACT	REDACT	REDACT	
YATES, EDDIE	REDACT		REDACT	REDACT	REDACT	
YDB SHOPRITE, LLC	C/O AVISON YOUNG-FLORIDA	515 EAST LAS OLAS BLVD	FT LAUDERDALE	FL	33301	
YDENIA MCNAIR	1223 SILAS CREEK PARKWAY		WINSTON-SALEM	NC	27127	
YMI	1155 S BOYLE AVE		LOS ANGELES	CA	90023	
YOKI FASHION INTERNATIONAL	YOKI SHOES	1407 BROADWAY	NEW YORK	NY	10018	
YOUNG, BILLY	REDACT		REDACT	REDACT	REDACT	
YOUNG, DEAN	REDACT		REDACT	REDACT	REDACT	
YOUNG, DIANA	REDACT		REDACT	REDACT	REDACT	
YOUNG, KENDAL	REDACT	REDACT	REDACT	REDACT	REDACT	
YOUNG, SHADENA	REDACT		REDACT	REDACT	REDACT	
YOUNG, SHADERIA	REDACT		REDACT	REDACT	REDACT	
YVONNE TOMPKINS	1041 LYNDHURST WAY S.E.		CONYERS	GA	30013	
ZACCAGNO, SELINA	REDACT		REDACT	REDACT	REDACT	
ZAJACZKOWSKI, LEIGH	REDACT		REDACT	REDACT	REDACT	
	CONSUMER SQUARE WEST/Store					
Zamias Services, Inc.	No563	300 Market Street	Johnstown	PA	15901	
ZAVALA, SARA	REDACT		REDACT	REDACT	REDACT	
ZEE MEDICAL, INC	PO BOX 781572		INDIANAPOLIS	IN	46278-8572	
ZENANA FASHIONS	1100 S SAN PEDRO, ROOM M10		LOS ANGELES	CA	90015	
ZENON, ANATASHIA	REDACT		REDACT	REDACT	REDACT	
ZEP SALES & SERVICE	PO BOX 404628		ATLANTA	GA	30384-4628	
ZEPEDA, MARLENY	REDACT		REDACT	REDACT	REDACT	
ZIEMINSKI, SARAH	REDACT	REDACT	REDACT	REDACT	REDACT	